Mitchell D. Roth Mayor

Deanna S. Sako Managing Director



Ramzi I. Mansour Director

Brenda Iokepa-Moses Deputy Director

County of Hawai'i

DEPARTMENT OF ENVIRONMENTAL MANAGEMENT

345 Kekūanāoʻa Street, Suite 41 · Hilo, Hawaiʻi 96720 · cohdem@hawaiicounty.gov Ph: (808) 961-8083 · Fax: (808) 961-8086

August 27, 2024

TO:	Mary Alice Evans
	Director, Office of Planning and Sustainable Development
	Environmental Review Program
FROM:	Ramzi Mansour, Director
	County of Hawai'i Department of Environmental Management
STIDIE CT.	Na jalehu Large Canacity Cessnool Closure Project
SUBJECT.	Na alchu Laige Capacity Cossport Closure i Tojeet
	Draft Environmental Information Document and Environmental Assessment
	Anticipated Finding of No Significant Impact (DEA-AFONSI)

The County of Hawai'i Department of Environmental Management herewith transmits the subject Draft Environmental Information Document (EID) / Environmental Assessment (DEA) for which there is an Anticipated Finding of No Significant Impact (DEA-AFONSI). The DEA-AFONSI has been prepared pursuant to Chapter 343, Hawai'i Revised Statutes, and Chapter 11-200.1, Hawai'i Administrative Rules. Please publish notice of this DEA-AFONSI in the upcoming issue of *The Environmental Notice*.

Included in the DEA-AFONSI are a Draft Archaeological Inventory Survey, a Natural Resource Assessment, and copies of comments received during pre-assessment consultation along with the corresponding responses regarding the subject project.

Please contact our consultant Mr. Keola Cheng at (808) 946-2277 if you have any questions.

Sincerely

Ramzi Mansour, Director County of Hawai'i Department of Environmental Management

Cc: Mark Grant, Wastewater Division Acting Deputy Chief Chris Sparber, Wastewater Division Acting Chief

From:	webmaster@hawaii.gov
То:	DBEDT OPSD Environmental Review Program
Subject:	New online submission for The Environmental Notice
Date:	Friday, August 30, 2024 11:54:51 AM

Action Name

Naalehu Large Capacity Cesspool Closure

Type of Document/Determination

Draft environmental assessment and anticipated finding of no significant impact (DEA-AFNSI)

HRS §343-5(a) Trigger(s)

• (1) Propose the use of state or county lands or the use of state or county funds

Judicial district

Ka'ū, Hawai'i

Tax Map Key(s) (TMK(s))

Portions of (3) 9-5-007, (3) 9-5-008, (3) 9-5-009, (3) 9-5-007, (3) 9-5-021, (3) 9-5-022, (3) 9-5-024, (3) 9-5-025, and (3) 9-5-026.

Action type

Agency

Other required permits and approvals

See EID / EA Section 5.3

Proposing/determining agency

County of Hawaii Department of Environmental Management

Agency contact name

Mark Grant

Agency contact email (for info about the action)

markj.grant@hawaiicounty.gov

Email address for receiving comments

publiccomment@wilsonokamoto.com

Agency contact phone

(808) 961-8083

Agency address

345 Kekūanāoʻa Street Suite 41 Hilo, HI 96720 United States <u>Map It</u>

Is there a consultant for this action?

Yes

Consultant

Wilson Okamoto Corporation

Consultant contact name

Harlee Meyers

Consultant contact email

hmeyers@wilsonokamoto.com

Consultant contact phone

(808) 946-2277

Consultant address

1907 South Beretania Street Suite 400 Honolulu, HI 96826 United States <u>Map It</u>

Action summary

This project involves the construction of new sewer collection systems or individual wastewater systems for treatment to serve portions of the existing community of Nāʿālehu in the Kaʿu District of the Island of Hawaiʿi. These improvements would allow for the closure of existing large capacity "gang" cesspools (LCC) currently serving these communities. The project will improve the longevity of the sewer systems in these communities as well as assure compliance with the EPA mandated conversion of LCCs.

Reasons supporting determination

See EID / EA Chapter 7

Attached documents (signed agency letter & EA/EIS)

- Draft-Naalehu-EID-EA.pdf
- 2024.08.27_Naalehu-Draft-EID-EA-Trans.-Letter-to-ERP.PDF

Action location map

• Naalehu-Project-Location.zip

Authorized individual

Harlee Meyers

Authorization

• The above named authorized individual hereby certifies that he/she has the authority to make this submission.

Draft Environmental Information Document & Environmental Assessment

NĀ'ĀLEHU LARGE CAPACITY CESSPOOL CLOSURE

Nā'ālehu, Hawai'i Island, Hawai'i September 2024

Prepared For: County of Hawai'i Department of Environmental Management

> Prepared By: Wilson Okamoto Corporation



DRAFT ENVIRONMENTAL INFORMATION DOCUMENT & ENVIRONMENTAL ASSESSMENT

NĀ'ĀLEHU LARGE CAPACITY CESSPOOL CLOSURE

Nāʿālehu, Hawaiʿi Island, Hawaiʿi Portions of TMK Plats: (3) 9-5-007, (3) 9-5-008, (3) 9-5-009, (3) 9-5-007, (3) 9-5-021, (3) 9-5-022, (3) 9-5-024, (3) 9-5-025, and (3) 9-5-026.

Prepared For:

The County of Hawai'i Department of Environmental Management (DEM) East Hawai'i Office: 345 Kekūanāo'a St., Suite 41 Hilo, Hawaii 96720

Prepared By:

Wilson Okamoto Corporation 1907 South Beretania Street, Suite 400 Honolulu, HI 96826

WOC Job No. 10345-05

September 2024

PREFACE

This Draft Environmental Information Document (EID) / Environmental Assessment (EA) Anticipated Finding of No Significant Impact (FONSI) has been prepared pursuant to Chapter 343, Hawai'i Revised Status (HRS), and Title 11, Chapter 200.1, Hawai'i Administrative Rules (HAR), Department of Health, State of Hawai'i.

The Hawai'i County (County) Department of Environmental Management (DEM) - Wastewater Division (WWD) is proposing to construct new sewer collection systems or individual wastewater systems for treatment to serve portions of the existing community of Nā'ālehu in the Ka'u District of the Island of Hawai'i. These improvements would allow for the closure of existing large capacity "gang" cesspools (LCC) currently serving these communities. The Proposed Action will improve the longevity of the sewer systems in these communities as well as assure compliance with the EPA mandated conversion of LCCs.

The subject EID / EA is intended to be consistent with requirements of the National Environmental Policy Act (NEPA) 42 U.S.C. § et seq. and will also provide documentation of necessary compliance with Section 7 of the Endangered Species Act and Section 106 of the National Historic Preservation Act. Further, the document shall be prepared pursuant to Hawai'i Revised Statutes (HRS) §343-5 (1) which outlines requirements for environmental disclosure for actions that propose the use of state or county lands or the use of state or county funds. Pursuant to HRS §343-5(b), the Proposed Action is being considered an "agency action" for which the County DEM shall be responsible for determining if the Final EA can be filed as a FONSI.

This Draft EID / EA includes an assessment of the potential environmental, social, cultural, and economic impacts associated with the Proposed Project. This Draft EID / EA has also been prepared in consideration of the comments received in response to the Early Consultation Package mailed out on March 29, 2024, to the respective stakeholders listed in Chapter 8.1 of this Draft EID / EA.



(This page intentionally left blank)



SUMMARY

Project Name:	Nā`ālehu Large Capacity Cesspool Closure
Project Applicant:	Hawai'i County Department of Environmental Management 345 Kekūanāo'a Street, Suite 41 Hilo, HI 96720
Location:	Nāʿālehu, Hawaiʿi Island, Hawaiʿi
Tax Map Keys (TMKs):	There are no TMK numbers for the County roadways right-of-way for which improvements are planned. TMK plat numbers for the various parcels being serviced by the new sewer collection and treatment systems are identified below:
	(3) 9-5-007, (3) 9-5-008, (3) 9-5-009, (3) 9-5-007, (3) 9-5-021, (3) 9-5-022, (3) 9-5-024, (3) 9-5-025, and (3) 9-5-026.
Recorded Fee Owner:	The sewer collection system improvements planned within Nāʿālehu would predominantly occur within the rights-of-way of existing County roadways. However, there are some segments of the collection system that would run through portions of privately- owned property. The IWS improvements planned to treat the wastewater would occur within privately-owned property or portions of such property. Easements for the sewer collection system would be obtained from the landowners of the private properties affected. Easements or acquisition of property needed for the wastewater treatment sites would also be obtained.
Existing Use:	Residential, Commercial, and undeveloped lands
State Land Use Classification:	Urban and Agricultural
County Zoning Designation:	Single-Family Residential District (RS-7.5, RS-10, and RS-15) Village Commercial District (CV-10) Limited Industrial District (ML-20) Multiple-Family Residential District (RM-3) Agricultural District (A-20) Open District
Flood Hazard Zone:	Zone X



Proposed Action:	This project involves the construction of new sewer collection systems or individual wastewater systems for treatment to serve portions of the existing community of Nā'ālehu in the Ka'u District of the Island of Hawai'i. These improvements would allow for the closure of existing large capacity "gang" cesspools (LCC) currently serving these communities. The project will improve the longevity of the sewer systems in these communities as well as assure compliance with the EPA mandated conversion of LCCs.
Impacts:	Short-term construction-related impacts to surrounding areas include fugitive dust, noise, and construction-related traffic. Impacts on air quality and noise levels will be mitigated through implementation of best management practices (BMPs) and adherence to federal, state, and county rules and regulations. Traffic impacts will be mitigated through traffic control measures developed during the design phase that will avoid or minimize disruptions to surrounding operations. Short-term construction-related impacts are anticipated to be negligible with mitigation. In the long-term, no significant impacts are anticipated as a result of the construction or operation of the proposed action.
Anticipated Determination:	Finding of No Significant Impact (FONSI)
Parties Consulted During Early	
Consultation:	Federal Agencies US Army Corps of Engineers, Honolulu District US Department of Agriculture Natural Resources Conservation Service US Environmental Protection Agency US Fish and Wildlife Service National Oceanic and Atmospheric Administration US Department of Transportation Federal Aviation Administration
	State AgenciesDepartment of AgricultureDepartment of Accounting and General ServicesDepartment of Business, Economic Development & Tourism(DBED&T)DBED&T Land Use CommissionDBED&T Office of Planning and Sustainable DevelopmentDBED&T State Energy OfficeDepartment of Hawaiian Home Lands (DHHL)DHHL – East Hawai'i District OfficeDepartment of Health (DOH)
	DOH – Clean Water Branch



DOH - Environmental Management Office DOH - Hazard Evaluation and Emergency Response DOH - Safe Drinking Water Branch DOH - Wastewater Branch Department of Land and Natural Resources (DLNR) DLNR State Historic Preservation Division DLNR Land Division Department of Transportation Office of Hawaiian Affairs University of Hawai'i

County of Hawai'i

Fire Department Police Department Planning Department Research and Development Department of Public Works Department of Parks and Recreation Department of Water Supply Environmental Management Commission Office of the Corporation Council Office of Housing and Community Development

Elected Officials

State Senator Dru Mamo Kanuha, Senate District 3 Representative Jeanne Kapela, House District 5 Councilmember Michelle Galimba, Council District 6

Public Utilities

Hawaiian Electric Company Hawaiian Telcom Spectrum Hawai'i Hawai'i Gas

Other Parties

Hawai'i State Library_ Nā'ālehu Public Library_ Hawaiian Civic Club of Ka'ū Ka'ū CDP Action Committee Ka'ū Calendar 'O Ka'ū Kākou



(This page intentionally left blank)



TABLE OF CONTENTS

PREF. SUMN	ACE 1ARY	
СНАР	TER 1:	INTRODUCTION
1.1	Backg	round
1.2	Projec	t Location
1.3	Purpos	se and Need1-4
СНАР	TER 2:	PROJECT DESCRIPTION
2.1	Propos	sed Action 2-1
	2.1.1	Alternative 1: Package Plant New Collection System
	2.1.2	Alternative 2: Package Plant Existing Collection System
	2.1.3	Alternative 3: Individual Wastewater System- Maintenance Contract Model . 2-15
	2.1.4	Alternative 4: Individual Wastewater System- Operating Permit Model 2-21
СНАР	TER 3:	CUMULATIVE EFFECTS
3.1	Scope	of Analysis
	3.1.1	Geographic Scope of Analysis 3-1
СНАР	TER 4:	EXISTING ENVIRONMENT, IMPACTS, AND MITIGATION
MEAS	URES	
4.1	Climat	e
4.2	Physio	grapny
	4.2.1	Topography
	4.2.2	Geology
4.2	4.2.3	Solis
4.3	water	Resources
	4.3.1	Surrace waters
	4.3.2	Groundwater
4.4	Agricu	Itural Lands
4.5	Natura	4-15
	4.5.1	Sea Level Rise
	4.5.2	Flood and Tsunami Threat
	4.5.3	Hurricane and Wind Hazard 4-16
	4.5.4	Seismic Hazard 4-18
	4.5.5	Volcanic Hazards 4-19
	4.5.6	Wildfire Hazards 4-19
4.6	Flora a	and Fauna 4-20



47	Cultural Historical and Archaoological Decources	4 2E
л./ Л Q	Air Quality and Odors	л_20
7.0 ∕ 0	Noico	4-29 4-30
4.10	Fineral and Natural Resources	4-31
4.10 4.11	Land Lice and Land Lice Plans	4-31 4-32
л.11 Д 17	Poodways and Traffic	4-32
4 13	Hazardous Materials	4-35
4 14	Socioeconomics and Environmental Justice	4-36
4 15	Sustainability	4-39
4 16	Human Health and Safety	4-41
4.17	Unresolved Issues	4-42
CHED	TER 5. RELATIONSHIP TO PLANS POLICIES AND CONTROLS	5-1
5.1	State of Hawai'ī I and Use Plans and Policies	5-1
5.1	5.1.1 Hawai'î State Plan	5-1
	5.1.2 State Functional Plans	5-29
	5.1.3 State Land Use Law, Chapter 205, Hawai'ī Revised Statutes	5-31
5.2	County of Hawai'i Land Use Plans and Policies	5-32
•	5.2.1 County of Hawai'i General Plan	5-32
	5.2.2 General Plan Land Use Allocation Guide and Zoning	5-75
	5.2.3 Ka'ū Community Development Plan	5-75
	5.2.4 County of Hawai'ī Special Management Area	5-76
5.3	List of Permits and Approvals	5-76
СНАР	TER 6: LEGAL FRAMEWORK AND REGULATING AUTHORITIES	6-1
6.1	National Environmental Policy Act (NEPA) of 1969	6-1
6.2	Archaeological and Historic Preservation Act	6-2
6.3	Bald and Golden Eagle Protection Act	6-3
6.4	Clean Air Act	6-4
6.5	Coastal Barrier Resources Act	6-5
6.6	Coastal Zone Management Act	6-5
6.7	Endangered Species Act	6-9
6.8	Environmental Justice Executive Order 12898	6-10
6.9	Farmland Protection Policy Act	6-11
6.10	Fish and Wildlife Coordination Act	6-11
6.11	Floodplain Management	6-12
6.12	Magnuson- Stevens Fishery Conservation Management Act	6-12
6.13	Marine Mammal Protection Act	6-13
6.14	Migratory Bird Treaty Act	6-13
6.15	National Historic Preservation Act	6-14



6.16	Protection of Wetlands	6-15
6.17	Rivers and Harbors	6-15
6.18	Safe Drinking Water Act	6-16
6.19	Wild and Scenic Rivers Act	6-16
6.20	Clean Water Act	6-16

CHAP1	FER 8: CONSULTATION	8-1
8.1	Early Consultation	8-1
8.2	Community Outreach	8-2
8.3	Additional Consultation	8-3

CHAP1	FER 9:	SELECTION OF A PREFERRED ALTERNATIVE	9-1
9.1	Recom	mendation Factors	9-1
	9.1.1	Regulatory Compliance	9-1
	9.1.2	Community Preference	9-2
	9.1.3	Environmental Risks	9-2
9.2	Action	Items Under the Preferred Alternative	9-2

CHAPTER 10: REFERENCES

LIST OF FIGURES

Figure 1-1	Project Location Map	1-3
Figure 2-1	Wastewater Treatment Plant Service Area	2-2
Figure 2-2	New Collection System Layout	2-3
Figure 2-3	Replacement of Existing Collection System	2-4
Figure 2-4	Wastewater Treatment Plant Site Plan	2-7
Figure 2-5	Operations Building Floor Plan	2-8
Figure 2-6	In-Channel Cylindrical Screen	2-10
Figure 2-7	Aerated Grit Removal Schematic	2-10
Figure 2-8	Activated Carbon SCrubber	2-11
Figure 2-9	Subsurface Drip System	2-13
Figure 2-10	Individual Wastewater System	2-17
Figure 2-11	Septic Tank	2-19
Figure 2-11	Common Septic Tanks in Hawai'i	2-19
Figure 2-12	Typical IWS Site Plan	2-22
Figure 2-13	Typical IWS Site Plan with Seepage Pit	2-22
Figure 4-1	Topographic Map	4-3
Figure 4-2	Soil Classification Map	4-7



Figure 4-3	Surface Waters Map	4-9
Figure 4-4	Aquifer Map	4-12
Figure 4-5	Flood Insurance Rate Map	4-17
Figure 4-6	Wildfire Risk Map	4-21

LIST OF TABLES

Table 2-1	DOH Required Setbacks for Wastewater Systems	
Table 4-1	Demographic, Economic, and Social Characteristics of	
	Nā`ālehu and Hawai`i County	4-37
Table 5-1	The Hawai'i State Plan	5-1
Table 6-1	Hawai'i Coastal Zone Management Act	

APPENDICES

Appendix A	Nāʿālehu Large Capacity Cesspool Closure Project Revised Preliminary
	Engineering Report, Brown and Caldwell
Appendix B	Natural Resource Assessment, AECOS Inc.
Appendix C	Section 7 Consultation Initiation Letter
Appendix D	Section 7 Consultation U.S. Fish and Wildlife Service Response
Appendix E	Draft Archeological Inventory Survey, Cultural Surveys Hawaii
Appendix F	Section 106 Consultation Initiation Letter
Appendix G	Early Consultation Comments and Responses



(This page intentionally left blank)



CHAPTER 1: INTRODUCTION

1.1 Background

The Hawai'i County (County) Department of Environmental Management (DEM) - Wastewater Division (WWD) is proposing to construct new sewer collection systems or individual wastewater systems for treatment (hereafter referred to as Proposed Action) to serve portions of the existing Nā'ālehu Community (Project Area) on the island of Hawai'i. The Proposed Action would allow for the closure of existing large capacity "gang" cesspools (LCC) currently serving the Project Area and is anticipated to improve the longevity of the sewer systems in Nā'ālehu as well as assure compliance with the mandated conversion of LCCs.

This Draft Environmental Information Document (EID) / Environmental Assessment (EA) is intended to address State and Federal Environmental Review Requirements of the Proposed Action. The Amended Administrative Order of Consent (Amended AOC) (Docket No. SDWA-UIC-AOC-2017-0002, effective February 14, 2024) Large Capacity Cesspool (LCC) Closure for Nā'ālehu outlines that an EID must be prepared by the County DEM for US Environmental Protection Agency (EPA) approval by July 30, 2024, to meet Federal Environmental Review Requirements.

The Amended AOC §31.a. requires evaluation of four feasible options:

- 1. A package plant and new collection system (Alternative 1)
- 2. A package plant connected to the existing collection system (Alternative 2)
- 3. A maintenance contract model Individual Wastewater System (IWS) program (Alternative 3)
- 4. A County issued voucher program with an operating permit model IWS program (Alternative 4)

In addition, to meet the requirements of the EPA, this EID will include:

5. A No Action Alternative.

In June 2017, the EPA and County entered into an AOC which required the WWTP to provide secondary treatment of the sewage. Secondary treatment was no longer required under the previous Revised AOC (effective date August 22, 2022) and is consistent with the Amended AOC. As such, the IWS alternatives provide a method to close the three LCCs without providing a secondary treatment process. The package plant discussed below will provide the secondary treatment WWTP.

This Draft EID / DEA was prepared in accordance with the National Environmental Policy Act (NEPA), EPA and Council on Environmental Quality NEPA regulations as well as the Hawai'i Environmental Policy Act (HEPA-Hawai'i Revised Statutes, Chapter 343) and implementing rules Hawai'i Administrative Rules (HAR), Title 11, State of Hawai'i Department of Health Chapter 200, Environmental Impact Statement Rules. The Proposed Action will also provide documentation of necessary compliance with Section 7 of the Endangered Species Act and Section 106 of the National Historic Preservation Act.



Supporting studies included herein within the subject EID / EA, encompass Archaeological and Cultural documentation, Botanical / Faunal Surveys and regulatory coordination. The County will seek to coordinate with the State of Hawai'i Department of Health (DOH) to ensure timely review and concurrence by the State of Hawai'i State Historic Preservation Division (SHPD) and the US Fish and Wildlife Service (FWS).

Within 60 days of receiving written approval of the Final EID, the County shall submit an implementation plan for EPA approval. The Nā'ālehu Implementation Plan shall include a schedule and completion dates for each step required to implement the selected alternative, provide wastewater services for 194 properties, and to close the Nā'ālehu Community Cesspools no later than December 31, 2027.

1.2 Project Location

The community of Nā'ālehu is located about 64 miles southwest of Hilo in the Ka'ū District. Māmalahoa Highway (State Route 11) runs through the community from east to west. The Nā'ālehu community was established as a result of the sugar operations of the C. Brewer Company (C. Brewer). Even though Ka'ū was one of the originally settled areas in the Hawaiian Islands, it remains a vast remote area. Only a fraction of a percent of the Ka'ū District has been developed with residential properties, and the remainder is largely used for agricultural purposes or is undeveloped. The Ka'ū District covers about 922 square miles (approximately 590,000 acres), with over 80 miles of virtually undeveloped coastline. Nearly two-thirds of its total land area is in the Conservation District. The Ka'ū District consists of several communities, including the Nā'ālehu community, which has a population of approximately 890 persons according to the US Census Bureau American Community Survey, 2022. The distance to the communities of Hilo and Kailua-Kona means that the Ka'ū District is relatively isolated from the major infrastructure systems found in those communities, including wastewater treatment and disposal facilities. The community of Nā'ālehu is about 2.06 miles from the shoreline. Most of the community lies approximately 740 feet above mean sea level.

The existing sewer system that services the community was privately owned and operated by the C. Brewer, which merged with Hawaiian Agriculture Company (HAC) in 1972. The sewer system is currently owned by the County DEM. The sewer system discharges wastewater into large capacity cesspools (LCCs). In total, 164 properties in the Nā'ālehu community are connected to LCCs.

The Project Area includes approximately 204 parcels (in whole or in part) and portions of multiple County streets in Nā'ālehu (Figure 1-3). The Proposed WWTP Site is located adjacent Māmalahoa Highway within a 14.9-acre portion of the Tax Map Key (TMK): (3) 9-5-007-016. The Proposed Wastewater Pump Station will be located on a portion of TMK (3) 9-5-008:048. The Proposed Collection System Area will include portions of Kaalaiki Road, Naalehu Spur Road, Melia Street, Opukea Street, Ohai Road, Lokelani Street, Kilika Street, Kukui Road, and Milo Road.

Two of the LCCs slated for closure are located within TMK parcels (3) 9-5-024:011 (por.) and 9-6-024:001 (por.). The third LCC slated for closure is situated between TMK parcels (3) 9-5-024:010 and 9-5-024:009 (See Figure 1-1).





1.3 Purpose and Need

A portion of the Nā'ālehu community is serviced by a sewer system that was privately built, owned, and operated by the C. Brewer. The C. Brewer built sewer system discharges sewage into three (3) LCCs. Around 2006, C. Brewer requested that the County construct and maintain a new and improved community sewer system. A County Council Resolution approved the C. Brewer request. In anticipation of C. Brewer's dissolution, C. Brewer proposed, and the County agreed, to enter into a formal agreement to not only construct and maintain a new and improved community sewer system but to assume ownership of the existing system including the LCCs by April 30, 2010.

As part of this agreement, for the majority of Nā'ālehu properties connected to the LCCs, C. Brewer committed to complete the line (called a lateral) between the residences and the property line at the edge of the public right-of-way adjacent to the new collection system. It was agreed, if the County did not complete its portion of the work by April 30, 2010, the County would assume pending and unfinished obligations to connect the new laterals installed by C. Brewer to the residences and new collection system when complete. Thus, because that date has passed and the County has not completed installation of the new collection system, this project includes connecting these C. Brewer laterals, which may now need to be replaced, or installing private laterals for currently connected properties if authorized by the property owner and approved by County Council.

In 1998, the U.S. Environmental Protection Agency (USEPA), promulgated regulations, 40 Code of Federal Regulations (CFR) 144.14, that require the elimination of LCCs. In 1999, EPA issued regulations under the Safe Drinking Water Act's (SDWA) Underground Injection Control (UIC) Program which prohibited the construction of new LCCs as of April 2000 and required the closure of all existing LCCs by April 5, 2005 (40 C.F.R. § 144.88). Under federal regulations, an LCC is a cesspool which serves multiple dwellings, or for non-residential facilities has the capacity to serve 20 or more persons per day. For the purpose of complying with EPA regulations, the existing LCC's within the Project Area must be upgraded.

In June 2017, EPA and the County entered into an Administrative Order on Consent (AOC) to close the LCCs serving the Pāhala community by April 2022. Options considered by the County to close the LCCs include construction of a new sewer collection system located within public right-of-way (ROW) and replacement of the existing LCCs with a wastewater treatment plant (WWTP) to address the wastewater treatment and disposal needs of the Nā'ālehu community. The recently Amended AOC that was proposed as of February 14, 2024, requires the LCCs to be closed no later than December 31, 2027.

The County and the EPA voluntarily entered into the initial AOC for the purpose of bringing the County into compliance with the requirements of the Safe Drinking Water Act (SDWA), 42 U.S.C. § 300f, et seq.

EPA has determined that the County, as the current owner and/or operator of three (3) LCCs that serve approximately 164 private residences in the community of Nā'ālehu, violated and continues



to violate the SDWA and its Underground Injection Control program requirements for existing LCCs.

A "cesspool" is a "drywell," which in turn is a "well," as those terms are defined in 40 C.F.R. § 144.3. LCCs include "multiple dwelling, community or regional cesspools, or other devices that receive sanitary wastes, containing human excreta, which have an open bottom and sometimes perforated sides.

Based on the above, the County has outlined that the purpose and need for the Proposed Action is to comply with the requirements and mandates of the SDWA and Amended AOC, and to ultimately close the three LCCs that serve Nā'ālehu. Thus, the core purpose of this exercise is to evaluate, gather community input, and make an informed decision on selecting an option or alternative that will allow the County to close the LCCs, and provide a new, SDWA compliant solution for handling wastewater generated by the Nā'ālehu Community. Closure of the LCCs will eliminate the disposal of untreated sewage into the subsurface which will serve County's mission to protect underground drinking water sources.

In summary, the goals and objectives of the Proposed Action are to:

- Improve the longevity of the sewer systems in Nā'ālehu;
- Construct new sewer collections systems or individual wastewater systems to serve portion of Nā'ālehu;
- Close the three existing LCCs currently serving the community in compliance with the SDWA; and
- Meet the compliance requirements of the Amended AOC and the applicable portions of the Clean Water Act by assessing four alternatives (further discussed in Section 2.1 of the EID-EA).



(This page intentionally left blank)



CHAPTER 2: PROJECT DESCRIPTION

2.1 Proposed Action

The County DEM WWD is proposing to construct new facilities for wastewater treatment to serve portions of Nā'ālehu as described in Section 1.1 of the EID-EA. The Proposed Action will allow the County to close the three existing LCCs in the Project Area, and thereby meet the compliance requirements of the Amended AOC and the applicable portions of the Clean Water Act. The Proposed Action would be achieved by any of the four alternatives set forth in the Amended AOC, further described below.

2.1.1 Alternative 1: Package Plant New Collection System

Under Alternative 1, the County would construct a new sewer collection system in the Project Area to replace the existing system of gravity lines that convey sewage to the three LCCs and connect it to the proposed wastewater treatment and disposal facility. The WWTP would serve the former Brewer lots as well as newly accessible parcels for future connection to the WWTP (See Figure 2-1). The potential connection of additional parcels may be further assessed by the County in the future.

Alternative 1 involves utilizing the existing collection system within the Brewer Company house lots and constructing new gravity sewers, wastewater pump station, and force main to transport sewage from the LCCs to the new WWTP (See Figure 2-2). Alternative 1 also involves the installation of gravity sewers to replace the existing collection system (See Figure 2-3). Construction may be conducted as one or multiple phases, and the County shall ensure that residential units can maintain access to the sewer system at all times.

Under this alternative, the County would perform the following actions:

- 1. Construct a new gravity sewer on Kaalaiki Road and Nāʿālehu Spur Road to the WWTP located on a portion of Tax Map Key (TMK) (3) 9-5-007:016, as the current proposed approach to bring sewer to the WWTP location
- 2. Implement drainage improvements within the vicinity of the WWTP within TMKs (3) 9-5-007:016 and (3) 9-5-021:015.
- 3. Construct a new pump station located on a portion of TMK (3) 9-5-008:048, and construct a new force main, which crosses an existing storm drainage channel at Melia Street, to connect to the Kaalaiki Road gravity sewer.
- 4. Construct a new gravity sewer on Opukea Street and Ohai Road to intercept existing flows entering the LCCs and divert sewage to the wastewater pump station (WWPS) and transport flows to the gravity sewer along Kaalaiki Road.
- 5. Install gravity sewers within the streets to replace the existing collection system.
- 6. Close and abandon the three LCCs.
- Accomodate future expansion of subsurface effluent disposal located within a portion of TMK (3) 9-5-022:001.









Package Plant

The Amended AOC allows for construction of a Package Plant to treat sewage currently being disposed in the 3 LCCs. In addition, after treatment of the incoming sewage flows, disposal of the treated effluent would occur using a subsurface irrigation system. The following sections describe the components and facilities which would comprise the package plant to treat the sewage and dispose the treated effluent which would be applicable to Alternative 1 and Alternative 2.

A Preliminary Engineering Report (PER), prepared by Brown and Caldwell in August 2023, provides technical information related to the analysis used by the County to select the package plant to be used to treat incoming sewage flows and a method to be used for disposal of effluent from the wastewater treatment plant (WWTP). As described in the PER, the package plant and effluent disposal method would be configured and developed within the 28.739-acre Project Area located makai of Māmalahoa Highway. Final design and configuration may differ from overall site plan depicted in the PER. For more information, see Appendix A.

The PER indicated accurately quantifying flow projections for the Nā'ālehu community is necessary to design an appropriately sized wastewater treatment and disposal facility. The WWTP design will need to provide sufficient capacity for the existing parcels within the service area, including newly accessible parcels reflecting currently developed portions of the Nā'ālehu community. This will allow the County to close the LCCS. The design will provide sufficient area within the WWTP site for future expansion of the package plant.

HAR Section 11-62-24(b) requires Counties to use their adopted wastewater flow standards to develop flow projections for WWTPs. Counties are to use the City and County of Honolulu (CCH) flow standards if they have not adopted their own standards. The County has not adopted its own flow standards, so wastewater flow projections were developed using the current CCH (2017) wastewater standards. However, flow projections based the current wastewater standards based on urban Honolulu are likely overly conservative for rural communities like Nā'ālehu.

The PER indicates the amount of wastewater generated within a residence will not exceed the amount of potable water used by the occupants. Therefore, potable water use records can be used to estimate wastewater generation rates within existing communities where no combined sewers are present. The County Department of Water Supply (DWS) provided potable water use records from November 2017 through October 2022 for the parcels located within the service area. Analysis of the potable water use records indicates that an 80,000 gallons per day (gpd) monthly wastewater generation rate would reflect the current needs of the service area. Using a 2.5 peaking factor to estimate the maximum wastewater flow into the collection system results in a maximum wastewater flow of 200,000 gpd.

As stated in the PER, groundwater can infiltrate into wastewater collection systems during dry weather, increasing flows to the WWTP. The 2017 CCH standards specify a dry weather infiltration and inflow (I/I) allowance of 35 gallons per capita per day (gpcd). The previous CCH standards (dated 1993) specified a dry weather I/I allowance of 5 gpcd for properties located above the groundwater table. Through the County's experience at the Honokaa WWTP evaluating dry weather, I/I for a rural collection system located in Hawai'i Island's well-drained geology, at elevations hundreds of feet above sea level and a significant distance from the shoreline,



continued use of the 1993 standard for dry weather I/I is appropriate for Nā`ālehu and using the 2017 standard would be overly-conservative.

The 2017 CCH standards specify a wet weather I/I allowance of 3,000 gallons per acre per day (gpad) were used for all wet weather I/I calculations. Evaluation has determined that a peak day wet weather peaking factor of 3.5 is recommended for the Nā'ālehu WWTP design.

HAR 11-62-23.1(i) requires the initiation of a facility planning process when the actual wastewater flows reach 75 percent of the design capacity of the WWTP, and implementation of the facility plan must be initiated when actual wastewater flows reach 90 percent of the design capacity. In anticipation of future development within the Nā'ālehu community, the PER recommend the WWTP design be rated to treat an average dry weather flow of 125,000 gpd to avoid the potential of having to initiate a facility plan shortly after the project is constructed. Note, the biological processes in the mechanical WWTP will need to be sized to treat the peak day dry weather flow of 212,000 gpd, not the average dry weather flow.

If the County pursues a WWTP approach to close the LCCs then a DOH variance from HAR 11-62 requirements will be needed. The variance will need to be renewed every 5 years. The WWTP capacity needs should be re-evaluated upon application for the variance renewal.

The PER provides a description of the package treatment plant that is recommended to be implemented at Nā'ālehu. Note, package plants typically consist of pre-manufactured treatment facilities/components that may be configured to treat wastewater in small communities or on individual properties. The Nā'ālehu WWTP would be configured within a 28.739-acre portion of TMK (3) 9-5-007:016. Pending final design, lands within this 28.739-acre Project Area which are not utilized to construct the package plant and related facilities or subsurface disposal of treated effluent shall be converted back to conservation land. A security fence would surround the site. The security fence would not include barbed wire stringers. Figure 2-4 shows the site plan for the WWTP.

The package plant includes the headworks, grit drying bed, potable water tank, utility or operations building which includes a blower room, an emergency generator room, electrical room with a monitor control center, a maintenance and storage room, and restroom, an above ground fuel storage tank, and an irrigation control tank. Figure 2-5 shows the operations building floor plan.

The Nā'ālehu package plant will include preliminary treatment, odor control and secondary treatment, and disposal of treated effluent. The preliminary treatment system will include influent flow measurement, influent sampling, screening, and grit removal. The narrative description of the proposed improvements contained herein mirrors the recommendations set forth in the subject PER, however, the CoH-DEM reserves the right / flexibility to utilize its discretion to utilize and implement alternative methods / technologies / processes that it deems superior / equivalent in regards to meeting the purpose and need of the Proposed Action. As the design phase progresses, modifications to the current conceptual site plan and operations building floor plan may occur. While the overall size of the structures is not anticipated to change significantly, adjustments to the site configuration could occur within the identified parcels.







PROPOSED WWTP SITE PLAN

FIGURE 2-4

NA'ALEHU WASTEWATER TREATMENT PLANT * Subject to change*



Preliminary Treatment

Influent flow measurement is recommended in the PER to allow assessment of flows and loads to the biological treatment process, and to assess the biological treatment process performance. A Parshall flume will be provided upstream of the screening system to continuously record influent flow rates. Parshall flumes work well for influent measurement because the flume can operate in an open-channel configuration, can accommodate wide ranges of flows, and is self-cleaning. A straight approach length of at least 20 times the flume throat width will be provided upstream of the flume to provide favorable hydraulic conditions.

An automatic refrigerated composite sampler is recommended to allow influent composite samples to be collected. Influent composite samples, when combined with influent flow measurement, can be used to calculate influent mass loading rates to the WWTP to assess the treatment performance and optimization of aeration rates in the biological treatment process. Periodic influent sampling is also recommended to monitor for changes in the influent characteristics.

Screening is recommended to protect the downstream system operation from large objects, debris, wipes, and rags that can be present in wastewater. The industry trend is towards finer screening systems that remove greater amounts of debris from the waste stream; screens with 6-millimeter (mm) (¼-inch) openings are frequently used for activated sludge treatment systems. Finer screens are used upstream of membrane bioreactors to remove hair that can foul the membranes. The screenings volume at the Nā'ālehu WWTP is expected to be small, subsequently screenings disposal is expected to be infrequent; weekly at most. Therefore, the screenings must be washed of organic debris to prevent the accumulation of nuisance odors and flies in the screenings barrel or bag between screening disposals events.

The PER recommended an in-channel cylindrical screen for this installation. The in-channel cylindrical screen combines screening, screenings washing, dewatering, compacting, and bagging/disposal within a single unit as shown in Figure 2-6. For this installation, the headworks will include one in-channel cylindrical screen, plus a bypass channel with manually cleaned bar rack. This system is recommended as it minimizes additional screenings handling and odor potential. The screening mechanisms provided in the final design of the Proposed Action may differ from what has been recommended by the PER depending on the selected manufacturer.

According to the PER, grit removal is crucial to help prevent wear to downstream equipment, costly service interruptions and repair. Grit is comprised of particles that are heavier than the organic biodegradable matter in wastewater. Grit particles can consist of sand, gravel, pebbles, silt, cinders, ground bone, eggshells, coffee grounds, and other materials. Grit in the wastewater collection and treatment system causes abrasive wear to mechanical equipment, piping, and appurtenances. Grit can also form deposits in pipelines, channels, and tanks, which reduces hydraulic capacity and can damage equipment.



2 Secondary Wash Wash Water Inlet 1 Primary Wash

Figure 2-6 In-Channel Cylindrical Screen

The PER recommended the sue of an aerated grit chamber for the Nā'ālehu WWTP. Aerated grit chambers are tanks that function specifically to remove inorganic solids from the wastewater stream as shown in Figure 2-7. Aerated grit tanks are designed to induce sufficient vertical velocity to separate organic and inorganic solids. In theory, inorganic solids have a higher specific gravity than organic solids, and therefore require higher vertical velocities to keep them in suspension. Air diffusers placed near one longitudinal tank wall induce a roll in the contents of the grit tank. This roll creates maximum velocities near the walls and lower velocities at the surface and bottom of the tank. The lower transverse horizontal velocities allow inorganic particles to settle out and be transported to the grit hopper by shear-induced currents. Grit removal mechanisms provided in the final design of the Proposed Action may differ from what has been recommended by the PER depending on the selected manufacturer.

The aerated grit chamber design is based on providing sufficient hydraulic detention time during peak wet weather flow conditions. The PER stated it is necessary to provide at least 10 minutes of detention time to achieve satisfactory grit removal.



Figure 2-7 Aerated Grit Removal Schematic



Nā'ālehu Large Capacity Cesspool Closure Draft Environmental Information Document /Environmental Assessment 2-10 Aerated grit tanks can provide excellent grit removal with minimal head loss, but the chambers themselves require a larger footprint than induced vortex systems. Proper operation of aerated grit tanks can be difficult under varying hydraulic loads due to the need to make fine adjustments to the air diffusers.

Odor Control

The headworks of a wastewater treatment plant is a typical location for foul odor. The odor is caused by hydrogen sulfide (H2S), which is formed under anaerobic conditions found in the wastewater collection system. Due to H2S low solubility in wastewater, when there is an excessive concentration of H2S or if there is turbulence, H2S gas escapes into the atmosphere. This release produces a distinct rotten egg smell. In addition to H2S, there are other foul odorous compounds that can be released from wastewater, such as ammonia, amines, diamines, mercaptans, skatole, and organic sulfides.

In order to control odors associated with the package plant, the PER recommends that covered channels with foul air collection and a granular activated carbon (GAC) scrubber be used at the Nā'ālehu headworks as shown in Figure 2-8. A GAC scrubber passes odorous air through a bed of activated carbon, which absorbs the odorous constituents within the pore spaces of the carbon. Chemical oxidation or reduction of some compounds can also occur. As pore spaces become occupied, efficiency degrades, and the carbon must be replaced or regenerated. Carbon is most effective on higher molecular weight molecules such as the organic sulfur compounds, which makes it the technology of choice. Package GAC scrubbers are available for small headworks and vessels can be situated vertically, horizontally or radially to optimize footprints and reduce structure elevation profiles. The County currently operates GAC scrubbers at other facilities and purchases the GAC media in bulk, which could reduce costs to the County. Odor control measures incorporated in the final design of the Proposed Action may differ from the PER recommendations depending on the selected manufacturer.







Secondary Treatment

Secondary treatment process provides 5-day biochemical oxygen demand (BOD5), total suspended solids (TSS) and nutrient removal via biological treatment. The PER provided descriptions of various secondary treatment options including advantages, disadvantages and applicability to the Nā'ālehu WWTP. Further, the treatment options were screened to identify technologies for further evaluation. Based on the analysis, the PER selected a membrane bioreactor (MBR), activated sludge with anoxic selector, and recirculating gravel filter for further consideration for use at Nā'ālehu. The PER evaluated these three alternatives based on cost considerations and non-economic criteria including level of service measures, regulatory considerations, operation and maintenance factors, and island factors.

According to the PER's evaluation of these alternatives, the costs of the three alternatives are nearly equivalent at the level of analysis provide through the PER; however, the MBR package plant had the highest non-economic score and is recommended for implementation if the County proceeds with a centralized sewer system and WWTP for the community; however, the secondary treatment process incorporated in the final design of the Proposed Action may differ from the PER recommendations depending on the selected manufacturer.

A membrane bioreactor (MBR) has the smallest footprint of the various biological treatment systems available and provides the highest quality effluent. An MBR basically combines an aeration basin with membrane filtration, eliminating the need for tertiary treatment if a very highquality effluent is desired for water reuse purposes.

Membranes provide an absolute barrier to large particles; total suspended solids (TSS) concentrations of the effluent (also known as "filtrate") are typically less than 1 mg/L. Effluent from an MBR process can meet stringent water recycling turbidity requirements without an additional filtration process.

The main difference between MBRs and other biological treatment technologies is the method of separating the bacteria from the clean water. MBRs have thin membranes with many thousands of micro-perforations. Depending on the manufacturer, these perforations are 0.04 to 0.2 microns (4 to 20 hundred-thousandths of a millimeter) in diameter, too small for the passage of most microorganisms or other particles present in the wastewater, but large enough to allow the passage of water molecules.

Disposal / Disinfection

Subsurface drip irrigation system is the recommended method of effluent disposal for the Nā'ālehu WWTP. Subsurface drip will incur lower capital cost and require less attention from WWTP operators with respect to vegetation maintenance than slow rate land treatment. Subsurface drip requires periodic maintenance chlorination to eliminate biofouling in the drip lines. The disposal system would be sized to handle the peak day wet weather flow of 322,000 gpd. An irrigation equalization and control tank are proposed to equalize higher peak flows and to allow discrete dosing of the irrigation zones.

During high flow conditions the irrigation control system would open multiple irrigation zones to accommodate the disposal needs. Additional drip lines will need to be added when the WWTP capacity is expanded. The minimum spacing between drip lines is 2 feet, so there will be sufficient


space between the initial drip lines to add additional drip lines as part of future expansion project(s).

HAR 11-62 requires a fully redundant subsurface disposal system. The design criteria listed in provided in the PER are based on providing a subsurface drip system that is two times larger than needed in order to satisfy the HAR 11-62 requirement for redundancy. The drip system would be divided into two separate systems so that the peak day wet weather flow can be disposed on the site using one system while the second system is out of service for maintenance.

The subsurface drip irrigation concept would retain the existing site topography along with the existing vegetation and use subsurface drip irrigation technology to apply the effluent within the effluent disposal area. The use of subsurface drip irrigation technology to disperse effluent at the site will allow the County to significantly reduce the amount of clearing, grubbing, and grading required to construct the facility, as compared to slow rate land treatment.

Drip irrigation technology has evolved to the point where non-clog emitters are available for subsurface applications of effluent. Non-clog subsurface emitters decrease the potential for the irrigation components to be clogged by roots. Drip tubing with integral emitters is buried 6 to 9 inches below ground. Effluent emitters are typically designed to operate at a flow rate of 1 gallon per hour (gph) and are typically spaced every 2 feet along a drip line. Pressure compensating drip systems typically operate under pressures ranging from 10 to 45 pounds per square inch (psi). Figure 2-9 illustrates the subsurface drip concept.

Figure 2-9 Subsurface Drip Irrigation Concept

Courtesy of Geoflow, Inc,

Subsurface drip irrigation technology incurs greater operation and maintenance cost than a surface irrigation system. The County will need to periodically flush the drip lines to remove debris. As described below, a significant number of drip lines will be necessary to accommodate peak flow rates. In addition, periodic chlorination will be required to remove biological growth from the drip lines. These operations and maintenance tasks will need to be completed on a regular schedule, because the drip system will be buried and not readily accessible or observable. During periods of dry soil conditions, the County will need to inspect the orchard for patches of wet soil that would indicate a localized failure that requires repair. Flow and pressure monitoring will also be useful tools for validating the status of the subsurface drip system. The land treatment area would be divided into multiple irrigation zones, allowing a zone to be taken out of service



for maintenance purposes. A fence will be constructed around the site to deter entry by humans and ungulates.

New Collection System

Under Alternative 1, the County would construct a new sewer collection system in the Nāʿālehu community to replace the existing system of gravity lines that convey sewage to the three LCCs and connect it to the proposed wastewater treatment and disposal facility. The WWTP would serve the former Brewer lots as well as newly accessible parcels for future connection to the WWTP (See Figure 2-10). The potential connection of additional parcels may be further assessed by the County in the future. The new system would be constructed within County streets to meet the County sewer standards and to allow the collection system to be owned and operated by the County. In addition, two backyard easements would be used to connect four properties to the collection system.

All accessible properties would be required to connect to the new wastewater collection system in accordance with Hawai'i County Code, HCC, § 21-5. The new collection system would be subject to HCC 21 (Sewers). Specifically, HCC § 21-5 states the following:

"(a) Owners of all dwellings, buildings, or properties used for human occupancy, employment, recreation, or other purposes, which are accessible to a sewer are required at their expense to connect directly with the public sewer within 180 days after date of official notice.

(b) If, due to rock, wastewater collection system depth, or other construction problems, a building cannot be practically served, the owner shall install, operate and maintain a residential pumping station.

(c) The director may grant a variance/exemption of the foregoing connection requirements to owners of single-family dwellings existing at the time of installation of the public wastewater system, if the following is found:

(1) There are special or unusual circumstances applying to the subject real property which exist that render the ability to connect to a wastewater system an extreme physical or financial hardship; and

(2) There are no other reasonable alternatives; and

(3) The variance is consistent with the general purpose of the chapter and will not be materially detrimental to public health, safety, or welfare."

Accordingly, additional newly accessible lots in Nā'ālehu would be required to connect to the new wastewater collection system after it becomes operational. These other lots are near the existing service area and are presently connected to individual wastewater systems or cesspools. The design of the new collection system would include stub-outs to accommodate the eventual connection of these newly accessible lots. However, the respective lot owners would be responsible for the design and completion of these connections and for the proper closure of their individual wastewater systems.

2.1.2 Alternative 2: Package Plant Existing Collection System

Under Alternative 2, the County would perform the same actions as described under Alternative 1; however, gravity sewers would not be installed within the streets to replace the existing



system. Instead, the new wastewater treatment and disposal facility and pump station would be connected to the existing 80-year old collection system.

Existing Collection System

In 2004, C. Brewer contracted M&E Pacific to conduct a sewer system evaluation for the town of Nā'ālehu. The results of this investigation determined that the existing sewer lines and manholes do not conform to County sewer design standards. The existing sewer system was not constructed in the streets, but instead runs through easements located on private properties, with many collection lines running adjacent to or beneath the houses. The results of a smoke test performed during the 2004 sewer system evaluation identified at least 13 locations of line breaks and/or pipe defects and 12 household units with defective sewer vents. In addition, the existing sewer system is over 80 years old. Long surpassing its expected lifespan and will require extensive repair and rehabilitation if chosen to be reused.

Nearly 20 years have passed since the 2004 study was completed. In order to reuse the existing collection system into the future, an updated condition assessment is recommended to better identify system deficiencies. Substantial improvements will likely be necessary due to the age of the system. Reusing the existing collection system would require the construction of new gravity sewers, wastewater pump station, and force main to transport sewage from the LCCs to the new WWTP.

Although reusing the existing appears to incur lower life cycle costs than constructing a new collection system, it is not recommended by the PER. Due to the advanced age of the existing collection system, the option would incur substantial financial, public health, and environmental risks to the County including:

- The piping is at the end of its useful service life; catastrophic failures are likely to increase • in frequency, resulting in greater risks to public health and the environment,
- Most of the system is located in backyard easements, making it difficult to access and • maintain,
- System expansion to accommodate sewering additional areas of the town (in accordance with the Kau CDP) would not be feasible.
- The option does not address the AOC requirement to connect additional properties that are not currently connected to the collection system to the WWTP.

The County would construct the collection system in two phases to ensure residential units can maintain access to the sewer system at all times. Phase 1 involves utilizing the existing collection system within the Brewer Company house lots and constructing new gravity sewers, wastewater pump station, and force main to transport sewage from the LCCs to the new WWTP. Specifically,

2.1.3 Alternative 3: Individual Wastewater System (IWS) – Maintenance Contract Model

In Hawai'i, an Individual Wastewater System (IWS) is defined by the Department of Health (DOH) as a wastewater system for an individual property that receives less than 1,000 gallons per day of wastewater flow or serves five bedrooms or less. An IWS can be a cesspool, a traditional septic system, an aerobic treatment unit (ATU), or other means of treatment achieving National Sanitation Foundation (NSF) 40 quality effluent.



As stated in the August 2023 PER, the State DOH Wastewater Branch is responsible for regulating IWS systems. The Amended AOC sets forth that the County Hawai'i must administer a more active management strategy than is typically found in Hawai'i IWS. Further, the Amended AOC states that either a Model 2 (Maintenance Contract) or a Model 3 (Operating Permit) must be used IWS systems at Nā`ālehu.

Under Alternative 3, the County is to:

- Fund design and manage project construction of the IWS systems;
- Administer ad manage a maintenance program for IWS;
- Develop a maintenance program which would entail establishing rules and regulations for monthly fees/penalties, County monitoring and reporting, and IWS educational information for homeowners;
- Operate the system and conduct routine maintenance, and response to any related trouble calls; and
- Prepare and submit related notices and reports.

Figure 2-10 show the properties where the IWS systems would be constructed.

Lots to be served in the Nā'ālehu community vary in size from 0.12 to 1.94 acres with a median size of 0.16 acres. HAR 11-62-3.1 (2)(A) states 10,000 square feet (0.23 acres) of usable land must be available for each IWS. Of the 174 properties to be served in this project, 81 have less than 10,000 square feet of total area. Space available for IWS installation on these properties is further limited by the presence of existing structures.

The actual location of treatment and disposal infrastructure is further limited by setback requirements. DOH-required setbacks are presented below (Table 2-1). From a system design perspective, it is recommended that systems should also be a minimum of 20 feet from any cutface slopes present on a site to avoid surfacing of treated effluent. This is a particular constriction to heavily sloped sites.

Table 2-1 DOH Required Setbacks for Wastewater Systems			
Features	Treatment Unit (ft)	Seepage Pit (ft)	Soil Absorption System (ft)
Structure Wall Line	5	5	5
Property Line	5	9	5
Surface Water Body	50	50	50
Large Trees	5	10	10
Treatment Unit	5	5	5
Seepage Pit	5	12	5
Soil Absorption System	5	5	5
Municipal Water Supply Well	500	1000	1000

Sites in Nā'ālehu have slopes that vary from 6-10%. This is likely to affect the constructability of absorption beds as a method of wastewater disposal. Per HAR 11-62-34, absorption beds shall not be installed on land with a slope gradient greater than 8%, while absorption trenches are





permitted on a slope of up to 12%. In case this slope requirement cannot be fulfilled, the DOH would allow a seepage pit to be installed instead of an absorption bed.

It is generally not good practice to install an IWS under a trafficked or otherwise concreted area. The presence of concrete or traffic compresses the soil in distribution systems and affects the accessibility of the system for maintenance. However, it is sometimes unavoidable on particularly spatially constrained properties. In this event, a system may be installed underneath a driveway or vehicle path of travel provided the system is designed to that end and traffic rated treatment components are used. These may include products such as concrete septic tanks and/or H-20 traffic related chambered disposal beds.

Septic Tanks

The PER stated septic tanks are the most common conversion treatment technology installed in Hawai'i. A septic tank is an underground chamber made of concrete, fiberglass reinforced polyester, or plastic, and used for treating and disposing of household wastewater. The tank contains a mixture of untreated sewage and anaerobic bacteria, which break down the waste and separate it into three layers: a top layer of scum, a middle layer of liquid effluent, and a bottom layer of sludge. Septic tanks operate without the need for electrical power. Contractors are familiar with the process of installing IWS systems, which minimizes potential installation mistakes.

The septic tank can have two chambers with sewage flowing from the home flowing into the first chamber where the heavy solids will settle to the bottom as sludge and the remaining liquid and lighter solids floating to the top as scum. The floating liquid will flow to the second chamber through an opening in the wall where any remaining solids will settle to the bottom and then effluent will flow to disposal system. Access to the tank will be sealed to retain the anerobic conditions which will help to control odor.

The liquid effluent flows out of the tank and into a means of disposal, where it is further treated and dispersed into the soil. According to the PER, the sludge and scum remain in the tank and must be periodically pumped out by a professional septic service approximately once every three to five years, depending on usage. Figure 2-11 shows a typical septic tank.



Please note: The number of compartments in a septic tank vary by State and region. Source: U.S. Environmental Protection Agency



The August 2023 PER stated there are several types of septic tanks from suppliers in Hawai'i (See Figure 2-12). The tanks can be made from concrete, plastic, and reinforced fiberglass polyester which come at of a variety of price points, each with a set of advantages and disadvantages and materials. Where a septic tank is located beneath a vehicular traffic area, a traffic rated concrete septic tank can be used or a structural concrete slab designed for H-20 loading spanning a non-traffic tank may be used.

Concrete tanks are durable and less susceptible to collapse. However, concrete tanks can be more expensive than plastic or reinforced fiberglass tanks and typically require a crane for installation and can corrode from the effects of acidic sewer gases.

Plastic or polyethylene tanks are less expensive than concrete; resistant to corrosion and may not require a crane for installation. However, plastic tanks may deform and, if not properly installed, can float if flooded.

Fiberglass reinforced polyester (FRP) tanks are less expensive than precast concrete tanks, primarily due to lower shipping and installation costs, typically resistant to corrosion, more rigid than plastic tanks, and may not require use of a crane for installation.

The August PER indicated the choice of septic tank material will depend on availability, budget, and site constraints. At a minimum, septic tanks in Hawai'i must comply with International Association of Plumbing and Mechanical Officials (IAPMO) material and property standards for septic tanks. Further, sizing and installation criteria are regulated by HAR 11-62-33. The minimum septic tank capacity is 1,000 gallons for a household of 4 bedrooms or less and 1,250 gallons minimum for households of 5 bedrooms. Septic tanks serving households greater than 5 bedrooms will require a variance from the DOH.

The PER discusses two commonly used IWS effluent disposal methods found in Hawai'i, absorption bed and absorption trench. Absorption beds are the most common form of IWS disposal method installed in Hawai'i today. The absorption bed will require excavation of 6 to 7 feet below grade to accommodate the network of perforated pipes, each a maximum of 100 feet long and laid in trenches 1.5-3 feet below the finished grade 4-6 feet apart. Each line is laid level to allow the gravity dispersal of the treated effluent through the length of the pipe before it filters out and percolates down into the soil. A minimum of 6 inches of gravel is provided below each pipe. If the percolation rate is faster than one minute per inch, a 3-foot soil replacement layer is installed under the entire absorption bed. The soil replacement is to be washed #4 sand or cindersoil mix with a percolation rate not faster than one minute per inch. The excavated material from the absorption bed will need to be removed off the site.

These systems are easy to maintain when connected to an effective treatment system and will rely on microorganisms in the soil for an added degree of treatment to the effluent as it filters through the upper toxic layers of the soil matrix. However, absorption beds have a significant space requirement with current Hawaiian regulations requiring a minimum of 350 square feet for a 4-bedroom home. This space requirement increases with decreasing hydraulic conductivity of the soil. Additionally, absorption beds can only be installed on a grade of less than 8 percent.







An absorption trench is a type of subsurface wastewater disposal system that utilizes a trench filled with gravel or other porous material to filter and distribute wastewater effluent into the ground. Wastewater is distributed into the trench through a network of pipes, typically made of PVC or other durable materials. The gravel in the trench acts as a natural filter, allowing the water to slowly seep into the surrounding soil while also removing impurities with adsorbed beneficial bacteria. The trench may be lined with a layer of filter fabric to prevent the gravel from becoming clogged with soil or other debris. Figure 2-13 shows a typical IWS site plan with an absorption bed.

The PER indicated, although not as common as an absorption bed or trench, use of a seepage pit should be considered for use in Nā'ālehu . Seepage pits are a vertical means of achieving the percolation area requirements for a disposal system. These systems typically consist of a 15-30-foot-deep pit lined with stacked precast perforated concrete rings or CMUs, to an internal diameter of 6-8 ft. Seepage pits are both less land area intensive and less expensive than absorption beds. Figure 2-14 shows a typical IWS site plan with seepage pit.



Source: Carollo, 2021

A seepage pit must include a cover which extends at least 12 inches beyond the seepage pit excavation or over a provided concrete lining. An access hatch must be provided in the concrete cover to allow inspection and maintenance of the pit. The seepage pit may be designed to be traffic rated by providing the sufficient strength required in the design of the concrete lining and cover.

The effective area of the seepage pit is equal to the vertical wall area corresponding to the effective depth of the pit. Slow percolation rates translate to a larger required absorption area or deeper pit.

While seepage pits are an approved means of disposal in Hawai'i, they are often only permitted when it can be demonstrated that an alternative means of disposal was not possible , i.e. insufficient land area, steep terrain (greater than 12 percent) or very slow percolation rates (less than 60 min/inch). Where slow percolation rates present, seepage pits will need to be dug through the basalt rock layer to reach more porous soils or a variance will be required from HAR 11-62-34 d(1)b:

Seepage pits shall not be constructed in soils having a percolation rate slower than ten minutes per inch (weighted average) or where rapid percolation through such soils may result in contamination of water-bearing formations or surface water.

2.1.4 Alternative 4: Individual Wastewater System (IWS) – Operating Permit Model Under Alternative 4, the County is to:

- Fund design and construction of the IWS systems;
- Administer an operating permit program for the IWS system to the homeowners;
- Issues maintenance notice to the homeowner.

Under Alternative 4, the homeowners would be responsible for maintenance scheduling, contracting and paying for a service provider to conduct the necessary maintenance and/or responding to trouble calls, monitoring and record keeping of maintenance.

Under Alternative 4, the County is evaluating the possibility of completing the project as a conventional Design/Bid/Build process or under a voucher program. The voucher program grant homeowners with funds to hire Professional Engineer to design new IWS and, then hire a Contractor to construct IWS. Under the voucher program homeowners will be responsible to hire and coordinate with a professional engineer for overall design and placement of the IWS.

Based on the above, regardless of the maintenance responsibility, the County will fund the design of the IWS systems for Nā'ālehu. The PER identifies two key considerations to ensure the IWS functions as intended;(1) system size, including the number of bedrooms or flow rate and (2) site considerations including soil type, slope, drainage patterns and accessibility to the IWS site.





Figure 2-13 Typical IWS with Absorption Bed



Figure 2-14 Typical IWS with Seepage Pit



Source: Engineering Partners PER, August 2023



(This page intentionally left blank)



CHAPTER 3: CUMULATIVE EFFECTS

The Proposed Action (construction of a new wastewater treatment and disposal facility and a new collection system, closure of three existing large capacity cesspools (LCCs), and connection of newly accessible properties to the sewer system), in combination with other past, present, or reasonably foreseeable actions at or near Nā'ālehu, could contribute to cumulative improvements and impacts on certain environmental resources. Cumulative effects can result from individually minor but collectively significant actions taking place over a period of time.

3.1 Scope of Analysis

This section identifies the other past, present, or reasonably foreseeable actions at or near Nā'ālehu that were considered and evaluated in a cumulative improvements and impacts analysis as related to Alternative 1 and Alternative 2. Since Alternative 3 and Alternative 4 would affect the individual parcels, these two alternatives are not included in the analysis.

3.1.1 Geographic Scope of Analysis

The extent of the cumulative effects analysis is generally limited to the geographic/natural boundaries of the affected resource areas. The Council on Environmental Quality (CEQ) handbook on Considering Cumulative Effects Under the National Environmental Policy Act indicates that the geographic extent for this analysis should be defined on a case-by-case basis and is dependent on the affected resources (CEQ, 1997).

In defining the geographic scope for consideration of cumulative effects, the DOH and County considered the resources that would be affected by the Proposed Action (i.e., within the project impact zone); the type and intensity of those effects; and whether those affected resources extend beyond the project impact zone. The effects of the Proposed Action would generally be limited to the immediate vicinity of the WWTP project site and related improvements plus minor transportation-related impacts during construction; the Proposed Action would not adversely affect protected species; it would not adversely affect surface waters that are part of a larger watershed (other than potential for temporary, minor construction-related runoff impacts that would be mitigated by adherence to BMPs). Historic or cultural resources identified in the area shall be maintained and preserved to the fullest extent feasible as the County considers ongoing communication and coordination with identified stakeholders. Based on these considerations, the County limited this cumulative effects assessment to include past, present, and reasonably foreseeable actions located within the Naialehu community or within 1 mile of the proposed location of the wastewater treatment and disposal facility and related improvements. This scope is expected to more than fully encompass the full extent of resource areas that would potentially experience discernable effects from the Proposed Action and is commensurate with the type and intensity of the effects of the Proposed Action.

Located approximately 13 miles to the northwest, the Pāhala Large Capacity Cesspools Closure Project (Pāhala Project) is similar in concept to the Proposed Action in that it proposes the closure of existing LCCs and the construction of a new system for a similarly sized community. The County has analyzed whether this and other similar projects throughout the Ka'ū District would have the potential to affect the same resources as the Proposed Action. A typical, similar construction project would be expected to result in temporary, localized impacts during construction including impacts from the use of construction-related vehicles and equipment (e.g., changes in traffic patterns and



increases in noise and air emissions), disturbance of soil and vegetation, and generation of construction and demolition debris' and potential long-term, localized impacts including changes in stormwater runoff and infiltration, removal of vegetation, and changes in visual resources. These direct and indirect effects, if managed in accordance with applicable environmental regulations, would not be expected to extend beyond the vicinity of the project construction sites and local communities; therefore, the Proposed Action is not expected to have any cumulative impacts with the Pāhala Project. The National Environmental Policy Act (NEPA) does not require consideration of socioeconomic impacts that are unrelated to an impact on the physical environment (30 CFR § 1508.14). The cumulative effects of the Proposed Action combined with the Pāhala Project on the County-wide economy, tax base, and borrowing capacity are not analyzed in this assessment.



(This page intentionally left blank)



CHAPTER 4: EXISTING ENVIRONMENT, IMPACTS AND MITIGATION MEASURES

4.1 Climate

Climate on the Island of Hawai'i and more broadly throughout the State can be characterized as having low day-to-day and month-to-month variability. Differences in the climate of various areas are generally attributed to local differences in geology and topography that create microclimates with different temperature, humidity, wind, and rainfall, and associated local ecosystems (Giambelluca et al. 2014). Hawai'i Island is characterized by a relatively uniform semi-tropical climate year-round with mild temperatures, moderate humidity, and relatively consistent northeasterly trade winds.

The climate of Nā'ālehu communities is typical of the Ka'ū District which is pre-dominantly dry. The National Ocean and Atmospheric Administration (NOAA) designates Ka'ū areas as a Humid Tropical Zone with transitional lowland areas in locations between windward and leeward regions. The Ka'ū area receives less orographic rainfall since it is not oriented to typical trade wind flow and exhibits a distinctive dry summer season.

Temperatures in the Ka'ū District generally range from approximately 70 to 80 degrees Fahrenheit during daylight hours and 60 to 70 degrees Fahrenheit during night hours.

Prevailing trade winds in the Ka'ū District area are from the southeast and usually dominate from April to November. Wind speeds average about 10 miles per hour and vary between approximately 10 - 15 miles per hour in Nā'ālehu. Winds from the southwest occur less frequently, mainly during the winter associated with "Kona" storms (Giambelluca et al. 2014).

Climate conditions in the Ka'ū District are most likely to change in the coming decades. Average annual precipitation is also likely to change, but climate models are uncertain regarding projections for Hawai'i. Cased on ensemble model projections available through the U.S. Environmental Protection Agency's (EPA's) Climate Resilience Evaluation and Awareness Tool (CREAT) Climate Scenarios Projection Map, projections for the area surrounding the Nā'ālehu range from a minor decrease in annual precipitation up to considerable increases in annual precipitation by 2060, depending on the model scenario (hot/dry vs. warm/wet). Other climate concerns include sea level rise to coastal areas.

Impacts and Mitigation Measures:

<u> Alternatives 1 and 2 – Package Plant:</u>

Under these alternatives, the construction phase may result in temporary greenhouse gas emissions due to heavy equipment operations and the transportation of supplies to the WWTP project site and along the streets for the new collection system. However, these emissions will be temporary during the construction period. Upon completion of the construction, emissions and greenhouse gases should return to current conditions.

In the long-term, the project's wastewater treatment and disposal facility does not discharge to surface water sources, therefore, it is unlikely to significantly impact local temperature or precipitation patterns. Climate models suggest variability in precipitation; however, this is not expected to be substantially impacted by the project.



The project's design takes into consideration the potential for increased storm intensity due to climate change. Drainage improvements will be implemented within the WWTP project site to manage stormwater during intense weather events, ensuring the facility's resilience to changing storm patterns.

Alternatives 3 and 4 – Individual Wastewater System Program:

Alternatives 3 and 4 involve the implementation of an IWS Maintenance Contract Model or Operating Permit model. This alternative would require construction activities on individual parcels including measures to contain storm water runoff during storm events. The climate impacts are like those in Alternative 1 and 2, with the project not significantly affecting local climate conditions. The design includes measures to address potential changes in storm intensity, ensuring the facility's resilience.

No Action Alternative:

Under the No-Action Alternative, the existing LCCs continue to operate without any modifications. These LCCs are at risk of impacts due to climate change, specifically changes in precipitation patterns, increased storm intensity, and potential sea level rise. Climate change related impacts on the existing LCCs could result in risks to groundwater and surface water quality. Increased storm intensity and altered precipitation patterns may exacerbate the challenges of managing wastewater in these LCCs, potentially leading to overflows or groundwater contamination.

4.2 Physiography

4.2.1 Topography

The Nā`ālehu community is situated on the slope of Mauna Loa. Sites in Nā`ālehu have slopes that vary from 6-10% and ranging in elevation from 780 ft to 645 ft above mean sea level (MSL) (See Figure 4-1). The existing topography in Nā`ālehu is characterized by gradual slopes, which significantly impact the community's layout and land use patterns.

The Project Area is approximately 33.3 acres and bounded by Kaalaiki Road to the west and Māmalahoa Highway to the south.

Impacts and Mitigation Measures:

<u> Alternatives 1 and 2 –Package Plant:</u>

The topographical conditions under Alternative 1 are consistent with the existing conditions. Construction activities for the new collection system and treatment facility will involve grading and earthwork at the package plant site. Erosion control measures will be implemented to prevent soil erosion and maintain the existing topographic conditions. Excavation depths of 4 to 10 feet would be necessary to place the various components of the package plant. The effluent disposal facility would require excavation of trenches of up to 3 to 4 feet to place the subsurface drip irrigation lines. The affected areas would be restored to approximately existing conditions.





Source: State OP and ESRI

The wastewater collection system would be constructed within the right-of-way of the County streets plus two segments within easements. Due to the existing topography, the collection system can be fed by gravity if the pipes are deep below the low point of low-lying parcels.

The sewer trenches are typically approximately 3 feet wide with depths between 10 to 15 feet to suite the gravity sewer lines. Once the line is placed in the trench, the affected area would be backfilled to restore the area to existing conditions which would have minimal effect on the topographic conditions of the area. Proper erosion control measures will be implemented to prevent potential soil erosion during the construction period. Construction of the package plant, subsurface irrigation system and collection system would not create significant changes to the existing topographic conditions of the Nā'ālehu area.

Alternatives 3 and 4 – Individual Wastewater System Program:

The topography for Alternatives 3 and 4 aligns with the existing conditions in Nā`ālehu. Construction activities for the IWS will involve excavation for placement of the septic tank and absorption bed. Once the tank and effluent disposal system have been put into place, minor grading will be needed to restore the affected area. Erosion control measures will be implemented during construction to prevent runoff and soil erosion during storm events. These measures will be designed to protect the existing topographic conditions of the surrounding area.

No-Action Alternative:

Under the No-Action Alternative, the existing LCCs are situated within the existing topography. There are no modifications or construction activities associated with this alternative, and the topography remains unchanged.

Proper erosion control measures will be implemented to minimize potential impacts on the topography during construction activities, ensuring its preservation. Overall, the topography in the Nā`ālehu area is not anticipated to undergo significant alterations as a result of the proposed project alternatives.

4.2.2 Geology

The Island of Hawai'i was formed by the activity of five shield volcanoes. These shield volcanoes are Kohala (extinct), Mauna Kea (active), Hualalai (active), Mauna Loa (active), and Kilauea (active) (USGS).

The Project Area is situated at the eastern end of the island and on the lower, southeastern flank of the Mauna Loa Volcano. This volcano appears to be made up of at least two huge shield volcanoes built around two separate eruptive centers, referred to as the Mauna Loa shield. The Mauna Loa shield was created by eruptions along two rift zones that extend in a southwest and east-northeast direction from the caldera. Rift zones are elongated areas of the ground fissures where volcanic activity such as earthquakes and volcanic eruptions are concentrated. In contrast, a few eruptions have taken place along the lower northeast rift zone.



The surrounding area consists of several interstratified beds of volcanic ash that sit upon the exposed bed rock. The Nā'ālehu area is known to contain lava tubes, which often occur in many places around the Island of Hawai'i. Lava tubes are natural conduits or voids that form when molten lava flows beneath the hardened surface of a previous lava flow. When the volcanic eruption stops, and the lava drains out, a lava tube forms in the void. Lava tubes can range in size from a few inches to more than two feet in diameter. The diameter and length can usually be identified through subsurface probing or geophysical surveys.

Fieldwork completed during an Archeological Inventory Survey (AIS) by Cultural Surveys Hawai'i (CSH) identified a depression caused by a collapsed lava tube system within the proposed WWTP project area where one lava tube opening was encountered.

Impacts and Mitigation Measures:

<u>Alternatives 1 and 2 – Package Plant</u>

Grading, excavating, and fill activities during the construction of the WWTP package plant and effluent disposal system and the new collection system would occur between 10 to 25 feet below grade. A preliminary layout of a gravity sewer system resulted in sewer depths of about 20 to 25 feet at the deepest point on the upper portions of Ohai Road and Nahele Street, and 15 to 18 feet at the deepest point on Lokelani Street and Kilika Street. The soil depth to rock in the area as reported by USGS and verified by past geotechnical investigations is 40 to 60 inches; therefore, excavation for the deep gravity sewers would be mainly in rock.

If bedrock is encountered during excavation for the proposed actions, removal would be accomplished using excavators or hydraulics hoe rams and/or pneumatic drill hammers consistent with other construction activities on the Hawaiian Islands. Archaeological monitoring is recommended during all project ground-disturbing activities to facilitate identification for information purposes, especially of any subsurface lava tubes that may contain historic properties.

Alternatives 3 and 4 – Individual Wastewater System Program:

Alternatives 3 and 4 involve the installation and maintenance of the IWS. While this alternative may require construction activities that could disturb deeper geological layers up to 10 feet, this depth of excavation should not affect the geology of the Nā'ālehu area. The August 2023 PER indicated construction/installation of an IWS can be relatively invasive process to a homeowner, involving large equipment such as excavators and cranes and removal of fencing, trees and landscaping and, in some cases, small structures. Determining a path to bring large equipment to the IWS site would need to account for building/structure footprints and roof overhangs and soffits.

No-Action Alternative:

Under the No-Action Alternative, the existing LCCs continue to operate without any modification. This alternative does not involve any construction activities or modifications to the existing conditions, and therefore would not have any impact on geology in the Nā'ālehu area.



4.2.3 Soils

According to the U.S. Department of Agriculture Natural Resource Conservation Service (NRCS) Soil Survey of the island, soil types within the Nā'ālehu area include Nā'ālehu medial silty clay loam which possess moderately high to high permeability characteristic and consist of well-drained soils formed of volcanic ash. The Nā'ālehu area also include Kanohina-Lava flows complex and Lava flows- Kanohina complex soils which are considered to have high surface runoff; however, flows reaching the lava area fan out and are known to percolate through tubes within the lava rock. Specifically, the Nā'ālehu area is comprised of the following soil profiles (See Figure 4-2):

- Nā'ālehu medial silty clay loam, 0 to 3 percent (Map Unit Symbol (MUSYM):538)
- Nā'ālehu medial silty clay loam, 3 to 10 percent slopes (MUSYM: 521)
- Nā'ālehu medial silty clay loam, 10 to 20 percent slopes (MUSYM: 522)
- Nā'ālehu medial silty clay loam, 20 to 35 percent slopes (MUSYM: 523)
- Kanohina Lava flows complex, 2 to 10 percent slopes (MUSYM: 734)
- Lava flows Kanohina complex, 2 to 20 percent slopes (MUSYM: 271)

Impacts and Mitigation Measures:

All Proposed Alternatives

Construction of the new collection system would occur below the travel ways or shoulders of the streets in the Nā'ālehu community. The collection system would not generate adverse impacts to soils in previously disturbed areas.

Construction of remaining proposed alternatives would require vegetation removal, clearing, and excavation. These activities would involve subsurface and surface disturbance to the soils of the affected areas. These impacts can be mitigated through incorporating appropriate stormwater and erosion control measures to ensure that soil erosion and transport during construction activities are minimized. Typically, the construction plans and documents would include erosion control plans which the construction contractor would need to follow.

No-Action Alternative

Under the No-Action Alternative, the existing LCCs would continue to operate without any modification. This alternative does not involve any construction activities or modification to existing conditions, and therefore would not cause any impacts to soils in the Nā'ālehu area.





4.3 Water Resources

4.3.1 Surface Waters

Water resources in the area are important for the existing wastewater treatment systems and the local community. Groundwater and surface water sources play a vital role in sustaining the environment and supporting agriculture.

The Nā`ālehu community is situated southeast of the Alapa`i Gulch. The southern point of the existing Nā`ālehu Wastewater System service area, north of Mamalahoa Highway, is located approximately 2 miles from the nearest coastline. According to USGS geologic maps, Honu`apo Bay is located approximately 2.5 miles from the Nā`ālehu community.

The nearest stream within the Project Area is Nāʿālehu Stream, a non-perennial stream which originates from flows that begin at Ka'ū Forest Reserve (Figure 4-3). A concrete drainage channel runs between Ka'alaiki Road and the residential community neighborhood along Melia Street. The drainage channel is utilized to carry storm runoff from upslope safely through the town and crosses under Mamalahoa Highway to the south. Further, an existing vegetated diversion channel and drainage outlet are located within the WWTP project site.

Impacts and Mitigation Measures:

<u> Alternatives 1 and 2 – Package Plant:</u>

Under Alternative 1, construction activities may temporarily impact water resources. The cumulative impacts of the wastewater treatment and disposal facility as well as the new collection system would require coverage under a National Pollutant Discharge Elimination System (NPDES) construction stormwater permit. Normally, this permit would mandate the implementation of best management practice (BMP) measures including silt fences, filter socks, and sediment traps to control sediment runoff. Since the NPDES permit requires detailed information about the means and method of construction, the selected contractor would need to submit the plans and documents to the DOH.

Construction trenches would be designed not to extend deeper than approximately 10 feet below grade when feasible, minimizing disturbance to the geology in the Nā'ālehu area. It should be noted that construction trenches may need to exceed 10 feet in depth in some design locations. A Site-Specific Construction BMP plan would be developed to prevent stormwater runoff along the collection system during construction.

The on-site drainage plan, as per Hawai'i County Code, Chapter 27, Section 20, would ensure that runoff caused by the construction activities in the WWTP project site area and effluent disposal area would need to account for expected one-hour, ten-year storm event, is retained within the site boundaries. Landscape buffers with dirt berms would act as secondary containment during large storm events, further safeguarding against adverse impacts on adjacent or downstream properties.





Source: State OP and ESRI

Overall, construction-related impacts on surface water resources under Alternatives 1 and 2 are expected to be temporary in nature, with BMPs effectively minimizing potential impacts. However, the proposed WWTP is anticipated to involve drainage improvements including the realignment of the vegetated drainage diversion channel and relocation of the drainage outlet which is considered to be a long-term impact. The drainage improvements shall be designed in accordance with the County Department of Public Works (DPW) drainage standards and will not alter existing stormwater discharge from the property.

Alternative 3 – Individual Wastewater System-Maintenance Contract Model:

Alternative 3 entails the implementation of an IWS Maintenance Contract Model which can be an effective means of wastewater management when properly designed and operated. However, IWS that are poorly designed and maintained have been nationally recognized as having high failure rates. In order to ensure that the systems function as intended, the design must take into account a variety of technical considerations including system size, site conditions, location, subsurface soil characteristics, influent characteristics, and level of treatment.

Should the IWS begin to fail, untreated sewage containing pathogens (e.g. E. coli), nutrients and other harmful substances could be discharged into the groundwater or into nearby surface waters.

In the event that the County pursues the IWS Alternatives, the implementation of BMP measures, including silt fences, filter socks, and sediment traps, would be required during construction at each IWS site to control sediment runoff. Construction trenches would generally not extend deeper than approximately 10 feet below grade. Further, the IWS designs will need to consider the ground slopes and drainage patterns to ensure that effluent does not seep through these surfaces.

Overall, construction-related impacts on surface water resources under Alternative 3 are anticipated to be temporary and localized, with BMPs effectively minimizing potential impacts.

Alternative 4 – Individual Wastewater System-Operating Permit to Homeowners:

Alternative 4 involves the implementation of an IWS Operating Permit model. As noted under Alternative 3, a properly designed and operated IWS can be an effective means of wastewater management; however, IWS that are poorly designed and maintained have been nationally recognized as having high failure rates and have been linked to contaminated groundwater resources. Under Alternative 4, the homeowner will be responsible for operating and maintaining the IWS while the County is responsible for issuing maintenance notifications to the homeowner.

Like the other alternatives, construction activities may temporarily impact water resources. The implementation of BMP measures, including silt fences, filter socks, and sediment traps, would be required during construction to control sediment runoff. Construction trenches would generally not exceed approximately 10 feet below grade. Further, the IWS



designs will need to consider the ground slopes and drainage patterns to ensure that effluent does not seep through these surfaces.

Construction-related impacts on surface water resources under Alternative 4 are expected to be temporarily and localized, with BMPs effectively minimizing potential impacts.

No-Action Alternative:

Under the No-Action Alternative, which involves the continued operation of the existing LCCs without modifications, there would be no construction activities or modifications to existing conditions. Therefore, this alternative would not cause any impacts to geology or surface water resources in the Nā'ālehu area.

It's important to note that the No-Action Alternative does not involve any mitigation measures to address potential stormwater-related risks associated with the existing LCCs.

4.3.2 Groundwater

Groundwater occurs within portions of geologic formations where aquifers receive and store water. Depending on the geologic conditions of the area, many communities and areas on the island rely on groundwater wells to obtain drinking water. To protect the quality of underground source of drinking water from contamination by subsurface disposal of fluids, Hawai'i has adopted the Underground Injection Control (UIC) program administered by the DOH Safe Drinking Water Branch. Hawai'i Revised Statues (HRS) 340 E and Hawai'i Administrative Rules (HAR) 11-23 (Underground Injection Control) set forth the requirements relate to protections of underground sources of drinking water. The proposed WWTP Site is located above the UIC line and, as such, on top of underground sources of drinking water. To avoid impacts to drinking water wells, sewage injection wells cannot be constructed above the UIC line.

Under HAR 11-62, Appendix F, a minimum separation of 1,000 feet from existing wells is required for wastewater treatment sites.

Under the State's Water Resource Protection Plan, aquifers of the island of Hawaii have been classified under an aquifer coding system to identify and describe these aquifers. This system is comprised of Aquifer Sectors, and then Aquifer Systems located within these sectors. The Nā'ālehu community is serviced by the Southeast Mauna Loa Aquifer Sector (Sector 805). Within this sector, there are four aquifer systems which are the 'Ōla'a, Kapapala, Nā'ālehu, and Ka Lae systems. The Project Area is situated within the Nā'ālehu aquifer system (80503) which has a sustainable yield of 118 million gallons per day (mgd).

The State Department of Land and Natural Resources (DNLR), Commission on Water Resource Management (CWRM) has established a groundwater hydraulic unit and coding system for groundwater resource management. The Nā'ālehu Community is situated within the Nā'ālehu Aquifer System (See Figure 4-4). According to the County of Hawai'i Department of Water Supply, the Nā'ālehu community is situated within the Waiohinu-Naalehu Water System. The system primarily depends on the New Mountain House Tunnel Spring and Haao Spring for its supply. The Naalehu Well supplements the tunnel and spring sources during dry weather. The service area is





widespread, covering the communities of Waiohinu, Naalehu and South Point, spanning six service zones through two booster pump stations and nine storage tanks.

Impacts and Mitigation Measures:

Alternative 1 and 2 – Package Plant:

The approximately 10-15 foot trenches required to support the collection system would be relatively shallow in relation to groundwater resources in the Nā'ālehu area. Thus, construction of the collection system would not affect groundwater resources in the area.

The separation between the Nā'ālehu WWTP site and the location of Hawai'i State Water Well, Nā'ālehu Deepwell (Well ID 8-0335-001), indicates that constructions and operation of the treatment and disposal facility would not affect potable groundwater resources in the Nā'ālehu area as the WWTP is located outside of the designated 1000-foot setback area.

While the use of three existing LCCs has not resulted in any documented impacts to groundwater drinking resources, abandonment of the LCCs would remove a potential source of such impacts. Abandonment of the existing wastewater collection system would not affect groundwater within the affected areas.

Alternative 3 and 4 – Individual Wastewater System Program:

Alternatives 3 and 4 entail the implementation of an IWS program which, when properly designed and operated, can be an effective means of wastewater management; however, IWS that are poorly designed and maintained have been nationally recognized as having high failure rates. To ensure that the systems function as intended, the design must take into account a variety of technical considerations including system size, site conditions, location, influent characteristics, and level of treatment.

Should the IWS begin to fail, untreated sewage containing pathogens (e.g. E. coli), nutrients and other harmful substances can be discharged into the groundwater or into nearby surface waters. As such, proper operation of the IWS is necessary to prevent any system failures.

In the event that the County pursues the IWS Alternatives, the implementation of BMP measures during construction would be required at each site during construction, including silt fences, filter socks, and sediment traps to control sediment runoff.

Construction-related impacts on groundwater resources under Alternatives 3 and 4 are anticipated to be temporary and localized, with BMPs effectively minimizing potential impacts.

No-Action Alternative:

The No-Action Alternative has the potential to adversely impact groundwater resources due to the continued operation of the existing LCCs. EPA regulations mandate the closure of LCCs to prevent potential impacts on groundwater resources.



4.4 **Agricultural Lands**

The Land Study Bureau (LSB) at the University of Hawai'i issued L.S. Bulletin No. 6, Detailed Land Classification-Island of Hawai'i in November 1965. The LSB compiled and interpreted data on geology, topography, climate, water resources, soils, and crops from field investigations to create a land classification for the island. Bulletin No. 6 assigned two types of ratings for each land type: the overall or master productivity rating, which reflects degree of overall suitability for agricultural use, ranging from A (Very Good) to E (Very Poor); and selected use ratings, which indicate the degree of suitability for selected use alternatives. Bulletin No. 6 has not been revised or re-issued and remains as the reference document for lands classified by LSB.

In addition to the LSB rating, the State of Hawai'i has developed the Agricultural Lands of Importance to the State of Hawai'i (ALISH) Classification System. This system was developed and compiled in 1977 by the State Department of Agriculture with assistance from the NCRS, U.S. Department of Agriculture (formerly the Soil Conservation Service) and the College of Tropical Agriculture at the University of Hawai'i as part of a national effort to inventory important farmlands. Lands not considered for classification within this system are developed urban lands (over ten acres), natural or artificial bodies of water (over ten acres), public use of lands, forest reserves, lands with slopes in excess of thirty-five percent, and military installations (except undeveloped areas over ten acres). The ALISH Classification identifies the following three categories of land (equivalent NRCS categories in parentheses):

- Prime Agricultural Lands (Prime Farmlands) Land that has the soil quality, growing • season, and moisture supply needed to produce sustained high yields of crops economically when treated and managed according to modern farming methods.
- Unique Agricultural Lands (Unique Farmlands) Land that has a special combination of soil quality, location, growing season, and moisture supply, and is used to produce sustained high-quality yields of a specific crop when treated and managed according to modern farming methods.
- Other Important Agricultural Land (Additional Farmland of Statewide and Local Importance) – Land other than Prime or Unique Agricultural Land that is also of statewide or local importance to agricultural use.

Impacts and Mitigation Measures:

Alternatives 1 and 2 – Package Plant:

The ALISH map shows the proposed WWTP and the collection system are located in "not rated" lands which is assigned to developed communities. Construction of the collection system within the County roads would not affect agricultural lands.

Abandonment of the three LCCs would reduce the potential for contamination of groundwater that is used for irrigation of agricultural lands. Otherwise, abandonment of the LCCs and the existing wastewater collection system would not affect agricultural lands within the affected areas.



Alternatives 3 and 4 – Individual Wastewater System Program:

As indicated by the LSB rating system, the lots identified for the IWS installation are "not rated" which is assigned to developed communities. Installation of the IWS would not affect agricultural lands.

No-Action Alternative

The No-Action Alternative would not impact agricultural lands. Continued operation of the existing LCCs could introduce pathogens and other contaminants to groundwater sources used for irrigation of agricultural lands.

4.5 Natural Hazards

The Disaster Mitigation Act of 2000, (Federal Emergency Management Agency (FEMA), 44 Code of Federal Regulations, Hazards Mitigation Planning required States and Counties to have approved hazard mitigation plans as of November 1, 2004 to receive Pre-Disaster Mitigation funding. The development of State and local hazard mitigation plans is critical for maintaining eligibility for future FEMA mitigation and disaster recovery funding.

Given Hawai'i's vulnerability to natural hazards and history of disasters, the State has maintained and implemented a comprehensive, multi-hazard mitigation strategy to reduce loss of life and property damage. This strategy is embodied in the 2018 State Multi-Hazard Mitigation Plan. This plan identifies the major natural hazards that affect the state's population, property, and infrastructure to the specific hazard, and recommends actions that can be taken to reduce the risk and vulnerability to the hazard. The State Hazard Mitigation Plan also contains a description of programs, policy, statutes, and regulations applicable to hazard mitigation. It should be noted that the 2023 update to this plan has begun and is expected to be released at the end of 2023. Identified major natural hazards that could affect the State, as well as the County are Climate Change Effects (including sea level rise (SLR)/coastal erosion), floods, tsunamis, strong, windstorms/hurricanes, earthquakes, landslides/rockfalls, volcanic activity, and wildfires.

4.5.1 Sea Level Rise

Sea level is rising at increasing rates due to climate change leading to the melting of glaciers and ice sheets. Rising sea level and projections of stronger and more frequent El Niño events and tropical cyclones in water surrounding Hawai'i indicate a growing vulnerability to coastal flooding and erosion. The Hawai'i Sea Level Rise Vulnerability and Adaptation Report (2017) modeled exposure to chronic coastal flooding and erosion using projections from the Intergovernmental Panel on Climate Change (IPCC) 5th Assessment Report (IPCC, 2014) where the high-end scenario was up to 3.2-ft of sea level rise by the end of the century (Courtney et al., 2020).

Impacts and Mitigation Measures:

All Proposed Alternatives:

No short-or long-term impacts are anticipated during the construction or operation of any of the alternatives. The Nā'ālehu community is located approximately 2 miles from the nearest coastline and at elevations from 600 to 800 feet above mean sea level (msl). The community is not anticipated to be impacted by sea level rise under any of the proposed alternatives.



4.5.2 Flood and Tsunami Threat

The Nā'ālehu community is located southeast of the Alapa'i Gulch. The Federal Emergency Management Agency (FEMA) Flood Insurance Rate Map (FIRM), Community Panel No. 155166 1925F, effective date September 28, 2017, shows no special flood hazard areas present in the Nā'ālehu WWTP site and that most of the Nā'ālehu area is located in *Zone X*, which designates areas determined to be outside of the 0.2 percent annual chance (500-year) floodplain. A small portion of the community of Nā'ālehu, including some land within the collection system project site, is located within *Zone X – Other Flood Areas* (See Figure 4-5), indicating areas within the 0.2 percent annual chance (500-year) floodplain, or areas with a 1-percent annual chance of flooding with average flood depths less than 1-foot.

According to the FIRM, all three existing LCCs are also located within Zone X.

Impacts and Mitigation Measures:

All Proposed Alternatives:

Based on the above, no significant impacts on flood hazards are anticipated to occur within the Nā'ālehu region as a result of the alternatives considered.

No-Action Alternative

The No-Action Alternative, specifically the continued operation of LCC 1, could lead to impacts during a flooding event. LCC 1 is located very close to an area mapped as within the 0.2-percent annual change (500-year) floodplain. The existing collection system is substandard and in poor condition. A large flood could potentially cause the collection system and/or LCC to overflow as a result of stormwater inflow and result in an uncontrolled release of raw sewage, thus potentially contaminating flooded areas and creating a public health hazard.

4.5.3 Hurricane and Wind Hazard

Hurricanes seasonally affect the Hawaiian Islands from late summer to early winter months. The State of Hawai'i has not been impacted by a hurricane rated Category 3 or higher since Iniki in 1992. Hurricane strikes are relatively rare with near misses that generate large swell and moderately high winds more common in Hawai'i.

During hurricanes and storm conditions, high winds cause strong uplift forces on structures, particularly on roofs. Wind-driven materials and debris can attain high velocity and cause devastating property damage and harm. Along the coastline, a surge of water, topped by battering of waves can move ashore into low lying coastal areas. Due to differences in atmospheric pressure, tidal stage, coastal topography, and location relative to the eye of the hurricane it is difficult to predict how hurricane-induced storm surge may impact specific location. It is reasonable to assume that future events will occur.

Impacts and Mitigation Measures:

All Proposed Alternatives:

Based on the above, no significant impacts on hurricane and wind hazards are anticipated to occur within the Nā'ālehu region as a result of the alternatives considered.





4.5.4 Seismic Hazard

Earthquakes in the Hawaiian Islands are primarily associated with volcanic eruptions resulting from the inflation or shrinkage of magma reservoirs beneath, which shift segments of the volcano. The Island of Hawai'i experiences thousands of earthquakes each year; however, most are so small that they can only be detected by instruments. Although difficult to predict, an earthquake of sufficient magnitude could cause structural or other damage to public facilities including wastewater treatment facilities. The seismic risk classification of the Island of Hawai'i is Zone 4 (County of Hawai'i, 2007).

Earthquakes may occur before or during an eruption or may result from the underground movement of magma that comes close to the surface. On the Island of Hawai'i, earthquakes directly associated with the movement of magma are concentrated beneath the active Kilauea and Mauna Loa Volcanoes. Typically, the risk of seismic activity and degree of ground movement decreases with the distance from these active volcanoes. A few of the island's earthquakes are less directly related to volcanism. These originate in the zones of structural weakness at the base of the volcanoes or deep within the earth beneath the island.

Several destructive earthquakes have occurred on the Island of Hawai'i. The locations of larger damaging on-island earthquakes since 1868 have generally occurred in the southeast portion of the island near Kilauea, with the most recent destructive earthquake on this south flank occurring on June 26, 1989 with a magnitude of 6.1. More recently, a magnitude 6.9 earthquake occurred on May 4, 2018 offshore and east of Kilauea, though this earthquake was classified as nondestructive.

Impacts and Mitigation Measures:

All Proposed Alternatives

Hawai'i County Code (HCC) § 5A indicates the "International Building Code, 2018 Edition" (IBC) – copyright and published in 2018 by the International Code Council, Incorporated - is adopted by the County. Chapter 5 is the applicable code for the construction of buildings, structures, and facilities in the County. The purpose of the seismic provisions in the IBC is primarily to safeguard against major structural failures and loss of life; limiting damage or maintaining functions is not a primary purpose. At a minimum, structures are to be designed and constructed to resist the effects of ground motions from seismic events. The seismic hazard characteristics described in the IBC are based on the seismic zone and proximity of the site to active seismic sources.

The proposed improvements would be designed and constructed to meet the requirements of the 2018 IBC and HCC Chapter 5 and would comply with seismic loadings established for the County of Hawai'i. This would minimize the potential for an uncontrolled release of untreated or partially treated sanitary wastewater, or diesel fuel from the tank being held for the emergency generator during a seismic event. The County would also develop a facility management plan in accordance with applicable rules and regulations.



No-Action Alternative:

The No-Action Alternative includes no construction or modification to existing conditions, and therefore would not impact seismic hazards in the Nā'ālehu area.

4.5.5 Volcanic Hazards

The U.S. Geologic Survey (USGS) prepared a volcanic hazard zone map for the island of Hawai'i which was last updated in 1997. The map shows lava flow hazard zones for the five on-island volcanoes. The map utilizes a 9-point ranking system which classifies zones ranked from 1 (highest hazard) through 9 (lowest hazard) based on the probability of coverage by lava flows. Nā'ālehu area has been assigned a rating of Zone 6. This designates areas which are less hazardous due to protection by topography.

Impacts and Mitigation Measures:

All Proposed Alternatives

Based on the volcanic hazard map, the potential for damage to buildings and structures is moderate, given the distance between the Nā'ālehu community and active vents and hazards. At this time, the County has no construction restrictions in Zone 6 areas. Therefore, the volcanic hazard designation would not affect the construction and operation of the collection system or the treatment and disposal facilities. Although the potential for volcanic activity in or around Nā'ālehu is present, the likelihood of that impact is relatively small. In the event of a volcanic eruption that threatens the Na alehu area, it is likely that damage would occur in residences, the treatment and disposal facility, the collection system and other assets in the area. There are no mitigation measures to prevent the potential impacts from volcanic activity, and the impacts would be similar regardless of the location of the treatment and disposal facility or treatment system employed.

No-Action Alternative

The No-Action Alternative includes no construction or modification to existing conditions, and therefore would not impact volcanic hazards in the Nā'ālehu area.

4.5.6 Wildfire Hazards

Wildfires can threaten life and property, but they can also harm the environment and threaten important natural resources such as endangered species. While sometimes caused by lightning, nine out of ten wildfires are human-caused. Put simply, "wildfire" is the term applied to any unwanted and unplanned fire burning in forest, shrub or grass regardless of whether it is naturally or human induced (DEM, 2020).

All of the Hawaiian Islands are susceptible to wildfires, especially during prolonged drought and high winds. In recent years, the average annual cost to suppress wildfires in Hawai'i is about \$1,100,000 - making it a Statewide risk (DEM, 2020). The greatest danger of fire is where wildlands border urban areas. Through August 2018, wildfires in Hawai'i have burned 30,000 acres (about double the annual average). Historically, the majority of these fires have been directly caused by humans, either directly or by negligence.



According to the Fire Risk Area Map, the Project Area is located in a medium risk area (See Figure 4-6). As further evidenced by recent events in West Maui, wildfires pose a significant threat to human health and human safety and must be taken very seriously.

Impacts and Mitigation Measures:

Alternatives 1 and 2

The proposed alternatives are not anticipated to have impacts that could result in wildfire. The State Department of Land and Natural Resources-Division of Forestry and Wildfire (DLNR-DOFAW) has adopted a Fire Management Handbook, which specifies its standards for prevention, pre-suppression, and suppression. The document provides a structured approach in providing for public/firefighter safety and minimizing damage to Hawai'i's environment. Additionally, the DLNR-DOFAW is a key agency within the State who can trigger provisions of the Stafford Act (Fire Suppression Assistance), which provides for FEMA funding assistance in situations where forest and grass fires on public or private lands threaten a major disaster to communities and economies.

The package plant and related facilities would be designed according to National Fire Prevention Association (NFPA) 820 "Standard for Fire Protection in Wastewater Treatment and Collection Facilities." In accordance with Hawai'i Fire Department requirements, Fire Department access and water supply to the proposed WWTP site would be designed to comply with Chapter 18 of NFPA 2006 Uniform Fire Code as amended by the County.

Alternatives 3 and 4

The IWS systems do not include construction of facilities which would be susceptible to fire hazards.

No-Action Alternative

The No-Action Alternative includes no construction or modification to existing conditions, and therefore would not impact wildfire hazards in the Nā'ālehu area.

4.6 **Flora and Fauna**

Two biological surveys were conducted on the sites: one in 2012 and one in 2021; both surveys were conducted by AECOS Inc. (AECOS, 2021). During the botanical, avian, and mammalian resources surveys no listed or proposed-for-listing species under federal statutes was recorded. The assessment observed avian diversity and densities present within the Project Area and its surrounding vicinity. A summary of the Natural Resources Assessment (NRA) is outlined below and included herein as Appendix B.

Survey Results

Vegetation

Vegetation across the site consists entirely of grass pasture (predominantly Guinea grass or Megathyrsus maximus) and areas of mostly low scrub growth, dominated by lantana. Trees are scattered around the site, especially along a shallow swale feature that transects the area from north to southeast. The site of the proposed SPS is a former pasture dominated by tall Guinea grass.




The NRA provides a listing of all the species of flowering plants (angiosperms) observed during the surveys with a total of 66 taxa identified from the three survey areas. The vast majority (91%) are naturalized or non-native) species. Three native indigenous plants (4.5%) and one early Polynesian species (3%). The native species are 'ilima (Sida fallax), 'ilie'e (Plumbago zeylanica), and 'uhaloa (Waltheria indica); all common plants. The early Polynesian introduction (so-called "canoe plant") is yellow wood sorrel ('ihi'ai or *Oxalis corniculata*), a very common small weed.

Avian Fauna

A total of 239 individual birds of 15 species, representing 10 separate families, were recorded during point counts. One species recorded, Pacific Golden-Plover (*Pluvialis fulva*), is an indigenous migratory shorebird species. The remaining 13 species recorded are all established alien or feral species. Avian diversity and densities were in keeping with the location of the three sites and the vegetation present. Three introduced species—Common Myna (Acridotheres tristis), House Finch (Haemorhous mexicanus), and Zebra Dove (Geopilia striata)-accounted for 44% of the total number of birds recorded. Common Myna was the most commonly tallied species, which accounted for 23 percent of the birds recorded during point counts.

Mammals

Terrestrian mammalian species detected during the survey include the domestic dog (*Canis lupus* familiaris), small Indian mongoose (Herpestes javanicus), house cat (Felis catus), pig (Sus scrofa), and domestic cattle (Bos taurus). All of which are alien species, introduced to the Hawaiian Islands by humans. The methods in which these species were detected include audio, scat, tracks, and signs.

Impacts and Mitigation Measures:

All Proposed Alternatives:

Based on the findings of the botanical and biological field surveys, construction activities associated with the new collection system and wastewater treatment and disposal facility are not anticipated to result in adverse impacts to botanical and faunal resources in the Nā`ālehu area.

The following minimization measures will be implemented during construction and following build out to ensure that the project minimizes all impacts to listed species to the maximum extent practicable.

Floral Resources

No plants of conservation concern or enjoying statutory protection (that is, listed as threatened or endangered; HDLNR, 1998; USFWS, 2020) were noted in the survey and given the highly disturbed nature of the site, would not be expected to be growing there. For comparative purposes, this survey revealed 66 species with 4 species as either indigenous or of early Polynesian introduction. The previous site located on the makai side of Māmalahoa Highway 1 mi (1.6 km) east (also pastureland) vielded 30 species of introduced plants and 1 species each of indigenous and early Polynesian introduction. A survey of the much smaller Wai'ohinu Transfer Station (AECOS, 2020) in Wai'ohinu, 2 mi



(3.4 km) to the west produced a list of 59 taxa. Although a much smaller area, the latter site is both highly disturbed (increasing the count of ruderal species) and subject to frequent introductions of species on vehicles and rubbish loads.

Faunal Resources

Insects

Several insects are now listed as endangered in the Hawaiian Islands: seven species of the yellow-faced bee (Hylaeus anthracinus, H. assimulans, H. facilis, H. hilaris, H. kuakea, H. longiceps, and H. mana) and Blackburn's sphinx moth (Manduca blackburnii).

No yellow-faced bee species was observed during the survey and no potential habitat or food sources were noted. The caterpillar of the sphinx moth feeds exclusively on plants in the Family Solanaceae. In particular, where the moth is found, caterpillars are most often associated with the widely distributed, non-native tree tobacco plant (*Nicotiana glauca*). The only plant observed representing the Family Solanaceae was a single wild cherry tomato plant (Solanum lycopersicum var. cerasiforme). This species is not known to be utilized by the caterpillar. We would deem any threat to these insect species due to Project activities as non-existent.

Terrestrial Birds

The findings of the avian survey are consistent with the vegetation present on the three sites. The WWTP site is mostly pasture that has been bulldozed in places. The other two smaller sites, located in town, have very limited plant species dominated by alien species. Fourteen of the recorded bird species are alien, naturalized bird species. The only native bird, the Pacific Golden-Plover, is an indigenous migratory shorebird commonly seen across the state between late September and the end of April each year. This species migrates to Arctic breeding grounds in late April or the first week of May.

Seabirds

It is possible that the endangered Hawaiian Petrel (*Puffinus sandwichesis*), Band-rumped Storm-Petrel (Hydrobates castro), and the threatened Newell's Shearwater (Puffinus newelli) over-fly the Project area between April and the middle of December each year in verv small numbers.

The potential impact that the Project poses to protected seabirds is the increased threat that birds will be downed during the seabird fledging season (from September 15 through December 15) after becoming disoriented by lights associated with the Project. The two main types of outdoor lighting posing a threat to nocturnally flying seabirds are: 1) nighttime construction lights; and 2) exterior lighting around Project infrastructure.

Fully shield all outdoor lights so the bulb can only be seen from below bulb height • and only use when necessary; install automatic motion sensor switches and timer



controls on all installed outdoor lights or turn off lights when human activity is not occurring in the lighted area. All external lighting structures should be fully "dark sky compliant" (HDLNR-DOFAW, 2016).

Waterbirds

No habitats suitable for waterbird species are present on any of the three sites. Once the WWTP is constructed it is possible that one or more species will be attracted to ponds associated with the WWTP. No species currently listed or proposed for listing as threatened or endangered under federal or state endangered species statutes is associated with the project site.

It should be mentioned that in the original plans for this project, which USFWS responded to in the TA letter (USFWS, 2019), standard wastewater open ponds were proposed for the new facility. Following public scoping meetings with the public the plan was revised to eliminate open wastewater lagoons. The current plan has the treatment plant being above ground and contained in what is referred to as a "Package Plant". This eliminates the attractive nuisance that open wastewater lagoons pose to endangered water obligate species in the Hawaiian Islands.

Mammalian Resources

Hawaiian Hoary Bat

No Hawaiian hoary bats or were detected during the course of our survey. It is within the realm of possibility that this species uses resources within the Project Area on a seasonal basis. The principal impact that the construction may pose to bats is during the clearing and grubbing phases as vegetation is removed and if proposed security fencing includes use of barbed wire. The removal of vegetation within the project site may temporarily displace individual bats using trees for roosting. As bats use multiple roosts within their home territories, the potential disturbance resulting from the removal of the vegetation is likely to be minimal. During the pupping season, females carrying their pups may be less able to rapidly vacate a roost site as the vegetation is cleared. Additionally, adult female bats sometimes leave their pups in a roost tree while they forage. Very small pups may be unable to flee a tree that is being felled.

- Potential adverse impacts from such disturbance can be avoided or minimized by not clearing woody vegetation taller than 4.6 m (15 ft) between June 1 and September 15, the period in which bats may have pups.
- No barbed wire will be installed on any fence or structures on the site to ensure that Hawaiian hoary bats do not get caught on the barbs.



Critical Habitat

No federally delineated Critical Habitat for any species occurs within the Project area (USFWS, 2024).

Jurisdictional Waters

No streams or wetlands occur in the Project area, although the WWTP site itself is located within a shallow drainage swale and runoff flow will be directed around the WWTP infrastructure in a new channel. Biological observed no indications of water flow in the swale at the Project site. A concrete drainage channel crossed by the proposed force main on Melia Street is similar in directing runoff through town into shallow swale further downslope. Neither of these drainageways direct flows to streams or outlets on the ocean shore. Annual rainfall in the Nā'ālehu area is on the order of 40 in (1020 mm) and the porous geology is not conducive to stream formation.

On May 31, 2024, the DOH initiated consultation with the U.S. Fish and Wildlife Service (FWS) pursuant to Section 7 of the Endangered Species Act (See Appendix C). The consultation requested the FWS' concurrence with the determination that the Proposed Action "may affect, but is not likely to adversely affect" federally listed species. In a letter dated June 21, 2024, the FWS provided concurrence with this determination. Additionally, the FWS provided avoidance and minimizations to be implemented to avoid adverse effects (See Appendix D).

It should be noted that for Alternatives 3 and 4, the area of disturbance for the proposed action will take place on private residential property. However, it is expected that Alternatives 3 and 4 will result in no impacts to botanical and faunal resources.

Generally, minimization measures discussed herein are intended to minimize any potential impacts on flora and fauna that could result from the construction and operations of the wastewater treatment and disposal facility and associated infrastructure. The proactive consultation with relevant authorities ensures compliance with regulations and protection of sensitive species. Additionally, the note regarding Alternatives 3 and 4 acknowledges the unique context of disturbance on private residential property while anticipating minimal impacts.

No-Action Alternative:

The No-Action Alternative involves no modifications to the existing LCC system and is not expected to impact flora or fauna.

4.7 Cultural, Historical, and Archaeological Resources

In May 2024, a Draft Archeological Inventory Survey (AIS) Report was completed by Cultural Surveys Hawai'i. The AIS is was designed to be compliant with both Federal and Hawai'i State environmental and historic preservation review legislation. The AIS was conducted to assess any historic properties for integrity and site significance in accordance with HAR §13-275-6. This report is also intended to support any project-related historic preservation consultation with



consulting parties, such as state and county agencies and interested Native Hawaiian Organizations (NHOs) and community groups, if applicable. A summary of the AIS is outlined below and included herein as Appendix E.

The AIS identified four previously documented and seven newly documented historic properties in the Project Area. The four previously documented sites include the following:

- SIHP # 50-10-74-29507 is a historic ranching complex. Pursuant to HAR §13-275-6 it is assessed as significant under Criterion d for the information it has yielded about historic ranching activity in the project area. It is evaluated as not eligible for listing on the National Register of Historic Places ("National Register").
- SIHP # 50-10-47-30187 comprises the former and present alignments of the Māmalahoa Highway, also known as the Belt Road (variably State Routes 11, 19, 180, and 190). Pursuant to HAR §13-275-6 it is assessed as significant under Criterion a for its association with events that have made an important contribution to the development of transportation routes on Hawaii Island, and under Criterion d for having yielded information about historic transportation in the project area. Pursuant to 36 CFR 60.4 is evaluated as eligible for listing on the National Register under Criterion A (for its association with events that have made a significant contribution to the broad patterns of our history) and Criterion D (for having yielded, or being likely to yield, information important in history).
- SIHP # 50-0-74-30929 is a pre-Contact activity area and historic ranching complex. It is assessed as significant under Criterion d for the information it has yielded about traditional and historic land use in the project area. It is evaluated as not eligible for listing on the National Register.
- SIHP # 50-10-74-30930 is a historic ranch wall. It is assessed as significant under Criterion d for the information it has yielded about historic ranching activity in the project area. It is evaluated as not eligible for listing on the National Register.

The seven newly documented historic properties in the project area include the following:

- SIHP# 50-10-74-31268 is a historic earthen drainage ditch used to channel storm water. It is assessed as significant under Criterion d for the information it has provided about historic water control activity in the project area. It is evaluated as not eligible for listing on the National Register.
- SIHP # 50-10-74-31269 is a historic ranching complex. It is assessed as significant under Criterion d for the information it has yielded about historic ranching activity in the project area. It is evaluated as not eligible for listing on the National Register.
- SIHP # 50-10-74-31270 is a historic jeep road. It is assessed as significant under Criterion d for the information it has yielded about historic transportation in the project area. It is evaluated as not eligible for listing on the National Register.



- SIHP # 50-10-74-31271 is the historic Nāʿālehu Spur Road. It is assessed as significant under Criterion d for the information it has yielded about historic transportation in the project area. It is evaluated as not eligible for listing on the National Register.
- SIHP # 50-10-74-31272 is the historic Ka'alaiki Road. It is assessed as significant under Criterion d for the information it has yielded about historic transportation in the project area. It is evaluated as not eligible for listing on the National Register.
- SIHP # 50-10-74-31273 is a historic complex comprising a concrete drainage channel and associated footbridge. It is assessed as significant under Criterion d for the information it has yielded about historic water control activity in the project area. It is evaluated as not eligible for listing on the National Register.
- SIHP # 50-10-74-31274 is a historic boundary wall. It is assessed as significant under Criterion d for the information it has yielded about historic land use in the project area. It is evaluated as not eligible for listing on the National Register.

As a part of previous environmental documentation efforts, a 2019 EA Pre-Assessment Package was distributed. In response the Pre-Assessment, the Ala Kahakai Trail Association provided further historical and cultural context on the 2013-acre parcel TMK (3) 9-5-007:016 in which the WWTP is proposed to be sited. According to the Ala Kahakai Trail Association (ATA), the parcel, known as the "Waikapuna Property" has exception cultural, historical, environmental, and natural resources. The property contains 2.3 miles of coastline that includes the ancient Alaloa footpath which once encircled the island, also known as the Alanui or Ala Kahakai National Historic Trail. The property is presently used by local fishermen, Native Hawaiian descendants and gatherers of various natural and marine resources for subsistence, recreational, or cultural purposes.

Impacts and Mitigation Measures:

In accordance with federal regulations (36 CFR 800.5), the results of the AIS are intended to support a future determination of "no adverse effect" by the Hawai'i State Department of Health (DOH) on behalf of the Environmental Protection Agency (EPA). Additionally, in accordance with HAR §13-275-7, the Hawai'i State DOH and County of Hawaii Department of Environmental Management (COH-DEM) are expected to determine the project effect as "effect, with proposed mitigation commitments." In a letter dated June 3, 2024, the DOH formally initiated consultation with the SHPD (See Appendix F). Future correspondence with the SHPD shall be included in future EID / EA efforts.

One historic property on the project area has been evaluated as eligible for inclusion on the National Register: SIHP # 50-10-47-30187 (Māmalahoa Highway). The 15-m (50-ft) portion of this historic property within the project area retains integrity of location only, as all the constructed elements of the original Māmalahoa Highway are no longer evident today. While the project will affect the modern roadway elements, these elements will be replaced following construction, and the integrity and significance of the highway will not be diminished. Therefore, the project will have no adverse effect on SIHP # -30187. This site is assessed as significant under Criterion a of the State of Hawai'i significance criteria



for its association with events that have made an important contribution to the development of transportation routes on Hawaii Island, and under Criterion d for having yielded information about historic transportation in the project area. Documentation of SIHP # -30187 in this AIS has included historical research, GPS data collection, photographs, and written description. This AIS has adequately documented the location, extent, function, age, and construction methods of the portion of SIHP # -30187 in the project area, mitigating project-related impacts pursuant to HAR §13-275-8.

The remaining ten historic properties in the project area (SIHP #s 50-10-74-29507, -30929, -30930, -31268, -31269, -31270, -31271, -31272, -31273, and -31274) may also be impacted by project development. These historic properties have been evaluated as not eligible for inclusion on the National Register. These sites are assessed as significant under Criterion d of the State of Hawai'i significance criteria because they have yielded information important for research on history. Documentation of these historic properties in this AIS has included historical research, GPS data collection, photographs, plan view maps, and written descriptions. This AIS has adequately documented the location, extent, function, age, and construction methods of these ten historic properties, mitigating project-related impacts pursuant to HAR §13-284-8.

The single historic property evaluated as eligible for listing on the National Register (SIHP # 50-10-47-30187, Māmalahoa Highway), will not be adversely impacted by the project as its integrity and significance will not be diminished. This AIS has adequately documented the location, extent, function, age, and construction methods of the 11 historic properties within the project area, mitigating any project-related impacts pursuant to HAR §13-284-8. No further work is recommended for these 11 historic properties.

Archaeological monitoring is recommended during all project ground-disturbing activities to facilitate identification for information purposes, especially of any subsurface lava tubes that may contain historic properties. Archaeological monitoring will proceed under an archaeological monitoring plan that meets the requirements of HAR §13-279-4, which will be submitted to the SHPD for review and acceptance prior to project initiation.

Although historic properties have been determined to require no further work, the County will consider further communication and coordination with ATA, on developing feasible and reasonable accommodations on either preserving historic properties or alternatives that integrate the broader preservation goals of the Waikapuna property. Further, for security and safety purposes, the COH-DEM plans to move forward with plans to construct the security fence around the perimeter of the facility.

No Action Alternative

Under the No Action Alternative, existing cultural, historical, and archeological resources in the project area will not be impacted.



4.8 Air Quality and Odors

The project area falls within the purview of ambient air quality standards (AAQS) at both national (NAAQS) and state levels, encompassing the criteria pollutants, including carbon monoxide, nitrogen dioxide, lead, ozone, and particulate matter (PM₁₀ and PM_{2.5}). It is noteworthy that the State of Hawaii has standards that align with the national standards in terms of stringency. The Hawai'i DOH operated an extensive network of air quality monitoring stations across the state. Ensuring that criteria pollutant levels have consistently remained below both federal and state AAOS thresholds throughout the state, include the Nā'ālehu area.

Air quality issues most applicable to the proposed project sites in the Nā'ālehu community concern short-term construction related emissions such as fugitive dust. Vehicular emissions (CO) from traffic occurring along the residential streets are not considered to be an important factor influencing the project areas. This project involves the construction of new underground sewer improvements and a WWTP that will not generate any long-term traffic in the area.

Impacts and Mitigation Measures:

Alternatives 1 and 2 - Package Plant:

During the construction phase of the wastewater collection system and treatment facility associated with Alternative 1, short-term impacts on air quality may manifest as fugitive dust emissions resulting from construction activities. These emissions would be effectively managed through the implementation of a comprehensive dust control plan. Measures within this plan would encompass the application of water to active work areas, the use of wind screens, vigilant maintenance of adjacent roads to prevent dust build-up, and the covering of open-bodied trucks. It is plausible that exhaust emissions from mobile construction equipment, in conjunction with traffic disruptions associated with construction activities, could exert a minor influence on air quality during this phase.

As previously discussed, the PER recommended a granular activated carbon (GAC) scrubber be used at Nā'ālehu WWTP headworks. A GAC scrubber passes odorous air through a bed of activated carbon, which absorbs the odorous constituents within the pore spaces of the carbon. Chemical oxidation or reduction of some compounds can also occur. Also, emissions would occur from trucks used to haul the solids to the County West Hawai'i Sanitary Landfill at Puuanahulu, located north of Kailua Kona. These trips are not expected to exceed federal or state ambient air quality standards for criteria pollutants. The presence of an emergency standby diesel-powered generator, operated periodically for testing and during the power outages, is unlikely to have a significant adverse impact on air quality.

Alternatives 3 and 4 - Individual Wastewater System Program:

Air quality impacts for Alternatives 3 and 4 primarily relate to the installation and maintenance of the IWS by homeowners. These activities may yield minor emissions associated with construction equipment and vehicles. However, any potential air quality impacts would be transient and localized to specific residential areas. Additionally, there is a slight risk of odors emanating from maintenance activities or system breakages within the IWS. Homeowners would be responsible for managing and mitigating these potential odors.



No-Action Alternative:

The No-Action Alternative entails no modifications to the existing LCC system and, therefore, is not projected to introduce new air quality impacts to the Nā'ālehu area. Historical air quality records indicate that the area has consistently met ambient standards during the operation of the existing LCCs.

4.9 Noise

The existing noise environment in the Project Area is generally associated with vehicular traffic and general activities associated with residences such as children playing, radio or television noise, or voices. Most of the proposed collection system routes will occur along residential roadways within the Project Area. Additional sources of noise include some commercial sites and schools. Existing noise is not expected to generate volumes that may exceed State DOH standards. Similarly, the schools and commercial sites should meet applicable noise limits.

Impacts and Mitigation Measures:

Alternatives 1 and 2 – Package Plant:

During the construction of the wastewater treatment facilities and collection systems, it is anticipated that there will be an increase in noise levels associated with the operation of heavy machinery, construction equipment, and from potential increased vehicle traffic. Construction activities can generate temporary noise impacts in the Project Area. Noise from construction activities is regulated under Title 11, Chapter 46 (Community Noise Control) of the State DOH's Administrative Rules. Under these regulations, the Project Area and immediate vicinity fall under the Class A zoning district. The district allows daytime (7 a.m. to 10 p.m.) maximum permissible sound levels up to 55 dBA. Construction activities are not planned at night, so the nighttime noise level restrictions should not be applicable.

To minimize construction-related noise impacts, the project can implement standard noise control measures such as avoiding noisy activities during quiet hours (e.g., evenings and weekends), and employing noise barriers or sound-reducing equipment where feasible. While construction may introduce short-term noise, the continuous operational noise from these facilities is generally localized and can be controlled with noise-reducing measures. Transportation noise may be minimal as wastewater can be treated onsite.

The operational phase of the wastewater treatment facilities may introduce continuous noise sources, such as equipment operation, pumps, and mechanical systems. While these noise sources are generally not excessively loud, they can contribute to ambient noise levels in the immediate vicinity of the facilities. It should also be noted that, in the case of a power outage, emergency generators shall be utilized and may generate additional noise.

To address operational noise, the project can consider noise-reducing designs for the treatment facilities, such as noise barriers or acoustic enclosures for noisy equipment. Additionally, maintenance schedules can be optimized to minimize noisy activities during sensitive times.



Vehicle traffic associated with travel to the Nā'ālehu WWTP by plant operators and with trucks needed to remove solids, based on the location of package plant would not affect noise in the residential areas.

<u> Alternatives 3 and 4 – Individual Wastewater System Program:</u>

During construction of the IWS, it is anticipated that there will be an increase in noise levels associated with the operation of heavy machinery, construction equipment, and due to vehicle traffic. These construction activities can generate temporary noise impacts in the Project Area.

To minimize construction-related noise impacts, the project can implement standard noise control measures such as scheduling construction activities during daytime hours, avoiding noisy activities during quiet hours (e.g., evenings and weekends), and employing noise barriers or sound-reducing equipment where feasible.

The installation and maintenance of the IWS may result in localized noise during construction and maintenance activities. However, these impacts are distributed across multiple resident properties.

No-Action Alternative:

This alternative maintains the existing LCC system. While it avoids construction-related noise, it may not address long-term noise concerns associated with the aging infrastructure.

4.10 Energy and Natural Resources

The Nā'ālehu area relies on a mix of energy sources for electricity, including fossil fuels (e.g., oil) and renewable sources (e.g., solar and wind). The specific energy mix can vary over time and may be influenced by state and county policies promoting renewable energy. Although the integration of renewable energy sources to reduce reliance on fossil fuels and promote sustainability can be considered, WWTP facilities typically rely on a consistent source of power. Electrical infrastructure in the Project Area is managed and maintained by the Hawaiian Electric company (HECO). The existing Punalu'u and South Point Substations and the 12,470-volt overhead distribution system currently serving the Project Area. Currently, the existing Punalu'u and South Point Substations and the 12,470-volt overhead distribution system serve the Project Area. It is anticipated that the capacity of the existing substation may be adequate to serve the anticipated load.

Impacts and Mitigation Measures:

All Proposed Alternatives:

The construction and operation of wastewater treatment facilities will require energy inputs. Construction equipment, pumps, aeration systems, and other mechanical components consume energy during installation and operation. Implementing energyefficient technologies and practices during facility construction and operation can help reduce energy consumption. Additionally, the proposed gravity flow system is intended to be require lower energy demands compared to centralized facilities.



Additionally, the feasibility of incorporating renewable energy systems into the wastewater treatment facilities to reduce carbon emissions and energy costs should be evaluated.

During the design phase, the County shall coordinate with HECO to provide detailed loading and civil plans to coordinate electrical power to the Project Site.

4.11 Land Use and Land Use Plans

The existing land use in the Project Area includes residential, agricultural, and undeveloped land. The Project Area is located within the Urban and Agricultural State Land Use Districts (SLUD). The WWTP site is entirely located within the Agricultural SLUD. As such, the Proposed Action would be required to comply with the regulations set forth in the State Land Use Law (HRS, Chapter 205).

Furthermore, the County of Hawai'i General Plan calls for the preparation of community development plans (CDPs) "to translate the broad General Plan statement to specific actions as they apply to specific geographical areas."

The Ka'ū CDP is one of nine CDPs for Hawai'i County. On October 17, 2017, the Ka'ū CDP was adopted as Ordinance No. 2017-66. The purpose of the CDPs is to implement the broad goals within the General Plan on a regional basis and to translate the broad General Plan statements into specific actions. CDPs are the forum for community input into managing growth and coordinating the delivery of government services to the community. CDPs designate detailed development patterns and direct physical development and public improvements by detailing land use policies and infrastructure priorities.

Section 5 of the CDP prioritizes improvements in the infrastructure, facilities, and services, including Section 5.8 applicable to Environmental Management which states:

"Policy 120 Extend the primary wastewater collection lines in Pāhala and Nā'ālehu so that infill development projects can connect wastewater systems built for new subdivisions to the County systems."

The collection system will be consistent with Policy 120 as the improvements for the Nā'ālehu LCC Replacement Project have been designed not to preclude expansion to accommodate the future needs of the Nā'ālehu community. Future subdivisions would be accommodated, as capacity allows, on a first-come, first-served basis.

Impacts and Mitigation Measures:

Alternative 1 and 2 – Package Plant:

Construction activities will result in land disturbance, affecting natural habitats and agricultural areas. Minimizing the affected area will act to footprint of construction activities and implementing erosion control measures can help mitigate land disturbance impacts.



Although the package plant in Alternative 2 would allow additional facilities, use of the existing collection system would not allow flows from the residential areas to the north and east to connect to the package plant and treatment disposal system.

As previously noted, the WWTP will be configured within a 28.74-acre portion of the subject TMK parcel. Following the final design of the WWTP, lands not utilized for the WWTP shall be converted back to conservation land.

<u>Alternatives 3 and 4 – Individual Wastewater System Program:</u>

Construction activities can result in land disturbance, affecting natural habitats and will affect the residential parcels, including affecting existing buildings, structures, and landscaping. Minimizing the footprint of construction activities and implementing erosion control measures can help mitigate land disturbance impacts.

These alternatives are designed to use gravity flows which will mean lower energy demands compared to centralized facilities.

No-Action Alternative:

The existing LCC system will not require energy consumption. No significant changes in energy use are anticipated with this alternative.

4.12 Roadways and Traffic

The existing roadway and traffic conditions in the Na'alehu area provide essential context for assessing the impacts and mitigation measures associated with the proposed wastewater treatment project. Key considerations include:

Road Network: Nā'ālehu is served by a network of roadways which are under the jurisdiction of the County and include Māmalahoa Highway and local roads in Nā'ālehu is generally low, reflecting the rural nature of the area. Limited vehicular traffic contributes to the low levels of congestion and a relatively peaceful road environment.

Access to Project Sites: The proposed project sites, including the preferred location for the wastewater treatment and disposal facility, are accessible via the existing road network. Consideration of the impact of construction and operational traffic on local roads is necessary. Safety: Road safety is a critical concern in the area. Ensuring the safety of residents, commuters, and workers during construction and operation is a primary focus. Safety measures may include signage, flaggers, and traffic control measures as needed.

Impacts and Mitigation Measures:

Alternative 1 and 2 – Package Plant:

During the construction phase of the wastewater treatment project, temporary disruptions to traffic flow on local roads may occur. Mitigation measures include the development of traffic management plans to minimize construction-related traffic impacts. These plans may include designated construction-related traffic impacts. These plans may include designated construction access points, scheduling work during off-peak hours and flaggers to ensure safe traffic flow.



The influx of construction vehicles, equipment, and workers to the Project Area may result in increased traffic volume on local roads. To mitigate this, construction logistics planning should aim to minimize the impact on existing road users and ensure the safety of all road users.

Safety measures, such as signage, temporary speed limits, and traffic control personnel, will be employed as necessary during construction activities to maintain the safety of both workers and the local community.

As part of the project, any necessary upgrades or improvements to local roads or intersections impacted by construction activities will be considered and implemented. This may include road repairs, resurfacing, or other enhancements to ensure the continued integrity of roadways.

During the operation of the wastewater treatment facility, regular visits by facility operators are expected. While these visits would introduce minimal traffic, safety remains a priority, and any potential traffic impacts will be mitigated through adherence to established safety protocols.

Continuous monitoring of traffic conditions and adherence to traffic management plans will be essential to address any unforeseen issues promptly. Compliance with local traffic regulations and safety standards will be enforced throughout the project's lifecycle.

In summary, the existing road network in Nā'ālehu serves as the backdrop for assessing potential impacts and mitigation measures associated with the proposed wastewater treatment project. During both the construction and operation phases, careful planning, safety measures, and infrastructure improvements will be implemented to minimize disruptions and maintain the safety and functionality of local roadways.

Additionally, the new collection system or methods of integration with the existing collection system may require careful planning and engineering to ensure compatibility and minimize impacts on the existing infrastructure.

Alternatives 3 and 4 – Individual Wastewater System Program:

Unlike the centralized package plant options, this alternative involves the installation and maintenance of the IWS at each residence within the service area. Logistically, this can be challenging due to the need for coordination with numerous property owners.

Obtaining access to private properties and ensuring compliance with installation and maintenance requirements for the IWS may pose logistical challenges. Coordinating schedules and ensuring proper installation and maintenance become complex tasks; however, it should be noted that septage trucks would only need to visit each property every three to five years to pump the septic tanks.



The ongoing operation and management of multiple individual systems can be logistically complex. Ensuring that all systems meet required standards and addressing any issues promptly is a continuous endeavor.

No-Action Alternative:

Maintenance of Existing Infrastructure: Under the No-Action Alternative, there would be no changes to the existing infrastructure. While this avoids the logistical challenges of the new construction, it does not address potential issues such as road closures or construction traffic which may be associated with maintenance and repairs of the existing and aging LCC infrastructure.

Long-Term Considerations: Continuing with the existing system may provide short-term stability, but it may not be a sustainable long-term solution for wastewater treatment in the area.

4.13 Hazardous Materials

The existing conditions in Nā`ālehu regarding hazardous materials primarily pertain to the operation of the Nā`ālehu LCCs, which historically managed wastewater treatment for the community. No chemicals are currently being used for treatment at the Nā`ālehu LCCs, however, wastewater treatment processes generate residual waste, including sludge and biosolids. Proper management and disposal of these materials are essential to prevent environmental contamination.

Impacts and Mitigation Measures:

All Proposed Alternatives:

The impacts and mitigation measures related to hazardous materials are primarily associated with the closure of the Nā'ālehu LCCs and the transition to alternative wastewater treatment methods:

- As the LCCs cease operation, residual waste, such as sludge and biosolids, must be managed appropriately. Mitigation involves planning for the safe removal and disposal of these materials to prevent adverse environmental impacts.
- Depending on the condition of the LCC sies, remediation efforts may be required to address any soil or groundwater contamination resulting from historical operations. Remediation plans and measures will ensure that the sites are restored to acceptable environmental standards.
- For proposed alternative wastewater treatment methods, any hazardous materials or chemicals used in the new treatment processes would be subject to stringent safety protocols, handling procedures, and storage requirements. Mitigation measures include compliance with safety regulations and ongoing staff training.

Continuous monitoring of the environment by the homeowners, including soil and water quality, will be essential to detect any potential impacts related to hazardous materials. Mitigation measures involve the implementation of robust monitoring programs to promptly address any issues that may arise.



The closure of the Na'alehu LCCs and the transition to alternative wastewater treatment methods in Nā'ālehu require careful management of hazardous materials and chemicals to prevent adverse environmental impacts. Proper disposal, remediation, and adherence to safety protocols are crucial mitigation measures to ensure the safe and responsible handling of hazardous materials throughout the project's lifecycle.

No-Action Alternative"

The No-Action Alternative does not involve any new construction or modification of the existing sewage system; however, this would not allow the County to meet the requirements of the AOC and SDWA.

4.14 Socioeconomics & Environmental Justice

The U.S. Census Bureau provides the American Community Survey (ACS) for Census Designated Places, which updates selected demographic, social, and economic, information for various years. The ACS shows age distribution, racial composition, and economic information, including employment and household income by Census Designated Place for various locations in Hawai'i County. The version of the ACS referenced is the 2021 5-Year Estimates, released in December 2022.

The population for the Na'ālehu census designated place (CDP), which includes the project service area, was 890 persons based upon the ACS 2022 5-year data Census data. There were 264 households and the median age of residents was 29.9 years (US Census Bureau 2022). Table 4-1 shows this CDP census information in comparison to both the County and State.

Overall, Na'alehu is characterized by a racial composition that includes a greater proportion of minorities compared to the County. The Nā'ālehu racial distribution includes a much lower proportion of White residents, a much higher proportion of Asian and Hispanic residents, and lower populations of other minority groups, including Native Hawaiians when compared to the County. There are also more residents of two or more races in Nā'ālehu than in the County.

Navalehu has a higher proportion of residents that have completed high school and some college than the County overall, but a lower proportion with college degrees (bachelor's and graduate or professional degrees). From an economic perspective, Nā'ālehu generally has a lower median household income than the County.

Lastly, Nā'ālehu had a higher proportion of employment in agriculture, forestry, fishing, hunting, and construction (6.2 percent), and retail trade (6.6 percent) compared to the County 1.7 percent and 5.0 percent respectively. Nā'ālehu had a lower proportion in education and health care (3.4 percent), compared to the County (9.6 percent).



Table 4-1 Demographic, Economic and Social Characteris	tics of Nā	`ālehu an	d Hawai`	i County
	Nā`ālehu	I	Hawai`i G	County
Item	Total	Percent	Total	Percent
Demographic Characteristics				
Total Population	890		206,315	
Under 5 to 19 years	324	36.5	46, 627	22.6
20 to 39 years	169	19.0	47, 039	22.8
40 to 59 years	220	24.8	49, 515	24.0
60 to 79 years	136	15.3	54, 054	26.2
80 years and older	40	4.5	9, 077	4.4
Median age	29.9		44.1	
Race				
White	39	4.4	61, 069	29.6
African American (inc. American Indian/Alaska Native)	44	0.5	1, 444	0.7
Asian	244	27.6	35, 279	17.1
Other Pacific Islander	53	6.0	23, 313	11.3
Two or more	267	30.0	54, 879	26.6
Hispanic	281	31.6	28, 677	13.9
Other	0	0.0	1, 237	0.6
Native Hawaiian	0	0.0	412	0.2
Social/Educational Characteristics				
Less than 9th grade	93	10.5	15, 267	7.4
High School to High School Graduate	391	44.0	59, 625	28.9
Some college to associate degrees	329	37.0	65, 195	31.6
Bachelor degree	42	4.8	38, 580	18.7
Graduate/professional degree	33	3.8	70, 147	13.4
Household Income Characteristics				
Less than \$24,999	104	11.7	27, 027	13.1
\$25,000 to \$49,999	121	13.7	23, 726	11.5
\$50,000 to \$99,999	240	27.1	55, 086	26.7
\$100,000 to \$199,999	281	31.6	64, 163	31.1
\$200,000 or more	141	15.9	31, 772	15.4
Median household income	\$54,688		\$72,568	
Employment Characteristics				
Agriculture, forestry, fishing and mining	55	6.2	3, 587	1.7
Construction	17	1.9	7, 344	3.5
Manufacturing and wholesale trade	11	1.2	4, 094	2.0
Retail trade	59	6.6	10, 457	5.0
Transportation, warehousing, and utilities	0	0.0	3, 960	1.9
Information tech, finance, insurance and real estate	3	0.3	6, 312	3.0
Professional, scientific and technical services	28	3.1	10, 306	5.0
Education and health care	30	3.4	19, 938	9.6
Arts, entertainment and recreation	47	5.3	14, 398	7.0
Other services, public administration	18	2.0	3, 894	1.9



A subset of social resources is environmental justice. Environmental justice considers sensitive populations, such as children, minorities, and low-income communities. Sensitive populations are identified in two Executive Orders (EOs):

- EO 12898, *Federal Actions to Address Environmental Justice in Minority and Low-Income Populations*, serves to avoid the disproportionate placement of adverse environmental, economic, social, or health impacts from federal actions and policies on minority and lowincome populations.
- EO 13045, *Protection of Children from Environmental Health Risks and Safety Risks*, states that federal agencies will identify and address environmental health and safety risks from their activities, policies, or programs that may disproportionately affect children.

As noted above, Nā'ālehu has a higher proportion of minority and children residents as compared to the County as a whole. For purposes of this assessment, and to correspond with the available ACS demographic characteristic data, "low income" is defined as having a household income of less than \$24,999; "minority" is defined as any race population other than White; and "children" is defined as the "Under 5 to 19" age category.

Impacts and Mitigation Measures:

<u> Alternatives 1 and 2 – Package Plant:</u>

In the short-term, construction projects would require a number of contractors and their subcontractors. Construction contract documents would reference HRS 103B, which requires the contractor (including subcontractors) to include not less than 80 percent Hawai'i residents in the work force. This would limit the importation of workers from outside the local area and the associated increase in demand for local housing.

The Alternative 1 would generate employment as the contractor would need workers to undertake construction of the improvements for the wastewater collection system and the WWTP and effluent disposal facility. This employment would generate wages and salaries paid to the contractor and subcontractor work forces. The wages and salaries paid to the work force would in turn generate purchases of goods and services, which would result in taxes paid to the State of Hawai'i. In addition, the contractor and their subcontractors would need to purchase equipment, supplies, and materials, some of which would be purchased from local suppliers and vendors. Direct purchases of equipment, supplies, and materials by the contractor would also generate taxes. Overall, Alternative 1 would result in positive employment benefits which would result in higher levels of income and overall economic benefits to the local economy.

The Alternative 1 is not likely to directly impact long-term employment or education trends since the wastewater plant operator would likely be based in Hilo or Kona, meaning the project would not involve long-term relocation of any staff to Nā`ālehu. Instead, the project is anticipated to create additional job opportunities for local community members. Additionally, Alternative 1 wastewater collection system and treatment and disposal facility would not be designed to encourage or accommodate substantial population growth in Nā`ālehu.



Alternative 3 and 4 – Individual Wastewater System Program:

Construction of the IWS would also create the need for workers to construct the septic system including the absorption bed or seepage pits. Since the IWS would not involve the complex treatment processes, the level of employment could be lower than Alternative 1 or 2. However, to construct over 100 IWS would require multiple crews or contractors. Overall, the economic impact may be similar to Alternatives 1 and 2.

No-Action Alternative:

The No-Action Alternative does not involve any new construction or modification of the existing sewage system, and therefore, it is not expected to impact socioeconomic or demographic conditions in the Nā'ālehu area.

No specific mitigation measures related to socioeconomic characteristics are required for this alternative.

These adapted impacts and mitigation measures address the potential socioeconomic impacts for all five alternatives, emphasizing employment and economic benefits while considering sensitive populations.

All Proposed Alternatives:

Despite the relatively high proportions of minority and children residents in Nā'ālehu compared to the County overall, the proposed alternatives would not result in disproportionately high and adverse human health or environmental effects on the minority or low income populations. The design and location of the proposed wastewater and disposal facility would minimize odor and air quality impacts. Construction of the wastewater collection system would result in intermittent and unavoidable noise from construction vehicles and equipment within the Nā'ālehu community, including noise associated with the removal of bedrock. However, as discussed in Section 3.18.2, construction activities within the community would comply with provisions of HAR 11-46 (Community Noise Control). This includes obtaining a noise permit for any activities that would generate noise exceeding the permissible sound levels from excessive noise sources; and would require the applicant to notify affected members of the public in advance of any planned nighttime construction activity (which must not exceed the permissible sound levels). Overall, the proposed alternatives are expected to result in positive human health and environmental effects to Na alehu residents by providing a cleaner and longer-lasting wastewater treatment system.

4.15 Sustainability

The concept of sustainability is vital in understanding the environmental, economic, and social conditions in Na'alehu and its surrounding areas. Sustainability encompasses the balance between meeting the needs of the present without compromising the ability of the future generations to meet their own needs. In the context of Nā'ālehu, several factors influence the existing conditions related to sustainability:

Nā'ālehu 's environmental sustainability is influenced by its unique natural surroundings, including its proximity to the Kilauea Volcano and the potential impact of volcanic activity on air and water



quality. The region's lush vegetation and agricultural activities contribute to its environmental diversity.

The local economy in Nā'ālehu is characterized by farming, fishing and forestry, a significant contributor to the region's economic sustainability. Additionally, employment opportunities in construction, education and healthcare play a role in the local economy.

Nā'ālehu's community consists of residents with diverse racial backgrounds, therefore factors such as education levels, access to healthcare services, and community engagement are influential in the region's social sustainability.

Impacts and Mitigation Measures:

Alternatives 1 and 2 - Package Plant:

The construction of a new wastewater collection system and treatment facility may have short-term environmental impacts, such as soil disturbance and potential disruption of local ecosystems. Mitigation measures include adherence to environmental regulations and best construction practices to minimize ecological disturbances.

The construction phase is expected to generate employment opportunities, contributing positively to economic sustainability in the short term. Long-term economic impacts include the potential for increased economic activity due to improved wastewater infrastructure.

Minimal direct social impacts are expected during construction. The long-term social benefits include improved access to wastewater services, contributing to the overall quality of life and social sustainability.

Alternatives 3 and 4 - Individual Wastewater System:

The installation and maintenance of the IWS may have minimal environmental impacts, mainly related to construction activities. Mitigation measures involve adherence to environmental regulations during installation.

This alternative provides opportunities for local residents to participate in the installation and maintenance of IWS, potentially benefiting economic sustainability at the community level.

The social impacts are localized, involving homeowners and their immediate surroundings. The long-term social benefit includes improved wastewater systems for individual households, enhancing overall social sustainability.

No-Action Alternative:

This alternative maintains the existing conditions, resulting in minimal changes to the environment.

The economic impacts of the No-Action Alternative are limited, as it does not involve new construction or economic development related to wastewater infrastructure.



This alternative does not introduce significant social changes, as it maintains the status quo in terms of wastewater services and community conditions.

4.16 Human Health and Safety

In Nā'ālehu, access to adequate wastewater treatment services is a fundamental necessity for maintaining public health and environmental quality. Existing conditions reseal the following:

- Proper wastewater treatment is vital for safeguarding public health. Inadequate treatment can result in the contamination of groundwater and surface water, posing risks to drinking water sources and recreational areas. It can also lead to the spread of waterborne diseases.
- Effective wastewater treatment is essential for preserving the local environment. Uncontrolled discharge of untreated sewage can harm aquatic ecosystems, damage coral reefs, and degrade coastal waters, impacting the region's biodiversity and natural beauty.
- Reliable wastewater treatment infrastructure contributes to the overall well-being of the community. It ensures that residents can enjoy a safe and healthy environment and minimizes nuisances such as foul odors and unsightly conditions associated with inadequate treatment.

The Closure of the LCCs and Improved Services:

The closure of the LCCs and the proposed wastewater treatment alternatives aim to address existing challenges and improve wastewater treatment services in Nā'ālehu:

- The closure of the LCCs signifies a shift towards more modern and effective wastewater treatment methods. The proposed alternatives include advanced treatment processes that can remove a higher percentage of contaminants from wastewater, resulting in cleaner effluent.
- Improved wastewater treatment aligns with the community's commitment to environmental stewardship. By ensuring that treated effluent meets stringent quality standards, the closure of the LCC supports the conservation of local ecosystems and marine life.
- The closure of the LCCs and the adoption of alternative wastewater treatment methods provide significant public health benefits. Treated wastewater reduces the risk of waterborne diseases and protects the health of residents and visitors.
- Adequate wastewater treatment services contribute to community satisfaction and quality of life. Residents can enjoy a cleaner and healthier environment, and in turn, can have positive social and economic impacts on the community.

The closure of the LCCs represents a critical step toward enhancing wastewater treatment services in Nā'ālehu. It reflects the community's commitment to protecting public health, conserving the environment, and improving overall community well-being. The proposed alternatives offer modern and effective solutions to ensure that wastewater is treated to the highest standards, addressing the pressing need for adequate wastewater treatment in the area.



Impacts and Mitigation Measures:

All Proposed Alternatives:

The Nāʿālehu LCC Closure project would not result in construction of new above-ground infrastructure within the 500-year floodplain. Although a small portion of the proposed collection system is located within the 500-year floodplain, the associated trenching operations would be temporary and would not alter the 500-year floodplain. Thus, no impacts to the existing floodplain are expected from the Proposed Action. For information related to stormwater management and impacts, please refer to Section 3.23.

Abandonment of the two LCCs and the existing wastewater collection system would not affect floodplains within the affected areas.

No-Action Alternative

The No-Action Alternative, specifically the continued operation of LCC 1, could lead to impacts during a flooding event. LCC 1 is located very close to an area mapped as within the 0.2-percent annual change (500-year) floodplain. The existing collection system is substandard and in poor condition. A large flood could potentially cause the collection system and/or LCC to overflow as a result of stormwater inflow and result in an uncontrolled release of raw sewage, thus potentially contaminating flooded areas and creating a public health hazard.

4.17 Unresolved Issues

The EID serves to assess the anticipated environmental impacts of each alternative on various environmental resources; however, there are still several unresolved issues that may affect the completion of the project under any of the Proposed Alternatives.

Construction Feasibility

Lots to be served in the Nā'ālehu community vary in size from 0.12 to 1.94 acres with a median size of 0.16 acres. Space available for IWS installation on these properties is further limited by the presence of existing structures. For properties that could accommodate a septic tank, the soils may have percolation rates that are too slow to allow for seepage pits based on HAR 11-62-34 regulations. Residents with insufficient space for a seepage pit may need to import fill soil to create elevated mound systems or convert to household aerobic treatment units. As a solution, seepage pits can be installed in lieu of absorption beds in space constrained lots, and IWS installations could be installed encroaching into County Right-of-Way, if permitted by Hawai'i County.

Additionally, as discussed throughout the EID, the IWS recommended by the PER also requires the installation of a septic tank with an absorption bed. The PER further outlines that this absorption shall not be installed on lands with a slope gradient of greater than 8 percent. Sites in Nā'ālehu have slopes that vary from 6-10%. This is likely to affect the constructability of absorption beds as a method of wastewater disposal. Per HAR 11-62-34, absorption beds shall not be installed on land with a slope gradient greater than 8%, while absorption trenches are permitted on a slope of up to 12%. In case this slope requirement cannot be fulfilled, the DOH would allow a seepage pit to be installed instead of an absorption bed.



Access to Properties

Under Alternative 3, the County will fund, design, and manage the project. Obtaining Right of Entry (ROE) to private properties for various purposes, such as infrastructure development or land surveys, can present several challenging issues. The process often involves negotiation, legal considerations, and respect for property rights. Failing to clarify these issues can lead to legal disputes and project delays. The most straightforward way to gain ROE is through the voluntary consent and permission of property owners. However, some property owners may be unwilling to grant access due to concerns about privacy, property damage, or other reasons.



(This page intentionally left blank)



CHAPTER 5: RELATIONSHIP TO PLANS, POLICIES, AND CONTROLS

5.1 State of Hawai'i Land Use Plans and Policies 5.1.1 Hawai'i State Plan

The Hawai'i State Plan, Chapter 226, Hawai'i Revised Statutes (HRS), as amended, provides goals, objectives, policies, and priorities for the State. The purpose of the Hawai'i State Plan is to set forth a plan that shall serve as a guide for the future long-range development of the State; identify the goals, objectives, policies, and priorities for the State; provide a basis for determining priorities and allocating limited resources, such as public funds, services, human resources, land, energy, water, and other resources; improve coordination of federal, state, and county plans, policies, programs, projects, and regulatory activities; and to establish a system for plan formulation and program coordination to provide for an integration of all major state, and county activities. The State Plan is divided into three sections. Part 1 is Overall Theme, Goals, Objectives and Policies. Part 2 is Planning Coordination and Implementation. Part 3 is Priority Guidelines. The Proposed Action's consistency with applicable goals, objectives and policies of Part 1 is discussed in Table 5-1, and an assessment of conformance with Part 3 is discussed in Table 5-2. Part 2 of the State Plan, which primarily covers internal government affairs, is not related to the Proposed Action.

To facilitate describing the relationships of the Proposed Action to the numerous land use and natural or cultural resource plans, policies, and controls for the affected area, some of those plans, policies, and controls are presented in tabular form, and are described with text and/or the following letter code:

Table 5-1: The Hawai'i State Plan	S	NS	N/A
§226-4 State goals. In order to ensure, for present and future generations, those elements of choice and mobility that ensure that individuals and groups may approach their desired levels of self-reliance and self-determination, it shall be the goal of the State to achieve:			
(1) A strong, viable economy, characterized by stability, diversity, and growth, that enables the fulfillment of the needs and expectations of Hawai'i's present and future generations.	x		
(2) A desired physical environment, characterized by beauty, cleanliness, quiet, stable natural systems, and uniqueness, that enhances the mental and physical well- being of the people.	x		
(3) Physical, social, and economic well-being, for individuals and families in Hawai'i, that nourishes a sense of community responsibility, of caring, and of participation in community life.	x		
Discussion: The Proposed Action will support the State's goals, for present and future generations, to ensure individuals and groups may approach their desired levels of self-reliance and self-determination. The Proposed Action involved the rehabilitation and replacement of existing plant facilities and construction of new facilities will ensure wastewater treatment efficiency for present and future generations.			

S = Supportive, NS = Not Supportive, N/A = Not Applicable



	Table 5-1: The Hawaili State Blan	6	NC	
A 11	Table 5-1: The Hawai I State Plan	5	CN1	N/A
A well-mainta and support t improved trea natural syster will also indire and protecting	the overall economic stability and growth of the region. Moreover, the Propose the overall economic stability and growth of the region. Moreover, the Propose the processes, reducing the impact on the surrounding environment and enhance ns. While the primary focus of the Proposed Action is on environmental and pul ectly contribute to the well-being of communities by ensuring the availability of c g public health.	reduce ed Acti ancing olic he lean w	on will the stat alth asp vater res	ead to bility of ects, it ources
§226-5 Obje	ectives and policies for population.			
(a) It sh cons	all be the objective in planning for the State's population to guide population gro istent with the achievement of physical, economic, and social objectives contain	owth t ed in t	o be his char	oter.
To achieve the	e population objective, it shall be the policy of this State to:			
(1)	Manage population growth statewide in a manner that provides increased opportunities for Hawai'i's people to pursue their physical, social, and economic aspirations while recognizing the unique needs of each county.			x
(2)	Encourage an increase in economic activities and employment opportunities on the Neighbor Islands consistent with community needs and desires.			X
(3)	Promote increased opportunities for Hawai'i's people to pursue their socio- economic aspirations throughout the islands.			X
(4)	Encourage research activities and public awareness programs to foster an understanding of Hawai'i's limited capacity to accommodate population needs and to address concerns resulting from an increase in Hawai'i's population.	x		
(5)	Encourage federal actions that will promote a more balanced distribution of immigrants among the states, provided that such actions do not prevent the reunion of immediate family members.			x
(6)	Pursue an increase in federal assistance for states with a greater proportion of foreign immigrants relative to their state's population.			X
(7)	Plan the development and availability of land and water resources in a coordinated manner so as to provide for the desired levels of growth in each geographic area.	x		
Discussion: Proposed Acti generate emp providing indi	The Proposed Action will support the objectives and policies of the State for pop on will not significantly contribute to the County's population growth. The Propo loyment opportunities both in the short-term construction phase and the long-te viduals with a means to pursue their socio-economic aspirations.	oulation sed Ad erm op	n. The ction wil peration,	 ,
§226-6 Obje	ectives and policies for the economy— in general.			
(a) Planı obje	ning for the State's economy in general shall be directed toward achievement of ctives:	the fo	ollowing	
(1)	Increased and diversified employment opportunities to achieve full employment and job choice, and improved living standards for Hawai'i's people.	, incre	eased in	come
(2)	A steady growing and diversified economic base that is not overly dependent or and includes the development and expansion of industries on the neighbor islar	n a fev nds.	v indust	ries,

(b) To achieve the general economic objectives, it shall be the policy of this State to:



Table 5-1: The Hawai`i State Plan	S	NS	N/A
 Promote and encourage entrepreneurship within Hawai'i by residents and nonresidents of the State. 			x
(2) Expand Hawai'i's national and international marketing, communication, and organizational ties, to increase the State's capacity to adjust to and capitalize upon economic changes and opportunities occurring outside the State.			x
(3) Promote Hawai'i as an attractive market for environmentally and socially sound investment activities that benefit Hawai'i's people.			x
(4) Transform and maintain Hawai'i as a place that welcomes and facilitates innovative activity that may lead to commercial opportunities.			x
(5) Promote innovative activity that may pose initial risks, but ultimately contribute to the economy of Hawaii.			x
(6) Seek broader outlets for new or expanded Hawai'i business investments.			X
(7) Expand existing markets and penetrate new markets for Hawai'i's products and services.			x
(8) Assure that the basic economic needs of Hawai'i's people are maintained in the event of disruptions in overseas transportation.			x
(9) Strive to achieve a level of construction activity responsive to, and consistent with, state growth objectives.	x		
(10) Encourage the formation of cooperatives and other favorable marketing arrangements at the local or regional level to assist Hawai'i's small scale producers, manufacturers, and distributors.			x
(11) Encourage labor-intensive activities that are economically satisfying and which offer opportunities for upward mobility.	x		
(12) Encourage innovative activities that may not be labor-intensive, but may otherwise contribute to the economy of Hawaii.			x
(13) Foster greater cooperation and coordination between the public and private sectors in developing Hawai'i's employment and economic growth opportunities.			x
(14) Stimulate the development and expansion of economic activities which will benefit areas with substantial or expected employment problems.			x
(15) Maintain acceptable working conditions and standards for Hawai'i's workers.	X		
(16) Provide equal employment opportunities for all segments of Hawai'i's population through affirmative action and non-discrimination measures.			x
(17) Stimulate the development and expansion of economic activities capitalizing on defense, dual-use, and science and technology assets, particularly on the neighbor islands where employment opportunities may be limited.			x
(18) Encourage businesses that have favorable financial multiplier effects within Hawai'i's economy, particularly with respect to emerging industries in science and technology.			x
(19) Promote and protect intangible resources in Hawai'i, such as scenic beauty and the aloha spirit, which are vital to a healthy economy.			x



Table 5-1: The Hawai'i State Plan	S	NS	N/A
(20) Increase effective communication between the educational community and the private sector to develop relevant curricula and training programs to meet future employment needs in general, and requirements of new, potential growth industries in particular.			x
(21) Foster a business climate in Hawai'i- including attitudes, tax and regulatory policies, and financial and technical assistance programs-that is conducive to the expansion of existing enterprises and the creation and attraction of new business and industry.			x
Discussion: The Proposed Action will support the objectives and policies of the State for the general.	econo	omy – in	
The Proposed Action will contribute to the local economy on Hawai'i Island, directly and i construction period. The construction of the Proposed Action will create expenditures, a por used towards the purchase of material from local suppliers. The employment of a local work use of income for local retail businesses. Furthermore, implementation of the Proposed Action w work experience to help build the local skilled labor workforce. The Proposed Action w acceptable working conditions and standards by adhering to relevant labor laws, regulation practices in terms of worker safety.	ndirect tion of force vill pro vill ma is, and	tly, duri f which will ena vide neo intain/ir l indust	ng the will be ble the cessary nprove ry best
§226-7 Objectives and policies for the economy— agriculture.			
(a) Planning for the State's economy with regard to agriculture shall be directed towards following objectives:	achie	evement	of the
 Viability of Hawaii's sugar and pineapple industries. Growth and development of diversified agriculture throughout the State. An agriculture industry that continues to constitute a dynamic and essential com strategic, economic, and social well-being 	poner	nt of Hav	waii's
To achieve the agriculture objectives, it shall be the policy of this State to:			
(1) Establish a clear direction for Hawaii's agriculture through stakeholder commitment and advocacy.			X
(2) Encourage agriculture by making the best use of natural resources.			X
(3) Provide the governor and the legislature with information and options needed for prudent decision-making for the development of agriculture.			X
 (4) Establish strong relationships between the agricultural and visitor industries for mutual marketing benefits. 			X
(5) Foster increased public awareness and understanding of the contributions and benefits of agriculture as a major sector of Hawai'i's economy			X
 (6) Seek the enactment and retention of federal and state legislation that benefits Hawai'i's agricultural industries. 			x
(7) Strengthen diversified agriculture by developing an effective promotion, marketing, and distribution system between Hawai'i's food producers and consumers in the State, nation, and world.			x
(8) Support research and development activities that strengthen economic productivity in agriculture, stimulate greater efficiency, and enhance the development of new products and agricultural by-products.			x
(9) Enhance agricultural growth by providing public incentives and encouraging private initiatives.			x



Table 5-1: The Hawai`i State Plan	S	NS	N/A
(10) Assure the availability of agriculturally suitable lands with adequate water to accommodate present and future needs.			x
(11) Increase the attractiveness and opportunities for an agricultural education and livelihood.			x
(12) In addition to the State's priority on food, expand Hawai'i's agricultural base by promoting growth and development of flowers, tropical fruits and plants, livestock, feed grains, forestry, food crops, aquaculture, and other potential enterprises.			x
(13) Promote economically competitive activities that increase Hawai'i's agricultural self- sufficiency, including the increased purchase and use of Hawaii-grown food and food products by residents, businesses, and governmental bodies as defined under section 103D-104.			x
(14) Promote and assist in the establishment of sound financial programs for diversified agriculture			x
(15) Institute and support programs and activities to assist the entry of displaced agricultural workers into alternative agricultural or other employment.			x
(16) Facilitate the transition of agricultural lands in economically non-feasible agricultural production to economically viable agricultural uses.			x
(17) Perpetuate, promote, and increase use of traditional Hawaiian farming systems, such as the use of loko i'a, māla, and irrigated lo'i, and growth of traditional Hawaiian crops, such as kalo, 'uala, and 'ulu.			x
(18) Increase and develop small-scale farms.			X
Discussion: The Proposed Action will not impact any of the objectives and policies outlined a economy related to agriculture.	above	for the	
226-8 Objective and policies for the economy— visitor industry.			
 (a) Planning for the State's economy with regard to the visitor industry shall be directed achievement of the objective of a visitor industry that constitutes a major component for Hawai'i's economy. 	towar It of st	ds the eady gr	owth
(b) To achieve the visitor industry objective, it shall be the policy of this State to:			×
(1) Support and assist in the promotion of Hawai is visitor attractions and facilities.	 		X
(2) Ensure that visitor industry activities are in keeping with the social, economic, and physical needs and aspirations of Hawai'i's people.			X
(3) Improve the quality of existing visitor destination areas by utilizing Hawaii's strengths in science and technology.			x
(4) Encourage cooperation between the public and private sectors in developing and maintaining well-designed, adequately serviced visitor industry and related developments which are sensitive to neighboring communities and activities.			x
(5) Develop the industry in a manner that will continue to provide new job opportunities and steady employment for Hawai'i's people.			x
(6) Provide opportunities for Hawai'i's people to obtain job training and education that will allow for upward mobility within the visitor industry.			x



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
(7)	Foster a recognition of the contribution of the visitor industry to Hawai'i's economy and the need to perpetuate the aloha spirit.			X
(8)	Foster an understanding by visitors of the aloha spirit and of the unique and sensitive character of Hawai'i's cultures and values.			X
Discuss econom	ion: The Proposed Action will not impact any of the objectives and policies outlined a y related to the visitor industry.	bove	for the	
§226 9	Objective and policies for the economy— federal expenditures.			
(a)	Planning for the State's economy with regard to federal expenditures shall be directed achievement of the objective of a stable federal investment base as an integral compleconomy.	d tow ponent	ards : of Haw	/ai`i's
(b)	To achieve the federal expenditures objective, it shall be the policy of this State to:	-		
(1)	Encourage the sustained flow of federal expenditures in Hawai'i that generates long-term government civilian employment.			x
(2)	Promote Hawaii's supportive role in national defense, in a manner consistent with Hawaii's social, environmental, and cultural goals by building upon dual-use and defense applications to develop thriving ocean engineering, aerospace research and development, and related dual-use technology sectors in Hawaii's economy.			x
(3)	Promote the development of federally supported activities in Hawai'i that respect statewide economic concerns, are sensitive to community needs, and minimize adverse impacts on Hawai'i's environment.	x		
(4)	Increase opportunities for entry and advancement of Hawai'i's people into federal government service.			X
(5)	Promote federal use of local commodities, services, and facilities available in Hawai'i.	х		
(6)	Strengthen federal-state-county communication and coordination in all federal activities that affect Hawai'i.			X
(7)	Pursue the return of federally controlled lands in Hawai'i that are not required for either the defense of the nation or for other purposes of national importance, and promote the mutually beneficial exchanges of land between federal agencies, the State, and the counties.			x
Discuss expendit through	tion: The Proposed Action will support the objectives and policies for the econom cures. The various improvements will be primarily funded by the County and may also the State of Hawai'i Department of Health (DOH) Clean Water State Revolving Fund (utilize (CWSF	ted to e federa (F) Prog	federal I funds ram.
§226-1	0 Objective and policies for the economy— potential growth and innovative	activ	ities.	
(a)	Planning for the State's economy with regard to potential growth and innovative actidirected towards achievement of the objective of development and expansion of poterinnovative activities that serve to increase and diversify Hawai'i's economic base.	vities : ential	shall be growth	and
(b)	To achieve the potential growth activity objective, it shall be the policy of this State t	:0:		
(1)	Facilitate investment and employment growth in economic activities that have the potential to expand and diversify Hawaii's economy, including but not limited to			X



Table 5-1: The Hawai`i State Plan	S	NS	N/A
diversified agriculture, aquaculture, renewable energy development, creative media, health care, and science and technology-based sectors.			
(2) Facilitate investment in innovative activity that may pose risks or be less labor- intensive than other traditional business activity, but if successful, will generate revenue in Hawai'i through the export of services or products or substitution of imported services or products.			x
(3) Encourage entrepreneurship in innovative activity by academic researchers and instructors who may not have the background, skill, or initial inclination to commercially exploit their discoveries or achievements.			x
(4) Recognize that innovative activity is not exclusively dependent upon individuals with advanced formal education, but that many self-taught, motivated individuals are able, willing, sufficiently knowledgeable, and equipped with the attitude necessary to undertake innovative activity.			x
(5) Increase the opportunities for investors in innovative activity and talent engaged in innovative activity to personally meet and interact at cultural, art, entertainment, culinary, athletic, or visitor-oriented events without a business focus.			x
(6) Expand Hawai'i's capacity to attract and service international programs and activities that generate employment for Hawai'i's people.			x
(7) Enhance and promote Hawai'i's role as a center for international relations, trade, finance, services, technology, education, culture, and the arts.			x
(8) Accelerate research and development of new energy-related industries based on wind, solar, ocean, and underground resources and solid waste.			x
(9) Promote Hawai'i's geographic, environmental, social, and technological advantages to attract new economic activities into the State.			x
(10) Provide public incentives and encourage private initiative to attract new industries that best support Hawai'i's social, economic, physical, and environmental objectives.			x
(11) Increase research and the development of ocean related economic activities such as mining, food production, and scientific research.			x
(12) Develop, promote, and support research and educational and training programs that will enhance Hawai'i's ability to attract and develop economic activities of benefit to Hawai'i.			x
(13)Foster a broader public recognition and understanding of the potential benefits of new, growth oriented industry in Hawai'i.			x
(14) Encourage the development and implementation of joint federal and state initiatives to attract federal programs and projects that will support Hawaii's social, economic, physical, and environmental objectives.			x
(15) Increase research and development of businesses and services in the telecommunications and information industries.			x
(16) Foster the research and development of nonfossil fuel and energy efficient modes of transportation			x



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
(17)	Recognize and promote health care and health care information technology as growth industries.			x
Discuss economy	ion: The Proposed Action will not impact any of the objectives and policies outlined a related to potential growth and innovative activities.	above	for the	
226-10	5 Objectives and policies for the economy— information industry.			
(a) (b)	Planning for the State's economy with regard to telecommunications and information directed toward recognizing that broadband and wireless communication capability a are foundations for an innovative economy and positioning Hawai'i as a leader in browireless communications and applications in the Pacific Region. To achieve the information industry objective, it shall be the policy of this State to:	n techr Ind infi Dadban	nology s rastruct nd and	hall be ure
(1)	Promote efforts to attain the highest speeds of electronic and wireless communication within Hawai'i and between Hawai'i and the world, and make high speed communication available to all residents and businesses in Hawaii			x
(2)	Encourage the continued development and expansion of the telecommunications infrastructure serving Hawai'i to accommodate future growth and innovation in Hawaii's economy.			x
(3)	Facilitate the development of new or innovative business and service ventures in the information industry which will provide employment opportunities for the people of Hawaii.			x
(4)	Encourage mainland- and foreign-based companies of all sizes, whether information technology-focused or not, to allow their principals, employees, or contractors to live in and work from Hawaii, using technology to communicate with their headquarters, offices, or customers located out-of-state.			x
(5)	Encourage greater cooperation between the public and private sectors in developing and maintaining a well-designed information industry.			x
(6)	Ensure that the development of new businesses and services in the industry are in keeping with the social, economic, and physical needs and aspirations of Hawaii's people.			x
(7)	Provide opportunities for Hawaii's people to obtain job training and education that will allow for upward mobility within the information industry.			x
(8)	Foster a recognition of the contribution of the information industry to Hawaii's economy.			x
(9)	Assist in the promotion of Hawai'i as a broker, creator, and processor of information in the Pacific.			x
Discuss economy	ion: The Proposed Action will not impact any of the objectives or policies outlined ab related to telecommunications and information technology industries.	ove fo	r the	
§226-1	1 Objectives and policies for the physical environment— land-based, shorel es.	ine, a	nd mai	rine
(a)	The land-based, shoreline, and marine resources objectives are:			
	(1) Prudent use of Hawai'i's land-based, shoreline, and marine resources.			
	(2) Effective protection of Hawai'i's unique and fragile environmental resources.			



Table 5-1: The Hawai'i State Plan	S	NS	N/A
(b) To achieve the land-based, shoreline, and marine resources objectives, it shall be to:	he polic	y of this	State
(1) Exercise an overall conservation ethic in the use of Hawai'i's natural resources.			X
(2) Ensure compatibility between land-based and water-based activities and natural resources and ecological systems.	x		
(3) Take into account the physical attributes of areas when planning and designing activities and facilities.	x		
(4) Manage natural resources and environs to encourage their beneficial and multiple use without generating costly or irreparable environmental damage.	x		
(5) Consider multiple uses in watershed areas, provided such uses do not detrimentally affect water quality and recharge functions.	x		
(6) Encourage the protection of rare or endangered plant and animal species and habitats native to Hawai'i.	x		
(7) Provide public incentives that encourage private actions to protect significant natural resources from degradation or unnecessary depletion.			x
(8) Pursue compatible relationships among activities, facilities, and natural resources.	X		
(9) Promote increased accessibility and prudent use of inland and shoreline areas for public recreational, educational, and scientific purposes.			x
Discussion: The Proposed Action will support the objective and policies for the physical environment related to land-based, shoreline, and marine resources.			
All native plants found on the site are relatively common. No significant adverse impa endangered species or their habitat are anticipated as a result of the Proposed Action. Project edge of the urban environment. The improvements will provide a facility for present and future The project site is located approximately two miles away from the nearest coastline.	cts on t Site is ure gen	threate locatec erations	ned or l at the s.
There are no delineated or proposed wetlands in the project area and there are no direct h between the project area and nearby surface waters. No impacts to these resources are anti-	iydrolog cipated	jic conn	ections
§226-12 Objective and policies for the physical environment— scenic, natural bear resources.	auty, a	nd hist	oric
 Planning for the State's physical environment shall be directed towards achievement enhancement of Hawai'i's scenic assets, natural beauty, and multi-cultural/historica 	it of the I resoui	e objecti rces	ve of
(b) To achieve the scenic, natural beauty, and historic resources objective, it shall be the to:	ne polic	y of this	s State
(1) Promote the preservation and restoration of significant natural and historic resources.	x		
(2) Provide incentives to maintain and enhance historic, cultural, and scenic amenities.			X
(3) Promote the preservation of views and vistas to enhance the visual and aesthetic enjoyment of mountains, ocean, scenic landscapes, and other natural features.	x		
 (4) Protect those special areas, structures, and elements that are an integral and functional part of Hawai'i's ethnic and cultural heritage. 	x		



Table 5-1: The Hawai'i State Plan	S	NS	N/A
(5) Encourage the design of developments and activities that complement the natural beauty of the islands.	x		
Discussion: The Proposed Action will support the objectives and policies for the physical envisence, natural beauty, and historic resources.	rironme	ent relat	ed to:
The Proposed Action will minimize and prevent disturbance to significant natural and historic resources to the fullest reasonable extents possible. Through the AIS conducted, one historic property that was evaluated as eligible for listing on the National Register, shall not be impacted by the Proposed Project as its integrity and significance shall not be diminished. Additionally,11 historic properties have been identified within the Project Area. The AIS has adequately documented the location, extent, function, age, and construction methods of these properties. Archeological monitoring has been recommended during all ground disturbing activities to facilitate identification for information purposes.			
Although historic properties have been determined to require no further work, the County will communication and coordination with identified stakeholders to develop feasible and reasonal on either preserving historic properties or alternatives that integrate the broader preservation Waikapuna property.	l consi ble acc goals	der furtl commod of the	ner ations
§226-13 Objectives and policies for the physical environment— land, air, and wate	er qua	lity.	
(a) Planning for the State's physical environment with regard to land, air, and water qua towards achievement of the following objectives:	llity sh	all be di	rected
(1) Maintenance and pursuit of improved quality in Hawai'i's land, air, and water rea	source	s.	
(2) Greater public awareness and appreciation of Hawai'i's environmental resources			
(b) To achieve the land, air, and water quality objectives, it shall be the policy of this St	ate to:		
 Foster educational activities that promote a better understanding of Hawai'i's limited environmental resources. 			X
(2) Promote the proper management of Hawai'i's land and water resources.	X		
(3) Promote effective measures to achieve desired quality in Hawai'i's surface, ground, and coastal waters.	x		
(4) Encourage actions to maintain or improve aural and air quality levels to enhance the health and well-being of Hawai'i's people.	x		
 (5) Reduce the threat to life and property from erosion, flooding, tsunamis, hurricanes, earthquakes, volcanic eruptions, and other natural or man-induced hazards and disasters. 			x
(6) Encourage design and construction practices that enhance the physical qualities of Hawai'i's communities.			x
(7) Encourage urban developments in close proximity to existing services and facilities.	X		
(8) Foster recognition of the importance and value of the land, air, and water resources to Hawai'i's people, their cultures and visitors.			X
Discussion: The Proposed Action supports the objectives and policies for the physical enviro land, air, and water quality.	nment	related	to

The Proposed Action will improve wastewater treatment facilities to ensure wastewater treatment efficiency to maintain water resources quality for present and future generations. This will enhance the health and well-being of Hawai'i's people.



Table 5-1: The Hawai'i State Plan	S	NS	N/A

Construction of the proposed action would involve land disturbing activities, which may result in short-term soil erosion impacts and generate temporary increases in fugitive dust. Construction activities must comply with the provisions of Hawai'i Administrative Rules, §11-60.1-33 on Fugitive Dust. Therefore, a dust control management plan will be developed which identifies and mitigates all activities that may generate airborne, visible fugitive dust. Measures to control airborne, visible fugitive dust from the road areas and during the various phases of construction will be incorporated into the project plans and specifications. Respective contractors would be responsible for adhering to air quality standards and minimizing air quality impacts during the various phases of construction. An air pollution control permit may be required to operate certain construction equipment, such as crushers or generator sets, which may be used during construction. If required, the contractor will be responsible for obtaining an air pollution control permit from the Clean Air Branch and complying with all applicable conditions and requirements. Air monitoring would also be required during construction. With the appropriate controls in place, no significant impacts on air quality are anticipated as a result of the proposed action.

Any potential impacts on air quality would be mitigated by complying with the State DOH Administrative Rules, Title 11, Chapter 60 "Air Pollution Control". In the long-term, GHG emissions in the project area are not expected to increase as future land use plans proposed by DEM are expected to involve consolidation of existing solid waste management program components from the surrounding area.

Soil erosion impacts would be mitigated by incorporating best management practices (BMP) and erosion control measures into the project plans and specifications. Specific measures may include but are not limited to phasing the project to minimize the total area of exposed soil at any given time, revegetating or stabilizing disturbed areas of soil as soon as possible after working, minimizing disturbance of soil during periods of heavy rain, applying protective covers to soil and material stockpiles, and installing appropriate erosion and sedimentation control devices during construction. In addition, the proposed action will comply with the requirements of Hawai'i County Code, Chapter 10, related to Erosion and Sedimentation Control. Following construction, exposed soils in the project area would be stabilized or re-vegetated to control erosion.

Soil erosion impacts would also be mitigated through coordination with the appropriate agencies during permitting and construction. A NPDES permit for storm water runoff from construction activities is anticipated to be required as individual and/or cumulative soil disturbances in the project area may exceed one acre of land area. Any discharges related to project construction or operation activities will comply with applicable State Water Quality Standards as specified in HAR Chapter 11-54 Water Quality Standards and HAR Chapter 11-55 Water Pollution Control, Department of Health.

The Proposed Action will be developed in an urbanized environment that is in close proximity to many available services and facilities within Nā'ālehu.

§226-14 Objective and policies for facility systems— in general.

(a) Planning for the State's facility systems in general shall be directed towards achievement of the objective of water, transportation, waste disposal, and energy and telecommunication systems that support statewide social, economic, and physical objectives.

(b) To achieve the general facility systems objective, it shall be the policy of this State to):	
 Accommodate the needs of Hawai'i's people through coordination of facility systems and capital improvement priorities in consonance with state and county plans. 	x	
(2) Encourage flexibility in the design and development of facility systems to promote prudent use of resources and accommodate changing public demands and priorities.	x	
(3) Ensure that required facility systems can be supported within resource capacities and at reasonable cost to the user.	x	



Table 5-1: The Hawai'i State Plan	S	NS	N/A			
(4) Pursue alternative methods of financing programs and projects and cost-saving techniques in the planning, construction, and maintenance of facility systems.	x					
Discussion: The Proposed Action supports the objectives and policies for facility systems in general.						
The Proposed Action will result in the closer of the LCCs in the Naalehu area and develop a new WWTP facility to ensure wastewater treatment efficiency accommodating the needs of Hawai'i's present and future generations. Furthermore, future expansion of subsurface drip irrigation are planned to improve operating efficiency and support maintenance activities. The Proposed Action shall allow for flexibility in the design and development of the WWTP to best suite public needs,						
§226-15 Objectives and policies for facility systems— solid and liquid wastes.						
(b) Planning for the State's facility systems with regard to solid and liquid wastes shall be directed towards the achievement of the following objectives:						
(1) Maintenance of basic public health and sanitation standards relating to treatment and disposal of solid and liquid wastes.						
(2) Provision of adequate sewerage facilities of physical and economic activities tha in housing, employment, mobility, and other areas.	t allevi	ate prol	olems			
(c) To achieve solid and liquid waste objectives, it shall be the policy of this State to:						
(1) Encourage the adequate development of sewerage facilities that complement planned growth.	x					
(2) Promote re-use and recycling to reduce solid and liquid wastes and employ a conservation ethic.	x					
(3) Promote research to develop more efficient and economical treatment and disposals of solid and liquid wastes.			x			
Discussion: The Proposed Action will support the objectives and policies for facility systems related to solid and liquid wastes.						
The Proposed Action will result in the closer of the LCCs in the Naalehu area and develop a new WWTP facility to ensure wastewater treatment efficiency accommodating the needs of Hawai'i's present and future generations. Furthermore, future expansion of subsurface drip irrigation are planned to improve operating efficiency and support maintenance activities.						
§226-16 Objective and policies for facility systems— water.						
(a) Planning for the State's facility systems with regard to water shall be directed towards achievement of the objective of the provision of water to adequately accommodate domestic, agricultural, commercial, industrial, recreational, and other needs within resource capacities.						
(b) To achieve the facility systems water objective, it shall be the policy of the State to:	1					
(1) Coordinate development of land use activities with existing and potential water supply.			X			
(2) Support research and development of alternative methods to meet future water requirements well in advance of anticipated needs.			x			
(3) Reclaim and encourage the productive use of runoff water and wastewater discharges.	x					
(4) Assist in improving the quality, efficiency, service, and storage capabilities of water systems for domestic and agricultural use.	x					



Table 5-1: The Hawai`i State Plan	S	NS	N/A
(5) Support water supply services to areas experiencing critical water problems.			X
(6) Promote water conservation programs and practices in government, private industry, and the general public to help ensure adequate water to meet long-t needs.	term		x
Discussion: The Proposed Action will support the objectives and policies for facility sy	stems relate	ed to wat	er.
In anticipation of future development within Nā'ālehu, it is recommended that the Prop rated to treat an average dry weather flow of 125,000 gpd. Drainage improvements sh maintain runoff conditions within the vicinity of the project area.	osed Actior all be made	n design b onsite to	be D
§226-17 Objectives and policies for facility systems— transportation.			
(a) Planning for the State's facility systems with regard to transportation shall be achievement of the following objectives:	directed to	wards the	2
(1) An integrated multi-modal transportation system that services statewide refficient, economical, safe, and convenient movement of people and good	needs and p ds.	promotes	the
(2) A statewide transportation system consistent with planned growth object(b) To achieve the transportation objectives, it shall be the policy of this State to:	ives througl	nout the s	State
(1) Design, program, and develop a multi-modal system in conformance with design growth and physical development as stated in this chapter.	ired		x
(2) Coordinate state, county, federal, and private transportation activities and programs toward the achievement of statewide objectives.			x
(3) Encourage a reasonable distribution of financial responsibilities for transportat among participating governmental and private parties.	ion		x
(4) Provide for improved accessibility to shipping, docking, and storage facilities.			X
(5) Promote a reasonable level and variety of mass transportation services that adequately meet statewide and community needs.			x
(6) Encourage transportation systems that serve to accommodate present and ful development needs of communities.	ture		x
(7) Encourage a variety of carriers to offer increased opportunities and advantage inter-island movement of people and goods.	es to		x
(8) Increase the capacities of airport and harbor systems and support facilities to effectively accommodate transshipment and storage needs.			x
(9) Encourage the development of transportation, systems and programs which w assist statewide economic growth and diversification.	vould		x
(10) Encourage the design and development of transportation systems sensitive to needs of affected communities and the quality of Hawai'i's natural environmer	the nt.		x
(11) Encourage safe and convenient uses of low-cost, energy-efficient, non-pollutir means of transportation.	ng		x
(12) Coordinate intergovernmental land use and transportation planning activities ensure the timely delivery of supporting transportation infrastructure in order accommodate planned growth objectives.	to to		x


	Table 5-1: The Hawai'i State Plan	S	NS	N/A
(13)	Encourage diversification of transportation modes and infrastructure to promote alternate fuels and energy efficiency.			X
Discuss systems	Discussion: The Proposed Action will not impact any of the objectives and policies outlined above for facility systems related to transportation.			ty
§226-1	8 Objectives and policies for facility systems— energy.			
(a)	Planning for the State's facility systems with regard to energy shall be directed toward of the following objectives, giving due consideration to all:	rd the	achieve	ment
	(1) Dependable, efficient, and economical statewide energy and telecommunication supporting the needs of the people.	syste	ns capa	ble of
	(2) Increased energy self-sufficiency through the reduction and ultimate elimination dependence on imported fuels for electrical generation and ground transportation	of Ha	waii's	
	 Greater diversification of energy generation in the face of threats to Hawaii's energy systems; 	ergy s	upplies	and
	(4) Reduction, avoidance, or sequestration of greenhouse gas emissions from energy and	jy sup	oly and	use;
	(5) Utility models that make the social and financial interests of Hawaii's utility custo	omers	a priorit	y
(b) (c)	To achieve the energy objectives, it shall be the policy of this State to ensure the pro- reasonably priced, and dependable energy services to accommodate demand To further achieve the energy objectives, it shall be the policy of this State to:	ovision	of adeo	quate,
(1)	Support research and development as well as promote the use of renewable energy sources.			x
(2)	Ensure a sufficient supply of energy to enable power systems to support the demands of growth.			X
(3)	Base decisions of least-cost supply-side and demand-side energy resource options on a comparison of their total costs and benefits when a least-cost is determined by a reasonably comprehensive, quantitative, and qualitative accounting of their long-term, direct and indirect economic, environmental, social, cultural, and public health costs and benefits.			x
(4)	 Promote all cost-effective conservation of power and fuel supplies through measures, including: (A) Development of cost-effective demand-side management programs; (B) Education; (C) Adoption of energy-efficient practices and technologies; and (D) Increasing energy efficiency and decreasing energy use in public infrastructure. 			x
(5)	Ensure, to the extent that new supply-side resources are needed, that the development or expansion of energy systems uses the least-cost energy supply option and maximizes efficient technologies.			X
(6)	Support research, development, demonstration, and use of energy efficiency, load management, and other demand-side management programs, practices, and technologies.			x
(7)	Promote alternate fuels and transportation energy efficiency.			X



Table 5-1: The Hawai'i State Plan	S	NS	N/A
(8) Support actions that reduce, avoid, or sequester greenhouse gases in utility, transportation, and industrial sector applications.			x
(9) Support actions that reduce, avoid, or sequester Hawaii's greenhouse gas emissions through agriculture and forestry initiatives.			x
(10) Provide priority handling and processing for all state and county permits required for renewable energy projects.			x
(11) Ensure that liquefied natural gas is used only as a cost-effective transitional, limited-term replacement of petroleum for electricity generation and does not impede the development and use of other cost-effective renewable energy sources.			x
(12) Promote the development of indigenous geothermal energy resources that are located on public trust land as an affordable and reliable source of firm power for Hawaii.			x
Discussion: The Proposed Action will not impact any of the objectives and policies outlined a systems related to energy.	bove	for facil	ity
§226-18.5 Objectives and policies for facility systems— telecommunications.			
(a) Planning for the State's telecommunications facility systems shall be directed towards the dependable, efficient, and economical statewide telecommunications systems capable of supp the people.(b) To achieve the telecommunications objective, it shall be the policy of this State to ensure	achie orting the p	the new	of eds of of
adequate, reasonably priced, and dependable telecommunications services to accommodate d	leman	d.	
(c) To further achieve the telecommunications objective, it shall be the policy of this State to: (1) Eacilitate research and development of telecommunication systems and resources	•		Y
 (2) Encourage public and private sector efforts to develop means for adequate, ongoing telecommunication planning. 			x
(3) Promote efficient management and use of existing telecommunication systems and services.			x
(4) Facilitate the development of education and training of telecommunication personnel.			x
Discussion: The Proposed Action will not impact any of the objectives and policies outlined a systems related to telecommunications.	bove	for facil	ity
§226-19 Objectives and policies for socio-cultural advancement— housing.			
(a) Planning for the State's socio-cultural advancement with regard to housing shall be d achievement of the following objectives:	lirecte	d towar	d the
(1) Greater opportunities for Hawaii's people to secure reasonably priced, safe, sanit homes, located in suitable environments that satisfactorily accommodate the new families and individuals, through collaboration and cooperation between governr and for-profit developers to ensure that more rental and for sale affordable hous available to extremely low-, very low-, lower-, moderate-, and above moderate- Hawaii's population.	tary, a eds ar ment a sing is incom	and lival nd desire and non made e segme	ole es of profit ents of

(2) The orderly development of residential areas sensitive to community needs and other land uses.



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
	(3) The development and provision of affordable rental housing by the State to mee of Hawaii's people.	et the	housing	needs
(b)	To achieve the housing objectives, it shall be the policy of this State to:			
(1)	Effectively accommodate the housing needs of Hawai'i's people.			X
(2)	Stimulate and promote feasible approaches that increase affordable rental and for sale housing choices for extremely low-, very low-, lower-, moderate-, and above moderate-income households.			x
(3)	Increase homeownership and rental opportunities and choices in terms of quality, location, cost, densities, style, and size of housing.			x
(4)	Promote appropriate improvement, rehabilitation, and maintenance of existing housing units and residential areas.			x
(5)	Promote design and location of housing developments taking into account the physical setting, accessibility to public facilities and services, and other concerns of existing communities and surrounding areas.			x
(6)	Facilitate the use of available vacant, developable, and underutilized urban lands for housing.			x
(7)	Foster a variety of lifestyles traditional to Hawai'i through the design and maintenance of neighborhoods that reflect the cultures and values of the community.			х
(8)	Promote research and development of methods to reduce the cost of housing construction in Hawai'i.			x
Discuss related t	sion: The Proposed Action will not impact the objectives and policies for socio-cultural to housing.	adva	ncemen	t
§226-2	0 Objectives and policies for socio-cultural advancement— health.			
(a)	Planning for the State's socio-cultural advancement with regard to health shall be dir achievement of the following objectives:	ected	towards	5
	(1) Fulfillment of basic individual health needs of the general public.			
	(2) Maintenance of sanitary and environmentally healthful conditions in Hawai'i's co	mmun	ities.	
	(3) Elimination of health disparities by identifying and addressing social determinant	s of h	ealth.	
(b)	To achieve the health objectives, it shall be the policy of this State to:			
(1)	Provide adequate and accessible services and facilities for prevention and treatment of physical and mental health problems, including substance abuse.			x
(2)	Encourage improved cooperation among public and private sectors in the provision of health care to accommodate the total health needs of individuals throughout the State.			х
(3)	Encourage public and private efforts to develop and promote statewide and local strategies to reduce health care and related insurance costs.			x
(4)	Foster an awareness of the need for personal health maintenance and preventive health care through education and other measures.			X



Table 5-1: The Hawai'i State Plan	S	NS	N/A
(5) Provide programs, services, and activities that ensure environmentally healthful and sanitary conditions.	x		
 (6) Improve the State's capabilities in preventing contamination by pesticides and other potentially hazardous substances through increased coordination, education, monitoring, and enforcement 			x
(7) Prioritize programs, services, interventions, and activities that address identified social determinants of health to improve native Hawaiian health and well-being consistent with the United States Congress' declaration of policy as codified in title 42 United States Code section 11702, and to reduce health disparities of disproportionately affected demographics, including native Hawaiians, other Pacific Islanders, and Filipinos. The prioritization of affected demographic groups other than native Hawaiians may be reviewed every ten years and revised based on the best available epidemiological and public health data.			x
Discussion: The Proposed Action will support the objectives and policies for socio-cultural ac regarding health.	lvance	ment	
The Proposed Action will undertake the construction of critical wastewater treatment facilities t processes and fulfill objectives and policies related to socio-cultural advancements with regard	o impi Is to h	rove tre ealth.	atment
§226-21 Objective and policies for socio-cultural advancement— education.			
(a) Planning for the State's socio-cultural advancement with regard to education shall be dire achievement of the objective of the provision of a variety of educational opportunities to enable fulfill their needs, responsibilities, and aspirations.	cted to le indi	owards ividuals	to
(b) To achieve the education objective, it shall be the policy of this State to:			
 Support educational programs and activities that enhance personal development, physical fitness, recreation, and cultural pursuits of all groups. 			x
(2) Ensure the provision of adequate and accessible educational services and facilities that are designed to meet individual and community needs.			x
(3) Provide appropriate educational opportunities for groups with special needs.			Х
(4) Promote educational programs which enhance understanding of Hawaii's cultural heritage.			x
(5) Provide higher educational opportunities that enable Hawaii's people to adapt to changing employment demands.			x
(6) Assist individuals, especially those experiencing critical employment problems or barriers, or undergoing employment transitions, by providing appropriate employment training programs and other related educational opportunities.			x
(7) Promote programs and activities that facilitate the acquisition of basic skills, such as reading, writing, computing, listening, speaking, and reasoning.			x
(8) Emphasize quality educational programs in Hawaii's institutions to promote academic excellence.			x
(9) Support research programs and activities that enhance the education programs of the State.			x
Discussion : The Proposed Action will not impact the objectives and policies of the State for t advancement regarding education.	he so	cio-cultu	ıral



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
\$226-2	2 Objective and policies for socio-sultural advancement— social services			
(a)	 (a) Planning for the State's socio-cultural advancement with regard to social services shall be directed towards the achievement of the objective of improved public and private social services and activities that enable individuals, families, and groups to become more self-reliant and confident to improve their well-being. (b) To achieve the social services objective, it shall be the policy of this State to: 			ies their
(1)	Assist individuals, especially those in need of attaining a minimally adequate			
(1)	standard of living and those confronted by social and economic hardship conditions, through social services and activities within the State's fiscal capacities.			X
(2)	Promote coordination and integrative approaches among public and private agencies and programs to jointly address social problems that will enable individuals, families, and groups to deal effectively with social problems and to enhance their participation in society.			x
(3)	Facilitate the adjustment of new residents, especially recently arrived immigrants, into Hawai'i's communities			x
(4)	Promote alternatives to institutional care in the provision of long-term care for elder and disabled populations.			x
(5)	Support public and private efforts to prevent domestic abuse and child molestation, and assist victims of abuse and neglect.			x
(6)	Promote programs which assist people in need of family planning services to enable them to meet their needs.			x
Discuss related t	sion: The Proposed Action will not impact the objectives and policies for socio-cultural to social services.	advar	ncemen	t
§226-2	3 Objective and policies for socio-cultural advancement— leisure.			
(a) (b)	Planning for the State's socio-cultural advancement with regard to leisure shall be dia achievement of the objective of the adequate provision of resources to accommodate artistic, and recreational needs for present and future generations. To achieve the leisure objective, it shall be the policy of this State to:	rected e diver	toward: se cultu	s the ıral,
(1)	Foster and preserve Hawai'i's multi-cultural heritage through supportive cultural.			
	artistic, recreational, and humanities-oriented programs and activities.			X
(2)	Provide a wide range of activities and facilities to fulfill the cultural, artistic, and recreational needs of all diverse and special groups effectively and efficiently.			X
(3)	Enhance the enjoyment of recreational experiences through safety and security measures, educational opportunities, and improved facility design and maintenance.			x
(4)	Promote the recreational and educational potential of natural resources having scenic, open space, cultural, historical, geological, or biological values while ensuring that their inherent values are preserved			x
(5)	Ensure opportunities for everyone to use and enjoy Hawai'i's recreational resources.			X



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
(6)	Assure the availability of sufficient resources to provide for future cultural, artistic, and recreational needs			x
(7)	Provide adequate and accessible physical fitness programs to promote the physical and mental well-being of Hawai'i's people.			x
(8)	Increase opportunities for appreciation and participation in the creative arts, including the literary, theatrical, visual, musical, folk, and traditional art forms.			x
(9)	Encourage the development of creative expression in the artistic disciplines to enable all segments of Hawai'i's population to participate in the creative arts.			x
(10)	Assure adequate access to significant natural and cultural resources in public ownership.			x
Discuss related t	ion: The Proposed Action will not impact the objectives and policies for socio-cultura to leisure.	l adva	ncemen	t
§226-24 well-be	4 Objective and policies for socio-cultural advancement-—-individual rights ing.	and	person	al
(a) (b)	Planning for the State's socio-cultural advancement with regard to individual rights a being shall be directed towards achievement of the objective of increased opportunit of individual rights to enable individuals to fulfill their socio-economic needs and asp To achieve the individual rights and personal wellbeing objective, it shall be the police	nd per ties an iration cy of th	rsonal w d protee s. nis State	vell- ction e to:
(1)	Provide effective services and activities that protect individuals from criminal acts and unfair practices and that alleviate the consequences of criminal acts in order to foster a safe and secure environment.			x
(2)	Uphold and protect the national and state constitutional rights of every individual.			X
(3)	Assure access to, and availability of, legal assistance, consumer protection, and other public services which strive to attain social justice.			x
(4)	Ensure equal opportunities for individual participation in society.			X
Discuss related t	sion: The Proposed Action will not impact the objectives and policies for socio-cultura to individual rights and personal well-being.	l adva	ncemen	t
§226-2	5 Objective and policies for socio-cultural advancement— culture.			
(a)	Planning for the State's socio-cultural advancement with regard to culture shall be di achievement of the objective of enhancement of cultural identities, traditions, values of Hawai'i's people.	irectec 5, custo	l toward oms, an	l the d arts
(b)	To achieve the culture objective, it shall be the policy of this State to:			
	(1) Foster increased knowledge and understanding of Hawai'i's ethnic and cultural heritages and the history of Hawai'i.			x
	(2) Support activities and conditions that promote cultural values, customs, and arts that enrich the lifestyles of Hawai'i's people and which are sensitive and responsive to family and community needs.			x
	(3) Encourage increased awareness of the effects of proposed public and private actions on the integrity and quality of cultural and community lifestyles in Hawai'i.			x



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
	(4) Encourage the essence of the aloha spirit in people's daily-activities to promote harmonious relationships among Hawai'i's people and visitors.			x
Discuss related t	ion: The Proposed Action will not impact the objectives and policies for socio-cultura o culture.	advar	ncement	t
§226-2	6 Objectives and policies for socio-cultural advancement— public safety.			
(a)	Planning for the State's socio-cultural advancement with regard to public safety shall towards the achievement of the following objectives:	be dir	rected	
	(1) Assurance of public safety and adequate protection of life and property for all pe	eople.		
	(2) Optimum organizational readiness and capability in all phases of emergency ma maintain the strength, resources, and social and economic well-being of the con of civil disruptions, wars, natural disasters, and other major disturbances.	nagem nmunit	ient to ty in the	event
	(3) Promotion of a sense of community responsibility for the welfare and safety of H $\!\!\!$	lawai`i	's	
(b)	To achieve the public safety programs objectives, it shall be the policy of this State t	o:		
(1)	Ensure that public safety programs are effective and responsive to community needs.			x
(2)	Encourage increased community awareness and participation in public safety programs.			х
(C)	To achieve the public safety programs objectives, it shall be the policy of this State t	0:		
(1)	Support criminal justice programs aimed at preventing and curtailing criminal activities.			х
(2)	Develop a coordinated, systematic approach to criminal justice administration among all criminal justice agencies.			х
(3)	Provide a range of correctional resources which may include facilities and alternatives to traditional incarceration in order to address the varied security needs of the community and successfully reintegrate offenders into the community.			x
(d)	To further achieve public safety objectives related to emergency management, it sha this State to:	all be t	he polic	y of
(1)	Ensure that responsible organizations are in a proper state of readiness to respond to major war related, natural, or technological disasters and civil disturbances at all times.			x
(2)	Enhance the coordination between emergency management programs throughout the State.			x
Discuss related t	ion: The Proposed Action will not impact the objectives and policies for socio-cultura o public safety.	advar	ncement	t
§226-2	7 Objectives and policies for socio-cultural advancement— government.			
(a)	Planning the State's socio-cultural advancement with regard to government shall be achievement of the following objectives:	directe	ed towa	ds the
	(1) Efficient, effective, and responsive government services at all levels in the State			



	Table 5-1: The Hawai'i State Plan	S	NS	N/A
	(2) Fiscal integrity, responsibility and efficiency in the state government and county	gover	nments	
(b)	To achieve the government objectives, it shall be the policy of this State to:			
(1)	Provide for necessary public goods and services not assumed by the private sector.	X		
(2)	Pursue an openness and responsiveness in government that permits the flow of public information, interaction, and response.			X
(3)	Minimize the size of government to that necessary to be effective.			X
(4)	Stimulate the responsibility in citizens to productively participate in government for a better Hawai'i.			X
(5)	Assure that government attitudes, actions, and services are sensitive to community needs and concerns.	x		
(6)	Provide for a balanced fiscal budget.			Х
(7)	Improve the fiscal budgeting and management system of the State.			X
(8)	Promote the consolidation of state and county governmental functions to increase the effective and efficient delivery of government programs and services and to eliminate duplicative services wherever feasible.			x
Discussion: The Proposed Action will support the objectives and policies for socio-cultural advancement related to government.				

The COH-DEM is proposing to undertake the construction of critical facilities to improve wastewater treatment processes in the Nā'ālehu community. Therefore, the project supports the achievements of the objectives and policies for socio-cultural advancement with regards to government.

PART III. PRIORITY GUIDELINES

Part III of the Hawai'i State Plan establishes the overall priority guidelines to address areas of statewide concern. Under HRS § 226-102, "*The State shall strive to improve the quality of life for Hawai'i's present and future population through the pursuit of desirable courses of action in seven major areas of Statewide concern which merit priority attention: economic development, population growth and land resource management, affordable housing, crime and criminal justice, quality education, principles of sustainability, and climate change adaptation.*"

	Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
§226-103 Eco	nomic priority guidelines.			
(a) Priorit provid	guidelines to stimulate economic growth and encourage business expansion a e needed jobs for Hawai'i's people and achieve a stable and diversified econom	nd dev y:	/elopme	nt to
(1) Seek a and ex	variety of means to increase the availability of investment capital for new panding enterprises.			x



Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
(2) Encourage the expansion of technological research to assist industry development and support the development and commercialization of technological advancements.			x
(3) Improve the quality, accessibility, and range of services provided by government to business, including data and reference services and assistance in complying with governmental regulations.			x
(4) Seek to ensure that state business tax and labor laws and administrative policies are equitable, rational, and predictable.			x
(5) Streamline the building and development permit and review process, and eliminate or consolidate other burdensome or duplicative governmental requirements imposed on business, where public health, safety, and welfare would not be adversely affected.			x
(6) Encourage the formation of cooperatives and other favorable marketing or distribution arrangements at the regional or local level to assist Hawai'i's small-scale producers, manufacturers, and distributors.			x
(7) Continue to seek legislation to protect Hawai'i from transportation interruptions between Hawai'i and the continental United States.			x
 (8) Provide public incentives and encourage private initiative to develop and attract industries which promise long-term growth potentials and which have the following characteristics: (a) An industry that can take advantage of Hawai'i's unique location and available physical and human resources. (b) A clean industry that would have minimal adverse effects on Hawai'i's environment. (c) An industry that is willing to hire and train Hawai'i's people to meet the industry's labor needs. (d) An industry that would provide reasonable income and steady employment. 			x
(9) Support and encourage, through educational and technical assistance programs and other means, expanded opportunities for employee ownership and participation in Hawai'i business.			x
 (10) Enhance the quality of Hawai'i's labor force and develop and maintain career opportunities for Hawai'i's people through the following actions: (a) Expand vocational training in diversified agriculture, aquaculture, and other areas where growth is desired and feasible. (b) Encourage more effective career counseling and guidance in high schools and post-secondary institutions to inform students of present and future career opportunities. (c) Allocate educational resources to career areas where high employment is expected and where growth of new industries is desired. (d) Promote career opportunities in all industries for Hawai'i's people by encouraging firms doing business in the State to hire residents. (e) Promote greater public and private sector cooperation in determining industrial training needs and in developing relevant curricula and on-the-job training opportunities. 			x



	Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
	(f) Provide retraining programs and other support services to assist entry of displaced workers into alternative employment.			
(b)	Priority guidelines to promote the economic health and quality of the visitor industry:	1		
(1)	Promote visitor satisfaction by fostering an environment which enhances the Aloha Spirit and minimizes inconveniences to Hawai'i's residents and visitors.			x
(2)	Encourage the development and maintenance of well-designed, adequately serviced hotels and resort destination areas which are sensitive to neighboring communities and activities and which provides for adequate shoreline setbacks and beach access.			x
(3)	Support appropriate capital improvements to enhance the quality of existing resort destination areas and provide incentives to encourage investment in upgrading, repair, and maintenance of visitor facilities.			x
(4)	Encourage visitor industry practices and activities which respect, preserve, and enhance Hawai'i's significant natural, scenic, historic, and cultural resources.			x
(5)	Develop and maintain career opportunities in the visitor industry for Hawai'i's people, with emphasis on managerial positions.			x
(6)	Support and coordinate tourism promotion abroad to enhance Hawai'i's share of existing and potential visitor markets.			x
(7)	Maintain and encourage a more favorable resort investment climate consistent with the objectives of this chapter.			x
(8)	Support law enforcement activities that provide a safer environment for both visitors and residents alike.			x
(C)	Priority guidelines to promote the continued viability of the sugar and pineapple indus	stries:		1
(1)	Provide adequate agricultural lands to support the economic viability of the sugar and pineapple industries.			x
(2)	Continue efforts to maintain federal support to provide stable sugar prices high enough to allow profitable operations in Hawai'i.			x
(3)	Support research and development, as appropriate, to improve the quality and production of sugar and pineapple crops.			x
(d)	Priority guidelines to promote the growth and development of diversified agriculture a	and aq	uacultu	re:
(1)	Identify, conserve, and protect agricultural and aquacultural lands of importance and initiate affirmative and comprehensive programs to promote economically productive agricultural and aquacultural uses of such lands.			x
(2)	Assist in providing adequate, reasonably priced water for agricultural activities.			X
(3)	Encourage public and private investment to increase water supply and to improve transmission, storage, and irrigation facilities in support of diversified agriculture and aquaculture.			x
(4)	Assist in the formation and operation of production and marketing associations and cooperatives to reduce production and marketing costs.			x



	Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
(5)	Encourage and assist with the development of a waterborne and airborne freight and cargo system capable of meeting the needs of Hawai'i's agricultural community			x
(6)	Seek favorable freight rates for Hawai'i's agricultural products from interisland and overseas transportation operators.			x
(7)	Encourage the development and expansion of agricultural and aquacultural activities which offer long-term economic growth potential and employment opportunities.			x
(8)	Continue the development of agricultural parks and other programs to assist small independent farmers in securing agricultural lands and loans.			x
(9)	Require agricultural uses in agricultural subdivisions and closely monitor the uses in these subdivisions.			x
(e)	Priority guidelines for water use and development:			-
(1)	Maintain and improve water conservation programs to reduce the overall water consumption rate.			x
(2)	Encourage the improvement of irrigation technology and promote the use of non- potable water for agricultural and landscaping purposes.			x
(3)	Increase the support for research and development of economically feasible alternative water sources.			x
(4)	Explore alternative funding sources and approaches to support future water development programs and water system improvements.			x
(f)	Priority guidelines for energy use and development:			
(1)	Encourage the development, demonstration, and commercialization of renewable energy sources.			x
(2)	Initiate, maintain, and improve energy conservation programs aimed at reducing energy waste and increasing public awareness of the need to conserve energy.			x
(3)	Provide incentives to encourage the use of energy conserving technology in residential, industrial, and other buildings.			x
(4)	Encourage the development and use of energy conserving and cost-efficient transportation systems.			x
(g)	Priority guidelines to promote the development of the information industry:			
(1)	Establish an information network, with an emphasis on broadband and wireless infrastructure and capability that will serve as the foundation of and catalyst for overall economic growth and diversification in Hawaii.			x
(2)	Encourage the development of services such as financial data processing, a products and services exchange, foreign language translations, telemarketing, teleconferencing, a twenty-four-hour international stock exchange, international banking, and a Pacific Rim management center.			x
(3)	Encourage the development of small businesses in the information field such as software development; the development of new information systems, peripherals,			X



	Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
	and applications; data conversion and data entry services; and home or cottage services such as computer programming, secretarial, and accounting services.			
(4)	Encourage the development or expansion of educational and training opportunities for residents in the information and telecommunications fields.			x
(5)	Encourage research activities, including legal research in the information and telecommunications fields.			x
(6)	Support promotional activities to market Hawai'i's information industry services.			X
(7)	Encourage the location or co-location of telecommunication or wireless information relay facilities in the community, including public areas, where scientific evidence indicates that the public health, safety, and welfare would not be adversely affected.			x
Discuss for econ	sion: The Proposed Action will not impact the objectives and policies outlined within th omic priority guidelines.	e Hav	/ai`i Stat	e plan:
§226-1	04 Population growth and land resources priority guidelines.			
(a)	Priority guidelines to effect desired statewide growth and distribution:			
(1)	Encourage planning and resource management to ensure that population growth rates throughout the State are consistent with available and planned resource capacities and reflect the needs and desires of Hawai'i's people.	x		
(2)	Manage a growth rate for Hawai'i's economy that will parallel future employment needs for Hawai'i's people.			x
(3)	Ensure that adequate support services and facilities are provided to accommodate the desired distribution of future growth throughout the State.	x		
(4)	Encourage major state and federal investments and services to promote economic development and private investment to the neighbor islands, as appropriate.			x
(5)	Explore the possibility of making available urban land, low-interest loans, and housing subsidies to encourage the provision of housing to support selective economic and population growth on the neighbor islands.			x
(6)	Seek federal funds and other funding sources outside the State for research, program development, and training to provide future employment opportunities on the neighbor islands.			x
(7)	Support the development of high technology parks on the neighbor islands.			X
(b)	Priority guidelines for regional growth distribution and land resource utilization:		1	
(1)	Encourage urban growth primarily to existing urban areas where adequate public facilities are already available or can be provided with reasonable public expenditures and away from areas where other important benefits are present, such as protection of important agricultural land or preservation of lifestyles.			x
(2)	Make available marginal or non-essential agricultural lands for appropriate urban uses while maintaining agricultural lands of importance in the agricultural district.			x
(3)	Restrict development when drafting of water would result in exceeding the sustainable yield or in significantly diminishing the recharge capacity of any groundwater area.			x



Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
(4) Encourage restriction of new urban development in areas where water is insufficient from any source for both agricultural and domestic use.	ent		x
(5) In order to preserve green belts, give priority to state capital improvement funds which encourage location of urban development within existing urban areas exce where compelling public interest dictates development of a non-contiguous new urban core.	pt		x
(6) Seek participation from the private sector for the cost of building infrastructure an utilities, and maintaining open spaces.	nd		x
(7) Pursue rehabilitation of appropriate urban areas.			X
(8) Support the redevelopment of Kaka'ako into a viable residential, industrial, and commercial community.			x
(9) Direct future urban development away from critical environmental areas or imposimitigating measures so that negative impacts on the environment would be minimized.	e x		
(10) Identify critical environmental areas in Hawai'i to include but not be limited to the following: watershed and recharge areas; wildlife habitats (on land and in the ocean); areas with endangered species of plants and wildlife; natural streams and water bodies; scenic and recreational shoreline resources; open space and natura areas; historic and cultural sites; areas particularly sensitive to reduction in water and air quality; and scenic resources.	: L		x
(11) Identify all areas where priority should be given to preserving rural character and lifestyle.			x
(12) Utilize Hawai'i's limited land resources wisely, providing adequate land to accommodate projected population and economic growth needs while ensuring th protection of the environment and the availability of the shoreline, conservation lands, and other limited resources for future generations.	^{ne} x		
(13) Protect and enhance Hawai'i's shoreline, open spaces, and scenic resources.			X
Discussion: The Proposed Action will support the objectives and policies for Population Gro Priority Guidelines. The Proposed Action aims to construct a new wastewater treatment community. The Proposed Action is necessary to ensure that adequate support services an	wth and L system fo d facilities	and Resort the Name	sources ā'ālehu ovided.
The Proposed Action will occur in an existing urban area in close proximity to othe Additionally, the construction and operation of the Proposed Action will implement appropriate to minimize and avoid any negative impacts on the environment.	r services iate mitiga	and fation me	acilities. easures
§226-105 Crime and criminal justice			
Priority guidelines in the area of crime and criminal justice:			
 Support law enforcement activities and other criminal justice efforts that are directed to provide a safer environment. 			x
(2) Target state and local resources on efforts to reduce the incidence of violent crim and on programs relating to the apprehension and prosecution of repeat offender	e rs.		x
(3) Support community and neighborhood program initiatives that enable residents to	2		x



assist law enforcement agencies in preventing criminal activities.

Та	ble 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
(4) Reduce overcro comprehensive sentencing law for persons whe	owding or substandard conditions in correctional facilities through a approach among all criminal justice agencies which may include revisions and use of alternative sanctions other than incarceration o pose no danger to their community.			x
(5) Provide a range community-bas	e of appropriate sanctions for juvenile offenders, including sed programs and other alternative sanctions.			х
(6) Increase public minimize the co	and private efforts to assist witnesses and victims of crimes and to osts of victimization.			X
Discussion: The Proposition related to crime and crime	sed Action will not impact the objectives and policies outlined within the hinal activity.	Hawa	ai`i Stato	e plan
§226-106 Affordable	housing			
Priority guidelines for the	e provision of affordable housing:			
(1) Seek to use ma housing needs	arginal or non-essential agricultural land and public land to meet of low and moderate-income and gap-group households.			x
(2) Encourage the means of reduce	use of alternative construction and development methods as a cing production costs.			x
(3) Improve inform housing.	nation and analysis relative to land availability and suitability for			x
(4) Create incentive rental opportun households, an	es for development which would increase home ownership and hities for Hawai'i's low and moderate-income households, gap-group d residents with special needs.			x
(5) Encourage cont provide low inte occupied housi	tinued support for government or private housing programs that erest mortgages to Hawai'i's people for the purchase of initial owner- ng.			x
(6) Encourage pub housing alterna	lic and private sector cooperation in the development of rental atives.			X
(7) Encourage impo government to	roved coordination between various agencies and levels of deal with housing policies and regulations.			X
(8) Give higher price residents and le individuals outs	prity to the provision of quality housing that is affordable for Hawaii's ess priority to development of housing intended primarily for side of Hawaii.			x
Discussion: The Proposition related to affordable hou	sed Action will not impact the objectives and policies outlined within the Husing.	Hawa	ai`i Stato	e plan
§226-107 Quality edu	ication.			
Priority guidelines to pro	mote quality education:			
(1) Pursue effective needs to streng	e programs which reflect the varied district, school, and student gthen basic skills achievement.			X
(2) Continue emph background to	asis on general education "core" requirements to provide common students and essential support to other university programs.			X
(3) Initiate efforts the education v	to improve the quality of education by improving the capabilities of work force.			X



Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
(4) Promote increased opportunities for greater autonomy and flexibility of educational institutions in their decision-making responsibilities.			x
 (5) Increase and improve the use of information technology in education by the availability of telecommunications equipment for: (A) The electronic exchange of information; (B) Statewide electronic mail; and (C) Access to the Internet. Encourage programs that increase the public's awareness and understanding of the impact of information technologies on our lives. 			x
(6) Pursue the establishment of Hawai'i's public and private universities and colleges as research and training centers of the Pacific.			x
(7) Develop resources and programs for early childhood education.			X
(8) Explore alternatives for funding and delivery of educational services to improve the overall quality of education.			x
(9) Strengthen and expand educational programs and services for students with special needs.			x
Discussion: The Proposed Action will not impact the objectives and policies outlined within th related to quality education.	e Haw	vai`i Stat	e plan
§226-108 Sustainability.			
Priority guidelines and principles to promote sustainability:			_
(1) Encouraging balanced economic, social, community, and environmental priorities.			X
(2) Encouraging planning that respects and promotes living within the natural resources and limits of the State.	x		
(3) Promoting a diversified and dynamic economy.	X		
(4) Encouraging respect for the host culture.			X
(5) Promoting decisions based on meeting the needs of the present without compromising the needs of future generations.	x		
(6) Considering the principles of the ahupua'a system.	X		
(7) Emphasizing that everyone, including individuals, families, communities, businesses, and government, has the responsibility for achieving a sustainable Hawai'i.			x
Discussion: The Proposed Action will support the priority guidelines and principles regarding infrastructure, such as wastewater systems, are essential to building a strong economy, protect and providing an enhanced quality of life. The Proposed Action seeks to rehabilitate aging waste to provide effective and efficient treatment of wastewater. The Proposed Action supports s development and growth through the creation of short-term construction and long-term oper opportunities.	sustai ing the ewate ustain eration	nability. e enviro r infrasti able ecc is emplo	Public nment, ructure onomic oyment
§226-109 Climate change adaptation.			
Priority guidelines for climate change adaptation:			
 Ensure that Hawaii's people are educated, informed, and aware of the impacts climate change may have on their communities. 			x



Table 5-2: Part III of the Hawai'i State Plan	S	NS	N/A
(2) Encourage community stewardship groups and local stakeholders to participat planning and implementation of climate change policies.	e in		x
(3) Invest in continued monitoring and research of Hawaii's climate and the impactimate change on the State.	cts of		x
(4) Consider native Hawaiian traditional knowledge and practices in planning for t impacts of climate change.	he		x
(5) Encourage the preservation and restoration of natural landscape features, suc coral reefs, beaches and dunes, forests, streams, floodplains, and wetlands th have the inherent capacity to avoid, minimize, or mitigate the impacts of clima change.	h as at ate		x
(6) Explore adaptation strategies that moderate harm or exploit beneficial opportu in response to actual or expected climate change impacts to the natural and b environments.	unities uilt		x
(7) Promote sector resilience in areas such as water, roads, airports, and public h by encouraging the identification of climate change threats, assessment of po- consequences, and evaluation of adaptation options.	ealth, tential X		
(8) Foster cross-jurisdictional collaboration between county, state, and federal age and partnerships between government and private entities and other nongovernmental entities, including nonprofit entities.	encies X		
(9) Use management and implementation approaches that encourage the continu collection, evaluation, and integration of new information and strategies into r and existing practices, policies, and plans.	al new		x
(10) Encourage planning and management of the natural and built environments the effectively integrate climate change policy.	nat		x
Discussion: The Proposed Action will support the priority guideline policies and objectives related to climate change adaptation. The Proposed Action serves to provide effective and efficient treatment of wastewater for the Nā'ālehu community. The various improvements will be primarily funded by the County and may also utilize federal funds through the State of Hawai'i Department of Health (DOH) Clean Water State Revolving Fund (CWSRF) Program.			

5.1.2 State Functional Plans

The Hawai'i State Plan directs appropriate State agencies to prepare Functional Plans which address statewide needs, problems, and issues, and recommend policies and actions to mitigate those problems. The Functional Plans are prepared to further define and implement statewide goals, objectives, policies, and priority guidelines contained in the Hawai'i State Plan. Thirteen Functional Plans were prepared to implement the State Plan provisions in the areas of agriculture, conservation lands, education, employment, energy, health, higher education, historic preservation, housing, human services, recreation, tourism, and transportation.

	Table 5-3: Hawai'i State Functional Plans	S	NS	N/A
1	Agricultural State Functional Plan (1991)			
Purpo	se: Continued viability of agriculture throughout the State.			X



Table 5-3: Hawai'i State Functional Plans	S	NS	N/A
Discussion: The Proposed Action is not directly appliable to the Agricultural State Functional F	่ ขอก.		
2 Conservation Lands State Functional Plan (1991)			
Purpose: Addresses issues of population and economic growth and its strain on current natural resources; broadening public use of natural resources while protecting lands and shorelines from overuse; additionally, promotes the aquaculture industry.	x		
Discussion: The Proposed Action will support the Conservation Lands State Functional Plan. T will not be applicable to the use of Conservation lands in the State. As previously discussed, th located about two miles from the shoreline and, as such, the project improvements will not aff	he Prop e Propo ect the	oosed A sed Act shorelir	ction ion is 1e.
Durnesse Improvements to House?'/ educational surriculum, quality of educational staff			
and access to adequate facilities.			X
Discussion: The Proposed Action is not directly applicable to the Education State Functional P	lan.		
4 Employment State Functional Plan (1990)			
Purpose: Improve the qualifications, productivity, and effectiveness of the State's workforce through better education and training of workers as well as efficient planning of economic development, employment opportunities, and training activities.			x
Discussion: The Proposed Action is not directly applicable to the Employment State Functiona	I Plan.		
5 Energy State Functional Plan (1991)			
Purpose: Lessen the reliance on petroleum and other fossil fuels in favor of alternative sources of energy so as to keep up with the State's increasing energy demands while also becoming a more sustainable island state; achieving dependable, efficient, and economical statewide energy systems.			x
Discussion: The Proposed Action is not directly applicable to the Energy State Functional Plan	i.		
6 Health State Functional Plan			
Purpose: Improve the health care system by providing for those who do not have access to private health care providers; increasing preventative health measures; addressing 'quality of care' elements in private and public sectors to cut increasing costs.			x
Discussion: The Proposed Action is not directly applicable to the Health State Functional Plan			
7 Higher Education Functional Plan (1984)			
Purpose: Prepare Hawai'i's citizens for the demands of an increasingly complex world through providing technical and intellectual tools.			x
Discussion: The Proposed Action is not directly applicable to the Higher Education Functional	Plan.		
8 Historic Preservation State Functional Plan (1991)			
Purpose: Preservation of historic properties, records, artifacts and oral histories; provide public with information/education on the ethnic and cultural heritages and history of Hawai'i			x
Discussion: The Proposed Action is not directly applicable to the Historic Preservation State F	unction	al Plan.	
9 Housing State Functional Plan (1989)			
Purpose: Provide affordable rental and for-sale housing; increase homeownership and amount of rental housing units; acquiring public and privately-owned lands for future residential development; maintain a statewide housing data system			x
Discussion: The Housing State Functional Plan is not directly applicable to the Proposed Action	n.		



Table 5-3: Hawai'i State Functional Plans	S	NS	N/A
10 Human Services State Functional Plan (1991)			
Purpose: Refining support systems for families and individuals by improving elderly care, increasing preventative measures to combat child/spousal abuse and neglect; providing means for `self-sufficiency'			x
Discussion: The Proposed Action is not directly applicable to the Human Services State Fun	ctional Pla	an.	
11 Recreation State Functional Plan (1991)			
Purpose: Manage the use of recreational resources via addressing issues: (1) ocean and shoreline recreation, (2) mauka, urban, and other recreation, (3) public access to shoreline and upland recreation areas, (4) resource conservation and management, (5) management of recreation programs/facilities/areas, and (6) wetlands protection and management.			x
Discussion: The Proposed Action is not directly applicable to the Recreation State Functiona	ıl Plan.		
12 Tourism State Functional Plan (1991)			
Purpose: Balance tourism/economic growth with environmental and community concerns; development that is cognizant of the limited land and water resources of the islands; maintaining friendly relations between tourists and community members; development of a productive workforce and enhancement of career and employment opportunities in the visitor industry.			x
Discussion: The Proposed Action is not directly applicable to the Tourism State Functional F	'lan.		
13 Transportation State Functional Plan (1991)			
Purpose: Development of a safer, more efficient transportation system that also is consistent with planned physical and economic growth of the state; construction of facility and infrastructure improvements; develop a transportation system balanced with new alternatives; pursue land use initiatives which help reduce travel demand.			x
Discussion: The Proposed Action is not directly applicable to the Transportation State Funct	ional Plar	า.	

5.1.3 State Land Use Law, Chapter 205, Hawai'i Revised Statutes

The State Land Use Law, Chapter 205, HRS, is intended to preserve, protect and encourage the development of lands in the State for uses which are best suited to the public health and welfare of Hawai'i's people. Under Chapter 205, HRS, all lands in the State are classified into four land use districts by the State Land Use Commission (LUC): Urban, Agricultural, Conservation, and Rural. The State Land Use Commission (LUC) is responsible for preserving and protecting Hawai'i's lands and encouraging those uses to which lands are best suited. Permitted uses within the districts are prescribed under Title 12, Chapter 205 (Land Use Commission), HRS, and the State Land Use Commission's Administrative Rules prescribed under Title 15, Subtitle 3, Chapter 15 HAR.

Discussion:

The State Land Use District Map for the island of Hawai'ī designates the project area as part of the Urban and Agricultural District. The existing and planned use are a permitted use for areas within the State Land Use Urban District. A Special Permit approved by the East Hawai'i Planning Commission is required for proposed uses within the State Land Use Agricultural District.



5.2 **County of Hawai'i Land Use Plans and Policies**

5.2.1 County of Hawai'i General Plan

The County of Hawai'i's General Plan is the blueprint that guides the long-term development of Hawai'i Island. It considers the needs of the entire island, and provides a sound growth strategy that directs future opportunities related to land use, zoning amendments and capital expenditures. The General Plan strives to position Hawai'i Island for economic progress while preserving the environment and strengthening community foundations. The current General Plan was adopted in 2005 and has guided County structure and future growth.

In September 2023, the Planning Department issued the first draft General Plan 2045. The purpose of the draft was to solicit feedback from the public, organizations, and agencies. The gathered information will be complied to form the basis for revisions to the draft General Plan prior or submission to the Planning Commission and the County Council. To date, since that process has not been completed, the existing General Plan adopted by the Hawai'i County Council in February 2005 (amended in December 2006), serves as the overall planning document outlining the long-range comprehensive development of Hawai'i Island. Since review of the General Plan 2040 has not been completed, the 2005 General Plan will be used for analysis. See below for further discussion related to the latest version of the General Plan. The Planning Department is in the process of revising the Draft General Plan 2045. A Final Recommended Draft Plan shall be released to the public in July 2024.

The Draft General Plan 2045 vision, goals, and objectives provide a high-level integrated direction for the community and a holistic perspective. The goals are divided into groups which together encompass the three sustainability pillars of community, economy, and environment. These groups include: 1) addressing climate change for island-wide health, 2) sustainable development and resilient communities, 3) thriving, diverse, and regenerative economy, and 4) collaborative biocultural stewardship.

Regulatory implementing actions are one of two types of approaches used in the General Plan to pursue the vision, goals, and objectives. Regulatory actions are controlling in that they define boundaries, development parameters, and measures intended to implement goals or objectives. The three regulatory implementing actions include: future land use maps, policies, and standards. In the 2045 General Plan, future land use maps, policies, and standards are specific to the actions through which zoning ordinances, subdivisions, and public improvements or projects are initiated or adopted because, as they must conform to and implement the general plan in accordance with the County Charter §3-15.

The second approach of implementing actions is not regulatory or controlling and requires subsequent decisions and/or the allocation of resources. The three types of non-regulatory implementing actions include: programs, projects, and interagency coordination.

A discussion of consistency with the goals, policies, and standards of the currently available Draft General Plan is provided in Table 5-5 below.



Table 5-5: County of Hawai'i Draft General Plan 2045	S	NS	N/A
Climate Change Goal			
Ensure a just transition to a climate resilient island by addressing the causes and impacts through incorporating equitable climate mitigation and adaptation priorities into policies, p infrastructure, and decision-making.	of clima rograms	te chang s,	e
Objective 1 : Ensure that climate actions are equitable and uplift historically marginalized communities.	and dis	advanta	ged
Policies			_
1.1 Prioritize and support existing community-led organizations, businesses, and programs through County purchasing and procurement policies.			x
1.2 Prioritize and support Low- and Moderate-Income (LMI) communities through tax incentives, grants, and financial support provided to community members			х
1.3 Prioritize projects in communities that experience disproportionate impacts of climate disasters.			x
1.4 Consider financial, time barriers, geographic constraints, and language accessibility when conducting community outreach.			x
1.5 Improve communication of climate risks and opportunities for adaptation.			х
1.6 Help communities become fire adapted as they prepare for climate change.			Х
1.7 Expand urban forestry benefits to disadvantaged communities.			Х
1.8 Support social science research and applications to help address environmental justice and intergenerational equity.			x
1.9 Include native Hawaiian traditional knowledge and practices in planning for the impacts of climate change.			x
Objective 2:			
Achieve a 100 percent renewable percent fleet by 2025 and 100 percent renewable	olo grou	nd	
transportation by 2045.	ne grou	nu	
Support the achievement of 70 percent renewable energy for the electricity sector by 2030 from renewables and 30 percent from efficiency, and 100 percent by 2045.), with 4	40 perce	nt
Policies			
2.1 Create codes and efficiencies that integrate affordable housing and carbon neutrality.			x
2.2 Require energy efficiency designs in all new County facilities and upgrade existing facilities with energy-efficient systems.			х
2.3 Support building code updates that incentivize energy-efficient designs and climate-neutral building methods and materials.			x
2.4 Implement the use of technologies, techniques, and materials in building design, construction, and removal that minimize the ecological footprint over the life cycle of the structure.			x
2.5 Encourage and incentivize large developments to meet energy sustainability certification standards.			x
2.6 Strive for energy sustainability certification for new County buildings or when renovating existing buildings for County use.			x



2.7 Prioritize operate officient designs, operate officient systems, and waste		
reduction and/or reuse at County facilities.		x
2.8 Evaluate Capital Improvement Projects for energy efficiency and carbon reduction, including the cost savings related to improved design		x
2.9 Partner with government, private and nonprofit agencies, and other stakeholders to develop comprehensive and coordinated strategies promoting energy and water conservation to strive for climate change resilience.		x
2.10 When evaluating public investments, including acquisition, siting, and design, consider the potential of natural areas for carbon sequestration and provide climate adaptation and mitigation opportunities.		x
2.11 Integrate urban forestry into all scales of planning.		Х
2.12 Develop carbon-emission standards and an incentive program aimed at achieving County. carbon-emission goals.		x
2.13 Mandate reporting of energy use of all County facilities annually.		Х
2.14 Program the upgrading of the vehicle fleet and equipment to zero emissions.		Х
2.15 The Mass Transit Agency shall operate a zero emissions fleet by 2035.		Х
2.16 Prioritize the installation of electric vehicle chargers at community facilities, for both County vehicles and public parking.		x
2.17 Mandate reporting of all new County vehicle purchases including average mpg, emissions equivalent per gallon of fuel (or kWh), and estimated annual operation and maintenance costs.		x
Objective 3 : Improve the identification of climate change threats, assessment of potenti evaluation of adaptation options.	al conseque	ences, and
Policies		
3.1 Use accurate and up-to-date scientific predictions and observations related to climate change impacts to guide adaptation policy and future land use decisions.	x	
3.2 Support local and regional climate change modeling and monitoring programs.		Х
3.3 Collaborate with government, private and nonprofit agencies, and other stakeholders to monitor impacts that may be specific to Hawai'i County due to its unique exposure to climate change and sea level rise impacts.		x
3.4 Improve assessments of climate change threats and potential consequences to		х
determine specific geographic areas impacted and projected financial cost.		
determine specific geographic areas impacted and projected financial cost.3.5 Climate change adaptation shall be considered in County budgetary, land use, and other decision-making processes.		x
 determine specific geographic areas impacted and projected financial cost. 3.5 Climate change adaptation shall be considered in County budgetary, land use, and other decision-making processes. 3.6 Implement a sea-level rise planning and policy benchmark of 4 feet by 2100 and 6 feet for public infrastructure projects and other projects with a low tolerance for risk. 		x
 determine specific geographic areas impacted and projected financial cost. 3.5 Climate change adaptation shall be considered in County budgetary, land use, and other decision-making processes. 3.6 Implement a sea-level rise planning and policy benchmark of 4 feet by 2100 and 6 feet for public infrastructure projects and other projects with a low tolerance for risk. 3.7 Prioritize the preservation and restoration of natural landscape features, such as coral reefs, beaches and dunes, forests, streams, floodplains, and wetlands, that have the inherent capacity to avoid, minimize, or mitigate the impacts of climate change. 		x x x
 determine specific geographic areas impacted and projected financial cost. 3.5 Climate change adaptation shall be considered in County budgetary, land use, and other decision-making processes. 3.6 Implement a sea-level rise planning and policy benchmark of 4 feet by 2100 and 6 feet for public infrastructure projects and other projects with a low tolerance for risk. 3.7 Prioritize the preservation and restoration of natural landscape features, such as coral reefs, beaches and dunes, forests, streams, floodplains, and wetlands, that have the inherent capacity to avoid, minimize, or mitigate the impacts of climate change. 3.8 Develop adaptation strategies (e.g., protection, accommodation, managed retreat, and preservation) for capital improvements to assure that the project's useful life and service expectations can be met in the face of projected climate change impacts. 	x	x x x



3.10 Partner with communities to develop adaptation strategies (e.g., protection, accommodation, managed retreat, and preservation) for vulnerable areas including		x
conducting vulnerability assessments and assessing land use and land availability.		
3.11 Partner with government (e.g., State Office of Planning and Sustainable		
Development [OPSD]), private and nonprofit agencies, and other stakeholders to		v
analyze conservation buffers to accommodate shifting native habitats impacted by		^
climate change particularly wetlands and high-elevation forests		

Discussion: The Proposed Action will not directly impact the goals, objectives, and policies outlined in the County of Hawai'i Draft General Plan regarding climate change. The design of the WWTP will not discharge to surface water sources and will take into consideration the potential for increased storm intensity. Drainage improvements will be implemented within the WWTP project site to manage stormwater during intense weather events, ensuring the facility's resilience to changing storm patterns.

In the short-term, the construction phase may result in temporary greenhouse gas emissions due to heavy equipment operations and the transportation of supplies to the WWTP project site and along the streets for the new collection system. However, these emissions will be temporary during the construction period. Upon completion of the construction, emissions and greenhouse gases should return to current conditions. Land Use Goal

Strategically apply progressive land use strategies incorporating indigenous and contemporary knowledge and place based practices to direct and manage growth for the health and safety of our communities.

Objective 4: Maintain community character and land use compatibility.

Policies		
4.1 The development of commercial facilities should be designed to reflect the character of the community while providing desired services and mitigating impacts on the surrounding environment.		x
4.2 In those cases where provisions of the zoning and subdivision code are inconsistent with the character of surrounding neighborhoods, variances, or Planned Unit Developments (PUD) that maintain consistent village/town character should be encouraged.		x
4.3 Zoning, subdivision, and other applicable ordinances shall provide for and protect open space areas.	x	
4.4 Support mechanisms, such as Planned Unit Development and Cluster Plan Development, that cluster density to preserve open space, recreational areas, or scenic viewsheds.	x	
4.5 Discretionary permit applications for uses that may impact view planes to and along the coastline, and areas of natural beauty should take into consideration visual impact assessments and propose conditions to mitigate scenic impacts where appropriate.	x	
4.6 In the review of discretionary permits, consider land use compatibility to ensure proximate compatible and complementary uses and appropriate mitigation measures.	x	
4.7 Power distribution should be placed underground when and where practical.		X
4.8 Encourage developers of new urban areas to place utilities underground.		X



		1	
4.9 Route selection for high-voltage transmission lines should include consideration for setbacks from major thoroughfares and residential areas. Where feasible, delineate energy corridors for such high-voltage transmission lines.			x
Objective 5 : Increase the integration of natural systems planning.			
Policies			
5.1 Protect and enhance Hawai'i's shoreline, open spaces, and scenic resources.	X		
5.2 During discretionary permit applications, the Planning Director may require a pedestrian, equestrian, and/or bicycle path when it is possible and safe to connect to existing or future drainage or active living corridors.			X
5.3 Proposed discretionary permits for large development projects (200+ units) in the North Kohala, South Kohala, North Kona, South Kona, and Ka'ū Districts should be designed to be as water neutral as reasonably possible through water conservation, recharge, and reuse measures to reduce the water footprint.			x
Objective 6 : Increase equitable planning and decision-making processes.			
Policies			
6.1 Ensure there is necessary and adequate on-site infrastructure for development projects.	X		
6.2 Ensure affordable housing requirements that meet the demand created by the development.		×	(
6.3 The County may impose incremental and conditional zoning that would be based on performance requirements that focus on addressing the impacts of the proposed development.	x		
Objective 7 : Reduce the threat to life and property from natural hazards and disaster environmental laws in order to conserve natural resources and the shoreline.	s. Enfoi	rce zoning a	and
Policies			
7.1 Enact additional land use and building structure regulations in areas vulnerable to severe damage due to the impact of waves or inundation.	X		
7.2 Review land use policy as it relates to floodplains, high surf, and tsunami hazard areas.	х		
7.3 Consider natural hazards in all land use planning and permitting.	X		
7.4 Discourage intensive development in areas of high volcanic hazard.			X
7.5 Discourage public investment/infrastructure that supports increases in density, while allowing such investment to support existing residents and facilities.	x		
7.6 Reduce development intensity in identified high-risk hazard areas.	X		



7.7 Incorporate hazard mitigation strategies into policies and planning decisions using the most conservative models in delineating hazard areas.	x		
Objective 8 : Increase the use of Smart Growth principles to focus development within e	xisting u	ırban cen	ters.
Policies			
8.1 Encourage flexibility in the design of residential sites, buildings, and related facilities to achieve a diversity of socio-economic housing mix and innovative means of meeting the market requirements.			x
8.2 Prioritize increase in density, rehabilitation, and redevelopment within existing zoned urban areas already served by basic infrastructure, or close to such areas.			x
8.3 Incentivize rehabilitation and adaptive reuse of existing buildings rather than demolition in urban areas characterized by vacant, abandoned, and underutilized older buildings.			x
8.4 Encourage the rehabilitation and/or utilization of maximum density in multi-family residential areas.			x
8.5 Rezonings that promote infill are encouraged and should be conditioned to ensure connectivity to the surrounding developments and, where applicable, to provide mixed-use opportunities to make the area more pedestrian-oriented.			x
8.6 The establishment of urban types of zoning may include additional acreages to account for acreages utilized for public benefits, such as historic sites, public access, parks, and open space.			x
8.7 Within the "high- and medium-density" area, commercial development shall be focused on major streets, while interior blocks should be zoned primarily for small lot single-family and multifamily residential use.			x
8.8 Focus on medium and high-density residential and commercial uses in communities that can sustain a higher intensity of uses and where these residential and commercial uses are consistent with the existing town character.			x
8.9 Support the rezoning of land to multiple residential near places of employment, shopping facilities, educational, recreational, and cultural facilities, and public facilities and utilities.			x
8.10 Development of TODs and TNDs are encouraged within locations of the centers shown on the General Plan Land Use Map. These locations are approximate and become fixed during rezoning.			x
8.11 Plan for and identify appropriate areas for business incubation/innovation districts and industrial/business parks.			X
8.12 Urban renewal, rehabilitation, and/or redevelopment programs should be undertaken in cooperation with communities, businesses, and governmental agencies.			X
8.13 Support master planning of present and proposed public and private institutions with an emphasis on TOD, affordable housing, and mixed-use development.			X



8.14 Low- to medium-density residential development and/or low-impact office uses within urban areas should serve as transitional densities between lower-density neighborhoods and more intensive commercial and residential uses.	x	Ľ
8.15 Encourage the use of more innovative types of housing development with respect to geologic and topographic conditions, such as zones of mix and cluster and planned unit developments.	x	Ľ.
8.16 Lots within proposed single-family residential subdivisions should not have direct vehicular access from major collector streets or larger roadway types.	x	ζ
8.17 Ensure flag lots are designed to support optimal density.	X	(
8.18 Large, oversized blocks in new subdivisions should be avoided in favor of smaller blocks and enhanced pedestrian networks. The determination of block size should be based on land use and the urban or rural character of the area.	x	[
8.19 Promote the redevelopment of aging and high-vacancy shopping centers, or strip- type developments into mixed-use developments with housing and public recreation facilities.	x	ζ.
8.20 Encourage the concentration of commercial uses within and surrounding a central core area adequately served by transportation, utilities, and other essential infrastructure.	X	Ľ
8.21 Infrastructure and design elements shall be incorporated into the review of commercial developments.	X	Ľ
8.22 Encourage a mix of uses near affordable housing and access to commercial and recreational opportunities.	X	<u> </u>
8.23 Industrial and commercial mixed-use districts may be provided in urban centers.	X	(
8.24 Distribution of commercial areas shall meet the demands of neighborhood, community, and regional needs.	X	Ľ
8.25 Discourage strip or spot commercial development on the highway outside of the Urban Growth Areas.	X	ζ
8.26 Discretionary permit applications for regional retail uses, including big box and regional shopping centers located adjacent to areas designated for low-density residential and rural uses, should be buffered to mitigate impacts.	x	r *
8.27 Encourage small-scale manufacturing and processing within retail establishments that enhance and are consistent with the surrounding community.	X	Ľ
8.28 Support the flexible design of commercial spaces to allow for transitional uses that serve the evolving needs of its users.	X	<u> </u>
8.29 Discretionary permit applications for new commercial developments adjacent to or within existing industrial designated lands shall be reviewed for the criteria of conversion of industrial lands.	X	<u>,</u>



8.30 Industrial development shall be in areas adequately served by transportation, utilities, and other essential infrastructure or adjacent to unique resources/projects.		X
8.31 Support the creation of industrial uses in appropriate locations as part of mixed-use districts/developments.		X
8.32 Support Industrial Project District zoning and flexibility of uses and lot sizes, depending on the needs of the industries and the communities.		X
8.33 Encourage Industrial Project Districts and Innovation Centers within the Urban Growth Area.		X
8.34 Industrial-commercial mixed-use districts should serve as transitional areas.	X	
8.35 Support land uses that locate industrial and warehouse sites near major transportation corridors and airports and harbors.	x	
8.36 Support the development of cottage recycling operations near transfer stations and County landfills.		x
8.37 Industrial uses may be permitted outside Urban Growth Areas through Special Permits when there is a clear community benefit or consistent with County, State, and Federal sustainability objectives.	x	
8.38 Encourage the rehabilitation of existing service-oriented industrial areas.	X	
8.39 Mitigate impacts of industrial development on surrounding uses by requiring landscaping, open spaces, buffer zones, and other appropriate conditions.	x	
8.40 Future land uses in the vicinity of industrial areas, including airports, should have an adequate open space buffer and/or be compatible with the anticipated aircraft noise exposure levels for that vicinity.		X
8.41 Resort development shall be in areas adequately served by transportation, utilities, and other essential infrastructure.		x
8.42 Promote and prioritize the rehabilitation and the optimum utilization of resort areas that are presently serviced by basic facilities and utilities before allowing new resorts.		x
8.43 Coastal resort developments shall provide public access to and parking for beach and shoreline areas.		x
8.44 The development or designation of new resort areas should complement the character of the area; protect the environment and natural beauty; respect existing lifestyles, cultural practices, and cultural resources; and provide shoreline public access.		X
8.45 Do not allow new Resort (V) zoning development along the ocean side of Ali'i Drive.		X
8.46 Resort development should be in balance with the social and physical goals as well as the economic desires of the residents of the area.		x
8.47 Encourage new developments to be water neutral and balance water supply and demand.		x



8.48 Provide for the establishment of new Retreat Resort areas in appropriate locations.		x
8.49 On-site affordable housing and workforce units shall not be included in visitor unit counts for existing and new resort developments		X
Objective 9 : Maximize the alignment and use of Rural designated lands to preserve rural	characte	r and lifestyle.
Policies		
9.1 Support the rezoning and State Land Use reclassification to Rural in alignment with the General Plan Rural designation.		X
9.2 Support reclassification/rezoning of appropriate General Plan Rural designated areas where an intermediate land use and a well-defined buffer between Urban and Productive Agricultural areas are consistent with the surrounding uses and rural character.		X
9.3 Support General Plan amendments and rezoning applications for the development of new Rural Neighborhood Centers with adequate infrastructure, as necessary, in or near presently underserved subdivisions, beginning with those experiencing higher rates of population growth.		x
9.4 Rural-style residential-agricultural developments, such as new small-scale rural communities or extensions of existing rural communities, should be incentivized to cluster in appropriate locations.		X
9.5 Support the development of small-scale visitor accommodations with heritage, agriculture, wellness, or similar themes in rural areas and near points of interest.		X
9.6 Provide flexibility in discretionary permit applications to maintain health and safety for rural small-scale visitor accommodations not serviced by public infrastructure.		X
Objective 10: Support the active use of Productive Agricultural lands.		
Policies		
10.1 Development in Productive Agriculture and Extensive Agriculture areas should include agricultural uses, related economic infrastructure and cottage industries, compatible renewable energy, open area recreational uses, and community facilities.		x
10.2 Special permit applications within Productive Agriculture designated land should support primary agriculture use	x	
10.3 Encourage buffer zones or compatible uses between Productive Agriculture and adjacent uses of land.	x	
10.4 Preserve agricultural character, including the open space preserved by agricultural land.	x	
10.5 Support the development of small-scale visitor accommodations that directly promote the agriculture industry, health and wellness industry, or are near points of interest		x



10.6 Any subdivision or farm labor housing complex developed on Productive Agricultural Lands should be clustered to minimize impact.)	X
10.7 Encourage and aid the agricultural industry in continuing to provide farm labor housing.		>	X
10.8 Encourage the use of agriculture, ranch, and forestry land preservation programs.		,	x
10.9 Promote the preservation and restoration of indigenous agricultural systems.	X		
10.10 Provide flexibility to allow for adjacent compatible uses for large-scale agriculture with industrial components, such as carbon sequestration.	x		
Discussion: The Proposed Action will support the goals, objectives, and policies out Hawai'ī Draft General Plan relating to Land Use.	lined in	the County	of
The existing land use in the Project Area includes residential, agricultural, and undeveloped Area is located within the Urban and Agricultural State Land Use Districts (SLUD). The located within the Agricultural SLUD. As such, the Proposed Action would be required regulations set forth in the State Land Use Law (HRS, Chapter 205).	oped lan WWTP d to co	d. The Proje site is entire mply with th	ect ely he
Furthermore, the County of Hawai'i General Plan calls for the preparation of commun (CDPs) "to translate the broad General Plan statement to specific actions as they apply to areas." The Ka'ū CDP is one of nine CDPs for the County of Hawai'ī. Section 5 of the CDP pr in the infrastructure, facilities, and services, including Section 5.8 applicable to Environment states:	ity deve o specifi ioritizes tal Mana	lopment plan c geographic improvemen agement whic	ns cal nts ch
"Policy 120 Extend the primary wastewater collection lines in Pāhala and Nā'ālehu so t projects can connect wastewater systems built for new subdivisions to the County system	that infil ns."	ll developme	ent
The Proposed Action has been designed not to preclude expansion to accommodate th Nā'ālehu community. Future subdivisions would be accommodated, as capacity allows, served basis.	ne future on a fir	e needs of th rst-come, firs	he st-
Transportation Goal			
Each community is connected by a multimodal and modernized transportation network t for safe, efficient, and comfortable movement of people and goods.	hat prov	vides a syste	em
	date pla	nned growth	۱.
Objective 11: Achieve a transportation system that is consistent with and will accommo	·		
Objective 11: Achieve a transportation system that is consistent with and will accommo Policies			
Objective 11: Achieve a transportation system that is consistent with and will accommon Policies 11.1 Encourage transportation systems that serve to accommodate the present and future development needs of communities.		, , , , , , , , , , , , , , , , , , ,	×
Objective 11: Achieve a transportation system that is consistent with and will accommon Policies 11.1 Encourage transportation systems that serve to accommodate the present and future development needs of communities. 11.2 Encourage safe and convenient use of low-cost, energy-efficient, non-polluting means of transportation.		>	x K



11.4 Transportation and land use planning shall be integrated to optimize the use, efficiency, and accessibility of existing mass transportation systems and future demand.	x
11.5 Establish a framework of transportation facilities that will influence desired land use and promote multimodal options.	x
11.6 Provide for present traffic and future demands, including the programmed development of mass transit programs for high-growth areas by both the private and public sectors.	x
11.7 Implement procedures for County departments to collaborate on defining short- and long-term transportation CIP projects in terms of scope, timing, proposed funding, and project performance measures required to optimally achieve transportation ambitions stated in County plans, standards, and laws.	x
11.8 Prioritize CIP investments consistent with General Plan goals and objectives and demands for roadway repaving, rehabilitation, and reconstruction. This prioritization should align with the six-year planning horizon and should include a long-term roadmap for future investments.	x
11.9 Support designing all transportation facilities, including airport, harbor, mass- transit stations, etc., to reflect local and/or Hawaiian architecture.	x
11.10 Identify and evaluate transportation strategies to address energy and climate issues.	x
11.11 Prioritize public and private transportation investments to expand the multimodal transportation system.	x
11.12 Require new developments to contribute their pro rata share of local and regional infrastructure costs.	x
11.13 There shall be coordinated planning of transportation systems for the funding of projects in areas of anticipated growth and to meet program goals of the other elements such as historic, recreational, environmental quality, and land use.	x
Objective 12: Increase transportation connectivity.	
Policies	
12.1 Ensure Native Hawaiian access rights are clearly expressed in County code, policies, and procedures.	x
12.2 Programmatically support the open space network concept with a methodology that includes criteria for establishing County department and other agency responsibilities, mapping requirements, financing strategies for implementation and maintenance, and standards for facilities that enhance the community experience.	x
12.3 Prior to disposing of, leasing, or transferring public lands through County Property Management procedures, the County shall assess, document, and protect access to existing active living corridors that are located on County-owned parcels.	x



12.4 Land use applications shall identify as early as possible any existing or potential active living corridors that should be incorporated into the County's open space network.		x
12.5 Ensure that existing active living corridors that are publicly owned or available by easement are properly identified and that their access elements are secured and documented.		
a) Primary examples include but are not limited to historic trails and roads, roads- in-limbo, 'paper roads', former sugar cane roads, train infrastructure remnants (Rails to Trails), and pedestrian and bicycling paths.		x
b) "Acceptance" by the County of the responsibilities detailed in the grant of easements should require County Council action and a dedicated funding source.		
12.6 Provide public pedestrian access opportunities to scenic places and vistas.		X
12.7 Establish public access to historic and modern active living corridors and facilities that provide an island-wide route and connect to major destinations.		x
Objective 13: Increase mass transit ridership by 50 percent.		
13.1 Ensure transit routes connect with other modes of active transportation consistent with the County Street Design Manual.		x
13.2 Provide more equitable mobility for youth, low-income, elderly, and people with disabilities.		x
13.3 Maximize regular and paratransit service to the following:		
a) Town centers, commercial districts, and employment centers.		
b) Airports and cruise ship terminals.		x
c) University and adult education centers.		
d) Accommodate school schedules such as after-school activities and sports.		
13.4 Bus maintenance facilities shall be developed at or near appropriate transit hubs.		x
13.5 Adopt Hub and spoke system including alternative first and last mile or door- to-door services.		x
13.6 Transit infrastructure (e.g., bus stops, bus pullouts, waiting benches and shelters, and signs) shall be adequate and upgraded along existing and future transit routes.		x
13.7 Data shall be collected and analyzed to optimize mass transit planning, operation, and overall performance.		x
13.8 Improve and expand public transportation in communities with the highest socioeconomic needs.		x
	<u> </u>	1



13.9 The County's public transit system assets shall be available to assist in transportation in emergency situations.		X
Objective 14: Reduce vehicle miles traveled (VMT).		
14.1 Encourage collaboration between the Planning Department, the Department of Public Works, the Department of Parks and Recreation, and the Mass Transit Agency to define the scope and priority of capital investment projects that achieve active transportation objectives and goals.		x
14.2 Increase arterial capacity through prioritization of alternative means of transportation, such as mass transit, bicycle, and pedestrian systems.		X
14.3 Incorporate bicycle routes, lanes, and paths within road rights-of-way in conformance with The Bikeway Plan for the County of Hawai'i.		X
14.4 Increase mobility for minors, non-licensed adults, low-income, elderly, and people with mobility limitations through prioritization of alternative means of transportation.		x
14.5 Roadway designs and improvements made by the Department of Public Works shall accommodate pedestrian-friendly, multimodal design, and on-street parking evaluations, when practical and feasible.		x
14.6 Use traffic demand management to aid in reducing traffic congestion by targeting an increase of active transportation mode share to 10 percent (bicycling, walking, micromobility).		x
14.7 Apply zoning reviews for concurrency that achieve sustainability and demand management goals and that prioritize reduction in vehicle miles traveled over impacts to the level of service.		x
Objective 15: Achieve a transportation system that employs all modes of transportation	at a community sca	ale.
Policies		
15.1 Encourage flexibility in applying the County of Hawai'i Street Design Manual when necessary to preserve the rural character of an area while maintaining a pedestrian-friendly design and desired landscaping solutions.		x
15.2 Allow for private, non-dedicated cul-de-sac and alley designs that are in accordance with national road standards.		X
15.3 In planning, designing, and constructing new roadways or modernizing improvements, transportation agencies should balance the conservation of the area's natural, historic, and scenic qualities with transportation objectives for traffic speed, safety, and traffic calming.		x
15.4 Support and provide technical assistance to assist in the development of road improvement districts to finance road improvements.		X



15.5 Preserve the unique character of an area by allowing flexibility in existing roadway improvements and maintenance while seeking a pedestrian-friendly design and desired landscaping solutions.	x
15.6 Incentivize subdivision roadway connectivity.	X
15.7 A corridor planning/management program shall be maintained to help prioritize various active transportation projects.	x
Objective 16: Incorporate green infrastructure to reduce stormwater runoff.	
Policies	
16.1 Incorporate low-impact development (LID), green infrastructure strategies, and pollution prevention procedures to address drainage in roadway design and update the operation and maintenance of these solutions to retain integrity.	x
16.2 Prioritize roadway drainage improvements in flood-prone areas.	X
16.4 Maintain an Adopt-a-Street program to encourage civic participation where moderate landscaping and roadside cleaning can be done by community groups.	x
16.5 At a minimum, the County shall plan, site, and develop roads, bridges, and highways to:a) Protect areas that provide important water quality benefits or are particularly susceptible to erosion or sediment loss;	x
b) Limit land disturbance such as clearing, grading, and cut and fill to reduce erosion and sediment loss; andc) Limit disturbance of natural drainage features and vegetation, including mitigating impacts of stream crossings.	
Objective 17: Increase transportation safety for transportation's most vulnerable use fatalities.	ers and reduce traffic
Policies	
17.1 Human life and public health are prioritized within all aspects of the transportation system.	x
17.2 Solutions should be prioritized in areas with the most vulnerable populations.	X
17.3 Engage communities in defining issues and developing solutions for their community, with a particular focus on engaging disadvantaged and vulnerable populations.	x
17.4 Prioritize interdepartmental coordination and accountability of traffic safety through education, enforcement, engineering, encouragement, and evaluation. Focus on policies, practices, staffing, and programs to improve road and pedestrian safety.	x



17.5 Incorporate traffic-calming features into arterial road and street designs to include vertical deflections, horizontal shifts, roadway narrowing, and closures to reduce speeding and increase safety. These may include techniques such as roundabouts, median barriers, speed humps, raised intersections, and other transportation industry practices.	x
17.6 Commit to an equitable approach and outcomes, including prioritizing engagement and investments in traditionally under-served communities and adopting equitable traffic enforcement practices.	x
17.7 Develop roadway standards to accommodate emerging technology for connected and automated vehicles.	x
17.8 Maintain dedicated roadway standards that are appropriate to roadway type and achieve active transportation and safety goals.	x
17.9 Engage and collaborate with the owners of private roads and local community groups to help identify and develop road management agreements that mitigate road closures to provide emergency.	x
Objective 18: Adequately maintain public transportation systems.	
Policies	
18.1 Maintain an Asset Management Program aimed at utilizing maintenance plans for pavement, bridges, and other road infrastructure to prolong the life of our transportation system as well as reduce its whole-life cost.	x
18.2 Maintain the unique features of historic bridges, while balancing safety needs and preserving historic and scenic character.	x
18.3 Prioritize the replacement of deficient and inadequate bridges and maintain pedestrian/bicycle access across bridges.	x
18.4 Design new bridges and bridge improvements to accommodate and not negatively impede identified scenic resources.	x
18.5 Evaluate freight routes identified in the State Freight Masterplan for required improvements to meet roadway standards.	x
Objective 19: Improve accessibility to airports, harbor systems, and support facilities.	
Policies	
19.1 Encourage the programmed improvement of existing terminals, including adequate provisions for control of pollution and appropriate and adequate covered storage facilities for agriculture products.	x
19.2 The State Department of Transportation should continue to implement its plans for transportation terminals and related facilities to promote and follow desired land	x



			-
19.3 Transportation terminals should be developed in conjunction with the different elements of the overall transportation system.			x
19.4 Encourage maximum use of the island's airport and harbor facilities.			x
19.5 Encourage the development, maintenance, and enhancement of Hilo and Kawaihae Harbors as detailed within the State's Hawai'i Commercial Harbors 2035 Master Plan.			x
19.6 Support the State's objectives to acquire rights within the runway clear-zones, limit heights within approach zones, and restrict noise-sensitive uses within designated noise contours determined by the State.			x
19.7 Future land uses in the vicinity of airports and harbors should have an adequate open space buffer and/or be compatible with the anticipated noise exposure and industrial nature in the vicinity.			x
19.8 Encourage pedestrian-oriented connectivity around harbors and small boat harbors.			X
19.9 Explore and encourage appropriate reuse of former airport facilities.			X
19.10 Encourage master planning of small boat harbors to accommodate commercial fishing and recreational fishing, tour boats, as well as residential and business activity, that balance economic vitality and environmental sensitivity.			x
Hawai'i Draft General Plan relating to transportation. Nā'ālehu is served by a network of r under the jurisdiction of the County and include Māmalahoa Highway and local roads in N low, reflecting the rural nature of the area. Limited vehicular traffic contributes to the low and a relatively peaceful road environment.	ed in the roadway ā`ālehu levels c	is gener of conges	of are ally tion
Public Utilities Goal			
Communities are adequately served by sustainable and efficient public infrastructure, utilities on existing and future growth needs, sound design principles, and effective maintenance	es, and practice	services l s.	based
Objective 20: Improve the efficiency, reliability, and sustainability of essential infrastruct	ture sys	tems.	
Policies			
20.1 Public utility facilities shall be designed at a scale that meets the needs of future development.	X		
20.2 Provide utilities and service facilities that minimize total cost to the public and effectively serve the needs of the community.	X		
20.3 Utility facilities shall be designed to complement adjacent land uses and minimize pollution or disturbance of the natural environment and natural resources.	X		
20.4 Improvement of existing utility services shall be encouraged to meet the needs of users.	X		



20.5 Encourage the clustering of developments to reduce the cost of providing utilities.	x	
20.6 Develop short- and long-range capital improvement programs and plans for public utilities within its jurisdiction that are consistent with the General Plan.	x	
Objective 21: Strive towards energy self-sufficiency.		
Policies		
21.1 Increase partnerships and interagency collaboration to ensure that energy facility production and distribution is adequate, efficient, and dependably available to each community to support present and future demands.		x
21.2 Promote and encourage the creation of a modern grid to support the use of distributed generation such as private photovoltaic systems connected to the grid.		x
21.3 The County shall remove barriers to energy systems that improve independence and resiliency, such as microgrids, combined heat and power (CHP), backup generation and storage, and other decentralized electricity systems.		x
21.4 Maintain tax incentives for renewable energy improvements and continue to revise incentives as energy technologies progress.		x
21.5 Continue to participate at the State level to provide feedback on all energy- related initiatives and proposed revisions to comprehensive Integrated Resource Plans.		x
21.6 Advocate to the Public Utility Commission (PUC) in support of the following types of strategies and initiatives:		
a) Programs and fee structures that promote renewable energy		
b) Consumer incentives to utilize renewable alternatives		X
c) Social Equity analysis of proposed energy projects to ensure residents are protected as energy consumers regarding rates, grid planning, utility compensation, and energy project siting		
Objective 22: Advance policies, programs, and initiatives for public and/or private invest and telecommunications infrastructure.	ment in t	proadband
22.1 Treat broadband access as a basic utility that is available to all communities.		x
22.2 Develop and support a program of free, public-use broadband services at appropriate County-owned facilities, mass transit facilities, and other community anchor institutions.		x
22.3 Collaborate with utility companies to incentivize the underground siting of electrical and telecommunications facilities within public rights-of-way.		x
22.4 Continuously improve the use of broadband communications and digital technology to educate and provide public services with a focus on digital access		x



22.5 Siting of new communications facilities shall comply with performance standards and site co-location as stated in HCC, Section 25-4-12 Note: HCC update will include details		x
22.6 Support projects that address service gaps in Hawai'i's broadband infrastructure.		x
22.7 Advocate for connectivity to businesses to protect the viability of businesses and the livelihoods of residents.		x
22.8 Promote and incentivize the landing of transpacific submarine fiber optic cables.		X
22.9 Alleviate barriers and assist broadband projects with navigating through the regulatory permitting process.		x
22.10 Maintain and improve cyber security and informational security of telecommunication facilities.		x
22.11 Advocate for service diversity, redundant network capacity, and provide improved communications to outlying rural areas and other underserved or unserved communities.		x
22.12 Plan for broadband infrastructure to support smart grid development.		x
Objective 23: Increase the protection of existing and potential sources of drinking water		
Objective 23: Increase the protection of existing and potential sources of drinking water Policies	•	
Objective 23: Increase the protection of existing and potential sources of drinking water Policies 23.1 All public water systems shall be designed and built to the Department of Water Supply dedication standards. All other systems shall meet all relevant health and safety regulations and be designed and constructed by a licensed engineer		x
Objective 23: Increase the protection of existing and potential sources of drinking water Policies 23.1 All public water systems shall be designed and built to the Department of Water Supply dedication standards. All other systems shall meet all relevant health and safety regulations and be designed and constructed by a licensed engineer 23.2 Water sources shall be adequately protected to prevent depletion and contamination from natural and man-made occurrences or events.	X	x
Objective 23: Increase the protection of existing and potential sources of drinking water Policies 23.1 All public water systems shall be designed and built to the Department of Water Supply dedication standards. All other systems shall meet all relevant health and safety regulations and be designed and constructed by a licensed engineer 23.2 Water sources shall be adequately protected to prevent depletion and contamination from natural and man-made occurrences or events. 23.3 A coordinated effort by County, State, and private interests shall be developed to identify sources of additional water supply and be implemented to ensure the development of sufficient quantities of water for existing and future needs of high-growth areas and agricultural production.	X	x x
Objective 23: Increase the protection of existing and potential sources of drinking water Policies 23.1 All public water systems shall be designed and built to the Department of Water Supply dedication standards. All other systems shall meet all relevant health and safety regulations and be designed and constructed by a licensed engineer 23.2 Water sources shall be adequately protected to prevent depletion and contamination from natural and man-made occurrences or events. 23.3 A coordinated effort by County, State, and private interests shall be developed to identify sources of additional water supply and be implemented to ensure the development of sufficient quantities of water for existing and future needs of high-growth areas and agricultural production. 23.4 Installation or rehabilitation of water distributions shall be sized to adequately meet fire protection.	X	x x x x
Objective 23: Increase the protection of existing and potential sources of drinking water Policies 23.1 All public water systems shall be designed and built to the Department of Water Supply dedication standards. All other systems shall meet all relevant health and safety regulations and be designed and constructed by a licensed engineer 23.2 Water sources shall be adequately protected to prevent depletion and contamination from natural and man-made occurrences or events. 23.3 A coordinated effort by County, State, and private interests shall be developed to identify sources of additional water supply and be implemented to ensure the development of sufficient quantities of water for existing and future needs of high-growth areas and agricultural production. 23.4 Installation or rehabilitation of water distributions shall be sized to adequately meet fire protection. 23.5 Ensure the highest quality of water is reserved for the most valuable end-use.	x x	x x x x


23.7 Promote best practices in sustainable water collection and use for private water systems	x	
23.8 Water system improvements, including exploratory wells, shall correlate with the County's desired land use development pattern.		x
23.9 The Department of Water Supply shall prioritize infill development and focus source development to serve designated Urban Growth Areas.	x	
23.10 Water demand projections shall include all consumptive and non-consumptive demands.	x	
23.11 The Department of Water Supply and the Planning Department shall coordinate priorities before the adoption of any new water development or County land use plans.	x	
23.12 All County potable water systems should have backup standby sources		x
23.13 Treat all water as a valuable resource in community design, and integrate designs for drinking water, stormwater, and recreational water needs.	x	
23.14 Manage water, stormwater, and wastewater as the same natural resource in collaboration with DWS, DEM, DPW, DOH, etc.	x	
23.15 New developments should be designed to reduce water demand, retain runoff, decrease flooding, and recharge groundwater.	x	
23.16 Support localized, small-scale solutions to water reuse and onsite systems.	X	
Objective 24: Planned and developed municipal sewer capacity is expanded to serve our and reduce sewage-related impacts on water quality.	Urban G	Frowth Areas
Policies		
21.1 A Sewerage Study for All Urban Areas, including appropriate water quality management strategies, shall be completed and used as guides for the general planning of sewerage disposal systems.		x
24.2 Private treatment systems shall be installed by land developers for major resorts and other developments along shorelines and sensitive higher inland areas, except where connection to nearby treatment facilities is feasible and compatible with the County's long-range plans, and in conformance with State and County requirements		x
24.3 Immediate steps shall be taken to designate treatment plant sites, sewerage pump station sites, and sewer easements according to the facility plans to facilitate their acquisition.	x	
24.4 The county shall obtain State and Federal funds to finance the construction of proposed sewer systems and improve existing systems.	X	
24.5 Plans for wastewater reclamation and reuse for irrigation and biosolids composting (remaining solids from the treatment of wastewater are processed into	X	



a reusable organic material) shall be utilized where topographically feasible and needed for landscaping purposes.		
24.6 Pollution shall be prevented, abated, and controlled at levels that will protect and preserve public health and wellbeing through the enforcement of appropriate Federal, State, and County standards.	x	
24.7 Ensure municipal wastewater systems serve designated Urban Growth Areas with the capacity to accommodate projected population growth.	X	
24.8 Average Daily Flow (ADF) for residential shall be 70 gallons per day per capita within any dwelling. For commercial and industrial uses (for example laundromats, food or beverage processing plants, et cetera) the ADF must be calculated based on the type of fixtures, number of fixtures, usage, and occupancy.	x	
24.9 The Department of Environmental Management and the Planning Department shall coordinate priorities before the adoption of any new wastewater development or land use plans.	x	
24.10 Prioritize developing a multipronged approach to wastewater infrastructure funding, including proactively seeking grant funding for wastewater system expansion, improvements, and new development.	x	
24.11 Ensure wastewater fees reflect actual costs for service, maintenance, and future improvements.	x	
24.12 Ensure that wastewater systems and improvements are designed and functioning to maximize system efficiencies, prevent accidental leaks or spills, and provide sanitary, reliable wastewater treatment that is not negatively impacting natural resources.	x	
24.13 Strive for an integrated approach to stormwater and wastewater, and water resource management that is comprehensive and as efficient as possible.		x
24.14 Encourage on-site water reuse solutions for large developments.		X
24.15 Encourage and incentivize the collection of rainfall for non-potable use.		x
24.16 Prioritize the use of gray water in areas connected to county water and not connected to county wastewater.		x
Objective 25: Increase green infrastructure practices.		
Policies		
25.1 Design to collect stormwater from streets, sidewalks, and other hard surfaces before it can enter the sewer system or cause local flooding; reduce the amount of stormwater that flows into the Sewer System.	x	
25.2 Control stormwater by using it as a resource rather than a waste.		X



"Drainage Master Plan for the County of Hawai'i" should be updated every for urban centers to incorporate new studies and reflect newly identified			x
ourage grassed shoulder and swale roadway design where climate and e conducive.			
ere applicable, natural drainage patterns shall be improved/restored to their capacity with special consideration for the practices of proper soil tion, and grassland and forestry management.	x		
lement nature-based solutions that manage stormwater onsite to reduce on on the storm sewer system and reduce flooding.			x
ritize drainage and flood studies for high-risk urban areas within the Urban rea.			x
nage standards shall incorporate cumulative upslope development patterns.	X	 	
elop an island-wide stormwater management program compatible with the S4 Phase II program.			Х
e County shall ensure sites are planned, designed, and developed to:			
ect, restore, or mimic the natural water cycle within built environments by ng, detaining, and/or treating stormwater runoff.			
gate direct impacts of the land development process through the use of nfrastructure or low-impact site planning techniques.			
tect areas that provide important water quality benefits and/or are larly susceptible to erosion and sediment loss.	X		
imize the integration of the existing landscape into green infrastructure ns.			
age post-construction stormwater runoff rates, through the use of green ucture or low-impact development stormwater management practices.			
e County shall ensure that golf course developments develop and implement nd site preparation plans to:			
elop nutrient management guidelines appropriate to Hawai'i for qualified tendents to implement so that nutrients are applied at rates necessary to sh and maintain vegetation without causing leaching into ground and waters.)
elop and implement an integrated pest management plan. Follow EPA nes for the proper storage and disposal of pesticides.			
elop and implement irrigation management practices to match the water of the turf.			
		1	I



	25.12 The County shall minimize impervious areas on County property, development sites, and parking areas and promote the use of permeable surfaces and landscaped areas in project designs including:			
	a) Porous materials			v
	b) Natural drainage			•
	c) Filtration pits			
	d) Infiltration basins, vegetated bioswales, permeable/porous paving			
Discussion: The Proposed Action will support the goals, objectives, and policies outlined in the County of Hawai'i Draft General Plan relating to Public Utilities. The Proposed Action will ensure the safe and efficient closure of the three LCCs in the Nā'ālehu community and provide a new, reliable collections system and WWTP. The WWTP shall be designed to accommodate future growth in the Nā'ālehu community. Future expansions would be accommodated, as capacity allows, on a first-come, first-served basis.				
	The Proposed Action will also incorporate drainage improvements to accommodate potenti term, the on-site drainage plan, as per Hawai'i County Code, Chapter 27, Section 20, wo caused by the construction activities in the WWTP project site area and effluent dispose account for expected one-hour, ten-year storm event, is retained within the site boundari with dirt berms would act as secondary containment during large storm events, further adverse impacts on adjacent or downstream properties.	al runof uld ensu al area es. Lano safegu	f. In the s ure that r would ne dscape bu arding ag	short- unoff ed to uffers jainst
	In the long-term, the proposed WWTP is anticipated to involve drainage improvements incl of the vegetated drainage diversion channel and relocation of the drainage outlet which	uding th is cons	ne realign idered to	ment be a

long-term impact. The drainage improvements shall be designed in accordance with the County Department of Public Works (DPW) drainage standards and will not alter existing stormwater discharge from the property.

Public Facilities and Services Goals

Our communities are safe and protected, and residents have access to quality, integrative health, education, and social services to support a high quality of life for residents of all ages.

Our communities are adequately served by sustainable and efficient public infrastructure and services based on existing and future growth needs, sound design principles, and effective maintenance practices.

Objective 26: Adequately maintain public facilities.

Policies		
26.1 Maintain an Asset Management Program aimed at utilizing maintenance plans to prolong the life of our facilities as well as reduce whole-life costs.	X	
26.2 Maintain the unique features of historic structures, while balancing maintenance and safety needs	X	
26.3 Prioritize the replacement of deficient and inadequate facilities.	X	
26.4 Encourage the adoption of innovative materials and methods that improve facility resilience.	X	



26.5 The development of County facilities should be designed to fit into the locale with minimal intrusion while providing the desired services. Implement protocols for receiving community input during Capital Improvement Project siting and design.	X	
Objective 27: Protect the health and wellbeing of residents and visitors		
27.1 Police and fire stations should be co-located whenever feasible.		X
27.2 The establishment of a police/fire facility shall consider site size and locations that permit quick and efficient vehicular access.		x
27.3 Strategically plan and locate volunteer fire facilities, which may include co- existing with fulltime Fire/EMS stations.		x
27.4 Police headquarters shall be near the geographic center of the service area and near concentrations of commercial and industrial use.		x
27.5 Hardened shelters shall be located within reasonable proximity to population centers.		x
27.6 Lifeguard stations should be located at all County and State beach parks.		X
27.7 Support and expand volunteer fire facilities and capacity.		X
27.8 Adequately support and fund the volunteer fire department and stations.		X
27.9 Maintain a level of service for response time that is consistent with National Fire Protection Association (NFPA) standards.		x
27.10 Ensure Hazardous Material service for both the windward and leeward sides of the island.		x
27.11 All fire stations should provide Emergency Medical Services (EMS).		X
27.12 Stations in rural areas should be based on the population to be served and response time rather than on geographic districts.		x
27.13 Accommodate flexibility in design and provisions for alternate water sources for fire protection when adequate public water is not available.		x
27.14 Incentivize the development of large, dedicated catchment tanks for firefighting access.		x
27.15 The Fire Department shall participate with other related governmental agencies and the involved landowners in the preparation of fire protection and prevention plans.		x
27.16 Crime Prevention through Environmental Design (CPTED) should be incorporated into planning and design.		x



27.17 Business Improvement Districts or other organizational tools, such as partnerships with local businesses, should be used to enhance security and orderliness in downtown areas.	x
27.18 Support bicycle patrol programs in urban areas.	X
27.19 Encourage the further development and expansion of community policing programs, school resource officers, and neighborhood/farm watch programs.	x
27.20 Ensure adequate training and capacity building for emergency response.	X
27.21 Prioritize hazard mitigation projects in the Capital Improvements Program.	X
27.22 County public expenditures within Volcanic Hazard Zones 1&2 should be limited to the restoration or enhancement of natural resources and parklands or expenditures required to serve existing development or necessary for public health and safety. However, other expenditures may be warranted based on a cost-benefit analysis.	x
27.23 Continue to participate in the National Flood Insurance Program (NFIP) Community Rating System (CRS) to the maximum extent possible and shall seek to improve its current CRS Class rating (to the maximum extent feasible to reduce insurance costs).	x
27.24 Ensure emergency warning sirens and communications coverage is adequate for each community.	x
27.25 All emergency response critical facilities and communication systems shall be designed and maintained to be resilient and remain operational during hazard events.	x
27.26 All new emergency facilities shall be designed to minimize and prevent loss.	X
27.27 In collaboration with State agencies, maintain shelter capacity and condition records to ensure that evacuation shelters are adequate for each community.	x
27.28 Increase public education related to hazard zones, including evacuation routes and procedures for visitor accommodations.	x
27.29 Develop the capacity for hazard preparedness of non-governmental organizations, businesses, and neighborhood groups.	x
27.30 Partner with government, private and nonprofit agencies, and other stakeholders to assess and plan for alternative routes and possible relocation of coastal roads.	x
Objective 28: Achieve Zero Waste in Hawai'i County by 2045	
Policies	



28.1 Appropriately designed and cost-effective solid waste transfer station sites shall be located in areas of convenience and easy access to the public.	X
28.2 Implement waste stream technology, such as recycling and upcycling and waste-to-energy to reduce the flow of refuse deposited in landfills.	x
28.3 Proactively pursue funding that will ensure continued progression toward zero- waste goals.	x
28.4 Optimize recoverable material diversion from landfill disposal by increasing percentage rates for diversion through waste reduction, recycling, and reuse.	x
28.5 Encourage and support composting at farms.	X
28.6 Encourage salvage and reuse of building materials and elements when demolition is necessary or appropriate.	x
28.7 Continue to develop and implement a green waste recycling program.	X
28.8 Incentivize opportunities for a circular economy, primarily upcycling and waste reuse by incorporating Resource Recovery Parks.	x
28.9 Ensure that redesign plans for landfill and transfer stations provide adequate space for Resource Recovery (RR) Stations.	x
28.10 Ensure waste and resource recovery facilities and equipment do not harbor, spread, or introduce harmful or invasive species.	x
28.11 Site new solid waste/resource recovery facilities in appropriate areas that serve the needs of population centers but that do not negatively impact the environment or surrounding neighborhood.	x
28.12 Reduce illegal dumping and littering.	X
28.13 Minimize the amount of waste generated by County facilities.	X
Objective 29: Each community has access to a wide range of educational opportunities.	
Policies	
29.1 Ensure educational facilities meet the needs of Hawai'i County.	X
29.2 In proposed communities, sufficient acreage, as determined by the Board of Education enrollment guidelines, shall be reserved for school facilities.	x
29.3 Sites shall be free from flooding and drainage problems, and excessive slope and shall incorporate appropriate street and driveway design and location to minimize traffic interference, pedestrian hazard, and enable safe and easy access for vehicles, bicycles, and pedestrians.	x
29.4 Continuous joint pre-planning of educational facilities shall be coordinated with the County, Department of Education, and the University of Hawai'i to ensure compatibility with public services, supporting infrastructure, and equitable mobility	X



access so that facilities are community-centered, designed for multiple uses, and serve as anchor institutions in the community.			
29.5 School facilities, such as playgrounds and gyms, should be combined with county parks to allow for afterschool use by the community for recreational, cultural, and other compatible uses.			X
29.6 The Hawai'i State Library System should co-locate public library facilities in public school libraries where a separate public library may not be feasible, promoting intentional proximity to other community facilities and assets that contribute to a high quality of life.			X
29.7 School buildings should be designed, or at times retrofitted, to serve as emergency shelters.			X
29.8 The County should facilitate the use of libraries to disseminate public information and engage civic participation. 29.9 Advocate to the State and private agencies to use educational facilities to offer placebased and distance education opportunities to adults.			X
29.10 Educational programs should be developed to provide opportunities in diversified industries and develop practices in sustainability and resiliency.			X
29.11 Encourage the State to provide student, faculty, and staff housing around University of Hawai'i sites.			X
29.12 Support the continued expansion of the University of Hawai'i at Hilo, Hawai'i Community College, and Pālamanui campuses, as well as encourage continuing education programs throughout the community.			X
29.13 Support and encourage the strengthening of the University of Hawai'i at Hilo through the transfer of appropriate colleges and departments from the University of Hawai'i at Mānoa to the University of Hawai'i at Hilo.			X
29.14 Encourage and support the active implementation of State and University of Hawai'i plans for post-secondary educational facilities, including the "Research and Technology Park," on Hawai'i Island.			X
29.15 Encourage the expansion of digital access and equity through the resilient buildout of broadband infrastructure and facilities.			X
29.16 Prioritize active transportation through the development of sidewalks, pedestrian walkways, and bike paths to and from educational facilities to increase walkability and pedestrian safety.			X
29.17 Require new developments in the vicinity of schools to provide safe pedestrian facilities and additional school zone signage.			X
Objective 30 : Park facilities are located within a 10-minute walk in urban areas and a 10 communities.	-minute	e drive in	rural
Policies			X



30.1 Diversify funding sources for recreational facilities.	X
30.2 State and County Capital Improvement Programs should continue to be coordinated to reflect recreational priorities.	x
30.3 Equitably allocate park dedications and in-lieu fees among the districts relative to the population.	X
30.4 Continue to improve parks and recreation outreach efforts to ensure program and facility information is adequately available, promoted online through accessible websites and other mediums, and kept up to date to facilitate maximum community participation.	x
30.5 Recreational facilities shall reflect the natural, historic, and cultural character of the area.	x
30.6 Equitably allocate facility-based parks among the districts relative to population, with public input to determine the locations and types of facilities.	X
30.7 Existing and new parks should be designed with features that accommodate and encourage meaningful levels of physical activity according to the level of service criteria.	x
30.8 Recreational facilities shall be assessed for dual use as emergency shelters and hardened as needed.	x
30.9 Establish, in cooperation with the State Department of Education, joint use of schoolyards, County parks, and other public facilities for community use for recreational, cultural, and other compatible uses.	x
30.10 Recreational facilities should be planned and located where they will best facilitate and support active-living communities.	X
30.11 Recreational sites should be planned and located within a 10-minute walk from residences in population centers, and serve with a minimum of five acres of land for park purposes per 1,000 resident population in non-population centers of Hawai'i Island.	x
30.12 Facilitate and prioritize the co-location of schools, parks, and senior centers to promote interactivity between community members of all ages.	x
30.13 Public lands with unique recreational and natural resources shall be maintained for public use.	x
30.14 Maintain and/or improve park facilities and programs based on community needs assessments to ensure County services are meeting the social, recreational, and activity needs of our communities.	x
30.15 Implement a proactive maintenance program to ensure that park facilities and trails are safely maintained for optimum usage.	X
30.16 Prioritize maintenance and necessary improvements at existing park facilities over developing new park facilities within each district (this does not preclude accepting lands for future park development or acquiring properties for the intent of	X



preserving open space, scenic areas, natural hazard areas, or cultural/historic areas from development).			
30.17 Combine recreation facility improvement projects with other needed facility improvements (e.g., ADA improvements with facility hardening, etc.)			X
30.18 Provide facilities and a broad recreational program for all age groups, with special considerations for the handicapped, the elderly, and young children.			X
30.19 Prioritize park acquisition and improvements that involve under-represented open recreation and healthy living activities (outside the scope of organized sports), such as:			
a) Walking and biking trails			
b) Skate/roller blade parks			
c) Dog-friendly parks			
d) Parks that offer camping opportunities			x
e) Botanical and community garden parks, pocket and art parks			
f) Equestrian/rodeo arenas			
g) Archery and shooting ranges			
h) ATV and motorized recreation areas			
i) Other types of active and passive recreation that enhance the quality of life for visitors and residents			
30.20 Support and enhance recreational facilities by developing additional recreational offerings in underutilized areas of County properties, such as the Pana'ewa Recreation Complex.			x
30.21 Continue to improve parks and recreation outreach efforts to ensure program and facility information is adequately available, promoted online through accessible websites and other mediums, and kept up to date to facilitate maximum community participation			x
30.22 Private and public cemeteries shall be compatible with surrounding land uses and provided with adequate access and drainage systems			X
Objective 31: Each community has access to healthcare facilities, programs, or commun	nity-base	ed care.	
Policies			
31.1 Ensure regular health service assessments identify and address the unique needs of the medically underserved population, especially in rural areas.			X
31.2 Partner with government, private and nonprofit agencies, and other stakeholders to ensure equitable access to healthcare services.			X



31.3 Ensure healthy communities through aligned land use and infrastructure policies.	x
31.4 Active living considerations should be integrated into the design of communities.	X
31.5 Advocate to the State to continue the operation of the rural hospitals.	X
31.6 Support the establishment of centrally located, 24-hour, full-service medical facilities, with trauma care, to service rural areas.	X
31.7 Hospitals should be on sites capable of handling moderate expansion of facilities. Quiet surroundings, convenient and adequate access, and compatibility with adjoining uses shall be required.	x
31.8 Hospitals shall be served by a public sewerage system or have selfcontained sewerage systems.	X
31.9 Ensure that hospitals are sufficiently hardened to remain in effective use through natural disasters.	X
31.10 Establish a comprehensive network of health and wellness services.	X
31.11 Integrate community health concerns in community planning.	X
31.12 Advocate for programs serving the elderly, disabled, and homeless persons.	X
31.13 Improve coordination and integration of services.	X
31.14 Support the establishment of home and community-based services (HCBS) that operate consistent with community character.	X
31.15 Increase opportunities and support for home-based care for aging in place.	X

Discussion: The Proposed Action will not directly impact the goals, objectives and policies outlined in the County of Hawai'i Draft General Plan relating to public facilities and services. The WWTP and collection system will be maintained by the COH-DEM and will provide improved wastewater treatment services to the Nā'ālehu community. Educational, recreational, healthcare, and emergency services in the community may indirectly benefit from the Proposed Action, but the policies and objectives outlined above will not be impacted by the Proposed Action.

Housing Goal

Residents have access to adequate and affordable housing to meet the needs of the population and provide equitable opportunities for household flexibility and mobility.

Objective 32 (Diversify and Expand Housing): Increase the number and variety of newly constructed housing units for rent and sale that addresses a range of Area Median Income (AMI).

Policies



32.1 County departments shall collaborate to identify and prioritize infrastructure requirements and public-private partnerships that support the desired density of housing types near mixed-use centers and transit centers in urban areas.	x
32.2 Incentivize a mix of diverse housing types, including missing middle housing and mixed-income communities.	x
32.3 Prioritize new housing including the missing middle in or near mixed-use developments, urban growth areas with infrastructure, and near existing and proposed transit centers.	x
32.4 Support experimental housing, energy efficiency, and compact housing communities in accordance with HRS 46-15.	x
32.5 Incentivize the use of universal design principles in new construction to create physically accessible housing for children, the aging, and those with mobility limitations.	x
32.6 Vacant lands in the urban growth boundary (UGB) should be prioritized for residential uses before additional agricultural lands outside the UGB are converted into urban uses.	x
32.7 Incentivize smaller housing options.	X
33.1 Enable data-driven research to support and maintain a housing inventory	
program to monitor existing housing inventory.	X
33.2 Incentivize the use of universal design principles for the rehabilitation of existing housing to create physical accessibility for those with mobility limitations.	X
33.3 Encourage the adaptive reuse of non-residential spaces for residential purposes in urban growth areas where supporting infrastructure exists.	X
33.4 Identify and support federal, State, and local housing assistance programs to rehabilitate existing housing for very low- to moderate-income residents.	x
Dbjective 34 (Create Housing Affordability): Prioritize providing quality affordable house residents.	sing for Hawai'i's
Policies	
34.1 Support affordable housing developments for all users including but not limited to the following groups: middle-income workforce, elderly, minimum wage workers, agriculture workers, individuals with special needs, individuals with disabilities, homeless, and retired individuals.	x
34.2 Support innovative and experimental housing types that address homelessness located near services, job centers, and transit hubs, while providing support services such as rent assistance	x



34.3 All affordable housing projects that receive development benefits from the County, such as land use/zoning approvals, special approvals (including 201-H), conditional uses, and density bonuses, shall be required to maintain the affordable rental units for not less than 20 years pursuant to deed restrictions or other mechanisms specified in the HCC.			x
34.4 Reduce the cost and time of processing land use and construction applications, particularly for affordable housing projects.			X
34.5 Allow for and apply property tax and land use regulations to incentivize and encourage private property owners to provide affordable housing rental units in mixed-use and urban areas and to disincentivize the land banking of unimproved properties.			x
34.6 Encourage public agencies and private organizations to participate in federal, state, and private programs to provide new and rehabilitated housing and rental opportunities for low- and moderate-income households.			X
34.7 Enable housing programs that implement a land trust strategy for publicly owned parcels.			X
34.8 Encourage the development of workforce housing within or near urban growth areas and employment centers and require large new developments that create a demand for housing to provide affordable workforce housing.			X
34.9 Enable and encourage the development of affordable retirement communities and aging-in-place opportunities that are located near services and activities for seniors			x
Discussion: The Proposed Action will not impact the goals, objectives, and housing outlin Hawai'i Draft General Plan relating to housing. The collection system and WWTP will serve Brewer lots and lots identified for potential future connection. While the Proposed Action impact the objectives and policies related to housing, the facilities shall be designed to ac growth and expansion of the Nā'ālehu community.	ned in the the exi will not a commod	ne County isting C. directly date futur	v of e
Integrated Systems Goal			
We are governed by integrated systems that are efficient, equitable, and organized to fac and collaboration.	ilitate co	oordinatio	'n
Objective 35: Increase collaboration and cooperation for efficiency, effectiveness, and re	esponsiv	/eness.	
Policies			
35.1 Maintain and adequately fund County government services at the level necessary to be effective.	X		
35.2 Ensure that government attitudes, actions, and services are sensitive to community needs and concerns.	X		
35.3 Sufficiently fund, and facilitate the timely preparation, maintenance, and update of public policies and plans to guide County programs and regulatory responsibilities.			X



35.4 Expand the adoption of technology across all County agencies to achieve greater efficiency, accessibility, and accountability to the general public throughout government operations.)
35.5 Continue to seek ways of improving public service through the coordination of service and maximizing the use of personnel and facilities.	X	
35.6 Promote alignment and consolidation of State and County functions whenever more efficient and effective delivery of government programs and services may be achieved.)
35.7 Collaborate with appropriate State agencies for the provision of public facilities to serve the needs of the community	x	
Djective 36: Maintain fiscal integrity, responsibility, and efficiency.		
licies		
36.1 Provide a balanced budget.		
36.2 Allocate fiscal resources to efficiently implement the objectives of the General Plan in addition to essential government operations.	x	
36.3 Ensure accountability in government operations.		
36.4 Calculate the cost of the different County services provided.	X	
36.5 Continue regular review of the County fee and fine schedules.		
36.6 Maintain a debt financing plan to schedule bond authorization.		
36.7 Leverage multiple sources of funding as part of the Capital Improvement Program (CIP).	x	
36.8 Encourage financing tools like Community Facilities Districts (CFD) to help fund off-site infrastructure improvements.		2
36.9 Develop short and long-range capital improvement programs and operating budgets for public facilities and services.		
36.10 Capital projects shall be analyzed for overlapping scopes.	X	
36.11 Projects involving more than one Department's assets shall be coordinated to define scoping, design, and construction needs.		
36.12 Improve the effectiveness of the Capital Improvement Program to maintain transparency of the status of all County CIP projects.		
		c



37.1 Promote policies that actively address and reduce disparate outcomes for historically underserved communities.		X
37.2 Seek equitable distribution of County investments towards promoting employment opportunities, infrastructure, and other community benefits.	X	
37.3 Provide resources for County employees to understand and actively advance equity solutions within all agencies of County government.		x
Discussion: The Proposed Action will not directly impact the goals, objectives, and housi County of Hawai'i Draft General Plan relating to integrated systems. As part of the EID / agencies, organizations, and stakeholders will be provided opportunities to comment on a Proposed Action. As such, the Proposed Action, as undertaken by the COH-DEM, shall con needs and concerns.	ng outli EA proc nd resp isider co	ned in the ess, identified ond to the ommunity
The Proposed Action is anticipated to utilize funding from the Clean Water State Revolving selection of a preferred alternative will take into account capital costs associate with opera maintenance of the Proposed Action. The August 2023 PER provides an evaluation of cost each of the four identified alternatives.	g Fund (ation an s assoc	(CWSRF). The d iated with
Economic Goal		
Our economy is diverse, regenerative, and innovative, improving and mainta wellbeing of our residents with a focus to increase local economic opportunitie	aining s.	the financial
Objective 38: Improve access at all levels fir education and training.		
Policies		
38.1 Support all levels of and forms of education.		X
38.2 Support programs and infrastructure that enables employees to telecommute or work in satellite locations.		X
38.3 Support apprenticeships and workforce training to strengthen leadership and entrepreneurial skillsets and networks		X
38.4 Support County apprenticeships, fellowships, and internships to strengthen skillsets, networks, and innovation.		X
Objective 39: Increase the growth and health of small businesses.		
Policies		
39.1 Establish Hawai'i Island as a business-friendly place.		X
39.2 Streamline regulatory processes associated with starting and operating a business.		x
39.3 Shared workspaces, including certified kitchens and industrial co-work buildings shall be supported.		x



39.4 Initiate and/or support programs to revitalize town centers and increase demand for local-serving businesses.	X
39.5 Maintain strong partnerships and effective communication with the business community to identify barriers and actions to improve the business climate.	X
39.6 Support creation of shared facilities and resources that can be utilized by multiple opportunity clusters, such as creative industries and technical services.	X
39.7 Promote creative industries through collaboration with local artists on the design and creation of public, livable spaces.	X
39.8 Support programs and initiatives that encourage manufacturing and support Hawai'i Island's small-scale independent manufacturers.	x
39.9 Support business development programs by reducing underwriting risks for the private sector such as industrial development bonds, tax abatement, and low-interest loan programs.	x
39.10 Support access to capital for small businesses and start-ups.	X
39.11 Promote the use of the incentives offered by federal and state programs such as opportunity zones and enterprise zones partnership programs to attract businesses.	x
39.12 Encourage the development of the Technology, Creative, Agribusiness, Health and Wellness, and Education targeted sectors.	X
jective 40: Incorporate resiliency, diversity, and innovation in County programs, plans, a port healthy economic development and revitalization.	nd research to
licies	
40.1 Increase County resources and actions devoted to strategic planning, interagency coordination, training and expertise, and capital improvements.	X
40.2 Capital improvements program shall improve and increase the capacity of existing and future commercial and industrial areas.	X
40.3 Maintain a program for updating zoning code to accommodate emerging industries and technologies consistent with other goals, objectives, and policies of the General Plan.	x
40.4 Support lease terms and extensions on State and DHHL lands that provide opportunities to improve or rehabilitate existing commercial and industrial zoned areas.	x
40.5 Encourage land uses that allow for small-scale manufacturers in retail	x



40.6 Maintain plans and programs to foster sustainable business development opportunities focusing on regenerative agriculture, green technologies and building, innovation and technology, creative industries, and regenerative tourism.	X
40.7 Improve opportunities for multi-modal transit that improve the quality of existing job centers.	x
40.8 Provide technological infrastructure that increases the competitiveness of businesses and allows them to thrive in all parts of the island.	X
40.9 Expand opportunities for innovation and tech-based businesses.	X
40.10 Promote a distinctive brand for the island of Hawai'i including distinctive, regional identities as an entity unique within the State of Hawai'i.	X
40.11 Encourage the development of economic opportunities through the utilization of byproducts from various industries.	x
40.12 Continue to encourage the research, development, and implementation of advanced technologies and processes.	x
40.13 Promote Hawai'i Island as a center for natural scientific research.	x
opportunities as the WWTP and collection system become operational. The Proposed Actieducational, local small-business, or research opportunities within the Nā'ālehu community Agriculture and Food Systems Goal Agriculture is a robust, diversified sector that achieves food security and includes a broad based businesses that highlight value.	on will not impact /. range of agricultural-
Objective 41:	
Policies	
41.1 Support urban agriculture uses including on-site home occupation sales.	x
41.2 Support innovative agriculture demonstration projects.	
 41.2 Support innovative agriculture demonstration projects. 41.3 Assist in the expansion of the agricultural industry through the efficient use of productive agricultural lands, capital improvements, and continued cooperation with appropriate State and Federal agencies. 	
 41.2 Support innovative agriculture demonstration projects. 41.3 Assist in the expansion of the agricultural industry through the efficient use of productive agricultural lands, capital improvements, and continued cooperation with appropriate State and Federal agencies. 41.4 Assist in the promotion of Hawai'i Island branding for local produce and agriculture products. 	
 41.2 Support innovative agriculture demonstration projects. 41.3 Assist in the expansion of the agricultural industry through the efficient use of productive agricultural lands, capital improvements, and continued cooperation with appropriate State and Federal agencies. 41.4 Assist in the promotion of Hawai'i Island branding for local produce and agriculture products. 41.5 Collaborate across County departments to engage in food systems planning, including the elimination of food deserts. 	



41.6 Assist in cooperative marketing and distribution endeavors to expand		
opportunities for local agricultural products for the local market as well as for exports.		X
41.7 Explore opportunities and methods to utilize local materials and byproducts from agriculture, agroforestry, silviculture, and aquaculture.		X
41.8 Support the research and development of the agriculture technology industry in collaboration with agriculture applications to optimize production.		X
41.9 Support regenerative agricultural practices and restoration of traditional knowledge and practices that offer multiple benefits, such as by improving agriculture and food system waste management that can reduce County greenhouse gas (GHG) emissions.		X
41.10 Support the development of private, County, and State agricultural parks to make land available and distributed equitably and proximate to infrastructure and housing.		x
41.11 Partner with government, private and nonprofit agencies, and other stakeholders for programs, training, and building community capacity in the promotion of the agricultural industry.		X
41.12 Increase public-private partnerships to develop and support community-based food systems.		x
41.13 Support the development of farm labor housing.		X
local agriculture infrastructure.	es that mpr	UVC
Policies		
Policies		
Policies 42.1 Support the creation of water cooperatives supported with financial sources, such as CFDs. Cooperative users should be responsible for the development, maintenance, and repair of agricultural non-potable water systems.		x
Policies 42.1 Support the creation of water cooperatives supported with financial sources, such as CFDs. Cooperative users should be responsible for the development, maintenance, and repair of agricultural non-potable water systems. 42.2 Where the County has replaced surface water sources with groundwater sources to meet Safe Drinking Water standards, the County should consider repurposing the surface water sources for agricultural use where the allocation is supportive of the ecosystem.		x
Policies 42.1 Support the creation of water cooperatives supported with financial sources, such as CFDs. Cooperative users should be responsible for the development, maintenance, and repair of agricultural non-potable water systems. 42.2 Where the County has replaced surface water sources with groundwater sources to meet Safe Drinking Water standards, the County should consider repurposing the surface water sources for agricultural use where the allocation is supportive of the ecosystem. 42.3 Advocate for more flexible and innovative wastewater systems to serve agriculture facilities.		x x x
 Policies 42.1 Support the creation of water cooperatives supported with financial sources, such as CFDs. Cooperative users should be responsible for the development, maintenance, and repair of agricultural non-potable water systems. 42.2 Where the County has replaced surface water sources with groundwater sources to meet Safe Drinking Water standards, the County should consider repurposing the surface water sources for agricultural use where the allocation is supportive of the ecosystem. 42.3 Advocate for more flexible and innovative wastewater systems to serve agriculture facilities. 42.4 Support the adaptive reuse or rehabilitation of existing infrastructure or buildings for agricultural processing, including but not limited to the development of commercial kitchens, processing, storage, or distribution facilities. 		x x x x
 Policies 42.1 Support the creation of water cooperatives supported with financial sources, such as CFDs. Cooperative users should be responsible for the development, maintenance, and repair of agricultural non-potable water systems. 42.2 Where the County has replaced surface water sources with groundwater sources to meet Safe Drinking Water standards, the County should consider repurposing the surface water sources for agricultural use where the allocation is supportive of the ecosystem. 42.3 Advocate for more flexible and innovative wastewater systems to serve agriculture facilities. 42.4 Support the adaptive reuse or rehabilitation of existing infrastructure or buildings for agricultural processing, including but not limited to the development of commercial kitchens, processing, storage, or distribution facilities. 42.5 Promote the development of a locally grown building material industry through streamlined permitting or building code flexibility. 		X X X X X X



42.7 Encourage the use and optimization of the export capacity of airports and harbors for local goods.	X
42.8 Support research and development of viable biofuel projects that will supply renewable transportation fuels or power for Hawai'i Island in ways that are community-supported, sustainable, ecologically sound, and complementary to food production.	X
42.9 Support the continued operation of the USDA's Daniel K. Inouye U.S. Pacific Basin Agricultural Research Center facility.	X
42.10 Support the development of processing and manufacturing facilities.	X

Discussion: The Proposed Action will directly impact the goals, objectives, and housing outlined in the County of Hawai'i Draft General Plan relating to agriculture and food systems. the proposed WWTP and the collection system are located in "not rated" lands which is assigned to developed communities. Construction of the collection system within the County roads would not affect agricultural lands.

Abandonment of the three LCCs would reduce the potential for contamination of groundwater that is used for irrigation of agricultural lands. Otherwise, abandonment of the LCCs and the existing wastewater collection system would not affect agricultural lands within the affected areas.

Visitor Industry Goal

A high quality of life for residents is maintained when a regenerative visitor industry balances the preservation of natural and cultural resources with responsible visitation.

Objective 43:

Policies	
43.1 Continue to monitor and adopt trends and standards for regenerative tourism.	X
43.2 County departments should integrate economic development, equity, and sustainability outcomes into their annual goals and reports to the Mayor.	X
43.3 Identify partnerships and resources with the visitor industry to ensure balance with the social, physical, and economic goals of the County.	X
43.4 Prioritize the maintenance of County properties and establish appropriate protocols for protection of wahi pana.	X
43.5 Ensure and expand equitable access to interpretive information about our wahi pana.	X
43.6 Support the coordination, collaboration, and improvement in public transportation services as well as eco-friendly options.	X
43.7 Support the coordination, collaboration, and improvement of public accessibility to natural resources with State agencies and private landowners.	X



43.8 Support partnerships to evaluate visitor industry impacts, develop mitigation strategies, and incorporate educational programs on native Hawaiian and community-based pono practices.			X
Objective 44: Increase authentic Hawai'i Island visitor experiences.			
Policies			
44.1 Integrate 'āina based and place-based values into Hawai'i Island's identity.			X
44.2 Strengthen the accessibility of creative industries and Hawai'i Island-made products such as fashion, food, and the arts to the visitor industry.			X
44.3 Sustain a visitor industry that promotes small business development.			X
44.4 Encourage eco-tourism and agricultural tourism as regional opportunities.			X
44.5 The visitor industry shall promote a high quality of life for residents.			X
44.6 Collaboratively create initiatives and improve existing efforts to provide social benefits through transportation, community assets, and housing.			X
44.7 Support the promotion and development of community-based programs, festivals, and events that celebrate our communities.			X
44.8 Maintain efforts to continue dialogue among stakeholders and tie 'āina-based and place-based values to Hawai'i Island's brand.			X
Environmental Goal			
Our natural and cultural resources are thriving and sustainably managed, preserved, and our unique and diverse environment.	1 restore	ed to mai	ntain
Objective 45: Increase the biodiversity and resilience of native habitats.			
Policies			
45.1 Minimize and mitigate significant impacts, such as degradation, incompatible uses, or other threats, to native Hawaiian habitats and public trust resources.	X		
45.2 Strive to improve the health of our island's forests, watersheds, nearshore environments, and coral reefs.			X
45.3 Encourage the preservation and restoration of natural landscape features, such as coral reefs, beaches and dunes, forests, streams, floodplains, and wetlands, or aquifer recharge areas that have the inherent capacity to avoid, minimize, or mitigate the impacts of climate change.	x		
45.4 Maintain the shoreline for recreational, cultural, educational, and/or scientific uses in a manner that is protective and respectful of resources and is of the maximum benefit to the general public.	x		



45.5 Increase collaborative efforts to improve coordination to conserve and manage wetlands, streams, and watersheds.		2
45.6 Encourage the preservation of native vegetation during development activities.	X	
45.7 Improve the use of native or non-native plants of cultural or environmental importance.	x	
45.8 Prioritize native landscaping for all (CoH) public projects.	X	
45.9 Limit the establishment of invasive or alien species.	X	
45.10 Maintain a continuing program to identify and protect exceptional trees, groves or stands of tree.		
45.11 Encourage and incentivize green belts, tree plantings, and landscape plans and designs in urban areas.		
45.12 Increase collaborative efforts to create and maintain community forests, food forests and other 'agroforestry'.		
45.13 Pursue the acquisition of lands for the protection of natural resources.		
a) Implement the Hawai'i State Wildlife Action Plan (SWAP)b) Better understand and model carrying capacities of the island's habitats and resourcesc) Improve the inventory of forested lands and associated ecosystem services		
d) Encourage the continued identification and inclusion of unique wildlife habitat areas of native Hawaiian habitat within the Natural Area Reserve System		
e) Anticipate future habitat migration, especially wetlands and coastal ecosystems		
f) Prioritize quantitative wetland assessment to identify wetlands		
g) Expand native/endemic forest cover		
h) Improve enforcement for illegal activities that harm or degrade endemic habitats		
45.15 Discretionary permit applications shall inventory the following and include appropriate mitigation measures of any impacts on the subject property:		
a) Groundwater recharge areas above 3,000 feet elevation	x	
b) Intact native habitats		
c) Critical habitat areas as identified by federal or state agencies		
		i i



d) Exceptional trees			
e) Historic, archaeological, or cultural sites or properties			
45.16 Any development shall be designed to not adversely impact the following resource asset(s):			
a) Rivers, streams, springs, and other naturally flowing surface water bodies	x		
b) Anchialine pools and estuaries			
c) Shoreline setback areas, beaches, and dunes			
45.17 Maintain shoreline setbacks to:			
a) Protect natural shoreline vegetation;			
b) Protect marine turtle nesting beaches/areas;			
c) Protect water quality;			x
d) Protect structures from the effects of long-term sea level rise;			
e) Protect beaches and shorelines from erosion; and			
f) Allow redevelopment of existing waterfront commercial structures consistent with the existing community character and preserve overwater views.			
45.18 Landscaping and irrigation shall be designed to maximize water use efficiency and native plants.	x		
Objective 46: Preserve the health of the watersheds by improving water quality and red	lucing ru	unoff.	
Policies			
46.1 Engage in comprehensive watershed planning to protect all watersheds and identify priority watershed areas to develop or complete watershed management plans and projects.			x
46.2 Address water best management practices and implement plans for non-point source discharges, such as irrigation flows, agricultural or urban runoff.	x		
46.3 Within mauka areas of high rainfall/fog-drip belt, ground disturbing activities such as excessive soil compaction and excessive removal of vegetative cover should be minimized and mitigated consistent with management strategies that encourage the retention of existing forested and pasture areas, reforestation, minimal coverage by impervious surfaces and other strategies that encourage effective infiltration to groundwater.	x		
46.4 Strengthen best management practices for wildfire management (erosion reduction, sediment control, fire management, storm water management, and natural riparian buffers along perennial and intermittent streams).	x		



46.5 Maintain participation in the development and implementation of the Ocean			
Resources Management Plan (ORMP), marine zoning plan(s), Marine Managed Areas (MMA's) or other appropriate tool(s).			X
46.6 Partner with government, private and nonprofit agencies, and other stakeholders to:			
a) Implement a comprehensive conservation plan that identifies priority watershed areas for habitat restoration and enhancement.			
b) Inventory lands considered necessary for the protection of watersheds, water sources and water supplies.			
c) Review and designate forest, river corridors, and watershed areas into the conservation district during State land use boundary comprehensive reviews.			
d) Monitor impacts to coral reefs and nearshore environments and address land- based sources of impacts.			
e) Restore wetlands and riparian corridors to decrease erosion, increase sediment management, groundwater infiltration, nutrient/pollutant uptake, soil moisture retention, stormwater abatement, and cultural/community connections.			x
f) Restore stream flows (volumes) to meet public trust purposes or where environmental quality could be improved.			
g) Develop reasonable standards to improve stream and coastal water quality monitoring and encourage local communities to develop such projects.			
h) Achieve a net increase in nearshore water quality with more AA designations, and/or fewer impaired inland freshwater bodies and impaired marine/coastal waterbodies.			
i) Prioritize water quality monitoring activities in areas of existing and potential urban growth.			
j) Document pollutant loads for Hawai'i Island streams and coastal waters.			
Objective 47: Increase direct community restoration and collaborative efforts to conservisland's biocultural resources.	/e and n	ourish the	e
Policies			
47.1 Encourage an overall conservation ethic in the use of Hawaii's resources by protecting, preserving, and conserving our critical and significant natural resources.			x
47.2 Foster recognition of the importance and value of the land, air, and water resources to Hawaii's people, their cultures, and visitors.	x		
47.3 Integrate progressive strategies incorporating indigenous and contemporary knowledge and practices to maintain environmental quality at the highest standards, address a changing climate, protect natural resources, and restore ecosystem health for the benefit of present and future generations.	x		



47.4 Protect the reasonable exercise of customarily and traditionally exercised rights of Hawaiians to the extent feasible.	x		
47.5 Promote resource management that is sustainable, responsible, and data driven.			x
47.6 Require the management of natural resources in a manner that fully minimizes adverse effects on the environment and depletion of energy and natural resources.	x		
47.7 Ensure that activities authorized or funded by the County do not irretrievably damage natural resources.	Х		
47.8 Increase public pedestrian access opportunities to scenic places and vistas.			X
47.9 Increase community stewardship partnerships that focus on management responsibilities and promote community education or shared learning.			x
47.10 Improve Community Planning capacity building efforts toward coordination, leadership, effective action, connectivity, and impact.	x		
47.11 Contribute to programs of collection and dissemination of data concerning cultural or natural resources.			x
47.12 Partner with government, private and nonprofit agencies, and other stakeholders to:			
a) Protect special areas, structures, and elements that are an integral and functional part of Hawai'i's ethnic and cultural heritage.			
b) Identify and protect wahi pana.			
c) Promote the preservation and restoration of significant natural and historic resources.			x
d) Aid in programmatic education concerning historic sites.			
e) Maintain the shoreline for recreational, cultural, educational, and/or scientific uses in a manner that is protective of resources and is of the maximum benefit to the public.			
f) Encourage the documentation and preservation of traditional ecological knowledge, identifying best management practices for integration.			
Objective 48: The historical integrity, character, scenic assets, and open spaces of our or protected, restored, and treated as unique assets with significant social and economic val perpetuity.	commun ue and r	ities are nanaged	in
Policies			
48.1 Require both public and private developers of land to provide historical and archaeological surveys and cultural assessments, where appropriate, prior to the clearing or development of land when there are indications that the land under consideration has historical significance.	x		



appropriate.	x	
48.3 Encourage the restoration of significant sites on private lands.		
48.4 Signs explaining historic sites, buildings and objects shall be in keeping with the character of the area or the cultural aspects of the feature.		
48.5 Historic Preservation shall represent the full range and diversity of the multi- cultural heritage of Hawai'i Island.		
48.6 Maintain an inventory of significant cultural and historic sites and districts compatible with that of the State Historic Preservation Division.	x	
48.7 Ensure that projects requiring preservation plans are identified on subdivision maps and plan approval site plans.		
48.8 The County shall develop a comprehensive management plan for historic and cultural resources that are on County owned properties or on properties managed by the County.		
48.9 The County shall use and promote the use of interpretive signage and/or other appropriate methods to recognize landscapes, sites, buildings, and objects of significant historical and cultural importance.		
48.10 Maintain the character of County-owned historic structures and bridges, as appropriate.		
48.11 Outstanding natural or cultural features, such as scenic resources, water courses, fine groves of trees, heiau, and historical sites and structures, shall be identified and preserved during subdivision.	x	
ective 49: Protect, restore, and enhance our communities' unique scenic character.		
cies		
49.1 Consider structural setback from major thoroughfares and highways and establish development and design guidelines to protect important view planes.	x	
49.2 Preserve transportation corridors that have important scenic, historic, recreational, cultural and/or natural resources that enhance the character and scenic resources of communities.		
49.3 Protect the views of areas endowed with natural beauty by carefully considering the effects of proposed construction and compatibility during all land use reviews.		
49.4 Encourage the design of developments and activities that complement the		
natural beauty of the island.		



49.6 Preserve and protect significant lava tubes, caves, or other geologic features determined to be significant by a governmental agency or plan.

Discussion: The Proposed Action will support the goals, objectives and policies outlined in the County of Hawai'i Draft General Plan relating to the environment. Chapter 4 of this EID / EA discusses the existing conditions and potential impacts to the natural, cultural, and historical resources identified in the Project Area. Various mitigation measures and BMPs as outlined in Chapter 4 will be implemented to ensure that long-term impacts as a result of the Proposed Action are avoided.

As presented in the EID / EA, two biological surveys have been undertaken in the Project Area. As a result, floral and faunal resources in the Project Area are not expected to be impacted by the Proposed Action. The DOH initiated consultation with the U.S. Fish and Wildlife Service (FWS) pursuant to Section 7 of the Endangered Species Act (See Appendix C). The consultation requested the FWS' concurrence with the determination that the Proposed Action "may affect, but is not likely to adversely affect" federally listed species. In a letter dated June 21, 2024, the FWS provided concurrence with this determination. Additionally, the FWS provided avoidance and minimizations to be implemented to avoid adverse effects

Additionally, an archeological inventory survey was completed for the Project Area. The survey identified four previously documented and seven newly documented historic properties in the Project Area. In accordance with federal regulations (36 CFR 800.5), the results of the AIS are intended to support a future determination of "no adverse effect" by the Hawai'i State Department of Health (DOH) on behalf of the Environmental Protection Agency (EPA). Additionally, in accordance with HAR §13-275-7, the Hawai'i State DOH and County of Hawaii Department of Environmental Management (COH-DEM) are expected to determine the project effect as "effect, with proposed mitigation commitments."

As proposed through the AIS, the identified properties have been thoroughly documented mitigating any possible effects which no further work being proposed. Although historic properties have been determined to require no further work, the County will consider further communication and coordination with identified stakeholders to develop feasible and reasonable accommodations on either preserving historic properties or alternatives that integrate the broader preservation goals of the Waikapuna property.

5.2.2 General Plan Land Use Pattern Allocation Guide and Zoning

The General Plan Land Use Pattern Allocation Guide Map (LUPAG) delineates broad-brush boundaries that are graphic expressions of the General Plan policies, particularly those relating to land uses. The land use pattern in a broad, flexible design intended to guide the direction and quality of future developments in a coordinated and rational manner. These maps delineate a number of land use categories for each area.

Discussion:

The General Plan Land Use Pattern Allocation Guide (LUPAG) designation for the project area includes Single-Family Residential District, Village Commercial District, Limited Industrial District, Multiple-Family Residential District, Agricultural District, and Open District designations. The existing and proposed use of the project area is consistent with the current LUPAG designation. The existing and proposed use of the project area is a permitted use for areas within the County zoning districts.

5.2.3 Ka'ū Community Development Plan

The County of Hawai'i General Plan calls for the preparation of community development plans (CDPs) "to translate the broad General Plan statement to specific actions as they apply to specific geographical areas." The Ka'ū CDP is one of nine CDPs for Hawai'i County. In October 2017, the



Ka'ū CDP was adopted as Ordinance No. 2017-66. The purpose of CDPs is to implement the broad goals within the General Plan on a regional basis and to translate the broad General Plan statements into specific actions. CDPs are the forum for community input into managing growth and coordinating the delivery of government services to the community. CDPs designate detailed development patterns and direct physical development and public improvements by detailing land use policies and infrastructure priorities.

The CDP Land Use Policies are designed to preserve the preferred future settlement pattern and achieve the Community Objectives as Ka'ū grows. There are Land Use Policies designed to protect coastal areas, agricultural lands, mauka forests, scenic areas, sensitive ecosystems, cultural resources, and public access.

The Proposed Action is consistent with land use policies as the improvements are designed to serve the designated areas shown in the Land Use Policy map which classifies the Project Area as primarily low and medium density urban. The WWTP will partially be constructed within lands classified as extensive agriculture. The collection system and the wastewater treatment and disposal facility will be consistent with the policy related to infill of commercial development within the Nā'ālehu community. The collection system improvements are consistent with the policy to maintain the community character as the improvements will retain the existing pavement, including retention of streets, shoulders, and drainage systems.

The collection system will be consistent with the CDPs policies which prioritize improvements in infrastructure, facilities, and services as the improvements for the Proposed Action been designed not to preclude expansion to accommodate the Nā'ālehu community. Similarly, the wastewater treatment and disposal facility has been designed not to preclude expansion to accommodate the future needs of the Pāhala community. Future subdivisions would be accommodated, as capacity allows, on a first-come, first-served basis.

5.2.4 County of Hawai'i Special Management Area

Pursuant to the Hawai'i CZM Program, HRS Chapter 205A, the counties have enacted ordinances establishing Special Management Areas (SMAs) that are in close proximity to the shoreline. Any "development" within the SMA requires an SMA Use permit administered by the County of Hawai'i Planning Department. Through the SMA permit system, the County assesses and regulates developments proposed for areas located within the SMA. The Proposed Action is located within the Nā'ālehu community which lies about two miles from the shoreline area and is not located within an SMA. As such, the project will not be subject to the requirements of an SMA Use permit.

5.3 List of Permits and Approvals

State of Hawai'i Department of Health

- Approval to Construct
- Approval to Use
- National Pollutant Discharge Elimination System Construction Stormwater Permit
- Underground Injection Well Abandonment
- Noise Permit
- Noise Variance (only if required)



County of Hawai'i

- Special Permit
- Plan Approval
- Grading Permit
- Building Permit
- Electrical Permit
- Plumbing Permits
- Fence Permit
- Sign Permit (only if required)
- Permit to Work Within County Right-of-Way



(This page intentionally left blank)



CHAPTER 6: LEGAL FRAMEWORK AND REGULATING AUTHORITIES

6. Legal Framework and Regulatory Authorities

As previously described, the County may use Clean Water State Revolving Fund (CWSRF) Program for construction of the Nā'ālehu Large Capacity Closure project. Since the State Revolving Fund receives annual funding from EPA, the National Environmental Policy Act (NEPA) of 1969, as amended (42 U.S.C. §§ 4321 – 4347), requires a federal agency proposing to undertake a project to consider the potential environmental impacts of the proposed project. Use of federal funds for a project is among the criteria set forth in NEPA that require preparation of environmental review documentation under NEPA and procedural requirements at 40 Code of Federal Regulations (CFR) Parts 1500-1508 (Council on Environmental Quality (CEQ) regulations), and 40 CFR Part 6 (U.S. Environmental Protection Agency (EPA) regulations). This EID / EA has been prepared under these guidelines.

The following regulatory requirements apply to this EID / EA and to federal cross cutting regulations necessary for compliance with the CWSRF program.

6.1 National Environmental Policy Act (NEPA) of 1969 (as Amended)

NEPA was passed in 1969 "to assure that all branches of government give proper consideration to the environment prior to undertaking any major federal action that significantly affects the environment." NEPA requires all federal agencies to prepare Environmental Information Documents (EIDs), Environmental Assessments (EA) and/or Environmental Impact Statements (EISs) to assess environmental impacts from project alternatives.

The purpose of NEPA is "to declare a national policy which will encourage productive and enjoyable harmony between man and his environment; to promote efforts which will prevent or eliminate damage to the environment and biosphere and stimulate the health and welfare of man, to enrich the understanding of the ecological systems and natural resources important to the Nation; and to establish a Council on Environmental Quality," Sec. 2 [42 USC § 4321].

According to NEPA, it is the continuing responsibility of the federal government to use all practicable means, consistent with other essential considerations of national policy, to improve and coordinate federal plans, functions, programs, and resources. NEPA, as amended in 1970, requires federal agencies to: (a) utilize a systematic, interdisciplinary approach which will ensure the integrated use of the natural and social sciences and the environmental design arts in planning and in decision making which may have an impact on man's environment; (b) identify and develop methods and procedures, in consultation with the Council on Environmental Quality established by Title II of this Act, which will ensure that presently un-quantifies environmental amenities and values may be given appropriate consideration in decision-making along with economic and technical considerations; and (c) include in every recommendation a detailed statement on the environmental impact of the Proposed Action; any adverse environmental effects which cannot be avoided should the proposal be implemented; alternatives to the Proposed Action; the relationship between local short-term uses of man's environment and the maintenance and enhancement of long-term productivity, and; any irreversible and irretrievable commitments of



resources which would be involved in the Proposed Action should it be implemented, Sec. 102 [42 USC § 4332].

This project may be funded by federal funds provided by U.S. Environmental Protection Agency (EPA) through the State of Hawai'i's Clean Water State Revolving Fund (CWSRF) Program. As such, the State of Hawai'i Department of Health (DOH) must conduct an environmental review of projects funded under the CWSRF as required under the Code of Federal Regulations (CFR), using the EPA-approved State Environmental Review Process. In addition, the State must comply with the federal cross-cutting authorities set forth in 40 CFR § 35.3145 for the CWSRF.

The CWSRF requirements are set forth as "cross cutters" described as follows.

In addition to the cross cutters required by the EPA-approved State Environmental Review Process, EPA guidance for conducting environmental reviews, and the Clean Water Act have been included.

6.2 Archaeological and Historic Preservation Act (54 U.S.C. § 312502)

The Archaeological and Historic Preservation Act (AHPA), also known as the Archaeological Recovery Act and the Moss-Bennett bill, was passed and signed into law in 1974. It amended and expanded the Reservoir Salvage Act of 1960. The AHPA built upon the national policy, set out in the Historic Sites Act of 1935, "*to provide for the preservation of historic American sites, buildings, objects, and antiquities of national significance.*" The AHPA expanded the policy by focusing attention on significant resources and data but does not require that they be shown to be of "national" significance. The AHPA required that federal agencies provide for *"...the preservation of historical and archeological data (including relics and specimens) which might otherwise be irreparably lost or destroyed as the result of...any alteration of the terrain caused as a result of any Federal construction project of federally licensed activity or program.*"

54 United States Code (U.S.C.) § 312502 (a)(1) states: "When any Federal agency finds, or is notified, in writing, by an appropriate historical or archeological authority, that its activities in connection with any Federal construction project or federally licensed project, activity, or program may cause irreparable loss or destruction of significant scientific, prehistorical, historical, or archeological data, the agency shall notify the Secretary, in writing, and shall provide the Secretary with appropriate information concerning the project, program, or activity."

54 U.S.C. § 312502 (b)(1) states: "When any Federal agency provides financial assistance by loan, grant, or otherwise to any private person, association, or public entity, the Secretary, if the Secretary determines that significant scientific, prehistorical, historical, or archeological data might be irrevocably lost or destroyed, may, with funds appropriated expressly for this purpose-

- (A) Conduct, with the consent of all persons, associations, or public entities having a legal interest in the property, a survey of the affected site; and
- (B) Undertake the recovery, protection, and preservation of the data (including analysis and publication)."



As discussed in Section 4.7 of this EA / EID, a Draft AIS Report was completed by Cultural Surveys Hawai'i in May 2024. The AIS is was designed to be compliant with both Federal and Hawai'i State environmental and historic preservation review legislation. The AIS was conducted to assess any historic properties for integrity and site significance in accordance with HAR §13-275-6. This report is also intended to support any project-related historic preservation consultation with consulting parties, such as state and county agencies and interested Native Hawaiian Organizations (NHOs) and community groups, if applicable.

The AIS identified four previously documented and seven newly documented historic properties in the Project Area. In accordance with federal regulations (36 CFR 800.5), the AIS results support the determination made by the Hawai'i State DOH on behalf of the EPA of "no adverse effect." In accordance with HAR §13-275-7, the Hawai'i State DOH and COH-DEM have determined the project effect is "effect, with proposed mitigation commitments." As stated in the AIS, the identified sites have been fully documented, mitigating any potential effect, and no further work is recommended.

Although historic properties have been determined to require no further work, the County will consider further communication and coordination with identified stakeholders to develop feasible and reasonable accommodations on either preserving historic properties or alternatives that integrate the broader preservation goals of the Waikapuna property.

Archaeological monitoring is recommended during all project ground-disturbing activities to facilitate identification for information purposes, especially of any subsurface lava tubes that may contain historic properties. Archaeological monitoring will proceed under an archaeological monitoring plan that meets the requirements of HAR §13-279-4, which will be submitted to the SHPD for review and acceptance prior to project initiation.

The contract drawings will state that, should archaeological sites such as walls, platforms, pavements or mounds, or remains such as artifacts, burials, concentrations of shell or charcoal be encountered during construction activities, work shall cease immediately and the find shall be protected from further damage. The contractor shall immediately contact the State Historic Preservation Division (SHPD), who will assess the significance of the find and recommend an appropriate mitigation measure, if necessary.

In addition to the AIS, the County is obligated to comply with the National Historic Preservation Act (NHPA). On June 3, 2024 the DOH, on behalf of the County, initiated consultation for this project in accordance with Section 106 of the NHPA. Consultation letters were sent to various Native Hawaiian Organizations inviting comments from organizations that may attach religious or cultural significance to properties affected by the proposed actions. NHPA Section 106 consultation correspondence can be found in Appendix F of this EA / EID. The SHPD has not yet responded to the consultation request. As such, any future consultation with the SHPD shall be incorporated into the Final EID / EA.



6.3 Bald and Golden Eagle Protection Act (16 U.S.C. § 668-668c)

The Bald Eagle Protection Act (16 U.S.C. § 668-668c) prohibits any act to take, possess, sell, purchase, barter, offer to sell, purchase or barter, transport, export or import, at any time or in any manner any bald eagle commonly known as the American eagle or any golden eagle, alive or dead, or any part, nest, or egg thereof of the foregoing eagles.

No bald or golden eagles are found in Hawai'i.

6.4 Clean Air Act (42 U.S.C. § 7401 et seq.)

Over the years, there has been a series of legislations affecting air quality and a number of amendments adopted related to air quality. The Air Pollution Control Act of 1955 was the first federal legislation involving air pollution and was followed by the Clean Air Acts of 1963 and 1970. The Clean Air Act of 1970 (1970 CAA, 42 U.S.C. § 7401 et seq.) authorized the development of comprehensive federal and state regulations to limit emissions from both stationary (industrial) sources and mobile sources.

The 1970 CAA set forth four major regulatory programs affecting stationary sources: the National Ambient Air Quality Standards (NAAQS), State Implementation Plans (SIPs), New Source Performance Standards, and National Emission Standards for Hazardous Air Pollutants. In Hawai'i, the DOH, Clean Air Branch, Air Quality program is defined by Hawai'i Administrative Rules (HAR) 11-60.1 and serves as the SIP approved by EPA.

The State DOH maintains 10 air monitoring stations on the island of Hawai'i. Established in 2007, the DOH maintains a monitoring station on the grounds of the N Elementary School to monitor SO_2 and $PM_{2.5}$ (in terms of micrograms per cubic meter (µg/m3)) from emissions from volcanic activities.

The quality of air in the general Nā'ālehu area is considered "Good" due to the typical flow of fairly constant northeasterly trade winds that disperse pollutants seaward. The rural nature of the Nā'ālehu area has no major stationary sources of air pollution. Existing sources of air pollution are emissions from motor vehicles traveling along Māmalahoa Highway and on the streets in the community; the low level of vehicle traffic tends to limit mobile sources of emissions.

Potential short-term effects from dust and exhaust due to construction activities will be minimized with BMPs such as water sprinkling and proper equipment maintenance. No long-term impacts on air quality resulting from operation of the collection system, the wastewater treatment and disposal facility, or the IWS are anticipated.

The DOH operates a network of air quality monitoring stations at various locations around the State. In September 2023, the DOH issued the Annual Summary 2022 Air Quality Data report (the most recent report) which provides the results from the network of air quality monitoring stations. The DOH maintains a monitoring station on the grounds of the Nā'ālehu. Established August 2007, the station was placed to monitor SO₂ and PM_{2.5} from volcanic emissions. In 2022,



Hawai'i was in attainment of the state annual SO2 standard. In 2015, Hawai'i was in attainment with the annual NAAQS for particulate matter with a diameter of 2.5 micrometers or less (PM2.5).

Volcanic eruptions are considered natural events and therefore EPA may exclude the exceedances of the 1-hour NAAQS from attainment determinations.

Potential short-term effects from dust and exhaust due to construction activities will be minimized with BMPs such as water sprinkling and proper equipment maintenance. No long-term impacts on air quality resulting from operation of the collection system, the wastewater treatment and disposal facility, or the IWS are anticipated.

6.5 Coastal Barrier Resources Act (16 U.S.C. § 3501)

In 1982, Congress passed the Coastal Barrier Resources Act (CBRA) (16 U.S.C. § 3501) to minimize the loss of human life; wasteful expenditure of federal revenues; and the damage to fish, wildlife, and other natural resources associated with the coastal barriers along the Atlantic and Gulf coasts and along the Great Lakes by restricting future federal expenditures and financial assistance which have the effect of encouraging development of coastal barriers, such as federal flood insurance through the National Flood Insurance Program.

The Coastal Barrier Resources Reauthorization Act of 2000 reauthorized the CBRA and directed the U.S. Fish and Wildlife Service (FWS) to complete a Digital Mapping Pilot Project that includes digitally produced draft maps for up to 75 John H. Chafee Coastal Barrier Resources System (CBRS) areas and a report to Congress that describes the feasibility and costs for completing digital maps for all CBRS areas.

Based on its location, the CBRA is not applicable to Hawai'i.

6.6 Coastal Zone Management Act (16 U.S.C. § 1451)

The Coastal Zone Management Act of 1972 (CZMA) (16 U.S.C §§ 1451-1464) was passed to establish a national policy to preserve, protect, develop, and where possible, restore or enhance, the resources of the Nation's coastal zone for this and succeeding generations and to encourage coastal states to develop and implement coastal zone management (CZM) programs. Each federal agency activity within or outside the coastal zone that affects any land or water use or natural resource of the coastal zone shall be carried out in a manner which is consistent to the maximum extent practicable with the enforceable policies of approved state management programs. Each federal agency carrying out an activity subject to the Act shall provide a consistency determination to the relevant state agency designated under § 1455(d)(6) of this title at the earliest practicable time.

In 1977, Hawai'i enacted HRS 205A (Coastal Zone Management). The CZM area encompasses the entire state, including all marine waters seaward to the extent of the state's police power and management authority, including the 12-mile U.S. territorial sea and all archipelagic waters. The objective and policies of the CZM Program and the Proposed Action's conformance are detailed in Table 6-1 below:



Table 6-1	Hawai`i Coastal Zone Management Act	S	NS	N/A
Recreational Resour	ces			
Objective: Provide co	astal recreational opportunities accessible to the public.			
Policies:	dination and funding of coastal regreational planning and	1		
(A) Improve coor management;	and and thinking of coastal recreational planning and			X
(B) Provide adequ	late, accessible, and diverse recreational opportunities in the			Х
i. Prote	ecting coastal resources uniquely suited for recreational			
activ	ities that cannot be provided in other areas;			X
ii. Requ recre reefs wher deve the desir	ativing restoration of coastal resources that have significant eational and ecosystem value including, but not limited to, coral is, surfing sites, fishponds, sand beaches, and coastal dune, in these resources will be unavoidably damaged by lopment; or requiring reasonable monetary compensation to State for recreation when restoration is not feasible or able;			x
iii. Provi cons recre	ding and managing adequate public access, consistent with ervation of natural resources, to and along shorelines with eational value;			x
iv. Provi recre	ding an adequate supply of shoreline parks and other actional facilities suitable public recreation;			Х
v. Ensu owne value natu	ring public recreational uses of county, state, and federally ed or controlled shoreline lands and waters having recreational e consistent with public safety standards and conservation of ral resources;			x
vi. Adop sourc recre	ting water quality standards and regulating point and nonpoint ces of pollution to protect, and where feasible, restore the ational value of coastal waters;			x
vii. Deve appro artifie	eloping new shoreline recreational opportunities, where opriate, such as artificial lagoons, artificial beaches, and cial reefs for surfing and fishing; and			x
viii. Enco recre or pe resol agair	uraging reasonable dedication of shoreline areas with eational value for public use as part of discretionary approvals ermits by the land use commission, board of land and natural urces, and county authorities; and crediting such dedication net the requirements of section 46-6.			x
Discussion: All project locations are at least 2 miles from the shoreline and, as such, the Proposed Action will not affect coastal recreational resources under any of the Proposed Alternatives.				
Historic Resources				
Objective : Protect, preserve, and, where desirable, restore those natural and manmade historic and prehistoric resources in the coastal zone management area that are significant in Hawaiian and American history and culture.				
Policies:	naluza significant archaeological resources	v		
(B) Maximize info	maryze significant archaeological resources;	^		
or salvage ope	erations; and			X
(C) Support state historic resour	goals for protection, restoration, interpretation, and display of rces.			x
Discussion: All project locations are at least 2 miles from the shoreline and, as such, the Proposed Action will not affect coastal historic resources under any of the Proposed Alternatives.				
Scenic and Open Space Resources				
Objective: Protect, preserve, and, where desirable, restore or improve the quality of coastal scenic and open space resources.				
Policies:				
(A) Identify value	d scenic resources in the coastal zone management area;			X



Table 6-1 Hawai'i Coastal Zone Management Act	S	NS	N/A
(B) Ensure that new developments are compatible with their visual environment			_
by designing and locating those developments to minimize the alteration of			Х
natural landforms and existing public views to and along the shoreline;			
(C) Preserve, maintain, and, where desirable, improve and restore shoreline			x
open space and scenic resources; and			~
(D) Encourage those developments that are not coastal dependent to locate in			х
Inland areas.			A
Discussion: All project locations are at least 2 miles from the shoreline and, as suc	n, the F	roposed	Action will
Coastal Ecosystems			
Objective: Protect valuable coastal ecosystems including reefs beaches and coastal	dunes	from disr	untion and
minimize adverse impacts on all coastal ecosystems	aunes,		aption and
Policies:			
(A) Exercise an overall conservation ethic and practice stewardship in the			
protection, use, and development of marine and coastal resources:			X
(B) Improve the technical basis for natural resource management:			x
(C) Preserve valuable coastal ecosystems of significant biological or economic			
importance, including reefs, beaches, and dunes:			X
(D) Minimize disruption or degradation of coastal water ecosystems by effective			
regulation of stream diversions, channelization, and similar land and water			х
uses, recognizing competing water needs; and			
(E) Promote water quantity and quality planning and management practices that			
reflect the tolerance of fresh water and marine ecosystems and maintain and			
enhance water quality through the development and implementation of point	X		
and nonpoint source water pollution control measures.			
Discussion: All project locations are at least 2 miles from the shoreline and, as suc	h, the F	roposed	Action will
not affect coastal recreational resources under any of the Proposed Alternatives.	,	•	
Economic Uses			
Objective: Provide public or private facilities and improvements important to the S	state's e	economy	in suitable
locations.			
Policies:			
(A) Concentrate coastal dependent development in appropriate areas;			Х
(B) Ensure that coastal dependent development and coastal related development are			X
located, designed, and constructed to minimize exposure to coastal hazards and			
adverse social, visual, and environmental impacts in the coastal zone management			
area; and			
(C) Direct the location and expansion of coastal developments to areas designated			Х
and used for such developments and permit reasonable long-term growth at those			
areas, and permit coastal development outside of presently designated areas when:			
i. Use of presently designated locations is not feasible;			X
ii. Adverse environmental effects and risks from coastal hazards are			X
minimized; and			
iii. The development is important to the State's economy;			X
Discussion : All project locations are at least 2 miles from the shoreline and, as suc	h, the F	Proposed	Action will
not affect coastal recreational resources under any of the Proposed Alternatives.			
Coastal Hazards			
Objective: Reduce hazard to life and property from coastal hazards.			
Policies:		1	ſ
(A) Develop and communicate adequate information about the risks of coastal			х
nazards;			-
(B) Control development, including planning and zoning control, in areas subject			X
to coastal nazaros;			
(c) Ensure that developments comply with requirements of the National Flood	X		
Insurance Program; and			


Table 6-1 Hawai'i Coastal Zone Management Act	S	NS	N/A
(D) Prevent coastal flooding from inland projects.			X
Discussion: All project locations are at least 2 miles from the shoreline and at least	t 600 fe	et above	mean sea
level (msl). Based on the location, the Proposed Action will not be subject to (and v	vill not	exacerbat	te) coastal
hazards and do not include improvements related to tsunami, storm waves, stream flo	oding e	rosion, s	ubsidence,
and pollution under any of the Proposed Alternatives.	5		
Managing Development			
Objective: Improve the development review process, communication, and public parti-	cipation	in the ma	nagement
of coastal resources and hazards.			
Policies:			
(A) Use, implement, and enforce existing law effectively to the maximum extent			
possible in managing present and future coastal zone development:			X
(B) Facilitate timely processing of applications for development permits and			
resolve overlapping or conflicting permit requirements; and			X
(C) Communicate the potential short and long-term impacts of proposed			
significant coastal developments early in their life cycle and in terms			
understandable to the public to facilitate public participation in the planning	X		
and review process			
Discussion: A total of five community outreach sessions to discuss the Revised A	NC wer	e conduc	ted in the
Pahala and Naalehu communities between March 2022 and Sentember 2023 In	late Or	toher 20	123 nublic
preference surveys to gather input on the four feasible ontions were mailed out to	194 lot	owners i	in the ICC
closure area A public meeting was held November 16 at the Nā'ālehu Community	Center	to engag	e with the
public to solicit feedback on the options explored in the PER and to provide a genera	l overvie	w of the	FID being
prepared A second round of surveys was mailed out to owners of affected lots in Feb	ruary 20)24 A se	mi-annual
community informational meeting was held on February 29, 2024 in Pāhala and the n	ext semi	-annual o	
informational meeting will be held in Δugust 2024 in Nā'ālehu	ext Serii		Communicy
The Project Area is located at least 2 miles from the coast, at least 600 feet above	io mol		
	e msi.	and do n	not involve
management of coastal resources and hazards under any of the Proposed Alternatives	6 msi, S.	and do n	ot involve
management of coastal resources and hazards under any of the Proposed Alternatives Public Participation	6.	and do n	iot involve
management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal management	e msi, 5.	and do n	ot involve
management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies:	jement.	and do n	not involve
management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal management processes: (A) Promote public involvement in coastal zone management processes:	gement.	and do n	not involve
management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal management Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of	jement.	and do n	not involve
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials publiced reports staff contact, and public workshops 	gement.	and do n	x
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues developments 	gement.	and do n	x X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities: and 	gement.	and do n	x X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops policy dialogues and site-specific mitigation to responde 	gement.	and do n	x X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts 	jement.	and do n	x X X X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Bevised A 			X X X X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pabala and Naalebu communities between March 2022 and Sentember 2023. In 	jement.	e conductor	X X X X ted in the
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to 	OC wer late Oc	e conductober 20	X X X X X tted in the 23, public in the LCC
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'šlehu Community 	OC wer late Oc 194 lot Center	e conductober 20 owners i	X X X X X x ted in the 23, public in the LCC e with the
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER and to provide a general 	OC wer late Oc 194 lot Center	e conductober 20 owners ito engag	X X X X X x x x x x x x x x x x x x x x
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a general prepared A second round of surveys was mailed out to owners of affected lots in Feb 	OC wer late Oc 194 lot Center l overvie	e conductober 20 owners into engage w of the	X X X X X x ted in the 23, public in the LCC e with the EID being EID being
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pābala and the period. 	OC wer late Oc 194 lot center l overvie ruary 20	e conductober 20 owners into engage w of the 024. A sec-	X X X X X X X X X X X X X X X X X X X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting was held on February 29, 2024 in Pāhala and the neuropartional meeting will be held in August 2024 in Nā'ālehu 	OC wer late Oc 194 lot Center lovervie ruary 20 ext semi	e conductober 20 owners in to engage w of the 024. A sec- annual c	X X X X X X X X X X X X X X X X X X X
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting was held in August 2024 in Nā'ālehu. 	OC wer late Oc 194 lot Center lovervie ruary 20 ext semi	e conductober 20 owners in to engagiew of the 24. A second	X X X X tted in the 23, public in the LCC e with the EID being emi-annual community
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal management Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting will be held in August 2024 in Nā'ālehu. Beach Protection and Coastal Dune Protection 	OC were late Oc 194 lot Center l overvie ruary 20 ext semi	e conductober 20 owners into engagew of the D24. A set annual co	X X X X tted in the 23, public in the LCC e with the EID being emi-annual community
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the netinformational meeting was held on February 29, 2024 in Pāhala and the netinformational meeting will be held in August 2024 in Nā'ālehu. Beach Protection and Coastal Dune Protection Objective: (A) Protect beaches and coastal dunes for; Public and recreation: 	OC wer late Oc 194 lot Center l overvie ruary 20 ext semi	e conductober 20 owners in to engage w of the 24. A second	X X X X ted in the 23, public in the LCC e with the EID being emi-annual community
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal management Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held non February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held coastal dunes for; i. Public and recreation; ii. The benefit of coastal ecosysteme 	OC wer late Oc 194 lot Center l overvie ruary 20 ext semi	e conductober 20 owners in to engage w of the 24. A set annual c	X X X X ted in the 23, public in the LCC e with the EID being emi-annual community
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on February 29, 2024 in Pāhala and the minformational meeting was held on Septruary 29, 2024 in Pāhala and the minformational meeting will be held in August 2024 in Nā'ālehu. Beach Protection and Coastal Dune Protection Objective: (A) Protect beaches and coastal dunes for; The benefit of coastal ecosystems Use as natural buffers against coastal barardst and 	gement. gem	e conductober 20 owners in to engage w of the D24. A set annual co	X X X X ted in the 23, public in the LCC e with the EID being emi-annual community
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting was held on Sebruary 29, 2024 in Pāhala and the net informational meeting was held on February 29, 2024 in Pāhala and the net informational meeting was held on Sebruary 29, 2024 in Pāhala and the net informational meeting was held on Sebruary 29, 2024 in Pāhala and the net informational meeting was held on Sebruary 29, 2024 in Pāhala and the net informational meeting will be held in August 2024 in Nā'ālehu. 	OC wer late Oc 194 lot Center l overvie ruary 20 ext semi	e conductober 20 owners ito engage w of the 24. A set annual c	X X X X ted in the 23, public in the LCC e with the EID being emi-annual community
 management of coastal resources and hazards under any of the Proposed Alternatives Public Participation Objective: Stimulate public awareness, education, and participation in coastal manage Policies: (A) Promote public involvement in coastal zone management processes; (B) Disseminate information on coastal management issues by means of educational materials, published reports, staff contact, and public workshops for persons and organizations concerned with coastal issues, developments, and government activities; and (C) Organize workshops, policy dialogues, and site-specific mitigation to respond to coastal issues and conflicts. Discussion: A total of five community outreach sessions to discuss the Revised A Pahala and Naalehu communities between March 2022 and September 2023. In preference surveys to gather input on the four feasible options were mailed out to closure area. A public meeting was held November 16, at the Nā'ālehu Community public to solicit feedback on the options explored in the PER, and to provide a genera prepared. A second round of surveys was mailed out to owners of affected lots in Feb community informational meeting was held on February 29, 2024 in Pāhala and the neinformational meeting was held on February 29, 2024 in Pāhala and the neinformational meeting was held no rebrary 29, 2024 in Pāhala and the neinformational meeting was held on February 29, 2024 in Pāhala and the neinformational meeting was held on setter 2023. In Pāhala and the neinformational meeting was held not be for; i. Public and recreation; ii. The benefit of coastal ecosystems iii. Use as natural buffers against coastal hazards; and, iv. Coordinate and fund beach management and protection. 	OC wer late Oc 194 lot Center l overvie ruary 20 ext semi	e conductor owners in to engage w of the 024. A sec i-annual c	X X X X ted in the 23, public in the LCC e with the EID being emi-annual community



Table 6-1 Hawai'i Coastal Zone Management Act	S	NS	N/A
 (A) Locate new structures inland from the shoreline setback to conserve oper space, minimize interference with natural shoreline processes, and minimize loss of improvements due to erosion; 			x
(B) Prohibit construction of private shoreline hardening structures including seawalls and revetments, at sites having sand beaches and at site where shoreline hardening structures interfere with existing recreational and waterline activities; and			x
(C) Minimize the construction of public shoreline hardening structures including seawalls and revetments, at sites having sand beaches and at sites where shoreline hardening structures interfere with existing recreational and waterline activities;			x
(D) Minimize grading of and damage to coastal dunes;			X
(E) Prohibit private property owners from creating a public nuisance by inducing or cultivating the private property owner's vegetation in a beach transi corridor; and			x
(F) Prohibit private property owners from creating a public nuisance by allowing the private property owner's unmaintained vegetation to interfere o encroach upon a beach transit corridor.	-		x
Discussion: All project locations are at least 2 miles from the shoreline. The Proprimprovements that would affect public use of beaches under any of the Proposed All	sed Actionsed Actionsed Actions	on does 1 5.	not include
Marine and Coastal Resources			
Objective: Promote the protection, use, and development of marine and coast sustainability.	al resour	ces to a	ssure their
Policies:			
 (A) Ensure that the use and development of marine and coastal resources are ecologically and environmentally sound and economically beneficial; 	•		x
 (B) Coordinate the management of marine and coastal resources and activities to improve effectiveness and efficiency; 	;		X
(C) Assert and articulate the interests of the State as a partner with federal agencies in the sound management of ocean resources within the United States exclusive economic zone;			x
(D) Promote research, study, and understanding of ocean and coastal processes impacts of climate change, marine life, and other ocean resources in orde to acquire and inventory information necessary to understand how coastal development activities relate to and impact ocean and coastal resources; and	-		x
(E) Encourage research and development of new, innovative technologies fo exploring, using, or protecting marine and coastal resources.			X
Discussion: All project locations are at least 2 miles from the shoreline. The Proprimprovements that would affect development of marine and coastal resources Alternatives.	osed Action under ar	on does i iy of the	not include Proposed

6.7 Endangered Species Act (16 U.S.C. § 1531)

On December 28, 1973, the Endangered Species Act (16 U.S.C. § 1531) was passed and, over the years, has been amended a number of times. The stated purpose of the original Act was to provide a means whereby the ecosystems upon which endangered species and threatened species depend may be conserved, to provide a program for the conservation of such endangered species and threatened species, and to take such steps as may be appropriate to achieve the purposes of various related treaties and conventions. The provisions of the Act are administered by the FWS and the National Oceanic and Atmospheric Administration (NOAA), National Marine Fisheries Service (NMFS). The FWS has primary responsibility for terrestrial and freshwater organisms, while NOAA/NMFS is mainly responsible for marine wildlife.



Section 7 of the Act, Interagency Cooperation (16 U.S.C. § 1536), states each federal agency shall, in consultation with and with the assistance of the Secretary of the Interior, ensure that any action authorized, funded, or carried out by such agency (an "agency action") is not likely to jeopardize the continued existence of any endangered species or threatened species or result in the destruction or adverse modification of habitat of such species which is determined, after consultation as appropriate with affected states, to be critical, unless such agency has been granted an exemption for such action.

As discussed in Section 4.6 of this EA/ EID, two biological surveys were conducted on the sites: one in 2012 and one in 2021; both surveys were conducted by AECOS Inc. (AECOS, 2021). During the botanical, avian, and mammalian resources surveys no listed or proposed-for-listing species under federal statutes was recorded. The assessment observed avian diversity and densities present within the Project Area and its surrounding vicinity.

Based on the findings of the botanical and biological field surveys, construction activities associated with the new collection system and wastewater treatment and disposal facility are not anticipated to result in adverse impacts to botanical and faunal resources in the Nā'ālehu area. Mitigation measures and BMPs as recommended by the FWS and AECOS will be implemented during construction and following build out to ensure that the project minimizes all impacts to listed species to the maximum extent practicable.

On May 31, 2024, the DOH initiated consultation with the U.S. Fish and Wildlife Service (FWS) pursuant to Section 7 of the Endangered Species Act (See Appendix C). The consultation requested the FWS' concurrence with the determination that the Proposed Action "may affect, but is not likely to adversely affect" federally listed species. In a letter dated June 21, 2024, the FWS provided concurrence with this determination. Additionally, the FWS provided avoidance and minimizations to be implemented to avoid adverse effects (See Appendix D).

6.8 Environmental Justice Executive Order 12898

Executive Order 12898, Environmental Justice (full title Federal Actions to Address Environmental Justice to Minority and Low Income Populations), was signed on February 11, 1994. The intent of Executive Order 12898 is to avoid disproportionately high adverse human health or environmental effects of projects on minority and low income populations. Executive Order 12898 also requires federal agencies to ensure that minority and low-income communities have adequate access to public information related to health and the environment.

The 2022 American Community Survey (ACS) (5-Year Estimates) is the most recent information related to socioeconomic conditions in the state and County. The 2022 ACS includes Hawai'i Geographic Area Profiles – Census Designated Places: Neighbor Islands. The ACS noted it is the Census Bureau's Population Estimates Program that produces and disseminates the official estimates of the population for the nation, states, counties, cities and towns and estimates of housing units for states and counties.

For purposes of this assessment, and to correspond with the available ACS demographic characteristics data, "low income" is defined as having a household income of less than \$24,999;



"minority" is defined as any race population other than White; and "children" is defined as the "Under 5 to 19" age category.

The Project Area is located within the Nā'ālehu Census Designated Place (CDP). The 2022 American Community Survey (ACS) was reviewed for the Nā'ālehu CDP and Hawai'i County in Section 4.14 of the EA.

6.9 Farmland Protection Policy Act (7 U.S.C. § 4201)

The Agriculture and Food Act was passed in 1981 and contained the Farmland Protection Policy Act (FPPA) (7 U.S.C. § 4201). The stated purposes of the FPPA are to: 1) minimize the extent to which federal programs contribute to the unnecessary and irreversible conversion of farmland to nonagricultural uses; and 2) assure that federal programs are administered in a manner that, to the extent practicable, will be compatible with state, unit of local government, and private programs and policies to protect farmland. "Farmland" subject to FPPA requirements does not have to be currently used for cropland.

The FPPA is administered by the U.S. Department of Agriculture (USDA), National Resources Conservation Service (NRCS). "Farmland", as used in the FPPA, includes prime farmland, unique farmland, and land of statewide or local importance, as defined by the State of Hawai'i Department of Agriculture.

Per the Agricultural Lands of Importance to the State of Hawai'i (ALISH) Classification System, all Project locations are located in "not rated" lands which is assigned to developed communities. Construction of the collection system within the County roads would not affect agricultural lands.

6.10 Fish and Wildlife Coordination Act (16 U.S.C § 661)

The Fish and Wildlife Coordination Act (16 U.S.C § 661), enacted on March 10, 1934, was amended on August 12, 1958. The purpose of the Act is to recognize the vital contribution of wildlife resources to the Nation, the increasing public interest and significance, and to provide that wildlife conservation shall receive equal consideration and be coordinated with other features of water-resource development programs through the effectual and harmonious planning, development, maintenance, and coordination of wildlife conservation. The Act defines wildlife and wildlife resources as birds, fishes, mammals and all other classes of wild animals, and all types of aquatics and land vegetation upon which wildlife is dependent (16 U.S.C. § 666b).

The Secretary of the Interior is authorized (1) to provide assistance to, and cooperate with, federal, state, and public or private agencies and organizations in the development, protection, rearing, and stocking of all species of wildlife, and their habitat; in controlling losses of the from disease or other causes; in minimizing damages from overabundant species; and in providing public shooting and fishing areas, including easements across public lands; (2) to make surveys and investigations of the wildlife of the public domain, including lands and waters acquired or controlled by any agency; and (3) to accept donations of land and contributions of funds in furtherance of the pupposes of the Act.



Specifically, the Act states that "whenever the waters of any stream or other body of water are proposed or authorized to be impounded, diverted, the channel deepened, or the stream or other body of water otherwise controlled or modified for any purpose whatever, including navigation and drainage, by any department or agency of the United States, or by any public or private agency under Federal permit or license, such department or agency first shall consult with the United States Fish and Wildlife Service" (16 U.S.C. § 662(a)). The consultation may result in a report of recommendations by FWS that should be adopted to prevent the loss of or damage to wildlife resources. The provisions of the Act do not apply to impoundments of water less than 10 acres.

The Proposed Action does not include any impoundment of water and therefore a Fish and Wildlife Coordination Act review and/or consultation pursuant to 16 U.S.C. § 662 is not required under any of the Proposed Alternatives.

6.11 Floodplain Management (Executive Order 11988, as amended by Executive Orders 12148 and 13690)

Executive Order 11988, Floodplain Management, dated May 24, 1977 requires federal agencies to avoid, to the extent possible, the long- and short-term adverse impacts associated with the occupancy and modification of floodplains and to avoid direct and indirect support of floodplain development wherever there is a practicable alternative.

In accomplishing this objective, "each agency shall provide leadership and shall take action to reduce the risk of flood loss, to minimize the impact of floods on human safety, health, and welfare, and to restore and preserve the natural and beneficial values served by floodplains in carrying out its responsibilities."

The Proposed Action is not located within a 100-year floodplain area, will incorporate stormwater BMPs to manage runoff in accordance with state requirements, and will be designed to ensure sufficient capacity for assimilation of peak effluent flow rates and precipitation from the design storm event. The Proposed Action therefore will not have an adverse impact on floodplains and will minimize the risk of flood-related impacts on surrounding properties under any of the Proposed Alternatives.

6.12 Magnuson-Stevens Fishery Conservation and Management Act (16 U.S.C. § 1801)

The 1996 Sustainable Fishery Act amendments to the Magnuson-Stevens Fishery Conservation and Management Act and subsequent Essential Fish Habitat (EFH) Regulatory Guidelines (NOAA, 2002) describe provisions to identify and protect habitats of federally managed marine and anadromous fish species. Under the various provisions, federal agencies that fund, permit, or undertake activities that may adversely affect EFH are required to consult with the NMFS.

Congress defines EFH as "those waters and substrate necessary to fish for spawning, breeding, feeding, or growth to maturity." EFH is further defined by the existing regulations (NOAA-NMFS, 2007; NOAA, 2002). "Waters" include aquatic areas and their associated physical, chemical, and



biological properties that are used by fish and may include aquatic areas historically used by fish where appropriate; "substrate" includes sediment, hard bottom, structures underlying the waters, and associated biological communities; "necessary" means the habitat required to support a sustainable fishery and the managed species' contribution to a healthy ecosystem; and "spawning, breeding, feeding, or growth to maturity" covers a species' full life cycle.

All project locations are at least 2.5 miles from the shoreline. The Proposed Action will not adversely impact EFH under any of the Proposed Alternatives.

6.13 Marine Mammal Protection Act (16 U.S.C. §§ 1361 et seq.)

The Marine Mammal Protection Act (MMPA) (16 U.S.C. §§ 1361 et seq.), protects all marine mammals. The MMPA includes a general moratorium on the taking and importing of marine mammals, and prohibits, with certain exceptions, the "take" of marine mammals in U.S. waters and by U.S. citizens on the high seas, and the importation of marine mammals and marine mammal products into the U.S. Jurisdiction for MMPA is shared by the FWS and NMFS. The FWS Branch of Permits is responsible for issuing take permits when exceptions are made to MMPA. Under the exception for incidental taking, the FWS or the NMFS must find that the total taking over the five-year period will have a "negligible impact" and will not adversely affect the availability of the marine mammal species or stock for subsistence use by natives.

All project locations are at least 2.5 miles from the shoreline. The Proposed Action will not adversely impact marine mammal communities and will not encourage any "take" of marine mammals under any of the Proposed Alternatives.

6.14 Migratory Bird Treaty Act (16 U.S.C. §§ 703 et seq.)

The Migratory Bird Treaty Act (MBTA) and Executive Order 13186 (Responsibilities of Federal Agencies to Protect Migratory Birds) provide for the protection of migratory birds. The MBTA of 1918, as amended (16 U.S.C. §§ 703-712) makes it unlawful to, among other things, pursue, hunt, take, capture, kill, transport or import any species listed under the Act. The Act implements conventions between the U.S., Great Britain, Mexico, Japan, and the former Soviet Union.

Executive Order 13186 was issued to assist federal agencies with their efforts to comply with the MBTA. It should be noted that the Executive Order does not constitute any legal authorization that in any way supersedes the requirements outlined in the MBTA. The Executive Order directs federal agencies undertaking actions that have or are likely to have a measurable adverse impact on migratory bird populations to develop and implement a Memorandum of Agreement with the FWS addressing the conservation of these populations.

As described in Section 4.6 of this EA / EID, biological surveys conducted in 2012 and 2021 by AECOS recorded one indigenous migratory shorebird species in the Project Area. The Pacific Golden-Plover (Pluvialis fulva) is commonly seen across the state between late September and the end of April each year. This species migrates to Arctic breeding grounds in late April or the first week of May. Minimization measures discussed herein are intended to minimize any potential impacts on flora and fauna that could result from the construction and operations of the



wastewater treatment and disposal facility and associated infrastructure. The proactive consultation with relevant authorities ensures compliance with regulations and protection of sensitive species.

6.15 National Historic Preservation Act (54 U.S.C. § 300101)

The National Historic Preservation Act (NHPA) of 1966 (54 U.S.C. § 300101) requires a federal agency undertaking an action/project consider of the effect of the project on any historic property defined as a district, site, building, structure, or object that is included in or eligible for inclusion in the National Register of Historic Places.

Section 106 of the NHPA (54 U.S.C. § 306108) requires a federal agency having direct or indirect jurisdiction over a federal or federally assisted undertaking to take into account the effect of the undertaking on any historic property. An "undertaking" includes a "project, activity, or program funded in whole or in part under the direct or indirect jurisdiction of a Federal agency" (54 U.S.C. § 300320). Because the Nā'ālehu LCC Replacement Project will be funded using federal funds, it is considered an "undertaking" and is subject to the NHPA.

The Act requires the federal agency's preservation-related activities to be carried out in consultation with other federal, state, and local agencies, Indian tribes, Native Hawaiian organizations (54 U.S.C § 306102).

As discussed in Section 4.7 of this EA / EID, an AIS Report was completed by Cultural Surveys Hawai'i in May 2024. The AIS is was designed to be compliant with both Federal and Hawai'i State environmental and historic preservation review legislation. The AIS was conducted to assess any historic properties for integrity and site significance in accordance with HAR §13-275-6. This report is also intended to support any project-related historic preservation consultation with consulting parties, such as state and county agencies and interested Native Hawaiian Organizations (NHOs) and community groups, if applicable.

The AIS identified four previously documented and seven newly documented historic properties in the Project Area. In accordance with federal regulations (36 CFR 800.5), the AIS results support the determination made by the Hawai'i State DOH on behalf of the EPA of "no adverse effect." In accordance with HAR §13-275-7, the Hawai'i State DOH and COH-DEM have determined the project effect is "effect, with proposed mitigation commitments." As stated in the AIS, the identified sites have been fully documented, mitigating any potential effect, and no further work is recommended.

Although historic properties have been determined to require no further work, the County will consider further communication and coordination with identified stakeholders to develop feasible and reasonable accommodations on either preserving historic properties or alternatives that integrate the broader preservation goals of the Waikapuna property.



Archaeological monitoring is recommended during all project ground-disturbing activities to facilitate identification for information purposes, especially of any subsurface lava tubes that may contain historic properties. Archaeological monitoring will proceed under an archaeological monitoring plan that meets the requirements of HAR §13-279-4, which will be submitted to the SHPD for review and acceptance prior to project initiation.

The contract drawings will state that, should archaeological sites such as walls, platforms, pavements or mounds, or remains such as artifacts, burials, concentrations of shell or charcoal be encountered during construction activities, work shall cease immediately and the find shall be protected from further damage. The contractor shall immediately contact the State Historic Preservation Division (SHPD), who will assess the significance of the find and recommend an appropriate mitigation measure, if necessary.

In addition to the AIS, the County is obligated to comply with the NHPA. On June 3, 2024 the DOH, on behalf of the County, initiated consultation for this project in accordance with Section 106 of the NHPA. Consultation letters were sent to various Native Hawaiian Organizations inviting comments from organizations that may attach religious or cultural significance to properties affected by the proposed actions. NHPA Section 106 consultation correspondence can be found in Appendix F of this EA / EID. The SHPD has not yet responded to the consultation request. As such, any future consultation with the SHPD shall be incorporated into the Final EA / EID.

6.16 6-15Protection of Wetlands (Executive Order 11990 (1977), as amended by Exec6-15utive Order 12608 (1997))

Executive Order 11990, Protection of Wetlands, dated 1977 requires federal agencies to avoid, preserve, or mitigate effects of new construction projects on lands which have been designated wetlands. Executive Order 11990 states in order to avoid to the extent possible the long- and short-term adverse impacts associated with the destruction or modification of wetlands and to avoid direct or indirect support of new construction 1. (a) Each agency shall provide leadership and shall take action to minimize the destruction, loss or degradation of wetlands, and to preserve and enhance the natural and beneficial values of wetlands in carrying out the agency's responsibilities for (1) acquiring, managing, and disposing of federal lands and facilities; and (2) providing federally undertaken, financed, or assisted construction and improvements; and (3) conducting federal activities and programs affecting land use, including but not limited to water and related land resources planning, regulating, and licensing activities.

The National Wetlands Inventory (NWI) Wetlands Mapper and U.S. Geological Survey (USGS) topographic maps identify no wetland features or streams within the Proposed WWTP Site, at the three LCCs, or within the Proposed Collection System Area.

6.17 Rivers and Harbors (33 U.S.C. § 403)

Originally enacted on March 3, 1899, the Rivers and Harbors Appropriation Act of 1899 affects navigable waters of the U.S. Section 10 of the Act states the creation of any obstruction not affirmatively authorized by Congress, to the navigable capacity of any of the waters of the United



States is prohibited; and it shall not be lawful to build or commence the building of any wharf, pier, dolphin, boom, weir, breakwater, bulkhead, jetty, or other structures in any port, roadstead, haven, harbor, canal, navigable river, or other water of the United States, outside established harbor lines, or where no harbor lines have been established, except on plans recommended by the Chief of Engineers and authorized by the Secretary of the Army; and it shall not be lawful to excavate or fill, or in any manner to alter or modify the course, location, condition, or capacity of, any port, roadstead, haven, harbor, canal, lake, harbor or refuge, or enclosure within the limits of any breakwater, or of the channel of any navigable water of the United States, unless the work has been recommended by the Chief of Engineers and authorized by the Chief of Engineers and authorized by the Chief of Engineers of the Secretary of the Army prior to beginning the same (33 U.S.C. § 403).

All project locations are at least 2 miles from the shoreline. According to USGS geologic maps, Honu'apo Bay is located approximately 2.5 miles from the Nā'ālehu community. The nearest stream within the Project Area is Nā'ālehu Stream (See Figure 5-3). Nā'ālehu Stream is non-perennial which originates from flows that begin at Ka'ū Forest Reserve. Proposed Action will not affect navigable waters under any of the Proposed Alternatives.

6.18 Safe Drinking Water Act (42 U.S.C. § 300f)

The Safe Drinking Water Act (SDWA) of 1974 (42 U.S.C. § 300f) was established to protect the quality of all waters actually or potentially designed for drinking use from both underground and aboveground sources. The SDWA authorizes EPA to establish minimum standards to protect potable water with which all owners or operators of public water systems must comply; to oversee the agencies which can be approved to implement these rules on EPA's behalf, such as state governments; and to encourage attainment of secondary standards (nuisance-related). Section 1424(e) of the SDWA of 1974 (Public Law 93-523, 42 U.S.C. 300 et. seq also established the Sole Source Aquifer program which states that no commitment for federal financial assistance (through a grant, contract, loan guarantee, or otherwise) may be entered into for any project which the EPA Administrator determines may contaminate such aquifer through a recharge zone so as to create a significant hazard to public health.

The Proposed Action does not establish a drinking water system, and no Sole Source Aquifers are present on the Island of Hawai'i. The Proposed Action will provide the infrastructure necessary to enable the County to comply with the SDWA by replacing the existing outdated and federally banned wastewater systems that pose a threat to underground sources of drinking water.

6.19 Wild and Scenic Rivers Act (16 U.S.C. §§ 1271-1287)

The Wild and Scenic Rivers Act, 16 U.S.C. §§ 1271-1287, declares that certain selected rivers with their immediate environments, which possess outstandingly remarkable scenic, recreational, geologic, fish and wildlife, historical, cultural, or other similar values, shall be preserved in their free-flowing condition for the enjoyment of present and future generations. The State of Hawai'i has no designated wild and scenic rivers.

The Wild and Scenic Rivers Act is not applicable to this project.



6.20 Clean Water Act (33 U.S.C. § 1251 et seq.)

The Clean Water Act established the basis for regulating discharges of pollutants into waters of the U.S. Enacted in 1948, it was originally called the Federal Water Pollution Control Act but became known as the Clean Water Act with the amendments of 1972. Section 404 of the Clean Water Act regulates the discharge of dredged or fill material into waters of the U.S. and adjacent wetlands from development, water resource projects, mining or other infrastructure projects. Activities are regulated through a permit process that is administered under the responsibility of the U.S. Army Corps of Engineers. Permits may be issued as either Individual Permits for projects with potentially significant impacts or general permits for projects with only minimal adverse effects.

The NWI Wetlands Mapper and USGS topographic maps identify no wetland features or streams within the WWTP Site, at the two LCCs, within the Proposed Collection System Area, or within the future IWS lots. Biological and archeological field survey reports do not indicate any standing water or evident wetland vegetation within the WWTP Site and Collection System Area.

As discussed in Section 4.6 of this EA / EID, no streams or wetlands occur in the Project Area, although the WWTP site itself is located within a shallow drainage swale and runoff flow will be directed around the WWTP infrastructure in a new channel. Biological surveys observed no indications of water flow in the swale at the Project site. A concrete drainage channel crossed by the proposed force main on Melia Street is similar in directing runoff through town into shallow swale further downslope. Neither of these drainageways direct flows to streams or outlets on the ocean shore. Annual rainfall in the Nā'ālehu area is on the order of 40 in (1020 mm) and the porous geology is not conducive to stream formation.

Because no wetland resources are present and no impacts to wetlands are anticipated due to the nature and design of the WWTP and collection, a Clean Water Act Section 404 permit is not required.

In addition to the above, the Clean Water Act was amended by the Federal Water Quality Act of 1987 which established provisions for a Clean Water State Revolving Fund (33 U.S.C. § 1383), a financial assistance program for water infrastructure projects. The program capitalizes on a partnership between EPA and states to provide loans to eligible recipients through state programs that act as environmental infrastructure banks providing low-interest loans. As stated in Section 2.1.2, the Naalehu LCC Replacement Project is being funded in part by the State of Hawai'i DOH Clean Water State Revolving Fund.



(This page intentionally left blank)



CHAPTER 7: ANTICIPATED DETERMINATION OF FONSI

Potential impacts of the Proposed Action on a range of environmental resource criteria have been evaluated in accordance with the significance criteria outlined in § 11-200.1-13, HAR. Discussion of the Proposed Project's conformance to the criteria is presented as follows: *1) Irrevocably commit a natural, cultural, or historic resource;*

The Proposed Action would not result in the loss or destruction of natural resources. The collection system will be constructed primarily within areas that were previously disturbed

during the construction of existing County streets, plus two backyard easements.

An AIS Report was completed by Cultural Surveys Hawai'i in May 2024. The AIS is was designed to be compliant with both Federal and Hawai'i State environmental and historic preservation review legislation. The AIS was conducted to assess any historic properties for integrity and site significance in accordance with HAR §13-275-6. This report is also intended to support any project-related historic preservation consultation with consulting parties, such as state and county agencies and interested Native Hawaiian Organizations (NHOs) and community groups, if applicable.

The AIS identified four previously documented and seven newly documented historic properties in the Project Area. In accordance with federal regulations (36 CFR 800.5), the AIS results support the determination made by the Hawai'i State DOH on behalf of the EPA of "no adverse effect." In accordance with HAR §13-275-7, the Hawai'i State DOH and COH-DEM have determined the project effect is "effect, with proposed mitigation commitments."

Archaeological monitoring is recommended during all project ground-disturbing activities to facilitate identification for information purposes, especially of any subsurface lava tubes that may contain historic properties. Archaeological monitoring will proceed under an archaeological monitoring plan that meets the requirements of HAR §13-279-4, which will be submitted to the SHPD for review and acceptance prior to project initiation.

Although historic properties have been determined to require no further work, the County will consider further communication and coordination with identified stakeholders to develop feasible and reasonable accommodations on either preserving historic properties or alternatives that integrate the broader preservation goals of the Waikapuna property.

2) Curtail the range of beneficial uses of the environment;

The Proposed Action will not curtail the range of beneficial uses of the environment. The proposed use of the site is a continuation of this existing use, which is compatible with the surrounding uses. No new uses or improvements are being proposed that would significantly intensify the land use, alter the land, or commit resources that would curtail the range of future beneficial uses of the environment.



3) Conflict with the State's environmental policies or long-term environmental goals established by law;

The Proposed Action will not conflict with the long-term environmental policies, goals, and guidelines of the State of Hawai'i as noted throughout Chapter 5 of the EID/EA.

Moreover, short-term impacts associated with various construction activities will be mitigated through best management practices noted throughout Chapter 5 of the EA.

4) Have a substantial adverse effect on the economic welfare, social welfare, or cultural practices of the community and State;

The Proposed Action will not have a significant adverse effect on the economic welfare, social welfare, or cultural practices of the State as discussed in Chapters 5 and 6 of the EID /EA.

In the short-term, construction jobs will be created to develop the Proposed Action and construction expenditures will provide positive benefits to the local economy but not at a level that would generate any significant population expansion. Further, long-term facility maintenance and operational jobs would be created for the WWTP.

In the long-term, the Proposed Action is intended to allow the County to provide wastewater collection, treatment, and disposal facilities that meet the needs of the Nā'ālehu community. It will be an integral part of the infrastructure needed to maintain the health and welfare of the community. Therefore, implementation and long-term operation of the Proposed Action is not anticipated to have a substantial short- nor long-term adverse impacts on public health. Typical short-term construction-related impacts (e.g., noise and air quality) are anticipated; however, they will be temporary in nature and will comply with Federal, State, and County regulations as discussed in Chapter 5 of the EID/EA.

5) Have substantial adverse effect on public health;

The Proposed Action will involve the design, construction, and operation of wastewater collection, treatment, and disposal facilities that will maintain and enhance the public health of the Nā'ālehu community. Thus, the Proposed Action shall have a beneficial effect on public health.

6) Involve adverse secondary impact, such as population changes or effects on public facilities;

The Proposed Action involves the design, construction, and operation of wastewater collection, treatment, and disposal facilities within the Nā'ālehu community. The proposed use is consistent with current use of the area and compatible with surrounding uses. The Proposed Action is not anticipated to induce population growth and, therefore, no secondary impacts associated with population changes or effects on public facilities are anticipated with implementation of the proposed action.



7) Involve a substantial degradation of environmental quality;

The Proposed Action is not anticipated to substantially degrade environmental quality. Long-term impacts to air and water quality, noise levels and natural resources will be minimal. Typical short-term construction-related impacts (e.g., noise and air quality) are anticipated, but will be temporary and will comply with State and County regulations as discussed in Chapter 5 of the EID / EA.

8) Be individually limited but cumulatively have substantial adverse effect upon the environment or involves a commitment for larger actions;

The Proposed Action is not anticipated to have a considerable effect upon the environment as discussed in Chapter 4 of the EID / EA. There are no commitments for further action beyond the scope presented within this EID / EA.

9) Have substantial adverse effect on rare, threatened, or endangered species, or its habitat;

As discussed in Section 4.6 of this EID / EA, two biological surveys were conducted on the sites: one in 2012 and one in 2021; both surveys were conducted by AECOS Inc. (AECOS, 2021). During the botanical, avian, and mammalian resources surveys no listed or proposed-for-listing species under federal statutes was recorded. The assessment observed avian diversity and densities present within the Project Area and its surrounding vicinity.

Based on the findings of the botanical and biological field surveys, construction activities associated with the new collection system and wastewater treatment and disposal facility are not anticipated to result in adverse impacts to botanical and faunal resources in the Nā'ālehu area. Mitigation measures and BMPs as recommended by the FWS and AECOS will be implemented during construction and following build out to ensure that the project minimizes all impacts to listed species to the maximum extent practicable.

On June XX, 2024, the designated non-federal representative for consultations under Section 7 of the Endangered Species Act, on behalf of EPA and the County of Hawai'i, requested concurrence from the FWS that the Nā'ālehu LCC Closure Project is not likely to adversely affect federally listed threatened and endangered species or critical habitat located within the Project Area.

10) Have a substantial adverse effect on air or water quality or ambient noise levels;

No long-term significant impacts to air quality, water quality, or noise levels within the project site are anticipated as a result of the construction and operation of the Proposed Action.

Respective contractors will be responsible for minimizing air quality impacts during the various phases of construction. Exhaust emissions from construction vehicles are anticipated to have negligible impacts on air quality in the project vicinity as the emissions would be relatively small and readily dissipated.

No short or long-term significant impacts on surface and/or coastal waters in the project vicinity are anticipated during construction or operation of the proposed action. There are no streams or wetlands on or within close proximity to the project site. For proposed



ground disturbing activities, applicable erosion control measures and best management practices will be implemented in order to mitigate any possible adverse effects relating to soil erosion and runoff. Permanent sediment control measures will be used once construction is completed.

Coordination will be undertaken with the appropriate agencies during permitting and construction to ensure that the proposed action will not result in significant impacts regarding surface and coastal waters. A NPDES permit for storm water runoff from construction activities would be required should individual and/or cumulative soil disturbances at the project site exceed one acre of land area. Any discharges related to project construction or operation activities will comply with applicable State Water Quality Standards as specified in HAR, Chapter 11-54 Water Quality Standards and Chapter 11-55 Water Pollution Control. Excavation and grading activities will be regulated by applicable provisions of the County's grading ordinance.

In the short and long-term, no significant impacts on noise levels are anticipated as a result of the construction and operation of the Proposed Action. Impacts from construction noise are not anticipated to be significant as the project site and much of the surrounding land uses are considered airport or industrial related. Any potential impacts would be mitigated by complying with the State DOH Administrative Rules, Title 11, Chapter 46 "Community Noise Control" regulations. Once construction has been completed, noise impacts would be consistent with existing conditions.

11) Have substantial adverse effect on or be likely to suffer damage by being located in an environmentally sensitive area such as a flood plain, tsunami zone, sea level rise exposure area, beach, erosion-prone area, geologically hazardous land, estuary, fresh water, or coastal waters;

No short or long-term significant impacts are anticipated as the project site is not located within an environmentally sensitive area. The project site is located about 2 miles from the nearest coastline. The nearest stream within the Project Area is Nā'ālehu Stream, a non-perennial which originates from flows that begin at Ka'ū Forest Reserve. No impacts to these resources are anticipated.

The new facilities and other improvements would be designed in consideration of the potential hazards posed by hurricanes and other natural hazards to the extent practical. The project site is outside of the 500-year flood plain and outside of the tsunami evacuation zone, with only small portions within the collection system site are within the 500-year flood plain. The WWTP would be designed to accommodate anticipated increases in runoff. The final drainage system design would be coordinated with the County of Hawai'i Department of Public Works (DPW) prior to construction.

According to the FIRM, the project site is designated Zone X, an area of minimal flood hazard. A small portion of the collection system area is designated as Zone X – Other Flood Areas, indicating areas within the 0.2 percent annual chance (500-year) floodplain, or areas with a 1-percent annual chance of flooding with average flood depths less than



1-foot. Generally, runoff at the project site from rainfall currently sheet flows and percolates within the upper layers of the site. Some temporary ponding may occur at natural low points in the topography during and immediately after heavy rainstorm events. However, this water quickly infiltrates into the ground once heavy rains have subsided.

12) Have substantial adverse effect on scenic vistas and view planes, during day or night, identified in County or State plan or studies; or

The project site is not located near any Kau natural scenic sites. View planes of the project site from the surrounding area is generally obstructed by the surrounding vegetation that provides a visual barrier for the project site. Construction of the Proposed Action would not significantly alter the topography in such a way that would diminish the aesthetic value of the area.

13) Require substantial energy consumption or emit substantial greenhouse gas.

The Proposed Action involves the development of new wastewater treatment facilities that will be planned and designed to minimize use of electrical power. Thus, it will not create a substantial increase in energy consumption.

The construction and operation of the Proposed Action will not require a significant level of energy consumption. Implementation of the Proposed Action will result in the short term irrevocable release of GHGs from construction activities associated with the development of the proposed improvements. However, these activities will be temporary and the quantities of GHGs released will be negligible.

In the long-term, the Proposed Action is not expected to have a significant impact on traffic operation; therefore, no significant impacts on air quality due to an increase in greenhouse gases is anticipated. Due to the minimal impact of the Proposed Action, further mitigation of any potential long-term impacts is not anticipated to be required.

Based on the finding and the assessment of potential impacts, the Proposed Action does not require preparation of an Environmental Impact Assessment and an anticipated FONSI is determined.



(This page intentionally left blank)



8.1 Early Consultation

The Early Consultation process included efforts to inform the community and solicit input in scoping the EID for the Proposed Action. The Early Consultation Package for the Proposed Action was mailed out on March 29, 2024, to the following agencies, organizations, and stakeholders listed below in preparation of the EID/ EA. Parties that formally replied during the Early Consultation/Pre-Assessment process are indicated by a " \checkmark " below. All written comments are reproduced in Appendix F.

Federal Agencies

US Army Corps of Engineers, Honolulu District US Department of Agriculture Natural Resources Conservation Service US Environmental Protection Agency ✓ US Fish and Wildlife Service National Oceanic and Atmospheric Administration US Department of Transportation Federal Aviation Administration State Agencies Department of Agriculture \checkmark Department of Accounting and General Services Department of Business, Economic Development & Tourism (DBED&T) DBED&T Land Use Commission DBED&T Office of Planning and Sustainable Development DBED&T State Energy Office Department of Hawaiian Home Lands (DHHL) DHHL - East Hawai'i District Office Department of Health (DOH) DOH - Clean Water Branch DOH - Environmental Management Office DOH – Hazard Evaluation and Emergency Response DOH - Safe Drinking Water Branch DOH – Wastewater Branch Department of Land and Natural Resources (DLNR) **DLNR State Historic Preservation Division DLNR Land Division DLNR Engineering Division** \checkmark DLNR Division of Forestry and Wildlife Department of Transportation

Office of Hawaiian Affairs

University of Hawai'i

County of Hawai'i

Fire Department



✓ Police Department

Planning Department Research and Development Department of Public Works Department of Parks and Recreation Department of Water Supply Environmental Management Commission Office of the Corporation Council Office of Housing and Community Development

Elected Officials

State Senator Dru Mamo Kanuha, Senate District 3 Representative Jeanne Kapela, House District 5 Councilmember Michelle Galimba, Council District 6

Public Utilities

 ✓ Hawaiian Electric Company Hawaiian Telcom
 Spectrum Hawai'i
 Hawai'i Gas

Other Parties

Hawai'i State Library_ Nā'ālehu Public Library_ Hawaiian Civic Club of Ka'ū Ka'ū CDP Action Committee Ka'ū Calendar 'O Ka'ū Kākou

8.2 Community Outreach

The County has conducted numerous public information meetings during the course of designing and documenting the Proposed Action. A total of six community outreach sessions to discuss the Revised AOC were conducted in the Pahala and Naalehu communities between March 2022 and December 2023. In late October 2023, public preference surveys to gather input on the four feasible options were mailed out to 194 lot owners in the Nā'ālehu community. A public meeting was held November 16, 2023, at the Nā'ālehu Community Center to engage with the public to solicit feedback on the options explored in the PER, and to provide a general overview of the EID being prepared. In February 2024, a second round of surveys was mailed out to owners of affected lots in Nā'ālehu. The survey collection period remains open.

A semi-annual meeting community informational meeting regarding the closures of the Pāhala and Nā'ālehu LCCs was held on February 29, 2024 in Pāhala. The next semi-annual community informational meeting will be held in August 2024 in Nā'ālehu.



8.3 Additional Consultation

The publication of this Draft EID / EA shall be followed by 30-day public review period in which various agencies, organizations, and other stakeholders will be invited to provide comments on the Proposed Action for consideration in the Final EID / EA. Those consulted during the Early Consultation process and other identified stakeholders will be provided with a Notice of Availability once the Draft EID / EA is published in the ERP's *The Environmental Notice*.



(This page intentionally left blank)



9.1 Selection of a Preferred Alternative

The Amended AOC (Docket No. SDWA-UIC-AOC-2017-0002, proposed February 14, 2024) LCC requirements for Naalehu outlines that an EID must be prepared by the County for US EPA approval to meet Federal Environmental Review Requirements. The Revised AOC requires evaluation of four feasible alternatives, evaluation of a No Action alternative, and the tentative selection of a preferred alternative. Further, pursuant to § 11-200.1-18, HAR, an EA is required to present a discussion of the consideration of project alternatives that could reasonably attain the goals and objectives of the Proposed Action.

In the process of identifying a preferred alternative, extensive community engagement has taken place, including numerous meetings and discussions with the EPA. Preliminary Engineering Reports have also been completed to assess the identified alternatives as discussed in Section 2 of this EID.

The PER initially recommended an IWS alternatives for cost-effectiveness as the alternatives involving a package plant were found to have overall higher capital costs. The Department of Environmental Management - Wastewater Division (DEM - WWD) has tentatively selected Alternative 1 as the preferred alternative based on further consideration of factors such as regulatory compliance, community preference, and perceived environmental impacts, outlined as follows:

9.1.1 Recommendation Factors

Regulatory Compliance

In the development of the PERs, public objections or legal ramifications were not considered in the recommendation of the IWS alternative. This includes barriers to property access for construction, existing agreements, and compliance with current County and State codes.

Obtaining Right of Entry (ROE) to private properties for various purposes, such as infrastructure development or land surveys, can present several challenging issues. The process often involves negotiation, legal considerations, and respect for property rights. Failing to clarify these issues can lead to legal disputes and project delays. The most straightforward way to gain ROE is through the voluntary consent and permission of property owners. However, some property owners may be unwilling to grant access due to concerns about privacy, property damage, or other reasons.

In addition, there is an existing agreement between the County and C. Brewer established on April 12, 2007, in which the County agreed to construct and maintain new improved community sewer systems, including new County treatment and disposal systems and elimination of large capacity cesspools. Following this agreement, the County Council has already approved, and resolutions (Resolution 72-05 and 290-06) have been obtained for pursuance of WWTP and new collection system per agreement.



Further, implementation of the IWS alternatives would require significant modifications to the County code. Current HRS and DOH administrative rules may not support the required modifications. As such, modification of multiple HRSs and HARs would likely be required, including County Council approval and obtaining of new resolutions which could have severe ramifications for the current project timeline.

Community Preference

The County has held multiple community meetings with the intention of gathering input and survey responses regarding the community's preferences regarding the LCC closures. Survey responses have indicated a strong community preference for the package plant option. In addition to the recent survey responses, it has been noted that previous efforts led by the County have promised to design and construct the Naalehu WWTP and collection system.

Environmental Risks

Under HAR 11-62-31.1(a), an IWS may be used as a temporary on-site means of wastewater disposal in lieu of wastewater treatment. In addition to being considered a temporary means of wastewater disposal, it is noted that the package plant facility and new collection system would provide higher level of wastewater treatment. The package plant facility would provide wastewater treatment and management within the community while the County remains protective of public health and safety.

9.1.2 Action Items Under the Preferred Alternative

Under the Preferred Alternative, the County of Hawai'i would perform the following actions:

- 1. Construct a new gravity sewer on Kaalaiki Road and Nā'ālehu Spur Road to the WWTP located on a portion of Tax Map Key (TMK) (3) 9-5-007:016, , as the current proposed approach to bring sewer to the WWTP location
- 2. Implement drainage improvements within the vicinity of the WWTP within TMKs (3) 9-5-007:016 and (3) 9-5-021:015.
- 3. Construct a new pump station located on a portion of TMK (3) 9-5-008:048, and construct a new force main, which crosses an existing storm drainage channel at Melia Street, to connect to the Kaalaiki Road gravity sewer.
- 4. Construct a new gravity sewer on Opukea Street and Ohai Road to intercept existing flows entering the LCCs and divert sewage to the wastewater pump station (WWPS) and transport flows to the gravity sewer along Kaalaiki Road.
- 5. Install gravity sewers within the streets to replace the existing collection system.
- 6. Close and abandon the three LCCs.
- Accomodate future expansion of subsurface effluent disposal located within a portion of TMK (3) 9-5-022:001

The current action items involve seeking the Mayor's concurrence with the Wastewater Division Recommendation, approval from the DEM Director, and the establishment of a consultant selection committee. The chosen consultant will be tasked with completing the WWTP design and implementation plan within 6-7 months, followed by the wastewater collection system design within the same timeframe. The project aims to adhere to a 2-year construction schedule.



In summary, Alternative 1, which involves the construction of a package plant with a new collection system, has been selected based on regulatory compliance, community preference and an assessment of environmental risks.

9.2 Alternatives Design Schemes

9.2.1 Aerated Lagoons

Previous planning and design efforts for the Naalehu LCC Closure project had focused on an aerated lagoon secondary WWTP. The former design included the implementation of approximately four aerated lagoons which would utilize artificial aeration to promote the biological oxidation of wastewater. Ultimately, the County decided not to pursue this design scheme further as the community had not been receptive to the concept based on feedback gathered through community meeting and outreach.

9.2.2 Alternative Access Routes

Based on the selection of a preferred alternative, the design scheme currently involves the utilization of Naalehu Spur Road and a connection to a proposed access road to the WWTP site through TMK 9-5-007:016. As the design of the Proposed Action progresses, alternative access routes may need to be utilized. As such, the County has considered two alternative access routes which utilize TMKs 9-5-021:015 or 9-5-022:001. The proposed conceptual site plan of the WWTP is partially sited within all three of these parcels and the alternative access routes shall be discussed further with the respective landowners.

9.2.3 Alternative WWTP Site Configuration

As noted, the WWTP is anticipated to be sited within a 28.739 acre area within portions of TMKs: 9-5-007:016, 9-5-021:015 and 9-5-022:001. As the design phase progresses, modifications to the current conceptual site plan may occur. While the overall size of the structures is not anticipated to change significantly, adjustments to the site configuration could occur within the identified parcels.



(This page intentionally left blank)



CHAPTER 10: REFERENCES

Atlas of Hawaiian Watersheds & Their Aquatic Resources, Island of Hawaii, Bishop Museum & Division of Aquatic Resources. 1262 p. (3 volumes). Parham, J.E., G. R. Higashi, E. K. Lapp, D.G. K. Kumolo, R. T. Nishimoto, S. Hau, J. M. Fitzsimons, D. A. Polhemus, and W. S. Devick. 2008.

Brown and Caldwell. Naalehu Large Capacity Cesspool Closure Project Revised Preliminary Engineering Report. August 2023

- C. Brewer Company. Ka'ū Sewer System Evaluation Report. December 2004.
- County of Hawai'i, Planning Department. County of Hawai'i Draft General Plan 2045. June 2024.
- County of Hawai'i, Planning Department. County of Hawai'i General Plan. Ordinance No. 05-025. February 2005.
- County of Hawai'i, Planning Department. Ka'ū Community Development Plan, Draft for Public Review, March 2015.
- County of Hawai'i, Planning Department. 2017 Ka'ū Community Development Plan, Ordinance 2017-66. October 2017.
- County of Hawai'i, Department of Environmental Management. Final Environmental Assessment – Finding of No Significant Impact, the Nā'ālehu-Pāhala Large Capacity Cesspool Conversion. August 2007.
- Executive Order No. 11990, 42 FR 26961, 3 CFR, p. 121, 1977, as amended by Executive Order No. 12608, 52 FR 2923, 1987.
- Executive Order No. 11988, 42 FR 26951, 3 CFR, p. 117, 1977, as amended by Executive Order Nos. 12148, 44 FR 43239, 3 CFR, p. 412, 1979 and 13690, 81 FR 57401, 44 CFR 9, p. 57401, 2016.

Executive Order No. 12898, 59 FR 7629. 1994.

Hawai'i County Code 1983 (2016 Edition, as amended). Chapter 21. Sewers.

Hawai'i County Code 1983 (2016 Edition, as amended). Chapter 25. Zoning.

Hawai'i Revised Statutes, Chapter 205, Land Use Commission. 2017.

National Oceanic and Atmospheric Administration (NOAA). 2002. Department of Commerce, National Oceanic and Atmospheric Administration. 50 CFR 600. Magnuson-Stevens Act Provisions; Essential Fish Habitat (EFH). Federal Register Volume 67, Number 12

National Oceanic and Atmospheric Administration – National Marine Fisheries Service (NOAA-



NMFS). 2007. Magnuson-Stevens Fishery Conservation and Management Act, Public Law 94- 265, as amended by the Magnuson-Stevens Fishery Conservation and Management Reauthorization Act (P.L. 109-479), an act to provide for the conservation and management of the fisheries, and for other purposes. 16 U.S.C. § 1801-1884.

- State of Hawai'i. Department of Agriculture. U.S. Department of Agriculture, National Agriculture Statistical Service 2017 Census of Agriculture – County Data 2017.
- State of Hawai'i. Department of Health. Annual Summary 2022 Air Quality Data. September 2023.
- State of Hawai'i. Department of Health. Noise Reference Manual Big Island Edition. Revised July 2017.
- State of Hawai'i. Hawai'i Administrative Rules, Title 11 (Department of Health), Chapter 23 (Underground Injection Control). October 11, 2011.
- State of Hawai'i. Hawai'i Administrative Rules, Title 11 (Department of Health), Chapter 46 (Community Noise Control). September 1996.
- State of Hawai'i. Hawai'i Administrative Rules, Title 11 (Department of Health), Chapter 54 (Water Quality Standards). Revised November 15, 2014.
- State of Hawai'i. Hawai'i Administrative Rules, Title 11 (Department of Health), Chapter 62 (Wastewater Systems). March 21, 2016.

University of Hawai'i at Hilo, Department of Geography. 1998. Atlas of Hawaii. Third Edition.

- U.S. Census Bureau, 2022 American Community Survey 5 year Census Designated Place, Hawai'i County. U.S. Census Bureau's American Community Survey Office, September 13, 2023.
- U.S. Department of Agriculture Soil Conservation Service. Ka'ū River Basin Study, County of Hawai'i. February 1994.
- U.S. Environmental Protection Agency and County of Hawai'i, Department of Environmental Management. Final Environmental Assessment for the Pāhala Large Capacity Cesspool (LCC) Replacement Project, EPA Grant XP-96942401. February 2020.
- U.S. Environmental Protection Agency (EPA). Process Design Manual: Land Treatment of Municipal Wastewater Effluents. EPA/625/R-06/016. Office of Research and Development. Cincinnati, Ohio. September 2006.
- U.S. Environmental Protection Agency (EPA). Climate Resilience Evaluation and Awareness Tool (CREAT) Climate Scenarios Projection Map. Web. Accessed February 6, 2020.



U.S. Environmental Protection Agency. Region IX. Revised Federal Administrative Order on Consent EPA Docket No. SDWA-UIC-AOC-09-2017-0002, Pāhala and Nāʿālehu Large Capacity Cesspool Closure Projects Effective Date: August 22, 2022.

7 U.S.C. § 4201. 1981.

16 U.S.C. §§ 1271-1287. 1968.

- 16 U.S.C. §§ 1361 et seq. 1972.
- 16 U.S.C. §§ 1451-1464. 1972.
- 16 U.S.C. § 1531. 1973.
- 16 U.S.C. § 1801. 1976.
- 16 U.S.C. § 3501. 1982.
- 16 U.S.C. § 661. 1934.
- 16 U.S.C. § 668-668c. 1940. 16 U.S.C. § 703 et seq. 1918.
- 33 U.S.C. § 403. 1899.
- 33 U.S.C. § 1251 et seq. 1948.
- 42 U.S.C. § 300f. 1974.
- 42 U.S.C. § 7401 et seq. 1970.
- 54 U.S.C. § 300101. 1966.
- 54 U.S.C. § 312502. 1974.



(This page intentionally left blank)



APPENDIX A:

Nāʿālehu Large Capacity Cesspool Closure Project Revised Preliminary Engineering Report Brown and Caldwell

Naalehu Large Capacity Cesspool Closure Project Revised Preliminary Engineering Report

Prepared for County of Hawaii, Department of Environmental Management August 2023

2261 Aupuni Street, Suite 201 Wailuku, Maui, HI 96793

T: 808.244.7005

August 24, 2023



Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division 108 Railroad Avenue Hilo, HI 96720 Subject: Naalehu Wastewater Treatment Plant Revised Preliminary Engineering Report

153740.008

Dear Ms. lokepa-Moses:

Brown and Caldwell (BC), in association with Engineering Partners, Inc. (EPI) is pleased to present the attached Revised Preliminary Engineering Report (PER) for the Naalehu Large Capacity Cesspool (LCC) Closure Project. Preparation of a Revised PER is required by the Revised Administrative Order on Consent (AOC) that became effective on August 22, 2022. The need for a Revised PER was precipitated by several items:

- A wastewater treatment process with a smaller and shallower footprint is preferable due to the presence of lava rock at the proposed wastewater treatment plant (WWTP) site. Mechanical treatment technologies in the form of package plants offer the opportunity to achieve these goals.
 - The community, through community meetings and outreach, has not been receptive to the aerated lagoon technology that was formerly proposed.
- The Revised AOC allows the possibility of implementing individual wastewater systems
 (IWS) to close the LCCs.

The Revised AOC requires evaluation of four feasible options:

- A package plant and new collection system.
- ii. A package plant connected to the existing collection system.
 - iii. A maintenance contract model IWS program.
 - iv. An operating permit model IWS program.

This Revised PER consists of three parts:

- This introductory summary that provides comparisons of the four feasible options.
 Part A, by BC, which presents updated analysis of feasible options i and ii that are based on using a package plant based WWTP to service the Naalehu community and does the LCCs. BC is a nation-wide environmental engineering firm with local Hawaii offices located in Kamuela, Wailku, and Honolulu. BC has been planning and designing WWTPs throughout the United States for over 75 years.
- Part B, by EPI, which presents a detailed analysis of feasible options iii and iv that are
 based on using IWS to service the Naalehu community and close the existing LCCs.
 EPI is a multi-discipline engineering and design firm based in Hilo. EPI has
 successfully designed IWS systems on Hawaii Island and is well-versed to address
 implementing IWS in the unique local soil and subsurface geological conditions in
 Nalehu.

Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division August 24, 2023 Page 2 Throughout this Revised PER the following terms are used:

"Feasible options" refers to the four specific options (i, ii, iii, iv) listed above and in paragraph V.A.31.a of the Revised AOC.

"Alternatives" and "project alternatives" refer to various combinations of systems or technologies that are evaluated within this Revised PER to determine preferences for the feasible options.

1. Comparison of Feasible Options

The four feasible options are compared below,

1.1 Protection of Human Health and the Environment

Table 1 compares the four feasible options with respect to protection of human health and the environment. The State of Hawaii Department of Health (DOH) regulates both WWTPs and IWS. All four feasible options are protective of human health and the environment when implemented in accordance with the applicable Hawaii Administrative Rules (HAR). Additional discussion is provided in Parts A and B.

Tab	e 1. Protection of Human	Health and the Environment	
Feasible Option	Regulatory Authority	Variances	Protective of Human Health and the Environment?
 Package plant and new collection system 	HAR 11-62, Subchapter 2	Variance required by DOH for WWTP flow capacity	Yes
ii. Package plant connected to the existing collection system	HAR 11-62 Subchapter 2	Variance required by DOH for WWTP flow capacity	Yes
iii. A maintenance contract model IWS program	HAR 11-62, Subchapter 3	Variances may be required for some lots for setback distances, etc.	Yes
iv. An operating permit model IWS program	HAR 11-62, Subchapter 3	Variances may be required for some lots for setback distances, etc.	Yes

1.2 Capital Cost Comparison of Feasible Options

Table 2 summarizes the capital costs for the four feasible options. Details of the capital cost estimates are provided in Parts A and B. Note that the IWS capital costs per lot are presented as ranges; greater precision will not be available until designs are complete due to the site-specific nature of IWS implementation on existing developed properties.

Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division August 24, 2023 Page 3

Table 2. Capital Cost	: Comparison	
Feasible Option	Capital Cost (\$ million)	Cost per Lot
i. Package plant and new collection system	\$84.3	\$413,000 ª
ii. Package plant connected to the existing collection system	\$74.2	\$364,000 ^a
iii. A maintenance contract model IWS program	\$5.8 - \$29.1	\$30,000 - \$150,000 ^b
iv. An operating permit model IWS program	\$5.8 - \$29.1	\$30,000 - \$150,000 b

a. Based on a total of 204 lots in the WWTP service area.

b. Based on a total of 194 lots converted per the AOC.

As shown in the table, the IWS feasible options incur significantly lower capital costs than the package plant alternatives.

1.3 Long-Term Recurring Costs Comparison

Long-term recurring costs include operations and maintenance (0&M) costs, examples include labor, electricity, chemicals, and maintenance materials like spare parts. Another recurring cost is the need to replace and refurbish (R&R) equipment or systems when they reach the end of their useful life.

A summary of the long-term recurring costs for the four feasible options is presented in Table 3 for comparison. Additional detail is included in Parts A and B.

lable 3. Long-lerm Kecurring	ig costs comparison	
Feasible Option	Annual 0&M Cost ^a	R&R Cost (after 20 years) ^a
i. Package plant and new collection system	\$886,000	\$6,000,000
ii. Package plant connected to the existing collection system	\$1,050,000	\$6,000,000
iii. A maintenance contract model IWS program	\$250,000	\$5,800,000
iv. An operating permit model IWS program	\$340,000	\$5,800,000

a. Expressed in current (2023) dollars.

1.4 Life-Cycle Cost Comparison

An economic evaluation was prepared to assess the potential life-cycle costs associated with each project alternative. The economic evaluation consists of a net present value comporation. The net present value analysis includes capital. O&M, and R&R costs. An appropriate initiationary factor and discount rate are applied to obtain the net present value over a 30-year planning period. The net present value of an alternative represents the amount of money that would need to be set asile today (at a given interest rate) to pay the costs associated with the alternative over the entire planning period. The amount of money that would need to be set asile today (at a given interest rate) to pay the costs associated with the alternative over the entire planning period. The economic persective Table 4 summarizes the life-cycle cost evaluation assumptions.

Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division August 24, 2023 Page 4

Table 4. Life-Cycle Econe	omic Assumptions
Description	Value
Year of analysis	2023
Planning period	30 years
Inflation rate	3.5 percent
Discount rate	5.0 percent
R&R cycle	20 years

Table 5 presents the life-cycle cost evaluation results.

I able 5. Life	P-UDE COST EV	aluation Kesuits	
Alternative	Capital Cost (\$M)	Net Present Value of O&M and R&R Costs (\$M)	Life-Cycle Cost (\$M)
i. Package plant and new collection system	\$84.3	\$ 25.4	\$ 109.8
 Package plant connected to the existing collection system 	\$74.2	\$ 29.3	\$ 103.5
iii. A maintenance contract model IWS program	\$ 29.1	\$ 10.2 a	\$ 39.3
iv. An operating permit model IWS program	\$ 29.1	\$ 12.3ª	\$ 41.4
Includes replacement costs and IMS 08-M costs	naid directly by I	standomotic	

Figure 1 shows the results graphically. The IVNS alternatives have significantly lower lifecycle costs than the package plant alternatives.



Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division August 24, 2023 Page 5

Figure 2 shows the cumulative cash flow projections for the four feasible alternatives over the planning period, expressed in current (2023) dollars.



Figure 2. Cumulative Cash Flow Projections

1.5 Schedule

and B include preliminary assessments of implementation schedules. Table 6 provides a procurement and fabrication can occur in parallel with design. However, compliance with The Revised AOC requires the LCCs be closed no later than December 31, 2027. Parts A summary of the preliminary implementation schedule assessments. As discussed in Part A, it will be difficult to implement the WWTP approach to close the LCCs by the deadline, the Revised AOC deadline will be a significant challenge with Feasible Options i and ii due to entitlement processes, environmental review, land acquisition, and materials supply challenges currently facing the Hawaii construction industry. A design/build approach could potentially reduce the implementation timeframe if equipment without a schedule extension.

property access and any alterations to County regulations deemed necessary by the deadline. The IWS approach assumes that the County can address Hawaii Revised Per Part B, the IWS approach may be able to be implemented by the Revised AOC Statues (HRS) 343 environmental review requirement via an exemption, and that County are achievable within the timeframe.

Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division August 24, 2023 Page 6

Table 6. Summary of P	reliminary Implen	nentation Schedule Feasible O	Assessments ptions	
Description	i. Package Plant New Collection System	ii. Package Plant Existing Collection System	iii. Maintenance Contract Model IWS Program	iv. Operating Permit Model IWS Program
Entitlements and permitting	Q4 2024	Q4 2024	Q2 2024	Q2 2024
Design and construction	Q3 2028	Q3 2028	Q2 2026	Q2 2026
Estimated LCC closure	Q3 2028	Q3 2028	Q3 2026	Q3 2026
Revised AOC LCC closure milestone		December 3.	1, 2027	
Risk of missing Revised AOC LCC closure milestone	High	High	Moderate	Moderate
Note: Q = quarter				

2. Revised AOC References

The Revised AOC paragraph V.30.A.a lists information that must be included in this Revised PER. Table 7 provides references to the information within.

Tabl	e 7. Revised AOC F	'aragraph V.30.A.a C	thecklist	
	Re	port Reference Section	on for Feasible Optio	us
Revised AOC Paragraph V. 30. A.a Description	i. Package Plant New Collection System	ii. Package Plant Existing Collection System	iii. Maintenance Contract Model IWS Program	iv. Operating Permit Model IWS Program
Description of project details for each feasible option	Part A, §2.3, §2.4, and §8	Part A, §2.3 and §8	Part B, §10	Part B, §10
Planning area description	Part A, Fig. 2-1	Part A, Fig. 2-1	Part A, Fig. 2-1	Part A, Fig. 2-1
Planning period	Part A, §7.2.3	Part A, §7.2.3	Part A, §7.2.3	Part A, §7.2.3
Description of planning phases	Part A, §7.2.3	Part A, §7.2.3	Part A, §7.2.3	Part A, §7.2.3
Owner and operator of facilities	Part A, §1-1	Part A, §1-1	County/In-house or 3rd-party service provider	Homeowner/3rd- party service provider
Location of facilities (including a map)	Part A, Fig. 2-1, Fig. 8-1	Part A, Fig. 2-1, Fig. 8-1	Part A, Fig. 2-1	Part A, Fig. 2-1
Design parameters for each feasible option	Part A, §2.3, §2.4, and §8	Part A, §2.3, and §8	Part B, Table 1.10	Part B, Table 1.10
Major unit processes:	Part A, §5	Part A, §5	Part B, §10	Part B, §10
Flow diagram	Part A, Fig. 8-2	Part A, Fig. 8-2	Part B, Fig. 1.1	Part B, Fig. 1.1
Pipe lengths, sizes, and locations	Part A, Appendix D	Part A, Appendix D	Part B, Appendix F	Part B, Appendix F
Design criteria	Part A, §8.3	Part A, §8.3	Part B, §6	Part B, §6
Project costs	Part A, §7	Part A, §7	Part B, §11, Annendix B	Part B, §11, Annendix B

Ms. Brenda lokepa-Moses County of Hawaii Wastewater Division August 24, 2023 Page 7

3. Conclusions

From a technical perspective, all four feasible options represent viable ways to close the LCCs in Naalehu and will be protective of human health and the environment. The IWS feasible options (iii and iv) present significantly lower capital. Jong-term recurring, and life-cycle costs than the WWT feasible options (ii and iv) offer greater potential to meet the LCC closure deadline contained options (iii and iv) offer greater potential to meet the LCC closure deadline contained with the Rewsed AOC. We anticipate that financing options, community engagement, burden on the Naalehu community, and alignment with County long term planning goals will need to be considered outside of this preliminary engineering report.

Brown and Caldwell appreciates that the County has requested our services in assisting with this project. Should you have any questions, please do not hesitate to call Craig Lekven at 808.442.3301.

Very truly yours,

Brown and Caldwell

aric. H

Craig C. Lekven, P.E., Project Manager Wailuku, Hawaii

PART A: WWTP Approach
Part A

Naalehu Wastewater Treatment Plant Revised Preliminary Engineering Report

Prepared for County of Hawaii, Department of Environmental Management August 2023 THIS WORK (PART A) WAS PREPARED BY ME OR UNDER MY SUPERVISION

and the

Signature

April 30, 2024 Expiration Date of the License



Brown Me Caldwell

Table of Contents-Part A

List	of Fig	ures	vi
List	of Tat	oles	Vii
List	of Abt	oreviation	s
ų	Introc	duction	1-1
	1.1	Backgro	1-1
	1.2	Existing	System
	1.3	Report	Contents
сi	Collec	ction Sys	tem
	2.1	Service	Area
	2.2	Existing	Collection System
	2.3	Phase 1	. Collection System Project2-3
	2.4	Phase 2	Collection System Project
	2.5	Collecti	on System Alternatives
	2.6	Collecti	n System Costs
	2.7	Recom	nendation
ю.	Flow	and Load	I Projections. 3-1
	3.1	Flow Pro	jections Based on City and County of Honolulu Standards
	3.2	Reduce	d Flows Based on Potable Water Records
		3.2.1	Dry Weather I/I Allowance
		3.2.2	Wet Weather I/I Allowance
		3.2.3	Reduced Flow Projections
		3.2.4	Flow Variance
	3.3	Influent	Characteristics
	3.4	Influent	Mass Loads
4.	Efflue	ent Mane	gement Alternatives and Regulatory Requirements4-1
	4.1	Effluent	Management Alternatives
		4.1.1	Ocean Discharge
		4.1.2	Subsurface Disposal via Injection Wells
		4.1.3	Water Recycling
		4.1.4	Slow Rate Land Treatment
		4.1.5	Subsurface Drip Irrigation Disposal4-3
		4.1.6	Leach Field
		4.1.7	Ramifications of United States Supreme Court Opinion on County of Maui,
		0	nawali V. nawali Wildlife Fund et. al
		4.1.0	Kecommendation

Brown AND Caldwell ≔

PER	
Revised	
WWTP	
Naalehu	

	4.2	Treatme	rt Requirements	4-8
2 2	Wast	ewater Tr	eatment Evaluations	5-1
	5.1	Prelimir	ary Treatment.	.5-1
		5.1.1	nfluent Flow Measurement	5-1
		5.1.2	Influent Flow Sampling.	.5-1
		5.1.3	Screening	5-1
		5.1.4	Grit Removal	5-2
		5.1.5	Odor Control	-2-2
		5.1.6	Recommendation	-5-8
	5.2	Second	ry Treatment	5-9
		5.2.1	Membrane Bioreactor	5-9
		5.2.2	Sequencing Batch Reactor	5-10
		5.2.3	Nereda (Granular Activated Sludge) Process 5	5-11
		5.2.4	Oxidation Ditch	5-11
		5.2.5	Extended Aeration Activated Sludge Package Plant	5-12
		5.2.6	Activated Sludge with Anoxic Selector E	5-13
		5.2.7	Recirculating Gravel Filter	5-13
		5.2.8	Secondary Treatment Technology Screening	5-14
	5.3	Disinfec	ion	5-16
		5.3.1	Sodium Hypochlorite	5-16
		5.3.2	Calcium Hypochlorite	5-18
		5.3.3	Ultraviolet Light Disinfection	5-19
		5.3.4	Cost Evaluation E	5-20
		5.3.5	Disinfection Recommendation	5-20
0.	Solid	s Manag(ment	.6-1
	6.1	Aerobic	Jigestion with Decant Thickening	.6-1
	6.2	Anaerot	c Digestion with Biogas Use	6-1
	6.3	Dewate	ng	6-2
		6.3.1	Centrifuge Dewatering	.6-2
		6.3.2	Screw Press Dewatering	.6-3
	6.4	Disposa		.6-3
7.	Proje	ct Alterné	tives Evaluations	7-1
	7.1	Project.	Iternative Descriptions	7-1
		7.1.1	Project Alternative 1: Activated Sludge with Anoxic Zone Package Plants	7-1
		7.1.2	Project Alternative 2: MBR Package Plants	7 3
		7.1.3	Project Alternative 3: Imhoff Tank/Recirculating Gravel Filter	7-4
	7.2	Cost Ev:	luations	7-5
		7.2.1	Capital Costs	7-5
		7.2.2	Operation and Maintenance Costs	.7-6
		7.2.3	Life-Oycle Costs	7-6

æ	
₽	
≷	
2	
ĥ	
ale	
Aa	
_	

Naa	lehu WV	WWTP Revised PER Table of	Contents
	7.3	3 Non-Economic Evaluation	7-8
		7.3.1 Approach	7-8
	7.4	4 Non-Economic Evaluation Criteria	6-2
	7.5	5 Non-Economic Evaluation Results	7-10
	7.6	3 Conclusions and Recommendation	7-11
ø	Prelir	eliminary Design of Improvements	8-1
	8.1	1 Site Plan	8-1
	8.2	2 Process Schematic	8-1
	8.3	3 Preliminary Design Criteria	8-4
	8.4	4 Preliminary Floor Plan	8-6
<u>о</u>	Imple	plementation Plan.	9-1
	9.1	1 Implementation Approach	9-1
		9.1.1 Design Bid Build Approach	9-1
		9.1.2 Design Build Approach	9-1
	9.2	2 Implementation Schedules	9-2
		9.2.1 Recent Change in State of Hawaii Land Use Commission Policy	9-2
		9.2.2 Equipment Procurement Time to Impact Construction Schedule	9-2
		9.2.3 Implementation Schedules	9-2
	9.3	3 Recommendation	9-2
10.	. Refe	ferences	10-1
App	bendix	lix A. Cost Estimates	A
App	bendix	lix B: Letter to Support Department of Hawaii Flow Variance Application	В
App	bendix	lix C: Non-Economic Evaluation	C
Apr	pendix	lix D: Naalehu Wastewater Collection System Improvements Technical Memorandum	D





Table of Contents

PER	
Revised	
WWTP F	
Naalehu	

List of Figures
Figure 1-1. Existing Naalehu Wastewater System1-3
Figure 2-1. Naalehu WWTP Service Area2-2
Figure 2-2. Naalehu Phase 1 Collection System Layout
Figure 2-3. Naalehu Phase 2 Collection System Layout-Former Brewer Company House Lots2-5
Figure 2-4. Life-Cycle Cost Comparison of Collection System Alternatives
Figure 4-1. Irrigation Demand Assessment
Figure 4-2. Subsurface Drip Irrigation Concept4-3
Figure 5-1. In-Channel Cylindrical Screen5-2
Figure 5-2. Sloped Bottom Vortex Grit Removal Cross Section
Figure 5-3. Flat Bottom PISTA® Grit Removal5-3
Figure 5-4. Aerated Grit Removal Schematic
Figure 5-5. HeadCell Process Schematic
Figure 5-6. Biotrickling Filter
Figure 5-7. Activated Carbon Scrubber5-8
Figure 5-8. Membrane Bioreactor Illustration
Figure 5-9. Membrane Cassettes at Johns Creek Environmental Campus
Figure 5-10. Nereda Process
Figure 5-11. Typical Oxidation Ditch Schematic 5-12
Figure 5-12. Extended Aeration Process Schematic
Figure 5-13. Activated Sludge with Anoxic Selector Process Schematic
Figure 5-14. Recirculating Gravel Filter for Treatment of Septic Tank Effluent5-14
Figure 5-15. Schematic Diagram of an On-site NaOCI Generator
Figure 5-16. Typical Calcium Hypochlorite Feed System
Figure 6-1. Centrifuge Dewatering6-2
Figure 6-2. Screw Press Diagram6-3
Figure 7-1. Project Alternative 1: Activated Sludge with Anoxic Zone Package Plants7-2
Figure 7-2. Project Alternative 2: MBR Package Plants7-3
Figure 7-3. Project Alternative 3: Imhoff Tank/Recirculating Gravel Filter
Figure 7-4. Life-Cycle Cost Evaluation Results7-8
Figure 7-5. Combined Economic and Non-Economic Results
Figure 8-1. Overall Site Plan
Figure 8-2. Process Schematic
Figure 8-3. Operations Building Preliminary Floor Plan
Figure 9-1. Implementation Schedules

Ш	
sed P	
Revis	
WTP	
≥	
lehu	
Naa	

Table of Contents

List of Tables	
Table 2-1. Naalehu WWTP Service Area Summary2	2
Table 2-2. Collection System Alternatives Cost Summary	မှ
Table 2-3. Collection System Alternatives Capital Cost Estimating Assumptions	9
Table 2-4. Life-Oycle Economic Assumptions2-	2-1
Table 3-1. Naalehu WWTP Flows Based on 2017 CCH Standards	4
Table 3-2. Naalehu WWTP Calculated Flow Capacity	N
Table 3-3. Recommended WWTP Capacity	က္
Table 3-4. Summary of Assumed Influent Characteristics	က္
Table 3-5. Projected Peak Dry Weather Day Influent Mass Loads	4-
Table 4-1. Nutrient Water Quality Standards for Class AA Embayments	Ţ
Table 4-2. Relative Risk of Being Functional Equivalent to Direct Discharge	2-1
Table 4-3. Recommended Subsurface Drip Disposal Design Criteria	ထု
Table 4-4. Applicable HAR 11-62 Land Disposal Requirements	ထု
Table 5-1. Induced Vortex-Advantages and Disadvantages5	4
Table 5-2. Aerated Grit Removal-Advantages and Disadvantages5	ŝ
Table 5-3. Lamella Plate Settling/HeadCell-Advantages and Disadvantages5	ဖု
Table 5-4. Grit Capture Size Comparison5	ဖု
Table 5-5. Screening of Secondary Treatment Options5-1	15
Table 5-6. Bulk Sodium Hypochlorite-Advantages and Disadvantages	16
Table 5-7. Onsite Sodium Hypochlorite Generation-Advantages and Disadvantages5-1	17
Table 5-8. Calcium Hypochlorite Summary5-1	18
Table 5-9. Chlorine Demand	19
Table 5-10. UV Disinfection Design Summary5-2	20
Table 5-11. Estimated Disinfection Costs5-2	20
Table 7-1. Capital Cost Estimating Assumptions7	μ
Table 7-2. Capital Cost Estimates Summary7	ĥ
Table 7-3. 0&M Cost Assumptions7	φ
Table 7-4. 0&M Cost Estimate Summary7	φ
Table 7-5. Life-Cycle Economic Assumptions7	2-
Table 7-6. Life-Cycle Cost Analysis Summary7	2-1
Table 7-7. Non-Economic Comparison Criteria7	ရ
Table 7-8. Non-Economic Comparison Criteria Weighting Factors	10
Table 7-9. Non-Economic Weighted Scores	10
Table 8-1. Preliminary Design Criteria	4





Table of Contents

Natural Resources Conservation Service operation and maintenance

List of Abbreviations

AACE	Advancement of Cost	NRCS
	Engineering International	0&M
AOC	Administrative Order on Consent	PER
BC	Brown and Caldwell	nsi
BOD ₅	5-day biochemical oxygen demand	PWWF
ССН	City and County of Honolulu	RGF
CDP	Kau Community Development Plan	RNA
CWA	Clean Water Act	SBRs
DB	design/build	SFS
DBA	District Boundary Amendment	SR
DBB	design/bid/build	SRT
DNA	deoxyribonucleic acid	TMK
НОП	Hawaii Department of Health	Z
ELLF	end-of-lamp-life	TSS
FAI	Fukunaga & Associates, Inc.	nc
ft²	square feet	USEPA
ft3	cubic feet	USSC
GAC	granular activated carbon	N
gpcd	gallons per capita per day	MOS
gpd	gallons per day	WWPS
gpm	gallons per minute	WWRF
H_2S	hydrogen sulfide	WWTP
HAC	Hawaiian Agriculture Company	
HAR	Hawaii Administrative Rules	
И	Infiltration and inflow	
_	Liter	
lbs	bounds	
LCCs	large capacity cesspools	
ГРНО	low pressure high output	
LUC	Land Use Commission	
MBR	membrane bioreactor	
Mga	million gallons	
mgd	million gallons per day	
mL	milliliter	
MLSS	mixed liquor suspended solids	
NaOCI	sodium hypochlorite	
z	nitrogen	
NPDES	National Pollutant Discharge Elimination System	
NPV	net present value	

 pounds per square inch PWWF Peak wetweather flow RGF recirculating gravel filter RNA ribonucleic acid SBRs sequencing batch reactors SES Sequencing batch reactors Second equivalent size Unc Underground Injection Control USSC United States Environmental Protection Åguss UV ultraviolet WVS water quality standards WVPS water quality standards
PWWF peak wet weather flow RGF recirculating gravel filter RNA ribonucleic acid SBRs sequencing batch reactors SES sand equivalent size SR slow rate SR solids retention time TMK tax map key TN total nitrogen TMK tax map key TN total suspended solids UIC Underground hjection Control UEEPA United States Environmental Protection Å USC United State Supreme Court UV ultraviolet WVPS wastewater pump stations
RGF recirculating gravel filter RNA ribonucleic acid SBRs sequencing batch reactors SES sand equivalent size SR slow rate SRT solids retention time TMK tax map key TN total nitrogen TN total nitrogen TN total suspended solids UIC Underground Injection Control USC Under States Environmental Protection A USC United States Supreme Court UV ultraviolet UV satewater quality standards WWPS wastewater punp station
RNA ribonucleic acid SBRs sequencing batch reactors SES sand equivalent size SR sonand equivalent size SR slow rate SR slow rate SR solids retention time TMK tax map key TN total nitrogen TS total suspended solids UIC Underground Injection Control USEPA United States Environmental Protection A US uttraviolet UV uttraviolet UV uttraviolet UV uttraviolet UV water quality standards WVPS water quality standards WVPS water quality standards
 SBRs sequencing batch reactors SES sand equivalent size SR slow rate SRT solids retention time TMK tax map key TMK tax map key TN total nitrogen TSS total suspended solids UIC Underground Injection Control USC United States Environmental Protection Agues UV ultraviolet UV ultraviolet WVPS wastewater pump station
SES sand equivalent size SR slow rate SRT solids retention time TMK tax map key TN total nitrogen TSS total suspended solids UIC Underground Injection Control USC Underground Injection Control USC United States Environmental Protection Ag USSC United States Environmental Protection Ag USSC United States Supreme Court UV ultraviolet WVPS wastewater programmers
 SR slow rate SRT solids retention time TMK tax map key TMK total nitrogen TSS total suspended solids UIC Underground Injection Control USC United States Environmental Protection Agues USC United States Supreme Court UV ultraviolet WVS wastewater pump station
 SRT solids retention time TMK tax map key TN total nitrogen TSS total suspended solids UIC Underground Injection Control USC United States Environmental Protection A USC United States Supreme Court UV ultraviolet WVS water quality standards WWPS wastewater pump station
TMK tax map key TN total nitrogen TSS total suspended solids UIC Underground Injection Control USEPA United States Environmental Protection A USSC United States Environmental Protection A UV ultraviolet WVS water quality standards WVPS wastewater punp station
TN total nitrogen TSS total suspended solids UIC Underground Injection Control USSC United States Environmental Protection A USSC United State Supreme Court UV ultraviolet WQS water quality standards WWPS wastewater punp station
TSS total suspended solids UIC Underground Injection Control USEPA United States Environmental Protection Å USC United State Supreme Court UV ultraviolet WQS water quality standards MWPS wastewater punp station
UIC Underground Injection Control USEPA United States Environmental Protection A USSC United State Supreme Court UV ultraviolet WQS water quality standards MWPS wastewater punp station
USEPA United States Environmental Protection A USSC United State Supreme Court UV ultraviolet WQS water quality standards WWPS wastewater pump station
USSC United State Supreme Court UV ultraviolet WQS water quality standards WWPS wastewater pump station
UV ultraviolet WQS water quality standards WWPS wastewater pump station
WQS water quality standards WWPS wastewater pump station
WWPS wastewater pump station
WWKF Wastewater Reclamation Facility
WWTP Wastewater Treatment Plant
WWTP Wastewater Treatment Plant

Section 1

Introduction

This section summarizes the project background and describes key elements of the existing Naalehu wastewater collection system.

1.1 Background

The town of Naalehu is located in the Kau district of the Island of Hawaii. According to the 2020 United States Census, the town population is approximately 1,007 persons. The Naalehu community was established as the result of the sugar operations of the C. Brewer Company. A portion of the community is serviced by a sewer system that was privately built, owned, and operated by the C. Brewer Company, which merged with Hawaiian Agriculture Company (HAC) in 1972. The wastewater collected by the sewer system discharges into large capacity "gang" cesspools (LCCs). Many years after its establishment, the private sewer system ownership was conveyed to the County of Hawaii Department of Environmental Management.

In 1998, the U.S. Environmental Protection Agency, promulgated regulations, 40 Code of Federal Regulations 144.14, that require the elimination of LCCs. Options to close the LCCs include construction of a new sewer collection system located within public right-of-way and replacement of the existing LCS with a wastewater treatment plant (WMTP) to address the wastewater treatment and disposal needs of the Naalehu community. These centralized WMTP options are the subject of this Preliminary Engineering Report (PER). A separater report is being concurrently prepared that evaluates additional options using individual wastewater systems in lieu of a new collection system and WWTP to close the LCCs.

This report revises the 2018 PER for the Naalehu WWTP and summarizes the proposed facilities needed to treat and dispose of wastewater flow currently discharged to the LCCs, plus additional sewer connections. The PER presents the existing and estimated future flows and loads to the treatment plant, describes the proposed treatment processes, recommends needed upgrades for the WWTP to meet the future treatment levels, and provides an initial opinion of the cost to construct, operate, and maintain the improvements project.

1.2 Existing System

The existing collection system is a network of gravity sewers that discharge to three existing LCGs. Figure 1-1 shows the existing Naalehu wastewater collection system and service areas for the LCGs. The LCGs in Naalehu are numbered 3, 4, and 5; LCGs 1 and 2 are located in the nearly town of Pahala. A detailed analysis of the existing wastewater collection system was completed by others (M&E Pacific, December 2004). The report concluded that the Naalehu community existing sewer system consists of approximately 5, L28 linear feet of 6-inch-diameter and 15,500 linear feet of 4-inch-diameter pipelines. Residential laterals connect to 4-inch sewers that discharge into 6-inch sewer mains, predominately found in easements on private property, which transmit wastewater to the three LCGs. There are approximately eight manholes in the sewer system. More recently available information notes the size of pingt to be between 3 and 8 inches with a few additional sewer manholes (Erkunaga and Associates, Inc., June 2013). There are no pump stations, and the system is not designed to collect stormwater.



Brown And Caldwell



Section 1

1.3 Report Contents

- The remainder of the revised PER is organized as follows: • Section 2 describes the service area and collection system options.
- Section 3 describes flow and load projections for the new Naalehu WWTP.
- Section 4 evaluates potential options for effluent management and details the treatment requirements for the preferred option.
- Section 5 describes the evaluations conducted in support of the preliminary design of the Naalehu WWTP.
- Section 6 summarizes the solids management approach.
- Section 7 describes alternative treatment options under consideration.
- Section 8 provides a preliminary design for the proposed improvements.
- Section 9 presents the proposed implementation plan for the new WWTP.





Section 2

Collection System

This section summarizes the existing and new collection system options for the Naalehu service area collection system layouts included within this section are reproduced from the 2020 FAI report, which was prepared in accordance with the original 2017 AOC. Should a new WWTP be the preferred option for implementation, an update to the collection system layout is anticipated to finalize the service area in accordance with the 2022 Revised AOC. FAI'S Naalehu report is provided (FAI) in May 2020 in efforts to close the existing large-capacity cesspools (LCCs). The figures and (LCC Closure Project compliance area), which was investigated by Fukunaga and Associates, Inc. in Appendix D.

2.1 Service Area

A concrete flood canal divides Naalehu into East (Hilo) and West (Kona) sides. The majority of the Hilo side properties are residential, while majority of Kona side properties are commercial. Within the town of Naalehu, there is an existing wastewater collection system that services approximately 164 former C. Brewer Company (Brewer Company) house lots on the mauka (mountain) side of Mamalahoa Highway. The collection system is currently located within easements in private properties and is discharged to three LCCs.

expanded to service the entire community; however, the collection system and wastewater treatment properties requesting connection Therefore, the proposed service area constitutes the LCC Closure plant (WWTP) presented in this report will service the former Brewer Company properties currently connected to the LCCs and properties adjacent to the new collection system, including three The Kau Community Development Plan (CDP) indicates the sewer system may eventually be Project compliance area.

Figure 2-1 shows the service area for the new WWTP, including the newly accessible properties.



Figure 2-1. Naalehu WWTP Service Area

Table 2-1 provides a summary of the WWTP service area (LCC Closure Project compliance area) property types.

Descett Time	Nur	ther of Parcels	
Froperty type	Existing C Brewer	Newly Accessible	Total
Residential	159	25	184
Commercia	2	7	6
Church	1	1	2
Industrial		2	2
Agricultural	1	2	е
Residential/Commercial	1	1	2
Residential/Agricultural		1	1
Park		1	1
Total	164	40	204
Note: Condoo aroo - 1 OC	Clocing Brologt comp	Concourt	





2-1

Section 2

2.2 Existing Collection System

In 2004, Brewer Company contracted M&E Pacific to perform a sewer system evaluation for the town of Naalehu. The results of this investigation determined that the existing sewer lines and manholes constructed in the streets, but instead runs through easements located on private properties, with sewer system is over 80 years old, long surpassing its expected lifespan, and will require extensive performed during the 2004 sewer system evaluation identified at least 13 locations of line breaks and/or pipe defects and 12 household units with defective sewer vents. In addition, the existing many collection lines running adjacent to or beneath the houses. The results of a smoke test do not conform to the County sewer design standards. The existing sewer system was not repair and rehabilitation if chosen to be reused.

consists of constructing a new sewer system in the streets to meet the County sewer standards and to allow the collection system to be owned and operated by the County (M&E Pacific, December 2004). The recommended alternative, which received overwhelming support from Naalehu voters in 2004,

collection system into the future an updated condition assessment study is recommended to better identify system deficiencies. Substantial improvements will likely be necessary due to the age of the Nearly 20 years have passed since the 2004 study was completed. In order to reuse the existing system. Reusing the existing collection system will require constructing the Phase 1 collection system scenario described below to connect to the proposed WWTP and close the LCCs.

2.3 Phase 1 Collection System Project

In efforts to close the existing LCCs, FAI conducted a collection system investigation in May 2020. wastewater pump station, and force main to transport sewage from the LCCs to the new WWTP. collection system within the Brewer Company house lots and constructing new gravity sewers, The investigation recommends a two-phase approach. Phase 1 involves utilizing the existing Phase 1 consists of the following:

- 1. Construct a new gravity sewer on Kaalaiki Road and Naalehu Spur Road to the WWTP located on a portion of Tax Map Key (TMK) (3) 9-5-007:016.
- Construct a new pump station located on a portion of TMK (3) 9-5-008:048, and construct a new force main, which crosses an existing storm drainage channel at Melia Street, to connect to the Kaalaiki Road gravity sewer. N
- entering the LCCs and divert sewage to the wastewater pump station (WWPS) and transport Construct a new gravity sewer on Opukea Street and Ohai Road to intercept existing flows flows to the gravity sewer along Kaalaiki Road. с.

Figure 2-2 illustrates the transmission system layout established by FAI in May 2020 for the Phase 1 collection system.

Naalehu WWTP Revised PER



Figure 2-2. Naalehu Phase 1 Collection System Layout





24

Section 2

2.4 Phase 2 Collection System Project

The Phase 2 collection system project consists of the installation of gravity sewers within the streets to replace the existing collection system. Figure 2-3 shows the former Brewer Company housing area Phase 2 collection system established by FAI in May 2020.



Figure 2-3. Naalehu Phase 2 Collection System Layout-Former Brewer Company House Lots

2.5 Collection System Alternatives

- Two collection system alternatives were evaluated for Naalehu:
- Phase 1 Only: Reuse Existing Collection System: This alternative would construct the Phase 1 project to close the LCS. The existing collection system would continue to be used.
 - Phase 1+2: All New Collection System: This alternative would construct both Phase 1 and Phase 2 to create an all-new collection system for the community.

Naalehu WWTP Revised PER

2.6 Collection System Costs

Table 2-2 summarizes the capital, operation and maintenance (0&M), equipment replacement, and life-cycle costs for the two collection system alternatives. The capital costs were calculated by FAI in their Naalehu Wastewater Collection System Improvements Technical Memorandum (Appendix D) conducted in May 2020, adjusted to current (June 2023) dollars. Phase 1 capital costs include an additional \$4.4 million from FAI's estimate to account for capital costs related to reusing the existing collection system, such as inspection, cleaning, and repairing of existing defects. Both scenarios include an estimated \$400,000 WWPS equipment replacement cost after 20 years. The life-cycle costs consist of the 30-year net present value (NPV) of the capital, 0&M, after 20 years equipment replacement costs. Additional detail is included in Appendix A.

Table 2-2	2. Collection S	iystem Alternative	s Cost Summary	
Collection System Alternative	Capital Cost (\$M)	Annual O&M Cost (\$/year)	WWPS Equipment Replacement (20-yr)	Life-Cycle Cost (\$M)
Phase 1 Only: Reuse Existing Collection System	\$30.0	\$240,000	\$400,000	\$35.9
Phase 1+ Phase 2: All New Collection System	\$40.2	\$74,000	\$400,000	\$42.2

Table 2-3 provides a summary of capital cost assumptions used for the collection system cost analysis.

Description	Value
Estimate date	June 2023
Engineering News Record 20-Cities Average Construction Cost Index	13,345
Engineering, administration, and legal markup	25 percent
Estimating contingency for unknowns	20 percent

The life-cycle cost evaluation consists of a NPV comparison of the two alternatives. The NPV analysis includes capital, annual 0&M, and periodic equipment replacement costs. The capital expenditure is assumed to occur in year 1, and annual 0&M costs are incurred during years 2 through 30. A pump station equipment replacement project is assumed to occur in year 20. An appropriate inflationary station equipment rate are applied to the cash flow projections to obtain the NPV over a 30-year planning period.

The NPV of an alternative represents the amount of money that would need to be set aside today (at a given interest rate) to pay the costs associated with the alternative over the entire planning period. The alternative with the lowest NPV is considered the most attractive from an economic perspective.





Table 2-4 summarizes the life-cycle cost evaluation assumptions.

Table 2-4. Life-Cycle Ec	onomic Assumptions
scription	Value
ar of analysis	2023
anning period	30 years
flation rate	3.5 percent
scount rate	5.0 percent
juipment replacement cycle	20 years

The life-cycle costs are shown graphically on Figure 2.4. Reusing the existing Brewer Company collection system has the lowest capital and life-cycle costs.



Figure 2-4. Life-Cycle Cost Comparison of Collection System Alternatives

Brown Act Caldwell

Section 2

2.7 Recommendation

Naalehu WWTP Revised PER

Although reusing the existing collection system appears to incur lower life-cycle costs than constructing an all-new collection system, it is not recommended. Due to the advanced age of the existing collection system, the option would incur substantial financial, public health, and

 The piping is at the end of its useful service life; catastrophic failures are likely to increase in frequency, resulting in greater risks to public health and the environment.

environmental risks to the County, as summarized below:

- Most of the system is located in backyard easements, making it difficult to access and maintain.
 - System expansion to accommodate sewering additional areas of the town (in accordance with the Kau CDP) would not be feasible.
 - The option does not address the Administrative Order on Consent requirement to connect additional properties that are currently not connected to the collection system to the WWTP.

A new conventional gravity sewer collection system constructed in the streets of the Brewer Company lot development (Phase 2) is a viable solution to meet the wastwater collection needs of the town of Naalehu. From a technical perspective, the Phase 1 + Phase 2 option is recommended for implementation should a new WWTP be constructed. It is anticipated that additional nonfechnical considerations such as financing options, community input, burden on the Naalehu community, and detailed alignment with County long term planning goals will be addressed outside of this preliminary engineering report.



Section 3

Flow and Load Projections

This section summarizes the wastewater flow and load projections for the new Naalehu wastewater treatment plant (WWTP).

3.1 Flow Projections Based on City and County of Honolulu Standards

Section 11-62-24(b) of the Hawaii Administrative Rules (HAR) requires Counties to use their adopted wastewater flow standards to develop flow projections for WWTPs. Counties are to use the City and County of Honolulu (CCH) flow standards if they have not adopted their own standards. The County of Hawaii has not adopted its own flow standards, so wastewater flow projections were developed using Lourent CCH wastewater its andards (2017). Table 3-1 summarizes the flow projections.

17 CCH Standards	Peaking Factor	1.0	2.0	2.5	3.1	
aalehu WWTP Flows Based on 201	Value (gallons per day)	225,000	446,000	563,000	480 gallons per minute (691,000 gallons per day)	
Table 3-1. N	Description	Average dry weather flow	Peak day dry weather flow	Peak day wet weather flow ^a	Peak hour wet weather flow	· · ·

 Peak day wet weather flow is not part of the CCH standards but is an important WWTP design paramet Peak day wet weather flow estimate was developed using an appropriate peaking factor. The CCH standards were established for a major metropolitan area that includes vast areas of residential, commercial, and industrial development, with significant proportions of service areas near sea level elevations. Wastewater generation rates are generally lower in rural areas than in urban areas. The County's experience with the CCH flow standards on other projects (e.g., Honokaa WTP) has illustrated that the standards are very conservative for small rural communities located at higher elevations on ethy conservative for small rural communities located Honolulu are likely overly conservative for rural communities like Naalehu.

3.2 Reduced Flows Based on Potable Water Records

The amount of wastewater generated within a residence will not exceed the amount of potable water used by the occupants. Therefore, potable water use records can be used to estimate wastewater generation rates within existing communities where no combined sewers are present. The County of Hawaii Department of Water Supply provided potable water use records for the parcell cocated within the service area from November 2017 through October 2022. Evaluation of the potable water we data is discussed in Appendix B. Analysis of the potable water use records indicates that an 80.000 gallons per day (gpd) monthly wastewater generation rate would reflect the current needs of



Naalehu WWTP Revised PER

the service area. Using a 2.5 peaking factor to estimate the maximum wastewater flow into the collection system results in a maximum wastewater flow of 200,000 gpd.

3.2.1 Dry Weather I/I Allowance

Groundwater can infiltrate into wastewater collection systems during dry weather, increasing flows to the WWTP. The 2D17 CCH standards specify a dry weather infiltration and inflow (//) allowance of 35 gallons per capita per day (gpcd). The previous CCH standards (dated 1993) specified a dry weather I/I allowance of 5 gpcd for properties located above the groundwater table. Through the County's experience at Honokaa evaluating dry weather I/I for a rural collection system located in Hawaii Island's well-drained geology, at elevations hundreds of feet above sea level and a significant distance from the shoreline, we conclude that continued use of the 1993 standard for dry weather discussion is provided in papendix B.

3.2.2 Wet Weather I/I Allowance

The 2017 CCH standards, which specify a wet weather I/I allowance of 3,000 gallons per acre per day, were used for all wet weather I/I calculations.

3.2.3 Reduced Flow Projections

Accurately quantifying flow projections for the Naalehu community is necessary to design an appropriately sized wastewater treatment and disposal facility. The WWTP design will provide sufficient capacity for the existing parcels within the service area including newly accessible parcels. The flocting current development. This will allow the County to close the three LCGs. Furthermore, the design will provide sufficient area within the WWTP site for future expansion. Table 3-2 provides a summary of the calculated WWTP capacities for the reduced flow projections and for the flow projections for future development based on the 2017 CCH Standards.

	Table 3-2. Naalehu WWTP Calculated F	low Capacity
Description	Reduced Flow Projections	How Projections Based on 2017 CCH Standards
Base sanitary flow	80,000 gpd	147,000 gpd
Peak hour sanitary flow	200,000 gpd	368,000 gpd
Dry weather I/I	12,000 gpd	78,000 gpd
Wet weather I/I	245,000 gpd	245,000 gpd
Average dry weather flow	92,000 gpd	225,000 gpd
Peak day dry weather flow	212,000 gpd	446,000 gpd
Peak day wet weather flow	322,000 gpd Peaking Factor = 3.5	563,000 gpd Peaking Factor = 2.5
Peak hour wet weather flow	317 gpm (457,000 gpd)	480 gpm (691,000 gpd)

HAR 11-62-23.1(i) requires the initiation of a facility planning process when the actual wastewater flows (measured at the WWTP) reach 75 percent of the design capacity of the WWTP), and implementation of the facility plan must be initiated when actual wastewater flows (measured at the WWTP) reach 90 percent of the design capacity. In anticipation of future development, we recommend the WWTP design be rated to treat an average dry weather flow of 125,000 gpd to avoid the potential of having to initiate a facility plan shortly after the project is constructed. Note that the



Section 3

biological processes in the mechanical WWTP will need to be sized to treat the peak day dry weather flow of 212,000 gpd, not the average dry weather flow.

The proposed WWTP design capacity is based on actual water use data to establish wastewater generation rates, and rational assumptions to establish dry weather *I/I* allowances, and we believe it is appropriate for the existing conditions, while providing limited capacity for growth. Table 3-3 presents the recommended design capacity for the reduced flow projections.

Table 3-	3. Recommended WWTP C	apacity
Description	Value	Peaking Factor
Average dry weather flow	125,000 gpd	1.0
Peak day dry weather flow	212,000 gpd	1.7
Peak day wet weather flow	322,000 gpd	2.6
Peak hour wet weather flow	457,000 gpd (317 gpm)	3.7

3.2.4 Flow Variance

If the County pursues a WWTP approach (using the recommended capacity in Table 3-3) to close the LCCs then a DOH variance from HAR 11-62 requirements will be needed. The variance will need to be renewed every 5 years. The WWTP capacity needs should be re-evaluated upon application for the variance renewal.

3.3 Influent Characteristics

The properties within the existing service area are primarily residential, but do include several commercial, agricultural, and industrial zoned parcels. The wastewater characteristics of the WWTP influent are assumed to be similar to typical domestic wastewater. Table 3.4 provides a summary of the assumed influent characteristics.

Table 3-4. Summary of Assumed Influent (Characteristics
Parameter	Value (mg/L)
5-day biochemical oxygen demand (BOD_5)	300
Total suspended solids (TSS)	300
Total nitrogen	40
Total phosphorus	7

3.4 Influent Mass Loads

Table 3-5 summarizes the projected loads to the WWTP, based on the proposed peak day dry weather capacity of 212,000 gpd and the influent characteristics presented above.

Naalehu WWTP Revised PER

Table 3-5. Projected Peak Drv W	eather Dav Influent Mass Loads
Description	Value (Ibs/day)
BOD5	530
TSS	530
Total nitrogen	71
Total phosphorus	12





				Naalehu WWTP Revised PER Section 4
Section 4. Effluer	it Mana	gement Alt	ternatives	 4.1.3 Water Recycling This section summarizes Brown and Caldwell's (BC's) evaluation of water recycling as the primary effluent management system. 4.1.3.1 Irrigation
and Re	gulator	'y Requiren	nents	An irrigation assessment was completed to determine the viability of water recycling as the primary effluent management system, assuming the recycled water would be used to irrigate nearby coffee trees or other agricultural crops.
Effluent managem regulatory requiren	ent alternatives are e nents for the recomm	valuated in this section, followe ended effluent management sy	ed by an assessment of ystem.	Figure 4-1 presents a summary of the assessment, which shows there is typically no irrigation demand for 3 months of the year (November through January) due to high rainfall. In addition, the DOH requires that all water recycling programs have a 100 percent backup disposal system in place to handle flow
4.1 Effluent	Management ,	Alternatives		that does not meet recycled water quality standards or when recycled water supply exceeds demand. Therefore, water recycling is not a viable primary or sole effluent management strategy for the
Effluent managem	ent alternatives are e	valuated below.		community at this time. However, water recycling treatment, storage, and distribution systems could be added in the future. In addition to nearby irrigated agricultural reuse, the Naalehu Park and Naalehu
4.1.1 Ocean Di	scharge			Elementary School have significant areas of turf that could be considered for future irrigation with
Ocean discharge o due to the long dis stringent receiving	of treated effluent is n tance to the shoreline water quality standar	ot considered a viable alternati a (approximately 2 miles), high ds, high receiving water monito viried harmite. The constal work	ive for this small community cost to construct an outfall, oring cost, and difficulty and	recycled water that meets the DUH 'K-L' standards.
classified as "AA" I Administrative Rulk	marine waters by Stat ss (HAR) 11-54 does	te of Hawaii Department of Hea not allow zones of mixing in war	alth (DOH). Hawaii ters up to a distance of	10 Notite
300 meters (1,000 18 meters (10 fath	D feet) offshore if ther toms). The water qual	e is no defined reef area and if lity criteria for Clas	f the depth is greater than ss AA embayments are listed	population of the second secon
in Table 4-1. If a m be required to trea levels is not feasib mixing zone and ar	it water to meet the a the with current techno le with current techno outfall at least 1,000	ided, then a WWTP discharging pplicable water quality criteria. Jogies. Therefore, ocean disch: 0 feet offshore would be requir	g to the coastal waters would Treatment to the specified arge is not feasible without a 'ed for Naalehu.	a to the separate of the separ
	Table 4-1. Nutrient W	ater Quality Standards for Class AA Em	ibayments	2 c noite
Parameter	Geometric mean not to exceed (µg/L)	Not to exceed the given value more than 10% of the time (µg/L)	Not to exceed the given value more than 2% of the time (µg/L)	1 1 1 1 1 1 000 1 1 1 0 1 1 0 00
Total nitrogen	200	350	500	
Ammonia nitrogen	9	13	20	JAN FEB MAR APR MAY JUN JUL AUG SEP OCT NOV DEC
Nitrate + nitrate nitrogen	80	20	35	
Total phosphorus	25	50	75	Figure 4-1. Irrigation Demand Assessment
4.1.2 Subsurfa	ce Disposal via Inje	ection Wells	- - - - -	4.1.3.2. Chords Wister
Per HAR, Title 11, the DOH Undergroi the shoreline. Sinc viable alternative. permits "for the co disposal options ar wells is not feasible	Chapter 23, disposal und Injection Control ie the town of Namer In addition, per Envirr nstruction of sewage ie not available, feasi a.	to groundwater via an injection (UC) line. The UIC line in the Ni. (UIC) line. The UIC line in the UIC lin un is located mauka of the UIC lin unerital Protection Act 133, D wastewater injection wells unle ble or practical". Therefore, sut	well is not allowed mauka of aalehu area is located along ine, an injection well is not a DOH is prohibited from issuing ess alternative wastewater ssurface disposal via injection	4.1.3.2. Stock water The proposed WMTP site is located on a large ranch parcel. The County evaluated recycled water for stock watering purposes in November 2019 (Bc, November 2019). The study concluded that recycled water that meets DDI R-1 standards could be used for stock watering purposes. The estimated peak stock water demand was 29,000 gallons per day (gpd), approximately 23 percent of the proposed average dry weather flow capacity of the WMTP. Stock water use appears to be feasible if a water recycling program is developed in the future.
		Brown And Caldwell		Brown Act Caldwell:

4-1

4-2

4.1.4 Slow Rate Land Treatment

A potential project effluent management concept consists of Type 1 slow rate (SR) land treatment, which involves irrigation of vegetation with effluent. Type 1 slow rate land treatment differs from which involves irrigation of vegetation with effluent. Type 1 slow rate land treatment differs from irrigation needs of the vegetation. The potential effluent management concept calls for grading the site to contain all precipitation and planting native Hawaiian trees within the effluent disposal area. Effluent would be applied using surface (flood) irrigation techniques. The soils at the proposed WWTP location are suitable for SR land treatment. The proposed WWTP effluent management system will make use of an area containing Naalehu medial sifty clay loam soil (Natural Resources Conservation Service (INRCS), 2018). This soil type is well drained with moderately high permeability. SR land treatment consists of irrigation of land and vegetation with effluent. Significant treatment is provided as the water percolates through the soil. The wegetation uses the nutritish in the effluent as fertilizer and transpires a portion of the applied water. SR land treatment serves as a means for final disposal of effluent.

4.1.5 Subsurface Drip Irrigation Disposal

Another effluent management concept is to retain the existing site topography along with the existing vegetation and use subsurface drip irrigation technology to apply the effluent within the effluent disposal area. The use of subsurface drip irrigation technology to disperse effluent at the site will allow the County to significantly reduce the amount of Clearing, grubbing, and grading required to allow the County to significantly reduce the amount of Clearing, grubbing, and grading required to construct the facility, as compared to slow rate land treatment.

Drip irrigation technology has evolved to the point where non-clog emitters are available for subsurface applications of effluent. Non-clog subsurface emitters decrease the potential for the irrigation components to be clogged by roots. Figure 4-2 illustrates the subsurface drip concept. Drip tubing with integral emitters is buried 6 to 9 inches below ground. Effluent emitters are typically designed to operate at a flow rate of 1 gallon per hour (gph) and are typically spaced every 2 feet along a drip line. Pressure compensating drip systems typically operate under pressures ranging from 10 to 45 pounds per square inch (psi).



Figure 4-2. Subsurface Drip Irrigation Concept (Courtesy of Geoflow, Inc.)

Brown we Caldwell

Naalehu WWTP Revised PER

Section 4

Subsurface drip irrigation technology incurs greater operation and maintenance than a surface irrigation system. The County would need to periodically flush the drip lines to remove debris. As

described below, a significant number of drip lines are necessary to accommodate peak flow rates. In addition, periodic chlorination would be required to remove biological growth from the drip lines. These O&M tasks would need to be completed on a regular schedule, because drip systems are buried and not readily accessible or observable. During periods of dry soil conditions, the County wuld need to inspect the disconsal area for parches of wer skill that would indicate a localized failure

would need to inspect the disposal area for patches of wet soil that would indicate a localized failure that requires repair. Flow and pressure monitoring would also be useful tools for validating the status of the subsurface drip system. The disposal area would be divided into multiple irrigation zones, allowing a zone to be taken out of service for maintenance purposes. A fence would be constructed around the site to deter entry by humans and ungulates.

The subsurface drip system would slowly disperse effluent 6 to 9 inches below the ground surface, therefore operating as a subsurface disposal system similar to a leach field. Effluent is not intended to surface with a properly operating subsurface drip system. Precipitation falling on the site would either percolate into the soil or run off as surface drainage. The amount of runoff from the site would vary with the storm intensity, precipitation rates in excess of the infiltrative capacity of the site soils would result in runoff. The implementation of a subsurface disposal system will allow the existing grading to be retained, because stormwater runoff will not come into contact with, and therefore will not contain, effluent.

4.1.6 Leach Field

A leach field could potentially be constructed for subsurface disposal of treated effluent. Preliminary assessment of the concept based on the site soil characteristics (NRCS, 2018) and HAR 11-62 indicate approximately 14 acres of leach fields would be required to accommodate the anticipated flow, which includes a 100-percent redundant drain field per the DOH requirements. There is insufficient soil area available at the proposed WVTP site to construct a leach field of this size; therefore, this alternative is considered to be not feasible.

4.1.7 Ramifications of United States Supreme Court Opinion on County of Maui, Hawaii v. Hawaii Wildlife Fund et. al.

The United States Supreme Court (USSC) has published its opinion on the case of Hawaii Wildlife Fund, et al. vs. County of Maui regarding whether the injection wells at the Lahaina Wastewater Reclamation Facility (WWRF) are subject to regulation under the Clean Water Act (CVN) National Pollutant Discharge Elimination System (NDES) program. The USSC ruled that a discWhN National pollutants that originate from a point source but are conveyed to navigable waters by a nonpoint source (in the Lahaina situation, groundwater) are subject to regulation under the NPES program if the nonpoint source is a "functional equivalent" of a point source discharge. Unfortunately, the term "functional equivalent" is not well defined at this time. The USSC decision offered two examples of what would and would not be considered functional equivalents to a direct discharge:

- A pipe ending a few feet from a navigable water and the pipe emits pollutants that travel those few feet through groundwater (or over the beach) to the navigable water would clearly be subject to NPDES regulation.
- If a pipe ends 50 miles from navigable waters and the pipe emits pollutants that travel with groundwater, mix with much other material, and end up in navigable waters only many years later, would not be subject to NPDES regulation.



groundwater that moves mauka to makai and into the ocean. Therefore, any pollutants that percolate to groundwater will eventually find their way to navigable waters.
A further complication is the State of Hawaii's water quality standards (WQS) that define ocean water quality criteria for discharge into the ocean. These standards are an order of magnitude lower (i.e., more restrictive) than what can be achieved with existing advanced treatment technology. Treatment technology is currently not available that can related by produce effluent that meets the WQS. While technology is currently not available that can related to where it is safe to drink (at great expense), inconcally that same purified water cannot not be discharged into the ocean in certain places or certain conditions because it would not meet the water quality standards the State has established for ocean waters.
4.1.7.3 United States District Court Ruling
Subsequent to the USSC decision, the U.S. District Court in Honolulu ruled the Lahaina WWRF injection wells are functionally equivalent to a direct discharge and are subject to the CWA and require an NPDES permit. The DOH is in the process of issuing an NPDES permit to the facility. The U.S. District Court Ruling establishes another data point in addition to the two provided by the USSC regarding whether a discharge is functionally equivalent to a direct discharge:
 The transit time from the Lahaina WWRF injection wells to the Pacific Ocean is a minimum of 84 days, reaches peak concentration in 9 to 10 months, and continues for 3 to 5 years after discharge to the injection wells. A dye tracer test was used to establish the transit time.
 The straight-line distance traveled from the injection wells to the Pacific Ocean is approximately 2,900 feet (0.55 miles), although the groundwater in the aquifer takes a non-linear route.
4.1.7.4 Functional Equivalent Risk
As discussed above, the USSC opinion establishes at least seven factors to be used to determine whether an indirect discharge is a "functional equivalent" to a direct discharge, and without regulatory action, legislative direction, or additional legal precedents, agencies that desire to implement an indirect discharge system are placed in a position of regulatory uncertainty. The risk can be eliminated by:
 Implementing a direct discharge system (e.g., ocean outfall) that is known to be subject to the NPDES program.
 Implementing a true zero-discharge system, i.e., evaporation.
All other forms of disposal carry risk of being a functional equivalent to a direct discharge. Site- specific factors need to be evaluated for each particular discharge system and location. However, each type of discharge system provides different degrees of treatment and environmental attenuation. Therefore, for a given site certain options present less risk than others.
Table 4-2 shows BC's opinion of the risk of various effluent management systems that have been discussed in relation to each other, assuming disposal systems that provide more treatment present less risk than disposal systems that provide less treatment.
Of the indirect discharge systems, discharge to groundwater via injection wells is shown as having the highest risk of being functionally equivalent to direct discharge because treated effluent is added directly to the groundwater in a discrete area.
Leach fields are shown to have lower risk than injection wells because effluent is applied near the surface over a large area and treatment occurs as the applied water percolates through the unsaturated soil.

Brown we Caldwell 4-5

Brown we Caldwell 4-6

guidance as to how situations between the two extremes could be handled. Specifically, seven

The two examples represent the two extreme situations. The USSC also offered some vague

Naalehu WWTP Revised PER

Section 4

relevant factors were listed that could be applied to specific situations: Transit time.

Distance travelled. N.

The nature of the material through which the pollutant travels. ю.

The extent to which the pollutant is diluted or chemically changed as it travels. 4

The amount of pollutant entering the navigable waters relative to the amount of the pollutant that leaves the point source. 5.

The manner by or area in which the pollutant enters the navigable waters. . Ö

7. The degree to which the pollution (at that point) has maintained its specific identity

cases, but not necessarily every case". The USSC was not able to provide additional guidance as to The USSC opinion further states that "time and distance will be the most important factors in most what would constitute a functional equivalent to a direct discharge, leaving substantial regulatory uncertainty that requires resolution.

4.1.7.1 The Road to Regulatory Clarity

often costly, and it is important for public agencies to have a level of confidence that a project will Unfortunately, the USSC opinion does not provide regulatory clarity. There are four potential ways environment by managing wastewater from communities. Environmental protection projects are Regulatory clarity is important to public agencies tasked with protecting public health and the meet regulatory requirements over the project lifecycle before public funds are expended that regulatory clarity can be achieved:

- regulatory guidance documents would be the fastest way to address the problem but would also Regulatory guidance documents: The U.S. Environmental Protection Agency (USEPA) or State of Hawaii could issue regulatory guidance documents that establish criteria for determining if a be the weakest approach because regulatory guidance documents aren't subject to a public particular discharge scenario is a functional equivalent to a direct discharge. Developing review process and don't carry the authority of the options listed below. Ч.
- whereby a discharge is a functional equivalent to a direct discharge. Development of regulations takes longer than development of regulatory guidance documents because a public participation Regulations: The USEPA and/or State of Hawaii could develop regulations that define conditions process is involved. сi
 - Legislation: The United States and/or State of Hawaii governments could enact legislation that defines what is functionally equivalent to a direct discharge. ю.
- Court rulings: The definition of what it means to be a functional equivalent to a direct discharge could be resolved via court cases. This would be the slowest pathway of the four. 4

Based on the above, it will likely take years to decades before the term "functional equivalent" is adequately defined to a point where agencies have regulatory clarity.

4.1.7.2 Complications of Discharges within the State of Hawaii

The USSC opinion is especially complicated for most forms of effluent discharge in the State of Hawaii, because of the unique hydrogeology of volcanic islands. Our islands are surrounded by the navigable waters of the Pacific Ocean, and water that percolates into the soil that does not evaporate or is not taken up by vegetation via evapotranspiration will eventually find its way to the

	Table 4-2. Relative Risk of Being	g Functional Equivalent to Direct Discharge
Classification	Risk of Being Functional Equivalent to Direct Discharge	Effluent Disposal System
Direct discharge	Not applicable	Direct discharge to ocean
	Highest risk	🗼 Discharge to groundwater via injection wells
:		Leach fields

Slow rate land treatment and subsurface drip disposal are shown as having lower risk because the effluent is applied over a large area at or near the surface, and the presence of managed vegetation allows a higher degree of treatment to occur before the water percolates to groundwater.

Slow rate land treatment and subsurface drip disposal systems

Water recycling

Lowest risk

Indirect discharge

Water recycling in the form of irrigation carries the lowest risk because it produces the smallest volume of deep percolate of the options shown.

4.1.7.5 Functional Equivalent Risk at Naalehu

At the current time we believe the risk of any of the feasible effluent management alternatives for Naalehu (water recycling, slow ratel land treatment, subsurface drip disposal, and leach field) being found to be the functional equivalent to a direct discharge are low due to the following considerations:

- The effluent management system(s) would be located at an elevation of approximately 680 feet above sea level, creating a large vadose zone that applied effluent would need to travel down before reaching the basal groundwater lens.
- The effluent disposal site is located nearly 2.5 miles from the shoreline, likely resulting in a long travel time.
- The effluent flows will be relatively small at this facility, and there are no documented groundwater quality problems in the area.

However, it must be noted that future court rulings, legal action, legislation, or regulatory actions could render the facility to be functionally equivalent to a direct discharge and subject to NPDES permitting requirements.

4 1.8 Recommendation

Subsurface drip irrigation system is the recommended method of effluent disposal for the Naalehu WWTP. Subsurface drip will incur lower capital cost and require less attention from WWTP operators with respect to vegetation maintenance than slow rate land treatment. Subsurface drip requires periodic maintenance chlorinate biofouling in the drip lines. Recommended design criteria for the subsurface drip irrigation system are presented in Table 4.3. The disposal system wild be sized to handle the peak day wet weather flow of 322,000 gpd. An irrigation equalization and control tank are proposed to equalize higher peak flows and to allow discrete dosing of the irrigation zones.

HAR 11-62 requires a fully redundant subsurface disposal system. The design criteria listed in Table 4-3 are based on providing a subsurface drip system that is two times larger than needed in order to satisfy the HAR 11-62 requirement for redundaroy. The drip system ould be divided into two separate systems so that the peak day we twather flow can be disposed on the site using one system while the second system is out of service for maintenance.



Section 4

Naalehu WWTP Revised PER

Table 4-3. Recommended Subsurf	ace Drip Disposal Design Criteria
Description	Value
Average dry weather flow	125,000 gpd (87 gpm)
Peak day wet weather flow	322,000 gpd (224 gpm)
Irrigation equalization and control tank volume	20,000 gallons
Disposal area	5.2 acres
Subsurface drip emitters	1 gallon per hour, pressure compensating
Number of emitters needed for peak day wet weather flow	13,417 emitters
Number of systems	2 (1 active, 1 redundant)
Number of emitters provided to provide 2x redundancy	26,833 total emitters
Emitter spacing	2 feet
Drip line length per system	26,833 feet
Total drip line length	53,667 feet
Drip line spacing	4 feet
Drip line depth	6 to 9 inches
Number of irrigation zones	4 (2 per system)
Length of drip line per zone	13,417 feet
How per irrigation zone	112 gpm

During high flow conditions the irrigation control system would open multiple irrigation zones to accommodate the disposal needs. Additional drip lines will need to be added when the WWTP capacity is expanded. The minimum spacing between drip lines is 2 feet, so there will be sufficient space between the initial drip lines to add additional drip lines as part of future expansion project(s)

Flow meter(s) and pressure indicators

Irrigation system monitoring

4.2 Treatment Requirements

The DOH regulates subsurface drip irrigation disposal as "land disposal" per HAR 11-62. Table 4-4 lists the applicable effluent requirements for land disposal applicable to the project in effect at the time this report was prepared.

Table 4-	4. Applicable HAR 11-62 Land Disposal Requirements	
Description	Value	HAR Reference
5-day biochemical oxygen demand (BOD5)	30 milligrams per Liter (mg/L) monthly average 60 mg/L peak	11-62-26
ISS	30 mg/L monthly average 60 mg/L peak	11-62-26
Disinfection	Except for subsurface disposal systems, continuous disinfection of the treated effluent shall be provided.	11-62-24
Setbacks	Treatment units shall be not less than 25 feet from property lines nor less than 10 feet from any building.	11-62-23.1
Public accessibility control	6-foot-high fence surrounding treatment units	11-62-08



Wastewater Treatment Evaluations

This section summarizes the evaluations conducted as part of developing the proposed wastewater treatment plant (WWTP).

5.1 Preliminary Treatment

The preliminary treatment system will include influent flow measurement, influent sampling equipment, screening, and grit removal.

5.1.1 Influent Flow Measurement

Influent flow measurement is recommended to allow assessment of flows and loads to the biological treatment process, and to assess the biological treatment process performance. A Parshall flume will length of at least 20 times the flume throat width will be provided upstream of the flume to provide be provided upstream of the screening system to continuously record influent flow rates. Parshall flumes work well for influent measurement because the flume can operate in an open-channel configuration, can accommodate wide ranges of flows, and is self-cleaning. A straight approach favorable hydraulic conditions.

5.1.2 Influent Flow Sampling

to be collected. Influent composite samples, when combined with influent flow measurement, can be used to calculate influent mass loading rates to the WWTP to assess the treatment performance and An automatic refrigerated composite sampler is recommended to allow influent composite samples optimization of aeration rates in the biological treatment process. Periodic influent sampling is also recommended to monitor for changes in the influent characteristics.

5.1.3 Screening

Screening is recommended to protect the downstream system operations from large objects, debris, screenings volume at the Naalehu WWTP is expected to be small, subsequently screenings disposal is expected to be infrequent; weekly at most. Therefore, the screenings must be washed of organic (1/4 inch) openings are frequently used for activated sludge treatment systems. Finer screens are wipes, and rags that can be present in wastewater. The industry trend is towards finer screening systems that remove greater amounts of debris from the waste stream; screens with 6-millimeter debris to prevent the accumulation of nuisance odors and flies in the screenings barrel or bag used upstream of membrane bioreactors to remove hair that can foul the membranes. The between screening disposal events.

5.1.3.1 In-Channel Cylindrical Screen

We recommend an in-channel cylindrical screen for this installation. The in-channel cylindrical screen single unit. The screening portion consists of an inclined screen basket inserted into the wastewater combines screening, screenings washing, dewatering, compacting, and bagging/disposal within a screening surface, allowing finer screening of the wastewater. Controlled by head loss, a rake arm channel. The screening basket can consist of bars, perforated plates or sieves, depending on the application and clear opening required. The controls can be set to allow a mat to build up on the



Naalehu WWTP Revised PER

conveys the screenings from the hopper through an inclined tube, which dewaters and compacts the screenings. The tube includes a perforated dewatering section. The discharged screenings are about starts rotating within the screen basket, pushing the screenings off the rake and into a perforated screenings hopper located at the screen's central axis. A shafted auger along the screen axis 40-percent dry and can be discharged into a bin or directly into a bagging system. Figure 5-1 illustrates the process. Manufacturers include Lakeside and Huber. The key benefit to this odor potential. For this installation, the headworks will include one in-channel cylindrical screen, plus system is the integrated screenings washing system, minimizing additional screenings handling and a bypass channel with manually cleaned bar rack.



Figure 5-1. In-Channel Cylindrical Screen

5.1.4 Grit Removal

Grit is comprised of particles that are heavier than the organic biodegradable matter in wastewater. abrasive wear to mechanical equipment, piping, and appurtenances. Grit can also form deposits in pipelines, channels, and tanks, which reduces hydraulic capacity and can damage equipment. Removal of grit is very important to help prevent wear to downstream equipment, costly service Grit particles can consist of sand, gravel, pebbles, silt, cinders, ground bone, eggshells, coffee grounds, and other materials. Grit in the wastewater collection and treatment system causes interruptions and repair

At this point, the largest materials have been removed by the screens and will not interfere with grit Grit removal systems usually are placed between screening and downstream treatment processes. handling equipment.

There are several types of grit removal methods, including induced vortex grit removal, aerated grit chambers, and lamella plate settlers. The type of grit removal chosen is mainly dependent on the size of the incoming grit particles and the desired capture rate. Removed grit must be washed, dewatered, and disposed.





5.1.4.1 Induced Vortex Grit Removal

Historically, vortex grit removal, or the circular grit chamber, has been the most widely used method for grit removal in the United States, Vortex grit removal relies on the principle that grit has a greater specific gravity than organic matter.

bottom funnels the settled grit into a hopper below the basin. A sloped bottom vortex grit unit cross There are two configurations of vortex grit removal systems: a sloped bottom unit and a flat bottom longer flow path is designed to achieve a sufficient retention time to allow grit to settle. The sloped unit. The sloped bottom unit relies on particle settling to remove grit. Flow enters the grit chamber tangentially to provide the longest flow path around the inside of the circular grit chamber. This section is shown in Figure 5-2.



Figure 5-2. Sloped Bottom Vortex Grit Removal Cross Section

The flat bottom vortex system relies on hydraulic removal instead of specific gravity alone to remove grit from the wastewater stream. Flat bottom vortex systems use two paddles within the interior of the grit chamber that induce a toroidal flow pattern to move grit along the bottom towards the center. Once collected at the center of the grit chamber, a propeller forces excess grit down into the hopper. A flat bottom PISTA® Grit unit is shown in Figure 5-3.



Figure 5-3. Flat Bottom PISTA® Grit Removal



Naalehu WWTP Revised PER

Section 5

5.1.4.2 Vortex Grit Removal Capture Rate

In BC's experience, it is necessary to de-rate vortex type grit removal units by a factor of 50 percent of the advertised capacity to achieve satisfactory performance, due to the short detention time in the chamber. At large flow rates, the small-chambered vortex type units tend to re-suspend smaller grit particles which become a problem for downstream processes. Table 5-1 lists some advantages and disadvantages of the vortex type grit removal.

Table 5-1. Induced Vortex-Ad	ivantages and Disadvantages
Advantages	Disadvantages
Low maintenance	 Re-suspends/low capture rate of fines
 Low headloss 	 Poor capture efficiency
Small footprint	

5.1.4.3 Aerated Grit Removal

wastewater stream. Aerated grit tanks are designed to induce sufficient vertical velocity in order to separate organic and inorganic solids. In theory, inorganic solids have a higher specific gravity than Aerated grit chambers are tanks that function specifically to remove inorganic solids from the organic solids, and therefore require higher vertical velocities to keep them in suspension. Air diffusers placed near one longitudinal tank wall induce a roll in the contents of the grit tank. This roll creates maximum velocities near the walls and lower velocities at the surface and bottom of the tank. The lower transverse horizontal velocities allow inorganic particles to settle out and be transported to the grit hopper by shear-induced currents.

Aerated grit chamber design is based on providing sufficient hydraulic detention time during peak wet weather flow (PWWF) conditions. In BC's experience it is necessary to provide at least 10 minutes of detention time to achieve satisfactory grit removal.

tanks can be difficult under varying hydraulic loads due to the need to make fine adjustments to the themselves require a larger footprint than induced vortex systems. Proper operation of aerated grit Aerated grit tanks can provide excellent grit removal with minimal headloss, but the chambers air diffusers. Figure 5-4 illustrates the particle settling action of an aerated grit chamber.



Figure 5-4. Aerated Grit Removal Schematic



PER	
Revised	
WWTP	
Naalehu	

Table 5-2 lists some advantages and disadvantages of aerated grit chambers.

-Advantages and Disadvantages	Disadvantages	 Large footprint Requires fine tuning diffuser airflow for optimal performance High capital cost High O&M cost due to blowers
Table 5-2. Aerated Grit Removal-	Advantages	 Low headloss Once airflow is dialed in, the maintenance is low Effective removal of fines Firective removal of fines Provides additional aeration; "freshens" sewage prior to primary darification. Reduces denitrification in primary clarifiers

A variation of aerated grit removal technology that can be used in small WWTPs like Naalehu is an aerated grit trap. A small, aerated tank is provided to allow grit to settle. Aeration is provided to maintain organic solids in suspension and to "freshen" the influent. Accumulated grit is periodically removed using a Vactor fruck.

5.1.4.4 Lamella Grit Removal

This proprietary technology from Eutek, called the HeadCell, consists of sloped trays stacked in deep tanks. Flow enters the tanks tangentially and establishes a vortex flow pattern. Solids settle onto each plate and fall toward an opening at the center of each plate. The grit collects at the cone staped bottom of the tank where it is pumped to be washed and dewatered. Effluent flows out of the trays, over a weir, and into an effluent.

Grit capture is all done hydraulically and there are no moving parts. The headloss through each HeadCell is around one foot. HeadCells can be sized to provide up to 50 mgd of capacity within a single unit. With the stacked tray design, the HeadCells can achieve a 95 percent capative rate of grit 75 microns and larger. The multiple trays provide a large surface area for settling multiple size particles. The treatment capacity of the HeadCell is greater than other technologies with the same footprint. Figure 5-5 is an illustration of a section cut through the HeadCell process.



Naalehu WWTP Revised PER

Section 5

Table 5-3 lists advantages and disadvantages of the HeadCell.

Table 5-3. Lame a Plate	Settling/HeadCel l –Advar	ntages and Disadvantages
Advantages		Disadvantages
 Effective removal of fines 	High capit	al cost
 Small footprint 	Short histo	ory of installations
 No moving parts 		
 Low operating cost 		

5.1.4.5 Grit Particle Size Considerations

Most grit technologies and literature assume that grit is a clean sand or silica particle with a specific gravity of 2.65. In reality, grit particles are often coated with facts, grease, and organic material that reduce the particle's specific gravity. Grit particles with lower specific gravity have lower settling velocities, behaving live lighter and smaller grit particles. The sand equivalent size (SES) is the size of a clean sand sphere that exhibits the same settling velocity as the coated grit particles. For example, a grease coated grit particle with a physical size of 200 microns may settle and behave like a clean particle with an SES of 150 microns.

5 1 4 6 Efficiency comparison

Each of the alternatives claims a minimum particle size and capture rate. These claims are based on the ideal, clean gitt particle. As previously discussed, in reality gift particles are coated with fats and greases and do not exhibit the behavior of ideal gift particles. The capture rates have to be derated to reflect the SES of the particles. Table 5-4 compares the claimed minimum particle size captured of the alternatives discussed.

Table 5-4. Grit C Alternative Induced Vortex Aerated Grit Removal HeadCell
--

The HeadCell is able to remove the finest particles, with up to 95 percent removal of particles with a physical size down to 75 microns.

5 1 4 7 Grit Removal Recommendation

A simple aerated grit trap located downstream of the screening process is recommended for the Naalehu WWTP. Accumulated grit would be periodically removed using a Vactor truck, and dried onsite in a small drying bed. The dewatered grit would be disposed at the landfill.

An aerated grit trap provides adequate performance with a relatively uncomplicated process. Although a Headcell grit removal system could potentially provide a slightly increased grit capture rate, that benefit is not likely to surpass its significantly higher costs and operational complexity. The capture rate of an aerated grit trap is sufficient to protect the downstream processes recommended in this report. High levels or grit removal are particularly important for anaerobic digestors, which are not anticipated for this facility.



5.1.5 Odor Control

A common location for foul odor is the headworks of a wastewater treatment plant. This odor is caused by hydrogen sulfice (H₂S), which is formed under anaerobic conditions of the wastewater collection system. Due to H₂S low solubility in wastewater, when there is an excessive concentration of H₂S in the wastewater or if there is turbulence, H₂S gas escapes into the atmosphere. This release throaces the distinct rottem egg small. In addition to H₂S, there are other foul odorus compounds that can be released from wastewater, such as ammonia, amines, diamines, mercaptans, skatole, and organic sulfides.

Treatment of foul odors can be approached in two ways: preventing odors through liquid treatment or controlling odors in the gas phase. While liquid treatment provides control of odors prior to their release, gas phase treatment involves the collection and treatment of gases once they have been eleased from wastewater. Treatment methods can be aimed at one type of odor or can treat a range of odors.

5.1.5.1 Biotrickling Filter

A biotrickling filter consists of a vessel containing plastic or foam media. Foul air is drawn through the media for treatment. A fixed film biomass is maintained on the media by circulating water over and media. If up for the filter must be added to the circulating water to provide the nutrients (nitogen and phosphorus) the biomass needs to grow. The biomass oxidizes odorous compounds from the foul air as it travels through the tower. A demister is provided to remove water droplets from the treated air stream. Figure 5-6 shows a schematic diagram of a biotrickling filter. Biotrickling filters work best when H₂S concentrations are greater than 10 to 25 parts per million, which is greater than what is expected at Naalehu, due to the small size of the collection system.





Section 5

5.1.5.2 Granular Activated Carbon

A granular activated carbon (GAC) scrubber is recommended for the Naalehu WWTP headworks. A GAC scrubber passes odorous air through a bed of activated carbon, which adsorbs the odorous constituents within the pore spaces of the carbon.

Chemical oxidation or reduction of some compounds can also occur. As pore spaces become occupied, efficiency degrades, and the carbon must be replaced or regenerated. Carbon is most effective on higher molecular weight molecules such as the organic suffur compounds, which makes it the technology of choice. Packaged GAC scrubbers are available for small headworks and vessels can be situated vertically, horizontally, or radially to optimize footprints and reduce structure elevation profiles. Figure 6-7: illustrates the process. The County ourrently operates GAC scrubbers at other facilities and purchases the GAC media in bulk to reduce costs.



5.1.6 Recommendation

- The following are recommended for the Naalehu WWTP headworks:
- Parshall flume influent flow measurement
- Refrigerated automatic composite sampler
- In-channel cylindrical screen with integrated washer
 - Aerated grit trap
- Covered channels with foul air collection and GAC scrubber



5.2 Secondary Treatment

Secondary treatment process provides 5-day biochemical oxygen demand (BOD $_{\rm S}$), total suspended various secondary treatment options including advantages, disadvantages and applicability to the solids (TSS), and nutrient removal via biological treatment. This section provides descriptions of Naalehu WWTP. The treatment options are then screened to identify technologies for further evaluation.

5.2.1 Membrane Bioreactor

A membrane bioreactor (MBR) has the smallest footprint of the various biological treatment systems with membrane filtration, eliminating the need for tertiary treatment if a very high-quality effluent is concentrations of the effluent (also known as "filtrate") are typically less than 1 milligrams per Liter available and provides the highest quality effluent. An MBR basically combines an aeration basin desired for water reuse purposes. Membranes provide an absolute barrier to large particles; TSS (mg/L). Effluent from an MBR process can meet stringent water recycling turbidity requirements without an additional filtration process.

separating the bacteria from the clean water. MBRs have thin membranes with many thousands of micro-perforations. Depending on the manufacturer, these perforations are 0.04 to 0.2 microns in The main difference between MBRs and other biological treatment technologies is the method of diameter, too small for the passage of most microorganisms or other particles present in the wastewater, but large enough to allow the passage of water molecules.

Figure 5-8 is an illustration of an MBR. Figure 5-9 shows submerged MBR membranes in clean water.





Brown we Caldwell 9-0 2-0

Naalehu WWTP Revised PER

Section 5



Figure 5-9. Membrane Cassettes at Johns Creek Environmental Campus Fulton County, Georgia

- Important considerations of an MBR system include:
- Small capacity MBRs can be purchased as a packaged treatment system.
- MBRs can be designed and programmed to achieve nutrient reduction.
- The membrane cost is significant, and membranes must be replaced every 10 to 15 years.
 - Membrane fouling can occur with wastewater with high fats, oils, and grease levels.
- MBRs require the use of membrane cleaning chemicals, typically NaOCI and citric acid
- The process requires a computer control system and is difficult to operate efficiently if the computer malfunctions.
- The process incurs high electrical power costs, relatively high costs for cleaning chemicals, and high overall 0&M costs. Highly skilled labor is required for some of the 0&M tasks.

The MBR process would produce an effluent that is of high quality and has a small footprint, but has high overall capital, 0&M, and life-cycle costs. MBR is retained for further evaluation.

Sequencing Batch Reactor 5.2.2

Sequencing batch reactors (SBRs) are fill-and-draw systems that combine the processes of activated Denitrification can be achieved by incorporating an anoxic fill step in the cycle or a separate anoxic sludge in a single reactor. The reactor is filled with wastewater, where aeration, settling, and decanting occurs. By combining these processes, the need for secondary settling is not required. zone. A minimum of two SBR reactors are typically used for the process. SBRs are capable of producing high quality effluent and are potentially space saving in that separate equalization (and thus increasing the footprint requirements). Considering these challenges, SBRs secondary sedimentation is not needed. However, SBRs are operated by a proprietary computer control system, cannot be operated in manual mode, and may require influent and/or effluent will not be considered further.







Section 5

5.2.3 Nereda (Granular Activated Sludge) Process

Naalehu WWTP Revised PER

The Nereda technology is a granular activated sludge process that utilizes proprietary granules in an SBR. Features of the process include simultaneous fill and draw, fast settling, and approximately 1/5 the footprint of traditional activated sludge systems. The process was developed in Europe and most current full-scale applications are located in Europe. In the U.S., the process is marketed by Aqua Aeron Systems, Inc, according to the supplier website, there are currently only two full-scale applicating systems than wastewater in the United States. One is a demonstration facility, and the othere is a 3.6 mgd facility in Alabama that began operation in early 2020.

Figure 5-10is a conceptual illustration of the Nereda process. Due to the challenges listed for an SBR and the lack of long-term operational experience in the United States, the Nereda process is considered not appropriate for the Naalehu WWTP application.



rigure o-tu, Nereua Process Courtesy: Aqua Aerobic Systems

5.2.4 Oxidation Ditch

An oxidation ditch is a variation of the complete-mix extended aeration activated-sludge process. The process generally has a long solids residence time (SRT) and high mixed liquor suspended solids (MLS) concentration, making it resilient to upset by park organic loads. The typical SRT for oxidation ditches ranges from 15 to 30 days, and the MLSS is generally between 2,000 and 5,000 mg/L. Oxidation ditches are often oval in shape and have been called "racetrack" reactors. The depth of the ditch typically ranges from 4 to 12 feet. Mechanical aerators in the ditch provide aeration and mixing. Strategic placement of the aerators creates aerobic and anoxic zones within the oxidation ditch, for effective nitrification and denitrification. Biological phosphorus removal is also possible.

Oxidation ditches are usually preceded by preliminary treatment, such as screening and grit removal. Primary settling is typically not included upstream of oxidation ditch systems. Return activated sludge is pumped from the secondary clarifier back into the ditch. Figure 5-11 presents a schematic of an oxidation ditch. Typically, rotating brush or disc mechanical aerators are used to move mixed liquor around the tank and to provide aeration. The aerators help mix scum into the water column for treatment. The rigorous mixing action of the mechanical aerators can generate off-spray. Oxidation ditches are not available as packaged treatment systems. Because of the large footprint requirements and non-available as packaged treatment units, the oxidation ditch process is eliminated from further evaluation.





5.2.6 Activated Sludge with Anoxic Selector

This process is similar to extended aeration but would employ a shorter SRT of less than 10 days and would operate at a MLSS concentration between 1,500 and 4,000 mg/L. Figure 5-13 shows a process schematic for this process. The Kihei WWFF and Wailuku-Kehului WWFFs on Maui operate with this process. The anoxic selector is typically sized to have a volume of approximately 10 to 30 percent of the total aeration basin volume. The process would not be as forgiving and resistant to shock loadings compared to the oxidation ditch and extended aeration processes due to the shorter SRT and lower MLSS concentration. But this option is available in a prefabricated package plants and would incur a smaller footprint than the oxidation ditch and extended aeration processes but would require operation and maintenance of blowers to provide air to the process. The fine bubble diffused aeration system would be more efficient than the mechanical aerators generally used in the oxidation ditch process. This process is retained for further evaluation.



Figure 5-13. Activated Sludge with Anoxic Selector Process Schematic

5.2.7 Recirculating Gravel Filter

Recirculating gravel filter (RGF) technology is an effective technology to treat septic tank effluent wastewater. After collection and conveyance, the wastewater is treated, in this case using a recirculating per gravel filter. RGFs are a relatively simple, but effective means to treat wastewater from small communities. RGFs have been used to treat flow rates up to 1.0 mgd. RGFs typically produce a nitrified effluent that contains less than 10 mg/L of BODs and TSS (Crites and Tchobanoglous, 1998).

A schematic diagram of a RGF is shown in Figure 5-14. A septic tank is used to capture settleable and floatable solids. The septic tank effluent enters a recirculation tank. A dosing pump is used to apply wastewater in small doses to the top of the filter. The wastewater is treated as it percolates through the pea gravel media. A network of drainage piping collects the water at the bottom of the filter and returns it to the recirculation tank. A floating ball recirculation valve controls the return flow back to the recirculation tank. A floating ball recirculation valve controls the return flow back to the recirculation tank or to the effluent disposal or reuse system. The dosing pump timer settings and recirculation tank volume are designed so that wastewater will typically flow through the filter for treatment an average of three to five times before being discharged. An example of a RGF system in use within a decentralized wastewater system can be found at the Stonehurst subdivision, located near Martinez, *Giltornia* (*Cires*, et. al. 1997).

For a community system with conventional sewers and Imhoff tank can be used in lieu of a septic tank. Imhoff tanks are designed to remove floatable and settleable solids, and also provides for some digestion of the removed materials.



Naalehu WWTP Revised PER

Section 5

Section 5



5.2.8 Secondary Treatment Technology Screening

Table 5-5 provides a screening evaluation of the secondary treatment technologies described above. MBR, activated sludge with anoxic selector, and recirculating gravel filter are carried forward as project alternatives in Section 7.



	Section 5
ill pathogens or mployed for thin nts, odor, aest ction of public n for wastewal ms, disinfection ms, disinfection	render them incapable of reproduction or harm to e purposes of protection of public health, reduction netics, and maintaining waste-assimilative capacity health through the control of disease-causing ter disinfection (WEF, 1996). As the last barrier of at WWTPs is an important process. To address system and a UV system were evaluated.
most commonly 15 percent. Ho anitizers usually commonly used duct of degrad duct of degrad tion should be	/ used form of liquid hypochlorite. It is an effective ionly known as "bleach," NaOCI can be found in usehold bleach usually has a NaOCI concentration have a NaOCI concentration of 11 to 15 percent. for wastewater treatment and will degrade to attion is chlorine gas. At low concentrations the stored in a cool, dark area in a non-corrosive
es used to trans be softened pri at) NaOCI.	sport diluted NaOCI if hardness is present in the or to mixing with the NaOCI or systems can be
NaOCI dissocia ous acid and hy h the free chlor me processes, h sinfection with vered in bulk or	tes to form hypochlorous acid and hypochlorite. pochlorite is referred to as free chlorine. When ine also have some disinfecting potential. These out are toxic to aquatic life if the processed water hypochlorite can form DBPs similar to chlorine generated on site.
s, drums, or sm to neighbor isle Table 5-6 lists a	aller packages as desired. Currently, NaOCI is ands in totes. NaOCI solutions are applied with a idvantages and disadvantages of bulk NaOCI.
Sodium Hypoch l ori	te-Advantages and Disadvantages
	Disadvantages
	 Cornsive and toxic Degrades quickly at high concentrations and temperatures Forms DBPs Forms DBPs Scaling in pipes and valves can occur High transportation cost due to weight of liquid High transportation cost due to weight of liquid 1.3.5% NaOCI sconsidered a hazardious chanical. A rehease of 100 lbs (approximately 12 gallons) or more is considered a reportable quantity. Air binding of pumps is possible at high temperatures due to off-gassing, but can be mitigated through proper system design.

5.3 Disinfection

humans. Disinfection at WWTPs is er Disinfection processes selectively kil of organic matter, inorganics, nutrier of receiving water bodies. The protect microorganisms is the primary reasc protection from pathogenic organisn disinfection, both sodium and calciu

5.3.1 Sodium Hypochlorite

of 3 to 6 percent. Swimming pool sar A concentration of 12.5 percent is co disinfectant at relatively low concent concentrations ranging from 1.5 to 1 10 percent in 6 to 8 weeks. The proc product is relatively stable. The solut container NaOCI is corrosive and toy Sodium hypochlorite (NaOCI) is the I

Scaling can occur in pipes and valves carrier water. The carrier water can b designed to transport undiluted (near

Not applicable, but largest overall footprint Not applicable, but largest overall footprint

Moderate

Large

Moderate

Small None High

>

×

×

>

Appropriate for remote island location

Total Nitrogen < 10 mg/L

Anoxic selector Nitrification

Appropriate for tropical clim

Aeration tank size

×

× ~ Low Low

Moderate Moderate Lange

> Higher Moderate

Moderate Largest Moderate

Moderate None Small

Moderate

Highest

None

Secondary clarifier size Operational complexity Energy requirement

High

High

Available as packaged treatment system

Largest Lange

×

×

Large footprint

Large footprint

Limited full scale installations in U.S.

Proprietary control systems

×

Carry forward in evaluations

Fatal flaw

The combined amount of hypochloro reacted, the compounds formed with chlorine residuals are desired in som is discharged to an open stream. Dis gas. For WWTPs, NaOCI can be delive Once added to the wastewater, the

5.3.1.1 Bulk Sodium Hypochlorite

Bulk NaOCI can be obtained in totes, produced on Oahu and transported t metering pump or suction injector. Ta

Table 5-6. Bulk Sodium Hypochlor	te-Advantages and Disadvantages
Advantages	Disadvantages
 Able to oxidize at relatively low concentrations 	Corrosive and toxic
Use is common at WWTPs	 Degrades quickly at high concentrations and temperatures
Readily available	Forms DBPs
 Simple Q&M-application with metering pumps 	 Scaling in pipes and values can occur
 Stored and used as a liquid 	 High transportation cost due to weight of liquid
	 12.5% NaOCI is considered a hazardous chemical. A release 100 lbs (approximately 12 gallons) or more is considered a reportable quantity.
	 Air binding of pumps is possible at high temperatures due to c gassing, but can be mitigated through proper system design.



Brown so Caldwell : 5-15

Section 5

Recirculating Gravel Filter

Activated Sludge with Anoxic Selector

Extended Aeration

Oxidation Ditch

Nereda

SBR

MBR

Criterion

B0D₅ ≤ 30 mg/L TSS ≤ 30 mg/L

Naalehu WWTP Revised PER

Table 5-5. Screening of Secondary Treatment Option

×

×

×

×

×

~

×

Section 5

The chlorine demand at the Naalehu WWTP is anticipated to be relatively small due to the small WWTP size and method of effluent disposal. Bulk NaOCI is not recommended due to cost and the chemical degradation rate in storage.

5.3.1.2 On-Site Generation

On site generation comprises mixing softened water with salt to form brine and then passing an electric current through the brine to produce NaOCI. When generated on site, the resulting solution is a relatively weak solution, usually around 0.8 percent. Because the process includes passing an electrical current through the solution, the electrical power demand can be quite high. Figure 5-15 is a schematic illustration of NaOCI generation system.



Figure 5-15. Schematic Diagram of an On-site NaOCI Generator

In addition to water hardness, the levels of silica present in the source water can cause build up problems in the on-site NaOCI generator. The level of pretreatment required is dependent on the water quality. A detailed water quality analysis would be necessary if this disinfection technology were chosen. Table 5-7 lists advantages and disadvantages of onsite NaOCI generation.

Table 5-1, Onsite Sodium Hypochlorite Generation-Advant Advantages Hypochlorite Generation-Advant Advantages addy available and inexpensive • High alectrical p • High maintenan by stable liquid product (pow concentration) • High maintenan • High maintenan • Plazardous chemical and is safer to handle re oncentrated solutions	Disadvantagas Al power demand ance requirements rocess generates flammable hydrogen gas by-products formed
ing in pumps due to off-gassing less likely to occur	

Onsite NaOCI generation is not recommended at the Naalehu WWTP due to the high maintenance requirements.

than with more concentrated solutions

Naalehu WWTP Revised PER

5.3.2 Calcium Hypochlorite

Calcium hypochlorite is the most common solid form of hypochlorite used for disinfection. It can be found as a powder, granules, pellets, or as tablets in concentrations up to 70 percent. Calcium hypochlorite will degrade in strength at a rate of 3 to 5 percent per year. Once applied to the astextearter, the chemistry is similar to that for NaOCI (i.e., bleach). Calcium hypochlorite decomposes in an exchemin reaction if exposed to moisture.

The solid can be directly applied to wastewater at small WWTPs. Figure 5-16 shows a typical calcium hypochlorite feed system.



Figure 5-16. Typical Calcium Hypochlorite Feed System

The advantages of using calcium hypochlorite for disinfection at small, remote WWTPs is that it is available in concentrated form as powder, pellets, or tablets. This makes the transportation and storage of disinfectant optimal for small WWTPs. Table 5-8 summaries calcium hypochlorite characteristics.

Description	Characteristic
Transported form	Solid
Typical transported concentration	70%
Largest transported volume available	55 pound pails
Decay rate	Decays 3-5% per year
Hazards	Toxic if ingested (usually through dust or liquid form)
Storage constraints	Must be stored in a cool, dry, dark place
Special equipment	Tablet feeder
Particular issues	Heats and combusts if not stored properly, scaling in pipes, off gassing

5.3 2.1 Dose and Contact Time

The effectiveness of a chlorination system is highly dependent on the characteristics of the wastewater, the initial mixing and contact time, and the chlorine dose used. For nitrified effluent, the recommended dose is between 4 and 8 mg/L (Crites and Tchobanoglous, 1998).



Brown we Caldwell

15 minutes (Ten States Standards Wastewater, Recommended Standards for Wastewater Facilities, chemical feed at a rate of up to 100 lbs/day, which will ensure an adequate chlorine dose for peak Table 5-9 lists the chlorine demand for various flow conditions. Equipment will be sized to provide 1997, Great Lakes-Upper Mississippi River Board of State and Provincial Public health and wet weather discharge flows. The recommended minimum contact time for chlorination is Environmental Managers).

Té	able 5-9. Chlorine	Demand
Description	Flow (mgd)	Chlorine Demand (lbs/day)
Average dry weather flow	0.125	4-8
Peak day wet weather flow	0.322	11-22
Peak hour wet weather flow	0.457 (317 gpm)	15-31

5.3.3 Ultraviolet Light Disinfection

control panel. Wastewater can flow either parallel or perpendicular to the lamps in the reactor, while the control box provides a starting voltage and maintains the continuous current needed. Currently, impeding their ability to reproduce. A UV disinfection system is comprised of lamps, a reactor, and most systems are equipped with an automated lamp cleaning system, to maintain lamp efficiency A common alternative to chlorine disinfection is ultraviolet light (UV). Ultraviolet systems destroy microorganisms by affecting their deoxyribonucleic acid (DNA) and ribonucleic acid (RNA) and evels. A UV system's effectiveness is dependent on the characteristics of the wastewater, the dose, and the microorganisms present at the required intensity. Ideally, the discharge undergoing treatment should water, which has a direct effect on the ability of UV light to penetrate through the liquid and reach exposure time. In the case of UV radiation, the most important factor is the transmittance of the not have a transmittance lower than 55 percent, with the intensity decreasing the farther the microorganisms are from the lamp. The optimum wavelength to effectively inactivate microorganisms is between 250 and 270 nanometers.

The main types of UV lamps used for wastewater disinfection are conventional low-pressure lamps, low pressure high output (LPHO) lamps and medium pressure lamps. Several UV systems include lamps with automated sleeve cleaning.

5.3.3.1 UV System Design Summary

assumed UV transmittance of 65 percent. The amalgam lamp used with the UV3000+ system has an electronic ballast. The electronic ballast allows the lamps to be dimmed to conserve power based on This ELLF has been tested and approved by the State of California and is also accepted by the State of Hawaii for reuse applications. The system would use LPHO lamps with automatic sleeve cleaning. end-of-lamp-life factor (ELLF) of 0.98 indicating little loss in UV light output over the life of the lamp. Disinfection occurs as the organism is exposed to the UV radiation as the water flows past the UV lightbulbs. The Trojan UV3000+ system is used at numerous facilities across the U.S., including cleaning devices to reduce labor requirements. Each UV lamp is enclosed in a quartz sleeve to separate it from the water medium. Each lamp draws 254 watts at full output and is driven by LPHO lamps are energy efficient and the UV3000+ system is furnished with automatic sleeve some treatment plants in Hawaii. The estimated cost included in this report are based on an A UV disinfection system requires about the same size footprint as chlorine at small WWTPs.



Naalehu WWTP Revised PER

Section 5

a control signal from a flow meter. The LPHO lamps will have a minimum life of 12,000 hours when Table 5-10 summarizes the size and design criteria for the UV system required to treat the WWTP operated in an automatic mode and limited to a maximum of 4 on/off cycles per 24 hours. discharge.

Table 5-10. UV Disinfe	ction Design Summary
Description	Value
Peak Hour Wet Weather Discharge	317 gpm
Minimum UV transmittance	65 percent
No. of UV channels	1
Design dose	35,000 µWs/cm2
Disinfection limit	30 MPN per 100 mL <i>(E.coli</i>)
Validation factors	0.98 end of lamp factor
MPN = most probable number	
and house and attention and any state	ouse continuter

5.3.4 Cost Evaluation

presented in Table 5-11 for comparison. Additional detail is included in Appendix A. The capital costs include the materials and equipment costs, construction costs, electrical, instrumentation and control, soft costs, and contingency. As shown in the table, the UV option incurs higher capital costs. The life-cycle costs look at the impact of the capital costs along with the annual operations and maintenance costs, including power, materials, chemicals, and labor costs over the next 30 years. The life-cycle costs for chlorination option appear to be about 55 percent of the UV option. A summary of capital and life-cycle estimated costs for both chlorination and UV disinfection is

Table	5-11. Estimated Disinfection Costs	
Description	Tablet Chlorination System	UV System
Capital Cost	\$150,000	\$1,100,000
Annual Operation and Maintenance	\$25,000	\$10,000
Life-cycle Cost (30-Year Net Present Value)	\$740,000	\$1,340,000
*Does not include annual labor.		

5.3.5 Disinfection Recommendation

Naalehu WWTP because it incurs lower capital and life-cycle costs. In addition, tablet chlorination will A tablet chlorination feed system is the recommended disinfection option over the UV system for the be more reliable than UV due to frequent "dirty power" conditions on the island. The County may elect to install a UV system at the Naalehu WWTP should they choose to pursue an R1 water recycling program in the future.

dispersed below the ground surface. However, periodic maintenance chlorination of the subsurface drip system will be required to reduce biofilm fouling within the drip lines. The proposed effluent management system (subsurface drip irrigation disposal) does not require a disinfection process to protect human health and the environment because the treated effluent is



Section 6

Solids Management

This section evaluates solids management options for the Naalehu WWTP.

6.1 Aerobic Digestion with Decant Thickening

aeration blowers, and therefore is not typically used at larger (i.e., greater than 10 million gallons per are oxidized in the process, stabilizing the sludge and reducing the total mass of solids that must be Aerobic digestion consists of aerating sludge in a tank for an extended period of time. Volatile solids produce biogas. The aerobic digestion process requires substantial energy input in the form of managed by recycling or disposal. Pathogen densities are also reduced. The process does not day [mgd]) wastewater reclamation facilities (WWRFs).

digestion to stabilize solids, due to its relatively low capital costs, simplicity, and compatibility with Many small (less than 5 mgd) wastewater treatment plants in the United States use aerobic the certain liquid treatment processes.

This process is repeated a few times until the sludge reaches approximately three percent solids. It is basin. The first stage includes a period of aerobic digestion as described above. In the second stage then decanted off the top. The blowers are turned back on to continue the aerobic digestion process the blowers are turned off for a period of time to allow sludge to settle and thicken. Supernatant is Aerobic digestion with decant thickening is a two-stage process that can be achieved in the same then pumped to the next process. Aerobic digestion with decant thickening is recommended for the proposed Naalehu WWTP due to its simplicity, low cost, and effectiveness for small WWRFs.

6.2 Anaerobic Digestion with Biogas Use

pathogen densities are reduced; and a stabilized sludge is produced. Modern high-rate digesters are Anaerobic digesters are covered tanks equipped with mixing, heating, and biogas collection systems. typically single-stage reactors. Mesophilic anaerobic digesters are typically operated at temperatures Anaerobic bacteria in the digesters convert organic matter into methane, carbon dioxide, and water; between 35 degrees Celsius (°C) and 38°C. Mesophilic digestion systems produce a Class B biosolids product if the solids retention time (SRT) is greater than 15 days. Two-stage mesophilic anaerobic digestion, where digesters are operated in series, improves process performance. The second-stage anaerobic digester generally has less SRT than the first stage. The advantages of this process configuration are slightly improved volatile solids reduction, a product with reduced pathogen content, and less product odor potential.

The anaerobic digestion process generates biogas that can be used for digester heating and generation of electricity. The mesophilic anaerobic digestion process requires primary sludge to operate effectively. Therefore, primary clarifiers are required for an anaerobic digestion process. WWRFs that do not have primary clarifiers must use other digestion technologies.

Anaerobic digestion is cost effective for facilities larger than 5 to 10 mgd. Anaerobic digestion is not considered to be an appropriate technology for a facility the size of the Naalehu WWTP



61

Naalehu WWTP Revised PER

6.3 Dewatering

A dewatering process is used to remove excess water from digested sludge to form a semi-solid 'cake" product.

6.3.1 Centrifuge Dewatering

have lower total solids content, on the order of 12 to 15 percent. The process is shown in Figure 6-1 cake of approximately 23 to 28 percent total solids content. Aerobically digested sludge will typically performance of the dewatering technologies presented in this report but require the highest energy input to do so. High-solids centrifuge machines typically achieve anaerobically digested dewatered Centrifuges are a commonly used dewatering technology. Centrifuges provide the best dewatering

-iquid sludge, conditioned with polymer, is introduced within the rotating bowl. The sludge spins with rotating bowl at a slightly faster or slower speed than the bowl to facilitate moving the solids fraction centrifuge, separating liquid from the solids. The centrifuge has a solid bowl that spins at a high rate the bowl, separating into liquid and solid fractions. A screw conveyor mechanism spins within the through another port. The process operates continuously. Required ancillary equipment includes towards one end of the bowl, where it is discharged. The centrate (removed liquid) is discharged Centrifugal force of 500 to 3,000 times the force of gravity is applied to the biosolids within the sludge feed pumps, polymer feed systems, and sludge cake conveyance systems.

Centrifuges are sized based on hydraulic and solids throughput. Machines are available to dewater sludge flow rates ranging from 25 gpm to 700 gpm. High-solids machines can produce a very welldewatered material, if anaerobic digestion is used. Centrifuges require a high level of operator due to the high rotational speed. For this reason, they are ypically not used at small WWTPs like Naalehu.





6-2

6.3.2 Screw Press Dewatering

The screw press represents a relatively new technology for dewatering municipal wastewater solids, although the technology has been used successfully in industrial, pulp and paper production, chemical, and food processing applications. Figure 6-2 shows a diagram of a screw press. Thickened sludge, conditioned with polymer, is introduced to the machine in the head box at the inlet end. The mixture is conveyed from the inlet end to the outlet end of the press by the rotating screw. As the material is conveyed along the length of the press it is squeazed between the tapered screw shell and the screen drums. The dewatered solids exit the press at the discharge end and fall down the discharge box. The adjustable pressure cone provides back pressure within the machine, particularly when the machine is initially filled. For municipal wastewater solids applications, provides sufficient back pressure. The idquid that was forced machine is filled; the dewatered sludge provides sufficient back pressure. The liquid that was forced machine is filled; the streme to the liquid treatment process.



The screw press operates at a very slow rotational speed. The screw rotation is usually one-half of a revolution per minute or less for municipal wastewater solids. Water is slowly forced from the sludge by squeezing action-similar to a belt filter press-but for much longer periods of time. The solids retention time in a screw press can be on the order of 2 hours. The simplicity of screw presses makes them practical for small wastewater treatment plants, such as Naalehu.

6.4 Disposal

Dewatered solids, grit, and screenings would be trucked to the West Hawaii Landfill for disposal.

Section 7

Section 6

Project Alternatives Evaluations

This section presents evaluations of the three project alternatives developed for the new wastewater treatment plant (WWTP).

7.1 Project Alternative Descriptions

Three project alternatives were developed as part of the tasks completed for this Preliminary Engineering Report. All three alternatives include a new C Brewer house lot collection system (i.e., Phase 1 + Phase 2), WWTP, and subsurface drip effluent disposal system.

7.1.1 Project Alternative 1: Activated Sludge with Anoxic Zone Package Plants

Project Alternative 1 is comprised of an activated sludge process with anoxic zone provided in the form of packaged treatment systems. A typical packaged treatment system of this type would include the following elements:

- Flow equalization
- Anoxic treatment zone
- Aerobic treatment zone
 - Secondary clarifier
- Aerobic digester with decant thickening





Figure 7-1 presents a sketch of Project Alternative 1. Wastewater would receive preliminary treatment in the headworks before flowing into the packaged treatment system. Two package treatment units would be provided, each with 62,500 gallons per day (gpd) capacity. Effluent would flow into an irrigation equalization tank before being applied to the subsurface drip disposal system. Digested solids would be dewatered using a screw press prior to disposal at the landfill.



Figure 7-1. Project Alternative 1: Activated Sludge with Anoxic Zone Package Plants

Naalehu WWTP Revised PER

Section 7

7.1.2 Project Alternative 2: MBR Package Plants

Project Alternative 2 is similar to Project Alternative 1 but includes two MBR package plants to provide treatment. Figure 7-2 provides an outline of Project Alternative 2. The MBR technology would create effluent that could be recycled for irrigation and/or stock watering purposes, if desired in the future. However, recycled water distribution costs are not included in the evaluations below to allow all alternatives to be considered on an equal basis.



Figure 7-2. Project Alternative 2: MBR Package Plants





<section-header><section-header><text><section-header><section-header><section-header><section-header></section-header></section-header></section-header></section-header></text></section-header></section-header>	ed PER Section 7	Naalehu WWTP Revised PER			Section 7
<section-header><text><text><text><text><text><text></text></text></text></text></text></text></section-header>	Alternative 3: Imhoff Tank/Recirculating Gravel Filter	7.2 Cost Evaluat	tions		
<section-header><text><section-header><text><text><text><text></text></text></text></text></section-header></text></section-header>	e 3 incorporates recirculating gravel filter treatment technology. Figure 7-3 provides e project alternative. An Imhoff tank would be provided downstream of the	Capital, operations and lection.	maintenance (0&M), and life-	sycle cost evaluations a	re presented in this
Survey and substantial production of the first would be the and intermed to the first would be the and the first would be the first would be the first would b	nove grease and setuleable solids prior to nowing into a recirculation tails. mps would distribute water from the recirculation tank over the surface of the pea	7.2.1 Capital Costs			
$ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \ \$	provides secondary treatment. Water collected at the bottom of the filter would flow subtion tank. On average water would flow through the filter five times before ibsurface drip irrigation system as previously described.	Conceptual cost estimat developed using constru devoloped using constru- bources. The costs were	tes were created for the three uction bids from similar projec e adjusted to account for econe	project alternatives. The ts, quantity take-offs, ve omies of scale and cons	e cost estimates were indor quotes, and other itruction inflation since
$\label{eq:product} Product of the $	Phase 2	adjustment to reflect Ha	mere nawall costs were unava awaii Island conditions.	liable, 0.5. Illalliaila co	Nere used arter
$\label{eq:production} \left(\begin{array}{c} \label{eq:production} \left(\begin{array}{c} \label{eq:production} \left(eq:$	System System Phase 1 Collection	In accordance with the A criteria, these are Class Viability Estimate. Typics to prepare planning leve planning.	Association for the Advanceme 5 estimates. A Class 5 estima ally, engineering is from 0 to 2 el cost scopes or evaluation of	int of Cost Engineering te is defined as a Cono percent complete. Clas alternative schemes, Ic	nternational (AACE) sptual Level or Project s 5 estimates are used ng range capital outlay
AnomeSolidsAnoth TarkSolidsAnoth TarkSolidsAnoth TarkSolidsAnoth TarkSolidsAnoth TarkSolidsAnoth TarkSolids <td>Grit & System Grit & Headworks</td> <td>Expected accuracy for Cl the technological comple appropriate contingency shown. Table 7-1 provid</td> <td>lass 5 estimates typically rang lexity of the project, appropriat determination. In unusual cir les a summary of capital cost :</td> <td>es from -50 to +100 pe e reference informatior cumstances, ranges co tssumptions used.</td> <td>rcent, depending on and the inclusion of an Ild exceed those</td>	Grit & System Grit & Headworks	Expected accuracy for Cl the technological comple appropriate contingency shown. Table 7-1 provid	lass 5 estimates typically rang lexity of the project, appropriat determination. In unusual cir les a summary of capital cost :	es from -50 to +100 pe e reference informatior cumstances, ranges co tssumptions used.	rcent, depending on and the inclusion of an Ild exceed those
Solution Legislation 			Table 7-1. Capital Cost Esti	nating Assumptions	
Entent of the Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Tark Entent and instrumention motion Entent and instrumention and instrumention motion Entent and instrumention and instrumention motion Entent and instrumention Entent and instrumentionFigure 7.3. Project Alternative 3.1 millionEntent and instrumention Entent and instrumention Entent and instrumention Entent and instrumentionFigure 7.3. Project Alternative 3.1 millionEntent and instrument and instrumention Entent and instrumention Entent and instrumention Entent and instrumention Entent and instrumentionFigure 7.3. millionEntent and Entent and Entent and Entent and Entent andFigure 7.3.	Solids Imhoff Tank		Description	Valu	Ð
Education Tark Tark Tark Encircleding Tark Encircleding Encircle		Estimate date		June 20	23
Frenchand Electrical and Instrumentation methy Endering administration, and regimenting administration, and regimenting Endering administration, and regimenting Electrical and Instrumentation methy Endering administration, and regimenting Estimating administration, and regimenting Electrical Estimating administration, and regimenting Estimating administration, and regimenting Electrical Estimating Ele		Engineering N	Vews Record 20-Cities Average Constructio	n Cost Index 13,34	5
Engineering, and ministration, and regulation Spreering Spreering Entiting contingent for uninoms Spreering Spreering Spreering Subsurface Subsurface Table 7-2 provides a sum any of the capital cost estimates, in current (June 2023) dollars. Detailed Subsurface Dip fingation Spreering Spreering Spreering Piper found Dip fingation Spreering Spreering Spreering Piper found Dip fingation Spreering Spreering Spreering Piper fingation Subsurface Disposed Spreering Spreering Piper fingation Spreering Spreering Spreering Spreering Piper financi (Spreering Spreering Spreering Spreering Spreering Piper found Advierter estimate Streering Spreering Spreering Piper found Streering Streering Spreering Spreering Piper found Streering Streering Streering Spreering Piper found Streering Streering Streering Streering Piper found Streering Stre	Tank Gravel Filter	Electrical and	l instrumentation markup	25 per	ent
Ententing contingent Detent Equalization Tank 20 prenet Equalization Tank 20 prenet Equalization Tank 20 prenet Disposition 20 prenet Subsurface Disposition Disposition 20 prenet Prevente 20 prenet Subsurface 20 prenet Disposition 20 prenet]	Engineering, a	administration, and legal markup	25 per	ent
Calle 7-2 provides a summary of the capital cost estimates, in current (June 2023) dollars. Detailed estimates DisposalSubsurtace DisposalOriginationDisposalFigure 7-3. Project Alternative 3:DisposalDisposalDisposalColspan="2">Table 7-2. Capital cost estimatesDisposalDisposalFigure 7-3. Project Alternative 3:DisposalDisposalDisposalColspan="2">Alternative 3:OriginationState of colspan="2">Alternative 3:OriginationState of colspan="2">Alternative 3:OriginationState of colspan="2">Alternative 3:OriginationState of colspan="2">Alternative 3:OriginationState of colspan="2">State of colspan="2">State of colspan="2">State of colspan="2">State of colspan="2"Figure 7:3: Project Alternative 3:Figure 7:3: Project Alternative 3:Figure 7:3: Project Alternative 3:State of colspan="2"Alternative 3:Figure 6:2: Project Alternative 3:State of colspan="2"State of colspan="2"State of colspan="2"State of colspan="2"State of colspan=	Irrigation	Estimating co	ontingency for unknowns	20 per	ent
Drip fingation Table 7.2. Capital Cost Stimates Stimaty Drip fingation Drip fingation Attentive 1: Attentive 2: Attentive 3: Disposal Pescription Attentive 1: Attentive 3: Attentive 3: Attentive 3: Figure 7-3. Project Alternative 3: Imboff Tank/Recirculating Gravel Filter Description \$40.2 million \$40.2 million </td <td>Equalization Tank</td> <td>Table 7-2 provides a sur estimates can be found</td> <td>mmary of the capital cost estir in Appendix A.</td> <td>nates, in current (June)</td> <td>2023) dollars. Detailed</td>	Equalization Tank	Table 7-2 provides a sur estimates can be found	mmary of the capital cost estir in Appendix A.	nates, in current (June)	2023) dollars. Detailed
Figure 7-3. Project Alternative 3: Inhoit Tank, Reinative 3: Memative 3:	Drip Irrigation		Table 7-2. Capital Cost Es	timates Summary	
Figure 7-3. Project Alternative 3: Imboff Tank/Recirculating Gravel Filter Collection system \$40.2 million \$40.2 million \$40.2 million Drainage channel improvement \$13.9 million \$13.9 million \$13.9 million \$13.9 million Wastewater freatment \$2.0.7 million \$20.3 million \$29.6 million \$29.6 million Matewater freatment \$2.0.7 million \$20.3 million \$29.6 million \$29.6 million Effluent disposal \$4.0 million \$20.3 million \$29.6 million \$29.6 million Anton \$2.0.7 million \$2.0.7 million \$20.3 million \$29.6 million Provention \$4.0 million \$4.0 million \$20.3 million \$20.6 million Provention \$4.0 million \$4.0 million \$4.0 million \$4.0 million Provention \$4.0 million \$4.0 million \$4.0 million \$4.0 million Anton \$4.0 million \$4.0 million \$4.0 million \$4.0 million Anton \$4.0 million \$4.0 million \$4.0 million \$4.0 million	recorder	Description	Alternative 1: Activated Sludge Package Plants	Alternative 2: MBR Package Plants	Alternative 3: Imhoff Tank/RGF
Drainage channel improvement\$19.9 million\$19.9 million\$19.9 millionWastewater treatment\$20.7 million\$20.3 million\$29.6 millionEffluent disposal\$4.0 million\$4.0 million\$4.0 millionTodal\$4.0 million\$4.0 million\$4.0 millionMacter disposal\$4.0 million\$4.0 million\$4.0 millionACCE class 5 estimate range\$4.2.4.5105. million\$42.4.5105. million\$48.8.8.100ACCE class 5 estimate range\$4.2.4.5105. million\$42.4.5187. million\$48.8.8.100	Figure 7-3. Project Alternative 3: Imhoff Tank/Recirculating Gravel Filter	Collection system	\$40.2 million	\$40.2 million	\$40.2 million
Wastewater treatment \$20.7 million \$20.3 million \$29.6 million Effluent disposal \$4.0 million \$4.0 million \$4.0 million Totals \$4.0 million \$4.0 million \$4.0 million ACC Class 5 estimate range \$4.2.4.5109.5 million \$4.2.4.5109.5 million \$42.8.8.6 million		Drainage channel improvement	\$19.9 million	\$19.9 million	\$19.9 million
Effluent disposal \$4.0 million \$4.0 million \$4.0 million Totals \$8.4.7 million \$4.3 million \$4.0 million ACCE class 5 estimate range \$8.4.7 million \$4.3.8.1 million \$43.8.1 million		Wastewater treatment	\$20.7 million	\$20.3 million	\$29.6 million
Totals \$84.7 million \$84.3 million \$83.6 million AACE Class 5 estimate range \$42.4-5169.5 million \$42.2-\$168.6 million \$46.8-\$187.2 million		Effluent disposal	\$4.0 million	\$4.0 million	\$4.0 million
AACE Class 5 estimate range \$42.4–\$169.5 million \$42.2–\$168.6 million \$46.8–\$187.2 million		Totals	\$84.7 million	\$84.3 million	\$93.6 million
		AACE Class 5 estimate range	\$42.4-\$169.5 million	\$42.2-\$168.6 million	\$46.8-\$187.2 million

7.1.3 Project Alternative 3: Imhoff Tank/Recin

Project Alternative 3 incorporates recirculating gravel filte a schematic of the project alternative. An Imhoff tank wou headworks to remove grease and settleable solids prior to Recirculation pumps would distribute water from the reci-gravel filter that provides secondary treatment. Water coll back to the recirculation tank. On average water would fit disposal in the subsurface drip irrigation system as previc

Section 7



Brown And Caldwell

7.2.2 Operation and Maintenance Costs

0&M costs estimates were developed for the three alternatives. The 0&M cost estimates include collection system maintenance, plus estimates of labor, electricity consumption, chemicals, maintenance materials and solids disposal for the WWTP. 0&M assumptions are listed in Table 7-3. The 0&M estimates are based on the WWTP average dry weather flow capacity.

Table 7-3. (J&M Cost Assumptions
Description	Value
Average dry weather flow	125,000 gpd
Labor cost, loaded	\$100,000/year/full time equivalent
Electricity cost	\$0.45/kWh
Landfill tip fee	\$116/wet ton
Maintenance materials	2 percent of equipment capital cost/year

The O&M estimates for the three project alternatives are summarized in Table 7-4. Details can be found in Appendix A. Per Table 7-4, Project Alternative 3: Imhoff Tank/Recirculating Gravel Filter incurs the lowest O&M cost, while Project Alternative 2: MBR Package Plants incurs the highest.

		Project Alternative 3: Imhoff Tank/RGF	\$74,000	\$300,000	\$120,000	\$27,000	\$84,000	\$67,000	\$672,000
t Estimate Summary	Annual Cost	Project Alternative 2: MBR Package Plants	\$74,000	\$300,000	\$300,000	\$32,000	\$113,000	\$67,000	\$886,000
Table 7-4. 0&M Cost		Project Alternative 1: Activated Sludge Package Plants	\$74,000	\$300,000	\$270,000	\$27,000	\$117,000	\$67,000	\$855,000
		Description	Collection system	Labor	Electricity	Chemicals	Maintenance materials	Solids disposal	Totals

7.2.3 Life-Cycle Costs

An economic evaluation was prepared to assess the potential life-cycle costs associated with each project alternative. The economic evaluation consists of a net present value (NPV) comparison. The NPV analysis includes capital, O&M, and equipment replacement costs. An appropriate inflationary factor and discount rate are applied to obtain the NPV over a 30-year planning period.

The NPV of an alternative represents the amount of money that would need to be set aside today (at a given interest rate) to pay the costs associated with the alternative over the entire planning period. The alternative with the lowest NPV is considered the most attractive from an economic perspective. The evaluation results are included in Appendix A.

Naalehu WWTP Revised PER

Section 7

Table 7-5 summarizes the life-cycle cost evaluation assumptions.

De D	able 7-5. Life-Cycle Economic Assumptions	Description Value	ysis 2023	riod, years 30	e, percent 3.5	te, percent 5.0	replacement cycle, years 20	replacement cycle, years 15
--	---	-------------------	-----------	----------------	----------------	-----------------	-----------------------------	-----------------------------

Table 7-6 summarizes the results of the life-cycle cost analysis.

	Table 7-6. Life-Cycle Cost Ana	lysis Summary	
Description	Alternative 1: Activated Sludge Package Plants	Alternative 2: MBR Package Plants	Alternative 3: Imhoff Tank/RGF
Capital cost	\$84.8 million	\$84.3 million	\$93.6 million
Annual O&M cost	\$855,000	\$886,000	\$672,000
Equipment replacement cost excluding membranes)	\$6.3 million	\$6.0 million	\$4.6 million
Membrane replacement cost	N/A	\$60,000	N/A
.ife-cycle cost	\$109.6 million	\$109.8 million	\$112.9 million
Comparison to lowest cost alternative	%0	+0.2%	+3.0%

Figure 7-4 shows the results in graphical form. As shown in the table and graph, the three project alternatives have similar capital costs. Alternative 1: Activated Sludge Package Plants and Alternative 2: MBR Package Plants incur nearly equivalent life-cycle costs due to the similarities in design and package plant costs. Project Alternative 3: Inoff Tank/Recirculating Gravel Filter incurs the greatest life-cycle costs. Iargely due to the high capital cost associated with the Imhoff incurs the greatest life-cycle costs. Iargely due to the high capital cost associated with the Imhoff considered to have similar life-cycle costs.







Figure 7-4. Life-Cycle Cost Evaluation Results

Capital Cost
 O&M and Replacement Cost

7.3 Non-Economic Evaluation

A non-economic evaluation was conducted to provide a qualitative comparison between the three alternatives.

7.3.1 Approach

Non-economic evaluations are generally subjective by necessity. Quantifiable measurements are used when available, but lack of information or difficulty and expense of obtaining information requires subjective assessments. The project alternatives were scored in relation to one another and according to an evaluation matrix, described below. Each alternative was scored from 1 to 5 (5 = high/desirable, 1 = low/less desirable) for each evaluation criteria. The alternatives were not ranked in the scoring; alternatives could receive the same scores for any given criteria. The evaluation criteria were weighted to reflect their overall significance for the project. The scores were multiplied by the criteria weights to develop a non-economic score for each alternative.

Naalehu WWTP Revised PER

Section 7

7.4 Non-Economic Evaluation Criteria

Table 7-7 shows the non-economic criteria chosen for comparing the three alternatives.

	Table 7-7	Non-Economic Comparison Criteria
Category	Criteria	Description
	Effluent quality	The quality of the effluent produced with respect to BODs, TSS, nutrients, and turbidity.
	Potential for capacity expansion	Ability of the system to be expanded should additional capacity be required.
Level of Service Measures	Water recycling feasibility	The relative extent of modifications that would be needed to create R-1 recycled water to support a future water recycling program.
	Public perception/community concerns	The community's impression of the project and the perceived support.
	Monitoring complexity	The relative difficulty of monitoring tasks required for the option chosen.
-	Treatment adjustment potential	The ability to increase treatment to comply with future permit requirements and/or growth.
Kegulatory	Safety regulations complexity	The relative difficulty to comply with safety regulations including staff training, reporting, maintenance procedures.
	Environmental concerns	The extent of the project's potential environmental impacts should failures occur.
	Footprint	The physical space that the processes will occupy (affecting land acquisition, subdivision, and permitting).
0&M Factors	Safe work environment	The relative health and safety risk operation of given option will have on the employees: includes equipment access, chemical hazards, confined spaces, dust, the extent of measures required to ensure the health and safety of the employees.
	Maintenance complexity	The relative intensity of equipment maintenance requirements.
	Operations complexity	The relative intensity of the operations requirements.
	Mainland delivery dependence	The relative dependence on regular deliveries of equipment, supplies, or spare parts from mainland sources.
	Mainland servicing dependence	The relative degree to which technology will require special servicing by mainland- based personnel.
ISIANG FACTORS	Power dependence	The relative degree to which the treatment processes depend on electrical power for operation.
	Chemical dependence	The relative dependence on chemical supplies, whether locally available or restricted by mainland delivery schedules and requirements.

The categories and criteria were developed using best engineering judgment and our understanding of the project, and the County of Hawaii Department of Environmental Management goals and concerns.





The weighting factors used for the non-economic comparison are listed in Table 7-8.

Catedory	Criteria
cureBond	2000
	Effluent quality (30%)
	 Potential for capacity expansion (20%)
Level of Service Measures	 Water recycling feasibility (30%)
101.0-2	 Public perception/community concerns (20%)
	Category Total (100%)
	Monitoring complexity (25%)
- 4	 Treatment adjustment potential (25%)
Kegulatory	 Safety regulations complexity (25%)
(or 0-2)	 Environmental concerns (25%)
	Category Total (100%)
	Footprint (30%)
L	 Safe work environment (25%)
Uwner Factors	 Maintenance complexity (25%)
(or n=1)	 Operations complexity (20%)
	Category Total (100%)
	 Mainland delivery dependence (25%)
	 Mainland servicing dependence (25%)
(25%)	 Power dependence (25%)
(or ~~)	 Chemical dependence (25%)
	Category Total (100%)
Overall Total 100%	

7.5 Non-Economic Evaluation Results

A score of 1 through 5 was given for each criterion, with 5 being the most favorable, and 1 representing the least desired option. The complete non-economic evaluation is included as Appendix 0. The non-economic evaluation results are summarized in Table 7-9.

	Rank	7	1	e
ighted Scores	Score	3.80	3.96	3.53
Table 7-9. Non-Economic Wei	Alternative	Project Alternative 1: Activated Sludge Package Plants	Project Alternative 2: MBR Package Plants	Project Alternative 3: Imhoff Tank/Recirculating Gravel Filter

As shown in Table 7-9, Project Alternative 2: MBR Package Plants, received the highest non-economic score. The higher score reflects the County's desire to standardize on MBR technology to provide the highest level of treatment at WWTP facilities and to facilitate future water recycling programs.

Section 7

Naalehu WWTP Revised PER

7.6 Conclusions and Recommendation

Figure 7-5 combines the economic and non-economic results into a single graph. As previously stated, the economic cost of the three project alternatives can be considered equivalent at this level of analysis. Project Alternative 2: MBR Package Plants, has the highest non-economic score and is recommended for implementation if the County proceeds with a centralized sewer system and WWTP for the community.



Alternative 1: Activated Sludge Package Plants

 Alternative 3: Imhoff Tank/Recirculating Filter ▲ Alternative 2: MBR Package Plants

Figure 7-5. Combined Economic and Non-Economic Results



7-10



Section 8

Preliminary Design of Improvements

This section presents an overview of preliminary design improvements proposed for the new Naalehu wastewater treatment plant (NWVTP) project.

8.1 Site Plan

Figure 8-1 provides a preliminary site plan of the WWTP project.

8.2 Process Schematic

Figure 8-2 provides a preliminary process schematic of the WWTP.





and the second s	FIGURE 8-2
	NAALEHU WASTEWATER TREATMENT PLANT PROCESS SCHEMATIC
and a second sec	Brown MD SCALE: NONE JOB NO: 153740

Patr Criteprid

WA 26, 2023 917 MM Cold Leer Richard Selana

Naalehu WWTP Revised PER

8.3 Preliminary Design Criteria Table 8-1 lists the preliminary design criteria for the proposed WWTP.

	lena
Description	Value
Influent Flow	
Average dry weather, gallons per day (gpd)	125,000
Peak day wet weather, gpd	322,000
Peak hour wet weather, gpd (gallons per minute [gpm])	457,000 (317)
Influent Characteristics	
Biochemical oxygen demand (BODs), mg/L	300
Total suspended solids (TSS), mg/L	300
Total Nitrogen (TN), mg/L	40
Odor Control-Granular Activated Carbon	
Airflow rate, air changes per hour	6
H ₂ S inlet concentration, parts per million	1-10
H ₂ S removal efficiency, percent	66
Media, type	High-capacity carbon
Mechanical Screens	
Number of units, each	1
Type	In-channel cylindrical
Screen opening size, inches/(millimeters)	0.125 / (3)
Maximum flow rate capacity, gpm	Greater than 480
Screening washing	Integral
Screening compaction	Integral
Bypass Screen	
Type	Manually-cleaned bar rack
Bar spacing, inches	1
Rake	Fabricated to Interlock with bars
Screenings Receptacle	
Type	55-gallon drum or bags
Screenings volume per million gallons (Mgal) treated, ft ³ /Mgal	5
Estimated screenings quantity, ft3/day	0.63
Disposal frequency, per week	1
Influent Flow Metering	
Type	Parshall flume
Maximum flow capacity, gpm	Greater than 480
Minimum straight upstream channel section	20 times the throat width
Influent flow sampling	Refrigerated automatic composite sampler

Brown Ann Caldwell

Table 8-1, Preliminary Design C	
Description	Value
Grit Chamber	
Number of units, each	1
Type	Aerated grit trap
Volume, gallons	4,800
Detention time, minutes at Peak Hour Wet Weather Flow	10
Air supply, ft³/minute	06
Removal	Vactor truck
Estimated average grit quantity, $\mathrm{ft}^3/\mathrm{day}$	1.4
MBR Package Plant	
Number of packaged biological treatment trains, each	2
Flow basis for biological design, gpd each	62,500
Anoxic tank working volume (excluding membranes), gallons each	2,000
Aerobic working volume, gallons each	10,000
Design SRT, days	7
Waste sludge removal, gpd each	1,500
Design mixed liquor suspended solids concentration in bioreactor, mg/L	≤ 8,000
Number of duty membrane blowers, per train	1
Number of duty process aeration blowers, per train	1
Aeration system, type	Coarse bubble diffused aeration
Mixed liquor recirculation rate, x Average Dry Weather Flow	4
Membrane cleaning dosing systems, types	Sodium hypochlorite, citric acid, and coagulant
Sludge Management System	
Number of units, each	1
Type	Incline screw press
Screw press capacity, gpm	45
Polymer dose, lbs/dry ton	20
Annual polymer use, Ibs	621
Average amount of dewatered sludge, wet tons/day	0.71
Disposal frequency, per week	1
Maintenance Disinfection system	
Type	Chlorine
Form	Calcium hypochlorite tablets
Design chlorine dose, mg/L	8
Irrigation Equalization (Control) Tank	
Number of units, each	1
Type	Glass lined bolted steel
Volume, gallons	20,000
Drip flow, gpm	112
Effluent Flow Metering	
Type	Magnetic
Effluent flow sampler	Refrigerated automatic composite

Section 8

How meter(s) and pressure indicators Subsurface drip irrigation disposal 2 (1 active, 1 redundant) Non-clog subsurface drip emitters Value 10-year, 1-hour storm Existing vegetation 26,833 53,667 56,630 112 1 2.50 <5
<55
<510
<0.2</pre> 5.2 6-9 Table 8-1. Preliminary Design Criteria 4 4.2 4 Description Design application rate, % of percolation rate Turbidity, nephelometric turbidity units Flow rate per zone, gallons per minute Drip rate per emitter, gallons per hour Design percolation rate, inches/day Total land application area, acres Total number of emitters, each Stormwater site management Irrigation system monitoring Effluent Management System Number of irrigation zones Total length of drip line, ft Depth of drip line, inches Number of units. each Area per zone, sq. ft. Drip line spacing, ft Distribution system Emitter spacing, ft BOD5, mg/L TSS, mg/L Effluent Quality Vegetation TN, mg/L Type

8.4 Preliminary Floor Plan

Figure 8-3 provides a preliminary floor plan for the proposed Operations Building.

Brown And Caldwell



Section 8

Naalehu WWTP Revised PER



Section 9

Implementation Plan

This section describes the proposed implementation plan for the Naalehu wastewater treatment plant (WWTP).

9.1 Implementation Approach

The WWTP and collection system projects could be implemented using either a traditional design/bid/build (DB) approach or a design/build (DB) approach, as discussed below.

9.1.1 Design Bid Build Approach

DBB is the traditional approach used by the County for implementing public works projects. The design is prepared by a consultant, and then bids are solicited from construction contractors. The County awards the contract to the lowest responsible bidder.

Advantages of the DBB approach are that the County retains maximum control over the design process, ensuring the project will meet its needs.

9.1.2 Design Build Approach

approach. The County would need to use a procurement process based on qualifications and cost to select the DB entity. The typical DB procurement process takes 9–12 months to complete. The DB bidders will need the County to complete the following prior to the DB procurement process: design and construct a facility that meets established project specifications. The combined WWTP and collection system projects are large enough monetarily for the County to consider a DB DB is an alternative delivery approach whereby the County would contract with an entity to both

- Complete geotechnical report
- Environmental assessment
- Land use entitlements
 - WWTP land purchase
- Advantages of DB implementation are:
 - - Possibility of reduced overall costs
- Design and construction can occur simultaneously, potentially reducing implementation time
 - DB entity assumes the performance liability for the project, as defined in the project specifications

Disadvantages of the DB approach are that the County has limited experience with it, and the County would not have as much control over how the project is designed


Section 9

9.2 Implementation Schedules

Planning level implementation schedules were developed for both approaches.

9.2.1 Recent Change in State of Hawaii Land Use Commission Policy

The State of Hawaii Land Use Commission (LUC) recently changed its policy regarding the use of Special Permits for non-conforming uses. The proposed WWTP site is located in the Agricultural District as defined by the LUC. A WWTP is not an allowable land use in the Agricultural District. In the past the LUC has allowed Special Permits to be used for non-conforming uses. However, in response to litigation the LUC has recently changed its policy and now recommends that project proponents for permanent facilities (like a WWTP) pursue a District Boundary Amendment (DBA) from the LUC. The LUCs rationale is that permanent entitlement (i.e., a DBA) is more appropriate for a permanent facility like a WWTP, rather than a temporary entitlement like a Special Permit. Since the WWTP aprocel is less that 15 acrea the DBA can be processed by the County of Hawaii. However, the action will likely take longer to implement than a Special Permit.

9.2.2 Equipment Procurement Time to Impact Construction Schedule

The COVID-19 pandemic continues to impact the construction industry due to increased time to deliver equipment and other materials. Most significantly, the time for the MBR package plant supplier to manufacture their equipment was quoted at 55 weeks instead of a typical pre-pandemic time of approximately 26 weeks. Similar delays are being experienced on other construction projects in Hawaii, and it is reasonable to assume that other equipment suppliers will quote extended supply in Hawaii, and it is reasonable to assume that other equipment suppliers will quote extended supply expectation.

9.2.3 Implementation Schedules

Figure 9-1 presents implementation schedules for both approaches. At this time, both approaches may not enable the County to meet the Administrative Order on Consent (AOC) milestone schedule to close the LCCs. The DB approach offers greater potential to meet the milestone, because equipment procurement can possibly occur in parallel with design within a DB contract.

9.3 Recommendation

Given the Revised AOC deadline to close the LCCs, and the equipment procurement time impact to the construction schedule. a DB approach may offer a better opportunity to meet the deadline, because a DB entity could initiate equipment procurement while design activities progress. If the U.S. Environmental Protection Agency was able to grant a time extension, then either a DBB or DB approach can be taken. A traditional DBB approach would give the County greater control over the project outcome, and is the County's standard method for implementing projects.

Naalehu WWTP Revised PER

Section 9



Brown and Caldwell



Naalehu WWTP Revised PER

Section 10

References

- Brown and Caldwell (November 2019). Feasibility Study: Naalehu WWTP R-1 Stock Water, Prepared for County of Hawaii Department of Environmental Management Wastewater Division.
- Crites, Ron, Craig Lekven, Steve Wert, George Tchobanoglous (1997). A Decentralized Wastewater System for a Small Residential Development in California. The Small Flows Journal. National Small Flows Clearinghouse. Volume 3, Issue 1, Winter 1997.
- Crites, P. W. and G. Tchobanoglous (1998) Small and Decentralized Wastewater Management Systems, McGraw-Hill, New York.
- Department of Wastewater Management, City and County of Honolulu, State of Hawaii. Design Standards of the Department of Wastewater Management, Volume 1 and 2. July 2017.
- Great Lakes-Upper Mississippi River Board of State and Provincial Public Health and Environmental Managers. Recommended Standards for Wastewater Facilities. 1997.
- Hawaii Administrative Rules, Title 11, Department of Health Administrative Rules.
- Masa Fujioka & Associates. Letter Report, Probing for Large Cavities (Lava Tubes), Naalehu and Pahala Large Capacity Cesspool Sewerage System. January 9, 2007.
- M&E Pacific, Inc. Kau Sewer System Evaluation, Kau, Island of Hawaii, Hawaii. December 2004.
- SSFM International. Inc. Final Preliminary Engineering Report for Naalehu and Pahala Large Capacity Cesspool Conversion Projects, July 2007.
- USEPA. *Process Design Manual, Land Treatment of Municipal Wastewater Effluents". EPA/625/R-06/016. September 2006.
- Water Environment Federation. Wastewater Disinfection, Manual of Practice FD-10. Water Environment Federation, 1996.

Appendix A: Cost Estimates





Naalehu WWTP Revised Preliminary Engineering Repo	Alternative Solutions Cost Summary
--	------------------------------------

Alternative #1 - RAS Package Plants / Subsurface Drip

+-

	Collection Sy Drainage Ch Wastewater Treat Effluent Dis	stem TOTAL annel TOTAL ment TOTAL posal TOTAL	\$40,165,000 \$19,865,000 \$20,705,000 \$4,016,000
ALTERI	VATIVE #1 CAPITAL	COST TOTAL	\$84.751,000
ANNUAL O&M COSTS			
Collec	tion system	\$74,000	
	Labor	\$300,000	
	Electricity	\$270,000	
	Chemicals	\$27,000	
Maintenan	ce materials	\$117,000	
So	lids disposal	\$67,000	
Total Annual Oper	ating Costs	\$855,000	
EQUIPMENT REPLACEMENT COST (20 YEAR)		\$6,238,000	

EQUIPMENT REPLACEMENT COST (20 YEAR)	\$6,238,000	
Alternative #2 - MBR Package Plants / Reuse /	Subsurface Dr	đ
Collection) System TOTAL Channel TOTAL	\$40,165,000 \$19,865,000
Wastewater T Effluent	eatment TOTAL Disposal TOTAL	\$20,283,000 \$4,016,000
ALTERNATIVE #2 CAPIT ANNUAL O&M COSTS	AL COST TOTAL	\$84,329,000
Collection system	\$74,000	
Labor	\$300,000	
Electricity	\$300,000	
Chemicals	\$32,000	
Maintenance materials	\$113,000	
Solids disposal	\$67,000	
Total Annual Operating Costs	\$886,000	
MEMBRANE REPLACEMENT COST (15 YEAR)	\$60,000	
EQUIPMENT REPLACEMENT COST (20 YEAR)	\$6,005,000	

/ Subsurface Drip
/ RGF ,
Alternative #3 - Imhoff Tank ,

	Collection Drainage C Wastewater Tre Effluent D	System TOTAL Channel TOTAL atment TOTAL Nisposal TOTAL	\$40,165,000 \$19,865,000 \$29,556,000 \$4,016,000
STSOD W&O IVIINN	ALTERNATIVE #3 CAPITA	IL COST TOTAL	\$93,602,000
	Collection system	\$74,000	
	Labor	\$300,000	
	Electricity	\$120,000	
	Chemicals	\$27,000	
	Maintenance materials	\$84,000	
	Solids disposal	\$67,000	
	Total Annual Operating Costs	\$672,000	

\$4,574,000

EQUIPMENT REPLACEMENT COST (20 YEAR)

Naa Revised Prelimir Unit C	ehu WWTP ary Engineering sst Estimates	Report
Electrical & Instrumentation	25.0%	
Engineering, Admin, & Legal	25.0%	
Contingency	20.0%	
ENR CCI	13,345.00	June, 2023
Naalehu WWTP Capital Unit Costs	Units	Unit Cost
Environmental protection, BMPs	ac	\$11,000
Site clearing	ac	\$20,000
Site roads, guard rails, pavement	ac	\$92,000
Perimeter fence	з	\$150
WWTP site grading	ac	\$30,000
WWTP site drainage improvements	ac	\$18,000
Drainage channel improvements	LS	\$13,700,000
Plant water catchment/collection system	LS	\$75,000
Process yard piping	LS	\$250,000
Headworks (includes site/civil, structures, equipment & piping)	SJ	\$1,024,000
Influent pump station	LS	\$1,670,000
Chlorine disinfection	LS	\$150,000
RAS package plants	LS	\$3,812,000
MBR package plants	LS	\$3,579,000
Irrigation equalization tank	gal	\$10
Effluent pump station	LS	\$1,250,000
Subsurface drip irrigation line	ιF	\$10
Irrigation piping & valves	LF	\$250
Imhoff tank	LS	\$1,184,000
Recirculation tank	LS	\$1,320,000
Recirculating gravel filter	LS	\$6,591,000
Plant drainage system	ac	\$40,000
Main generator (including process piping)	LS	\$500,000
Maintenance/operations/electrical building	SF	\$1,000
Phase 1 existing gravity collection system w/ new WWPS & force main	LS	\$17,650,000
Reuse existing Brewer area gravity collection system	LS	\$3,700,000
Phase 2 new gravity collection system w/ new WWPS & force main	LS	\$27,700,000

-				
Cost	Number of Units	Unit Cost	Units	Recirculation Tank
\$6,591,000	TOTAL			
\$140,000	1,400	\$100	С	6 in sand media under liner
\$1,575,000	10,500	\$150	ς	Gravel media
\$16,000	200	\$80	Ч	8-inch PVC recirculation pipe
\$222,000	3,700	\$60	Ч	6-inch PVC drainage pipe
\$417,000	13,900	\$30	Ч	3-inch PVC lateral pipe
\$80,000	400	\$200	Ц	20-inch PVC manifold pipe
\$600,800	75,100	\$\$	SF	Bed liner
\$3,540,000	11,800	\$300	Ç	RGF bed excavation
Cast	Number of Units	Unit Cost	Units	Recirculating Gravel Filter
1				
\$1,184,000	TOTAL			
\$240,000	3,000	\$80	SF	Epoxy Coating
\$500,000	1	\$500,000	rs	Odor control
\$15,000	75	\$200	SF	Cover plates
\$50,000	1	\$50,000	LS	Piping & valves
\$240,000	160	\$1,500	ç	Concrete
\$3,500	35	\$100	ς	Bedding & backfill
\$135,000	450	\$300	<u></u>	Excavation

1,480 \$444,000 200 \$20,000 290 \$435,000 210 \$21,000 4 \$41,000 4 \$1,320,000

 CY
 S300

 CY
 5100

 CY
 51,500

 LF
 51,000

 ea
 \$100,000

Recirculation tank excavation Bedding & backfill Concrete Handrail Pumps & valves

		Cost	
P eering Report	mates	Number of Units	
laalehu WWT minary Engine	Sum Cost Esti	Unit Cost	
N vised Preli	Lump	Units	
Re			

Imhoff Tank

Naalehu WWTP Operation, Maintenance, & Replacement Unit Costs	Unit Cost	Units
Equipment replacement cost	25%	of process capital cost
Package plant replacement cost	100%	of package plant captial cost
Maintenance materials	2%	of equipment capital cost
Gravity mainline collection system maintenance cost	\$16,000.00	per mi
Force main collection system maintenance cost	\$10,000.00	per mi
Membrane replacement cost	\$1,950.00	per module + S&H & install
Solids disposal dumpster rental fee	\$500.00	per week
Sanitary landfill tipping fee	\$116.00	per wet ton
Hypochlorite tablet cost	\$8.00	per lb
Dewatering polymer cost	\$3.00	per lb
Diesel price	\$6.06	per gallon

Naalehu WWTP Revised Preliminary Engineering Report Unit Cost Estimates Naalehu WWTP Revised Preliminary Engineering Report Cost Estimate

Alternative #1 - RAS Package Plants / Subsurface Drip

Capital (Cost Estimat	9		
Naalehu WWTP Capital Cost Item Description	Units	General Unit Cost	Number of Units	COST
Collection System				
Phase 2 new gravity collection system w/ new WWPS & force main	SI	\$27,700,000		\$27,700,000
			Subtotal	\$27,700,000
			Contingency @ 20%	\$5,540,000
		Engineering, Ad	min, & Legal @ 25%	\$6,925,000
		Col	lection System Total	\$40,165,000
Drainage Channel				
Drainage channel improvements	LS	\$13,700,000	1	\$13,700,000
	-		Subtotal	\$13,700,000
		Enaineerina. Ad	Contingency @ 20% min_& Lead @ 25%	\$2,740,000 \$3.425.000
		Dra	inage Channel Total	\$19,865,000
Wastewater Treatment				
Environmental protection, BMPs	ac	\$11,000	1.5	\$16,500
Site clearing	ac	\$20,000	1.5	\$30,000
Site roads, guard rails, pavement	ac	\$92,000	1.5	\$138,000
Perimeter fence	F	\$150	4,100	\$615,000
WWTP site grading	ac	\$30,000	1.5	\$45,000
WWTP site drainage improvements	ac	\$18,000	1.5	\$27,000
Plant water catchment/collection system	SI	\$75,000	-1	\$75,000
Process yard piping	S :	\$250,000		\$250,000
Headworks (includes site/civil, structures, equipment & piping)	S .	\$1,024,000	-	51,024,000
intiuent pump station Chlorino dirinfontion	2 -	000'0/9'T¢	-	61E0.000
Difference distribution	3 2	000/0016	-	000,00040
NAS package piants Plant drainage system	3 %	000/770/55	т Т	000/710/06
Main generator (including process piping)	s s	\$500.000		\$500.000
Maintenance/operations/electrical building	SF	\$1,000	2,150	\$2,150,000
Sludge dewatering system	LS	\$860,000	1	\$860,000
			Subtotal	\$11,422,500
		Electrical & Inst	rumentation @ 25% Subtotal	000,025,25
			Contingency @ 20%	\$2.856.000
		Engineering, Ad	min, & Legal @ 25%	\$3,570,000
		Wastewa	iter Treatment Total	\$20,705,000
Effluent Disposal				
Irrigation equalization tank	leg	\$10	20,000	\$200,000
Effluent pump station	LS	\$1,250,000	1	\$1,250,000
Subsurface drip irrigation line	LF	\$10	54,000	\$540,000
Irrigation piping & valves	Ľ	\$250	006	\$225,000
		1	Subtotal	\$2,215,000
		Electrical & Inst	rumentation @ 25% Subtotal	52.769.000
			Contingency @ 20%	\$554,000
		Engineering, Ad	min, & Legal @ 25% Tuent Disposal Total	\$693,000 \$4 016 000
		17		
			IVIOL 14 oldsteines	CON 7E1 000

300 HP diesel dump truck	รา	\$300,000	1	\$300,000
Dewatering screw press	เ	\$300,000	1	\$300,000
Incline screw conveyor	LS.	\$80,000	4	\$80,000
Polymer system	ป	\$80,000	T	\$80,000
Sludge feed pump & piping	SJ	\$100,000	T	\$100,000
			TOTAL	\$860,000
Reuse Existing Gravity Collection System	Units	Unit Cost	Number of Units	Cost
Inspection & cleaning	LS	\$1,500,000	1	\$1,500,000
Repair defects	SJ	\$2,000,000	1	\$2,000,000
Archaeological monitoring	SJ	\$50,000	1	\$50,000
BMPs	SJ	\$50,000	1	\$50,000
Traffic control measures	LS	\$50,000	1	\$50,000
Pre- & post-construction inspections & documentation	LS	\$50,000	1	\$50,000
			TOTAL	\$3,700,000
New Drainage Channel	Units	Unit Cost	Number of Units	Cost
Earthwork	ς	\$250	49500	\$12,375,000
Concrete culverts	SJ	\$1,250,000	1	\$1,250,000
Archaeological monitoring	SJ	\$25,000	1	\$25,000
Clear and grub	LS	\$25,000	1	\$25,000
BMPs	LS L	\$25,000	7	\$25,000

\$25,000 \$25,000 \$25,000 TOTAL \$13,700,000

Units Unit Cost Number of Units Cost

Sludge Dewatering System

ed Preliminary Engineering Report	Lump Sum Cost Estimates
Revised F	Lu

Naalehu WWTP

Naalehu WWTP Revised Preliminary Engineering Report Cost Estimate

Alternative #2 - MBR Package Plants / Reuse / Subsurface Drip

Naalehu WWTP Capital Cost Item Description	Units	General Unit Cost	Number of Units	COST
Collection System				
Phase 2 new gravity collection system w/ new WWPS & force main	LS	\$27,700,000	1	\$27,700,000
			Subtotal	\$27,700,00
			Contingency @ 20%	\$5,540,00
		Enaineerina. Ao	lmin. & Leaal @ 25%	\$6.925.00
		Co	lection System Total	\$40,165,00
Drainage Channel				
Drainage channel improvements	LS I	\$13,700,000	1	\$13,700,00
-			Subtrated	\$13,700,00
			contingency @ 20%	72,740,00
		Engineering, Aa Dro	lmin, & Legal @ 25% iinage Channel Total	\$3,425,00 \$19,865,00
Nactemater Treatment				
invironmental notertion RMPs	ar	\$11 DDC	15	¢16 50
site clearing	- Le	\$20,000		00025
site roads guard rails navement	- Se	\$42,000	1.0	\$138.00
arc round barra rand, parament Darimatar fance	3 4	\$150	4 100	\$615,00
WWTD site grading	j a	200 065	15	\$45.00
WWTP site drainage improvements	e e	\$18,000	15	00/225
lant water catchment/collection system	2	\$75,000		\$75,00
Process vard bibing	2	\$250.000		\$250.00
Haadworks fincludes site/sixil_structures_equipment & nining)	2			¢1 07 00
redaworks (intelades site/civit, structures, equipment & piping)	2 5	51,024,000 61 670 000		00,420,14
niuent pump station	2 5		-	00'0/0'T¢
	2			nn'nsts
VIBR package plants	L5	\$3,579,000	1	\$3,579,000
Plant drainage system	ac	\$40,000	1.5	\$60,000
Vlain generator (including process piping)	LS	\$500,000	1	\$500,000
Vlaintenance/operations/electrical building	SF	\$1,000	2,150	\$2,150,000
sludge dewatering system	LS	\$860,000	1	\$860,00
			Subtotal	\$11,189,50
		Electrical & Insi	trumentation @ 25%	100,867,24
			Subtotal	00,888,614
			Contingency @ 20%	\$2,798,000
		Engineering, Aa	lmin, & Legal @ 25%	\$3,497,000
		Wastewa	ater Treatment Total	\$20,283,000
Effluent Disposal				
rrigation equalization tank	gal	\$10	20,000	\$200,00
Effluent pump station	LS I	\$1,250,000	1	\$1,250,000
Subsurface drip irrigation line	Ч	\$10	54,000	\$540,000
Irrigation piping & valves	Ľ	\$250	006	\$225,000
			Subtotal	\$2,215,000
		Electrical & Inst	trumentation @ 25%	\$554,000
			Subtotal	\$2,769,000
			Contingency @ 20%	\$554,000
		Engineering, Aa	lmin, & Legal @ 25%	\$693,000
		1 <u>1</u>	fluent Disposal Total	\$4,016,00
			Nternative #2 TOTAL	\$84.329.00
		•	The second	

Naalehu WWTP Revised Preliminary Engineering Report Cost Estimate

Alternative #3 - Imhoff Tank / RGF / Subsurface Drip

Capital	ost Estimat	a		
Naalehu WWTP Capital Cost Item Description	Units	General Unit Cost	Number of Units	COST
entrature entration				
Collection System				
Phase 2 new gravity collection system w/ new WWPS & force main	S	\$27,700,000	-	\$27,700,000
			Subtotal	\$27,700,000
		0	ontingency @ 20%	\$5,540,000
		Engineering, Adn	nin, & Legal @ 25%	\$6,925,000
		Colle	ection System Total	\$40,165,000
Drainage Channel				
Drainage channel improvements	LS	\$13,700,000	1	\$13,700,000
			Subtotal	\$13,700,000
		0	ontingency @ 20%	\$2,740,000
		Engineering, Adm Desit	nin, & Legal @ 25% accord homoel Total	\$3,425,000
				000'000'01¢
Wastewater Treatment				
Environmental protection. BMPs	ac	\$11.000	2	\$22.000
Site clearing	ac	\$20,000	2	\$40,000
Site roads, guard rails, pavement	ac	\$92,000	2	\$184,000
Perimeter fence	ч	\$150	4,100	\$615,000
WWTP site grading	ac	\$30,000	2	\$60,000
WWTP site drainage improvements	ac	\$18,000	2	\$36,000
Plant water catchment/collection system	LS	\$75,000	1	\$75,000
Process yard piping	LS	\$250,000	1	\$250,000
Headworks (includes site/civil, structures, equipment & piping)	SJ	\$1,024,000	1	\$1,024,000
Influent pump station	SJ	\$1,670,000	1	\$1,670,000
Imhoff tank	LS	\$1,184,000	1	\$1,184,000
Recirculation tank	LS	\$1,320,000	1	\$1,320,000
Recirculating gravel filter	SJ	\$6,591,000	-1	\$6,591,000
Chlorine disinfection	LS	\$150,000	1	\$150,000
Plant drainage system	ac	\$40,000	2	\$80,000
Main generator (including process piping)	LS	\$500,000	1	\$500,000
Maintenance/operations/electrical building	SF	\$1,000	1,645	\$1,645,000
Sludge dewatering system	SI	\$860,000	1	\$860,000
			Subtotal	\$16,306,000
		Electrical & Instri	umentation @ 25%	\$4,077,000
			Subtotal	\$20,383,000
			ontingency @ 20%	54,077,000
		Engineering, Aan	nn, & Legal (v 25%) se Testinoni Testi	000,950,65
		wastewat	er ireatment iotal	000,000,62¢
Effluent Disposal				
Irrigation equalization tank	lag	\$10	20,000	\$200,000
Effluent pump station	IJ	\$1,250,000	1	\$1,250,000
Subsurface drip irrigation line	ч	\$10	54,000	\$540,000
Irrigation piping & valves	н	\$250	006	\$225,000
			Subtotal	\$2,215,000
		Electrical & Instri	umentation @ 25%	\$554,000
			Subtotal	\$2,769,000
			ontingency @ 20%	\$554,000
		Engineering, Aan	nn, & Legar (* 25%) unit Disassel Total	000/550¢
		chu	into insodero tuan	000,010,44
		Ah	ternative #3 TOTAL	\$93,602,000

<u>MBR deaning chemicals</u> Sodium hypochlorite & citric acid cost: \$5,000 per yr Alternative #2 only	Membrane replacement (Alt #2) Membrane cost per module 51.950 material costs only	Estimated additional costs per module SS00 shipping & hondling + instollation costs Number of membrane modules 24 modules (12 per unit) Membrane replacement cost (15 year): \$60,000	WWTP maintenance materials	RAS PP MBR PP RGF Package plant captial cost \$3,579,000 M/A Process equipment capital cost \$8,104,000 \$8,104,000 splot point Finiment relatement capital cost 58,104,000 \$5,000 not including pockage plant	process equipment replacement cost <u>\$2,026,000</u> \$2,026,000 \$4,173,500 not including package plants Total NUVP equipment replacement cost: \$5,83,000 \$5,65,000 \$4,173,500 includes 100% package plant replacement Maintenance materials cost factors <u>2,0%</u> 2,0%	Total WWTP annual maintenance materials cost \$117,000 \$113,000 \$84,000	Studge dispose1 680 Both Both	Onsite disposal roll off dumpster size <u>10</u> cu yds Dumpster rental fer even far rental fer <u>5500.00</u> per week	Annual ountyster fentar lee <u>>65,000.000</u> Disposal frequency 7 days Disposal frequency 7 days 56.06 per gallon 56.06 per gallon	Employee labor cost per hour 548.08 per hour based on 100K annual salary Distance Naalehu to Landfill (roundtrip) 165 mi Dump truck fuel economy 50 mpg	Annual studge disposal cost (truck to landfill) Annual fuel cost Annal landfill tipping fee 510,000	Amfual Jumpser Ferraintee	<u>Collection sostem</u> <u>Phase 1: Reuse Brewer gravity collection system (Close LCCs) mointenance</u> Existing Brewer gravity sever mainine <u>20,788</u> (t 2004 Kou Sever System Evoluation Report	Existing gravity mainline multiplier 3.94 mi Gravity mainline maintenance cost 548.000 Annual mainline maintenance cost: 5188.990	New gravity sever mainine (FM to WWTP) 2,500 ft New gravity sever mainine (ICCs to 5Ps) 3,500 ft New gravity sever mainine 712,510 ft 4450 ft	Gravity mainline maintenance cost 0.84 mi Annual gravity mainline maintenance cost: \$16,000 per mi	Force main length 2,800 it Fukunaga estimate Force main maintenance cost 510 000 lene mi	Annual force main maintenance cost: 55,310	WMPS process equipment replical cost 31,670,000 Assume equal to influent pump station WWPS process equipment replacement cost: \$400,000 20 year replacement	Total annual maintenance materials cost: 510,000 Total annual WWPS electricity cost \$20,000	
					Alt 3 RGF (kWh)	2,613 5.323	N/N N/N	N/A 13,065	32,662 19,597 39,194	N/N	N/A N/A	58,791	9,798 3,919 3,919	3,266 N/A 13,065	265,213 \$120.000						
					Alt 2 MBR PP (kWh)	2,613	65,323 176,373	84,920 N/A	N/A 19,597 39,194	N/A	19,597 65,323	58,791	9,798 3,919 3,919	3,266 32,662 13,065	663,685 \$300.000		ц				
					Alt 1 RAS PP (kWh)	2,613	65,323 176,373	84,920 N/A	N/A 19,597 39,194	13,065	N/A N/A	58,791	9,798 3,919 3,919	3,266 32,662 13,065	591,829 \$270.000		ime @ PHWW				
			or		Annual Power (KWh)	2,613	65,323 176,373	13,065	32,662 19,597 39,194	13,065	19,597	58,791	9,798 3,919 3,919	3,266 32,662 13,065	umption kWh: ectricity cost:		min contact ti	tives		250	tives
Report			īri + 1 supervis		Equivalent Continuous Load (hp)	0.7	21 10					ŭ	11.0	0	ectricity consi Annual el		ng/L dose, 15	oss all alterna	r	n nan ƙimikani	oss all alterna
u WWTP y Engineering t Estimates			ige benefits - Sat / Mon -		Use Factor	20%	100%	100%	100% 30% 40%	100%	30% 100%	\$06	30%	10% 100% 100%	Annual e		assuming 8	common ac		v= hillincep	соттоп асі
Naaleh ed Preliminar O&M Cos	/kWh	mgd cfs	including frir 2 shifts: Weo	_	Motor Size (hp)	2	5 27	13	10 10	1	5 5	5	2 2	1 2 2			lbs/d bs/tr	per lb	1.1	lbs/yr per lb	
Revis	\$0.45	0.125	\$100,000	\$300,000	Duty Unit Count	1	2	F F 7		2	2	2	1 1	1 1 2			8.3 3.030	\$24,300	*	621.0 53	\$1,900
	Electricity cost	Flow ADWF:	Labor (common across all alternatives) COH WWTP operator annual salary Number of employees/operators	Annual labor cost: Elorreicieu	load	<u>Headworks</u> Screens Grit blower	Process tanks Anoxic zone mixers Aeration blower (main)	Aeration blower (flow equalization) Imhoff tank odor control	Recirculation tank pump Influent pump Effluent pump	<u>Secondary clarifier</u> Clarifier mechanisms	<u>Membranes</u> Membrane blower Permeate pumps	<u>Aerobic digestion</u> Digester blowers	<u>Sludge dewatering</u> Screw press feed pump Screw press Cake conveyor	<u>Miscellaneous</u> Drainage return pumps Plant water pumps Fans		Chemicals	Hypochlorite Tablets Daily chlorine demand @ ADWF Annual hworchlorite demand @ ADWF	Hypochlorite tablet unit cost Total annual hypochlorite tablet cost:	Dewatering polymer	uany dewatering punyner use Annual dewatering polymer use Dewatering polymer unit cost	Total annual dewatering polymer cost:

Page 1 of 3

Page 2 of 3

\$238,000

Total annual maintenance cost:

Phase 2: New Brewer gravity collection system n	naintenance	
New Brewer gravity sewer mainline	10,140	Ŧ
New gravity sewer mainline (FM to WWTP)	2,500	Ŧ
New gravity sewer mainline TOTAL	12,640	Ŧ
	2.39	Ē
Gravity mainline maintenance cost	\$16,000	per mi
Annual gravity mainline maintenance cost:	\$38,310	
Force main length	2,800	Ŧ
	0.53	Ē
Force main maintenance cost	\$10,000	per mi
Annual force main maintenance cost:	\$5,310	

Fukunaga estimate

WWPS process equ	uipment capital cost	\$1,67
WWPS process equ	uipment replacement cost:	\$4

Total annual maintenance materials cost:	\$10,000
Total annual WWPS electricity cost	\$20,000
Total annual maintenance cost:	\$74,000

 cost
 \$1,670,000
 Assume equal to influent pump station

 ement cost:
 \$400,000
 20 year replacement

 ials cost:
 \$10,000
 20 year replacement

	-	

	Nas Alternati	County (lehu WWT ves Net Pre	of Hawaii P Revise sent Val	d PER ue Analy	sis		
Agency:	County of Hawaii		Sensitivit	y Adjustme	ents (%)		Results
Project/Problem:	Naalehu Collection System	Risk Premium	Benefits	Capital Costs	Other Costs	Capital Cost	30-year NPV
Alternative 1	Phase 1 - Reuse C Brewer Collection System					\$30,032,500	(\$35,930,046)
Alternative 2	Phase 2 - New C Brewer Collection System					\$40,165,000	(\$42,197,986)
Alternative 3							
Alternative 4							
Alternative 5							
Year of analysis:	2023		Select one			Note: "Status quo	" refers to
Escalation rate:	3.50%	0	All entries i	n dollars		Alternative	-
Discount rate:	5.00%	0	O All entries i	n thousands	of dollars		

entries in yellow cells only

Make

Troject/Problem: Nateleful Disinfection System Risk Reins Rank Capital Cost Cuther 30-Jear Northal Cost Alternative 1 Tablet Chlorination Feed System Enerlini Costs Costs Costs 5150 000 (5135.400) Alternative 3 Utraviolet Light (UV) System Enerlini Costs Costs 51100 000 (5135.400) Alternative 3 Utraviolet Light (UV) System Enerlini Costs Costs 51100 000 (5135.400) Alternative 3 Alternative 3 Alternative 4 Enerlini Costs S1100 000 (5135.400) Alternative 3 Alternative 3 Alternative 3 Alternative 4 Enclini Costs S1100 000 (5135.400) Alternative 3 Costs Costs Costs Costs Costs S1100 000 (51353.400) Alternative 3 Costs Costs Costs Costs S1100 000 (51353.400) Fear of analysis Costs Costs Costs Costs S1100 000 (51353.400) Fear of analysis Costs Costs Costs Costs Costs Costs Costs Fear of analysis Costs Costs Costs Costs Costs Costs Costs Fear of analysis Cos	Agency:	County of Hawaii		Sensitivit	ty Adjustmu	ents (%)		Results
Alternative 1 Tatent Chlorination Feed System 5150,000 (5733,500) Alternative 2 Unraviolet Light (UV) System 51,100,000 (5733,500) Alternative 3 Alternative 3 51,000,000 (5733,500) Alternative 3 Alternative 3 Alternative 4 Alternative 1 Discount rate: 5.00% Alternative at otolars Alternative 1	Project/Problem:	Naalehu Disinfection System	Risk Premium	Benefits	Capital Costs	Other Costs	Capital Cost	30-year NPV
Atternative 2 Utraviolet Light (UV) System \$1,100 000 \$1,333.400 Atternative 3 Atternative 3 \$1,000 000 \$1,333.400 Atternative 4 Atternative 4 \$1,000 000 \$1,333.400 Atternative 5 Atternative 6 \$1,000 000 \$1,333.400 Atternative 5 Atternative 6 \$1,000 000 \$1,333.400 Atternative 5 Atternative 1 \$1,000 000 \$1,333.400 Atternative 5 Atternative 1 Atternative 1 Atternative 1	Alternative 1	Tablet Chlorination Feed System					\$150,000	(\$738,500
Alternative 3 Alternative 3 Alternative 4 Alternative 4 Alternative 4 Alternative 4 Alternative 5 Alternative 4 Alternative 5 Alternative 4	Alternative 2	Ultraviolet Light (UV) System					\$1,100,000	(\$1,335,400
Atternative 4 Atternative 4 Atternative 4 Atternative 5 Atternative 5 Select one Atternative 5 Select one Atternative 5 Atternative 1	Alternative 3							
Atternative 5 Atternative 5 ear of analysis: 2023 Escalation rate: 3.50% Discount rate: 6.00%	Alternative 4							
ear of analysis: 2023 Select one Escalation rate: 3.50% M entries in dolars Discount rate: 5.00% O all entries in brousends of dolars	Alternative 5							
Cear of analysis: 2023 elect of analysis: 2023 electron analysis: 2029 d'a entre in datas Automotion rate: 3.50% All entries in thousands of dollars Alternative 1 Discount rate: 5.00%								
Escalation rate: 3.50% All entries in thousands of dollars Alternative 1 Discount rate: 5.00%	ear of analysis:	2023	eg S	ct one Mi antriac in c	follare		Note: "Status quo	' refers to
Discount rate: 5.00%	Escalation rate:	3.50%) (Alternative	-
	Discount rate:	5.00%	C	All entries in t	thousands of	dollars		
	-	Make entries in vellow cells only						
Make entries in vellow cells only	-	6						

	Naa Alternativ	County c lehu WWTI es Net Pre	of Hawaii P Revise sent Vall	d PER ue Analy	sis			
Agency:	County of Hawaii		Sensitivi	ty Adjustme	ents (%)		Results	
Project/Problem:	Naalehu WWTP Revised PER	Risk Premium	Benefits	Capital Costs	Other Costs	Capital Cost	30-year NPV	
Alternative 1	RAS Package Plants / Subsurface Drip					\$84,751,000	(\$109,555,755)	
Alternative 2	MBR Package Plants / Subsurface Drip					\$84,329,000	(\$109,776,643)	
Alternative 3	Imhoff Tank / RGF / Subsurface Drip					\$93,602,000	(\$112,851,053)	
Alternative 4								
Alternative 5								
Year of analysis:	2023	See See	t one All antriac in c	dollare		Note: "Status quo	o" refers to	
Escalation rate:	3.50%) (Alternative	L.	
Discount rate:	5.00%	C	All entries in 1	chousands of	dollars			
	Make entries in yellow cells only							

Naalehu WWTP Revised PER

	Naalehu I Naalehu I Alternatives Nei	ог паwa Revised / Present	N DEM NOC PER Value Ar	alysis			
							:
Agency:	County of Hawaii DEM		Sensitivity	Adjustme	ents (%)		Results
Project/Problem:	Naalehu Revised AOC PER	Risk Premium	Benefits	Capital Costs	Other Costs	Capital Cost	30-year NPV
Alternative 1	i. Package plant and new collection system					\$84,329,000	(\$109,776,643)
Alternative 2	ii. Package plant connected to existing collection system					\$74,197,000	(\$103,505,201)
Alternative 3	IWS management model 2A					\$29,100,000	(\$38,859,287)
Alternative 4	iii. A maintenance contract model IWS program					\$29,100,000	(\$39,321,375)
Alternative 5	IWS management model 3A					\$29,100,000	(\$39,093,038)
Alternative 6	iv. An operating permit model IWS program					\$29,100,000	(\$41,422,061)
Alternative 7							
Alternative 8							
Alternative 9							
Alternative 10							
Alternative 11							
Alternative 12							
Year of analysis:	2023	 Select o All enti 	ne ies in dollars			Note: "Status qu	o" refers to
Escalation rate:	3.50%	a Allent	ies in thousar	nds of dollar	5	Alternativ	e 1
Discount rate:	5.00%				1		

Make entries in

Appendix B: Letter to Support Department of Hawaii Flow Variance Application



2261 Aupuni Street, Suite 201 Wailuku, Maui, HI 96793

T: 808.244.7005

July 18, 2023

Mr. Mark Grant County of Hawaii Wastewater Division 108 Railroad Ave Hilo, HI 96729-4252 Subject: Naalehu Wastewater Treatment Plant - Capacity Phasing

Dear Mr. Grant,

The County of Hawaii (County) is investigating options to reduce the size and impact of the Naalehu Wastewater Treatment Plant (WWTP). While previous planning and design efforts focused on an aerated lagoon secondary WWTP, the County is now evaluating a mechanical secondary treatment process to strive to meet the EPA Administrative Order on Consent (AOC) deadlines. This letter provides information regarding a phased capacity innovative approach to the project scope that will further the County's goals. A phased WWTP capacity approach will require Department of Health (DOH) acceptance as it potentially falls outside of normally accepted practices and regulatory guidance. Phased WWTP capacity implementation is proposed to reduce initial WWTP size while protecting human health and the environment. The phasing approach that the County is proposing requires an explanation of deviations from Hawaii Administrative Rules (HAR) 1-62.; including utilization of potable water use data to estimate wastewater flows, peak flow prediction, and phasing implementation.

1.1 Background

HAR 11-62 requires WWTPs to be designed in accordance with County standards, or the City and County of Honolulu (CCH) standards if no applicable County standards exist. The County does not have its own WWTP standards, and therefore relies on the CCH standards.

Project planning and design to date has followed the CCH flow standards that were updated in 2017. Table 1 provides a summary of the flows to the Naalehu WWTP based on the 2017 CCH standards.

Mr. Mark Grant County of Hawaii Wastewater Division July 18, 2023 Page 2

Table 1. Naalehu	I WWTP Flows Based on 2017 CO	CH Standards
Description	Peaking Factor	Value
Average dry weather flow	1.0	225,000 gpd
Peak day dry weather flow	2.0	446,000 gpd
Peak day wet weather flow ^a	2.5	563,000 gpd
Peak hour wet weather flow	3.1	480 gpm (691,000 gpd)
Notes: and = dallons per day an	um = dallons ner minute	

rotes: gor = galoris: per oay, gun = galoris per minute • Paak day wet weather flow is not and the CCH standards but is an important WVTP design parameter. Peak day wet weather flow estimate was devoloped using an apportante peaking factor.

The 2017 CCH flow standards include three elements:

- Wastewater flow generation estimates based on equivalent population estimates (70 gallons per capita per day (good)). This reflects the amount of wastewater that is expected to enter the sewer from residences and businesses and provide the main source of organic material mass to be removed by the WVTP process.
- Dry weather infiltration and inflow (*J*/I) estimates based on equivalent population estimates (35 gpcd).
- Wet weather I/I estimates based on service area acreage (3,000 gallons per acre per day (gpad)).

1.2 County Experience with the CCH Flow Standards

The CCH standards were established for a major metropolitan area that includes vast areas of residential, commercial, and industrial development, with significant proportions of service areas near sea level elevations. Wastewater generation rates are genertions of service areas near sea level elevations. Wastewater generation rates are generally lower in rural areas than in urban areas. Typical ruban areas and the form of the nurban areas. Typical ruban areas range between 50 good in rural areas and 120 good in typical urban areas (Tchobanoglous, George, F. L. Burton, and H.D. Stensel, Wastewater Engineering: Treatment and Reuse / Metcalf & Eddy, Inc., 4th edition, 2003). The County's experience with the CCH flow standards on other projects (e.g., Honokaa WWTP) has illustrated that the standards are very conservative for small rural communities located at higher elevations on Hawaii Islain. The observed monthly average dry weather flow design capacity as designed per the 1993 CCH standards that were near the WWTP facility capacity as designed per the 1993 CCH standards that were in effect at the time (1,250 gpad for sewers above the water table). Therefore, the current wastewater standards based on urban thonolulu are likely overly conservative for runal communities like Naalehu, as subtrantiated by Tkey overly conservative for runal communities like Naalehu, as subtrantiated by Tkey overly conservative for runal communities like Naalehu, as subtrantiated by Tkey overly conservative for runa communities like Naalehu, as subtrantiated by Tkey overly conservative for runal communities like Naalehu, as subtrantiated by Tkey overly conservative for runal communities like Naalehu, as subsubariated by Tkey overly conservative for runal communities like Naalehu, as subtrantiated by Tkey overly conservative for runal communities like Naalehu, as subsubariated by Tkey overly conservative for runal communities like Naalehu, as subsubariated by Tkey overly conservative for runal communities lik



1.3 Service Area

Fukunaga & Associates, Inc (FAI) prepared a preliminary engineering report for the Naalehu wastewater collection system improvements in May 2020. The results of FAI's investigation established an updated service area for the proposed Naalehu WMTP that incorporates both the town's and County's needs. Figure 1 shows the updated WMTP service area, which includes properties that are currently connected to the three large capacity cesspools (LCC 3, 4, and 5), and properties that will be "newly accessible" to the collection system after the replacement collection system is constructed. LCC 1 and 2 are located in the neighboring town of Pahala.



Figure 1. Naalehu WWTP Service Area Established by Fukunaga & Associates, Inc

Mr. Mark Grant County of Hawaii Wastewater Division July 18, 2023 Page 4

Table 2 provides a summary of the WWTP service area.

Та	ole 2. Naalehu WWTP Servi	ce Area Summary	
December Tenno		Number of Parcels	
	Existing C Brewer Lots	Newly Accessible Lots	New Collection Total
Residential	159	25	184
Commercial	2	7	6
Church	1	1	2
Industrial	I	2	2
Agricultural	1	2	3
Residential/Commercial	1	1	2
Residential/Agricultural	•	1	1
Park	I	1	1
Total	164	40	204

1.4 Potable Water Use

The amount of wastewater generated within a residence will not exceed the amount of potable water used by the occupants. Therefore, potable water use records can be used to estimate wastewater generation rates within existing communities where no combined sewers are present. The County of Hawaii Department of Water Suppk (DWS) provided potable water use records for the parcels located within the service area from November 2017 through October 2022. Figure 2 provides an analysis of the potable water use records, with adjustments for properties for which no water use data was available. As shown figure 2, the maximum monthly average water use during the period of record was approximately 78,000 gpd in May and June 2018.



Figure 2. Average Potable Water Use in Naalehu WWTP Service Area, Nov 2017 - Oct 2022

conclude that Naalehu residents do not use significant volumes of potable water for outgust, but inspection of potable water demands in Figure 2 shows little to no increase in uses enter the sewer system. Figure 3 shows the results of an irrigation demand analydoor irrigation purposes, and most of the potable water that is supplied to the commusis for Naalehu. Peak irrigation demands occur during the months of May through Aupotable water use during the driest months of the year. Therefore, it is reasonable to Potable water is used both indoors and outdoors in residential areas, but only indoor nity by DWS ends up in the sewer.

County of Hawaii Wastewater Division July 18, 2023 Mr. Mark Grant Page 6



Based on the above analyses, we believe assuming an 80,000 gpd monthly average wastewater generation rate in the capacity calculations reflect the current needs of the service area

collection system. The 2.5 peaking factor is appropriate for Naalehu, resulting in a maxistandards use a 2.5 peaking factor to estimate the maximum wastewater flow into the Daily wastewater flows can vary substantially from the monthly average flow. The CCH mum wastewater flow of 200,000 gpd.

1.5 Dry Weather I/I Allowance

County's experience at Honokaa has been that the 1993 dry weather I/I allowance is adcreasing the flow to the WWTP. The 2017 CCH standards specify a dry weather I/I allowplan for increased dry weather I/I due to sea level rise effects in Honolulu. The Honokaa The significant increase in the dry weather I/I allowance in the latest CCH standard will help the CCH to include dry weather *V*1. The high flows in July 2016 and August 2018 were due to Hurri-canes Darby and Lane, respectively. We conclude that continued use of the 1993 ance of 35 gpcd. The previous CCH standards (dated 1993) specified a dry weather I/I line. Figure 4 summarizes annual average effluent flows at the Honokaa WWTP, which WWTP was designed for an average dry weather flow (ADWF) capacity of 200,000 gpd, elevations hundreds of feet above sea level, and a significant distance from the shoreequate for a rural collection system located in Hawaii Island's well-drained geology, at Groundwater can infiltrate into wastewater collection systems during dry weather, inwhich includes a dry weather I/I allowance based on the 1993 CCH standards. The allowance of 5 gpcd for properties located above the groundwater table.

Mr. Mark Grant County of Hawaii Wastewater Division July 18, 2023 Page 8

standard for dry weather I/I is appropriate for Naalehu and using the 2017 standard would be overly-conservative.



Figure 4. Honokaa WWTP Monthly Average Effluent Flows, Apr 2016 – Apr 2021

1.6 Wet Weather I/I Allowance

The 2017 CCH standards, which specify a wet weather \ensuremath{V} allowance of 3,000 gpad, are used for all wet weather \ensuremath{N} calculations.

1.7 Phased Capacity Implementation Recommendations

A phased approach towards implementing WWTP capacity is recommended to provide a flexible and appropriately sized wastewater treatment and disposal facility for the Naalehu community:

- Phase 1: Provide sufficient capacity for the existing parcels within the service area. including newly accessible parcels, reflecting current development. This will allow the County to close the LCCs. Provide sufficient area within the WWTP site for future expansion.
- **Phase 2:** When needed, expand the WWTP capacity to reflect the full CCH flow standards requirements for the existing service area. HAR 11-62 requires development of a facility plan when flows reach 75 percent of capacity and implementation of capacity-increasing measures at 90 percent.
- Phase 3: When needed, expand the WWTP capacity to reflect the recommendations included in the Kau Community Development Plan to sewer the entire community. This is not anticipated to occur within the near future and is not part of the current project's environmental assessment.

Table 3 provides a summary of the calculated capacities for the three-phase approach.

	Table 3. Calculated	l Capacity Phasing	
Description	Phase 1	Phase 2	Phase 3
Base sanitary flow	80,000 gpd	147,000 gpd	211,000 gpd
Peak base sanitary flow ^a	200,000 gpd	368,000 gpd	528,000 gpd
Dry weather I/I b	12,000 gpd	78,000 gpd	119,000 gpd
Wet weather I/I °	245,000 gpd	245,000 gpd	931,000 gpd
Average dry weather flow ^d	92,000 gpd	225,000 gpd	330,000 gpd
Peak day dry weather flow ^e	212,000 gpd	446,000 gpd	647,000 gpd
Peak day wet weather flow ^f	322,000 gpd PF=3.5	563,000 gpd PF=2.5	825,000 gpd PF=2.5
Peak hour wet weather flow ^h	317 gpm (457,000 gpd)	480 gpm (691,000 gpd)	1096 gpm (1,578,000 gpd)

* Base sanitary flow x 2.5 peaking factor.

 $^{\mathrm{b}}$ 5 gpcd for Phase 1, 35 gpcd for Phase 2 and Phase 3.

3,000 gpad
 ^d Base sanitary flow + dry weather I/I.

^a Base sanitary flow + dry weather I/I.
^b Peak base sanitary flow + dry weather I/I.

 Average dry weather flow x peaking factor (PF) shown. Peak day wet weather flow estimate was developed using an appropriate peaking factor.

Peak base sanitary flow + dry weather I/I + wet weather I/I.

HAR 11-62-23.1(i) requires the initiation of a facility planning process when the actual wastewater flows reach 75 percent of the design opapacity of the WWTP, and implementation of the facility plan must be initiated when actual wastewater flows reach 90 percent of the design capacity. We recommend the Phase 1 WWTP be rated to be recent of the design capacity of the capacity and an area an earting to the design capacity. We recommend the Phase 1 WWTP be rated to the variant and the point of the design capacity of the post earting to initiate a facility plan shortly after the project is constructed. The peak day and peak hour flow capacities can remain as shown in Table 3. Note that the biological processes in the me-

gpd, not the average dry weather flow.

Per capita wastewater generation rate estimates were considered to check the validity of the proposed Phase 1 average dry weather flow capacity. The non-residential propertues in the service area are expected to contribute approximately 50 percent of the total flows. Therefore, 50 percent of the average dry weather flow capacity can be allocated to the residential properties in the service area (92,000 gpd x 50% = 46,000 gpd). If we consider only the 184 residential properties in the service area, and assume four residents per residential lot, we have a total residential flow capacity by the population x 4 persons/lot = 736 persons. Dividing the residential flow capacity by the population yidds awastewater flow capacity of approximately 63 gpcd (46,000 gpd / 736 capita = 3 gpcd), which is within the typical values presented in Section 1.2 above.

The proposed Phase 1 capacity is based on actual water use data to establish wastewater generation rates, and rational assumptions to establish dry weather // al-lowances, and we believe it is appropriate for the Phase 1 existing conditions, while providing limited capacity for growth. Table 4 presents the recommended Phase 1 capacity.

	Peaking Factor	1.0	1.7	2.6	3.7
e 4. Recommended Phase 1 Capacity	Value	125,000 gpd	212,000 gpd	322,000 gpd	457,000 gpd (317 gpm)
Tabl	Description	Average dry weather flow	Peak day dry weather flow	Peak day wet weather flow	Peak hour wet weather flow

If the County pursues a WWTP approach (using the Phase 1 capacity recommendation above) to close the LCCs then a DOH variance from HAR 11-62 requirements will be needed. The variance will need to be renewed every five years. The WWTP capacity needs should be re-evaluated upon application for the variance renewal.

Please call me at (808) 442-3301 if you have any questions.

Very truly yours,

Brown and Caldwell

K.JYao

Craig Lekven, P.E. Project Manager

Naalehu WWTP Revised PER

Appendix C: Non-Economic Evaluation



					Raw Scores		-	We	ighted Scol	es
Category	Category Weight	Criteria	Criteria	Alt #1 RAS	Alt #2 MBR	Alt #3 RGF		Alt #1 RAS	Alt #2 MRR	Alt #3 RGF
Level of Service	25%	Effluent quality Potential for capacity expansion Water recycling feasibility Public perception / community concerns	30% 20% 20%	9 9 Q 9	ບບບບ			0.90 0.90 0.90 0.90 0.90 0.90 0.90 0.90	1.50 1.50 1.50	0.90
Regulatory	25%	Montroring complexity Treatment adjustment potential Safety regulations complexity Environmental concerns	25% 25% 25%	4044	ლი 4 ო	ωσ≠ ω		1.00 1.25 1.00	0.75 1.25 1.00	1.25 0.75 1.00 0.75
O&M Factors	25%	Footprint Safe work environment Maintenance complexity Operations complexity	30% 25% 20%	r0 4 4 4	မကကက	0440	I	1.50 1.00 0.80	1.50 0.75 0.75 0.60	0.60 1.00 1.00
Island Factors	25%	Mainland delivery dependence Mainland servicing dependence Power dependence Chemical dependence	25% 25% 25% 25%	n n n 4		4404	I	0.75 0.75 0.75 1.00	0.75 0.75 0.75 0.75	1.00 1.25 1.00
***Note : 5 = high/des	irable, 1 = low/nc	it-desirable				Overall S	core:	3.80	3.96	3.53

Appendix D: Naalehu Wastewater Collection System Improvements Technical Memorandum

Fukunaga & Associates, Inc., May 2020





MEMORANDUM

- TO: Michelle Sorensen, Brown and Caldwell Craig Lekven, Brown and Caldwell
- FROM: Andrew Amuro
- DATE: May 1, 2020
- SUBJECT: Naalehu Wastewater Collection System Improvements Final Preliminary Engineering

1. GENERAL PROJECT DESCRIPTION

The County of Hawaii (COH) is scheduled to close three large capacity cesspools (LCCs) in the town of Naalehu on the southeast side of the Big Island. The Naalehu town area is shown on **Figure 1**. To accomplish the closure, the COH has tasked Brown and Caldwell (B&C) with the planning and "preliminary regimering for a wastewater treatment plant (WVTP) to serve the properties impacted by the LLC closure. Fukunaga and Associates, Inc. (FAI) has been tasked with investigating options for the collection system to convey the wastewater from the impacted properties to the properties updated by the LCC closure of the project is to close the LCC as expediently and economically as possible; however, the Consent of the Project is to close the LCC as expediently and economically as possible; properties currently served by the LCCs.

Three properties requested connection to the County sewer system and are presented in Figure 1. These properties are:

- Iglesia ni Cristo Church (TMK 9-5-009:079)
- Punalu'u Bakery (TMK 9-5-025:037; owned by La'i LLC)
- Hana Hou Restaurant (TMK 9-5-021:005; owned by Fujimoto, Drake S.)

These properties are not connected to the former Brewer sewer system. The church property is located on the Hilo side of the former Brewer house lots and can be connected to the collection system by adding a sewer within the State right-of-way (ROW) (see Section 2) or by adding a sewer within an easement behind the propert if the County chooses to extend the sewer to this property. The Pumalu'u Bakery and Hana Hou Restaurant are located on the Kon side of the former Brewer house lots. This side of Naalehu town, to the west of the concrete-lined drainage canal, is considered the "commercial" side of Naulehu town. The vest of the concrete-lined drainage to accommedate connection of these two commercial properties and will serve to define the western textent of the collection system.

fukunagaengineers.com office@fukunagaengineers.com

Memorandum Naalehu Collection System Improvements May 1, 2020 Page 2

2. GENERAL CRITERIA

The COH will be the owner of the collection system; therefore, the sewer system will meet COH standards and be accessible for maintenance. Preference is to construct sewers within the County ROW as much as possible unless other factors make placing the pipes within easements on private property or in the State ROW much more practical from economic and engineering standpoints. The COH prefers sewers not be within easements due to future access challenges when maintenance and repairs are needed. It is also recommended that the collection system not be located in a State ROW if possible due to the lengthy review and approval time needed for the Use and Occupancy Agreement (UOA). This was emphasized by the State Department of Transportation Highways Division (DOT-H) in their Environmental Assessment (EA) pre-consultation comment to stay out of the State ROW unless there is no other fassible means of routing the sewer. The collection system will require a crossing of the State ROW to convey the sewage from the community Mauka of the highway to the WWTP site that is Makai of the highway; however, DOT-H tends to take less time when reviewing ROW crossings, especially if they are done using trenchless installation techniques.

3. TOPOGRAPHY

The topography of the Naalehu area is presented in Figure 2. The proposed location of the wastewater treatment plant is upgradient of the lowest of the former Brewer properties. The ground levation of the WWTP site is approximately 690 ft MSL while the ground elevation of the lowest Brewer house lot is about 645 ft MSL, an elevation difference of 45 feet. A gravity sever at this depth is technologically possible using micro tunneling techniques for construction; however, a wastewater pump station will be more feasible to convey the wastewater to the WWTP due to the cost of micro tunneling a relatively short distance. The micro tunneling option can be investigated in the event that an acceptable wastewater pump station location cannot be found.

4. POTENTIAL WASTEWATER PUMP STATION LOCATIONS

As noted in the previous section, the former Brewer properties in Naalehu would most likely be served by a pump station due to the topography. When investigating potential sites for a wastewater pump station (WWPS), priority was placed on sites that are on County or State lands to minimize land acquisition issues, or large private parcels where a portion can be purchased by the County in hopes of minimizing impact to the landowner. The potential WWPS sites are listed in Table 1, and shown on Figure 3.

Table 1: Potential WWPS Sites

TMK	Owner	Current Use	Class	Size (sq ft)	Notes
9-5-008:012	Chinese Society Church	Church	Agricultural	186,001	Requires land from the owner and/or searcenst for WWPS sife, utilities and road. Requires sever and force main in State ROW. WVPS can be set back away from houses. Located at lowest point of the collection system. Within 550 feet of school property line. Ungest distance for transmission to proposed WWTP location.
9-5-008:048	Kuahiwi Contractors Inc.	Storage/ Baseyard	Agricultural	534,307	Requires land from the owner and/or searcenst for WVPS site, utilities and road. WVPS can be set back away from most residents. Within 300 feet of school property line. Located mear lowest point of the collection system requiring approx, 15 foot wet well.
9-5-009:006	State of Hawaii	Naalehu School	Residential	534,481	Site is frequented by the public. Residents have indicated sensitivity to wastewater facility at a school. Located at lowest point of the collection system.
9-5-021:023	County of Hawaii	Naalehu Park	Residential	279,176	Site is frequented by the public. Residents may be sensitive to wastewater facility at a park. Requires sewer in State ROW. Located upgradent of some of the former Brewer properties.
9-5-021:035	Thy Word Ministries – Waikoloa Faith Ctr	Vacant	Residential	134,949	Requires land acquisition from the owner and access easement or dedicated County read. Routing would most likely have sewer within State ROW. WWPS can be set back away from most houses.
9-5-024:007	'O Ka'u Kakou Inc.	Senior Center	Residential/ Commercial	84,680	Facility currently in construction, timing to coordinate location could be difficult. Residents may be sensitive to wastewater facility within senior center campus.
9-5-024:011	County of Hawaii	LCC Location	Residential	10,347	Site is surrounded by houses. Facilities will be relatively close to property line. Residents may be sensitive to wastewater facility in residential area. concided upgradient of the lowest Brewer proceedies requiring 24.0 food down wer well

Memorandum Nauehu Collection System Improvements May 1, 2020 Page 4 Two of the potential sites appear to be more favorable than the others, the County site located within the former Brewer house lots (TMK 9-5-024:011) and the Kuahiwi Contractors site (TMK 9-5-008:048). The remaining parcels were eliminated from consideration at this time but can be re-visited in the future if needed. The Naalehu School and Naalehu Park sites would likely face the most public opposition due to proximity to areas frequented by the general population for recreation and education. The Chinese Society Church and Thy Word Ministries sites would require land acquisition after negotiation with the land work and TUA within the State ROW. Both processes would be time consuming and the land expusition depends on the landowner's willingness to cede some of their land to the County. The new senior center parcel is currently in construction, making the timing to negotiate the WWPS site difficult and may possibly incur construction delays that would also factor into the land negotiations.

A. Pump Station on County Parcel TMK 9-5-024:011

The County site would be the most favorable from a timing standpoint. It does not require land acquisition and would minimize work in the State ROW by limiting construction to a single crossing of Mamalahoa Highway. However, since this site is up-gradient from the collection system low point, this site wull require deeper sewers and a very deep pump station wet well. Construction on this site would be more challenging due to the narrow site and the existing sever stand LCC that would have to remain in operation until the collection system and WWTP are completed. These utilites, especially the existing sever pipes, would have to be protected during construction as heavy tenders and machinery enter and leave the site. The condition of the existing pipes may require additional protection such as supplemental concrete cover and/or partial replacement depending on condition. A proposed layout of a pump station on the County parcel is shown in **Figure 4**.

The need to trench for the deep influent sever and the deep wet well may result in flow from the LCC being diverted to these trenches due to the lower resistance to flow through the ground to those exeavated areas. Using mud walls for groundwater control would hinder the flow of sewage from the LCC into the ground and probably reduce the capacity of the LCC. Therefore, septage pumping for the duration of the pump station construction would most likely be required until the collection system and WVTP are complete. This would be a significant cost to the construction. WVTP, the addition, should here be undiressend delays in completion of the collection system and/or wWTP, the additional septage handling due to the delay could result in a significant construction change order for the additional pumping.

The on-site stormwater may be more difficult to contain due to the small area. Pervious pavement, if geotechnical investigation shows this type of pavement is permissible, would be used to minimize hard surfaces on the site. The only hard pavement area required would be the area designated for refueling of the generator. The grade of the site would be adjusted to minimize flow of off-site stormwater. Grade adjustment walls are anticipated along the property lines on the south and west sides of the parcel to raise the ground elevation.

This site is surrounded by residences directly adjacent to the property line on the east, south and west. The north side is bordered by Opukea Street. Due to the narrow site, the house structures are only about 60 feet from the proposed pump station location. The ambient noise from odor control ventilation fans and electrical switches could be a nuisance to the residents if not mitigated. The standby generator would also produce noise during testing and operation. This would require additional sound insulation for the generator and other equipment on the site. A passive odor control system can be used to eliminate the need for a mechanical fan; however, passive systems rely on the collection system to be at a higher pressure than atmosphere to force the foul air through the odor control system. Nhen the collection system becomes pressurized, odors can begin to leak out of the wet well hatches and nearby manholes so they must also be gasketed or scaled. A forced-air odor control system using a mechanical fan to draw air out of the pump station would keep the wet well and collection system at negative pressure, thus minimizing the chance that odors scape from nearby manholes and hatches. Electrical and water utilities are located just off-site on Opukea Road. The HELCo transformer for the site can be located within an easement just off the County road. Water for fire protection could be from existing hydrants (pending analysis by a fire protection engineer). Water service is anticipated to be similar to a residential unit to provide wash down and general housekeeping hose bibbs.

This site poses several risks during construction that could add significantly to the cost and time to construct the pump station. The site is also narrow and cannot be configured optimally to mitigate noise, which could be significant to the homeowners directly adjacent to the site.

B. Pump Station on Portion of Kuahiwi Contractors, Inc. Parcel TMK 9-5-008:048

The Kuahiwi Contractors site is private property and would require time to negotiate the subdivision or easement for the WWPS site and access roadway; however, similar to the County site, work in the State ROW can be limited to a single crossing of the highway. The anticipated size of the site is about 8000 to 9000 square feet. A proposed pump station layout and site are presented in **Figure 5**. A larger area may be needed if soil conditions require a larger area to mitigate on-site stomwater flows. This is smaller than the County parcel above but the dimensions can be optimized to provide the area needed for the facility. This site is located at the low point of the collection system, which would keep the severs and pump station wet well at more typical depths. Locating the WWPS on a portion of the Kuahiwi Contractors site would allow severing of the Iglesia in Cristo Church that requested connection but will place. It and allow severing of the use would also be needed to the site. A gravel drive would allow surface area but can be susceptible to erosion during heavy rains.

The pump station would be placed close to the south property line, the low side of the parcel. The far south east or Hilo corner of the parcel is the lowest point where stormwater from this parcel and possibly some adjacent parcels collect before entering a drainage culvert under Mamalahoa

Memorandum Naalchu Collection System Improvements May 1, 2020 Page 6 Highway. Anecdotal reports from residents indicate the highway floods at the storm drain during heavy or sustained rains. There is also a swale that appears to divert stormwater from cattle and horse pens on the site toward the far south east corner. It is recommended that the County not encreach into the far south east corner to maintain the current storm drainage pattern from the site. An aerial view of the area is shown in **Figure 6**. On-site stormwater can be contained by making the site large enough for impoundments and by using pervious pavement where possible depending on geotechnical investigation data. The only road area requiring hard surfacing the perimeter of the pump station site to raise the grade on the south side and lower the grade on the north side to even out the elevation of the site and minimize inflow of off-site storm water.

The site would be bordered on the south by residences unless the pump station is moved further interior of the parcel. If the pump station site is located against the south property line, the pump station could be as close as 80 feet from the house structures. Similar to the County parcel, the ambient noise from equipment could be a nuisance to the residents if not mitigated. This parcel would allow the County to obtain a more optimally configured site that would allow an additional motor control center (MCC) noom that would not only mitigate noise issues from the effectival switches, but also allow the valuable electrical equipment to be secured in a locked room. The MCC room can be integrated into the generator building. Noise from the odor control system could allo as be an size, especially from the fan that would operate continuously. However, the site can be optimally sized to allow sound attenuating enclosures around the fan or around the county parcel site is probably not needed at this site.

Running electrical utility and communication lines would be more costly for this site since the closest pole would be on Ohai Road. HELCo would charge the County to run utility lines to their transformer near the pump station site. As an alternative, the HELCo transformer and metre can be placed within an easement near Ohai Road and the County would own the power lines going into the site. This option is mentioned in case HELCo prefers not to own the power lines going into the site. This option is mentioned in case HELCo prefers not to own the power lines going into the site or requires road improvements that would not be favorable to the County. This would require discussion with and approval by HELCo and would have to be analyzed further by an electrical engineer. A potable water line would be required at the site for wash down and general housekeeping. In addition, a hydrant for fire protection would be needed. The potable water pipe in Naalehu appear to be 6-inch diameter. The new potable water line to the site would maintain the 6-inch size to maximize the water flow in case of fire or other emergency requiring water.

This site is preferred from an engineering standpoint because it is at the low side of the collection system and does not entail unknowns during construction such as breakage of existing utilities or need for septage pumping for unknown duration. It can also be optimally configured to mitigate noise and odors.

5 COLLECTION SYSTEM DESIGN

The town of Naalehu is divided by a concrete flood control canal into an east side (Hilo side) where the majority of the properties are residences, and a west side (Kona side) where the majority of the properties are commercial. The town is also divided in the Mauka (mountain) and Makai (ocean) directions by the Mamalahoa Highway. The collection system will convey the wastewater to the proposed WWTP that is located on the Makai-Kona side of Naalehu. The Naalehu collection system will be analyzed as two distinct systems, one serving the former Brewer house lots located on the Mauka-Hilo side to allow closure of the LCCs, and the gravity sewer located on the Kona side to transport the sewage to the proposed WWTP. An added benefit would be provision of sewer service to the two restaurants requesting connection. The church on the Hilo side that requestic service falls outside of this area. The County has expressed a desire to accommodate their request; however, depending on the location of the WVPS, connection of the church may have to be deferred.

C. Former Brewer House Lots

The former Brewer house lots are located on the Mauka-Hilo side as shown in Figure 7. The houses are on a hillside with a slope of 8% to 9%, ranging in elevation from about 780 ft MSL to 645 ft MSL. The steep hillside results in houses located on the downhill side of the street being pelow the elevation of the street. Sewage from houses on the uphill lide can flow easily by gravity to a sewer in the street. On the downhill side; however, the lateral serving the home is lower than a typical depth sewer (typically designed with 4 feet to 5 feet cover). Four options were investigated adpents expertise; use of easternents on private property, vacuum sewers, individual grinder pump stations at low-lying properties, and deep sewers to allow from the low-lying properties.

i. Easements

The use of easements was eliminated as a viable option at this time. An initial investigation of the house lots along the backyard property lines where an easement would be most beneficial revealed extensive homeowner improvements. As part of sewer construction and easement terms, the improvements can be removed and the homeowners can be ordered to keep improvements off of the easements. However, over time it is likely that the homeowners' memory of the easement locations would fade and various improvements may be placed over the easement, hindering access by the County for repairs and maintenance.

ii. Vacuum Sewers

The use of vacuum sewers was also eliminated at this time. Vacuum sewers use a sealed sewer system that produces suction to move the wastewater from low or flat areas. Research into these types of systems revealed that it is not cost effective to sewer the uphill houses by gravity and the downhill houses by vacuum because of the need for a sealed piping system

Memorandum Naalehu Collection System Improvements May 1, 2020 Page 8 for vacuum service. Serving some house by conventional gravity and some by vacuum would require two sewer collection systems in the County ROW. The vacuum interface valve that would be located on each property requires frequent maintenance checks to ensure proper operation. These checks should be done by trained maintenance personnel but since these valves would be on private property, the responsibility would fall on the homeowner to hire trained personnel or obtain training on the system. Another drawback is that the vacuum interface valves typically fail open to minimize the chance of spills. A failed valve will therefore result in the County-owned vacuum pumping system to run constantly (normal operation would be intermitten to save energy). County maintenance personnel would not be able to readily find the faulty valve since they would not have right-of-entry onto the individual properties. Also, since there is no spill or "problem" with the sewer at the individual property with the faulty valve, there is no incentive to correct the problem in a timely manner. Remote telemetry can be added to the vacuum interface valves to monitor the valves via SCADA but these would also require periodic maintenance and to the valves via SCADA but these would also require periodic maintenance variate property and would ad to the cost of the system for monitoring at every property.

iii. Individual Grinder Pump Stations

The original concept developed for Brewer before the company dissolved was to have individual grinder pump stations at the low-lying properties. This type of system consists of a small pump station at each downhill home. As the pump station fills with sewage, the high wastewater level starts a pump that evacuates the station and discharges into the gravity sewer system on the street. Houses on the uphill side of the street would discharge sewage by gravity into the collection system. The advantage of individual grinder pump stations is the ability to keep the collection system as shallow as possible. Disadvantages of this system are the need to maintain the pumps and valves on private property and the energy used to pump the wastewater. Another disadvantage is the potential for spills when there are mechanical failures such as, power outages, pump breakdowns, level switch malfunctions, and clogs within the pump. Similar to the vacuum sever system, the mechanically intensive nature of individual grinder pump stations requires trained minitance personnel to perform regular checks on the system. However, since the system belongs to the homeowner, it is their responsibility to either hire a maintenance service or obtain the necessary training. Unlike the vacuum system, the pump stations are not "fail safa" would prompt them to seek corrective action. A nough life-cycle cost analysis of an individual grinder pump unit resulted in an annual cost burden of \$\$50 per house on the homeowners.

The subsurface conditions of the area at the depths required for the sewers is yet to be investigated. If the deep gravity sewers (see below) are not feasible due to unfavorable conditions; such as, lava tubes, then the sewers may have to be kept as shallow as possible. This option is feasible and will be kept in consideration in the event that subsurface

conditions require it, but is not the preferred option due to the disadvantages, especially the cost burden on the homeowners.

iv. Deep Gravity Sewers

The majority of the collection system sewer mains can be kept within the County or State ROW and fed by gravity if the pipes are deep below the low point of the low-lying parcels. Sewers constructed in the County of Hawaii must adhere to the *Wastewater System Design Standards of the City and County of Honolulu*, which requires review and approval for sewers deeper than 15 feet due to higher soil loading and potential difficulty when repairs are needed. A preliminary layout of a gravity sewer system, as shown in **Figure 7**, resulted in sewer depths of about 20 to 25 feet at the deepest point on the upper portions of Ohai Road and Nahele Street, and 15 to 18 feet at the deepest point on Lokelani Street and Klikk Street. These depths are based on running a lateral from the lowest point on a house lot up to the back of the lot that are asured to be connected to the existing collection system. The soil depth to rock in the area as reported by USGS and verified by past geotechnical investigations is 40 to 60 inches; therefore, excavation for the deep gravity sewers would be mainly in rock. A sever along the highway or easements would be needed to service the houses on the State highway between Kukui Road and Ohai Road. The depth, length and direction of flow for this sever would depend on the location of a pump station. The depth of the sever on Opukea Street depends on the location of a sevage pump station. The depth of the sever on Opukea this sever would depend on the location of a sevage pump station. The depth of the sever on Opukea Street depends on the location of a sevage pump station is est escient 4). As a worse case, the manhole fronting the County-owned parcel (TMK 9-5-024:011) on Opukea Street would have an invert about 30 feet deep if a pump station is constructed on that site due to the elevation of the house at the corner of Ohai Road and Mamalahoa Hwy (TMK 9-5-024:001), the lowest point of the former Brewer community. Although 30 feet is rather deep, it is not unheard of at the entrance to pump stations in Honolulu which are often 3 stories deep. At the Kuahiwi Contractors site (TMK 9-5-008:048) the ground elevation is about the same as the house at Ohai Road and Mamalahoa Hwy resulting in severs that are at a more reasonable depth of about 15 feet.

This option is feasible and will be considered further. Despite the depth of the sewers in the County ROW, the advantage of not burdening the homeowners with the cost of maintaining the pumping system in Section 5-A-iii, above, makes this option preferred over the individual grinder pump option.

D. Gravity Sewer from Force Main to WWTP

The primary purpose of the sewer on the Kona side of Naalehu is to convey the wastewater from the Brewer house lots to the WWTP. A gravity sewer on Kaalaiki Road can be used to collect the wastewater discharged from a pump station servicing the housing area on the Hilo side of the

Memorandum Naalehu Collection System Improvements May 1, 2020 Page 10 drainage canal at either the County parcel on Opukea Street or the Kuahiwi Contractors site. The force main can be run entirely Mauka of the highway, cross the drainage canal, and discharge into a manhole when it reaches Kaalaiki Road. An easement would be required from the drainage canal to the County road. There is an existing access and utility easement located on a parcel owned by La'i LLC (TMK 9-5-025:039) that gaccess and utility easement located on a parcel be the recommended method of conveying the sewage to the WWTP to keep construction in the State ROW to a minimum. The force main route is discussed further in the next section.

Two commercial properties requested connection to the new collection system. These were the Punaluu Bake Shop (TMK 9-5-025:037) and Hana Hou Restaurant (TMK 9-5-021:005). These properties can be connected to the collection system via the gravity sewer running down Kaalaik Road, across the highway, and down Naalelu Spur Road to the WWTP site. There are potentially an additional eight properties that would be newly accessible along this sewer route depending on the disposition of Naalelu Spur Road, which is privately owned by Kuahiwi Contractors, Inc. If the road is dedicated to the County on a easement in favor of the County spans the entire road width, then both sides of Naalelu Spur Road would be newly accessible. If the sewer easement is only on one side of the road, assumed to be the Hilo side to allow connection of Hana Hou Restaurant (TMK 9-5-021:005), then only the Hilo-side properties would be newly accessible. The proposed Kona-side collection system is presented in **Figure 8**.

E. Force Main from Brewer House Lots to Gravity Sewer on Kona Side

A force main will be needed to convey the wastewater from the WWPS that serves the former Brewer house lots to the gravity sewer going to the WWTP on the Kona side. The force main can be kept out of the State ROW by running the pipe along easements and County roads. **Figure 8** shows the possible force main routes from the County parcel and the Kuahiwi Contractors sites. The approximate force main lengths from the County parcel and Kuahiwi Contractors parcel are 1300 LF and 2800 LF, respectively. Both WWPS options will require crossing the concrete drainage canal that divides the Kona side of Naalehu from the Hilo side. The force main can go either above the canal on a support structure, or under the canal. There is an existing pedestrian bridge crossing the canal that can be reconstructed and used to support the new force main; however, the County DEM is not in favor of this option due to the possibility of the structure impeding flow through the canal and ownership of the bridge structure being a burden on DEM personnel who are not set up for bridge inspection and maintenance duties. Routing the force main under the canal can be end by either cutting the concrete canal structure being a burden on DEM personnel who are not set up for bridge cutting the concrete canal structure and laying the pipe in an open trench or tunneling under the canal. Cutting the structure poses some risk should a rain storm occur while the trench is open or repaired concrete is curing. Tunneling would be costly due to the depth and equipment needed to micro-tunnel or the shoring required to excavate under the structure. Further discussions with a geotechnical engineer will be needed to determine the best method to cross the canal while

Naalehu Collection System Improvements May 1, 2020 Page 11 Memorandum

limiting risk during construction. For planning purposes, it will be assumed that the canal crossing will be approximately 100 LF using micro-tunneling as a worse case.

6 COLLECTION SYSTEM COSTS AND RECOMMENDATIONS

Planning level capital costs were developed for the collection system options developed above. The costs are presented in Table 2. Basis for the costs are presented in Table 3. Cost basis numbers are from recent past project construction and bid costs inflated to the current year.

Table 2: Collection System Costs

Deep Gravity Sewer at House Lots	\$12 million
Individual Grinder Pumps at House Lots	\$9 million
WWPS at County Parcel	1.1 million + 1.3 million FM = 8.4 million
WWPS at Kuahiwi Contractors	4.9 million + 2.8 million FM = 7.7 million
Canal Crossing (Tunneled)	\$0.7 million
Gravity Sewer from FM to Proposed WWTP	\$3.3 million

Table 3: Cost Basis

	Unit Cost
Deep WWPS at County Parcel	\$7.1 million
Typical WWPS at Kuahiwi Contractors Parcel	\$4.9 million
Gravity Sewers	\$1100/LF
Deep Gravity Sewers in Rock	\$1600/LF
Deep Sewers (Tunneled)	\$7000/LF
Sewer Force Mains (not including tunneled section under canal)	\$1000/LF
Note: Cost of individual grinder pump system based on design cost estimate of \$5.6 million prepar	ed in 2012 inflated to
2020 at 6% average inflation (\$5.6m x 1.06 $(2020-2012) = $ \$9 million)	

highway, and down Naalehu Spur Road. There are up to an additional fifteen properties that would be the County parcel, the overall estimated cost of the system would be \$24.4 million. If the the WWPS collection system. These easements have already been surveyed and are located on the south side of accessible properties along the sewer to serve the commercial lots brings the total newly accessible properties to 39, exceeding the number identified in the Consent Order. If the WWPS is located on The overall recommendation for the collection system is to use deep gravity sewers for the former properties requested connection to the new collection system. These were the Punaluu Bake Shop avoiding the need for a sewer in the State ROW to serve three of the properties. Two commercial TMK 9-5-024:009 and TMK 9-5-024:010. These easements would expedite implementation by Brewer housing area. Two backyard easements would be used to connect four properties to the connected to the collection system via a gravity sewer running down Kaalaiki Road, across the newly accessible along this sewer route. The newly accessible house lots along with the newly (TMK 9-5-025:037) and Hana Hou Restaurant (TMK 9-5-021:005). These properties can be

Naalehu Collection System Improvements May 1, 2020 Page 12 Memorandum

is located on the Kuahiwi Contractors site, overall estimated cost would be \$23.7 million. The recommended collection system options are presented in Figure 9 and 10. If using deep sewers in the former Brewer house lots is not feasible due to subsurface conditions, the estimated cost of the collection system would be \$21.4 million if the WWPS is on the County parcel, and \$20.7 million if the WWPS is on the Kuahiwi Contractors site. However, there would be added cost for the homeowners with parcels below the road of approximately \$850 per year per house to maintain the pumps. A summary of the overall costs is presented in Table 4.

Fable 4: Overall Planning Level Cost Estimates

	Deep Gravity Sewers	Individual Pumps
County WWPS at County Parcel	\$24.4 million	\$21.4 million
County WWPS at Kuahiwi Contractors Site	\$23.7 million	\$20.7 million

COLLECTION SYSTEM PHASING ~

construct the permanent WWPS as part of Phase 1. Phase 1 would therefore consist of the following: minimum, the collection system would have to intercept the existing sewers entering the LCCs and convey the sewage to a pump station. To minimize the need for temporary facilities that would be removed when the collection system is completed in a second phase, it would be beneficial to The collection system can be constructed in phases to expedite the closure of the LCCs. As a

- a. Construct Gravity Sewer on Kaalaiki Road and Naalehu Spur Road to the WWTP
 b. Construct the WWPS and force main
- c. Construct the gravity sewer on Opukea Street and Ohai Road to intercept existing sewers and connect to the WWPS

main WWPS and means of conveying the flow to the WWTP will be in place so that bypass pumping to the WWPS can be used if needed. If the main WWPS and conveyance sewers to the WWTP were "b" can be done concurrently; however, intercepting the existing sewer system should be done at the not in place, the contractor would have a more difficult time with bypass pumping to the WWTP or end of Phase 1. This is so that the if there are issues encountered when constructing the sewers, the The sewer system for the proposed Phase 1 is presented in Figures 11 and 12. Sub-phases "a" and would have to truck sewage if a problem occurs. The new WWPS would be close enough to allow temporary piping to be routed from the problem area into the wet well.

The cost to implement Phase 1 would be about \$15.3 million with the WWPS on the County parcel, and about \$15.1 million with the WWPS on the Kuahiwi Contractors site (not including land acquisition costs). A phased cost summary is presented in Table 5.

Table 5: Phased Cost Summary

	Deen Gra	vity Sewers	Individu	al Pumos
	County Parcel	Kuahiwi Parcel	County Parcel	Kuahiwi Parcel
Phase 1	\$15.3 million	\$15.1 million	\$12.3 million	\$12.1 million
Phase 2	\$9.1 million	\$8.6 million	\$6.1 million	\$5.6 million
Total	\$24.4 million	\$23.7 million	\$18.4 million	\$17.7 million







Figure 2: Naalehu Topography



Figure 4 Naalehu Wastewater Collection System Pump Station Layout on County Parcel











Appendix Pump Station Cost Estimates



\$ 800,000 \$ 110,000 \$ 60,000 \$ 1,050,000 \$ 4,830 \$ 24,600 \$ 34,800 4,830 24,600 39,740 26,350 5,940 5,940 5,940 115,260 9,960 9,960 9,960 9,960 9,960 115,260 9,960 15,760 13,560 1,355 1,355 1,355 1,355 1,355 1,355 2,660 3,560 1,355 1,3 TOTAL ŝ \$ \$ s s \$ 5 s v 0 5 3.00 5 11.00 5 0 5 3.00000 5 8.00000 5 0 5 3.00000 5 3.00000 5 0 5 1.50000 5 1.240000 5 0 5 1.50000 5 1.240000 5 0 5 3.00000 5 1.240000 5 0 5 3.00000 5 2.85000 5 0 5 3.0000 5 2.85000 5 0 5 3.0000 5 2.85000 5 0 5 3.0000 5 2.85000 5 0 5 3.0000 5 2.85000 5 0 5 3.0000 5 2.85000 5 0 5 1.000 5 1.3500 5 0 5 1.000 5 1.3500 5 0 LABOR EQUIPMENT UNIT COST \$ 800,000.00 5 12,00000 5 3 5 5,000 5 3 5 5,44000 5 3 5 5,2500 5 325.00 5 2,2500 5 3000 5 5 800.00 5 6 5 460.00 5 7 5 450.00 5 9 5 460.00 5 10 5 2000 5 10 5 2000 5 10 5 2000 5 10 5 10.00 5 10 5 10.00 5 10 5 1750.00 5 10 5 1750.00 5 10 5 1750.00 5 10 5 1750.00 5 10 5 1750.00 5 10 5 1750.00 5 10</ 8.00 15,000.00 6,000.00 MATERIAL ŝ 1 1 1 10 1 1 35 1 1 35 1 1 35 1 1 35 1 1 35 1 1 2 0 2 31 1 1 1 1 1 2 0 1 <tr/td> 1 <td QTY UM Haul Spoils Septage pumping (keep LCC low) Septage puming Concrete mat foundation Concrete mat foundation Wet well stave: 8' dia, 9' wall Wet well back, sa ad aluminum Wet well harch, sa and aluminum Wet well interior coating Wet well interior coating Wet well interior coating Wet well exterior molsture prot Engineered backfill Subm Pumrsy, doing, 0ft TDH Disch elbows and guides Valve vault structure Emergency pump conn vault Gate Valve, 6" Check valve, 6" Quick connect hose coupling Clearing and grubbing 90 bend, FExFE, 6" Tee, FExFE, 6" 45 bend, MJ, 6" Wye, MJ, 6" Plug, MJ, 6" DI pipe, FE, 6" Machine setup Wet well shaft DESCRIPTION Mobilization

Vertical Shaft Machine (Herrenknecht VSM 8000 at 4.5 m diameter)

County Parcel

Pump Station Option Summary Table

County Parcel Vertical Shaft Machine	Ş	7,700,000	*	
County Parcel Dig and Shore	Ŷ	7,100,000		
Kuahiwi Site Minimum VSM	ŝ	6,100,000	# *	
Kuahiwi Site Minimum Dig and Shore	Ş	4,900,000	#	
Kuahiwi Site Large, Dig and Shore	Ŷ	5,600,000	#	

Notes: • Includes full cost of mobilizing VSM to Hawaii. If VSM is used for any other purposes (e.g., translespipe installation, deep SMHs etc.), then mobilization cost will get distributed. # Does not mudue cost of stormwater control that may be required due to flooding issues at 5E corner of site

01 pipe, MJ, 6"	150	۳	ŝ	180.00	\$ 25.00	\$ (15.00	\$ 220.00	ŝ	33,000
						_				
sewer trench excavation, soft	270	2			\$ 25.00	\$ (7.00	\$ 32.00	ŝ	8,640
sewer trench excavation, rock	110	2			\$ 175.00	Ş	35.00	\$ 210.00	ŝ	23,100
Haul Spoils	380	S			\$ 13.00	ŝ	8.00	\$ 21.00	ŝ	7,980
French Shoring, 0-10 ft	2200	sf	Ş	2.00	\$ 5.00	\$ (3.00	\$ 10.00	ŝ	22,000
rench Shoring 10-20 ft	2200	s,	ŝ	2.50	\$ 7.00	\$ 0	3.00	\$ 12.50	ŝ	27,500
french Shoring 20-27 ft, rock	1540	s	ŝ	3.50	\$ 9.00	ŝ	5.00	\$ 17.50	ŝ	26,950
eptage pumping (keep LCC low)	1.5	ê	ŝ	6,000.00	\$ 4,800.00	\$ 0	1,500.00	\$ 12,300.00	ŝ	18,450
septage hauling	1.5	ĥ			\$ 14,400.00	ŝ	3,000.00	\$ 17,400.00	ŝ	26,100
nfluent sewer, 14" DI, lined	110	ц	ŝ	260.00	\$ 45.00	\$ 0	25.00	\$ 330.00	ŝ	36,300
teinf conc jkt	6.3	5	ŝ	950.00	\$ 350.00	\$ 0	55.00	\$ 1,355.00	ŝ	8,537
CLSM Backfill	285	S	ŝ	300.00	\$ 5.00	\$ 0	10.00	\$ 315.00	ŝ	89,775
3 ackfill	41	S	Ş	20.00	\$ 42.00	\$ (10.00	\$ 72.00	ŝ	2,952
Deep manhole (27 ft)	7	ea	ŝ	7,500.00	\$ 3,300.00	\$ 0	1,500.00	\$ 12,300.00	ŝ	12,300
Manhole interior coating	355	sf	Ś	85.00	\$ 20.00	\$ 0	10.00	\$ 115.00	ŝ	40,825
MH exterior moisture protection	410	sf	ŝ	12.00	\$ 10.00	\$ (2.00	\$ 24.00	ŝ	9,840
						_				
Odor control unit	1	ea	ŝ	7,150.00	\$ 6,000.00	\$ (400.00	\$ 13,550.00	ŝ	13,550
Senerator Bldg	320	sf	ŝ	250.00	\$ 265.00	\$ (10.00	\$ 525.00	ŝ	168,000
Roofing membrane	5.8	s	ŝ	3,000.00	\$ 2,000.00	\$ 0	170.00	\$ 5,170.00	ŝ	29,986
slock sealant and paint	700	sf	ŝ	11.00	\$ 10.00	\$ (4.00	\$ 25.00	ŝ	17,500
Acoustic insulation	700	sf	ŝ	90.00	\$ 20.00	\$ (5.00	\$ 115.00	ŝ	80,500
Acoustic louvers	144	s	ŝ	245.00	\$ 50.00	\$ 0	60.00	\$ 355.00	ŝ	51,120
Senerator w/base mounted tank	1	ea	ŝ	127,000.00	\$ 35,000.00	ŝ	800.00	\$ 162,800.00	ŝ	162,800
Conc inertia pad for generator	9	Ş	ŝ	950.00	\$ 450.00	\$ (85.00	\$ 1,485.00	ŝ	8,910
Conc fueling pad	3.5	S	Ş	750.00	\$ 225.00	\$ (12.00	\$ 987.00	ŝ	3,455
Pervious pavement road	5000	s	ŝ	50.00	\$ 30.00	\$ 0	9.00	\$ 89.00	ŝ	445,000
' Grade adjustment wall	360	ъ	ŝ	625.00	\$ 475.00	\$ 0	5.00	\$ 1,105.00	ŝ	397,800
ence	426	۳	ŝ	45.00	\$ 35.00	ŝ	5.00	\$ 85.00	ŝ	36,210
Sate	1	ea	ŝ	950.00	\$ 500.00	\$ 0	30.00	\$ 1,480.00	ŝ	1,480
2" water service	120	۳	ŝ	75.00	\$ 9.00	ŝ	2.00	\$ 86.00	ŝ	10,320
2" water meter box	1	ea	ŝ	250.00	\$ 200.00	\$ 0	15.00	\$ 465.00	ŝ	465
Vater meter	1	S						\$ 1,500.00	ŝ	1,500

General civil site work	10000	sf						\$ 20.00	ŝ	200,000	
HELCO xfrmr and fees	H	S						\$ 125,000.00	ŝ	125,000	
SCADA and security system	1	ป						\$ 95,000.00	s	95,000	
Electrical, incl instrumentation	1	ป						\$ 377,310.60	ŝ	377,311	
Adder for deeper sewers	2000	Ŧ	\$ 105.00	ŝ	235.00	Ş	125.00	\$ 465.00	ŝ	930,000	

\$ 930,000	\$ 5,906,945	\$ 1,772,083	\$ 7,679,028	
\$ 465.00	SUBTOTAL	NGENCY (30%)	TOTAL	
\$ 125.00		CONTIL		
235.00				

County Parcel Conventional Dig and Shore

DESCRIPTION	QTγ	Σ	MA	TERIAL	د	ABOR	EQUIPN	1ENT	INN	T COST	Ĺ	TOTAL
Mobilization	1	S							Ş		ŝ	
Clearing and grubbing	10,000	sf			ş	8.00	Ş	3.00	Ş	11.00	ŝ	110,000
Mud walls for water control	75	S	Ş	600.00	ş	200.00	\$ 20	00.00	Ş	1,000.00	ŝ	75,000
Excavation soft material	296	C C			s	164.71	\$ 8	5.00	ş	249.71	ŝ	73,914
Excavation rock	80	сy			ş	500.00	\$ 17	5.00	ş	675.00	ŝ	54,000
Haul Spoils	376	с			ş	13.00	Ş	8.00	Ş	21.00	ŝ	7,896
Septage pumping (keep LCC low)	4	ê	ŝ	6,000.00	Ş	1,800.00	\$ 1,50	00.00	\$ 1.	2,300.00	ŝ	49,200
Septage hauling	4	ê			\$ 14	1,400.00	\$ 3,00	00.00	\$ 1.	7,400.00	ŝ	69,600
Shoring, 0-10 ft	800	sf	ş	45.00	ŝ	200.00	\$ 12	0.00	ş	365.00	ŝ	292,000
Shoring 10-20 ft	640	sf	Ş	65.00	ş	320.00	\$ 20	0.00	ş	585.00	s	374,400
Shoring 20-35 ft, rock	720	s,	ŝ	160.00	ş	320.00	\$ 22	5.00	Ş	705.00	ŝ	507,600
Concrete mat foundation	20	Ş	Ş	750.00	ş	225.00	\$ 1	2.00	ş	987.00	ŝ	19,740
Wet well structure; 8' dia, 9" wall	31	£	Ş	500.00	ş	100.00	\$ 25	0.00	ş	850.00	s	26,350
Wet well base	1	ea	ŝ	1,750.00	ŝ	800.00	\$ 30	0.00	ŝ	2,850.00	ŝ	2,850
Wet well top slab, 12'x12'x10"thk	4	S	Ş	950.00	ş	450.00	\$ 8	5.00	\$	1,485.00	ŝ	5,940
Wet well hatch, ss and aluminum	1	ea	Ş	2,300.00	ş	500.00	\$ 9	0.00	Ş	2,890.00	ŝ	2,890
Wet well interior coating	1000	sf	Ş	85.00	ŝ	20.00	\$ 1	0.00	ş	115.00	s	115,000
Wet well exterior moisture prot	1045	sf	Ş	12.00	ş	10.00	Ş	2.00	Ş	24.00	ŝ	25,080
Engineered backfill	295	Ş	ş	47.00	ŝ	42.00	\$ 2	0.00	Ş	109.00	ŝ	32,155
Subm Pumps, 400 gpm, 80 ft TDH	2	ea	Ş	3,000.00	\$ 1	1,800.00	\$ 18	0.00	\$ 7	1,980.00	ŝ	9,960
Disch elbows and guides	2	ea	Ŷ	1,500.00	\$ 1	L,200.00	\$ 18	0.00	ŝ	2,880.00	ŝ	5,760
Valve vault structure	1	ea	ş	925.00	ŝ	250.00	\$ 18	0.00	Ş	1,355.00	ŝ	1,355
Emergency pump conn vault	1	ea	Ş	925.00	ş	250.00	\$ 18	0.00	\$ 1	1,355.00	ŝ	1,355
Gate Valve, 6"	S	ea	Ş	910.00	ş	175.00	\$ 8	5.00	Ş	1,170.00	ŝ	5,850
Check valve, 6"	2	ea	Ş	1,050.00	ş	175.00	\$ 8	5.00	\$	1,310.00	ŝ	2,620
Quick connect hose coupling	1	ea	Ş	400.00	ş	100.00	\$ 8	5.00	Ş	585.00	ŝ	585
90 bend, FExFE, 6"	3	ea	Ş	1,005.00	ş	150.00	\$ 8	5.00	\$ 1	1,240.00	ŝ	3,720
Tee, FExFE, 6"	1	ea	Ş	1,250.00	ş	225.00	\$ 8	5.00	Ş	1,560.00	ŝ	1,560

45 bend, MJ, 6"	4	ea	ŝ	935.00	\$ 150.0	\$ 0	85.00	\$ 1,170	8	ŝ	4,680
Wye, MJ, 6"	2	ea	ŝ	1,270.00	\$ 225.0	ŝ	85.00	\$ 1,580	8	ŝ	3,160
Plug, MJ, 6"	1	ea	ş	685.00	\$ 75.0	ŝ	85.00	\$ 845	8	s	845
DI pipe, FE, 6"	80	Г	ŝ	185.00	\$ 25.0	ŝ	15.00	\$ 225	8	ŝ	18,000
DI pipe, MJ, 6"	150	Г	ŝ	180.00	\$ 25.0	\$ 0	15.00	\$ 220	8	ŝ	33,000
Sewer trench excavation, soft	270	сy			\$ 25.0	ŝ	7.00	\$ 32	8	ŝ	8,640
Sewer trench excavation, rock	110	cy			\$ 175.0	s o	35.00	\$ 210	8	s	23,100
Haul Spoils	380	Q			\$ 13.0	ŝ	8.00	\$ 21	8	ŝ	7,980
Trench Shoring, 0-10 ft	2200	sf	ŝ	2.00	\$ 5.0	\$ 0	3.00	\$ 10	8	ŝ	22,000
Trench Shoring 10-20 ft	2200	sf	ş	2.50	\$ 7.0	ŝ	3.00	\$ 12	50	s	27,500
Trench Shoring 20-27 ft, rock	1540	sf	ŝ	3.50	\$ 9.0	\$ 0	5.00	\$ 17	50	ŝ	26,950
Septage pumping (keep LCC low)	1.5	mo	ŝ	6,000.00	\$ 4,800.0	ŝ	1,500.00	\$ 12,300	8	ŝ	18,450
Septage hauling	1.5	mo			\$ 14,400.0	\$ 0	3,000.00	\$ 17,400	0.	Ş	26,100
Influent sewer, 14" DI, lined	110	LF	ŝ	260.00	\$ 45.0	ŝ	25.00	\$ 330	8	ŝ	36,300
Reinf conc jkt	6.3	сy	s	950.00	\$ 350.0	ŝ	55.00	\$ 1,355	8	s	8,537
CLSM Backfill	285	S	ŝ	300.00	\$ 5.0	ŝ	10.00	\$ 315	8	ŝ	89,775
Backfill	41	cy	ŝ	20.00	\$ 42.0	s o	10.00	\$ 72	8	ŝ	2,952
Deep manhole (27 ft)	1	ea	s	7,500.00	\$ 3,300.0	\$ 0	1,500.00	\$ 12,300	0.	s	12,300
Manhole interior coating	355	sf	ŝ	85.00	\$ 20.0	\$ 0	10.00	\$ 115	8	ŝ	40,825
SMH exterior moisture protection	410	sf	ŝ	12.00	\$ 10.0	s o	2.00	\$ 24	8	s	9,840
						_					
Odor control unit	1	ea	ŝ	7,150.00	\$ 6,000.0	\$ 0	400.00	\$ 13,550	0.	ŝ	13,550
Generator Bldg	320	sf	ŝ	250.00	\$ 265.0	\$ 0	10.00	\$ 525	8	ŝ	168,000
Roofing membrane	5.8	sq	ŝ	3,000.00	\$ 2,000.0	\$ 0	170.00	\$ 5,170	0.	ŝ	29,986
Block sealant and paint	700	sf	ŝ	11.00	\$ 10.0	\$ 0	4.00	\$ 25	0.	ŝ	17,500
Acoustic insulation	700	sf	ŝ	90.00	\$ 20.0	\$ 0	5.00	\$ 115	8	ŝ	80,500
Acoustic louvers	144	sf	ŝ	245.00	\$ 50.0	\$ 0	60.00	\$ 355	8	ŝ	51,120
Generator w/base mounted tank	1	ea	ŝ	127,000.00	\$ 35,000.0	\$ 0	800.00	\$ 162,800	0	ŝ	162,800
Conc inertia pad for generator	9	сy	ŝ	950.00	\$ 450.0	ŝ	85.00	\$ 1,485	8	ŝ	8,910
Conc fueling pad	3.5	cy	ŝ	750.00	\$ 225.0	s o	12.00	\$ 987	8	ŝ	3,455
Pervious pavement road	5000	sf	ŝ	50.00	\$ 30.0	\$ 0	9.00	\$ 89	8	ŝ	145,000
3' Grade adjustment wall	360	LF	ŝ	625.00	\$ 475.0	\$ 0	5.00	\$ 1,105	8	ŝ	397,800
Fence	426	LF	ŝ	45.00	\$ 35.0	\$ 0	5.00	\$ 85	8	s	36,210

Gate	1	ea	ŝ	950.00	Ş	500.00	ŝ	30.00	\$ 1	,480.00	ŝ	1,480
2" water service	120	5	ş	75.00	Ş	9.00	ş	2.00	ş	86.00	ş	10,320
2" water meter box	1	ea	ş	250.00	ş	200.00	ş	15.00	ş	465.00	ş	465
Water meter	1	S							\$ 1	.,500.00	ŝ	1,500
General civil site work	10000	sf							ş	20.00	s	200,000
HELCO xfrmr and fees	1	LS							\$ 125	,000.000	ş	125,000
SCADA and security system	1	S							\$ 95	,000.00	ŝ	95,000
Electrical, incl instrumentation	1	S							\$ 375	,844.85	s	379,845
Adder for deeper sewers	2000	ff	Ş	105.00	Ş	235.00	Ş	125.00	Ş	465.00	Ş	930,000
									SU	BTOTAL	\$ 5,	455,754
								CONTIN	NGENC	.Y (30%)	\$ 1,	636,726
										TOTAL	\$ 7,	092,480

DESCRIPTION	QTY	MU	MATERIAL	LABOR	EQUIPMENT	UNIT COST	TOTAL
Mobilization	1	ป				\$800,000.00	\$800,000
Clearing and grubbing	8,000	sf		\$8.00	\$3.00	\$11.00	\$88,000
Machine setup	1	ป				\$60,000.00	\$60,000
Wet well shaft	18	¥	\$15,000.00	\$12,000.00	\$3,000.00	\$30,000.00	\$540,000
Haul Spoils	103	S		\$13.00	\$8.00	\$21.00	\$2,163
Concrete mat foundation	20	2	\$750.00	\$225.00	\$12.00	\$987.00	\$19,740
Wet well structure; 8' dia, 9" wall	15	æ	\$500.00	\$100.00	\$250.00	\$850.00	\$12,750
Wet well base	1	ea	\$1,750.00	\$800.00	\$300.00	\$2,850.00	\$2,850
Wet well top slab, 12'x12'x10"thk	4	2	\$950.00	\$450.00	\$85.00	\$1,485.00	\$5,940
Wet well hatch, ss and aluminum	1	ea	\$2,300.00	\$500.00	\$90.00	\$2,890.00	\$2,890
Wet well interior coating	500	sf	\$85.00	\$20.00	\$10.00	\$115.00	\$57,500
Wet well exterior moisture prot	440	sf	\$12.00	\$10.00	\$2.00	\$24.00	\$10,560
Engineered backfill	64	S	\$47.00	\$42.00	\$20.00	\$109.00	\$6,976
Subm Pumps, 400 gpm, 80 ft TDH	2	ea	\$3,000.00	\$1,800.00	\$180.00	\$4,980.00	\$9,960
Disch elbows and guides	2	ea	\$1,500.00	\$1,200.00	\$180.00	\$2,880.00	\$5,760
Valve vault structure	1	ea	\$925.00	\$250.00	\$180.00	\$1,355.00	\$1,355
Emergency pump conn vault	1	ea	\$925.00	\$250.00	\$180.00	\$1,355.00	\$1,355
Gate Valve, 6"	ŝ	ea	\$910.00	\$175.00	\$85.00	\$1,170.00	\$5,850
Check valve, 6"	2	ea	\$1,050.00	\$175.00	\$85.00	\$1,310.00	\$2,620
Quick connect hose coupling	1	ea	\$400.00	\$100.00	\$85.00	\$585.00	\$585
90 bend, FExFE, 6"	ŝ	ea	\$1,005.00	\$150.00	\$85.00	\$1,240.00	\$3,720
Tee, FExFE, 6"	1	ea	\$1,250.00	\$225.00	\$85.00	\$1,560.00	\$1,560
45 bend, MJ, 6"	4	ea	\$935.00	\$150.00	\$85.00	\$1,170.00	\$4,680
Wye, MJ, 6"	2	ea	\$1,270.00	\$225.00	\$85.00	\$1,580.00	\$3,160
Plug, MJ, 6"	1	ea	\$685.00	\$75.00	\$85.00	\$845.00	\$845
DI pipe, FE, 6"	80	5	\$185.00	\$25.00	\$15.00	\$225.00	\$18,000
DI pipe, MJ, 6"	150	ъ	\$180.00	\$25.00	\$15.00	\$220.00	\$33,000

Gate	1	ea		\$950.00		\$500.00		\$30.00	ŝ	1,480.00		\$1,480
6" water service	500	5	ŝ	120.00	ŝ	25.00	ş	15.00	ş	160.00	ŝ	80,000
6" detector check in vault	1	ea	ŝ	8,000.00	s	6,000.00	\$ 1	000.000	\$ 15	,000.000	ŝ	15,000
Fire standpipe	1	ea	ŝ	1,500.00	ŝ	900.006	ş	20.00	\$ 2	,420.00	ŝ	2,420
2" water service	120	5	ŝ	75.00	s	9.00	s	2.00	s	86.00	ŝ	10,320
2" water meter box	1	ea	ŝ	250.00	s	200.00	Ş	15.00	Ş	465.00	ŝ	465
Water meter	1	പ							\$ 1	,500.00	ŝ	1,500
General civil site work	8000	sf								\$20.00		\$160,000
HELCO xfmr and fees	1	ป							\$22	5,000.00		\$225,000
SCADA and security system	1	ป							\$9	5,000.00		\$95,000
Electrical, incl instrumentation	1	ป							\$40	5,364.33		\$406,364
Additional gravel road	17,000	sf		\$1.75		\$1.60		\$0.75		\$4.10		\$69,700
									S	JBTOTAL	ŵ	1,675,323
								CONTI	NGEN	CY (30%)	ŝ	1,402,597
						•						

TOTAL \$6,077,920

	Ś	2		_	\$ 25.00	ŝ	7.00	v	32.00	ĥ	004/77
Sewer trench excavation, rock	0	S			\$ 175.00	ŝ	35.00	ŝ	210.00	ŝ	
Haul Spoils	700	S			\$ 13.00	ŝ	8.00	ŝ	21.00	ŝ	14,700
Trench Shoring, 0-10 ft	9400	sf	\$ 2	8	\$ 5.00	ŝ	3.00	ŝ	10.00	ŝ	94,000
Trench Shoring 10-20 ft	1880	sf	\$ 2	50	\$ 7.00	ŝ	3.00	ŝ	12.50	s	23,500
Trench Shoring 20-27 ft, rock	0	sf	\$	50	\$ 9.00	ŝ	5.00	ŝ	17.50	s	
Septage pumping (keep LCC low)	0	0 m	\$ 6,000	8	\$ 4,800.00	ŝ	1,500.00	ŝ	12,300.00	ŝ	•
Septage hauling	0	Ê			\$ 14,400.00	ŝ	3,000.00	ŝ	17,400.00	ŝ	
Influent sewer, 14" DI, lined	470	5	\$ 260	00.	\$ 45.00	ŝ	25.00	ŝ	330.00	ŝ	155,100
Reinf conc jkt	0	2	\$ 950	8	\$ 350.00	ŝ	55.00	ŝ	1,355.00	ŝ	
CLSM Backfill	0	2	\$ 300	8	\$ 5.00	ŝ	10.00	ŝ	315.00	ŝ	
Engineered Backfill	41	2	\$ 47	0.	\$ 42.00	ŝ	20.00	ŝ	109.00	ŝ	4,469
Manhole (10 ft)	m	ea	\$ 3,900	8	\$ 1,800.00	ŝ	600.009	ŝ	6,300.00	ŝ	18,900
Manhole interior coating	180	sf	\$ 85	8	\$ 20.00	ŝ	10.00	ŝ	115.00	ŝ	20,700
SMH exterior moisture protection	152	sf	\$ 12	8	\$ 10.00	ŝ	2.00	ŝ	24.00	ŝ	3,648
SFM trench excavation, soft	0	2			\$ 25.00	ŝ	7.00	ŝ	32.00	ŝ	
Haul Spoils	0	2			\$ 13.00	ŝ	8.00	ŝ	21.00	ŝ	
Trench Shoring, 0-10 ft	0	sf	\$ 2.	8	\$ 5.00	ŝ	3.00	ŝ	10.00	s	
SFM, 6" DI, coated and lined	0	ч	\$ 180	8	\$ 25.00	ŝ	15.00	ŝ	220.00	ŝ	
Engineered Backfill	0	2	\$ 47	8	\$ 42.00	ŝ	20.00	ŝ	109.00	ŝ	
Odor control unit	1	ea	\$7,150	0.00	\$6,000.00		\$400.00		\$13,550.00		\$13,550
Generator Bldg	520	sf	\$250	00.0	\$265.00		\$10.00		\$525.00		\$273,000
Roofing membrane	8.4	sq	\$3,000	0.00	\$2,000.00		\$170.00		\$5,170.00		\$43,428
Block sealant and paint	1105	sf	\$11	1.00	\$10.00		\$4.00		\$25.00		\$27,625
Acoustic insulation	700	sf	\$90	0.0	\$20.00		\$5.00		\$115.00		\$80,500
Acoustic louvers	144	sf	\$245	00.5	\$50.00		\$60.00		\$355.00		\$51,120
Generator w/base mounted tank	1	ea	\$127,000	0.00	\$35,000.00		\$800.00	ŝ	162,800.00		\$162,800
Conc inertia pad for generator	9	S	\$95C	0.00	\$450.00		\$85.00		\$1,485.00		\$8,910
Conc fueling pad	3.5	c	\$750	00.0	\$225.00		\$12.00		\$987.00		\$3,455
Pervious pavement road	5000	sf	\$ 50	8	\$ 30.00	ŝ	9.00		\$89.00		\$445,000
3' Grade adjustment wall	370	5	\$ 625	0.0	\$ 475.00	ŝ	5.00		\$1,105.00		\$408,850
Fence	360	٤	\$45	00.0	\$35.00		\$5.00		\$85.00		\$30,600

Kuahiwi Contractors Site, Minimum Conventional Dig and Shore

DESCRIPTION	ΔT	Σ	MA	TERIAL	-	LABOR	EQUI	PMENT	INN	T COST	Ĺ	TOTAL
Mobilization	1	S							Ş		ŝ	
Clearing and grubbing	8,000	sf			ŝ	8.00	ş	3.00	ş	11.00	Ş	88,000
Excavation soft material	175	сy			ŝ	164.71	Ş	85.00	Ş	249.71	Ŷ	43,699
Excavation rock	0	с			ŝ	500.00	ş	175.00	ŝ	675.00	ŝ	
Haul Spoils	175	Ş			ŝ	13.00	ŝ	8.00	ş	21.00	ŝ	3,675
Shoring, 0-10 ft	640	s	ş	45.00	ŝ	200.00	ş	120.00	ş	365.00	ŝ	233,600
Shoring 10-18 ft	384	s	Ş	65.00	ŝ	320.00	ŝ	200.00	ŝ	585.00	ŝ	224,640
Shoring 20-35 ft, rock	0	sf	Ş	160.00	ŝ	320.00	ş	225.00	ş	705.00	ŝ	
Concrete mat foundation	20	2	Ş	750.00	ŝ	225.00	ş	12.00	ş	987.00	ŝ	19,740
Wet well structure; 8' dia, 9" wall	15	Ĥ	Ş	500.00	s	100.00	ş	250.00	Ş	850.00	Ş	12,750
Wet well base	1	ea	Ş	1,750.00	ŝ	800.00	ş	300.00	ŝ	2,850.00	ŝ	2,850
Wet well top slab, 12'x12'x10"thk	4	2	Ş	950.00	ŝ	450.00	ş	85.00	ŝ	1,485.00	ŝ	5,940
Wet well hatch, ss and aluminum	1	ea	Ş	2,300.00	s	500.00	ş	90.00	ŝ	2,890.00	Ş	2,890
Wet well interior coating	480	sf	ş	85.00	ŝ	20.00	ŝ	10.00	ŝ	115.00	ŝ	55,200
Wet well exterior moisture prot	445	sf	Ş	12.00	ŝ	10.00	Ş	2.00	Ş	24.00	ŝ	10,680
Engineered backfill	136	ç	Ş	47.00	ŝ	42.00	Ş	20.00	Ş	109.00	ŝ	14,824
Subm Pumps, 400 gpm, 80 ft TDH	2	ea	Ş	3,000.00	ŝ	1,800.00	ş	180.00	ŝ	4,980.00	ŝ	9,960
Disch elbows and guides	2	ea	Ş	1,500.00	ŝ	1,200.00	Ş	180.00	Ş	2,880.00	ŝ	5,760
Valve vault structure	1	ea	Ş	925.00	ŝ	250.00	ş	180.00	ŝ	1,355.00	ŝ	1,355
Emergency pump conn vault	1	ea	Ş	925.00	ŝ	250.00	Ş	180.00	Ş	1,355.00	ŝ	1,355
Gate Valve, 6"	0	ea	ş	910.00	ŝ	175.00	ŝ	85.00	ŝ	1,170.00	ŝ	5,850
Check valve, 6"	2	ea	Ş	1,050.00	ŝ	175.00	ş	85.00	ŝ	1,310.00	ŝ	2,620
Quick connect hose coupling	1	ea	Ş	400.00	ŝ	100.00	Ş	85.00	Ş	585.00	ŝ	585
90 bend, FExFE, 6"	m	ea	Ş	1,005.00	ŝ	150.00	ş	85.00	ŝ	1,240.00	ŝ	3,720
Tee, FExFE, 6"	1	ea	Ş	1,250.00	ŝ	225.00	Ş	85.00	Ş	1,560.00	ŝ	1,560
45 bend, MJ, 6"	4	ea	Ş	935.00	ŝ	150.00	Ş	85.00	ŝ	1,170.00	ŝ	4,680
Wye, MJ, 6"	2	ea	ş	1,270.00	ŝ	225.00	ŝ	85.00	ŝ	1,580.00	ŝ	3,160
Plug, MJ, 6"	1	ea	Ş	685.00	ŝ	75.00	ş	85.00	ş	845.00	ŝ	845

DI pipe, FE, 6"	80	ч	Ş	185.00	ŝ	25.00	ŝ	15.00	ŝ	225.00	ŝ	18,000
DI pipe, MJ, 6"	150	ъ	ş	180.00	ŝ	25.00	ŝ	15.00	ŝ	220.00	ŝ	33,00
Sewer trench excavation, soft	700	5			ŝ	25.00	ş	7.00	ŝ	32.00	ŝ	22,400
Sewer trench excavation, rock	0	2			ŝ	175.00	ŝ	35.00	s	210.00	ŝ	1
Haul Spoils	700	Ş			ŝ	13.00	Ş	8.00	ş	21.00	s	14,700
Trench Shoring, 0-10 ft	9400	s	ş	2.00	ŝ	5.00	ş	3.00	ş	10.00	s	94,000
Trench Shoring 10-20 ft	1880	sf	ş	2.50	ŝ	7.00	ş	3.00	s	12.50	ŝ	23,500
Trench Shoring 20-27 ft, rock	0	sf	ŝ	3.50	ŝ	9.00	ş	5.00	ŝ	17.50	ŝ	1
Septage pumping (keep LCC low)	0	ĥ	s	6,000.00	ŝ	4,800.00	\$ 1,5	00.00	\$ 12,	300.00	s	1
Septage hauling	0	0 m			\$ 1	4,400.00	\$ 3,01	00.00	\$ 17,	400.00	ŝ	•
Influent sewer, 14" DI, lined	470	5	ş	260.00	ŝ	45.00	ŝ	25.00	ŝ	330.00	ŝ	155,100
Reinf conc jkt	0	S	s	950.00	ŝ	350.00	ş	55.00	\$ 1,	355.00	ŝ	•
CLSM Backfill	0	Ş	ş	300.00	ŝ	5.00	ş	10.00	ŝ	315.00	ŝ	•
Engineered Backfill	41	S	ŝ	47.00	ŝ	42.00	ŝ	20.00	ŝ	109.00	ŝ	4,469
Manhole (10 ft)	m	ea	ş	3,900.00	ŝ	1,800.00	\$ 6	00.00	\$ 6'	300.00	s	18,900
Manhole interior coating	180	sf	ş	85.00	ŝ	20.00	Ş	10.00	ŝ	115.00	ŝ	20,700
SMH exterior moisture protection	152	s	ŝ	12.00	ŝ	10.00	ş	2.00	ŝ	24.00	s	3,648
SFM trench excavation, soft	0	Ş			ŝ	25.00	ş	7.00	ş	32.00	ŝ	1
Haul Spoils	0	S			ŝ	13.00	ş	8.00	ş	21.00	s	1
Trench Shoring, 0-10 ft	0	sf	ŝ	2.00	ŝ	5.00	Ş	3.00	Ş	10.00	ŝ	1
SFM, 6" DI, coated and lined	0	5	s	180.00	ŝ	25.00	s	15.00	ŝ	220.00	s	1
Engineered Backfill	0	ç	ş	47.00	Ş	42.00	Ş	20.00	ş	109.00	s	1
Odor control unit	1	ea	ŝ	7,150.00	ŝ	6,000.00	\$ 4	00.00	\$ 13,	550.00	ŝ	13,550
Generator Bldg	520	sf	ŝ	250.00	Ş	265.00	ş	10.00	ŝ	525.00	ŝ	273,000
Roofing membrane	8.4	ß	ŝ	3,000.00	ŝ	2,000.00	\$	70.00	ŝ	170.00	ŝ	43,428
Block sealant and paint	1105	sf	ş	11.00	ŝ	10.00	ş	4.00	ş	25.00	s	27,625
Acoustic insulation	700	sf	ŝ	90.06	ŝ	20.00	ş	5.00	ŝ	115.00	ŝ	80,500
Acoustic louvers	144	s	ş	245.00	ŝ	50.00	ş	60.00	ŝ	355.00	ŝ	51,120
Generator w/base mounted tank	1	ea	Ş	27,000.00	\$ 3	5,000.00	\$ 8	00.00	\$ 162,8	300.00	s	162,800
Conc inertia pad for generator	9	Ş	ŝ	950.00	ŝ	450.00	ŝ	85.00	\$ 1'	485.00	ŝ	8,910
Conc fueling pad	3.5	ç	Ş	750.00	ŝ	225.00	ş	12.00	ŝ	987.00	ŝ	3,455

Pervious pavement road	5000	sf	Ş	50.00	ŝ	30.00	Ş	9.00		89.00		\$445,000
3' Grade adjustment wall	370	5	Ş	625.00	ŝ	475.00	Ş	5.00	\$1,1	105.00		\$408,850
Fence	360	۳	ş	45.00	ŝ	35.00	Ş	5.00	ş	85.00	ŝ	30,600
Gate	1	ea	ŝ	950.00	ŝ	500.00	ş	30.00	\$ 1,4	80.00	ŝ	1,480
6" water service	500	5	ş	120.00	ŝ	25.00	ŝ	15.00	\$ 1	60.00	ŝ	80,000
6" detector check in vault	1	ea	Ş	8,000.00	s	6,000.00	\$ 1,00	00.00	\$ 15,0	00.00	s	15,000
Fire standpipe	1	ea	Ş	1,500.00	ŝ	900.006	ŝ	20.00	\$ 2,4	20.00	ŝ	2,420
2" water service	120	5	ş	75.00	ŝ	9.00	Ş	2.00	ŝ	86.00	ŝ	10,320
2" water meter box	1	ea	ŝ	250.00	ŝ	200.00	ŝ	L5.00	\$ 4	65.00	ŝ	465
Water meter	1	S							\$ 1,5	00.00	ŝ	1,500
General civil site work	8000	sf							ŝ	20.00	ŝ	160,000
HELCO xfrmr and fees	1	S							\$ 225,0	00.00	ŝ	225,000
SCADA and security system	1	S							\$ 95,0	00.00	ŝ	95,000
Electrical, incl instrumentation	1	LS							\$ 407,2	14.53	s	407,215
Additional gravel road	17,000	sf	Ş	1.75	ŝ	1.60	Ş	0.75	Ş	4.10	Ş	69,700
									SUB.	TOTAL	\$ 3	,785,292

\$ 3,785,292	\$ 1,135,588	\$ 4,920,880	
SUBTOTAL	CONTINGENCY (30%)	TOTAL	

Kuahiwi Contractors Site, Large

DESCRIPTION	QTY	ΜŊ	MATERIAL	LABOR	EQUIPMENT	UNIT COST	TOT	AL			
Mobilization	1	รา				\$0.00	Ş	÷			
Clearing and grubbing	8,000	sf		\$8.00	\$3.00	\$11.00	\$ 88	8,000			
Excavation soft material	175	Ş		\$164.71	\$85.00	\$249.71	\$ 43	3,699			
Excavation rock	0	Ş		\$500.00	\$175.00	\$675.00	Ş				
Haul Spoils	175	Ş		\$13.00	\$8.00	\$21.00	ş	3,675			
Shoring, 0-10 ft	640	sf	\$45.00	\$200.00	\$120.00	\$365.00	\$ 233	3,600			
Shoring 10-18 ft	384	sf	\$65.00	\$320.00	\$200.00	\$585.00	\$ 224	4,640			
Shoring 20-35 ft, rock	0	sf	\$160.00	\$320.00	\$225.00	\$705.00	Ş				
Concrete mat foundation	20	Ş	\$750.00	\$225.00	\$12.00	\$987.00	\$ 15	9,740			
Wet well structure; 8' dia, 9" wall	15	£	\$500.00	\$100.00	\$250.00	\$850.00	\$ 12	2,750			
Wet well base	1	ea	\$1,750.00	\$800.00	\$300.00	\$2,850.00	ş	2,850			
Wet well top slab, 12'x12'x10"thk	4	Ş	\$950.00	\$450.00	\$85.00	\$1,485.00	Ş	5,940			
Wet well hatch, ss and aluminum	1	ea	\$2,300.00	\$500.00	00 [.] 06\$	\$2,890.00	Ş	2,890			
Wet well interior coating	480	sf	\$85.00	\$20.00	\$10.00	\$115.00	\$ 56	5,200			
Wet well exterior moisture prot	445	sf	\$12.00	\$10.00	\$2.00	\$24.00	\$ 10	0,680			
Engineered backfill	136	Ş	\$47.00	\$42.00	\$20.00	\$109.00	\$ 1/	4,824			
Subm Pumps, 400 gpm, 80 ft TDH	2	ea	\$3,000.00	\$1,800.00	\$180.00	\$4,980.00	Ş	9,960			
Disch elbows and guides	2	ea	\$1,500.00	\$1,200.00	\$180.00	\$2,880.00	ş	5,760			
Valve vault structure	1	ea	\$925.00	\$250.00	\$180.00	\$1,355.00	Ş	1,355			
Emergency pump conn vault	1	ea	\$925.00	\$250.00	\$180.00	\$1,355.00	Ş	1,355			
Gate Valve, 6"	2	ea	\$910.00	\$175.00	\$85.00	\$1,170.00	ş	5,850			
Check valve, 6"	2	ea	\$1,050.00	\$175.00	\$85.00	\$1,310.00	Ş	2,620			
Quick connect hose coupling	1	ea	\$400.00	\$100.00	\$85.00	\$585.00	ş	585			
90 bend, FExFE, 6"	33	ea	\$1,005.00	\$150.00	\$85.00	\$1,240.00	ş	3,720			
Tee, FExFE, 6"	1	ea	\$1,250.00	\$225.00	\$85.00	\$1,560.00	Ş	1,560			
45 bend, MJ, 6"	4	ea	\$935.00	\$150.00	\$85.00	\$1,170.00	\$	4,680			
Wye, MJ, 6"	2	ea	\$1,270.00	\$225.00	\$85.00	\$1,580.00	ş	3,160			
Plug, MJ, 6"	1	ea	\$685.00	\$75.00	\$85.00	\$845.00	Ş	845			
DI pipe, FE, 6"	80	5	\$185.00	\$25.00	\$15.00	\$225.00	\$ 1E	8,000			
3' Grade adjustment wall	484	5	ŝ	625.00	ŝ	475.00	Ş	5.00	\$1,105.00		\$534,820
----------------------------------	--------	----	---	----------	---	----------	--------	--------	--------------	------	-----------
Fence	480	5		\$45.00		\$35.00		\$5.00	\$85.00	ŝ	40,800
Gate	1	ea		\$950.00		\$500.00		30.00	\$1,480.00	ŝ	1,480
6" water service	500	ъ	ŝ	120.00	ŝ	25.00	Ş	15.00	\$ 160.00	ŝ	80,000
6" detector check in vault	1	ea	ŝ	8,000.00	s	6,000.00	\$ 1,0	00.00	\$ 15,000.00	ŝ	15,000
Fire standpipe	1	ea	ŝ	1,500.00	s	900.006	ş	20.00	\$ 2,420.00	ŝ	2,420
2" water service	120	ъ	ŝ	75.00	ŝ	9.00	Ş	2.00	\$ 86.00	ŝ	10,320
2" water meter box	1	ea	ş	250.00	ŝ	200.00	ŝ	15.00	\$ 465.00	ŝ	465
Water meter	1	പ							\$ 1,500.00	ŝ	1,500
General civil site work	15000	£							\$20.00	ŝ	300,000
HELCO xfrmr and fees	1	ম							\$225,000.00	ŝ	225,000
SCADA and security system	1	പ							\$95,000.00	ŝ	95,000
Electrical, incl instrumentation	1	പ							\$475,340.03	ŝ	475,340
Additional gravel road	17,000	sf		\$1.75		\$1.60		\$0.75	\$4.10	Ş	69,700
									SUBTOTAL	\$ 4	1,307,588
								CONTI	NGENCY (30%)	\$1	.,292,276
									TOTAL	ŝ	,599,864

Di sino Mil 6"	100	-		¢1 00 00	Ş	2	¢1E OC		00 0CC2	v	000 0 0
	PCT	5		DO:DOTC	¢.	8	DO'CTC		00.0220	'n	nnn'ee
Sewer trench excavation, soft	700	δ			\$ 2!	00.5	7.00	Ş	32.00	Ş	22,400
Sewer trench excavation, rock	0	Ş			\$ 179	5:00	35.00	Ŷ	210.00	ŝ	•
Haul Spoils	700	ç			\$ 10	00.8	8.00	ŝ	21.00	s	14,700
Trench Shoring, 0-10 ft	9400	sf	Ş	2.00	ş	00.5	3.00	Ş	10.00	ş	94,000
Trench Shoring 10-20 ft	1880	sf	ŝ	2.50	ŝ	00.7	3.00	ŝ	12.50	ŝ	23,500
Trench Shoring 20-27 ft, rock	0	s	ş	3.50	\$.	00.6	5.00	ŝ	17.50	ŝ	•
Septage pumping (keep LCC low)	0	ош	Ŷ	6,000.00	\$ 4,800	00.0	\$ 1,500.00	\$	2,300.00	ŝ	•
Septage hauling	0	ĥ			\$ 14,400	00.0	3,000.00	\$ 1.	7,400.00	ŝ	•
Influent sewer, 14" DI, lined	470	٤	Ş	260.00	\$ 4!	00.5	\$ 25.00	ŝ	330.00	ŝ	155,100
Reinf conc jkt	0	2	Ş	950.00	\$ 35(00.0	55.00	ŝ	1,355.00	ŝ	•
CLSM Backfill	0	2	ŝ	300.00	ş	00.5	10.00	ŝ	315.00	ŝ	•
Engineered Backfill	41	S	ŝ	47.00	\$	00.2	\$ 20.00	ŝ	109.00	ŝ	4,469
Manhole (10 ft)	e	ea	ŝ	3,900.00	\$ 1,800	00.0	\$ 600.00	ŝ	6,300.00	ŝ	18,900
Manhole interior coating	180	sf	ŝ	85.00	\$ 2(00.0	3 10.00	ş	115.00	s	20,700
SMH exterior moisture protection	152	sf	ŝ	12.00	\$ 1(00.0	\$ 2.00	ŝ	24.00	ŝ	3,648
SFM trench excavation, soft	0	2			\$ 2!	00.9	5 7.00	ŝ	32.00	ş	
Haul Spoils	0	S			\$ 10	3.00	8.00	ŝ	21.00	ŝ	•
Trench Shoring, 0-10 ft	0	sf	Ş	2.00	ş	00.5	3.00	ŝ	10.00	s	•
SFM, 6" DI, coated and lined	0	5	Ŷ	180.00	\$ 25	00.5	\$ 15.00	ŝ	220.00	ŝ	•
Engineered Backfill	0	2	ş	47.00	\$	00.2	20.00	ŝ	109.00	ŝ	•
Odor control unit	1	ea		\$7,150.00	\$6,0C	00.00	\$400.0I	5 51	13,550.00	ŝ	13,550
Generator Bldg	520	sf		\$250.00	\$26	5.00	\$10.0	0	\$525.00	ŝ	273,000
Roofing membrane	8.4	sq		\$3,000.00	\$2,00	00.00	\$170.0I	ŝ	\$5,170.00	ŝ	43,428
Block sealant and paint	1105	sf		\$11.00	\$1	0.00	\$4.0I	0	\$25.00	ŝ	27,625
Acoustic insulation	700	sf		\$90.00	\$2	00.00	\$5.0I	0	\$115.00	ŝ	80,500
Acoustic louvers	144	sf		\$245.00	\$5	0.00	\$60.0I	0	\$355.00	ŝ	51,120
Generator w/base mounted tank	1	ea		\$127,000.00	\$35,00	00.00	\$800.0	0 \$16	52,800.00	ŝ	162,800
Conc inertia pad for generator	9	c		\$950.00	\$45	0.00	\$85.0I	ŝ	\$1,485.00	ŝ	8,910
Conc fueling pad	3.5	Ş		\$750.00	\$22	5.00	\$12.0I	0	\$987.00	ŝ	3,455
Pervious pavement road	7000	sf	ŝ	50.00	\$ 3(00.0	9.00		\$89.00		\$623,000

PART B Nāʻālehu Individual Wastewater System Preliminary Engineering Report	Prepared for Brown and Caldwell & County of Hawai'i, Department of Environmental Management August 2023	THIS WORK (PART B) WAS PREPARED BY ME OR UNDER MY SUPERVISION	Yen Wen Fang
	PART B: IWS Approach		

Part B - IWS Approach

Table of Contents

÷	Intr	oduction7
2.	Diff	erent Types of IWS7
	2.1	Septic System7
	2.2	Residential Aerobic Treatment Unit (ATU)9
	2.3	Intermittent Sand Filter (ISF)11
	2.4	Passive Treatment Beds13
	2.5	Composting Toilet16
	2.6	Incineration Toilet18
	2.7	Greywater Reuse
ы.	Typ€	ss of IWS Disposal Systems20
	3.1	Absorption Bed
	3.2	Leach Field21
	3.3	Seepage Pits22
	3.4	Subsurface Drip Irrigation
	3.5	Evaporation Beds24
	3.6	Mound Septic System25
4	Trac	litional IWS Recommendation
ъ.	Trac	litional IWS Components28
	5.1	Types of IWS Tanks
9.	Des	ign of IWS: DOH Chapter 11-62 Guidelines31
	6.1	Percolation Testing31
	6.2	Sizing of Absorption Beds and Disposal Systems
	6.3	DOH Required Setbacks32
	6.4	DOH Variance
7.	Typi	cal IWS Permitting and Construction Process
	7.1	Project Schedule
ø.	IWS (Constructability Challenges & Solutions38
	8.1	Lack of Yard Space
	8.2	Landscaping
	8.3	Ground Slope
	8.4	Cut Slope

1. Economic Considerations52	Ч
0. Operational Considerations	÷
9. Technically Challenging Sites	01
8.13 Availability of Resources and Contractors	
8.12 Absorption Beds vs. Seepage Pits	
8.11 Coordination with Landowners or Tenants	
8.10 Access of Maintenance Equipment 43	
8.9 Access of Large Construction Equipment	
8.8 Existing Building Structures42	
8.7 Traffic Area	
8.6 Soil Permeability: Percolation Test Results	
8.5 Site Geology	

References
Appendices
Appendix A - LCC Closure Properties A-1
Appendix B - Cost Calculations A-5
topendix C - TopographyA-10
Appendix D - USGS Soil Survey A-12
Appendix E - Geotech Report A-14
Appendix F - Typical IWS Layout & ComponentsA-23
Appendix G – Lava Tube Backfill Detail A-31
Appendix H – Percolation Test ResultsA-33

List of Figures

)
Figure 1.1: Typical IWS flow diagram7
Figure 1.2: Side-view of a typical two-chambered septic tank
Figure 1.3: Side-view of typical ATU10
Figure 1.4: Effluent quality of septic systems vs. ATUs in Hawaifi
Figure 1.5: Pumped discharge ISF following a septic tank1
Figure 1.6: Generic, open intermittent sand filter12
Figure 1.7: Schematic of Presby's Enviro-Fin Passive Treatment Bed
Figure 1.8: AES, Enviro-Septic, and Simple Septic14
Figure 1.9: Component view of a typical individual composting toilet
Figure 1.10: Top view of a typical urine separating toilet16
Figure 1.11: Top view of a central composting unit17
Figure 1.12: Side view of a typical incinerating toilet18
Figure 1.13: Typical greywater reuse system installed in parallel with septic tank19
Figure 1.14: Typical absorption filed installed following a septic tank
Figure 1.15: Leach field schematic21
Figure 1.16: Absorption Trenches Diagram22
Figure 1.17: Typical subsurface drip irrigation installed following a septic tank24
Figure 1.18: Evapotranspiration (ET) bed25
Figure 1.19: Elevated Mound System26
Figure 1.20: Common septic tank materials and shapes in Hawaii
Figure 1.21: HAR 11-62 Table III Absorption bed sizing table
Figure 1.22: Existing Large Capacity Cesspools
Figure 1.23: EPA Manual recommendation for placement of SWIS
Figure 1.24: Boring hole locations in Na'alehu40
Figure 1.25: Percolation test locations in Na'alehu41
Figure 1.26: Problematic sites45

es
P
പ
ſ
Ö
S
Ë

able 1.1: Typical septic system performance in Hawai'i
able 1.2: Advantages and Disadvantages of Septic Tank Materials
able 1.3: Common single family home septic tank products in Hawaii
able 1.4: Septic sizing guide per DOH31
able 1.5: DOH required setbacks for wastewater systems per HAR 11-62
able 1.6: Hawai'i Permitting and Construction Process
able 1.7: Na'alehu LCC Replacement Schedule37
able 1.8: Percolation test results42
able 1.9: Na'alehu Problematic Sites46
able 1.10: The five management models for IWS maintenance
able 1.11: County and homeowner responsibilities under variations of the EPA
able 1.12: Installation cost estimates52
able 1.13: Costs associated with four IWS management models53

Individual Wastewater System (IWS)

1. Introduction

In Hawai'i, Individual Wastewater System (IWS) is defined by the Department of Health (DOH) as a wastewater system for an individual property that receives less than 1,000 gallons per day of wastewater flow or serves five bedrooms or less. An IWS can be a cesspool, a traditional septic system, an aerobic treatment unit (ATU), or other means of treatment achieving National Sanitation Foundation (NSF) 40 quality effluent. Since 2016, the State of Hawai'i banned the use of new cesspools due to their poor treatment quality. Both ATUs and NSF 40 type systems are considered secondary treatment and will result in higher operating and maintenance requirements.

2. Different Types of IWS

2.1 Septic System

A typical IWS system consists of three main components: the septic tank, distribution box, and leach field (also known as absorption bed or drain field). The septic tank is typically buried underground and receives wastewater from the building's plumbing fixtures (Figure 1.1). From the septic tank which has one or two compartments, effluent flows to a distribution box, which evenly distributes the wastewater to the leach field. The leach field or absorption bed is a network of perforated pipes or chambers laid in trenches or beds, buried in the soil. A seepage pit, which is a vertical excavation typically lined with concrete perforated rings, can be used in place of an absorption bed when a variance application has been approved by the DOH. Typical IWS layout with an absorption bed, seepage pit etc. is outlined in Appendix F.



Figure 1.1: Typical IWS flow diagram.



Report	
ngineering	
reliminary E	
r System P	
Wastewate	
Individual	
Nā'ālehu	



Figure 1.2: Side-view of a typical two-chambered septic tank (Carollo, 2021).

tank, solid waste settles to the bottom forming sludge, while lighter materials like grease and as a primary settling chamber to remove bulk solids and floatable substances such as oil and The underground septic tank is typically made of concrete, fiberglass, or plastic, and serves grease. Septic tanks usually have an inlet pipe where wastewater enters and an outlet pipe through which partially treated effluent flows out to the distribution box. Inside the septic oils float to the top as scum. The middle layer consists of partially clarified effluent.

The size of the septic tank is determined based on the number of bedrooms in the house or the daily wastewater flow rate. In Hawaii, typical septic tank sizes for residential properties wastewater generation. Septic tanks are the most common conversion treatment technology in may range from 1,000 gallons to 1,250 gallons, or more, depending on the household's Hawai'i.

completed. If solids overflow into the drainfield, they can clog up the soil pores physically and flows out of the tank and into a disposal system, where it is further treated by means of soil anaerobic bacteria further break down the solids that remain in the tank. The liquid effluent The tank contains a mixture of wastewater and anaerobic bacteria. While the primary function of the septic tank is to physically separate the solids and liquid into three layers: a top layer of filtration and dispersed into the soil. The sludge and scum remain in the tank and must be periodically pumped out by a professional septic service approximately every three to five years, depending on usage. The drainfield can fail prematurely if periodic pumping is not scum, a middle layer of liquid effluent, and a bottom layer of sludge (Figure 1.2), the biologically due to excessive bio-mat growth.

The local IWS contractors are familiar with the nuances of septic system construction, which minimizes potential installation mistakes. The other advantages of the Traditional IWS are



Nā'ālehu Individual Wastewater System Preliminary Engineering Report

that it operates without the need for electricity, so operational cost is minimal and there are no mechanical or electrical components to repair or replace. Residential septic systems on their own remove about half of the organics in the wastewater stream and none of the nitrogen. The overall level of treatment is thought to be much higher when wellmaintained and operating with a fully functional absorption bed (Table 1.1).

drainfield to simulate a good soil filtration layer when percolation rate is expected to be faster than from site to site, it is a common practice in Hawai'i to utilize a 3-foot soil replacement below the Because the level of treatment is highly dependable on the soil filtration which can vary greatly one minute per inch.

project and will be evaluated further. Septic systems also cause less disturbance to the public Right-of-Way (ROW) in comparison to the installation of a new wastewater collection system. Due to their simple design, septic systems are applicable for consideration to be used for the

Table 1.1: Typical septic system performance in Hawaii (Carollo, 2021).

Contaminant	Typical Raw Residential Wastewater ¹	Typical Septic Tank Effluent Quality ²	Typical Effluent Quality Following Soil Absorption System ²
Total Nitrogen, mg N/L ⁴	14-40	39-82	~1
TSS (mg/L)	100-400	49-161	4~
BOD (mg/L)	100-400	132-217	<30
Fecal Coliform, MPN/100ml ³	~106	1-10 ⁶	~13
¹ From Table 2-1 (Water Resources Cent	er (WRRC) University of Hawaii-Manoa, 2	2008).	r

² From Table 4-1 in the Onsite Wastewater Treatment Survey and Assessment Study (WRRC, 2008).

³ MPN/100mL = most probably number per 100 milliliters.

2.2 Residential Aerobic Treatment Units (ATU)

An ATU is a type of wastewater treatment system that utilizes oxygen and aerobic bacteria to but at a minimum, systems typically include a primary settling chamber (similar to a septic break down and treat household sewage. ATUs come in many proprietary shapes and sizes microbes are allowed to settle out of the water (Figure 1.3). The effluent from the ATU will suspended bacteria/microbes are able to grow and thrive and a clarifier chamber, where tank), an aeration chamber, where the wastewater is blended with air or oxygen while discharge into the drainfield similar to the Traditional IWS.





Figure 1.3: Side-view of typical aerobic treatment unit (Carollo, 2021)

With the introduction of air and/or oxygen into the wastewater within the settling chamber, ATU systems provide higher organic and nutrient removal rates than traditional septic systems, making ATUs optimal for operation upstream of sensitive receiving environments. Conversely, the added mechanical and electrical componentry leads to more frequent system downtime. Downtime can occur when mechanical parts such as blowers, pumps, and control systems need to be replaced, and when the ATU system requires cleaning, monitoring, and airflow adjustments. In addition, power outages can halt the treatment process due to the mechanical components' reliance of electricity, which can result in downtime until power is restored.

Without an effective maintenance strategy, the performance of ATUs in Hawai'i has been proven to be similar to septic tanks (Figure 1.4). ATUs are less affordable than septic tanks, as they have a higher up-front cost and a much higher O&M cost due to the requirement of contracting a certified wastewater treatment operator by the DOH, and due to the reliance of electricity. The continuous operation of mechanical components and exposure to moisture and organic matter can lead to ATUs having a shorter service lifetime than septic systems, as the components are more prone to wear and malfunction over time.

Installation of ATUs may be applicable for this project when a single dwelling has more than five bedrooms, in accordance with the DOH Chapter 11-62 guidelines.

Nā'ālehu Individual Wastewater System Preliminary Engineering Report



Figure 1.4: Effluent quality of septic systems vs. ATUs in Hawaii (Babcock, 2012)

2.3 Intermittent Sand Filter (ISF)

Intermittent Sand Filter (ISF) systems provide secondary treatment that effectively eliminate contaminants through the means of physical, chemical and biological treatment processes for primary treated wastewater or septic tank effluent (Figure 1.5). ISFs are made in a variety of packed-bed filters composed of sand or other granular materials (EPA, 2002). Septic tanks are used as a preliminary treatment step



Figure 1.5: Pumped discharge ISF following a septic tank (EPA, 2023)

before sending wastewater to an ISF. The septic tank's primary function is to separate and settle out solids from the incoming wastewater. This helps to reduce the organic load and solids entering the ISF, thus extending its lifespan and improving its overall performance. Following the septic tank, the effluent is either pumped or gravity fed to the ISF.





Nā'ālehu Individual Wastewater System Preliminary En	ngineering Report	Nā'ālehu Individual Wastewater System Preliminary	Engineering Report
Wastewater is dosed onto the sand surface through a di through the sand layer to reach the underdrain system.	listribution network, percolating . Filtered effluent is collected for	could be challenging in Hawaifi.	
further treatment or discharged to a leach field through	a system of perforated pipes. These	For the Nā'ālehu LCC Replacement project, ISFs are n	ot the most suitable IWS choice for
systems involve an excavation or structure lined with im	mpermeable PVC on sand bedding	residential installation. This is due to each dwelling he	aving limited yard space to allow area
Insulated Cover in monimed. Designed fateriturion	filled with washed sand.	for a septic tank, ISF, drainfield, and potentially a pur for ISEC would be difficult to course in Homoria	mp station. In addition, the sand needed
	Most biochemical treatment occurs in	TOL TOLS WOULD BE UTILICULE TO SOULCE IN LIAMAN.	
	the top six inches, effectively	Usage Case: Sand filters are a practical alternative v	when site conditions hinder proper
	removing suspended solids and	wastewater treatment and disposal through percolativ	ve beds/trenches. They are suitable for
	carbonaceous biochemical oxygen	sites with shallow soil, poor permeability, high ground	dwater, and limited land.
24-50 INCI	demand (DUU). Microorganisms absorb wastewater constituents,	2.4 Passive Treatment Beds	
	nearly eliminating BOD. Decreasing carbonaceous BOD allows nitrifving	Passive treatment beds are a type of secondary, dece technology designed to treat domestic or small-scale	entralized wastewater treatment
	microorganisms to thrive deeper,	treatment beds need to be installed together with a s	eptic tank and do not require the use of
30 mil line or rigid containment Underdrain	promoting nitrification.	electricity. This technology uses a natural process to I	prevent suspended solids from sealing
Figure 1.6: Generic, open intermittent sand filter (EPA,		the underlying soil by incorporating aeration and a lar than traditional systems (Preshy Environmental Thr	rger surtace area for bacterial treatment 2017) Treated water is tvnically
2002)		discharged directly to the soil below the treatment sy	stem via a soil absorption system (New
There are several different types of ISFs:		,	Zealand Distributors, 2018).
Constitution Constraints (Charles Constitution)	the distribution optimization of the second s	Perforated Pipe (Air Duct) Fin Distribution Unit (FDU) Skimmer Tabs - Alawa at the winghmea - Alawa at the winghmea - Prevent -	Preshv Environmental Tuc
• Gravity Discriarge LSFS: These Inters use gravity of	/ to distribute wastewater office the	No promote bacterial growth the second second subsection between subsected with units	specializes in the development and
salid surface usually oil a sloped surface. Gravity and hottom of the sand filter, where it is transferred to	allows for the enfuenc to exit the b a drainfield through a system of	Optimized etherit term term term Optimized etherit	manufacturing of several passive
perforated pipes. The bottom of the sand filter need	eds to be several feet higher than the	Contraction of the second seco	wastewater treatment technologies,
drainfield area.	ı	perfectie	including Enviro-Fin (EF), Advanced
			Enviro-Septic (AES), and Enviro-
Pumped Discharge ISFs: Pumped discharge ISFs	s utilize a separate pump station to		Septic (ES), and Simple-Septic (SS).
distribute treated effluent from the filter to a drainfi	ifield, allowing for more controlled and		The Enviro-Fin system (Figure 1.7)
flexible effluent distribution regardless of topograph	hy, unlike gravity discharge ISFs. The	iment Fins	consists of a centrally located Fin Distribution Unit (FDUI) a basin and a
ווונכאוור) סו נווב סמווח וווכו ווובו וז זמובאתמותכת אונוו	ו מוסכוומוקב אואוווק אססונוטוובת מאסעב ור.		sump made of plastic. The sump has
Bottomless ISFs: The bottomless ISF lacks an imp	npermeable liner and doesn't release		a waterproof area at the bottom and
wastewater to a drainfield. Instead, it directly infiltr	trates into the soil beneath the sand.		small holes around it. Inside the
		Course Green Flbers	sump, there is a pipe that helps
ISFs present a more intricate setup that includes various	us components like a dosing tank,	Air Ports suspended solids	equalize the wastewater between
pump or siphon, distribution network, and a filter bed ec	equipped with an underdrain system.	Treatment Firs to promote bucknait treatment surface aerobic tacterial growth - Contrast massive tacteria	different parts of the system. The centic tank affluent flows into the
Regular maintenance will increase the cost que to the el	siectnoity and labor for		EDII and then rate distributed to
puriprig/uosing, and overall own labor. There is a pote intermittent sand filters becoming clogged, and finding li	enual risk of the meet meeta in local replacements for the media	Figure 1.7: Schematic of Presby's Enviro-Fin Passive Treatment Bed (Presby Environmental)	eight Treatment Fins that extrude
ENGINEERING		ENGINEERING C C T N C C C	
LAKINEKS	12	LAK - NEKU	13

۲
ō
B
≃
p
÷E
e
.⊆
g
ш
≥
g
Ē
. <u> </u>
<u>n</u>
<u>с</u>
F
Ъ
Š
Ľ,
Ę
۲a
ē
ß
Ň
4
Ë
ġ
≓
Ĕ
Ξ
Ę
E E
un m
ž

radially from the center and treat the wastewater. Small holes in the plastic air ducts supply oxygen to the surfaces containing bacteria, which help break down the suspended solids in the wastewater. The system also has green plastic fibers packed underneath the air ducts, and specially designed fabric around the Treatment Fins and FDU, which increases the surface area for bacterial growth and wastewater treatment.

effluent but also accelerates the treatment processes, ensures even distribution, and provides square feet of total surface area for bacterial activity. The overall sizes of the AES system are within the pipe that efficiently treats the wastewater (Presby Environmental, Inc. 2017). The AES technology represents the evolution of Presby's ES system, incorporating the proprietary Bio-Accelerator® enhancement. This enhancement not only filters additional solids from the similar to an absorption bed and depend on the number of bedrooms in a dwelling, and the percolation rate of the soil. Due to the higher rate of filtration, the pipes of an AES system can be installed in closer proximity compared to the perforated pipes in an absorption bed, a larger surface area for bacterial activity. Each foot of the AES pipe offers more than 40 plastic fibers and sewn geo-textile fabric. Together, these materials create an ecosystem Accelerator® fabric along its bottom. This fabric is surrounded by a layer of randomized AES (Figure 1.8) consists of a pipe made of corrugated, perforated plastic and a Biowhich decreases the overall width of the disposal surface area needed.



- thes the pipe and is conied to ground lare a from the cooled liquid effluent
- and suspended solids frem the existing affluent cuplure grease
- end. from the inted. ators fabric screens additional solids uce start-up after periods of non-use, een distribution, and further protects o effluent to Flow units allow the
- plide from the effluent (AES &
 - see into the geo-textile failetcs and grows a protected wicks
 - liquid from the geo-leatile fabrics and erables air to transfer to the bac The fabrics
 - and fibers provide a large bacterial surface to break down in supply and fuctuating louid levels increase bacterial affe

Figure 1.8: AES, Enviro-Septic, and Simple-Septic 10 stages of treatment.

The ES technology is composed in a similar fashion to the AES pipe but offers only 25 square feet of surface area for bacterial activity for every foot of pipe. The SS pipe is comprised of a single-layer geo-textile fabric, large diameter, gravelless pipe (LDGP) system, coupled with



Nā'ālehu Individual Wastewater System Preliminary Engineering Report

surface area of the fabric. For every foot of SS pipe, there is 15 square feet of total surface Presby's patented skimmer tabs and cooling ridges which increase bacterial growth on the area for bacteria to grow.

Administrative Rules (HAR) by the Department of Health (DOH). The DOH acknowledges this is an innovative solution that could be beneficial to the State. In addition, the DOH requires Presby has claimed that each treatment system output effluent that has been tested and certified to NSF 40, Class I standards. The use of the Presby Passive Onsite Wastewater Treatment & Dispersal System is permitted under Section 11-62-35(b) of the Hawai'i that operational data be submitted upon request within the first twelve months after installation when Presby products are used in place of soil absorption systems.

operation can differ between the two systems. ATUs typically require ongoing electrical power parts, which can lead to lower long-term maintenance and operational costs. Enviro-Septic or installation. However, it is important to consider that the long-term costs of maintenance and electricity or mechanical components. They rely on natural processes and have fewer moving than that of an ATU. Both systems require upfront investment for equipment, materials, and In general, the cost of an AES or Enviro-Fin system may be comparable to or slightly higher monitoring systems. These factors can contribute to higher operational costs over time. On technologies and are generally less expensive alternatives but require a bigger footprint. for aeration and additional maintenance, such as replacing mechanical components and Simple-Septic systems have a lower system complexity than AES or Enviro-Fin Presby the other hand, AES systems are designed to operate passively without the need for

Presby offers both in-person and online training options, covering various aspects of system techniques, system components, troubleshooting, and best practices for achieving optimal design, installation, maintenance, and regulatory requirements. Proper installation performance of wastewater treatment systems are provided in the training.

Sand" which is a granular material with a very specific gradation. At the time of this report, While these passive treatment bed systems offer higher effluent quality without mechanical it is not known if the local rock quarry can produce aggregates that meet that specification. Furthermore, one of the key components of these systems is the requirement of "System components, they are new to the market and are not familiar to local contractors.

Based on the increased regulatory requirements, unfamiliarity of local contractors, and supply chain limitations for the aggregates needed, AES, EF, SS, and ES solutions may not be suitable treatment processes for the project.



2.5 Composting Toilet

action to break down human waste into a nutrient-rich compost. Composting toilets typically A composting toilet is a type of toilet that uses a combination of heat and aerobic microbial come in three varieties:

- Individual Composting Toilets:
- exhaust fan, to aid in the breakdown human waste with bulking material 1.9) The waste dries and composts in-situ until the container fills and such as sawdust, leaves, or peat These waterless toilets combine moss in a single chamber (Figure mechanical systems, such as an toilets do not require water or a waste in accordance with DOH of waste and limit odors. These landfill that can process solid then needs to be hauled to a connection to a sewer system, alternative to traditional flush standards. Some composting making them an eco-friendly toilets also use electrical or

Figure 1.9: Component view of a typical individual composting toilet (Sun-Mar, 2023)

and one for solid waste (Figure while the solid waste is broken Urine Diverting Toilets: Urine diverting toilets typically have stored and used as a fertilizer, two chambers: one for urine 1.10). The urine is typically down into compost.

toilets.

Figure 1.10: Top view of a typical urine separating toilet (Separett, 2023)

Central Composting Units: These

home and work in conjunction with pint-flush toilets. Toilet blackwater (along with a scoop of wood chips) flows through a 4" gravity sewer to the composter composters are installed outside the



liquids to drain into the base of the enclosure (Figure 1.11). The drum must be where it is introduced to a horizontal drum that collects the solids and allows rotated in the forward direction once

every two days to fluff the retained solids and rotated in the reverse direction once every two months to drop the retained composting process is completed. The solids into an aging drawer where the entire home and solid wastes and any central composting unit can serve an manure and drained liquid. A single outputs are a drawer of composted odors are kept entirely outdoors.

toilets that are land applied must meet the requirements of HAR 11-62 Subchapter 4: Wastewater Sludge Use and Disposal. This basis. Solids generated from composting approved by the DOH on a case-by-case Composting toilets are reviewed and



Figure 1.11: Top view of a central composting unit (Sun-Mar, 2023)

treatment process may not be applicable for this project due to the regulations pertaining to tanks require less active involvement from homeowners and do not typically require reqular emptying if properly sized and maintained. Composting toilets can also emit strong odors if unit in compliance with the sludge related conditions in a permit issued by the DOH. Septic the disposal of solid waste, which could require disposal at a municipal solid waste landfill not properly maintained.







Nāʻālehu Individual Wastewater System Preliminary Engineering Report	2.7 Graywater Reuse Graywater systems are systems that collect and reuse wastewater from sources such as sinks, showers, and laundry machines. The collected water is then treated and can be used for landscape irrigation once separated from blackwater. The wastewater streams can be separated through the implementation of two separate wastewater piping systems in the household. Graywater can be treated to a re-use level for distribution strictly below the soil surface or should be disposed of to an IWS or county sewer system (CS). In Hawairi, the use of graywater systems is regulated by the State Department of Health, which sets guidelines for the treatment and reuse of graywater. Overall, these systems are typically expensive to install and maintain as they are installed in addition to a traditional wastewater treatment system that's still required to treat household blackwater. For simplicity of design, graywater systems are less applicable for this project than a traditional septic tank system, as access to a CS is not readily available in Näfälehu, which would require the reuse system to have both a septic and graywater holding tank. Usage Case: Used on lots with source separated plumbing that prioritizes wastewater reuse.	Figure 1.13: Typical gravities the solution of
Nā'ālehu Individual Wastewater System Preliminary Engineering Report	Figure 1.12: Side view of a typical incinerating tolicit (Cindensila, 2023)	2.6 Incineration Toilet An incineration toilet, also known as a thermal toilet, is a type of toilet that uses heat to turn human waste into ash (Figure 1.12). The waste is placed into a combustion chamber where it is heated to high temperatures, typically around 800-1000 degrees Fahrenheit, by a gas or electric burner. This process kills any harmful bacteria and viruses and reduces the volume of waste by up to 90%. The ash that is left can be safely disposed of in a landfill or used as a fertilizer. Incineration toilets do not require any water, making them suitable for remote locations or areas with limited water resources. They also produce very little smell and have no need for a septic system or connection to a sewer. However, they do require electricity or gas to operate, and can be relatively expensive to purchase and maintain. Further, they do not comprise a complete treatment, a septic tank or other treatment unit is still required to treat kitchen blackwater, eliminating the economic advantage. No matter the variety, compositing toilets and incineration toilet technologies are significantly lower cost than septic tanks and other IWS. Unfortunately, they do not comprise a complete treatment solution, making them not applicable for this project.

ENGINEERING P A R T N E R S

E PARTNER S

3. Types of IWS Disposal Systems

3.1 Absorption Bed

Absorption beds are the most common form of IWS disposal system installed in Hawai'i today. They consist of a network of perforated pipes, each a maximum of 100 feet long and laid in trenches 1.5-3 feet below the finished grade 4-6 feet apart (Figure 1.14). Each line is laid level to allow the gravity dispersal of treated effluent below and above each pipe, with a filter fabric covering the top of gravel, to prevent the gravel from clogging. A minimum of 6 inches of gravel is provided below each pipe. If the percolation rate is faster than one minute per inch, a depth of 3-foot soil replacement shall be installed to underlie the entire absorption bed. Soil replacement shall be washed #4 sand or cinder-soil mix with a percolation rate not faster than one minute per inch. These systems are easy to maintain when following an effective treatment system and microorganisms in the soil offer an added degree of treatment to the effluent as it filters through the upper oxic layers of the soil matrix. Absorption beds, however, have a significant space requirement, which increases with decreasing hydraulic conductivity of the soil. Additionally, absorption beds can only be installed on a grade of less than 8%.

While conventional perforated pipe adsorption beds are not traffic rated, companies such as Infiltrator offer a chambered dispersion product with an H-20 load rating that can also reduce the absorption bed space requirement by 17%. Absorption beds are applicable for this project when space permits, as they utilize natural soil and microbial processes to treat and filter wastewater, minimizing the environmental impact. Additionally, the relatively simple design and lower maintenance requirements make them costeffective and practical for residential properties.

Usage Case: Used on typical lots without spatial, groundwater level, grade, or percolation rate constraints. Installed on flat/mild sloped terrain.



Figure 1.14: Typical absorption field installed following a septic tank (Babcock, 2019)

3.2 Leach Filed

Leach fields consist of absorption trenches installed parallel to the contour line on ground slopes up to 12%, making them suitable for steeper terrain. These trenches are non-traffic rated, and consist of gravel, which serves as a natural filter that permits gradual seepage into the adjacent soil. Perforated PVC pipes are commonly used to distribute wastewater into the trenches. Additionally, filter fabric lining can be installed to prevent the gravel from becoming clogged with soil or debris.









Nāʻālehu Individual Wastewater System Preliminary Engineering Report	Nā'ālehu Individual Wastewater System Preliminary Engineering Report
Treatment system	insufficient land area, steep terrain (>12%) or very slow percolation rates (>60 min/inch). Where slow percolation rates present, seepage pits will need to be dug through the basalt rock layer to reach more porous soils or a variance will be required from HAR 11-62-34 d(1)b:
VO free max	Seepage pits shall not be constructed in soils having a percolation rate slower than ten minutes per inch (weighted average) or where rapid percolation through such soils may result in contamination of water-bearing formations or surface water.
Trenches	Seepage pits are applicable for consideration to be used for this project for site conditions mentioned above and will be evaluated further.
Figure 1.16: Absorption Trenches Diagram (Water Resources Research Center, & Engineering Solutions, Inc. (2008).	Usage Case: Used on highly spatially constrained, slope constrained, or geologically constrained lots where sufficient percolation rates can be achieved.
The DOH design requirements state absorption trenches should have a width ranging from eighteen to thirty-six inches. Additionally, there should be a minimum vertical separation of three feet between the bottom of the trenches and the underlying groundwater or bedrock.	3.4 Subsurface Drip Irrigation Subsurface drip irrigation is an extremely water efficient means of wastewater disposal, slowly delivering effluent into the existing vegetation present on site within the disposal area, promoting delivering effluent into the existing vegetation present on site within the disposal area, promoting
Leach fields with absorption trenches are applicable for this project when the site has a grade between 8%-12% and will be discussed further in the report.	encient uptake or nutrients by the microbes and plants in the soil medium (Figure 1.17). Installation of subsurface systems often requires less disruption to the absorption area though in Hawaii, they traditionally cost more. Regular maintenance is required to ensure the continued operation of the irrigation pump and manage biofouling in the distribution lines.
Usage Case: Absorption trenches are typically used on steeper terrain.	Subsurface drip irrigation may not be suitable for the project, as more O&M is required
3.3 Seepage Pits	compared to alternative wastewater disposal systems, in order to flush the drip lines to mitigate clossing from dobrin Chlorinship to among biological arouth from the drip lines would also be
Seepage pits are a vertical means of achieving the percolation area requirements for a disposal system. These systems typically consist of a 15-30-foot-deep pit lined with stacked precast perforated concrete rings or CMUs, to an internal diameter of 6-8 ft. Seepage pits are both less area intensive and less expensive than absorption beds, if converted from an existing cessbool. A	clogging from debris. Cinomiation to remove biological growth from the urp intes would also be needed, which has a greater environmental impact than other means of disposal. Additionally, each IWS with subsurface drip irrigation would require the incorporation of an ATU to properly treat wastewater for reuse.
seepage pit must include a cover which extends at least 12 inches beyond the seepage pit excavation or over a provided concrete lining. An access hatch must be provided in the concrete cover to allow inspection and maintenance of the pit. The seepage pit may be designed to be traffic rated by providing the sufficient strength required in the design of the concrete lining and cover.	Usage Case: Used on lots that prioritize wastewater reuse, are served by an ATU, and have a robust maintenance strategy.
The effective area of the seepage pit is equal to the vertical wall area corresponding to the effective depth of the pit. Slow percolation rates translate to a larger required absorption area and deeper pit (Figure 1.21).	
While seepage pits are an approved means of disposal in Hawai'i, they are often only permitted when it can be demonstrated that an alternative means of disposal was not possible, i.e.,	
ENGINEERING P A R T N E R S	ENGINEERING P A R T N E R S



Figure 1.17: Typical subsurface drip irrigation installed following a septic tank (EPA, 2023)

3.5 Evaporation Beds

Evaporation beds, or evapotranspiration (ET) wastewater treatment systems, are utilized to remove effluent by direct evaporation and by plant transpiration. If soil infiltration is desired, evapotranspiration/infiltration (ETI) process can dispose of effluent by employing both evapotranspiration and soil penetration. ET and ETI systems follow primary pretreatment units which filter out settleable and floatable soils.

Nā'ālehu Individual Wastewater System Preliminary Engineering Report



Figure 1.18: Evapotranspiration (ET) bed (EPA, 2018)

The septic tank effluent enters the ET or ETI units through distribution pipes and reaches a porous bed. To prevent water from seeping out, a liner is used below the bed in ET systems. Aquatic plants are grown on the surface of the sand bed to aid in transpiration of the effluent. In ETI systems, the effluent can percolate into the underlying soil with the absence of a liner.

Evaporation beds are sizable disposal systems that may experience overflow if the evapotranspiration rate does not exceed the monthly precipitation by more than 2 inches a month. ET/ETIs are reviewed on a case-by-case basis by the Department of Health. Evaporation beds may not be appropriate for this project, as Nă'ălehu has a tropical climate with abundant rainfall throughout the year. Additional maintenance is also required to prune and monitor plant health to ensure the system functions effectively.

Usage Case: Arid climates where evaporation exceeds precipitation. Sensitive locations where no subsurface disposal is desired/allowed. These disposal units are most appropriate following ATUs but are occasionally employed after septic tanks.

3.6 Mound Septic System

When greater vertical distance from groundwater is needed, or when the soil quality or slope is not conducive for an absorption bed or trench system, above-ground mounds of sand or soil can be implemented to achieve desired conditions (Water Resources Research Center, & Engineering Solutions, Inc., 2008). The process begins by preparing the land where the



ENGINEERING P A R T N E R S

mound will be situated through tilling. Afterward, a layer of sand is spread across the tilled area along with a distribution system. Next, the top of the mound is covered with soil from the surrounding area and given an aesthetically pleasing landscape. A pump system is necessary due to the influent disposal point being at a higher elevation than the treatment system. This factor raises the overall cost of the system since electricity is required.	household IWS could cost between \$30,000-\$150,000 to install and roughly \$1,000 per year to operate and maintain. However, associated O&M fees for household IWS are dependent on the management model implemented, which is further discussed in Section 11. Based on the EPA requirements for this project, the installation of traditional septic tanks with standard absorption bed is recommended for Nā ⁷ ãlehu. Per the revised AOC, IWS is a possible
Mound Septic System Renew Manual System	alternative to be considered, however technical and environmental analysis will still need to be conducted for Hawai'i County to make an informed decision. Septic tanks coupled with absorption beds offer the lowest cost, least complex solution, with low maintenance requirements in comparison to the other options previously discussed. Where space and grading constraints prevent the installation of an absorption bed, an existing cesspool (if present) can be repurposed as a seepage pit for disposal, or a new seepage pit can be installed. This alternative is being considered for three reasons:
the first fi	 Passive Operation: The traditional septic IWS offers a passive operation with minimal O&M cost. While the ATUs offer the promise of improved performance, in practice in Hawaii, the added mechanical and electrical complexity and the operator requirement results in higher-than-average O&M costs or facing the risk of falling into similar treatment performance comparable to septic tanks. Adaptive to Small Lots: The option of disposing to seepage pits allows septic systems to
Figure 1.19: Elevated Mound System (EPA, 2018)	 be installed particularly on spatially and geographically constrained lots. Familiarity: Hawair's engineers, regulators, contractors, and septic pumpers are familiar with septic tanks, absorption beds, and seepage pits. The ATU and CTDS, on the other hand, are less known. Furthermore, the lack of certified wastewater operators theorem the other hand, are less known. Furthermore, the lack of certified wastewater operators
The Department of Health (DOH) evaluates the design criteria for mound disposal systems on a case-by-case basis when systems are outside the percolation rate range of trenches and beds. There is a three feet minimum required separation from any existing groundwater.	 CTDS options. Cost: Traditional septic tanks don't have moving parts or require significant mechanical components, which can reduce maintenance and repair costs over
Mound septic systems may not be suitable for this project as the cost can be more expensive to install and maintain compared to traditional wastewater disposal systems. This is due to the additional design components required, such as the construction of the mound itself and specialized fill material. Mound systems require a larger land area than other disposal systems which can be a limitation on smaller lots.	 time. Energy cost savings are also significant since septic tanks rely on natural processes for wastewater treatment, compared to some other IWS that involve mechanical or electrical components. O&M: The absence of complex components in septic tanks simplifies maintenance. There are no pumps, blowers, or electronic controls to manage or repair, making maintenance tasks more straightforward.
 Usage Case: In areas where limitations exist due to poor soil or proximity to groundwater. Traditional IWS Recommendation When considering an IWS alternative, existing regulations, environmental concerns, site constraints, economics, and performance shall be taken into consideration. Overall, each 	• Environmental: Septic tanks are more energy efficient than other IWS alternatives such as ATUs, which require energy-intensive processes (aeration) for effective treatment. Additionally, other IWS alternatives may require the use of chemicals for treatment processes. Septic tanks generally involve fewer chemical inputs, contributing to a lower environmental impact.

Nā'ālehu Individual Wastewater System Preliminary Engineering Report



E PARTNER S

Traditional IWS Components പ

5.1 Types of IWS Tanks

points and materials. Septic tanks can be made from concrete, plastic, and fiberglass (Figure 1.20), There are several septic tank providers commonly used in Hawai'i offering tanks of a variety of price each of which having its own set of pros and cons (Table 1.2). Where a septic tank is located beneath a vehicular traffic area, a traffic rated concrete septic tank can be used or a structural concrete slab designed for H-20 loading spanning a non-traffic tank may be used.

material is flimsy, it is prone to be installed improperly, and has more restrictions on the location The yellow plastic tank displayed in Figure 1.20 is manufactured locally by Chemtainer in Keaau, Hawai'i. This HDPE tank is the most economical type of septic tank. However, because the to which it can be installed.

Septic tank manufacturers and distributers in Hawai'i include Jensen Precast, Ferguson (Infiltrator Chambers), Chemtainer, and Orenco.

Figure 1.20: Common septic tank materials and shapes in Hawai'i (Carollo, 2021).



Rectangular, Concrete Tank Tank



Oval, Concrete Tank



Cylindrical, Concrete







Steel, Horizontal, Cylindrical Tank



Table 1.2: Advantages and Disadvantages of Septic Tank Materials (Carollo, 2021).

Septic Tank Material	Advantages	Disadvantages
Concrete	 Durable Suitable for installation in traffic area Less susceptible to collapse and floatation May be cast-in-place for custom shape 	 Precast tanks can be more expensive than plastic or FRP due to shipping and installation costs Typically requires use of a crane for installation Concrete may corrode over time due to acidic sewer gases
Plastic (polyethylene)	 Less expensive than precast concrete tanks (lower shipping and installation costs) Manufactured locally on the island Plastica are typically resistant to corrosion May not require a crane for installation 	 Plastic tanks may deform depending upon quality of the plastic and potential structural weaknesses of the material If not installed properly, plastic tanks can float if flooded
Plastic (polypropylene)	 Less expensive than precast concrete tanks (lower shipping and installation costs) Plastics are typically resistant to corrosion Locally stocked Higher tank rigidity. More tolerant impact and backfill loads May not require a crane for installation 	 Not a traffic rated tank without additional structural slab A two half-clam shell construction. Requires proper factory assembly to achieve watertightness
Fiberglass-reinforced polyester (FRP)	 Less expensive than precast concrete tanks (lower shipping and installation costs) Variety of manufacturers and sizes for desired footprint Fiberglass is typically resistant to corrosion May not require a crane for installation More rigid and sturdy than plastic tanks 	 Less structurally strong than concrete tanks If not installed properly, fiberglass tanks can float if flooded

constraints (Table 1.3). At a minimum, septic tanks in Hawaii must comply with International Association of Plumbing and Mechanical Officials (IAPMO) material and property standards for Ultimately, the choice of septic tank material will depend on availability, budget, and site septic tanks. Further, sizing and installation criteria are regulated by HAR 11-62-33. The





Rectangular, Plastic Tank



28

ENGINEERING PARTNERS

Ĕ
Rep
ering
ginee
Ē
inar
elim
m Pr
yste
Ň
vate
astev
Ň
iduā
.indi
hu I
a'āle
ž

minimum septic tank capacity is 1,000 gallons for a household of 4 bedrooms or less and 1250 gallons minimum for households of 5 bedrooms. Septic tanks serving households greater than 5 bedrooms will require a variance from the DOH.

Table 1.3: Common single family home septic tank products in Hawai'i.

Product	Material	Traffic Rated	Capacity (Gal)	Length (in)	Width (in)	Height (in)	Weight (Ibs)	Local List Price ¹
Chem- tainer	HDPE	No	1250	96	58	62	400	\$3,189
Infiltrator (via Ferguson)	dд	No	1287	127	62.2	54.7	320	\$2,863
Orenco (via Custom Concrete & Septic)	DCPD	N	1500	168	72	64.5	620	\$6,850
Jensen Precast	concrete	Yes	1250	138	70	57	16,700	\$6,850

Nā'ālehu Individual Wastewater System Preliminary Engineering Report

6. Design of IWS: DOH Chapter 11-62 Guidelines

Per the EPA Revised Administrative Order on Consent (AOC), SDWA-UIC-AOC-2017-002, dated August 22, 2022, the County must provide wastewater services for 194 properties, and close the Nāʿālehu Community Cesspools. The Department of Health's Hawai'i Administrative Rules, HAR 11-62-31.1, outlines general requirements for individual wastewater systems used in lieu of wastewater treatment works. Lots to be served in the community vary in size from 0.12 to 1.94 acres with a median size of 0.16 acres. The current Hawai'i Administrative Rules for INS installation require a minimum of 10,000 ft² (0.23 acres) of usable land for each system for lots constructed and documented after August 30, 1991. The Nāʿālehu considered grandfathered in.

The maximum permitted wastewater volume entering a single wastewater system must not surpass one thousand gallons, and this system is not allowed to cater to more than five bedrooms, regardless of whether these bedrooms are in one dwelling unit or distributed across two.

Table 1.4: Septic sizing guide per DOH

ık Sizing Guide	Minimum Capacity (Gallons)	1000	1250
DOH Septic Tar	No. of Bedrooms	4 or less	5

If a single household has more than five bedrooms and requires an IWS to handle the wastewater, then an Aerobic Treatment Units (ATUs) needs to be installed, which can handle a larger wastewater load.

6.1 Percolation Testing

Prior to installing the IWS, a percolation test is performed to assess the soil's ability to accept and treat the effluent. The results of this test help determine the appropriate size and design of the system. Per HAR 11-62, soil percolation tests must be carried out at a depth of at least three feet. If the soil composition varies between the three-foot and five-foot depths during construction, an additional percolation test should be conducted at the level corresponding to the bottom of the absorption system.





6.2 Sizing of Absorption Beds and Disposal Systems

The sizing for absorption beds and disposal systems is outlined in Figure 1.21 from HAR 11-62. Absorption beds cannot be built in soil with a percolation rate slower than sixty minutes per inch. Appendix A contains preliminary absorption bed sizing calculations for the properties in Na^{*}älehu using the HAR11-62 Absorption Bed Sizing Table. Percolation rates ranging from 10-15 min/in per site were used for the basis of this calculation, which is reflective of the higher end of the percolation test results.

Attentition Arwa (ft2/badroom of 200 gellone)	253	257	260	263	267	270	273	277	280	283	287	290	293	297	300	302	304	306	308	310	312	314	316	318	320	322	324	326	328	330
thin of equal to	31	32	33	34	35	36	37	36	39	40	41	42	43	44	45	46	47	48	49	50	51	52	53	54	55	56	57	58	59	60
Absorption Arma (ft2/hedrams at 200 gallens)	70	85	100	115	125	133	141	149	157	165	170	175	180	185	190	194	198	202	206	210	214	218	222	226	230	234	238	242	246	250
fmin/inch) Less than Jr équal to	1	2	57	4	S	9	7	B	6	10	- 11	12	13	14	15	36	17	16	19	20	21	22	23	24	25	26	27	28	29	30

Figure 1.21: HAR 11-62 Table III Absorption bed sizing table.

6.3 DOH Required Setbacks

The actual location of treatment and disposal infrastructure is limited by setback requirements. DOH-required setbacks are presented below (Table 1.5). From a system design perspective, it is recommended that systems should also be a minimum of 20 feet from any cut-face slopes present on a site to avoid surfacing of treated effluent. This is a particular constriction to heavily sloped sites.

ENGINEERING P A R T N E R S

Nā'ālehu Individual Wastewater System Preliminary Engineering Report

Table 1.5: DOH required setbacks for wastewater systems: Table II, HAR 11-62.

inimum orizontal istance rom	Cesspool (ft)	Treatment Unit (ft)	Seepage Pit	Soil Absorption System (ft)
all line of any structure or building	ŝ	ŝ	ŝ	ŝ
roperty line	σι	ŝ	6	ŝ
at the ocean at the shoreline certification, pond, lake, or other surface water body	50	ŝ	50	co S
arge trees	10	ŝ	10	10
reatment unit	ŝ	ŝ	ŝ	S
eepage pit	18	ŝ	12	ŝ
esspool	18	ŝ	18	ŝ
oil absorption system	ŝ	ŝ	ŝ	ŝ
otable water sources serving public water systems	1000	500	1000	1000

6.4 DOH Variance

The Na'älehu LCC Replacement project will likely encounter the following situations where a DOH Variance request would be needed:

- Absorption field placement due to percolation rates and property size.
- Seepage pit installation when space is limited.
- Septic tanks serving households with more than 5 bedrooms.

Variance application and review usually adds approximately two months to the permitting process following design package review but the DOH has expressed willingness to allow the request for variance to be included with the initial design package.

7. Typical IWS Permitting and Construction Process

The permitting and construction process for IWS begins with the Engineer of Record (EoR) preparing and submitting a design package for DOH approval. Once DOH reviews and approves the design package, DOH issues an "Approval for Construction" letter. While during DOH review



process, the plumbing contractor submits the Plumbing Permit application with County DPW Building Division for building sewer line modification. Once both permits are issued, the contractor can start excavation of the septic tank, drainfield and pipe trenches, while the existing sewer lines are protected in place, or a temporary bypass is installed to keep the existing system in service. The EoR will perform a site-specific percolation test at the bottom of the drainfield excavation to further verify the actual percolation rate and adjust drainfield sizing as necessary. Per HAR 11-62, retesting is only required however when the soil profile at 5 feet depth is different than the soil profile that was performed during the percolation tests. After the installation of drain rocks, drainfield piping and the septic tank, the engineer performs a final inspection and document asbuilt of the installed system before final backfilling and files a final inspection report to DOH for review. If everything is in order, the DOH returns an "Approval for Use" letter and the building waste line can be changed over to the newly installed IWS.

The Na'alehu LCC Replacement project will use a project specific approach in three principal ways:

- Properties to be Permitted in Bulk: The DOH has expressed the capacity to receive packages in groups of 10 or more properties. This will serve to expedite the DOH review process.
- Variance Requests: Due to preliminary data on percolation rates and property sizes, it's expected that DOH variances to setback constraints will be required to accommodate the IWS installation. Where space is still overly constrained, DOH variances will be required to allow for seepage pit installation. During the design phase, the specific lots requiring variances will be identified, which the variance application can be submitted prior to submitting the design application to construct, which will enable the permitting process to go more efficiently.
- **Optioned Design:** The geological conditions present at a given site will not be known until construction on that site commences. These conditions will have a significant impact on the disposal system design. To accommodate this uncertainty in the design and facilitate permitting, an optioned design package shall be submitted to the DOH to allow for a field determination according to the actual percolation rates and soil composition encountered. We would expect the in-situ soil stratum to change from 4-5 feet deep. The percolation rates between the volcanic ash soil and the basalt rock are great, so as comparing fractured and unfractured rock. Since the absorption bed sizing greatly depends on the percolation rate, location and the depth of required excavation, it is best to perform the percolation test at the bottom of the excavation to represent actual conditions of design.

The timeline and deliverables for this procedure is outlined on the following page (Table 1.6).



Nāʻālehu Individual Wastewater System Preliminary Engineering Report

Table 1.6: Hawai'i Permitting and Construction Process

Step	Timeline	Party	Deliverables
Design Package + Variance Application Preparation	4-6 months	Engineer	 Site Evaluation/Percolation Test Result IWS Calculation Parcel map Plot Plan Plot Plan Simple Building Floor Plan for number of bedroom determination IWS Leyout IWS Details UWS Details Owner Certification form \$100 IWS Application Fee
Design Package Review	2-4 weeks	рон	Letter of Approval
Variance Application Review	2 months	РОН	Letter of Approval for Construction
Construction	1-2 weeks/property	Contractor	Completed IWS
Final Inspection and Final Inspection Report (FIR)	1 day/property 1 month/report	Engineer	 Final Inspection Report (FIR) As-Builts
FIR Review	1 month	рон	Approval for Use

¹ Timelines are estimates, as it needs to be clarified if the design/DOH review is for the group of 10 lots or all 194 lots.

7.1 Project Schedule

Phasing for this project should reflect deadlines as well as workforce and product availability. The AOC requires the County to implement the selected wastewater treatment alternative for the 194 properties and to close the Nă'älehu Community Cesspools no later than December 31, 2027. The County still needs to make a decision if the project will be bid out in groups of IWS (multiple bid packages) or if the project will be bid out in its entirety. Two different options for the project schedule are proposed as follows:

 Scheduling of the sites could be in accordance with the three currently open LCCs, which are displayed in Figure 1.22 below. The smaller two LCCs, LCC 4 and 5, which serve 16 and 8 lots respectively, could be grouped together to streamline the bidding process and timeframe. The construction of the smaller LCCs could also be treated as a pilot stage, with lessons learned to be integrated into the design of LCC 3, which serves approximately 138 lots. This option would streamline the amount of coordination required to inspect each of the three groups of connected properties and is displayed in Table 1.7.



During construction, lots should be grouped and approached with a top-down approach, starting with lots on the north end of Na^{*}ālehu and working down, keeping the previous sewer mains in mind and keeping the existing collective system active. This will enable the households downstream to not be affected by the construction. Multiple contractors could be awarded construction contracts for each bidding cycle which overlaps for increased time efficiency. The County could also choose to do the procurement and construction process separately for each of the three LCCs, starting with the smallest, LCC 5 with 8 lots as a pilot phase, and assign multiple contractors to each LCC. However, this would increase the total time of the project due to the additional round of bidding that would occur from ungrouping the construction of LCC 4 and 5.

A pilot consisting of 10 sites consisting of LCC connected properties could be first permitted and constructed prior to the remaining parcels. Lessons learned from the pilot would be integrated into the design and permitting of the next phase of 154 LCC connected properties. Multiple contractors would be awarded construction contracts in 10-lot bundles that may be completed simultaneously.



Figure 1.22: Existing Large Capacity Cesspools: LCC 4, LCC 3, and LCC 5 (Brown & Caldwell, 2023).



Nā'ālehu Individual Wastewater System Preliminary Engineering Report



Jā'ālehu Individual Wastewater System Preliminary Engineering Report	Nā'ālehu Individual Wastewater System Preliminary Engineering Report
8. IWS Constructability Challenges & Solutions	5-5. General considerations for locating a SWIS on a
8.1 Lack of Yard Space	alts Burdos
Lots to be served in the Nā'ālehu community vary in size from 0.12 to 1.94 acres with a median size of 0.16 acres. Space available for IWS installation on these properties is further limited by the presence of existing structures. As a solution, seepage pits can be installed in lieu of absorption beds in space constrained lots, and IWS installations could be installed encroaching into Cumbr Rinch-or-Waw if normality and IWS installations could be installed encroaching in the Cumbr Rinch Route Reserved in Second bin into Cumbr Rinch-or-Waw if normality and IWS installations could be installed	
8.2 Landscaping	A Company of the second
Wastewater infiltration and dispersion can be affected by landform position, and thus the selection of the IWS location needs to account for each site's unique landscape during the design and construction phase.	AVOID AREAS WHERE SURFACE FLOWS CONVERCE Source Pudde (Montaly, 1988
Additionally, disturbance of existing landscape can occur during the installation of an IWS, which can be a relatively invasive process requiring large equipment like excavators, dump	Figure 1.23: EP4 Manual recommendation for placement of subsurface wastewater infiltration system (SlydS)
trucks and cranes. Front-yard IWS installations can be utilized as an alternative to back-yard installations if this helps to minimize the disturbance of landscape. Thorough site assessments can help identify existing landscape features such as vegetation, trees, retaining walls, fence walls and natural drainage patterns which can help preserve key elements in the design. Trees may be temporarily removed and properly stored, and then later reinstalled to restore vegetation back to the original location. Speed and care in the restoration of any displaced	Sites in Narateriu have slopes that vary from 6-10% (Appendix C). This is likely to affect the constructability of absorption beds as a method of wastewater disposal. Per HAR 11-62-34, absorption beds shall not be installed on land with a slope gradient greater than 8%, while absorption trenches are permitted on a slope of up to 12%. In case this slope requirement cannot be fulfilled, the DOH would allow a seepage pit to be installed instead of an absorption bed.
and disturbed landscapes should be taken to increase nomeowner satisfaction and to restore the site to pre-construction conditions or better.	8.4 Cut Slope
8.3 Ground slope	Constructing an individual wastewater system (IWS) near a cut face can present challenges, particularly in relation to potential side seepage. The excavation or cut face may pose a risk of
When analyzing the topography of the site, key features offer more design flexibility than others. Long, planar slopes or plateaus should be prioritized when constructing an IWS over	allowing wastewater to permeate the surrounding soil or potentially contaminate nearby groundwater sources.
ridges, knolls, or other mounded or steeply sloping sites. Considering the ground slope is crucial in the design of an absorption bed to be downstream of the dwelling and septic tank. According to the EPA Manual, subsurface flows can accumulate in the presence of swales, depressions, or floodplains, and optimal landscape positions consist of convex slopes and flat areas with deep, permeable soils (EPA, 2002).	To mitigate potential side seepage into adjacent cut face slopes, the IWS should be designed to consider the proximity to the cut face by adjusting the location and depth of the components, such as septic tanks or absorption beds. Installing the bottom of the absorption bed 20 feet away from the cut face slope ensures adequate distance for potential side seepage not to occur.
	8.5 Site Geology
	The National Resources Conservation Service Soil Survey was consulted for site soils information (Appendix D). The site is principally composed of Naʿālehu medial silty clay loam and Puueo-Naʿālehu complex. The surface of both compositions features silty loam, but the Puueo-Naʿālehu complex gives way to lithic bedrock at a depth of approximately 20-40 inches.
ENGINEERING PARTNERS	ENGINEERING PARTNERS

Nā'ālehu Individual Wastewater System Preliminary Engineering Report	Nā'ālehu Individual Wastewater System Preliminary Engineering Report
This data was reinforced by the boring logs from a 2007 Geotechnical Investigation Report conducted by Masa Fujioka & Associates (Appendix E):	The underlying basalt found at the site will significantly increase the size and installation cost
"The Naalehu site has been mapped by the SCS as consisting of NaC and NaD, Naalehu series soils.	or I.W.S. As a solution, the engineer could show in the bid document/construction drawing that basalt rock layer is likely to be encountered, where the depth of the rock layer varies. The
NaC and NaD are Naalehu series soils, which consists of well-drained sith clay loams that formed in volaanic ash. This soil normally overlies pahoehoe or a'a lava flows at depths of more than 40 inches.	Contractor should price in dealing with rock layer in the bid. Further, it will be important to exercise caution during excavation due to the potential to encounter underground cavities and lava tubes. Appendix G outlines the proposed solution if lava tubes are encountered.
During our site reconnaissance, we noted that the Naalehu site had been graded to form terraces with a road and adjacent homes on each terrace. Between terraces, we noted that the between terrace slopes	8.6 Soil Permeability: Percolation Test Results
consist of relatively thin near surface volcanic ash soils overlying basaltic rock. The boring at the Septic Tank site area was drilled to a depth of 35 feet. This boring encountered topsoil (one foot) overlying ash soil to a depth of 4.5 feet. Broken and highly vesicular basaltic rock was encountered for the first 1.5 feet. The deeper rock appeared to become moderately fractured and vesicular, with higher recovery and RQD."	IWS sizing is based on the percolation rate of the receiving soil. Percolation rates were conducted on July 31^{st} and August 7^{th} , 2023, at the Nāʿālehu site and ranged from 7-15 minutes/inch. The tests were conducted to a depth of 4-4.5 feet at four sites distributed across the project (Figure 1.25). Mainly soft dirt with occasional rocks with varying sizes was encountered during the tests with the full percolation test reports included in Appendix H.
A THAT IN THE TANK	TUAC: (3) 9-5-026-071
	Accel and accel
	Agents, damagered and Apple Processing Apple per VACIAS Sol Map
WAWAL HO HIGHWAY	THK (D) 9 5 202011 Control Provide The Control
Figure 1.24: Boring hole locations in Nāʿālehu, adapted from Masa Fujioka & Associates.	A OVERALL MAP
Figure 1.24 displays the boring hole locations from the analysis conducted by Masa Fujioka & Associates in Ná'ālehu. The detailed boring logs of these seven locations is located in Appendix E.	Figure 1.25: Percolation test locations in Nā'ālehu.
ENGINEERING PARTNERS	ENGINEERING PARTNERS
•	!

The A F A F A F A F A F A F A F A F A F A					
Nome Tech large Commerciance	Table 1.8: Percolation	test results			8.9 Access of Large Construction Equipment
rear (3) 9-5-03-03 10 et 4. (07)/13.13 tolow rear (3) 9-5-03-031 10 et 4. (07)/13.13 tolow rear (3) 9-5-03-031 10 et 4. (07)/13.13 tolow rear (3) 9-5-03-031 15 et 4. (07)/13.13 tolow rear (3) 9-5-03-031 (00) we at the the the the the the the the the th	Measure	TMK Number	2023 EPI Nă'ălehu LCC Replacement PER (min/in)	Test Date	Excavators, loaders, and cranes are examples of some of the large construction equipme needed to install an IWS. Access for this equipment is often obstructed by low roof overh blockage by permanent structures, overhead utilities, vegetation, or from insufficient pat
test (3) 9-5-04:011 7 0-4.5 h (7)31/23 130 m Test (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (3) 9-5-04:031 15 0 4.1 (4) 0 0 1.1	Test 1	(3) 9-5-025:019	10 @ 4 ft	07/31/23 8:00am	widths. Accommodating such equipment on especially small lots often requires temporaril removing and reconstructing of fencing, walls, existing landscaping, and in some cases sn
reast (3) 9-5-03/660 (6) 9/123 (00m) (10) 1/23 (0m)	Test 2	(3) 9-5-024:011	7 @ 4.5 ft	07/31/23 1:30pm	structures.
 Tartical and the statistic of the sector and the statistic and the stat	Test 3	(3) 9-5-024:068	8 @ 4 ft	08/07/23 12:00pm	Access should be considered for installation as well as future maintenance activities. Opportunii to resolve access issues include:
 1 Tatik tas 1 Tatik tas 1 Tatik tas 1 Tatik tas parter for the system is concreted area. The greater of concrete or tradific compresses the solin (distribution systems and affects the greater stratement of properties. In this source or tradific compresses the solin (distribution systems and affects the greater of concrete or tradific compresses the solin (distribution systems and affects the greater of concrete or tradific compresses the solin (distribution systems and affects the greater of concrete or tradific compresses the solin (distribution systems and affects the greater of concrete or tradific compresses the solin (distribution systems) to the system and a facts the greater of compressions with the system is concrete a settic task and/or 1+20. 2 Basing subfacts action a settice and solicity, etc.) to prevent which a stating and track and the constructions in task and the system as and or 1+20. 3 Basing structures limit available speed for limit available speed for limit available speed for limit and set and a limit available speed for limit and set and a limit and set and a limit and set and a limit and a set and a limit a solin (the pump hose with) while a structures in the solure of a limit and set and a limit a constructures and a limit and set and a limit and set and a limit a solure of the solure of a limit and set and a limit and set and a limit and set and a limit and and set and a limit and a limit a set and a limit and a limit a set and a limit a constructures and a limit a constructures and a limit a set and set and a limit a limit limit a limit a limit limit limit limit a limit limit li	Test 4	(3) 9-5-026:071	15 @ 4ft	08/07/23 8:00pm	INS Placement: Front-yard installations are recommended for homes without sufficient p
 It is generally not good practice bineses the solin derivative system for mathemance. However, it is sometimes unavoidable poptications are need to obtain a frection or participation or protection. For the system in derivative system is and/orth-20 transfer stated chambered disposal beds. A threat Access Routes. Access to a property's backyard may spatialy or the system is and/orth-20 transfer stated chambered disposal beds. A threat Access Routes. Access to a property's backyard may appending threat and and taffic rated undermeating and and taffic rated chambered disposal beds. A threat Access Routes. Access the a property's backyard may appending threat access y to incorporate traffic or the system allowing the need for obtain mark access to a property's backyard may appending threat access points to the provide the pumped every 3 to 5 years depending on the size of the building actures in an analysis access points to the provide the pumped every 3 to 5 years depending on the size of the building actures in an access point to be building actures in an access point to the pumped every 3 to 5 years depending on the size of the building actures in an access point to the pumped activity actes and mainters access points to the pumped activity actes and mainters access points to the pumped activity access points and any access points to the pumped activity access points and any access points access and and acceleration and accession pactorements and any access points and any access points and any access points and acceleration access and and acceleration access and and acceleration access and and acceleration access and acceleration access and and acceleration acceleration acceleration accelerateration acceleration acceleration acceleration acceleration a	8.7 Traffic Area				 widths to accommodate equipment access into the backyard. Lightweight or Cast-in-Place Technologies: The use of a crane can be avoided by spec cast-in-place concrete septic tanks instead of precast varieties for particularly inaccessible
 Attende Access from a propertify backyard may a present is and reprire and target careful and target or a system may be installed in the end to bat and target to the part of the property is backyard may appearly that end and target target and target to the system is designed to that end and target targ	It is generally not good presence of concrete o	d practice to install an IWS u or traffic compresses the so	under a trafficked or othe	rwise concreted area. The	locations. Alternatively, plastic and fiberglass offer lightweight alternatives for simplifie installation.
 adjoint careful protects. Such access will need to obtain adjoint of neal protects. Such access will need to obtain adjoint of neal protects acreate adjoint of neal protects. Such access will need to obtain adjoint of neal protects acreated the adjoint of neal protects. Such access will need to obtain adjoint of neal protects acreated the adjoint of neal protects. Such access will need to obtain adjoint of neal protects acreated the adjoint of neal protects. Such access will need to obtain adjoint of neal protects acreated the adjoint of neal protects. Such access will need to obtain a diport of near the adjoint of neal protects acreated the adjoint of neal protects. Such access will need to the adjoint of neal protect and access and healts (EA) 20 wastewater and/or such access points to the protect and and access points that the design of the access point of neal protect and and any account for the number of permitted barding adding adding actures the wastewater affort of the such prosess and the adjoint of near black and any account for the number of permitted barding adding acting adjoint access points the adjoint of the such prosess are typically includes the the prop for a success and healts (EA) 20 wastewater and/or such access points the access properties of the permitted accurates will explicit access and adving the construction prace. The access properties during the construction prace access properties access properties of the properties access properties access properties access properties during the construction prace access properties accourding to access and adving the construction prace access p	accessibility of the syst	tem for maintenance. Howe	ever, it is sometimes unav	voidable on particularly	Alternate Access Routes: Access to a property's backyard may be accessed from an
 Turthermore, for front-yard IWS installations, it would be necessary to incorporate trafficulate related channels expert and on strategies of the purple deviated. Turthermore, for front-yard IWS installations, it would be necessary to incorporate trafficulate related inspections, septage pratries (wall, boliards, etc.) to prevent vehicle traffic over the system, allowing the need for the purple deviated. 3.8 Existing Structures init available space for IWS installation. There is a possibility that some of the sating structures init available space for IWS installation. There is a possibility that some of the sating structures internated structures of the purple deviated. 3.8 Existing structures into available space for IWS installation. There is a possibility that some of the sating structures into available structures of the amount/extent of unpermitted structures of the watterwater and/or studge from the task, the ill must be removed the structures apoints to the purple deviated. 3.8 Existing structures into available space for IWS installation. There is a possibility that some of the sating structures are unpload design of the WS beyond its origing additons can increase the watterwater and/or studge from the task, the ill must be removed and the tracted canone with the State of Hawaii Department of Health have concluded that the design of the WS beyond its origing performance and on the INS beyond its origing performance and on the ison increase the watterwater and on the INS beyond its origing performance and on the INS beyond its origing performance and on the ison increase the structures and includes estates that homeowners on the instructures and includes estates are thoreowners on the ison increase the structures and includes estates are thoreowners on the ison increase. Unpermitted structures and includes estates are the purp beserver and or with the INS beyond its origing performance and on this project. Increasely and includes est	spatially constrained pr vehicle path of travel p	properties. In this event, a symptotic provided the system is desig	ystem may be installed ur jned to that end and traffi	derneath a driveway or c rated treatment	adjoining neighbor's property. Such access will need to obtain permission of the neighboring homeowner.
Furthermore, for front-yard WS installations, it would be necessary to incorporate traffic- rated INS components to be alleviated. IWS maintenance typically includes regular inspections, septage p cleaning when a traditional septic tank is used. If systems are not straffic-rated INS components to be alleviated. 8.8 Existing Building Structures intravel intravellable space for INS installation. There is a possibility that some of traffic-rated INS components, present is unserning and incore the system. Allowing the need for the building structures intravellable for the pump hose which will pump wastewater for the pump hose which will pump wastewater to the pum prostructures are unpermitted. Just the amount/extent of unpermitted structures and incorporate to the INS soon of the INS components, making maintenance and servicing more challenging, unpermitted structures increase the wastewater for the pump hose which will pump wastewater to the pum prostructures and increases points to the INS components, making maintenance and servicing more challenging, unpermitted bachmonk the asting structures and increases points to the INS components, making maintenance and servicing more challenging, unpermitted bachmonk from the County Right-of-Way where the pump truck is prostructures and increases are typically 150 feet long and can usually prostructures and increases are typically 150 feet long and can usually prostructures and increases are the assertance of the INS capacity. The successful completion of this project, necessitates that homeowners of each in Master and the county to continuoush notify indowners of maintenance and servicing process, and drink INS system will be insti- notify and the constructed. Intermitted structures and includes regular inspections. Intermisted structures and instinter and homeowners on the county to conthousely noth increases structures and increasing acting	traffic related chamber	irred disposal beds.	ים אחרוו מא רטוורו ברב אבארוו		8.10 Access of Maintenance Equipment
 ated IWS components in potential parking areas. One alternative could be constructing barriers (wall, Delards, etc.) to prevent vehicle traffic over the system, allowing the need for traffic rated IWS components to be alleviated. 8. Existing Building Structures 9. Existing Building Structures 9. Existing Building Structures 9. Existing Structures 9. Existing Building Structures 9. Existing 	Furthermore, for from	it-yard IWS installations, i	it would be necessary to	incorporate traffic-	IWS maintenance typically includes regular inspections, septage pumping, and periodic
 barrers (wall, bollards, etc.) to prevent vehicle traffic over the system, allowing the need for traffic over the system must be restricted in the traffic over the system must be restricted in the traffic over the system must be restricted in the traffic over the system must be restricted in the traffic over the system must be restricted in the traffic over the public structures and nononents, the existing structures and servicing more challenging, unpermitted structures and nononents with the structures will require similar structures and for signal and the service and servicing more challenging, unpermitted structures and servicing more challenging, unpermitted structures and nononents, making additions can in Xiefailen to the two the INVS beyond its original design of the INVS capacity. Previous on the success for the pump hoses are typically 150 feet long and can usually propersy from the county light-of-furty in the counts usually project necessitates that homeover or non addressing unpermitted structures and including existing permitting process, and during the construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to maintain a construction phase. The County will need to ma	rated IWS component	ts in potential parking are	as. One alternative coul	ld be constructing	cleaning when a traditional septic tank is used. If systems are not inspected, septic tanks
 traffic-rated IWS components to be alleviated. Existing Building Structures Existing Structures Existing Structures Building Structures 	barriers (wall, bollard	is, etc.) to prevent vehicle	e traffic over the system	،, allowing the need for	should be pumped every 3 to 5 years depending on the size of the tank, the number of
8. Existing Building Structures 8. Existing Building Structures in the system must be restring structures immust be restring structures are unpermitted, but the amount/extent of unpermitted structures are unpermitted building additions can increase the wastewater flow to the IWS beyond its original design capacity. Previous conversations with the State of Hawaii Department of Health have concluded that the design increases the wastewater flow to the IWS beyond its original design capacity. Unpermitted structures will require further coordination and discussions between the County indoneover on addressing unpermitted structures and neuloding existing permitted in the total that the design of the pump function prive indoneover on the structures and neulom only account for the IWS capacity. Unpermitted structures and neulom only account for the IWS capacity. Unpermitted structures and neulom and discussions between the County and homeowners on addressing unpermitted structures and neulom and discussions between the County and homeowners on addressing unpermitted structures and neulon and discussions between the County and homeowners on addressing unpermitted structures and neulon and discussions between the County and homeowners. The success properties during the construction phase increases proved and necolation and discussions between the County and homeowners. The success properties during the construction phase. The County will be used and unpermitting accounts of the design of the partition addition will be used as a succease addition of the parting add	traffic-rated IWS com	nponents to be alleviated.			building occupants, and household appliances and habits (EPA, 2002). In order to remove
Existing structures limit available space for IWS installation. There is a possibility that some of the existing structures are unpermitted, but the amount/extent of unpermitted structures obstruct access points to the IWS components, making maintenance and servicing more challenging, unpermitted building additions can increase the antiser and servicing more challenging, unpermitted building additions can increase the antiser and servicing more challenging, unpermitted building additions can increase the antiser and servicing more challenging, unpermitted building additions can increase the avaitent part from the County Right-of-Way where the pump truck is provining maintenance and servicing more challenging, unpermitted building additions can increase the avaitent and nere constructions with the State of Hawaii Department of Health have concluded that the design can astwater to the Bagin of the IWS beyond its organic structures will require further coordination and discussions between the County and unit and unit and baneowner on addressing unpermitted structures and including existing permitted and homeowner on addressing unpermitted structures and including existing permitted structures and including existing permitted and homeowner on addressing unpermitted structures and including existing permitted and homeowner on addressing unpermitted structures and including existing permitted and homeowner on addressing unpermitted structures and including existing permitted and homeowner on addressing unpermitted structures. The County will he design, permitting for continuously notify landowners of a milestones, such as the selection of which IWS system will be obtained for during the construction phase. The County will he doration of the successful and structures than and the successful and structures and including existing permitting account for the unpervise during the successful and anting the successful and antistin a construction phase. The County will be during the successful and antistin a construction phas	8.8 Existing Building St	tructures			wastewater and/or sludge from the tank, the lid must be removed, and the tank is pumpe through an access port. Debris surrounding the system must be removed, and access mu:
 the visiting structures are unpermitted, but the amount/extent of unpermitted structures obstruct access points to the IWS components, making maintenance and servicing more challenging, unpermitted building additions can making maintenance and servicing more challenging, unpermitted building additions can in Nãaileiu should only account for the number of permitted buildings will not contribute to the design of the IWS capacity. The successful completion of this project necessitates that homeov of each TMK restructures will require further coordination with landowners or more challenging, unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted structures will require further coordination and discussions between the County and homeowner on addressing unpermitted structures that may need to be temporarily removed and reconstructed. Unpermitted structures that may need to be temporarily removed and reconstructed. 	Existing structures lin	nit available snace for IW ^G	S installation There is a	nossibility that some of	available for the pump hose which will pump wastewater to the pump truck parked in clos
 making maintenance and servicing more challenging, unpermitted building additions can making maintenance and servicing more challenging, unpermitted building unpermitted building with the State of Hawair' Department of Health have concluded that the design of the IWS exponds that the design of the IWS capacity. The successful completion of this project necessitates that homeov of each INK process, and during the construction phase the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. The successful completion of the design unpermitted buildings will not contribute to the design of the IWS capacity. The success properties during the construction phase. The County will need to maintain a coordination with homeowners/tenants through the duration of the be successful. 	the existing structure present is unknown. ¹	ss are unpermitted, but the Not only can structures ob	e amount/extent of unp sstruct access points to	ermitted structures the IWS components,	proximity. Pump hoses are typically 150 feet long and can usually access the backyard of property from the County Right-of-Way where the pump truck is parked.
conversations with the State of Hawairi Department of Health have concluded that the design of each in Näřalehu should only account for the number of permitted bedrooms for each TMK parcel. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will require further coordination and discussions between the County and homeowner on addressing unpermitted structures and including permitted obstructing structures that may need to be temporarily removed and reconstructed. Second and the construction phase. The County will need to maintain a coordination with homeowners/tenants through the duration of the be successful.	making maintenance increase the wastewa	and servicing more challe iter flow to the IWS beyon	enging, unpermitted buil of its original design car	lding additions can pacity. Previous	8.11 Coordination with Landowners or Tenants
of each in Nărălehu should only account for the number of permitted bedrooms for each TMK the design, permitted buildings will not contribute to the design of the IWS capacity. Dipermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity. Unpermitted buildings will not contribute to the design of the IWS capacity to continuously notify landowners or unpermitted buildings will not contribute to the design of the IWS capacity. Individual Right-of-Entry (ROE) forms to access properties during the and homeowners that may need to be temporarily removed and reconstructed. Individual Right-of-Entry (ROE) forms to access properties during the and topographic surveys. Separate ROE forms will need to maintain a coordination with homeowners/tenants through the duration of the Bencine RING	conversations with th	ie State of Hawai'i Departr	ment of Health have cor	icluded that the design	The successful completion of this project necessitates that homeowners/tenants cooperat
Unpermitted structures will require further coordination and discussions between the County Unpermitted structures will require further coordination setween the County and homeowner on addressing unpermitted structures and including existing permitted and homeowner on addressing unpermitted structures and including existing permitted obstructing structures that may need to be temporarily removed and reconstructed. adding the construction phase. The County will need to maintain a coordination with homeowners/tenants through the duration of the be successful.	of each in Nā'ālehu st	hould only account for the	e number of permitted b	edrooms for each TMK	the design, permitting process, and during the construction phase of the project. It is recommended for the County to continuously notify landowners or tangate of project
and noncoversion addressing unperimed and reconstructed. and topographic structures that may need to be temporarily removed and reconstructed. and topographic surveys. Separate ROE forms will be obtained for during the construction phase. The County will need to maintain a coordination with homeowners/tenants through the duration of the be successful. Image: Find the Right of the construction phase. The County will need to maintain a coordination with homeowners/tenants through the duration of the be successful.	Unpermitted structure	es will require further cool	rdination and discussion	we capacity. Is between the County viting normitted	recommended for the obtained of continuously notify another of the market of project milestones, such as the selection of which IWS system will be installed. The County will of individual brick-of Extru-/OECS forms to accord another during the docing offer accord
ENGINEERING ENGINEERING ENGINEERING ENGINEERING ENGINEERING ENGINEERING ENGINEERING	obstructing structures	s that may need to be terr	notation and including e	econstructed.	and topographic surveys. Separate ROE forms will be obtained for construction activities
ENGINEERING PARTNERS PARTNERS					during the construction phase. The County will need to maintain a continuous dialogue an coordination with homeowners/tenants through the duration of the project for the project be successful.
PARTNERS PARTNERS	< ENGINEE	FRING			ENGINEERING
	PARTN	IER S		9	PARTNERS

Potential challenges can occur if the homeowner has a preference on the location of the IWS. If the IWS is not in a favorable location for the homeowner, the IWS layout can be moved to a location with less disturbance to the homeowner to try to accommodate this request. Solutions may include altering the system layout by adding pumps, excavating deeper, etc.

8.12 Absorption Beds vs. Seepage Pits

When setback requirements and absorption bed sizing requirements cannot be reached due to lack of horizontal yard space, seepage pits can be employed to dispose of wastewater in a vertical manner. Based on a visual walk through within the ROW roadway of the site on July 20, 2023 (Section 9), a preliminary estimate of 83 seepage pits will need to be installed in lieu of absorption beds. This estimate was made by visual inspection and would need to be confirmed by further surveys, as backyard space was not always visual from the public roads. Sites lacking the necessary space in the backyard for an absorption bed were assumed to require a seepage pit, unless significant space was present in the side or front yard. Appendix A estimates the yard area required for an absorption bed, which ranged from 330 sq. ft. to 1,330 sq. ft., based on the upper range of the percolation rate of the soil found (10 and 15 minutes/inch) and the number of permitted bedrooms in each dwelling.

8.13 Availability of Resources and Contractors

As an island state, Hawai'i faces unique challenges when it comes to the availability of IWS. The entire IWS market in Hawai'i grew from 1192 units per year in 2018 to 1414 units per year in 2021. Based on permits issued, sourcing the 194 treatment units required for this project will require an increase in statewide treatment unit supply and workforce size by 14% (DOH, 2022). This will require advanced planning to overcome this logistic hurdle. Locally based septic manufacturers include Jensen Precast on O'ahu and Chemtainer on Hawai'i Island. At present, Jensen produces approximately 40 concrete septic tanks per year. With that said, the company has stated that they have the capacity to build one septic tank per day to keep up with the needs of the project. The bottleneck to their current production rate has been cited as a shortage of inspectors, contractors, and engineers in the local market. Chemtainer manufactures polyethylene septic systems on Hawai'i Island but was unwilling to share their annual production rates. Mainland treatment system manufacturers like Orenco and Infiltrator have significantly higher production rates but will also face increased shipping costs in transporting the units to Hawai'i. Material availability and delivery will be the critical path item during construction, and the design of the IWS needs to ensure the possibility of different manufacturers being utilized is accommodated.

Nā'ālehu Individual Wastewater System Preliminary Engineering Report

9. Technically Challenging Sites

A visual walk-through within the ROW was conducted at Naʿalehu by EPI on July 20th, 2023. The aim of this site visit was to identify common conditions that are present in the community that would render an IWS installation technically difficult, due to the challenges of IWS construction. Five properties were initially flagged during the walk-through to have such challenging conditions, however the total number of properties with potential technical challenges is not limited to following households identified. Each dwelling may or may not have its own unique challenges and this section serves as an example of which challenges may be encountered during IWS design and construction.



Figure 1.26: Examples of sites with technically challenging conditions are marked in yellow (but are not limited to these five properties).





Table 1.9: Examples of Nā alehu Problematic Sites

Problem Description

Site

elevation than the house, which may require a pump for IWS installation and will be costly to remove and reinstall. The IWS components will likely need to be also need to be traffic-rated. The IWS components retaining walls and stairs that limit available space to be included in the system design. The IWS will could also be installed 3-4 feet deeper than usual under the driveway to ensure proper wastewater installed under the driveway, which is at a higher drainage from the household, and to negate the Site 1 has physical barriers such as concrete need for a pump.

be installed in addition to the septic tank. Installing a seepage pit in lieu of an absorption bed will require a variance application to the DOH. The seepage pit will the property line and existing structures and will thus absorption bed, and thus a seepage pit will need to also not meet the defined setback requirements to The driveway will not allow sufficient space for an require an additional variance application.

be relocated or removed to ensure space for the IWS Site 2 has limited available accessible space for IWS installation. Communication with the homeowner will need to be made to see which household items can components.

existing structures and will thus require an additional tank. Installing a seepage pit in lieu of an absorption space for an absorption bed, and thus a seepage pit bed will require a variance application to the DOH will need to be installed in addition to the septic The front and side yards do not allow sufficient The seepage pit will also not meet the defined setback requirements to the property line and variance application.

2

yard. Physical barriers such as the estimated 6-7 feet high retaining wall may need to be removed to access the available yard space. In addition, this site equipment to be able to reach the lower level of the Site 3 is situated on a slope, which may be difficult to maneuver with installation equipment. During below the street grade, which will require deeper constructed with material such as gravel for the contains a basement that is estimated to 6 feet installation, a temporary ramp may need to be nstallation of IWS components or a pump.

m



Address: 95-1204 Kukui Rd, Nā'ālehu, HI 96772



Address: 95-5586 Nahele St, Nā'ālehu, HI 96772



Address: 95-5581 Nahele St, Nā'ālehu, HI 96772

PARTNER S

W







10. Operational Considerations

An effective Individual Wastewater System (IWS) management strategy is crucial to ensuring distributed treatment systems are maintained and operated in a way that ensures they are functioning properly and effectively treating wastewater. This strategy may include but is not limited to:

- Monitoring: Regular inspections of system components, such as septic tanks and drain fields, ensure they are functioning properly and identify and address any issues that may arise.
- Maintenance: Proper maintenance of the system is also crucial, including regular pumping of septic tanks, cleaning and maintenance of the distribution systems, and proper maintenance of the treatment components. Regular maintenance can help prevent issues such as clogs and backups, which can lead to costly repairs and potential health hazards.
- Regulatory Compliance: Necessary permits and licenses are obtained for the system, and required inspection and reporting schedules are met with the local regulator.
- Community Education: Information and training on proper usage and maintenance of the systems are provided to homeowners, and any concerns or questions that may arise are addressed.

In Hawai'i, centralized wastewater treatment plants and cluster systems are regulated and inspected by the Department of Health (DOH) Wastewater Branch (Hawai'i Administrative Rules 11-62). State-licensed WWTP operators are required for oversight of Wastewater Treatment plants to ensure that systems are inspected, operated, and maintained as required. A similar regulatory requirement does not exist for IWS in Hawaif. The State DOH Wastewater Branch is responsible for regulating IWS while operation and maintenance are currently the responsibility of the individual homeowner. If IWS were selected to serve Nā'ālehu Community, maintenance responsibilities could be distributed in a number of ways. Per Voluntary National Guidelines for Management of Onsite and Clustered (Decentralized) Wastewater Treatment System, the EPA outlined five management models that can be used for the operation and maintenance of IWS (Table 1.10).

When selecting an appropriate management model for a network of IWS it is important to take into account the regulatory and cultural framework within which the IWS is situated. As it stands in Hawai'i, IWS are currently managed similar to a combination of management Models 1 and 2 (DOH, 2016):

- Model 1: Homeowner Awareness. The DOH allows septic systems to be managed by the homeowner under this model. Homeowners own and operate their own IWS and are responsible for keeping the system in good working order.
- Model 2: Maintenance Contracts. The DOH requires that Aerobic Treatment Units



Nā'ālehu Individual Wastewater System Preliminary Engineering Report

(ATUs) are managed by a state licensed wastewater operator using this model. Homeowners are required to have an active service contract with a certified operator or factory certified representative, and a copy of that active service contract must be submitted annually to the DOH (DOH, 2016). Elevated regulation around the operation of ATUs is a reflection of their increased mechanical complexity and associated maintenance demands.

Table 1.10: The five management models for IWS maintenance (EPA, 2003)

	The Fiv	e Management	Models	
Model 1	Model 2	Model 3	Model 4	Model 5
Homeowner Awareness:	Maintenance Contracts:	Operating Permits:	Responsible Management Entity (R:ME) Operation and Maintenance:	RME Ownership:
specifies appropriate program elements and activities where treatment systems are owned and operated by individual property owners in areas of low environmental sensitivity. This program is adequate where treatment technologies are limited to converticioal systems that require little owner attention. To help ensure that timely maintenance is performed, the regulatory authority mails maintenance is performets to owners at appropriate	specifies program elements and activities where more complex designs are employed to enhance the capacity of conventional systems to accept and treat wastewater. Because of treatment complexity, contracts with qualified technicians are needed to ensure proper and timely maintenance.	specifies program elements and activities where sustained performance of treatment systems is critical to protect public health and water quality. Limited- term operating permits are issued to the owner and another term if the owner demonstrates that the terms and conditions of the permit. Performance- based designs may be incorporated into programs with	specifies program elements and activities where frequent and highly reliable operation and maintenance of decentralized systems is required to ensure water resource is required to ensure the operating permit is issued to an RME instead of the property owner to provide that the appropriate maintenance is performed.	specifies that program elements and activities for treatment systems are owned, operated, and maintained by the RME, which removes the property owner from responsibility for the system. This program is analogous to central severage and provides the greatest assurance of system performance in the most sensitive of environments.
intervals.		management controls at this level.		



The AOC stipulates that the County of Hawai'i must administer a more active management strategy than is typical in Hawai'i, either a Model 2 (Maintenance Contract) or Model 3 (Operating Permit) management strategy for a network of IWS at Nă'älehu. These models reflect varying degrees of responsibility to the County and homeowner (Table 1.11). Four potential variations of these models are outlined here for implementation on this project:

- Management Model 24: Maintenance Contract with County In-House Staff The County employs and trains an in-house IWS management team; purchases and maintains its own pumping/hauling equipment; and administers the management program. The homeowner pays a monthly sewer fee that covers a portion of the costs.
- Management Model 28: Maintenance Contract with Third-Party Service The County administers the management program, keeps an operations and maintenance (O&M) schedule, and contracts out O&M activities to a third-party service provider. The homeowner pays a monthly sewer fee that covers a portion of the costs.
- Management Model 3A: Operating Permits with O&M by Users

The County issues an operating permit to the homeowner; keeps an O&M schedule; and sends out maintenance reminders to homeowners. The homeowner is responsible for contracting a third-party service provider to conduct maintenance.

Management Model 38: Operating Permits with O&M Voucher by County The County issues an operating permit to the homeowner; keeps an O&M schedule; and sends out maintenance reminders with service vouchers to homeowners. The homeowner pays a sewer fee and is responsible for contracting a third-party service provider to conduct amual maintenance using the voucher.

•

These management strategies presented here are required by the AOC but are also unique to Hawai'i and will present a number of barriers for implementation at the legislative, regulatory, and public levels. The Nă'ălehu Community and Hawai'i's stakeholders at large are accustomed to Management Model 1, which is the standard practice across the State of Hawai'i.

Table 1.11: County and homeowner responsibilities under variations of the EPA Management Models 2 and 3

Management Model	Brief Description of Management Model	County's Responsibility	Homeowner / User's Responsibility	Pros	Cons
ZA	Maintenance Contract w/ County in- house staff	 Funds design and construction of IWS Purchase equipment & train M/S operator OBM of IVWS including truble calls OBM of IVWS includers to Keeping record of OBM log Send out ordex and terminders to newores Submit WS inspection reports and variance renewals to State DOH 	 Report IWS problem to Country Coopers and allows Coopers and allows Country staff to enter private property and provide matternance of IWS Meintain cleanance to IWS for ess y access 	 Best control on O&M schedule Ensure best IWS performance 	 High est cost May not receive cooperation from some homeowners/weeks Homeowner may have more trouble calls between homeowner & County on homeowner & County on phrmbig graph cost & 1WS repair cost & 1WS
58	Maintenance Contract w/ 3rd Party Service	 Funds design and construction of IVIS Select/prequality certain 3rd party service provider (formper) Issue DO Punnersk B junner for annual inspection and trouble calls Keeping record O 08M tog Keeping record O 08M tog Send out notices and reminders to homeowies and eminders to homeowies State DOH Subm TMS State DOH 	Report IWS problem to Country Coopers and allows Service providers to enter private property and provide mattenance of IWS Meintain cleanance to IWS for ess y access	Better control on O&M schedule Schedule Schedule Schedule Schedule Less County staff to terformance Less County staff to the outpring to the outpring to burchase & maintain	I. Higher cost May not receive cooperation May not receive cooperation Homeowner may have more Homeowner may have more torould entitie between homeowner & County on homeowner & County on phrabig graph cost & WS repair cost & WS
ЗА	Operating Permits w/ O&M by Users	 Funds design and construction of IWS Keeping record of OBM INg Send out notices and reminders to neavores Enforce rules and regulations Ensues permit to homeowner to use, operate & maintain the IWS 	 Contracts with preferred purper/purper/purper/purper/ tet WS Pay for the OBM service Submit OBM record to County Submit UWS inspection reports and variance enemals to State DOH 	 Least cost to County No O&M staff or equipment No trouble calls 	 Least control for TWS complance & performance Conflict with non-complaint homeowners Highlest cost to homeowner
æ	Operating Permits w/ O&M Voucher by County	 Funds design and construction of IVS Keeping record of Rol Migg Send out notes and termidets to nonewners Enforce vides and regulations Pre-select qualifying service providers Issue vouchers to homeowner for annual inspections and promovines for annual inspections and promovines for annual inspections and promovines for operate & marking the IVS 	 Contracts with preferred pre- qualitiest pumper/pumper/ participant the IWS Pay for the annual D&M service with voucher Submit UMS inspection County by pumper Submit IVMS inspection reports and variance reveals to State DOH 	 Reasonable control on 08M Reasonable IWS performance Less County staff to train Nopumping/brauing equipment by prudvase & mantain For the control of the by baid County exist to be paid 	1. High cast to County 2. Less control of all IWS



11. Economic Considerations

When assessing the overall cost of a given system, it is important to consider the net present value lifecycle cost taking system lifetime and installation, maintenance, and operation costs into account. These costs can be affected by a variety offactors, including: Type of treatment and disposal system: The selection of different types of treatment systems such as traffic rated tanks or aerobic treatment significantly affects overall installed cost. For disposal, seepage pits are significantly lower cost than absorption fields when it is possible to convert an existing cesspool. Site specific conditions will control which options are required. For residential IWS installations subject to State procurement regulations, capital costs per household are typically in the range of \$30,000-\$100,000 (Table 1.12). Due to the potential need for repair and reconstruction of property site elements (fencing, walls, structures, etc.) that may need to be removed during IWS installation, the cost per household in Na⁷afelu can be up to \$150,000. At this stage of the project, it is fully unknown the extent of reconstruction of existing site elements, and this would need to be furthed during the design phase of the project, which can impact the stated estimated cost.

Table 1.12: Installation cost estimates for a standard septic tank installed in conjunction with an absorption bed (left) and seepage pit (right). Figures are based on a 3-bedroom house and a percolation rate no slower than 5 min/inch.

	Standard Abs	sorption Bed	Seepa	ige Pit
	Low (non-traffic)	High (Traffic Rated)	Low (non-traffic)	High (Traffic Rated)
Septic Tank	3,000.00	7,000.00	3,000.00	7,000.00
D-Box	750.00	2000.00	I	Ì
Sewer pipe	250.00	250.00	250.00	250.00
Leach field-pipe/ chamber	500.00	3,000.00	I	I
Leach field-gravel	1,000.00	500.00	1	Ì
Cone. Ring	1	I	3,000.00	3,000.00
Cone. Cover	I	I	2,500.00	4,000.00
Soil replacement	1,500.00	1,500.00	I	Ē
Inspection ports	500.00	500.00	I	į
Misc. material	2,000.00	2,000.00	2,000.00	2,000.00
Material Total	\$ 9,500.00	\$ 16,750.00	\$ 10,750.00	\$ 16,250.00
Labor / Equipment	7,500.00	15,000.00	7,500.00	15,000.00
Remoteness	5,000.00	5,000.00	5,000.00	5,000.00
Trucking for spoils	3,000.00	3,000.00	3,000.00	3,000.00
Tight working space	3,000.00	10,000.00	3,000.00	10,000.00
Relocate/reinstall/ repair	5,000.00	100,000.00	5,000.00	100,000.00
TOTAL	\$ 33,000.00	\$ 149,750.00	\$ 34,250.00	\$ 149,250.00

ENGINEERING P A R T N E R S

Nā'ālehu Individual Wastewater System Preliminary Engineering Report

 Operations and Maintenance cost: Operations and maintenance cost also play a big role in the overall cost of IWS. Annual maintenance costs to the County and homeowner vary depending on the management strategy. It is estimated that bringing maintenance in-house is the most affordable option (Table 1.13). Annual costs over a 20-year service lifetime are further expounded in Appendix B.

Table 1.13: The costs associated with four IWS management models, assuming septic systems with leach fields (Appendix B).

	Average Ani Cou	nual Cost to Inty	Average An Home	nual Cost to owner	Net	Total	
Management Model	Third-Party Service Provider	In-House	Third-Party Service Provider	County Sewer Bill ¹	Cost to County	Dollars Spent	
2A: Maintenance Contract w/ County in- house staff	I	(\$956)	·	\$600	(\$356)	(\$956)	
2B: Maintenance Contract w/ 3rd Party Service	(\$783)	(\$572)	ı	\$600	(\$755)	(\$1,355)	
3A: Operating Permits w/ 0&M by Users	I	(\$572)	(\$733)	,	(\$572)	(\$1,305)	
3B: Operating Permits w/ O&M Voucher by County	(\$533)	(\$572)	1	\$600	(\$505)	(\$1,105)	

 1 The annual average County sewer bill is based on an average of \$50/month sewer bill for house



Nā'ālehu Individual Wastewater System Preliminary Engineering Report	Nā'ālehu Individual Wastewater System Preliminary Engineering Report
	files/2016/03/14_Gray_Water_GL.pdf
References	Hawaii State Department of Health. (2016) HAR 11-62 Wastewater Systems. https://health.hawaii. gov/opppd/files/2015/06/11-62-Wastewater-Systems.pdf
Babcock, R., Barnes, M., Fung, A., Goodell, W., & Oleson, K. (2019). <i>Investigation of Cesspool</i> <i>Upgrade Alternatives in Upcountry Maui Final Report</i> . University of Hawaii Manoa.	Hawaii State Department of Health. (2017). Relating to Cesspools and Prioritization for Replacement.
Babcock, R., Lamichhane, K. M., Cummings, M. J., & Cheong, G. H. (2014). Condition assessment	nttps://neatm.nawaii.gov/oppdg/nies/201//12/Att-125-Hb1244-HD1-5D3-CD1-23m- Legislature-Cesspool-Report.pdf
survey or unsite sewage uisposal systems (OSDSS) in navani. water saence and	Hawaii State Department of Health. (2022). <i>Iws Final Approvai Date Report (04/01/2017-04/01/2022).</i>
Technology , 70(6), 1083–1089. https://doi.org/10.2166/wst.2014.336	Hawaii State Department of Health Wastewater Branch.
Carollo Engineers (2021). <i>Cesspool Conversion Technologies Research Summary Report</i> . Hawaii	Incinerating Toilets Inc. (2023). <i>Cinderella Comfort Incinerating Toilets</i> . Incinerating Toilets Inc.
State Department of Health.	https://incineratingtoilets.com/us/product/cinderella-comfort/
Hawaii County Code, Chapter 21 Sewers (2016). https://www.hawaiicounty.gov/home/	Infiltrator. (2023). Advanced Enviro-Septic. Infiltrator Water Technologies. https://www.
showpublisheddocument/46/637032750278570000	infiltratorwater.com/products/presby-environmental/advanced-enviro-septic/
Hawaii State Department of Business, Economic Development and Tourism. (2021). The State of Hawaii Data Book: A statistical Abstract. State of Hawaii Department of Business, Economic Development, and Tourism. https://files.hawaii.gov/dbedt/economic/databook/db2021/ DB2021_final.pdf	Lekven, C. (2019). <i>Pahala Wastewater Treatment Plant Preliminary Engineering Report</i> . Brown and Caldwell. https://www.epa.gov/sites/default/files/2020-03/documents/pahala_final_ea_vol_2_508_2020-02-24.pdf
Environmental Protection Agency. (1996, June). <i>Wastewater Treatment: Alternatives to Septic</i>	Mezzacapo, M., & Shuler, C. (2021). 2021 <i>Hawaii Cesspool Hazard Assessment and Prioritization</i>
<i>Systems</i> . Environmental Protection Agency.	<i>Tool</i> . University of Hawaii. https://health.hawaii.gov/wastewater/files/2022/01/
https://nepis.epa.gov/Exe/ZyPURL.cgi?Dockey=20013JJ8.txt	priortizationtoolreport.pdf
Environmental Protection Agency. (2002). USEPA Onsite Wastewater Treatment Systems Manual.	Mohamed, R. (2009). Why households in the United States do not maintain their septic systems
In United States Environmental Protection Agency (pp. 1–367).	and why state-led regulations are necessary: Explanations from public goods theory.
https://www.epa.gov/sites/default/files/2015-	<i>International Journal of Sustainable Development and Planning</i> , 4, 143–157. https://doi.
06/documents/2004_07_07_septics_septic_2002_osdm_all.pdf	org/10.2495/SDP-V4-N2-143-157
Environmental Protection Agency (2003). Voluntary National Guidelines for Management of Onsite	New Zealand Distributors. (2018). AES Homeowners Manual AES (Advanced Enviro-
and Clustered (Decentralized) Wastewater Treatment Systems. Environmental Protection	Septic TM) Owners Manual (pp. 1-12).
Agency. https://www.epa.gov/sites/default/files/2015-06/documents/septic_guidelines.pdf	Nuflow Wide Bay. (2015, September 15). <i>Blocked Drains Bundaberg</i> . Nuflow Wide Bay.
Environmental Protection Agency. (2018) <i>Types of Septic Systems</i> [Overviews and Factsheets].	https://www.nuflowwidebay.com.au/understanding-the-mysteries-of-septic-
https://www.epa.gov/septic/types-septic-systems	tanks/
Environmental Protection Agency (2010). Code of Federal Regulations, 40 C.F.R §144.80 Title 40—	Peterson, A. (2010, May 2). Plantation Era Is Gone, and Pahala Lives On: Historic Preservationist
Protection of Environment. https://www.govinfo.gov/content/pkg/CFR-2014-title40-vol23/	Julia Neal Uses Hammer and Wood to Pull the Community Together. <i>Ie Ola Magazine</i> .
xml/CFR-2014-title40-vol23-part144.xml	https://keolamagazine.com/community/plantation-era-gone-pahala-lives-on/
Evergreen. (2023). <i>Separett Villa urine separating toilet</i> . Evergreen. https://evergreen.eco/en-us/	Pine Creek Structures. (2023). <i>Excel NE Sun-Mar Composting Toilet.</i> https://www.storageshedspa.
products/urine-separating-toilet-villa-9020-12v-230v	com/sheds/other-products/17759/excel-ne-sun-mar-composting-toilet
Hawaii State Department of Health. (2009). <i>Guidelines for the Reuse of Gray Water</i> . Hawaii Department of Health Wastewater Branch. https://health.hawaii.gov/wastewater/	Presby Environmental, Inc. (2017). The Presby Wastewater Treatment System Design and Installation Specifications for Advanced Enviro-Septic ®, Enviro-Septic ® and Simple-Septic ® Wastewater Treatment Systems Minimizes the Expense
PARTNERS	ENGINEERING
PARTNERS	PARTNERS

Nā'ālehu Individual Wastewater System Preliminary Engineering Report	A Appendix A - LCC Closure Properties		A-1
Nā'ālehu Individual Wastewater System Preliminary Engineering Report	 Protects the Environment Preserves the Site. wwww.PresbyEnvironmental.com Presby Environmental, Inc. (n.d.). The Next Generation of Wastewater Treatment Technology Brochure. ShopTinyHouses. (2023). Sun-Mar Centrex 3000 Composting Toilet System. ShopTinyHouses. Com. https://www.shoptinyhouses.com/products/sun-mar-centrex-3000-central-composting-toilet-system Water Resources Research Center, & Engineering Solutions, Inc. (2008). onsist wastewater Mater Resources Research Center, & Engineering Solutions, Inc. (2008). onsist wastewater https://health.hawaii.gov/wastewater/files/2013/06/onsitesurvey.pdf 	PARTNERS	20

Nā'ālehu Individual Wastewater System Preliminary Engineering Report

TMMS 95-008-022 95-008-022 95-008-022 95-008-022 95-008-031 95-008-031 95-008-013 95-008-018-008-018 95-008-018-008-018-018-008-018-008-008-008	Address 5-31214 KUKI IROAD 5-31214 KUKI IROAD 5-31214 KUKI IROAD 5-31214 KUKI IROAD 5-31204 KUKI IROAD 5-31204 KUKI IROAD 5-3124 KUKI IROAD 5-3125 KUKI IROAD 5-3126 KUKI IROAD 5-3126 KUKI IROAD 5-3126 KUKI IROAD 5-3126 KUKI IROAD	Land Anna (acres) 1.0540 1.1810 0.1818 0.1818 0.1818 0.01818 0.2673 0.2673 0.2673 0.2673 0.2673 0.2710 0.2182 0.2182 0.2182 0.2184 0.11866 0.11867 0.11866 0.11867 0.11866 0.11867 0.11866 0.11867 0.11866 0.11867 0.1	Bidge Beetrof	Et. WW Et. WW Gan 3 2	Saptic Tank [gal] 1000 1000 1000 1000 1000 1000 1000 10	Absorption Absorption 10000000 100000 100000 1000000 1000000 1000000 1000000 1000000 1000000 1000000 100000000	Teak Size (apr A) Teak Size (apr A) Teak Size (apr A) 36 Standardina 37 Standardin	
95.0084 95.0084022 95.0084022 95.0084023 95.0084023 95.0084023 95.0084012 95.00840000000000000000000000000000000000	CADA (CALOR 10, 10, 10, 10, 10, 10, 10, 10, 10, 10,	(acreal) 10540 110540 110540 110540 110540 110540 110540 013818 01388180000000000	Ridge Bedra	w Lpcdl 1 4 4 8 600 3 3 5 600 6 6 3 3 5 600 6 <td< th=""><th>Tank [20] 10000 1000 1000 1000 1000 1000 1000 1000 1000 1000</th><th>10min/in. 313 313 313 314 315 316 405 405 405 405 405 405 405 405</th><th>Affinity/Inc. Affinity/Inc. 0<th>Bernarts</th></th></td<>	Tank [20] 10000 1000 1000 1000 1000 1000 1000 1000 1000 1000	10min/in. 313 313 313 314 315 316 405 405 405 405 405 405 405 405	Affinity/Inc. Affinity/Inc. 0 <th>Bernarts</th>	Bernarts
95.008.022 95.008.022 95.008.022 95.008.022 95.008.021 95.005.031 95.005.031 95.005.031 95.005.030 95.005.030 95.005.030 95.005.030 95.005.031 95.005.013 95.005.013 95.005.013 95.005.013 95.005.013 95.005.013 95.005.013 95.005.015 95.005.005 95.005.005 95.005.005 95.005 95.005.005 95.005	55 - 1214 (KUKI) ROAD 59 - 1214 (KUKI) ROAD 59 - 1210 (KUKI) ROAD 59 - 1210 (KUKI) ROAD 59 - 1207 (KUKI) ROAD 59 - 1207 (KUKI) ROAD 59 - 1128 (KUKI) ROAD 51 - 1158 (KUKI) ROAD	1.0540 1.01810 0.1818 0.1818 0.1818 0.01818 0.01318 0.01310 0.22310 0.22310 0.22310 0.22310 0.22310 0.2335 0.23555 0.23555 0.23555 0.23555 0.23555 0.235555 0.2355555 0.23555555555555555555555555555555555555		4 4 8 9 3 3 3 5 600 3 3 5 600 5 4 4 8 600 5 5 5 5 5 600 5	2000 2000 2000 2000 2000 2000 2000 200	900 1000 1000 1000 1000 1000 1000 1000	36 5	
9 5 035 037 9 5 035 037 9 5 035 039 9 5 035 039 9 5 035 039 9 5 035 031 9 5 035 031 9 5 035 012 9 5 035 013 9 5 035 013 9 5 035 013 9 5 035 023 9 5 035 025 035 025 025 025 02	 55-1136 AMD (PAOL) 55-1136 (XUKU RAD) 55-1206 (XUKU RAD) 	0.13810 0.13818 0.13818 0.01382 0.01382 0.04037 0.2673 0.2673 0.2673 0.2753 0.2753 0.2753 0.2753 0.2753 0.2754 0.2865 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1866 0.1784 0.1786 0.1866 0.1784 0.1786 0.18866 0.18866 0.18866 0.18866 0.18866 0.18866 0.18866 0.18866 0.18866 0	~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~ ~	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	1000 1000 1000 1000 1000 1000 1000 100	3580 4950 4950 4950 4950 4950 4950 4950 495		
95.005.001 95.005.009 95.005.009 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.001 95.005.002 95.002.002 95.002.002 95.002.002 95	25 - 21201 KUKUR POLICE 56 - 2201 KUKUR POLICE 56 - 2201 KUKUR POLICE 56 - 1201 KUKUR POLICE 50 - 1155 MILO ROAD 56 - 1155 MILO ROAD 56 - 1156 MILO ROAD 56 - 1158 MI	0.1818 0.1818 0.4198 0.4037 0.2673 0.2673 0.2355 0.2355 0.2355 0.2182 0.2182 0.2182 0.2182 0.2182 0.2182 0.2182 0.21636 0.1566 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.16660 0.166600 0.166600 0.166600 0.166600 0.16660000000000		3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 3 5 500 3 5 500 5 3 5 500 5 3 5 5 5 5 5 5 5 5 5 5 5 5	2000 2000 2000 2000 2000 2000 2000 200	900 900 900 900 900 900 900 900 900 900	9 10 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
e 5 035 039 e 5 035 038 e 5 035 038 e 5 035 033 9 5 035 013 9 5 035 015 9 5 035 015 0 5 05 015 0 5 05 015 0 5 05	6 1307 KUKI ROLD 54 1307 KUKI ROLD 54 1307 KUKI ROLD 55 1307 KUKI ROLD 56 1308 MILO ROLD 57 1308 MILO ROLD 57 1308 MILO ROLD 57 1308 MILO ROLD 56 1308 MILO ROLD 56 1308 MILO ROLD 57 1308 MILO R	0.11818 0.4198 0.4198 0.4198 0.4198 0.2335 0.23566 0.23566 0.23566 0.23566 0.23566 0.23566 0.23566 0.23566 0.23566 0.23566 0.235		3 5 600 3 3 4 800 3 4 800 900 3 5 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 3 600 900 3 600 900 900 5 900 900 900 5 900 900 900 5 900 900 900 5 900 900 900 5 900 900 900 6 900 900 900 1000 900 900 9	1000 1000 1000 1000 1000 1000 1000 100	1000 100 1000 1	9 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	
95.025.028 95.025.028 95.025.011 95.025.012 95.025.012 95.025.010 95.025.010 95.025.010 95.025.010 95.025.016 95.025.016 95.025.018	55 1101 (XXX) R0X01 55 1100 (XXX) R0X01 55 1199 (XXX) R0X01 55 1199 (XXX) R0X01 55 1191 (XXX) R0X01 55 1110 (XXX) R0X01 55 110 (XXX	0.4158 0.4037 0.2673 0.2673 0.2673 0.2673 0.267 0.2182 0.1182 0.1187 0.1187 0.1637 0.1637 0.1637 0.1637 0.1637 0.1636 0.1656 0.1533 0.1560		1 4 4 800 3 3 500 500 500 3 3 500 500 500 500 3 3 500 3 500	1000 1000 1000 1000 1000 1000 1000 100	6660 4950 4950 4950 4950 4950 4950 4950 495	No No<	
9 5 035 01/1 9 5 035 01/1 9 5 035 01/1 9 5 035 01/2 9 5 035 01/2 9 5 035 01/2 9 5 035 01/2 9 5 035 01/2 9 5 035 01/2 9 5 035 01/6 9 5 00/6 9 5	CLAND RUMO RUMO RUMO RUMO RUMO RUMO RUMO RUMO	0.4037 0.2673 0.2673 0.23355 0.2310 0.2017 0.2182 0.116450 0.116450 0.116450 0.1164500000000000000000000000000000000000		4 4 800 3 3 610 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 600 9 3 600 9 3 600 9 3 600 9 3 600 9 3 600 9 5 1000 9 600 9 1000	2000 2000 2000 2000 2000 2000 2000 200	4956 4959 4959 4959 4959 4956 4956 4956	70 70 51 51 52 52 53 53 54 53 55 54 56 53 57 54 58 54 59 54 50 55 56 54 57 54 56 54 57 54 56 54 57 54 56 54 57 54 56 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 57 54 54 <	
9 5 035 013 9 5 035 013 9 5 035 013 9 5 035 011 9 9 5 035 011 9 9 5 035 011 9 9 5 035 011 9 9 5 035 011 9 9 5 035 015 9 9 5 035 015 9 9 5 035 017 9 9 5 035 017 9 9 5 035 017 9 9 5 035 017 9 9 5 035 017 9 9 5 035 012 9 9 5 035 021 9 9 5 035 022 9 9 9 5 035 022 9 9 9 035 022 9 9 5 035 022 9 9 5 035 022 9 9 5 035 022 9 9 5 035 022 9 9 5 035 022 9 9 5 035 022 9 9 5 035 022 9 9 9 5 035 022 9 9 9 5 035 022 9 9 9 5 035 022 9 9 9 5 035 022 9 9 9 5 035 022 9 9 9 5 035 022 9 9 9 5 035 022 9 9 035 022 9 9 035 022 9 9 035 022 9 035 025 025 025 025 025 025 025 025 025 02	055 1136 MLO ROMO OTIMA POLI 280 0136 MLO ROMO OTIMA POLI 280 055 1120 MLO ROMO OTIMA POLI 280 0100 ROMO OTIMA POLI 280	0.2673 0.2335 0.2335 0.2187 0.2187 0.1187 0.11876 0.11806 0.11806 0.11806 0.11900 0.11900		3 600 5 600 5 1000	2000 2000 2000 2000 2000 2000 2000 200	4955 4956 4956 4956 4956 4956 4956 4956	9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,9,	
9 5 035 012 9 9 5 035 012 9 9 5 035 012 9 9 5 035 010 9 9 5 035 010 9 9 5 035 010 9 9 5 035 010 9 9 9 5 035 015 9 9 5 035 015 9 9 5 035 015 9 9 5 035 015 9 9 5 035 012 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 5 035 020 9 9 9 9 5 035 020 9 9 9 9 020 9 9 9 9 020 9 9 9 9 020 9 9 9 9	95 1198 MLO ROAD 95 1198 MLO ROAD 95 1188 MLO ROAD 95 1187 MLO ROAD 95 1197 MLO ROAD 95 1192 MLO ROAD 95 1192 MLO ROAD 95 1192 MLO ROAD 95 1193 MLO ROAD 95 1596 MLLA STREET 13 371 CALL	0.2335 0.2210 0.2017 0.2182 0.1282 0.1285 0.1606 0.1606 0.1606 0.1533 0.1500 0.1500 0.1900		600 3 3 600 600 600 600 600 600 600 600	2000 2000 2000 2000 2000 2000 2000 200	4995 4995 4995 4995 4995 4995 4995 4995	97 97 97 97 97 97 97 97 97 97	
9 5 035 011 9 5 035 011 9 5 035 011 9 5 035 009 9 5 035 009 9 5 035 009 9 5 035 012 9 5 035 012 9 5 035 012 9 5 035 012 9 5 035 013 013 0000000000000000000000000	CLAND RUND RUND RUND RUND RUND RUND RUND RU	0.2210 0.2017 0.2182 0.1744 0.1835 0.1635 0.1605 0.1605 0.1606 0.1533 0.1500 0.1900		3 3 600 2 3 3 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 5 5 600 5 5 5 5 5 5 5 5 5	2000 2000 2000 2000 2000 2000 2000 200	495 495 495 495 495 495 495 495 495 495	9.9 9.9 9.9 9.9 9.9 9.9 9.9 9.9	
9 5.035.010 9 5.035.010 9 5.035.015 9 5.035.015 9 5.035.015 9 5.035.018 9 5.035.018 9 5.035.021 9 5.035.022 9 5.035.025 9 5.055.025 9 5.055.055.055.055.055.055.055.055.055.0	55.1188 MILO RAND 55.1202 MILO RAND 55.1202 MILO RAND 55.1202 MILO RAND 55.1218 MILO RAND 55.1288 MILO RAND 55.1288 MILO RAND 55.528 MILA STREE 55.528 MILA STREE 55.558 MILA	0.2017 0.2182 0.1744 0.1616 0.1637 0.1606 0.1606 0.1533 0.1606 0.1896 0.16900 0.16600		3 3 600 2 2 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 3 3 600 5 1000 5 1000 5 1000	2000 2000 2000 2000 2000 2000 2000 200	495 330 495 495 4955 4955 4955 4955 4955 4955	9 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	
95-035-009 95-035-009 95-035-016 95-035-016 95-035-019 95-035-019 95-035-019 95-035-020 95-0200000000000000000	55.1128 MILO ROAD 55.128 MILO ROAD 55.1199 MILO ROAD 55.1198 MILO ROAD 55.1188 MILO ROAD 55.1188 MILO ROAD 55.1184 MILO ROAD 55.1158 KUKIN ROAD	0.2182 0.1744 0.1826 0.1637 0.1616 0.1616 0.1606 0.1533 0.1533 0.1533 0.1533		2 2 400 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 3 3 5 600 5 5 1000 1000 5 5 1000 1000	1000 1000 1000 1000 1000 1000 1000 100	830 4959 4959 4956 4956 4956 4956 4956 4956	2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2 2	
9-502-015 9-502-017 9-502-017 9-502-018 9-502-018 9-502-018 9-502-020 9-502-022 9-502-022 9-502-023 9-502-020 9-502-020 9-502-020 9-502-020 9-502-020 9-502-0200 9-500	5-5-13/2 MILO ROAD 5-13/2 MILO ROAD 55-13/2 MILO ROAD 55-13/8 MILO ROAD 55-13/8 MILO ROAD 55-55/8 MILIA STREET 55-55/8 MILIA STREET 55-55/8 MILIA STREET	0.1/44 0.1826 0.1637 0.1616 0.1606 0.1533 0.1533 0.1596 0.1900		600 000 000 000 000 000 000 000 000 000	1000 1000 1000 1000 1000 1000 1000 100	495 495 495 495 495 495 495 495 495 495		
9-5-03-01b 9-5-03-01b 9-5-035-018 9-5-035-019 9-5-035-019 9-5-035-021 9-5-035-021 9-5-035-028 9-5-035-028 9-5-035-028 9-5-035-028 9-5-035-028 9-5-035-028 9-5-035-028 9-5-035-028 9-5-035-028	25-1134 MLO ROAD 25-1132 MLO ROAD 25-1188 MLO ROAD 55-1186 MLO ROAD 55-1186 MLO ROAD 55-5598 MELA STREET 55-5588 MELA STREET 55-5588 MELA STREET 55-1158 KUKUI ROAD	0.1826 0.1637 0.1616 0.1616 0.1826 0.1826 0.1826 0.1866		600 2 2 2 2 6 0 0 0 0 0 0 0 0 0 0 0 0 0	1000 1000 1000 1000 1000 1000 1000 100	495 495 495 495 495 495 495 495 495 495		
9:5-045-041 9:5-045-049 9:5-045-049 9:5-045-040 9:5-045-042 9:5-045-043 9:5-045-043 9:5-045-043 9:5-045-043 9:5-045-045 9:5-045-045 9:5-045-045 9:5-045-045 9:5-045-045	95-11192 MILO ROMOU 95-1186 MILO ROMO 95-1186 MILO ROMO 95-5596 MILLA STREET 95-5596 MILLA STREET 95-5596 MILLA STREET 95-1185 KIRKUI ROAD	90910 96810 96810 91910 91910		600 600 600 600 600 600 600 600	1000 1000 1000 1000 1000 1000 1000 100	495 495 495 495 495 495 495 495	2 2 2 2 2 3 8 2 2 2 2 2 2 2 2 2 2 2 2 2	
95-025-019 95-025-020 95-025-020 95-025-022 95-025-023 95-025-028 95-025-028 95-025-028 95-025-028 95-025-028 95-025-028	95-1186 MILO ROAD 95-1184 MILO ROAD 95-5598 MILIA STREET 95-5596 MELIA STREET 95-1185 KUKUII ROAD	90910 96810 90910	-	3 600 3 600 3 600 3 600 3 600 5 1000	1000 1000 1000 1000 1000 1000	495 495 495 495 495 495 495	23 88 21 21 28 20 20 20 20 20 20 20 20 20 20 20 20 20	0000000
95-025-020 9-5-025-020 9-5-025-022 9-5-025-023 9-5-025-024 9-5-025-024 9-5-025-025 9-5-025-025 9-5-025-025 9-5-025-025	95-1184 MILO ROAD 95-5598 MELIA STREET 95-5596 MELIA STREET 95-1185 KUKUI ROAD	0061.0 01666		3 600 3 600 3 600 5 1000	1000 1000 1000 1000 1250	495 495 495 495 825	5 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8 8	00000
95,025,021 95,025,022 95,025,023 95,025,024 95,025,025 95,025,025 95,025,025 95,025,025 95,025,025	95-5596 MELIA STREET 95-5596 MELIA STREET 95-1185 KUKUI ROAD	0061.0 0021.0		2 600 3 600 3 600 5 1000	1000 1000 1000 1250	495 3300 495 495 495 825 825	5 8 2 2 2 8 2	00000
9.5.025.022 9.5.025.023 9.5.025.024 9.5.025.025 9.5.025.025 9.5.025.027 9.5.025.027	95-5596 MELIA STREET 95-1185 KUKUI ROAD	0.1900		2 400 3 600 5 1000	1000 1000 1000 1250	330 495 495 495 895 825	855565	0000
9.5-025-023 9.5-025-024 9.5-025-025 9.5-025-026 9.5-025-027 9.5-025-027 0.5-025-027	95-1185 KUKUI ROAD	0.1666		3 600 3 600 5 1000 600	1000 1000 1250	495 495 825 825	5 6 6 8 8 8 6	0
9-5-025-024 9-5-025-025 9-5-025-026 9-5-025-027 9-5-025-027		000710		3 600 3 5 1000 5 1000	1000 1250 1250	495 495 825	5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5 5	0
9-5-025-025 9-5-025-026 9-5-025-027 9-5-025-027	95-1187 KUKUI ROAD	0.1590		3 600 5 1000	1250	825	5 27 57 57	0
9-5-025-026 9-5-025-027 0-5-025-027	95-1191 KUKUI ROAD	0,1569		3 1000	1250	825	5 57	
9-5-025-027	95-1193 KUKUI ROAD	0,1751	-	000	000.	JOK	21	0 Note: 1250 gel Septic Tank
0.6.036.000	95-1206 MILO ROAD	0,1765	-	200	0.0.0.8	fet		0
0.00000000	95-5599 MELIA STREET	0.2129	-	3 600	1000	495	25	0
100-670-6-6	90-5595 MELIA SIREEI	71170		Mb 7	0001	350	30	0
000-570-5-6	92-11/3 KUKUI NUAU	51070	-	1 200	0001	102	13	0 0
C00-C70-C-6	05-11/1 NUNUI INNUI POAD	901510	-	2 000	0007	Soft.		0.0
100-020-0-6	95.1163 KIIKI BOAD	LUSED	-	2 600	000,	305	15	
02000000	95-1213 KHKHH BOAD	0.1721	-	0	000.	- Contraction of the second se		figuration of
0-5-076-079	95-124D OHAI ROAD	01722		0	000.			0 no structure
9-5-026-078	95-1238 OHAI ROAD	0.1722	-	3 600	0007	495	LS IS	0
9-5-026-077	OHAI ROAD	0.1722		0	1000	0		0 no structure
9-5-026-076	95-1234 OHAI ROAD	0.1722	-	3 600	1000	495	5 57	0
9-5-026-075	95-1232 OHAI ROAD	0,1722	1	2 400	0007	330	38	0
9-5-026-074	95-1228 OHAI ROAD	0.1722	1	3 600	1000	495	5 57	0
9-5-026-073	95-1226 OHAI ROAD	0.1722	F	3 600	1000	495	57	0
9-5-026-072	95-1224 OHAI ROAD	0.1722	T	3 600	1000	495	5 57	0
9-5-026-071	95-1222 OHAI ROAD	0.1722	-	3 600	1000	495	5 57	0
9-5-026-070	95-1220 OHAI ROAD	0.1722	T	3 600	1000	495	51	0
9-5-026-069	95-1216 OHAI ROAD	0.1722	1	3 600	1000	495	5 51	0
9-5-026-068	95-1214 OHAI ROAD	0.1722	1	2 400	1000	330	38	0
9-5-026-067	95-1212 OHAI ROAD	0.1722	1	3 600	1000	495	21 21	0
9-5-026-066	CHAI ROAD	0.1835	0	0 0	1000	0		0
9-5-026-065	95-1208 OHAI ROAD	0.2703	-	3 600	1000	495	57	0
9-5-026-064	95-1206 OHAI ROAD	0.2399	1	3 600	1000	495	5 57	0
9-5-026-063	95-1204 OHAI ROAD	0.2077	-	3 600	0007	495	57	0
9-5-026-062	95-1202 OHAI ROAD	0.2204	1	3 600	1000	495	57	0
9-5-026-061	95-1198 OHAI ROAD	0.2146	1	3 600	1000	495	5 57	0
9-5-026-060	95-1196 OHAI ROAD	0.2066	F	3 600	1000	495	57	0
9-5-026-059	95-5548 LOKELANI STREET	0.2046	F	3 600	2000	495	57	0
9-5-024-075		0.1417	0	0 0	2000	0	0	0
9-5-024-074	95-5547 LOKELANI STREET	0.2046	I	3 600	1000	495	570	0
9-5-024-073	95-1184 OHM ROAD	0.2066	-	3 600	1000	465	5 57	0
9-5-024-072	95-1182 OHM ROAD	0.2066	-	3 600	1000	495	5 51	0
9-5-024-071	95-1178 OHAI ROAD	0.2111	-	3 600	1000	495	5 51	0
9-5-024-070	95-1174 OHAI ROAD	0.2301	-	3 600	0007	495	2 21	0
9-5-024-069	OHAI ROAD	0.1286	0	0	000			0
9-5-024-068	95-1106 UHAI KUAU	15570	-	2 000	000	464	10	
9-5-026-031	95-5554 NAHELE SIKEET	0.1421	-	2 400	10001	330	200	0.0
9-5-020-032	NAHELE SIREE!	0.1550	-	3 000	NOOT	CIC+	0	D
9-5-026-033	95-5558 NAHELE SIREET	0.1336	-	3 600	1000	495	20	D

9-5-026-034	95-5562 NAHELE STREET	0.1703	1	3 600	1000	495	570	
9-5-026-035	95-5564 NAHELE STREET	0.1703	=	3 600	1000	495	570	
9-5-026-036	95-5568 NAHELE STREET	0.1871	1	6 1200	1500	066	1140 with DOH	Variance
15-026-037	95-5570 NAHELE STREET	0.1336	1	3 600	1000	495	570	
-5-026-038	95-5572 NAHELE STREET	0.1336	FT	3 600	1000	495	570	
-5-026-039	95-5576 NAHELE STREET	0.1336	1	3 600	1000	495	570	
-5-026-040	95-5578 NAHELE STREET	0.1425	1	3 600	1000	495	570	
-5-026-041	95-5580 NAHELE STREET	0.1247		3 600	1000	495	570	
20120-042	95-5562 NAHELE SIKEET	01226	-	000	1000	495	0/0	
5006.048	95-1206 KHKUI BUAD	01361		000	1000	495	570	
5-026-045	95-1241 OHM ROAD	0.1494		2 400	1000	330	380	
5-026-046	95-1239 OHM ROAD	0.1603	-	3 600	1000	495	570	
-5-026-047	95-1237 OHAI ROAD	0.1603	1	3 600	1000	495	570	
-5-026-048	95-1235 OHAI ROAD	0.1336	1	3 600	1000	495	570	
5-026.049	95-1231 OHAI ROAD	0.1336	-	3 600	1000	495	570	
-5-026-050	95-1227 OHAI ROAD	0.1336	-	000	1000	495	5/0	
150 970 5	05-1722 CHAN BOAR	01236		000	1000	495	0/6	
5-026.053	95-1221 OHAI ROAD	91336	-	009 8	1000	507	570	
-5-026-054	95-1219 OHAI ROAD	0.1336		3 600	1000	495	570	
5-026-055	95-1217 OHAI ROAD	0.1336	1	3 600	1000	495	570	
5-026-056	95-1215 OHM ROAD	0.2316	1	3 600	1000	495	570	
5-026-058	95-12D7 OHAI ROAD	0.1990	-	3 600	1000	495	570	
100 920 5	95-5554 LORELANI STREET	24410		000 0	1000	495	0/6	
5-026-003	95-5558 LOKELANI STREET	0.1336	-	3 600	1000	495	570	
5-026-004	95-5560 LOKELANI STREET	0.1336	1	3 600	1000	495	570	
5-026-005	95-5552 LOKELANI STREET	0.1336	-	3 600	1000	495	570	
5-026-006	95-5566 LOKELANI STREET	0.1336	-	3 600	1000	495	5/0	
2-026-008	95-5520 LOKELANI STREET	91210	-	000	1000	405	0/0	
5-026-009	95-5574 LOKELANI STREET	0.2120	-	3 600	1000	495	570	
5-026-010	95-5576 LOKELANI STREET	0.1386	1	3 600	1000	495	570	
5-026-011	95-5580 LOKELANI STREET	0.1336		3 600	1000	495	570	
5-026-012	95-5582 LOKELANI STREET OK 5524 LOKELANI STREET	0.1336		8 600	1000	495	570	
5-026-014	95-5586 LOKELANI STREET	01336		3 600	1000	495	570	
5-026-015	95-5588 LOKELANI STREET	0.1316	1	3 600	1000	495	570	
5-026-016	95-1198 KUKUI ROAD	0.1316	1	2 400	1000	330	380	
5-026-017	95-5583 NAHELE STREET	0.1336	-	3 600	1000	495	570	
5-026-018	95-5581 NAHELE STREET	0.1336	1	7 1400	1750	1155	1330 with DOH	Variance
5-026-019	95-5579 NAHELE STREET	0.1336	1	3 600	1000	495	570	
5.026.020	95 5577 NAHELE STREET	0.1269		000 m	1000	495	570	
5-026-022	95-5573 NAHELE STREET	0.1336	-	3 600	1000	495	570	
5-026-023	95-5569 NAHELE STREET	0.1336	1	3 600	1000	495	570	
5-026-024	95-5557 NAHELE STREET	0.1336	I	3 600	1000	495	570	
50.020-5	95-5565 NAHELE SIREET	0.1356	-	2000	1000	495	0/6	
5-026-027	95-559 NAHELE STREET	0.1336	-	S 1000	1250	825	950 Note: 1250	0 gal Septic Tank
5-026-028	95-5557 NAHELE STREET	0.1336	1	3 600	1000	495	570	
5-026-029	95-5555 NAHELE STREET	0.1336	-	3 600	1000	495	570	
5-026-030	95-5551 NAHELE SIRLET OC.5552 LOVELANI STREET	76310		8 000	1000	495	570	
5 024 053	95 5555 LOKELANI STREET	0.1336	-	009	1000	495	570	
5-024-054	95-5557 LOKELANI STREET	0.2530	1	3 600	1000	495	570	
5-024-056	95-5561 LOKELANI STREET	0.1336	1	3 600	1000	495	570	
5-024-057	95-5555 LOKELANI STREET	0.1336	-	009 6	1000	495	570	
5-024-059	95-5569 LOKELANI STREET	01336		009 8	1250	564	570	
5-024-060	95-5571 LOKELANI STREET	0.1336	1	3 600	1000	495	570	
- 614 1164	APP PERSON A AND A AND CONCEPT					-	Note: 1500	0 gel Septic Tank
2-024-062	95-5577 LOKELANI STREET	01336		007 6	1000	201	570 WELLING	variance
5-024-063	95-5579 LONELANI STREET	0.1336	-	009	1000	495	570	
5-024-064	95-5581 LOKELANI STREET	0.1336	H	3 600	1000	495	570	
S-024.065	95-5583 LOKELANI STREET	0.1336	1	3 600	1000	495	570	
5-024-066	95-5585 LOKELANI STREET	0.1336	-	3 600	1000	495	570	

sport
ď
sering
Ē
/ Eng
Preliminary
ystem
Ņ
Wastewater
a
Ъ
.≥
g
Ξ
Nā'ālehu

Appendix B - Cost Calculations

ß

100-120-0-1	URUN IUNUN 0011-06	6/ST.0	-	n	200	Inn	CGt	A/C
-5-024-036	95-1184 KUKUI ROAD	0.1379	-	æ	600	1000	495	570
-5-024-037	95-5588 KILIKA STREET	0.1336	1	m	600	1000	495	570
9-5-024-038	95-5586 KUKA STREET	0.1336	-	m	009	1000	495	570
0-5-024-039	95-5584 KILIKA STREET	0.1336	1	-0	1000	1250	825	950 Note: 1250 gal Septic Tank
0-5-024-040	95-5582 KUIKA STREET	0.1336		8	600	1000	495	570
-5-024-041	95-5578 KILIKA STREET	0.1247	-	3	600	1000	495	570
-5-024-042	95-5576 KILIKA STREET	0.1425	1	3	600	1000	495	570
-5-024-043	95-5574 KILIKA STREET	0.1336	1	3	600	1000	495	570
-5-024-044	95-5572 KILIKA STREET	01336	T	æ	600	1000	495	570
9-5-024-045	95-5570 KIUKA STREET	01336	1	m	009	1000	495	570
-5-024-046	95-5568 KIUKA STREET	0.1336	1	m	600	1000	495	570
-5-024-047	95-5564 KILIKA STREET	0.1336	1	e	600	1000	495	570
-5-024-048	95-5562 KILIKA STREET	0.1305	1	e	600	1000	495	570
-5-024-049	95-5560 KILIKA STREET	0.1336	1	m	600	1000	2612	570
NAM AND	BK SSS9 VILIVA STORET	35610	-	4	100.0	1500	CORT	Note: 1500 gal Septic Tank
NOT A PLAN	OF FEEA VILLAA STREET	2102.0		-	area a	-	270	STORES AND THE OTTO
100-100 3	22-2224 NUMA SINELI	0,9650	-	t /	000	1000	AAC	100
STO-POP-C-	95.5500 ODI IKEA STREET	01778	-		and up	1000	495	570
-5-024-016	95-5586 OPUKEA STREET	01768	-	m	909	1000	495	570
-5-024-017	95-5582 OPUKEA STREET	01688	-	m	600	1000	495	570
5-024-018	95-5578 OPUKEA STREET	01800	-	m	600	1000	495	570
5-024-019	95-5574 OPUKEA STREET	01725	-	m	600	1000	495	570
-5-024-020	95-5570 OPUKEA STREET	0.1795	1	m	600	1000	495	570
-5-024-021	95-5564 OPUKEA STREET	0.1779	-	m	600	1000	495	570
-5-024-022	95-5560 OPUKEA STREET	0.1771	1	m	600	1000	495	570
5 024 023	95-5558 OPUKEA STREET	0,1605	-1	m	600	1000	495	570
5 024 024	95-5556 OPUKEA STREET	0,1433	1	m	600	1000	495	570
5 024 025	95-5553 KILIKA STREET	0.1958	-	m	600	1000	495	570
5-024-026	95-5555 KILIKA STREET	0.1228	-	m	900	1000	495	570
5-024-027	95-5559 GUKA SHEET	1/8/0	-	-	809	1000	495	5/0
070-670-0	22-3303 NUMA STREET		+	-	200	0001	100	0/0
5-024-020	92-5500 KUKA SIREI	10710	-	2 1	000	1000	264	5/0
000-620-0-20	132010 MUNA STORET	07770	-	2	000	1000	1995	2/0
COV VOID	05.5572 WHAA STOCT	01530	•	1	100	0001	58V	0.0
5-024-033	138LE VILLA STREET	01633	-		009	1000	204	570
-5-024-034	95-5581 GUKA STREET	01507	-	4	800	1000	660	760
-5-024-035	95-5583 KILIKA STREET	0.1570	-	m	600	1000	495	570
-5-024-001	95-1163 OHAI ROAD	03013	-	m	600	1000	495	570
-5-024-002	95-5558 MAMALAHOA HIGHW	1.0000	-	2	400	1000	330	380
-5-024-003	95-1167 OHAI ROAD	0,2502	-	3	009	1000	495	570
-5-024-004	95-5555 OPUKEA STREET	0.2503	1	8	600	1000	495	570
-5-024-005	95-5559 OPUKEA STREET	0,2655		e	009	1000	495	570
-5-024-006	95-5561 OPUKEA STREET	0.2842	-	m	600	1000	495	570
-5-C24-007	95-5572 MAMALAHOA HIGHW	19440		-	0	-	0	0
-5-024-008	59-553 KAALA ROAD	06650	-	m	600	1000	495	570
-5-024-009	95-5575 OPUKEA STREET	0.2916	1	m	600	1000	495	570
-5-024-010	95-5579 OPUKEA STREET	0.3293	1	m	600	1000	495	570
-5-024-011	OPUKEO STREET	0.2375	0	0	0	1	0	0 no structure
-5-024-012	95-1168 KUKUI ROAD	02040	1	m.	600	1000	495	570
-5-024-013	95-1172 KUKUI ROAD	03329	-	2	400	1000	330	380
-5-024-076	95-5586 MAMALAHOA HIGHW	04231	-	4	800	1000	660	760
-5-024-077	95-5582 MAMALAHOA HIGHW	03950	-	mi	009	1000	495	570
9-5-024-078	95-5580 MAMALAHOA HIGHW	0.3217	-	-	900	10001	495	5/0

A-4

IWS Management Model 2A County In-House Maintenance Cost $^{\scriptscriptstyle 1}$

Year	Tasks		County O&N	l Staff Cost
0	IWS Installation		Capital Cost/I	Household
0	Pumping & Hauling equipment	\$250,000.00	Ŷ	1,470.59
0	Personnel Training	\$ 50,000.00	Ş	294.12
	Annual Inspection by & Trouble Calls by County staff – two			
-	IWS Operators / Plumbers		Ş	822.00
	Annual Inspection by & Trouble Calls by County staff - two			
2	IWS Operators / Plumbers		Ŷ	822.00
e	Septic sludge pumping & disposal by County staff		Ŷ	1,250.00
	Annual Inspection by & Trouble Calls by County staff - two			
4	IWS Operators / Plumbers		Ŷ	822.00
	Annual Inspection by & Trouble Calls by County staff - two			
2	IWS Operators / Plumbers		Ŷ	822.00
9	Septic sludge pumping & disposal by County staff		Ş	1,250.00
	Annual Inspection by & Trouble Calls by County staff - two			
7	IWS Operators / Plumbers		Ş	822.00
	Annual Inspection by & Trouble Calls by County staff - two			
8	IWS Operators / Plumbers		Ŷ	822.00
6	Septic sludge pumping & disposal by County staff		Ŷ	1,250.00
	Annual Inspection by & Trouble Calls by County staff - two			
10	IWS Operators / Plumbers		Ş	822.00
	Annual Inspection by & Trouble Calls by County staff - two			
11	IWS Operators / Plumbers		Ŷ	822.00
12	Septic sludge pumping & disposal by County staff		Ş	1,250.00
	Annual Inspection by & Trouble Calls by County staff - two			
13	IWS Operators / Plumbers		Ş	822.00
	Annual Inspection by & Trouble Calls by County staff - two			
14	IWS Operators / Plumbers		Ŷ	822.00
15	Septic sludge pumping & disposal by County staff		Ş	1,250.00
	Annual Inspection by & Trouble Calls by County staff - two			
16	IWS Operators / Plumbers		Ş	822.00
	Annual Inspection by & Trouble Calls by County staff - two			
17	IWS Operators / Plumbers		Ş	822.00
18	Septic sludge pumping & disposal by County staff		Ş	1,250.00
	Annual Inspection by & Trouble Calls by County staff - two			
19	IWS Operators / Plumbers		Ŷ	822.00
20	Absorption bed replacement		Ş	30,000.00
20	Pumping & Hauling equipment Replacement	\$250,000.00	Ş	1,470.59
	Annual Inspection by & Trouble Calls by County staff - two			
21	IWS Operators / Plumbers		Ş	822.00
	Average Annual Maintenance Cost		Ş	956.44

1 Cost of services are estimated from 2023 USD rates and would increase due to inflation.

IWS Management Model 2B Outsource Maintenance Cost

ш

m

				County	Admin
Year	Tasks	Outsource O&M (Cost	Staff	
0	IWS Installation	Capital Cost			
1	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
2	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ş	572.00
£	Septic sludge pumping & disposal	\$ 1,25	00.00	Ş	572.00
4	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ş	572.00
ß	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ş	572.00
9	Septic sludge pumping & disposal	\$ 1,25	00.00	ş	572.00
7	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ş	572.00
8	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ş	572.00
6	Septic sludge pumping & disposal	\$ 1,25	00.06	Ŷ	572.00
10	IWS annual inspection & Trouble Calls	\$ 55	00.06	Ş	572.00
11	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
12	Septic sludge pumping & disposal	\$ 1,25	00.00	Ŷ	572.00
13	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
14	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
15	Septic sludge pumping & disposal	\$ 1,25	00.00	Ŷ	572.00
16	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
17	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
18	Septic sludge pumping & disposal	\$ 1,25	50.00	Ŷ	572.00
19	IWS annual inspection & Trouble Calls	\$ 55	00.00	Ŷ	572.00
20	Absorption bed replacement	\$ 30,00	00.00	Ş	572.00
21	IWS annual inspection (repeat as Year 1)	\$ 55	00.03	Ŷ	572.00
	Average Annual Maintenance Cost	\$ 78	33.33	ŝ	572.00
County W	WD Admin Personnel Cost for Record Keeping and				
administe	ring	Based on \$10000)/175 =	= \$572	
Trouble c	alls, emergency repairs per IWS per year	\$ 50	0.00		

IWS Management Model 3A Operating Permit User O&M Cost

	unty Admin Cost	Capital Cost	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	572.00	300.00	30,000.00	572.00	572.00	- Å700 - Å700
	CO		Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	ŝ	0000
& M Cost to	ieowner/User	0	500.00	500.00	1,200.00	500.00	500.00	1,200.00	500.00	500.00	1,200.00	500.00	500.00	1,200.00	500.00	500.00	1,200.00	500.00	500.00	1,200.00	500.00	,	500.00	733.33	
õ	Hom		Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ŷ	Ş	Ŷ	
	Tasks	IWS Installation	IWS annual inspection & Trouble calls	IWS annual inspection	Septic sludge pumping & disposal	IWS annual inspection	IWS annual inspection	Septic sludge pumping & disposal	IWS annual inspection	IWS annual inspection	Septic sludge pumping & disposal	IWS annual inspection	IWS annual inspection	Septic sludge pumping & disposal	IWS annual inspection	IWS annual inspection	Septic sludge pumping & disposal	IWS annual inspection	IWS annual inspection	Septic sludge pumping & disposal	IWS annual inspection	Absorption bed replacement	IWS annual inspection (repeat as Year 1)	Annual Average Cost	
	Year	0	1	2	ε	4	ß	9	7	∞	6	10	11	12	13	14	15	16	17	18	19	20	21		

O&M Cost to Homeowner / User includes trouble calls County admin staff to provide recording keeping, regulation and enforcing

\$300 + \$200 = \$500 Based on \$100000/175

IWS Management Model 3B County Voucher O&M Cost

ш

ш

Year	Tasks	Trouble call Cost to Homeowner/User	County Voucher Cost (present value)
0	IWS Installation	0	Capital Cost
1	IWS annual inspection & Trouble calls	\$ 600.00	\$ 872.00
2	IWS annual inspection	\$ 600.00	\$ 872.00
æ	Septic sludge pumping & disposal	\$ 600.00	\$ 1,572.00
4	IWS annual inspection	\$ 600.00	\$ 872.00
ß	IWS annual inspection	\$ 600.00	\$ 872.00
9	Septic sludge pumping & disposal	\$ 600.00	\$ 1,572.00
7	IWS annual inspection	\$ 600.00	\$ 872.00
8	IWS annual inspection	\$ 600.00	\$ 872.00
6	Septic sludge pumping & disposal	\$ 600.00	\$ 1,572.00
10	IWS annual inspection	\$ 600.00	\$ 872.00
11	IWS annual inspection	\$ 600.00	\$ 872.00
12	Septic sludge pumping & disposal	\$ 600.00	\$ 1,572.00
13	IWS annual inspection	\$ 600.00	\$ 872.00
14	IWS annual inspection	\$ 600.00	\$ 872.00
15	Septic sludge pumping & disposal	\$ 600.00	\$ 1,572.00
16	IWS annual inspection	\$ 600.00	\$ 872.00
17	IWS annual inspection	\$ 600.00	\$ 872.00
18	Septic sludge pumping & disposal	\$ 600.00	\$ 1,572.00
19	IWS annual inspection	\$ 600.00	\$ 872.00
20	Absorption bed replacement	\$	\$ 30,000.00
21	IWS annual inspection (repeat as Year 1)	\$ 600.00	\$ 872.00
	Annual Average Cost	\$ 600.00	\$ 1,105.33
County vo	ucher cost: \$572 + \$300 Admin Hoff cost \$100000/17E		
Ddseu UII,			

A-8





Appendix D – USGS Soil Survey

۵



MAP LEGEND

MAP INFORMATION

The soil surveys that comprise your AOI were mapped at 1:24,000.

Warning: Soil Map may not be valid at this scale.

Enhangement of maps beyond the scale of mapping can sause misurderstanding of the detail of mapping and securacy of soil line placement. The maps of not show the small areas of contrasting soils that could have been shown at a more detailed scale.

Please rely on the bar scale on each map sheet for map measurements.

Source of Map. Natural Resources Conservation Service Web Soil Survey URL: Coordinate System: Web Mercator (EPSG:3857)

Maps from the Web Soil Survey are based on the Web Mercator projection, which preserves direction and shape but distorts distorts and asserves direction and shape but distorts affaitors and area A projection high prevense area, such Albers equal area conic projection, should be used if more accurate calculations of distance or area are required.

This product is generated from the USDA-NRCS certified data as of the version date(s) listed below.

Soil Survey Area: Island of Hawaii Area, Hawaii Survey Area Data: Version 15, Aug 30, 2022

Soil map units are labeled (as space allows) for map scales 1:50,000 or larger.

Date(s) aerial images were photographed: Jan 3, 2019—Jun 28, 2022

The orthophoto or other base map on which the soil lines were compiled and digitized probably differs from the background imagery displayed on these maps. As a result, some minor shifting of map unt boundaries may be evident.

Map Unit Legend

Map Unit Symbol	Map Unit Name	Acres In AOI	Percent of AOI
12	Naalehu medial aity day loam 3 to 10 percent slopes	32.4	47.2%
55	Naalehu medial siity day loam 10 to 20 percent slopes	214	31.2%
36	Namichlu medial suit loami 0 to 3 percent slopes	8,24	52'E
otals for Area of Interest		68.8	100.03%




ш









Masa Fujio 98-021 Kameha Ale ph (308/4845	Diiller: Geolabs	Dill Method: Con	Sample Method: 3	Boring Diameter:	κα Βαρ		30%	16%		10% 0% 0
6-091				MF	Test Results					
Page 6 Job Number: 05086 Elevation:	Date:	11/13/06	11/13/06	Checked By:			r, well fractured and	jociace (s'a)	dvater encountered	
VG LOG: N-6 sspool Conversion nd of Hawaii	Drilling	Started:	Finished:	Logged By: JR	rial Description	ste pavement	ALT, moderately vestoria	ulghtly vosioular, 3% plag	ed @ 10 ft bgs, no groun	
ssociates BORI vay #337 Na'atehu Ce Na'atehu, Ista				Water Level : nía	USCS	AS Asphaltic-concre	Diokan Linokan	Grey BASALT, a	Boring terminat	
ijioka & A mehameha High Aleo, Hawaii 484-5366, FAX 43	abs	: Core Barrel	nod: 2.5" Core	ieter: 2.5"	(Fi) digage (Fi) 9947 (Fi)			∞ ⁵	2 2	
Masa Fu 98-021 Ka ph (808)	Driller: Geol	Drill Method	Sample Meti	Boring Diam	каם Весоνеιγ		10% 0% 28% 15%			

	16-091					MF	Test Results			MC = 40 % CBR = 11 % Exp = 1.4 %				
e 7	Number: 050	ation:		13/06	13/06	Checked By								
Pag	dot	Elev	Date:	111	11/			ols			ular	ar clinker		
IG LOG: N-7	snool Conversion	nd of Hawaii, Hawaii	Drilling	Started:	Finished:	Logged By: JR	lal Description	soil), with grass and re		Own sdjacent test pit	roken and highly vesk	SASALT, highly vesicul	noderately vesicular	
BORIN	Na'alehu Ces	Na'alehu, Islar				el : n/a	Mater	Cleyey SILT (top	Brown clayey Sil	colors reddish-b Sample collected from	Black BASALT, I	Blackish-brown	Black BASALT.	
soriates	ay #337	4-0007				Water Lev	USCS	HM	HM	Bulk		E CONTRACTOR		
itoka & As	nehameha Highw	AJ63, Hawau 184-5366, FAX 48	bs	Core Barrel	od: 2.5" Core	eter: 2.5"	Sample Type Depth (Ft)	-		Core 1				
e Ha	021 Kar	h (308)4	: Geola	Aethod:	le Meth	g Diame	RQD			%0		%0		%0
Mae	86	۵	Dillet	DHILE	Samp	Borin	Βεσονείγ			30%		16%		10%



F Appendix F - Typical IWS Layout & Components





TYPICAL IWS LAYOUT W/ SEEPAGE PIT See Schedule A for Required Setback Dimensions and Variance Request



SCHEDULE A

VARIANCE REQUEST (SEE NOTE 1)	Z' < "a" < 5'	1' < "b" < 5'	2' < c < 5'	.S > "P" > 0	2' < "e" < 5'	1 ⁻ < "f" < 5 ⁻	Z' < "g" < 5'	1' < "h" < 5'	6' < "I" < 12'
DESCRIPTIONMIN. PER DOH, TABLE (SEE NOTE 1)	5' MIN	DIST. BTW PROPERTY LINE & SEPTIC TANK / 5' MIN.	5' MIN	DIST. BTW. PROPERTY LINE & ABSORPTION BED / 5' MIN.	DIST. BTW BLDG & SEEPAGE PIT / 5' MIN.	DIST BTW PROPERTY LINE & SEEPAGE PIT/ 9' MIN.	DIST. BTW SEPTIC TANK & ABSORPTION BED / 5' MIN.	DIST. BTW NEIGHBORING ABSORPTION BEDS / 5' MIN.	DIST BTW NEIGHBORING SEEPAGE PITS / 12' MIN
SETBACK TYPE	a		υ	"P.	"G"	ight.	"Ô"		

ABSORPTION BED & SEEPAGE PIT SIZING

E.		0.0				
SEEPAGE PIT SI2 (SEE NOTE 4)	s'r/J X 12' DEEF	6'r/J X 15' DEEF 8'r// X 12' DEEF	<i>3'r/!</i> × 14' DEEF	8'r/! X 17' DEEF	8'r/! X 20' DEEF	
AFFIC RATED SORPTION BED SIZE (E NOTE 3)	24' 18' 16'	30' 24' 18'	40' 28' 24'	48' 36' 28'	52' 40' 32'	
IC RATED TR/ N BED SIZE AB: 3) (SE	9. 12	9' 12 15	9' 15	9, 12	9 15 15	
NON-TRAFF ABSORPTIO (SEE NOTE	10' X 24' 12' X 20' 15' X 16'	10' X 32' 12' X 27' 15' X 22'	10' X 40' 12' X 34' 15' X 27'	10' X 48' 12' X 40' 15' X 32'	10' X 56' 12' X 47' 15' X 38'	
# OF BEDROMS SEPTIC TANK SIZE	3 1000 GAL	4 1000 OR 1250 GAL	5 1250 GAL.	6 1250 GAL. PER DWS (SEE NOTE 2)	7 1250 GAL. PER DWS (SEE NOTE 2)	NOTES:

The request is for Variance from Section 11-62-22 Spacing of Individual Wastewater Systems. Table II in Appendix D and Section 11-62-31.1(1)(D) where states that one IWS cannot serve more than 5 bedrooms. ÷

- For dwellings with more than 5 bedrooms, we request a Variance to base the IWS design on the DWS water consumption record rather than based on number of bedrooms. ¢.
- Absorption Bed for standard perforated pipe with gravel bed installation (non-rafific rated) a percellation rate of 2 min. Amb. is a supmed. For graveless relatation (Infiltrator Chambers) or infiltre rated chambers. T7% reduction is taken for the required area of absorption bed.
- For sizing of seepage pit, a percolation rate of 1 min/inch is assumed because the soil condition is likely to be granular or rocky type at that depth.



15, WIN'

VARIES

.92



A31



Terk Functional by Austin A) Hee Due True, 731/33 130 pm Terk Foremond by Austin A) Hee DNK: 3 5 5 024 : 010 DNK: 3 5 5 024 : 010 DNK: 3 5 5 024 : 010 DNK: 3 10 1000 10000 1000 100	ALTH - WASTEWATER BRANCH [IWS] - SITE EVALUATION / PERCOLATION TEST	DEPARTMENT INDIVIDUAL WASTEWATER SY	OF HEALTH - WASTEWATER BRANCH STEM (IWS) - SITE EVALU/ATION / PERCOLATION TEST
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Arredin Alt (Las	Date / Time: 7/3 1/23 1:30 pm	Test Performed by: Austin Ah Hee
TMK: (3,) 9, -5, -055; 0.9 Eventor: COM Eventor: COM Eventor:	st Performed by: Austrin An rice	Owner: County of Hawai'i	TMK: (3) 9 - 5 - 024 : 011
Test Test Test Test Test beins gate Deption in direction (indexer of link). Eet beins gate Test beins gate Eet beins gate Eet beins gate Eet beins gate Eet beins gate Sult brotht catana. entral Eet beins gate Eet beins gate Eet beins gate Sult brotht catana. entral Eet beins gate Eet beins gate Eet beins gate The brown diff. Eet beins gate Eet beins gate Eet beins gate The brown diff. Eet beins gate Eet beins gate Eet beins gate The brown diff. Eet beins gate Eet beins gate Eet beins gate The brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. The brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. The brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. The brown diff. Eet beins eet beins eet of the brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. The brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. Eet beins eet of the brown diff. The brown diff. Eet beet of the brown diff. Eet beet of the brown di	MK:(3)) 9 - 5 - 025 : 019	Elevation: 670	feet
Image: Intervent parted in the forw option in the for		Depth to Groundwater Table: N/A	feet below grade
Image: Image: Image: Image: Image: Image: Image: (choice grade (choice grade (choice grade (choice grade (choice grade Salt Profit (color extrant of all (choice grade (choice grade (choice grade (choice grade Salt Profit (color extrant of all (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (choice grade (choice grade (choice grade (choice grade (choice grade Image: (cho	East below grade	Depth to Bedrock (if observed): N/A	feet below grade
Tech before gate Destity induction Soil Destity field concreation Soil Destity field colore extrate orbit Soil Destity field concreation Decomposition 0-34 ERCOLATION READINGS Interest in an extrate or exp avery: Soil Destity field concreation Decomposition 0-34 ERCOLATION READINGS Interest in an extrate or exp avery: Soil Destity field concreation of the interval and water dops at least every 10 minutes. Interval Unterval in the interval and water dops at least every 10 minutes. ERCOLATION READINGS Interval interval Data of the soil or exp avery: Soil Destity field concreation of the interval and water dops at least every 10 minutes. Interval inter	inches	Depth to Hole Bottom: 45	finctes feet below grade
minutes Description Description Description Description information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information information inf	feet below grade	Depth, inches below grade	Soil Profile (color, texture, other)
minutes minutes	sou rioue tootor texture, oner proven dirt	0-04	Soft brown dig
Time 12 techs of vater to seep away: Bit mittage minutes Time 12 techs of vater to seep away: Bit mittage minutes Time 12 techs of vater to seep away: Bit mittage minutes There of the interval graph are to a test to a testo a testo a test to a test to a testo a test to a test to a test			
minus intrust intrust intrust intret i		PERCOLATION READINGS:	
intervals and water drops at least every 10 minutes for at least every 10 minutes for 1 mount of tracellation tests in search soils, recorded time intervals and water drops at least every 10 minutes for 1 hours of trags for at least 4 hours. Recorded time intervals and the fort 6 minutes for 1 hours of trags for the first 6 miches to seep away in greater than 30 drops at least every 30 minutes for 1 hours of trags for the first 6 miches to seep away in greater than 30 drops at least every 30 minutes for 1 hours of trags for the first 6 miches to seep away in greater than 30 drops at least every 30 minutes for 1 hours of trags for the first 6 miches to seep away in greater than 30 drops at least every 30 minutes for 1 hours or trang for the minutes for the min	minutes	Time 12 incress of water to seep away: Time 12 inches of water to seep away:	8 minuss.
d fine intervals and water drops at least every 10 minutes for 1 hours. Recorded time intervals and water drops at least every 30 minutes for 1 hours of minutes for 1 hours of a drops at least every 30 minutes for 1 hours of minutes in the minutes in the minutes in the minutes in the minutes of the for the for the hours of the minutes in the minutes	minutes	Check one:	
And the test hole for at least 4 hours. Recorded time intervals and the test hole for at least 4 hours. Recorded time intervals and to the first 6 inches to seep away in greater than 30 thous or until 2 successive. X Precentation tests in no-sumly soils, presonded the test \$9/6 for at least 4 hours. Recorded the test \$9/6 for at least 4 hours. Recorded the test \$9/6 for at least 4 hours. Recorded the test \$9/6 for at least 4 hours or until 2 successive. A hour or the first 6 inches to seep away in greater than 30 thous or until 2 successive. A increment in the first 6 inches to seep away in greater than 30 thous or until 2 successive. A increment in the set of a hours or until 2 successive. A increment in the set of a hours or until 2 successive. A increment in the set of a hours or until 2 successive. A increment in the set of a hours or until 2 successive. A increment in the set of a hour or until 2 successive. A increment in the set of a hour or until 2 successive. A increment in the set of a hour or or until 2 successive. A increment in the set or and or any other fination and precolation test is the set of a hour or or the soil absorption system and the groundwater table or any other fination and precolation test table or any other fination and precolation test table or any other fination and precolation system and the groundwater table or any other fination and precolation system and the groundwater table or any other finating a set of table or any other fination are an a	ne intervals and water drops at least every 10 minutes for at least	Percolation tests in sandy soils, reco	orded time intervals ssiid water drops at least every 10 minutes for at le
$\frac{1}{10000000000000000000000000000000000$	ed the test hole for at least 4 hours. Recorded time intervals and hour of time for the first 6 inches to seep away in greater than 30 ps at lest every 30 minutes for 4 hours or until 2 successive	X Percolation tests in no-sandy soils, 1 water drops at least every 10 minute minutes record time intervals and w drops do not vary by more than 1/1	presoaked the test ŝtaĝi for at least 4 hours. Recorded time intervals a es for 1 hour of ting fig the first 6 inches to seep avay in greater than rater drops at lest every 30 minutes for 4 hours or until 2 successive 6 inch.
Time Interval Drow in Inchs 30 $4 \cdot 1/2"$ $4 - 1/2"$ $\sim 10^{-1}$ minutes/inches 30 $4 - 1/2"$ $a - 1/2"$ $\sim 10^{-1}$ minutes/inches 30 $4 - 1/2"$ $a - 1/2"$ $\sim 10^{-1}$ minutes/inches $a - 1/2"$ $a - 1/2"$ $a - 1/2"$ $\sim 10^{-1}$ minutes/inches $a - 1/2"$ $a - 1/2"$ $a - 1/2"$ $\sim 10^{-1}$ minutes/inches $a - 1/2"$ $a - 1/2"$ $a - 1/2"$ $\sim 10^{-1}$ minutes/inches $a - 1/2"$ $a - 1/2"$ $a - 1/2"$		Time Interval Drop in Inc	ches Time Interval Drop in Inches
 →10 minutes/inches →11 minutes/inch	Time Interval Drop in Inches	30 4-1/2"	
 →10 minutes/inches →10 minutes/inches Percolation Rate (time/final water level drop): ~7 minutes/inches As the engineer responsible for gathering and percolation and percolation test result fact that above site information is accurate and that the site evaluation was conducted in accordanc provisions of Chapter 11-62. "Wastewater Systems" and the groundwater table or a layer table or any other limiting The site evaluation was conducted in accordance with the maximum system and the groundwater table or any other limiting The site evaluation system and the groundwater table or any other limiting Encineer's Signature/Siamp 			
<u>~10</u> minutes/mches As the engineer responsible for gathering and providing site information and percolation test result accolation site information and percolation test results. I attest to the provisions of Chapter 11-62, "Wastewater Systems" and the results were accordance provisions of Chapter 11-62, "Wastewater Systems" and the groundwater table or a suitable soil exist between the bottom of the soil absorption system and the groundwater table or a provisions of Chapter 11-62, "Wastewater Systems" and the groundwater table or a provisions of Chapter 11-62, "Wastewater Systems" and the groundwater table or any other limiting the site exists were accordance provisions of the soil absorption system and the groundwater table or any other limiting Engineer's Signature/Siamp Date Date Date Date Date Date Date Date		Percolation Rate (time/final water level drop): ~7 minutes/inches
Engineer's Signature/Stamp Date	~10 minutes/inches ling site information and percolation test results, I attest to the the site evaluation was conducted in incordance with the and the results were acceptable. I also attest that three feet of orption system and the groundwater table or any other limiting.	As the engineer responsible for gathering an fact that above itin information is accurate provisions of Chapter 11-62, "Wastewater Sys suitable soil exist between the bottom of the layer.	d providing site information and percolation test results. I artest to the aid that the site evaluation was conducted in accordance with the systems' and the results were acceptable. I also attest that three feet of soli absorption system and the groundwater table or any other limitin soil absorption system and the groundwater table or any other limitin
		Engineer's Signature/Stamp	Date
Date	Date		

DEPARTMENT OF HEALTH - WASTEWATE INDIVIDUAL WASTEWATER SYSTEM (IWS) - SITE EVALUAT

Date / Time: 7/31/23 8:00 am

Owner: Karl Stewart Ragan

Elevation: 700

Depth to Groundwater Table: <u>N/A</u> Depth to Bedrock (if observed): <u>N/A</u>

93 PERCOLATION READINGS: Time 12 inches of water to seep away: Time 12 inches of water to seep away:

Soft brown dirt

Depth, inches below grade

0-48

Depth to Hole Bottom: 4 Diameter of Hole: 12

Check one: Percolation tests in sandy soils, recorded time intervals and water dro 1 hour.

Fercolation tests in no-sandy soils, presoaked the test hole for at least voter drops at least east, vory 10 minutes for 1 hour of time for the first 6 minutes record time irtervals and water drops at lest every 30 minutes drops do not vary by more than 1/16 inch. ×

Drop in Inches	
Time Interval	
<u>3-1/4"</u> 3-1/4"	
Time Interval 30	

minutes/inches Percolation Rate (time/final water level drop): ~10 As the engineer responsible for gathering and providing site information and p fact that above site information accurate ascurate and that the fact evaluation was con provisions of Chapter 11-62, "Wastwater Systems" and the results were accept suitable soil exist between the bottom of the soil absorption system and the gro layer.

Engineer's Signature/Stamp

Date / Time: 8/7/23 8:00 pm Test Performed by: Austin Ah Hee	Date/Time: 8.7/23 12:00 pm Test Performed by: Au	istin Ah Hee
Owner: Sophia Rhonda Lorenzo TMK;(3)9-5-026;071	Owner: Baron Yamamoto Trust TMK: (<u>3</u>) 9 -	5 . 024 : 068
Elementary 760 for the former of the former	Elevation: 665	4
Depth to Groundwater Table: N/A feet below grade	Depth to Groundwater Table: N/A for	ct below grade
Depth to Bedrock (if observed): N/A feet below grade	Depth to Bedrock (if observed): N/A fee Diameter of Hole- 12 inc	et below grade ches
Diameter of Hote: <u>1.2</u> metres Depth to Hote Bottom: 4 feet below grade	Depth to Hole Bottom: 4 for	t below grade
Depth, hishes below grade Soil Profile (color, texture, other) 0-48 Hard brown dirt, some rocks	Depth. inches below grade Soil Profile fcolur. 0-48 Seft brown dirt, some i	texture other) small rocks
PERCOLATION READINGS. Time 12 indues of water to seep away: 180 minutes Time 12 indues of water to seep away: 180 minutes	PERCOLATION READINGS: 110 minutes Time 12 inches of water to seep away: 94 minutes	
Check one: Perepádion tests in sandy soils, recorded time intervals and water drops at least every 10 minutes for at least 1 from:	Check one: Percolation tests in sanly soils, recorded time intervals and wate 1 hour.	r drops at least every 10 minutes for at lea
X Persolution tests in no-sandy soils, presoaked the test hole for at least 4 hours. Recorded time intervals and waxer dops at least every 10 minutes for 1 hour of time for the first 6 inches to seep away in greater than 30 mistafica record time intervals and ware drops at least every 30 minutes for 4 hours or until 2 successive drops do not vary by more than 1/16 inch.	X Percolution tests in no-sandy soils, presoaked the test hole for at water dops at least every 10 minutes for 1 hour of time for the 1 minutes record time intervals and water drops at lest every 30 m drops do not vary by more than 1/16 inch.	cleast 4 yours. Recorded time intervals an first 6 inshes to seep away in greater than 3 inutes for 4 hours or until 2 successive
$ \frac{1}{30} \frac{\text{Time Interval}}{2^{"}} \frac{\text{Drop in Inches}}{2^{"}} \frac{\text{Time Interval}}{2^{"}} $	Time IntervalDroo in IachesTime Inte 30 $4.1/4^{\circ\circ}$ $4.1/4^{\circ\circ}$	rval Drop in Inches
Percolation Rate (time/final water level drop): minutes/inches	Percolation Rate (time/final water level drop): minutes/in	ches
As the engineer responsible for gathering and providing site information and percolation test results. I attest to the fact that above site information is accurate and that the site evaluation was conducted in accordance with the provisions of Chapter 11-62, "Wastewater Systems" and the results were acceptable. I also attest that three feet of suitable soil exist between the bottom of the soil absorption system and the groundwater table or any other limiting layer.	As the engineer responsible for gathering and providing site information fact that above site information is accurate and that the site evaluation wa provisions of Chapter 11-62, "Wastewater Systems" and the results were suitable soil exit between the bottom of the soil absorption system and th layer.	and percolation test results, I attest to the s conducted in accordance with the acceptable. I also artest that these feet of he groundwater table or any other limiting
Envincer's Signature Stann	Engineer's Signature/Stamp	Date

DEPARTMENT OF HEALTH - WASTEWATER BRANCH INDIVIDUAL WASTEWATER SYSTEM

FALLING HEAD TEST PROCEDURE

- ÷
- Preparing Percolation Test Hole(s) 1. Dig or bore a hole, four to twelve inches in diameter with vertical walls to the approximate
 - depth of the soil absorption system (bottom of trench or bed). Scratch the side wall and bottom to remove any smeared soil and remove loose material. Place one inch of coarse sand or gravel on bottom.
 - ci m
- Determine Percolation Rate B.
- Place twelve inches of water in hole and determine time to seep away. Record this time on the site evaluation form. -
 - Repeat step B.1. above. Also record this time on the site evaluation form. ci m
- If the time of the second test is less than 10 minutes go to Step C, if not skip to Step D.
- (granular) Soils Sandy (ċ
- Establish a fixed reference point, add water to six inches above gravel and measure water
 - level drops every ten minutes for 1 hour. Use a shorter time interval if first six inclues seeps away in ten minutes or less. Use fill when necessary, do not exceed six inches of water. Record time intervals and water drops on site evaluation form.
 - cimiti si
- Use final water level drop interval to calculate percolation rate. (Step E)
- Ū.
- Other Soils (non-granular, e.g. silt, loarns & clays)
 Maintain at least twelve inches of water in the hole for at least four hours to presoak soil.
 Do not remove water remaining after four hours (Dy clayey soils should be soaked and permitted to 3. Permit soil to swell at reast 12 hours. (Dy clayey soils should be soaked and permitted to swell for longer periods to obtain stabilized percolation mtes).
 Alter swelling, remove loose material on top of gravel.
 - 4. 2
- Use fixed referenced point, adjust water level to six inches above gravel and measure water
- level drop. If the first six inches of water seeps away in less than 30 minutes, measur water level drops every ten-minutes and run for one hour. If the first six inches of water takes longer than 20 minutes to seep away, use 30 minute time è.
 - intervals for four hours or until two successive drops do not vary by more than one-sixteenth L.
- inch (stabilized rate). Refill with water only when necessary, but no adjustment during last three readings except to the limit of the last drop. Do no exceed six inches of water. ŝ
- Use final drop interval to calculate percolation rate and record on site evaluation form: цí

Time Interval Water Level Drop

= Percolation Rate

APPENDIX B:

Natural Resource Assessment

AECOS, Inc.

A natural resources assessment for the Nāʿālehu WWTP, Nāʿālehu, Kaʿu District Island of Hawaiʿi (TMK 9-5-005:001 por.)



AECOS Inc. 45-939 Kamehameha Highway Suite 104 Kāne'ohe, Hawai'i 96744

June 24, 2021

NĂ'ĂLEHU WWTP, KA'U

(sland of Hawai'i (TMK 9-5-005:001 por.) Nā'ālehu WWTP, Nā'ālehu, Ka'u District A natural resources assessment for the

AECOS No. 1613 DRAFT E Guinther, D. Miranda, and R. David June 24, 2021 AECOS Inc.

Kāne'ohe, Hawai'i 96744 Phone: (808) 234-7770 ~ Email: guinther@aecos.com 45-939 Kamehameha Highway Suite 104

Introduction

allow the closure of three large capacity cesspools currently used to dispose of the existing site drainage, beginning from a vegetated diversion channel at a The Hawai'i County Department of Environmental Management, Wastewater Division is proposing to construct a wastewater treatment and disposal system to treat sewage collected in Nā'ālehu, Ka'u District, Island of Hawai'i (the "Project"; Figure 1). Construction of the treatment and disposal system will untreated sewage. The treatment and disposal facility will be located on an Resources (DLNR) land to the southwest of town (makai and downslope; see Figure 2). Site work includes grading of a drainage diversion swale to realign point some 350 ft (100 m) downslope of Māmalahoa Highway to a point downslope and east of the Project site. Road access and a utility corridor would approximately 28-ac (11.3-ha) portion of State Department of Land and Natural follow from off the existing Nā'ālehu Spur Road.

includes a force main pump station (SPS) and a force main connecting to the WWTP The latter was surveyed specifically where the force main would cross A collector system will initially service residences on the north (mauka) side of the main street, the Hawaii Belt Road (Mamalahoa Highway). The Project also a drainage channel. Otherwise, the force main route follows existing roads.

AECOS Inc. was contracted by Wilson Okamoto Corporation to conduct a natural resources survey and assessment of the subject property (TMK 9-5-005:001

AECOS Inc. [Report No, 1613]

Page | 1



por.) and prepare this report of findings.¹ We make reference herein to a previous survey of a nearby location proposed for the WWTP nearly a decade collector system as presently proposed, the latter covered by driving all of the streets to be impacted to establish that no environmental resources of ago (David & Guinther, 2012). As in 2012, our survey included the sewerage conservation interest occur in those areas.

Site Description

The Project site is low sloping ground covered by pasture, in places mostly of cropped grasses (Figure 3), in other places a mixture of grasses and low growing shrubs (Figure 4); along the southern boundary, the vegetation merges

¹ This document is produced for inclusion in an EA for the subject Project and will become part of the public record.

NĀ'ĀLEHU WWTP, KA'U



Natural Resources Survey



Figure 3. Grassland pasture area of cropped Guinea grass.



Figure 4. Pasture area of cropped grass and low shrubs.

AECOS Inc. [Report No, 1613]



AECOS Inc. [Report: 1613—page 4]

Page | 5

Image: With Guine a gras. Image: With Guine a gras.	Vagner & Herbst, 1999) for native and naturalized flowering plants, <i>Hawaif's</i> "errors and Fern Allies (Palmer, 2003) for ferns, and A Tropical Garden Flora Staples & Herbst, 2005) for ornamental plants. More recent name changes for aturalized plant species follow Imada (2019). Terrestrial Vertebrates Survey is a conducted on the morning of errestrial Vertebrates Survey is solved to the morning of the norming the Project areas and a single eight-minute avian noint-count made at each station. The avian counts were conducted in the norming hours between 0730 and 1030 am with the aid of Leica 8 X 42 inoculars and by listening for vocalizations. Weather conditions were ideal, SInc. [ReportNo.1613] Page 16
	ant species were identified as they were encountered. Species names follow <i>mual of the Flowering Plants of Hawaii</i> (Wagner, Herbst, & Sohmer, 1990; agner & Herbst, 1999) for native and naturalized flowering plants, <i>Hawaii's</i> <i>rns and Fern Allies</i> (Palmer, 2003) for ferns, and <i>A Tropical Garden Flora</i> taples & Herbst, 2005) for ornamental plants. More recent name changes for turalized plant species follow Imada (2019).
	e natural resources survey was undertaken by the botany team on February , 2020 and again on November 5, 2020. A survey area map (corresponding to e red outline on Fig. 2) was loaded on a Trimble 6000 Series GNSS unit rimble GeoXH and GeoXT for use during the surveys). The GNSS units corded the progress tracks of the botanists, providing real time feedback on action and adequacy of coverage during a wandering (pedestrian) transect. e November survey encompassed the site for a sewerage pump station east of an Road, the force main cossing of a drainage channel off Melia Street, and additional look along the southwest side of the WWTP site (Figure 2) that iv not have been completely covered by the February visit owing to later justments in Project footprint.
period of time. Negative results cannot confirm the absence of this mammal. At a location. Mammal scientific names follow <i>Mammal species of the world: a</i> <i>taxonomic and geographic reference</i> (Wilson and Reeder, 2005).	ods vtanical Survey
The survey of mammals was limited to visual and auditory detection, coupled with visual observation of scat, tracks, and other animal sign. We did not conduct a survey for Hawaiian hoary bat or <i>'āpe'ape'a</i> (<i>Lasiurus cinereus semotus</i>), the only native land mammal in the Islands. Bats are solitary and potentially widespread throughout Hawai', and detection requires night survey and dedelovment of special detection equipers a low	it of the intersection with Opukea Street. The force main runs west on ukea Street to Melia Street then south on Ka'ai'aiki Rd and Na'alehu Spur ad to the proposed WWTP site. The pipe crosses a drainage channel on Melia eet ("Area No. 2").
with unlimited visibility, light winds, and no rainfall. The avian phylogenetic order and nomenclature used in this report follow the AOU <i>Check-List of North and Middle American Birds</i> 2019 and the Sixtieth first Supplement to the Check-list of North American Birds (Chesser et al., 2019, 2020)	o scrub growth (cover photo). Much of the grassland is dominated by Guinea ass (<i>Megathyrsus maximus</i>), Low shrub growth areas are dominated by tana (<i>Lantana camara</i>). Taller shrub/scrub growth is entirely <i>koa haole</i> <i>ucaema leucocephala</i>). The proposed pump station site ("Area No. 3") is on <i>e mauka</i> side of Māmalahoa Highway in an overgrown pasture off "Ôhai Road
Natural Resources Survey NA'ALEHU WWTP, KA'U	Resources Survey NA'ALEHU WWTP, KA'U

AECOS Inc. [Report No, 1613]

ı WWTP. KA'u N ā', ā'r c'u

NĂ/ALEHU WWTP, KA'U	Natural Resources Survey		Ń	₽,ĂLEHU	WWTP, K/	<u>,,</u>
	Flora					
sists entirely of grass pasture (predominantly <i>aximus</i>) and areas of mostly low scrub growth, are scattered around the site, especially along a sects the area from north to southeast.	Table 1 is a listing of all the sp during the surveys with a tol areas. The vast majority (919 native indigenous plants (4.55 native species are <i>'ilima</i> (<i>Sida</i> (<i>Waltheria indica</i>); all commo called "canoe plant") is yellow common small weed.	ecies of flowering plan tal of 66 taxa identif (6) are naturalized or 1 (7) and one early Poly (7) allax), 'ilie'e (Plumb 1 plants. The early P n plants. Twood sorrel ('ihi'ai or	its (angio ied from non-nativ nesian s ggo zeyla olynesian c Oxalis c	ssperm the t ve) spe ve) spe pecies mica), n intro ornicu	s) observintee survicies. Thr cies. Thr (3%). T and 'uhal and 'uhal duction (s lata), a ve	ed ee oo- ry
re 2) of the proposed SPS is a former pasture (Figure 5, above). "Area No. 2" is an open area idge crossing a concrete-lined drainage channel	Table 1. Plant spec	cies observed at the P	roject si	ites.		
ses (rigure o).	Species listed by family	Common name	Status	Abun	dance Not	es
	FLC	DWERING PLANTS MONOCOTS				
	Commelina benghalensis L. MIISACEAE	hairy <i>honohono</i>	Nat	Ч		
	Musa x paradisiaca L POACEAE	banana	Pol	:	<3>	
	Bothriochloa pertusa (L.) A. Camis	pitted beardgrass	Nat	Oa		
	Chloris barbata L. Cynodon dactylon (L.) Pers.	swollen fingergrass Bermuda grass	Nat Nat	: V	<2, 3> <2, 3>	
	Digiteria insularis (L.) Mez ex Fkman	sourgrass	Nat	0		
	Eleusine indica (L.) Gaertn.	wiregrass	Nat	0	<2, 3>	
	Eragrostis pectinacea (Michx.) Nees	Carolina lovegrass	Nat	U	<2>	
- July	Megathyrcus maximus (Jacq.) R K Simon & WI I Jacobe	Guinea grass	Nat	AA	<2, 3>	
	Paspalum conjugatum Bergius	Hilo Grass	Nat	;	<2, 3>	
	Sporobolus indicus	rattail grass EUDICOTS	Nat	С		
r survey is a section of concrete-lined	ACANTHACEAE Justica betonica L.	white shrimp plant	Nat	:	<2,3>	
channel at Melia Street.	Thunbergia fragrans Roxb.	sweet clock vine	Nat	N		
	ANACARDIACEAE <i>Manaifera indica</i> 1.	mango	Nat	П		
	Schinus terebinthefolius Raddi	Christmas berry	Nat	0c	<3>	
Page 8	AECOS Inc. [Report No, 1613]				Page	6

Results

Natural Resources Survey

Vegetation

Vegetation across the site consi: Guinea grass or *Megathyrsus max* dominated by lantana. Trees are shallow swale feature that transec

The site ("Area No. 3" in Figure dominated by tall Guinea grass (F on Melia Street with a foot bridg and surrounded by private houses

ł



Figure 6. "Area 2" in our drainage c'

Natural Resources Survey		z	JH∃TĚ,Ý	J WWTP, KA		Natural Resources Survey		N	₫,ĨLEHU	WWTP, KA't
Table 1 (continued).						Table 1 (continued).				
Species listed by family	Common name	Status	Abun	dance Note	S	Species listed by family	Common name	Status	Abune	lance Notes
ASCLEPIADACEAE Asclepias physocarpa (E. Mey.) Schlechter	balloon plant	Nat	Þ			FABACEAE (cont.) Neonotonia wightii (Wight & Arnorth Lackev	glycine vine	Nat	U	<3>
ASTERACEAE (COMPOSITAE) Acanthospermum hispidum DC	1	Nat	Я			Prosopis pallida (Humb. & Bonnl. ex Willd.) Kunth	kiawe	Nat	0	
Ageratum conyzoides L	maile hohono	Nat	0	<3>		Senna occidentalis (L.) Link	coffee senna	Nat	0	
Bidens pilosa L. Convert boundriencie Crone	Spanish needle	Nat	: :	<2,3>		Senna surattensis (N.L. Burm.)	scrambled egg plant	Nat	0	
Pluchea carolinensis G.Don	sourbush	Nat	: :	<2, 3/ <3>		п. Irwin & barneby LAMIACEAE	•			
Senecio madagascariensis Poir	fireweed	Nat	n			Hyptis pectinata (L.) Poit.	comb hyptis	Nat	0 0	<2, 3>
AMARANTHACEAE						Leonous neperyona (L.) K. Br. MALVACEAE	11011 S CAI	Nat	D	<75
Amaranthus spinosus L. BRASSICACEAE	spiny amaranth	Nat	R	<2>		Abutilon grandifolium (Willd.) Sweet	hairy abutilon	Nat	0	<3>
Lepidium virginicum L.		Nat	;	<2>		Sida fallax Walp	ʻilima	Ind	0	(
CAPPARACEAE						Sida rhombifolia L.	1	Nat	0	<2>
<i>Cleome gynandra</i> L. CARICACEAE	wild spider flower	Nat	Ŋ	<2>		Sida spinosa L. Urena lobata L.	prickly sida	Nat Nat	ж ж	
Carica papaya L.	papaya	Nat	:	<3>		Waltheria indica L.	'uhaloa	Ind	0c	
CHENOPODIACEAE						MELIACEAE				
Chenopodium murale L. CONVOLVULACEAE	ʻaheahea	Nat	К			<i>Melia azedarach</i> L. MYRTACEAE	Chinabery	Nat	ы	
Ipomoea obscura (L.) Ker-Gawl.		Nat	0			Syzygium cumini (L.) Skeels	Java plum	Nat	N	
Ipomea trilobata L	little bell	Nat	Ŋ	<2, 3>		NYCTAGINACEAE	-			
CUCURBITACEAE		;		0		Boerhavia coccinea Mill.		Nat	;	<3>
<i>Momordica charantia</i> L. EIIPHORRIACEAE	balsam pear	Nat	:	<2, 3>		OXALIDACEAE Ovalis conniculata I	امسمو المستامين	Dol	ď	
Euphorbia cyathophora]. A.	wild noinsettia	Nat	;	<3>		PLUMBAGINACEAE	jelluw wuuu sullel	5	4	
Murray	with pulliseting	ואמר	:	è,		Plumbago zeylanica	ilie'e	Ind	Π	
Ricinus communis L Faraceae	castor bean	Nat	∍			POLYGONACEAE				
Clitoria ternatea L	butterfly pea	Nat	;	<3>		Antigonon leptopus Hook. & Arnott	Mexican creeper	Nat	:	<2>
Crotalaria incanum L	fuzzy rattlepod	Nat	;	<2>		PORTULACEAE				
Chamaecrista nictitans L	partridge pea	Nat	К	<2, 3>		Portulaca pilosa L		Nat	Я	<2, 3>
Desmodium incanum DC.	Spanish clover	Nat	0	<3>		PROTEACEAE				
Indigofera suffruticosa Mill.	indigo, ' <i>ĩnikō</i>	Nat	U	<3>		<i>Grevillea robusta</i> A. Cunn. ex	silk oak	Met	Ξ	
Leucaena leucocephala (Lam.) Aowit	koa haole	Nat	c	<2,3>		R.Br.		Nat	C	
Lotus subbiflowing Los		No+	۵			RUTACEAE				
ьошь зиријгот из Бав. Mimosa pudica L.	 sensitive plant	Nat	2 22			Murraya paniculata W. Jack	mock orange	Nat	:	<2>
AECOS Inc. [Report No, 1613]				Page 1	0	AECOS Inc. [Report No, 1613]				Page 11
)						J

Page | 11

Natural Resources Survey		NĀ'ĀI	EHU WW	TP, KA'u	Natural Resources Su	rvey	NĂ'ĂLEHU V	WWTP	, KA'U
Table 1 (continued).						W whole, 50 to boto the doto of M5'51 with	M/TD cito		
Species listed by family Co	mmon name Stat	/ sn	Abundance	Notes		2. Avial species uelected at Na alellu w On November 5, 2020.	wir sine		
SOLANACEAE Solanum linnaeanum Hepper Apple & P. Jaever	e of Sodom Na	t R			Соттоп Name	Scientific Name	, v	27	8
TILIACEAE									
Triumfetta rhomboidea Jacq VERBENACEAE	Na	t C	ŝ			Phasianinae - Pheasants & Allies	ŝ		
Lantana camara L	na Na	it A	<3>		Domestic Chicken	Gallus gallus	-	4	06:
Stachytarpheta urticifolia (Salisb.) Sims	Na	it 0	-			COLUMBIFORMES			
Verbena litoralis Kunth ōwi	Na	it R	- 4			COLUMBIDAE - Pigeons & Doves			
					Spotted Dove	Streptopelia chinensis		A 1	.70
Legend to STATUS – distributional status for the Ha	o Table 1 awaiian Islands:				Zebra Dove	Geopelia striata		A 2	.50
Ind = indigenous; native to Haw	awallall islallus: vaii, but not unique to th	e Hawai	ian Islan	S.					
Nat = naturalized, exotic, plan	it introduced to the Ha	awaiian	Islands	since the					
arrival of Cook Expedicultivation	ition in 1778, and we	ell-estab	lished o	utside of		CHAKAUKIIUAE - LAPWINGS & Plovers Charadriimaa - Dlavars			
Orn = A cultivated plant; a spec:	cies not thought to be na	turalize	d (spread	ing on its			:	-	0
own) in Hawai'i.	5		;	0	Pacific Golden-Plover	Pluvialis Julva	=	⊃ ∑	Dč.
ABUNDANCE = occurrence ratings for pla	ant species:	location				DACCEDIEODMEC			
U - Uncommon seen at m	nuy one or pernaps two nost in several locations	IUCAUUII	'n			AI AI IDI DAF - I arks			
0 - Occasional seen with	ch some regularity.							•	ç
C - Common observed	d numerous times during	g the sur	vey.		eurasian skylark	Alauaa arvensis		T A	.40
A - Abundant found in .	large numbers; may be l	locally d	ominant			2051EKUPIDAE - White-eyes			
NOTES: Not observed in Area No					Warbling White-eye	Zosterops japonicus		A 1	.20
<2> - Noted present in Area.	No. 2. Mo. 2					STURNIDAE - Starlings			
<3> - Noteu present III Area <45 - Plant lacking key diagn	INO. J. Anstis charactaristics (fls	mar fri	it).		Common Myna	Acridotheres tristis	`	A 5	.20
identification, therefore	re, uncertain.	n 11 (17 M)	(1)			ESTRILDIDAE - Estrildid Finches			
	×				Common Waxbill	Estrilda astrild	1	A 1	<u>.</u> 90
				1	African Silverbill	Euodice cantans		0 4	.60
					Java Sparrow	Lonchura oryzivora		0 4	.70
					Scaly-breasted Munia	Lonchura punctulata		A 1	.10
						PASSERIDAE - Old World Sparrows			
Avian Fanna					House Sparrow	Passer domesticus		0 4	.60
						EPINGULUAE - Eringilling and Carduling Einche	ac 8. Alliac		
م 2016 مفتار المناط المناط المناطقة من 15 م	onitaconance ociocan	10 00	a croto			Carduelinae - Carduline Finches and Hawa	co examico alian		
A WEAT OF 233 IIIUIVIAUAI DII US OF 13 S	peuco, representant o		palate						
were recorded during point counts.	. Une species reco	rded,	Pacific	Golden-	:				ŝ
Plover (<i>Pluvialis fulva</i>), is an indige	enous migratory sh	orebir	d speci	es. The	House Finch	Haemorhous mexicanus	-	A 2	07.
remaining 13 species recorded are all	ll established alien c	or feral	specie	s (Table	Yellow-fronted Canary	Ceithagra mozambica	-	A 1	.60
<i>2</i> J.									

Page | 12

AECOS Inc. [Report No, 1613]

Page | 13

latural Resources Survey		NĂ'ĀLEHU W'	WTP, KA'U	Natural Resources Survey NATALEHU WV	WWTP, KA'U
Table 2 (continued)				Table 3 (continued).	
Common Name	Scientific Name	ST	84	Common Name Scientific Name ST	Detection
	THRAUPIDAE - Tanagers			ATRIODACTYLA - EVEN-TOED UNGULATES Suicidae - Old World Swine	
	Thraupinae - Core Tanagers			Pig Sus scrofa A	V, Sc, Tr, Si
Saffron Finch	Sicalis flaveola	A	0.50	Bovidae-Hollow-horned Ruminants	
	Key to Table 2			Domestic cattle Bos taurus A	V, A, Ir, Sc
37 Status				Key to Table 3	
A Alien – Introduced	d to the Hawaiian Islands by humans			57 Status	
IM Indigenous migrat	tory- Native migratory shorebirds species which br	seds in the high	Arctic	A Alien – Introduced to the Hawaiian Islands by humans	
R4 Relative Abundance	e - Number of birds detected divided by the numb	er of point cour	nts (10)	V Visual – the animal was seen	
				Detection type	
				A Audio – the anima; was heard	
				Sc Scat – feces of the anima; were observed	
Avian diversity and	densities were in keening with the loca	tion of the th	rree sites	Tr Tracks – tracks of the animals were seen	
and the vegetatic	on present. Three introduced speci	es—Commo	in Myna	Si Sign – browsing's signs, beds, rooting, dust baths etc. were observed	
(Acridotheres tristis (Geopilia striata)—	s), House Finch (Haemorhous mexican accounted for 44% of the total numb accounted for 44% of the total numb	us), and Zel	ora Dove recorded.		
common myna was percent of the birds	s the most commonly tained species, wi s recorded during point counts.	nen account	c7 101 ba	Discussion and Recommendations	
Mammals				Recommendations are narrly based on 11.5. Fish and Wildlife Service	re Animal
Terrestrial mamma are presented in Tal	alian species detected along with the r ble 3.	nethods of (detection	Avoidance and Minimization Measures (USFWS-PIFWO, nd). Implement the recommendations (provided below as bulleted items) by the contractor will minimize impacts to protected species to the maximur	entation of he Project um extent
				practicable.	
Table 3. Mamm	ialian species detected at Nā'ālehu W	WTP site in	2020.	Floral Resources	
Common Name	Scientific Name	ST	Detection	No plants of conservation concern or enjoying statutory protection	n (that is,
	CARNIVORA- FLESH EATERS			listed as threatened or endangered; HDLNR, 1998; USFWS, 2020) were	re noted in
	Canidae - Wolves, Jackals & Allies			the survey and given the highly disturbed nature of the site, would	uld not be
Domestic dog	Canis lupus familiaris	A	V, A, Sc	expected to be growing there. For comparative purposes, this survey in the survey of early Do	ty revealed Polynesian
Construction and the second	VIVERTIGAE - LIVETS & AIIIES	<	~ ~	introduction. The previous site-located on the <i>makai</i> side of Mā	fāmalahoa
	Felidae-Cats	¢	£ `>	Highway 1 mi (1.6 km) east (also pastureland)—yielded 30 sp	species of
House cat	Fells catus	٩	>	introduced plants and 1 species each of indigenous and early Po introduction. A survey of the much smaller Wai'ōhinu Transfer Station 2020) in Wai'ōhinu, 2 mi (3.4 km) to the west produced a list of	Polynesian on (AECOS, i 59 taxa.
AECOS Inc. [Report No, 1613]			Page 14	AECOS Inc. [Report No, 1613]	Page 15

Natural Resources Survey

Saffron Finch

AECOS Inc. [Report No, 1613]

AECOS Inc. [Report No, 1613] Page 17	Page 16	AECOS Inc. [Report No, 1613]
No riawanan noary bats or were detected during the course of our survey. It is within the realm of possibility that this species uses resources within the	the summer and fall, can	seabirds, especially fledglings on their way to sea in t
LITCLIA.	man-made structures is	Hodges, 1998; Ainley et al., 2001). Collision with
deleterious to native ecosystems and the native faunal species dependent on thom	t to be predation by allen FWS, 1983; Simons and	mortaiity in these listed seabirds in Hawai'i is thought mammalian species at inland nesting colonies (USF
habitation and activity. All of these introduced mammalian species are	ers. A primary cause of	middle of December each year in very small numbe
Project area. These human commensal species are drawn to areas of human	ea between April and the	Shearwater (Puffinus newelli) over-fly the Project are
small Indian mongoose (<i>Herpestes javanicus</i>) use resources in the general	the threatened Newell's	Band-rumped Storm-Petrel (Hydrobates castro), and
<i>norvegicus)</i> , and black rat (<i>kattus exulans hawanensis</i>)—use various resources found within the general project area on a seasonal basis. It is also likely that	(Puffinus sandwichesis),	It is possible that the endangered Hawaiian Petrel
house mouse (Mus musculus), roof rat (Rattus rattus), brown rat (Rattus		breeding grounds in late April or the first week of May.
more of the other four established alien Muridae found on Hawai'iEuropean	species migrates to Arctic	late September and the end of April each year. This s
introduced species. Although no rodents were recorded, it is likely that one or	across the state between	an indigenous migratory shorebird commonly seen a
The five mammalian species recorded during these surveys are all common	e Pacific Golden-Plover, is	species upfinition by anen species. Four teen of the right
Mammalian Resources	, have very limited plant	places. The other two smaller sites, located in town,
	at has been bulldozed in	the three sites. The WWTP site is mostly pasture the
Undergeted species secures is associated with the project site (112 prov. 2013). USFWS, nd-a).	he vegetation nresent on	The findings of the avian survey are consistent with th
proposed for listing as threatened or endangered under federal or state		Terrestrial Birds
attracted to ponds associated with the WWTP. No species currently listed or		uue to Froject acuvities as non-existent.
No habitats suitable for waterbird species are present on any of the three sites.	at to these insect species	utilized by the caterpillar. We would deem any threa
	tild tilenty without plant	(Solanum lycopersicum var. cerasiforme). This speci
external lignung structures snould be fully dark sky compliant (HDLNR-D0FAW, 2016).	he only plant observed	native tree tobacco plant (<i>Nicotiana glauca</i>). Th
off lights when human activity is not occurring in the lighted area. All	; widely distributed, non-	found, caterpillars are most often associated with the
sensor switches and timer controls on all installed outdoor lights or turn	of the sphinx moth feeds ticular where the moth is	habitat or food sources were noted. The caterpillar c exclusively on plants in the Family Solanaceae. In parti
 Fully shield all outdoor lights so the bulb can only be seen from below bulb height and only use when necessary install automatic motion 	survey and no potential	No yellow-faced bee species was observed during the
2) exterior lighting around Project infrastructure.	o, zuloj and biackourn s 5).	niuris, n. kuakea, n. iongiceps, and n. mana; USFWS, sphinx moth (<i>Manduca blackburnii</i> ; USFWS, 2000, 2005
lights associated with the Project. The two main types of outdoor lighting posing a threat to nocturnally flying seabirds are: 1) night-time construction lights: and	H. assimulans, H. facilis, H.	species of the yellow-faced bee (Hylaeus anthracinus, H
(from September 15 through December 15) after becoming disoriented by	Hawaiian Islands: seven	Several insects are now listed as endangered in the
The potential impact that the Project poses to protected seabirds is the increased threat that birds will be downed during the seabird fledging season		Insects
Ainley et al., 2001; Hue et al., 2001; Day et al., 2003).		Faunal Resources
man-made structures and, it not kliled outright, become easy targets of opportunity for feral mammals (Hadley, 1961; Telfer, 1979; Sincock, 1981; Reed et al., 1983: Telfer et al., 1983: Conner and Dav. 1998: Podolsky et al., 1998;	to frequent introductions	(increasing the count of ruderal species) and subject t of species on vehicles and rubbish loads.
become disoriented by exterior lighting. Disoriented seabirds may collide with	s both highly disturbed	Although a much smaller area, the latter site is
Natural Resources Survey Natural Resources Survey	NĂ'ĀLEHU WWTP, KA'U	Natural Resources Survey

	24, 2021.		
Chen, A.G. Frazier, J.P. Price, YL. Chen, PS. Chu, J.K. M. Delparte. 2013: Online Rainfall Atlas of Hawai'i. <i>Bull.</i> <i>Soc</i> 94: 313-316, doi: 10.1175/BAMS-D-11-00228.1. http://rainfall.geography.hawaii.edu//.last.acressed_on_lune	Giambelluca, T. W., Q. Eischeid, and D <i>Amer. Meteor.</i> Online af IIIL:	ns or outlets on the ocean shore. Annual e order of 40 in (1020 mm; Giambelluca et ot conducive to stream formation.	drainageways direct flows to strea rainfall in the Naʿālehu area is on th al, 2013) and the porous geology is r
	120: 669-679,	on Melia Street is similar in directing runoff e further downslope. Neither of these	crossed by the proposed force main through town into shallow swal
nd T. C. Telfer, 2003, Decline of Townsend's (Newell's uffinus auricularis newelli) on Kauai, Hawaii, The Auk,	Day, K. H., B. Cooper, Shearwaters (P	Project site. A concrete drainage channel	of water flow in the swale at the
		e swale and runoff flow will be directed	is located within a shallow drainag
. Day. 1998. Summer Behavior and Mortality of Dark- and Newells' Shearwaters at Power Lines on Kauai.	Cooper, B. A. and R. F rumped Petrels Colonial Waterby	Project area, although the WWTP site itself	Jurisdictional Waters No streams or wetlands occur in the
ological Consulting, Inc. 18 pp.	Hawai'i. Rana B	Project.	Conservation Districts occur near the
Guinther. 2012. Biological Surveys Conducted for the water Treatment Plant Project, Ka'ū District, Island of	David, R. E., and E. B Nā'ālehu Waste	sts under state law. Conservation zoning in level by state Conservation Districts No	nd-b). No equivalent designation exi Hawai'i is promulgated at the state
tp://checklist.aou.org/taxa; last retrieved July 31, 2020.	online at URL: h	outer of the provident of the provident of the second second second second second second second second second s	test de la factoria de la constante de la const
K Winker 2020. Sixty-first Supplement to the Check-list can Birds. American Ornithological Society. Available	D. F. Stotz, and of North Amer		Critical Habitat
J. V. Remsen Jr.		ncern	Other Resources of Potential Co
dvances, 136: 1-23.	Ornithological A	ıd Bonaccorso, 2010).	should be avoided (Zimpfer a
lason, P. C. Rassmusen, and K. Winker. 2019. Sixtieth the Check-list of North American Birds. <i>The Auk</i>	Lovette, N. A. N Supplement to	fence lines may entangle flying bats and	The use of barbed wire to top
erman, K. J. Burns, C. Cicero, J. L. Dunn, A. W. Fratter, I. J.	Chesser, R. T., S. M. Bil		September 15 (USFWS, nd-c).
остеку, лиси и гезэ, дамтенсе, манзаз, т р. 100-120.		upping season between June 1 and	removed during the bat p
una. Studies in Avian Biology No. 22. Cooper's Aviety Allen Press Lawrence Vanses Dr. 108-123	Vanishing Avif Ormithological S	i impacts to roosting bats with pups, it is vegetation taller than 4.6 m (15 ft) be	 Io avoid potential deleteriou: recommended that no woody
in. Sout, J. M. S. Contair, and C. Van Ripel III (eurors) <i>JJ</i> , Conservation, and Management of Hawaiian Birds: A	Evolution, Ecolo		- - - -
Trends of the Newell's Shearwater on Kaua'i: Insights	and Population	ey themselves forage, and very small pups in رامالمرا	their pups in the roost tree while th may be unable to flee a tree that is be
y, L. Deforest, G. Spencer, and N. Nur. 2001. The Status	Ainley, D. G, R. Podols	tionally, adult female bats sometimes leave	site while vegetation is cleared. Addi
	14 pp.	s may be less able to rapidly vacate a roost	pupping season, female carrying pup
rep. for Wilson Okamoto Corporation. AECOS No. 1618:	005:001 por.). I	e territory, so the potential disturbance	multiple roosts within their hom resulting from the removal of the ver-
20. A natural resources assessment for improvements to Transfer Station, Wai'ohinu, Ka'u, Hawai'i (TMK 9-5-	AECOS Inc. (AECOS). 20 the Wai'ōhinu	nstruction areas may temporarily displace cation as a roost. However, a bat will use	of foliage and/or trees within the cc an individual bat utilizing the vege
	References Cited	e potential impact to bats would be during he construction. The trimming or removal	Project area on a seasonal basis. The the clearing and grubbing phase of the clearing area.
NĂĂLEHU WWTP, KA'U	Natural Resources Survey	NĂ'ĂLEHU WWTP, KA'U	Natural Resources Survey

Natural Resources Survey NĂLEHU W	WTP, Kaʻu	Natural Resources Survey NA'ALEHU WWTP, KA'
Hadley, T. H. 1961. Shearwater calamity on Kauai. ' <i>Elepaio</i> , 21: 60.		Sincock, J. L. 1981. Saving the Newell's Shearwater. Pages 76-78 in Proceeding
Hawaii Department of Land and Natural Resources (HDLNR). 1998. In Wildlife, Enderstood, And Theoreman, Wildlife, And Dhee	digenous	of the frawari rolessly and whithing contretence, 2-4 occourt 2000 Department of Land and Natural Resources, State of Hawaii, Honolulu.
Withine, Entangereu And Inteactureu Withine And Fiat Introduced Wild Birds. Department of Land and Natural Resour of Hawaii. Administrative Rule §13-134-1 through §13-134-1 March 02, 1998.	us, Anu ces State 10, dated	Staples, G. W. and D. R. Herbst. 2005. A Tropical Garden Flora. Plants Cultivate in the Hawaiian Islands and other Tropical Places. Bishop Museun Honolulu. 908 pp.
2015. Hawai'i Administrative Rules, Title 13, Department of I Natural Resources, Subtitle 5, Forestry and Wildlife, Part 2	Land and Wildlife,	Telfer, T. C. 1979. Successful Newell's Shearwater Salvage on Kauai. ' <i>Elepai</i> 39:71
Chapter 124, Indigenous Wildlife, Endangered and Threatened Injurious Wildlife, Introduced Wild Birds, and Introduced February 27, 2015. 16 pp.	Wildlife. Wildlife	J. L. Sincock, G. V. Byrd, and J. R. Reed. 1987. Attraction of Hawaiia seabirds to lights: Conservation efforts and effects of moon phase Wildlife Soc. Bull., 15: 406-413.
Hawai'i Department of Land and Natural Resources-Division of Fore Wildlife (HDLNR-D0FAW). 2016. Wildlife Lighting. PDF ave URL: http://dlnr.hawaii.gov/wildlife/files/2016/03/D0C439.pdf, last February 21, 2020.	sstry and ailable at retrieved	U.S. Fish & Wildlife Service (USFWS). 1983. Hawaiian Dark-Rumped Petrel Newell's Manx Shearwater Recovery Plan. USFWS, Portland, Oregoi February 1983.
Hue, D., C. Glidden, J. Lippert, L. Schnell, J. MacIvor and J. Meisler. 2003 Use and Limiting Factors in a Population of Hawaiian Dark Petrels on Mauna Loa, Hawai'i, <i>in:</i> :Scott, J. M. S. Conant, and C. V	Habitat <-rumped /an Riper	2000. Endangered and Threatened Wildlife and Plants; Determination c Endangered Status for Blackburn's Sphinx Moth from the Hawaiia Islands, <i>Federal Register</i> , 65 (21; February 1, 2000); 4770-4779.
III (editors) Evolution, Ecology, Conservation, and Manage Hawaiian Birds: A Vanishing Avifauna. Studies in Avian Biolog Cooper's Ornithological Society, Allen Press, Lawrence, Kansas 2020	sment of y No. 22. (Pg. 234-	
242). Imada, C. T. 2019. Hawaiian Naturalized Vascular Plants Checklist (2019 update). <i>Bishop Museum Tech. Rept.</i> 69. 209 pp.	February	2016. 50 CFR 17, Final Rule: Endangered and Threatened Wildlife an Plants; Endangered Status for 49 Species from the Hawaiian Island. Federal Register, 81 (190; Friday, September 30, 2016): 67786-67860.
Podolsky, R., D. G. Ainley, G. Spencer, L. de Forest, and N. Nur. 1998. of Newell's Shearwaters Caused by Collisions with Urban Struc Kana'i, Colonial Waterbirds 21: 20-34.	Mortality ctures on	
Reed, J. R., J. L Sincock, and J. P. Hailman 1985. Light Attraction in En Procellariform Birds: Reduction by Shielding Upward Radiation. 100-277-383	dangered . <i>The Auk</i> ,	—. Undated (nd-b). Critical Habitat Portal. Available online at URI https://ecosfws.gov/ecp/report/table/critical-habitat.html; last retrieved Ma 20, 2020.
Simons, T. R., and C. N. Hodges. 1998. Dark-rumped Petrel (<i>Pte phaeopygia</i>). <i>In:</i> A. Poole and F. Gill (editors). The Birds America, No. 345. The Academy of Natural Sciences, Philadel and the American Ornithologists Union, Washington, D.C.	<i>erodroma</i> of North phia, PA.	U.S. Fish & Wildlife Service-Pacific Islands Fish and Wildlife Office (USFWS PIFWO). No date (nd). Avoidance and Minimization Measures. Availabl online at URL: https://www.fws.gov/pacificislands/articles.cfm? id=14948972 last retrieved July 31, 2020.
AECOS Inc. [Report No, 1613]	Page 20	AECOS Inc. [Report No. 1613] Page 2

AECOS Inc. [Report No, 1613]

- Wagner, W. L., D. R. Herbst, and S. H. Sohmer. 1990. Manual of the Flowering Plants of Hawai'i: Volume 1 and II. Bishop Museum Special Publication 83. University of Hawai'i Press. 1853 pp.
- and ______ 1999. Supplement to the Manual of the flowering plants of Hawai'i, pp. 1855-1918. In: Wagner, W. L., D. R. Herbst, and S. H. Sohmer, Manual of the flowering plants of Hawai'i. Revised edition. 2 vols. University of Hawaii Press and B.P. Bishop Museum.
- Wilson, D. E. and D. M. Reeder (eds.). 2005. Wilson & Reeder's Mammal Species of the World (Third Edition). Available online at URL: http://www.departments.bucknell.edu/biology/resources/msw3/browse.asp; last retrieved December 9, 2019.

APPENDIX C:

Section 7 Consultation

Initiation Letter

JOSH GREEN, M.D. OOVERNOR OF HAVIAT KERAVAINA OKA MORU AINA 'O HAVIAT



KENNETH S. FINK, MD, MGA, MPH DIRECTOR OF HEALTH KALUNA HOTKELE

> STATE OF HAWAII DEPARTMENT OF HEAL TH KA 'OIHANA OLAKINO P. O. BOX 3378 HONOLULU, HI 96801-3378

in reply, please refer to: File:

90-04, 90-07 S7 Itr (initial).docx

May 31, 2024

Dr. Earl Campbell, Field Supervisor U.S. Department of the Interior U.S. Fish and Wildlife Service Pacific Islands Fish and Wildlife Office 300 Ala Moana Boulevard, Room 3-122 Honolulu, HI 96850 Email: <u>pitwo_admin@fws.gov</u>

Dear Dr. Campbell:

- Subject: Initiate Informal Consultation and Request Concurrence of Proposed Determination Under Section 7 of the Endangered Species Act Nä'älehu Wastewater Collection System and Nä'älehu Wastewater Treatment and Disposal System and Drainage Modifications
 - Usposel system and Unamage modifications Clean Water State Revolving Fund Project No. C150090-04 and C150090-07 Information for Planning and Consultation (IPaC) Project Code: 2024-0078740

The U.S. Environmental Protection Agency (EPA) has designated the State of Hawai' Department of Health (DOH) as its non-federal representative pursuant to 50 CFR Section 402.08 for purposes of initiating the consultation process and preparing a biological assessment, if necessary, under Section 7 of the federal Endangered Species Act (ESA) for certain projects funded under the Clean Water State Revolving Fund (CWSRF) program. The County of Hawai'i Department of Environmental Management (COH-DEM) is planning to undertake the "Närälehu Wastewater Collection System and Närälehu Wastewater Treatment and Disposal System and Drainage Modifications" project. These projects plan to use CWSRF program funds and are contracting your office to initiate the informal consultation process under Section 7 of the ESA.

We are contacting your office to initiate the informal consultation process and request concurrence of the proposed determination under Section 7 of the ESA, request a list of threatened and endangered plant and animal species and critical habitats within the project and/or action area, and determine whether there are any other service trust resources, which may include, but not limited to the Migratory Bird Treaty Act (MBTA), beyond the species described and disterts.

Project Description

The COH-DEM is proposing to construct, wastewater system improvements to replace existing collection, treatment, and disposal systems that service the community of Na'älehu in the Kā'u District on the Island of Hawai'I. The wastewater system improvements would allow the Country to comply with the EPA regulations requiring the closure of the Large Capacity Cesspools (LLC)

Dr. Earl Campbell, Field Supervisor May 31, 2024 Page 2 currently in service and to construct a new system meeting the DOH and DEM design guidelines for collecting, treating, and disposing of community wastewater. The project would route the new collection and transmission systems primarily within existing rights-of-way, with easements where that isn't feasible or practical, and would provide a treatment and disposal system to allow closure of the three LCCs to meet EPA requirements. The proposed improvements will occur on lands identified as TMK: 9-5-007-:016, 9-5-022:001, 9-5-015:015, and various other parcels within Nä"älehu. The above-ground package wastewater treatment plant will be sited on the south side of Māmalahoa Highway (State Route vasteture turs east-west through the community. Collection and delivery systems will primarily the on the north (upslope) side of the highway within the developed community (*Attachment A*). The Proposed Action includes four alternatives all using the above-ground package wastewater treatment plant but with different options for collection and delivery—these alternatives are discussed in detail in the attached DEM project package (*Attachment B*).

Consultation History

The project consultants sent a letter to the US Fish and Wildlife Service (USFWS) dated June 24. 2019, requesting a Technical Assistance (TA) review of the Draft Environmental Assistance (TA) review of the Draft Environmental Assessment for the Na^{*}alehu LLC dosure project. A response was received from USFWS in a letter dated July 15, 2019, assigning a TA consultation No. 01EPIF00-2019-TA-0388 for the project and including a list of species that could potentially be impacted by the project *Attachment* C).

The official species list was obtained through the IPaC system on April 17, 2024 and identified 23 species that could potentially be present in the area and be affected by the proposed action. Of the species mentioned—one mammal, seven birds, one repille, one insect, twelve flowering plants, and one ferm—the only ones that could potentially be impacted by the action are the Hawaiian heary bat (Lasiurus semotus), Band-rumped Simm Petrel (Hydrobatis castro). Hawaiian Petrel (Pterodroma sandwichensis), and Newell's Shearwater (Puffinus newelli) based on the biological surveys.

Biological Surveys

Two biological surveys were conducted on the sites: one in 2012 and one in 2021; both surveys were conducted by AECOS Inc. (*Attachment D*). During the botanical, avian, and mammalian resources surveys no listed or proposed-for-listing species under federal statutes were recorded. Also, a Natural Resources Assessment letter was prepared for the project in 2020 (*Attachment E*). To avoid and minimize the impacts, the following minimization measures will be incorporated into the project's plans and/or specifications.

<u>Hawaiian Hoary Bat</u>

To avoid or minimize effects to the Hawaiian hoary bat, the following conservation measures will be incorporated into the project where applicable:

- No barbed wire will be used for fencing.
- A provision will be included in the construction bid documents requiring the contractor to provide a qualified biologist to schedule surveys during the Hawaiian hoary bat birthing and pup rearing season between June 1 to September 15 should woody plants taller than 15 feet be trimmed, removed, or disturbed prior to the start of construction activities. If woody plants taller than 15 feet are removed outside of that time frame, the

Dr. Earl Campbell, Field Supervisor May 31, 2024 Page 3 surveys will not be necessary as work will fall outside of the peak birthing and pup rearing season.

Hawaiian Seabirds (Band-rumped Storm-petrel, Newell's shearwater, Hawaiian Petrel) To avoid or minimize effects to Hawaiian seabirds, the following conservation measures will be incorporated into the project where applicable:

- Fully shield all outdoor lights so the bulb can only be seen from below bulb height and only use when necessary.
- Install automatic motion sensor switches and timer controls on all outdoor lights or turn off lights when human activity is not occurring in the lighted area.
 - Avoid nightime construction during the seabird fledging period. September 15 to December 15 on all islands.
 Install any required fencing extending above vegetation with three strands of polytape
 - Install any required fencing extending above vegetation with three strands of polytak into the fence to increase visibility.

The project will **not affect** the remaining species identified in the official species list as biological surveys (*Attachment D*) concluded these species were neither present nor likely to be present in the project area.

Request for USFWS Assistance

Based on the project description along with the proposed avoidance and minimization measures that will be implemented, the project may affect, not likely to adversely affect these endangered or threatened species in the area as they will not be disturbed.

We are contracting your office to initiate the informal consultation process and request the concurrence of the proposed determination under Section 7 of the ESA and request a list of threatened and endangered plant and animal species and critical habitats within the project and/or action area that has not been mentioned above and/or in the natural resource assessment.

Your response within sixty (60) calendar days of receipt of this letter is greatly appreciated. Please address your written response to the following email or mailing addresses: <u>Chane.Hayashida@doh.hawaii.gov</u>

Attn: Chane Hayashida Department of Health, Wastewater Branch 2827 Waimano Home Road, Room 207 Pearl City, HI 96782

Dr. Earl Campbell, Field Supervisor May 31, 2024 Page 4 Should you have any questions, please contact Chane Hayashida at (808) 586-4294.

Sincerely,

JONATHAN NAGATO, P.E., ACTING CHIEF Wastewater Branch

Attachments CH:jn C: Ramzi Mansour (via email at <u>Ramzi Maansour@hawaiicountv.gou</u>) Chris Sparber (via email at <u>Chris Sparber@hawaiicountv.gou</u>) Mark Grant (via email at <u>MarkJ.Grant@hawaiicountv.gou</u>)



ATTACHMENT A



ATTACHMENT B

NĂ ÂLEHU LCC CLOSURE ENVIRONMENTAL INFORMATION DOCUMENT & ENVIRONMENTAL ASSESSMENT CONSULTATION PACKAGE	
ACKGROUND	
thalf of the County of Hawai'i (County) Department of Environmental Management (DEM), n Okamoto Corporation has prepared this Consultation Package to notify stakeholders of the ration of a Draft Environmental Information Document (EID) / Environmental Assessment for the Năãlehu Large Capacity Cesspool (LCC) Closure (Proposed Action), as well as to scoping input. The Proposed Action will be situated within the community of Năãlehu in	serve approximately 164 private residences in the community of NaʾAlehu, violated and continues to violate the SDWA and its Underground Injection Control program requirements for existing LCCs.
a'ū District, Island of Hawai'i (See Figure 1).	The Amended AOC (Docket No. SDWA-UIC-AOC-09-2017-0002) LCC closure requirements for the
lä alehu community was established as the result of the sugar operations of the C. Brewer any. A portion of the community is serviced by a sewer system that was privately built.	Natalehu community mandates that an ELD be prepared by the County of Hawai'i for the EPA approval to meet Federal Environmental Review Requirements.
1, and operated by the C. Brewer Company, which merged with Hawaiian Agriculture	The Amended AOC 31.a. requires evaluation of four feasible options:
any (TAC) in 1972. The wasteward contected by the seven system disularges into the capacity "gang" cesspools (LCCs). Many years after its establishment, the private sever	 A package plant and new collection system (Alternative 1) A package plant connected to the existing collection system (Alternative 2)
n ownership was conveyed to the County of Hawaii Department of Environmental gement. In anticipation of C. Brewer's dissolution, C. Brewer proposed, and the County	3. A maintenance contract model Individual Wastewater System (IWS) program (Alternative
d, to enter into a formal agreement not only to construct and maintain a new and improved unity sewer system but to assume ownership of the existing system including the LCC's by	4. An operating permit model IWS program (Alternative 4)
30, 2010.	In addition, to meet the requirements of the EPA, this EID / EA will also evaluate:
while, in 1998, the U.S. Environmental Protection Agency (EPA), promulgated regulations,	5. A NU ALUON AICENTAUVE.
de of Federal Regulations (CFR) 144.14, that require the elimination of LCCs. In 1999, EPA 1 reculations under the Safe Drinking Water Act's (SDWA) Underground Injection Control	The subject EID / EA is intended to be consistent with requirements of the National Environmental Policy Act (NEPA) 42 U.S.C. § et seq. and will also provide documentation of necessary compliance
Program which prohibited the construction of new LCCs as of April 2000 and required the	with Section 7 of the Endangered Species Act and Section 106 of the National Historic Preservation
e of all existing LCCs by April 5, 2005 (40 C.F.R. § 144.88). Under federal regulations, an	Act. Further, the document shall be prepared pursuant to Hawai'i Revised Statutes (HRS) §343-5 (1) which outlines remuirements for environmental disclosure for actions that propose the use of
a cesspour which serves manape avenings, or nor non-resuctivita radinates has the departy ve 20 or more persons per day.	state or county lands or the use of state or county funds.
ie 2017, the EPA and the County entered into an Administrative Order on Consent (AOC) to	This project may also be funded by the DOH Gean Water State Revolving Fund (CWSRF) Program. Under the CWSRF program, the project consists of two parts: Natalenu Large Capacity Cessbool
the LCCs serving the Navalehu community by June 2021. Options considered by the County	Conversion and Na ale Wastewater Collection System. The CWSRF Program was created by the
se the LLLs include construction of a new sewer collection system located within public of-way (ROW) and replacement of the existing LCCs with a wastewater treatment plant	federal Water Quality Act of 1987 and authorizes low interest loans for the construction of publicly owned wastewater treatment works
P) to address the wastewater treatment and disposal needs of the Nataleiu community. A	
up Americaed AOC was agreed upon between EFA and County, and took in effect August 22, which requires the LCCs to be closed no later than December 31, 2027.	Supporting studies, to be included within the EID / EA, are anticipated to include Archaeological and Cultural documentation, including a Ka Pa'akai Analysis, as well as Botanical / Faunal Surveys
August 22, 2022, the County of Hawai'i and the EPA voluntarily entered into an Amended or the purpose of bringing the County into compliance with the requirements of the SDWA, 5.C. § 300f, et seq.	and regulatory coordination. The County will seek to coordinate with the State of Hawai'i Department of Health (DOH) to ensure timely review and concurrence by the State of Hawai'i State Historic Preservation Division (SHPD) and the US Fish and Wildlife Service (FWS).
ias determined that the County, as the current owner and/or operator of two LCCs that	
approximately 109 private residences in the community of Pāhala and three (3) LCCs that	Nă fălehu LCC Closure EIN / EA Concultation Declana – March 2024
hu LCC Closure A Early Consultation Package – March 2024	EIU / EA CONSUMATION FACABLE - MARICH 2024

1. BACKGROU

Wilson Okamoto C preparation of a D (EA) for the Naïāl solicit scoping inpu the Ka'ū District, Is On behalf of the

The Nă'ālehu com Company. A porti owned, and oper: large capacity "ga system ownership agreed, to enter ir community sewer April 30, 2010. Company (HAC) Management. In

Meanwhile, in 199 40 Code of Federa issued regulations (UIC) Program wh closure of all exist LCC is a cesspool to to serve 20 or mo In June 2017, the close the LCCs ser to close the LCCs right-of-way (ROW (WWTP) to addres recently Amended 2022, which requir As of August 22, 2 AOC for the purpo 42 U.S.C. § 300f,

EPA has determi serve approxima

Nā'ālehu LCC Closu EID/ EA Early Consu

2. PROJECT DESCRIPTION AND ALTERNATIVES	<u>Alternative 2 - Package Plant with Existing Collection System</u>
The Proposed Action includes the construction of facilities which would allow the County to close the three LCCs in Naãalehu and thereby meet the compliance requirements of the Amended AOC and the applicable portions of the Cean Water Act. The Proposed Action would be achieved by any of the 4 alternatives set forth in the Revised AOC and described below.	Under Alternative 2, the County would perform the same actions as described under Alternative 1; however, gravity sewers would not be installed within the streets to replave the existing system. Instead, the new wastewater treatment and disposal facility and pump station would be connected to the existing 80-year old collection system.
Alternative 1 - Package Plant with New Collection System	<u> Alternative 3 - Individual Wastewater System-Maintenance Contract Model</u>
Under Alternative 1, the County would construct a new sewer collection system in the Nă'ālehu community to replace the existing system of gravity lines that convey sewage to the three LCCs and connect it to the proposed wastewater treatment and disposal facility. The WWTP would serve the former Brewer lots as well as newly accessible parcels for future connection to the WWTP (See Figure 2). The potential connection of additional parcels may be further assessed by	Hawai'i Revised Statutes, Title 14, Taxation, Chapter 235, Income Tax Law, § 235-16.5 defines a septic system as an individual wastewater system (IWS) that typically consists of a septic tank, piping, and a drainage field where there is natural biological decontamination as wastewater discharged into the system is filtered through soil.
the County in the Inture. During constructuon, the County shall ensure that residential units can maintain access to the sewer system at all times. Alternative 1 involves utilizing the existing collection system within the Brewer Company house lots and constructing new gravity sewers, wastewater pump station, and force main to transport sewage from the LCCs to the new WWTP (See Figure 3). Alternative 1	The August 2023 PER, stated the State DOH Wastewater Branch is responsible for regulating IWS systems. The Revised AOC sets forth that the County Hawai'i must administer a more active management strategy than is typically found in Hawai'i IWS systems. The Revised AOC states either a Model 2 (Maintenance Contract) or a Model 3 (Operating Permit) must be used IWS systems at Pahala. Figure 4 shows the lots where the IWS systems shall be constructed.
also involves the installation of gravity sewers to replace the existing collection system (See Figure 4)	The April 2023 PER indicates for a Model 2 Alternative, the County is to:
 Under this alternative, the County would perform the following actions: Construct a new gravity sewer on Kaalaiki Road and Nă'ālehu Spur Road to the WWTP located on a portion of Tax Map Key (TMK) (3) 9-5-007:016. Implement drainage improvements within the vicinity of the WWTP within TMKs (3) 9-5-007:016 and (3) 9-5-021:015. Construct a new purpostation located on a portion of TMK (3) 9-5-008:048, and construct a new pravity sewer. Construct a new gravity sewer on Optica Street and Ohai Road to intercept existing flows entering the LCCs and divert sewer along kalaliki Road. Install gravity sewer and gives the wastewater pump station (WWPS) and transport flows to the gravity sewer along the existing collection system. Install gravity sewer solity cublace the existing collection system. Cose and abandon the three LCCs. 	 rund design and manage project construction or up the systems; Administer and manage project construction or up the systems; The maintenance program would entail establishing rules and regulations, for monthly fees/penalties, County monitoring and reporting, and IWS educational information for homeowners Conduct routine maintenance and operate system, and respond to related trouble calls; and Prepare and submit related notices and reports. Prepare and submit related notices and reports. Under Alternative 3, the County funds and manages project construction of the IWS which typically consist of a septic tank, disposal systems (deach field or seepage pit), and interconnecting piping between the IWS. The maintenance program will establish rules and regulations, monthly fees/ penalties, County monitoring and reporting, and IWS educational information for homeowners.
Ná'álehu LCC Closure EID / EA Consultation Package – March 2024 3	Ná'álehu LCC Closure EID / EA Consultation Package – March 2024

Alternative 4 - Individual Wastewater System-Operating Permit to Homeowners	alternative that will allow the County to close the LCCs, and provide a new, SDWA compliant
The April 2023 PER indicates for a Model 3 Alternative, the County is to:	solution for hardening wastewater generated by the valateru community. Cuosure of the LCCs will eliminate the disposal of untreated sewage into the subsurface which will serve the County's minimate common distribution cutors of the subsurface which will serve the County's
 Fund design and construction of the IWS systems; Administer an operating permit program for the IWS system to the homeowners; 	THISSION TO PROJECT UTION OF THISTING WATCH SOUTCES. 3. ENVIRONMENTAL CRITERIA TO BE ANALYZED
Issues maintenance notice to the homeowner.	The EID / EA will evaluate and compare the anticipated environmental impacts of each alternative
under Alternative 4, the nomeowners would be responsible for maintenance scheduling, contracting and paving for a service provider to conduct the necessary maintenance and/or	
responding to trouble calls, monitoring and record keeping of maintenance.	 Climate Physiography
Under Alternative 4, County is evaluating the possibility of either completing the project as a	o Topography
conventional Design/ Bid/ Build process, or under a voucher program, which the County would	o Soils
administer a voucher program which will grant homeowners with funding to hire Professional Engineer to design new TWS and hire Contractor to construct TWS. Horder the volucher program	Water Resources
homeowners will be responsible to hire and coordinate with Professional engineer for placement	Natural Hazards Cea Level Pice
of IWS and overall design of IWS.	 Flood and Tsunami Threat
2 1 BBOJECT OCATION & SETTING	 Hurricane and Wind Hazard
	 Seismic Hazard
The community of Nă'ālehu is located about 64 miles southwest of Hilo in the Ka'ū District.	 Volcanic Hazard
Māmalahoa Highway (State Route 11) runs through the community from East to West. The	
community of Nariaeth Is about 2.06 miles from the shoreline. Most of the community lies	 Clift rel Literation and Archarological Decisions
between approximately /40 reet above mean sea level.	 ditrollegity & Done ditrollegity & Done
Even though Ka'ū was one of the originally settled areas in the Hawaiian Islands, it remains a	
vast remote area. Only a fraction of a percent of the Karu District has been developed with	Energy and Natural Resources
residential propertices, and the remainder is largely used for adjustmentation purposes or is individually the Value Director and the Adjustmentation of the	Energy Sources
undevended. The said District Lovels about 222 square mixer stapt swith over 80 millse of virtuality undeveloped.	 Water Resources
with over optimized of microstructure documents, meany two mirus on the over and and access in the Concentration District The Kayin District consists of coveral communities including the Missishi	 Land Use
are conservator restored in the adjustment of approximately R10 persons according to the US Census community. Which has a population of approximately R10 persons according to the US Census	Roadways And Traffic
Bureau American Community Survey, 2020. The distance to the communities of Hilo and Kailua-	Hazardous Materials
Kona means that the Ka'ū District is relatively isolated from the major infrastructure systems	
found in those communities, including wastewater treatment and disposal facilities.	Human Health and Safety
2.2. PURPOSE AND NEED	4. CONSULTATION
DEM has outlined that the purpose and need for the Proposed Action is to comply with the	This Consultation Package has been circulated to the following parties:
requirements and mandates of the SDWA and AOC, and to ultimately close the three LCCs that	Enders Anamiae
serve Na ⁷ älehu in accordance with the Revised AOC. Thus, the core purpose of this exercise is to evaluate, gather community input, and make an informed decision in selecting an option or	US Department of Agriculture Natural Resources Conservation Service
	-
Na'álehu LCC Closure EID / EA Consultation Package – March 2024	Na'álehu LCC Closure EID / EA Consultation Package – March 2024
5	0.
US Environmental Froecom Agency US Fish and Wildlife Service National Oceanic and Atmospheric Administration US Department of Transportation Federal Aviation Administration	Elected Officials State Senator Dru Mamo Kanuha, Senate District 3 Representative Jeanne Kapela, House District 5 Councilmember Michelle Galimba, Council District 6
--	---
State Agencies Department of Agriculture Department of Accounting and General Services Department of Business, Economic Development & Tourism (DBED&T) DEFD&T I and Les Commission	Public Utilities Hawaiian Electric Company Hawaiian Telcom Spectrum Hawai'i Hawai'i Gas
DBLBAT can use commission DBEDAT Office of Planning and Sustainable Development DBEDAT State Energy Office Department of Hawaiian Home Lands (DHHL) DHHL – East Hawaii District Office Department of Health (DOH) DOH – Clean Water Branch	Other Parties Hawai'i State Library Nă'ălehu Public Library Hawaiian Civic Club of Ka'ū Ka'ŭ Colpo Action Committee
DUH - Hazard Evaluation and Emergency Response DUH - Hazard Evaluation and Emergency Response DUH - Safe Drinking Water Branch DOH - Wastewater Branch DOH - Wastewatewatewatewatewatewatewatewatewatewa	'O Ka'u Kākou
County of Hawaii Fire Department Police Department Planning Department Research and Development Department of Parks and Recreation Department of Water Supply Environmental Management Commission Office of the Corporation Commission Office of Housing and Community Development	
Nă fălehu LCC Closure	Nă falehu LCC Closure















ATTACHMENT C

100
3
-
÷.
5
10
Σ
T
10
Ξ.
Ľ.
Σ

comments are provided under the authorities of the Endangered Species Act of 1973 (ESA), as amended (16 U.S.C 1531 et seq.).

Based on information you provided and pertinent information in our files, including data compiled by the Hawaii Biodiversity and Mapping Project, there are eight (8) listed species that have the potential to either be in or fly through the vicinity of the project area: the federally endangered Hawaiian hoary but (*Lasiurus cinereus semotus*), Hawaiian hawk (*Buteo solitarius*), *Poten Bronta sondvicensis*, Hawaiian petrel (*Prevorma sandwichensis*), band-rumped stormpetrel (*Oceandorma caracio*), Hawaiian petrel (*Prevorma sandwichensis*), hawaiian coot (*Fulica alai*), and the threatened Newell's shearvater (*Pufiltna auricularis newell*).

Avoidance and Minimization Measures

Hawaiian hoary bat

The Hawaiian hoary bat roosts in both exotic and native woody vegetation across all islands and will leave young unattended in trees and shrubs when they forage. If trees or shrubs 15 feet or taller are cleared during the pupping season, there is a risk that young bats could inadvertently be harmed or killed since they are too young to fly or may not move away. Additionally, Hawaiian harmed to reare for insects from as thore feet to higher than 500 feet above the ground and can become entangled in barbed wire used for fencing.

To avoid and minimize impacts to the endangered Hawaiian hoary bat we recommend incorporating the following applicable measures into your project description:

- Do not disturb, remove, or trim woody plants greater than 15 feet tall during the bat birthing and pup rearing season (June 1 through September 15).
 - Do not use barbed wire for fencing.

Hawaiian hawk

The Hawaiian hawk is known to occur across a broad range of forest habitats throughout the Island of Hawaii. Loud, irregular and unpredictable activities, such as using heavy equipment or building a structure, near an endangered Hawaiian hawk nest may cause nest failure. Harassment of Hawaiian hawk nesting siles can alter feeding and breeding patterns or result in Harassment of chick abandonment. Nest disturbance can also increase exposure of chicks and juveniles to inclement weather or predators.

To avoid and minimize impacts to Hawaiian hawks we recommend you consider incorporating the following applicable measures into your project description:

- If work must be conducted during the March 1 through September 30 Hawaiian hawk breeding season, have a biologist familiar with the species conduct a nest search of the project footprint and surrounding areas immediately prior to the start of construction activities.
- Pre-disturbance surveys for Hawaiian hawks are only valid for 14 days. If disturbance for the specific location does not occur within 14 days of the survey, conduct another survey.

Mr. Earl Matsukawa

N

-

- No clearing of vegetation or construction activities within 1,600 feet of any active Hawaiian hawk nest during the breeding season until the young have fledged.
 Repardless of the time of vest no trimming or cutting trees containing a havk nest a
- Regardless of the time of year, no trimming or cutting trees containing a hawk nest, as nests may be re-used during consecutive breeding seasons.

Nana

Nene are found on the islands of Hawaii, Maui, Molokai, and Kauai predominately, with a small population on Oahu. They are observed in a variety of habitats, but prefer open areas, such as pastures, golf courses, weltands, natural grasslands and shrublands, and lava flows. Threats to the species include introduced mammalian and avian predators, wind facilities, and vehicle strikes.

To avoid and minimize potential project impacts to Nene we recommend incorporating the following applicable measures into your project description:

- Do not approach, feed, or disturb Nene.
- If Nene are observed loafing or foraging within the project area during the Nene breeding season (September through April), have a biologist familiar with the nesting behavior of Nene survey for nests in and around the project area prior to the resumption of any work. Repeat surveys after any subsequent delay of work of three or more days (during which the birds may attempt to nest).
 - O Cease all work immediately and contact the Service for further guidance if a nest is discovered within a radius of 150 feet of proposed work, or a previously undiscovered nest is found within said radius after work begins.
 - In areas where Nene are known to be present, post and implement reduced speed limits, and inform project personnel and contractors about the presence of endangered species on-site.

Hawaiian petrel, Band-rumped storm-petrel, and Newell's shearwater

Hawaiian scabirds may traverse the project area at night during the breeding, nesting and fledging seasons (March 1 to December 15). Outdoor lighting could result in scabird disorientation, fallout, and injury or mortality. Scabirds are attracted to lights and after circling the lights they may become exhausted and collide with nearby wires, buildings, or other structures or they may land on the ground. Downed scabirds are subject to increased mortality due to collision with automobiles, starvation, and predation by dogs, cats, and Obeenber 15, in their first flights from their mountain nests to the sea, are particularly vulnerable. To avoid and minimize potential project impacts to seabitds we recommend you incorporate the following applicable measures into your project description:

- Fully shield all outdoor lights so the bulb can only be seen from below bulb height and only use when necessary.
- Install automatic motion sensor switches and controls on all outdoor lights or turn off lights when human activity is not occurring in the lighted area.
- Avoid nightime construction during the seabird fledging period, September 15 through December 15.

Mr. Earl Matsukawa

Hawaiian stilt and Hawaiian coot

Listed Hawaiian waterbirds are found in fresh and brackish-water marshes and natural or manmade ponds. Hawaiian stilts may also be found wherever ephemeral or persistent standing water may occur. Threats to these species include non-native predators, habitat loss, and habitat degradation. Hawaiian ducks are also subject to threats from hybridization with introduced mallards.

Based on the project details provided, our information suggests that your project may result in standing water or the creation of open water, thus attracting Hawaiian waterbirds to the site. In particular, the Hawaiian sult is known to nest in sub-optimal holitat may suffer adverse if water is present. Hawaiian waterbirds attracted to sub-optimal habitat may suffer adverse impacts, such as predation and reduced reproductive success, and thus the project may create an attractive misance. Therefore, we recommend you work with our office during project planning so that we may assist you in developing measures to avoid impacts to listed species (e.g., fercing, vegetation control, predator management).

To avoid and minimize potential project impacts to Hawaiian waterbirds we recommend you incorporate the following applicable measures into your project description:

- In areas where waterbirds are known to be present, post and implement reduced speed limits, and inform project personnel and contractors about the presence of endangered species on-site.
- If watter resources are located within or adjacent to the project site, incorporate applicable best management practices regarding work in aquatic environments into the project
 - design.
 Have a biological monitor that is familiar with the species' biology conduct Hawaiian waterbird nest surveys where appropriate habitat occurs within the vicinity of the proposed project site prior to project initiation. Repeat surveys again within 3 days of project initiation and after any subsequent delay of work of 3 or more days (during which
 - the birds may attempt to nest). If a nest or active brood is found: o Contact the Service within 48 hours for further guidance.
- Establish and maintain a 100-foot buffer around all active nests and/or broods until the chicks/ducklings have fledged. Do not conduct potentially distruptive activities or habitat alteration within this buffer.

Have a biological monitor that is familiar with the species' biology present on the project site during all construction or earth moving activities until the chicks/ducklings fledge to ensure that Hawaiian waterbirds and nests are not adversely impacted.

Native plants

Where disturbed areas do not need to be maintained as an open area, the Service recommends restoring disturbed areas using native plants as appropriate for the location.

The following websites are good resources to use when choosing landscaping plants:

Pacific Island Ecosystems at Risk (www.hear.org/Pier/), Hawaii-Pacific Weed Risk Assessment (www.botany.hawaii.edu/faculty/daehler/wra/full_table.asp.html) and Global Compendium of Weeds (www.hear.org/gcw).

Mr. Earl Matsukawa

Ŧ

ŝ

Compliance with the Endangered Species Act

If this project will receive any federal funding, federal permits, or any federal authorization, it is that Federal agency's responsibility to consult with the Service pursuant to section 7 of the ESA on the listed species and designated critical habitat. The Service only conducts Section 7 consultations with the federal action agency or their designated representative. If there is no federal action agency involved, but the take of listed species cannot be fully avoided, the project forgonoment should apply for an incidental take permit from the Service by developing a Habitat Conservation Plan.

Thank you for participating with us in the protection of our endangered species. If you have any further questions or concerns regarding this consultation, please contact Eldridge Naboa, Fish and Wildlife Biologist, 808-284-0037, e-mail: <u>eldridge naboa@fivs.gov</u>. When referring to this project, please include this reference number: *01EP1F00-2019-7A-0388*.

Sincerely,

MICHELLE Digitally 19med by BOGARDUS Date 2019/215 1021/05-10201

Michelle Bogardus Island Team Leader Maui Nui and Hawaii Island

Mr. Earl Matsukawa

BIOSECURTY PROTOCOL – HAWAII ISLAND (JULY 2018)

The following biosecurity protocol (based on National Park Service, State of Hawaii, U.S. Fish and Wildlife, U.S. Geological Survey, and the DOI Office of Native Hawaiian Relations guidance) should be followed when operating on Hawaii Island to prevent the introduction of hamful invasive species including frogs, ands, weeds, and fungi into local <u>natural areas</u> (e.g., Hawaii Volcances National Park, Hakalau Forest National Wildlife Refuge, State of Hawaii "Natural Areas") and areas with <u>native habitat</u> (habitat that is primarily composed of native vegetation), other islands in Hawaiian archipelago, or the U.S. mainland. The protocol also includes suggestions for keeping field staff safe from octain invasive species.

All work vehicles, machinery, and equipment should be cleaned, inspected by its user, and found free of mud, dirt, debris and invasive species prior to entry into the natural areas or native habitat.

a. Vehicles, machinery, and equipment must be thoroughly pressure washed in a designated cleaning area and visibly free of mud, dirt, plant debris, insects, frogs (including frog eggs) and other vertebrate species such as rats, mice and non-vegetative debris. A hot water wash is preferred. Areas of particular concern include bumpers, grills, hood compartments, areas under the battery, wheel wells, undercarring, eabs, and truck beds (ruck beds with accumulated material (intentionally placed or fallen from trees) are prime sites for hitchhikens).

b. The interior and exterior of vehicles, machinery, and equipment must be free of rubbish and food. The interiors of vehicles and the cabs of machinery must be vacuumed clean. Floor mats shall be sanitized with a solution of >70% isopropyl alcohol or a freshly mixed 10% bleach solution.

c. Any machinery, vehicles, equipment, or other supplies found to be infested with ants (or other invasive species) must not enter natural areas or native habitat. Treatment is the responsibility of the equipment or vehicle owner and operator.

2. Little Fire Ants – All work vehicles, machinery, and equipment should be inspected for invasive ants prior to entering the natural areas or native habitat.
a. A visual inspection for little fire ants should be conducted prior to entry into natural areas or native habitat. b. Hygiene is paramount but even the cleanest vehicle can pick up a little fire ant. Place MaxForce Complete Brand Granular Insect Bait (1.0% Hydramethylnon; http://ittlefireants.com/Maxforce%20Complete.pdf) into refillable tamper resistant bait stations. An example of a commercially available refillable tamper resistant bait station is the <u>Ant Café</u> <u>Pro (https://www.antrefic.on/</u>). Place a bait station (or stations) in vehicle. Note larger vehicles, such as trucks, may require multiple stations. Monitor bait stations frequently (every week at a minimum) and replace bait as needed. If the station does not have a sticker to identify the contents, apply a sticker listing contents to the station.

c. Any machinery, vehicles, equipment, or other supplies found to be infested with ants (or other invasive species) must not enter natural areas or native habitat until it is sanitized and re-tested following a resting period. Infested vehicles must be sanitized following recommendations by the Hawaii Ant Lab (http://www.littlefireanis.com/) or other ant control expert and in accordance

Mr. Earl Matsukawa

9

with all State and Federal laws. Treatment is the responsibility of the equipment or vehicle owner. d. Gravel, building materials, br other equipment such as portable buildings should be baited using MaxForce Complete Brand Granular Insect Bait (1.0% Hydramethylnon; <u>http://littlefireants.com/Maxforce%20Complete.pdf</u>) or AmdroPro (0.73% Hydramethylnon; http://littlefireants.com/Amdro%20Pro.pdf) following label guidance. Storage areas that hold field tools, especially tents, tarps, and clothing should be baited using MaxForce Complete Brand Granular Insect Bait (1.0% Hydramethylnon; <u>http://littlefireants.com/Maxforce%c20Complete.pdf</u>) or AmdroPro (0.73% Hydramethylnon; http://littlefireants.com/Amdro%20Pro.pdf) following label guidance.

Base yards and staging areas inside and outside areas must be kept free of invasive species.

a. Base yards and staging areas should be inspected at least weekly for invasive species and any found invasive removed immediately. Pay particular attention to where vehicles are parked overnight, keeping areas within 10-meters of vehicles free of debris. Parking on pavement and not under trees, while not always practical is best.

b. Project vehicles or equipment stored outside of a base yard or staging area, such as a private residence, should be kept in a pest free area.

4. All cutting tools must be subitized to prevent the Rapid Ohia Death (ROD) fungus. a. Avoid wounding ohia trees and roots with mowers, chainsaws, weed eaters, and other tools. Cut only the minimum amount of trees and branches as approved for the project.

b. All cutting tools, including machetes, chainsaws, and loppers must be sanitized to remove visible dirt and other contaminants prior to entry into natural areas or areas with native habitat, and when moving to a new project area within the native habita area. Tools may be sanitized using a solution of >70% isopropyl alcohol or a freshly mixed 10% bleach solution. One minute after sanitizing, you may apply an oil based lubricant to chainsaw chains or other metallic parts to prevent corrosion.

c. Only dedicated tools and chainsaws should be used to sample known or suspected ROD infected trees.

d. Vehicles, machinery, and equipment must be cleaned as described in (1) above.

Imported firewood, logs, and ohia parts:
 Ohia firewood, ohia logs, and ohia parts should not be transported.

denominant and and an and an and the second second for a second for a second second second second second second

6. For individuals working in the field:

a. Before going into the field, visually inspect and clean your clothes, boots, pack, radio tharness, tools and other personal gear and equipment, for seeds, soil, plant parts, insects, and other debris. A small brush is handy for cleaning boots, equipment and gear. Soles of shoes should be sanitized using a solution of >70% isopropyl alcohol or a freshly mixed 10% bleach solution.

Mr. Earl Matsukawa

b. Immediately before leaving the field, visually inspect and clean your clothes, boots, pack, radio harness, tools, and other personnel gear and equipment, for seeds, soil, plant parts, inseets, and other debris. Soles of shoes should be sanitized using a solution of >70% isopropyl alcohol or a freshly mixed 10% bleach solution.

~

c. Little fire ants nest in trees. If you are under a tree and that tree is bumped or somehow stressed, the threat response of the ants is to fall from the leaves and sting the person under the tree. If you are subject to an ant attack, do not panic. The ants are extremely small but their stings are publical so make sure you remove all ants from your body and clothing. The stings are publically and that are itchy and painful, and can last for weeks. Treat stings as you would other insect stings. In some persons thing can produce life threatening reactions. Stocking antihistamine in the first aid kit is a reasonable precaution.

d. Rat Lungworm disease is caused by a parasite that can infect humans who consume raw or undercooked infected smalls or slugs or consume raw produce that contains a small infected small or slug. Infection is rare but can be serious. Symptoms can include severe headache, neck stiffness, low grade fever, nausea, and vomiting anywhere from 1-6 weeks after exposure. The disease is not spread person to person. Anyone who handles smalls or slugs should wear gloves and/or wash hands. Eating unwashed produce is discouraged.

ATTACHMENT D

the trict	A natural resources assessment for the Nā'ālehu WWTP, Ná'ālehu, Ka'u District Island of Hawai'i (TMK 9-5-005:001 por.)
101·J	June 24, 2021 DRAFT AECOS No. 1613
134	E. Guinther, D. Miranda, and R. David AECOS Inc. 45-939 Kamehameha Highway Suite 104 Kāné'ohe, Hawai'i 96744 Phone: (808) 234-7770 ~ Email: guinther@aecos.com
	Introduction
	The Hawai'i County Department of Environmental Management, Wastewater Division is proposing to construct a wastewater treatment and disposal system to treat sewage collected in Ná'álehu, Ka'u District, Island of Hawai'i (the "Project"; Figure 1). Construction of the treatment and disposal system will
	allow the closure of three large capacity cesspools currently used to dispose of untreated sewage. The treatment and disposal facility will be located on an approximately 28-ac (11.3-ha) portion of State Department of Land and Natural Resources (DLNR) land to the southwest of town (<i>makai</i> and downslope; see
	Figure 2). Site work includes grading of a drainage diversion swale to realign the existing site drainage, beginning from a vegetated diversion channel at a point some 350 ft (100 m) downslope of Māmalahoa Highway to a point downslope and east of the Project site. Road access and a utility corridor would follow from off the existing Năʿălehu Spur Road.
	A collector system will initially service residences on the north (<i>mauka</i>) side of the main street, the Hawaii Belt Road (Mamalahoa Highway). The Project also includes a force main pump station (SPS) and a force main connecting to the WWTP. The latter was surveyed specifically where the force main would cross a drainage channel. Otherwise, the force main route follows existing roads.
	AECOS Inc. was contracted by Wilson Okamoto Corporation to conduct a natural resources survey and assessment of the subject property (TMK 9-5-005:001
	AECOS Inc. [Report No. 1613] Page 1

A natural resources assessment for th Na'alehu WWTP, Na'alehu, Ka'u Distri Island of Hawai'i (TMK 9-5-005:001 pc



AECOS Inc. 45-939 Kamehameha Highway Suite 104 Kāne'ohe, Hawai'i 96744

June 24, 2021



por.) and prepare this report of findings.¹ We make reference herein to a previous survey of a nearby location proposed for the WWTP nearly a decade ago (David & Guinther, 2012). As in 2012, our survey included the sewerage collector system as presently proposed, the latter covered by driving all of the streets to be impacted to establish that no environmental resources of conservation interest occur in those areas.

Site Description

The Project site is low sloping ground covered by pasture, in places mostly of cropped grasses (Figure 3), in other places a mixture of grasses and low growing shrubs (Figure 4); along the southern boundary, the vegetation merges

¹ This document is produced for inclusion in an EA for the subject Project and will become part of the public record.





AECOS Inc. [Report: 1613-page 4]

Natural Resources Survey	NĂ'ĂLEHU WW'TP, KA'U	Natural Resources Survey NA'ALEHU WWTP, KA'U
		into scrub growth (cover photo). Much of the grassland is dominated by Guinea grass (<i>Megathyrsus maximus</i>). Low shrub growth areas are dominated by lantana (<i>Lantan camara</i>). Taller shrub/scrub growth is entirely <i>koa haole</i> (<i>Leaceana leucocephala</i>). The proposed pump station site ("Area No. 3") is on the <i>mulawa</i> is disclosed pump station site ("Area No. 3") is no the <i>mulawa</i> is disclosed pump station site ("Area No. 3") is no the <i>mulawa</i> is disclosed pump station state of "Öhal Road east of the intersection with Opukea Street. The force main runs west on Opukea Street to Melia Street then south on Ka'aï'äiki Rd and Na'aïlehu Spur Road to the proposed WWTP site. The pipe crosses a drainage channel on Melia Street ("Area No. 2").
		Methods Botanical Survey
Figure 3. Grassland pasture area of cropped (iuinea grass.	The natural resources survey was undertaken by the botany team on February 18, 2020 and again on November 5, 2020. A survey area map (corresponding to the red outline on Fig. 2) was loaded on a Trimble 6000 Series GNSS unit (Trimble GeoXH and GeoXT for use during the surveys). The GNSS units
ž		recorded the progress tracks of the botanists, providing real time feedback on location and adequacy of coverage during a wandering (pedestrian) transect. The November survey encompassed the site for a sewerage pump station east of 'Õhai Road, the force main crossing of a drainage channel off Melia Street, and an additional look along the southwest side of the WWTP site (Figure 2) that may not have been completely covered by the February visit owing to later adjustments in Project footprint.
		Plant species were identified as they were encountered. Species names follow <i>Manual of the Flowering Plants of Hawai'i</i> (Wagner, Herbst, & Sohmer, 1990; Wagner & Herbst, 1999) for native and naturalized flowering plants, <i>Hawai'i's Ferns and Fern Allies</i> (Palmer, 2003) for ferns, and <i>A Tropical Garden Flora</i> (Staples & Herbst, 2005) for ornamental plants. More recent name changes for naturalized plant species follow Imada (2019).
		Terrestrial Vertebrates Survey A survey of extant birds and mammals was conducted on the morning of February 18, 2020 and the morning of November 5, 2020. Only the results from
Figure 4. Pasture area of cropped grass and	low shrubs.	the November 5 survey are reported herein. Ten avian point-count stations were established covering the Project areas and a single eight-minute avian point-count made at each station. The avian counts were conducted in the morning hours between 0730 and 1030 am with the aid of Leica 8 X 42 binoculars and by listening for vocalizations. Weather conditions were ideal,
AECOS Inc. [Report No, 1613]	Page 5	AECOS Inc. [Report No. 1613] Page 6

Vatural Resources Survey Ni	А́А́LEHU WWTP, КА́ ^U	Natural Resources Survey NA'ALEHU WWTP, KA'
with unlimited visibility, light winds, and no rainfall. The a order and nomenclature used in this report follow the AOU <i>c</i> and Middle American Birds 2019 and the Sixtieth first Suppler	avian phylogenetic <i>Check-List of North</i> ment to the Check-	Results
list of North American Birds (Chesser et al., 2019, 2020)		Vegetation
The survey of mammals was limited to visual and auditory with visual observation of scat, tracks, and other animal s conduct a survey for Hawaiian hoary bat or <i>'õpe'ape'a</i> (<i>semotus</i>), the only native land mammal in the Islands. Bat	detection, coupled sign. We did not (<i>Lasiurus cinereus</i> ts are solitary and	Vegetation across the site consists entirely of grass pasture (predominant) Guinea grass or <i>Megathyrsus maximus</i>) and areas of mostly low scrub growth dominated by lantana. Trees are scattered around the site, especially along shallow swale feature that transects the area from north to southeast.
potentiary whete preduction in our manuary, and recection surveys and deployment of special detection equipment rect period of time. Negative results cannot confirm the absence of a location. Mammal scientific names follow <i>Mammal specia</i> <i>taxonomic and geographic reference</i> (Wilson and Reeder, 2005	m requires man ording over a long of the world: a 5).	The site ("Area No. 3" in Figure 2) of the proposed SPS is a former pasture dominated by tall Guinea grass (Figure 5, above). "Area No. 2" is an open aree on Melia Street with a foot bridge crossing a concrete-lined drainage channe and surrounded by private houses (Figure 6).
Figure 5. "Area 3" proposed pump station site overgrown with Guinea grass.		Figure 6. "Area 2" in our survey is a section of concrete-lined drainage channel at Melia Street.
AECOS1nc. [Report No, 1613]	Page 7	AECOS Inc. [Report No. 1613] Page

AECOS Inc. [Report No, 1613]

		z	IĀ'ĀLEHU	u WWTP, KA'u	Natural Resources Surv.	ey		N	Ā'ĀLEHU	WWTP, KA'U
					Table 1 (continued).					
Il the spec	cies of flowering plan	ts (angio	spern +	1s) observed	Species listed by famil	ily	Common name	Status	Abund	ance Notes
ity (91%) nts (4.5%) are naturalized or r) and one early Poly	ion-nativ nesian s	ve) sp(species	ecies. Three (3%). The	ASCLEPIADACEAE Asclepias physoca Schlachter	arpa (E. Mey.)	balloon plant	Nat	n	
(nnic) nii common	plants The early Po	yo zeyu Iynesiar	, (החווות n intro י	duction (so-	ASTERACEAE	SITAE) 2 hisnidum DC	-	Nat	ď	
s yellow v	vood sorrel (<i>ini'a</i> i or	UXalis c	cornicu	ilata), a very	Ageratum conyzo	nides L.	maile hohono	Nat	: 0	<3>
					Bidens pilosa L.		Spanish needle	Nat	;	<2, 3>
					Conyza bonariens Pluchea carolinen	sis Cronq asis G Don	hairy horseweed sourbush	Nat Nat	: :	<2, 3><3>
ant speci	es observed at the P	roject si	ites.		Senecto madagas	cariensis	fireweed	Nat	N	1
					L POIF. AMARANTHACEAE					
	Common name	Status	Abur	ndance Notes	Amaranthus spine	osus L.	spiny amaranth	Nat	Я	<2>
1014 1	WERING PLANTS				BKASNICACEAE Lepidum virginic CADDA DA CEAE	cum L.		Nat	ł	<2>
is I.	hairv honohono	Nat	В		CARTCACEAE CARTCACEAE	r L.	wild spider flower	Nat	Ŋ	<2>
	hanana	Pol	1	<3>			papaya	Nat	:	<3>
. A []	nitted heardorass		c			ırale L.	ʻaheahea	Nat	Я	
		Nat	Ua			(L.) Ker-Gawl.	1	Nat	0	
Pers.	swollen ingergrass Bermuda grass	Nat	-	<2, 3> <2, 3>	<i>Ipomea trilobata</i> CUCURBITACEAE	Ŀ	little bell	Nat	N	<2, 3>
Mez ex	sourgrass	Nat	0		Momordica chara	antia L.	balsam pear	Nat	1	<2, 3>
ertn. Miseland	wiregrass	Nat	0	<2, 3>	EURINDIALEAE Euphorbia cyatho	ophora J. A.	wild noinsettia	Nat	1	<3>
(MICUX.)	Carolina lovegrass	Nat	C	<2>	Murray	-		Mat	=	
(Jacq.)	Guinea grass	Nat	AA	<2, 3>	KICINUS COMMUNIS FABACEAE	S L	castor bean	Nat	5	^ ??
L. Jacobs Reroins	Hilo Grass	Nat	;	3	Clitoria ternatea	L.	butterfly pea	Nat	;	<3>
DCI BIUS	rattail grass	Nat	J	201	Crotalaria incanu	ım L.	fuzzy rattlepod	Nat	:	<2>
	EUDICOTS		I		Chamaecrista nici	titans L	partridge pea	Nat	ж	<2, 3>
					Desmodium incan	num DC.	Spanish clover	Nat	0 1	~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~~
	white shrimp plant	Nat	;	<2,3>	Inaigojera sujirut Leucaena leucoce	ricosa Mill. Phala (Lam.)	inaigo, <i>iniko</i>	Nat	ر	^ ?
loxb.	sweet clock vine	Nat	D		deWit	(mmr) mmrd	koa haole	Nat	C	<2,3>
		Not	П		Lotus subbiflorus	Lag		Nat	R	
s Raddi	Christmas berry	Nat	° O	<3>	Mimosa pudica L.	_	sensitive plant	Nat	ж	
										1
				Page 9	AECOS Inc. [Report No, 16]	13]				Page 10

AECOS Inc. [Report No, 1613]

Flora

Natural Resources Survey

Table 1 is a listing of all t during the surveys with areas. The vast majority native indigenous plants native species are *'llima* (*Waltheria indica*); all co called "canoe plant") is y common small weed.

Table 1. Pla

Species listed by family

ACANTHACEAE Justica betonica L. Thunbergia fragrans Roxt ANACARDIACEAE Mangifera indica L. Schinus terebinthefolius R Megathyrsus maximus (Ja B.K. Simon & W.L. J Paspalum conjugatum B Sporobolus indicus Musa x paradisiaca L POACEAE Bothriochloa pertusa (L.) Eleusine indica (L.) Gaer Eragrostis pectinacea (N Nees Commelina benghalens Cynodon dactylon (L.) Digiteria insularis (L.) Chloris barbata L Ekman COMMELINACEAE Camus MUSACEAE

AECOS Inc. [Report No, 1613]

Page | 10

Natural Resources Survey		z	₫,ĀLEHU	WWTP,	<u> (, , y)</u>	Natural Resources Survey		N/	й'ālehu WWTP, Kaʻl	٦
Table 1 (continued).						Table 1 (continued).				
Species listed by family	Common name	Status	Abune	lance N	btes	Species listed by family	Common name	Status	Abundance Notes	s
FABACEAE (cont.) <i>Neonotonia wightii</i> (Wight & Arnott) Lackey	glycine vine	Nat	U	ŝ	I	SOLANACEAE Solanum linnaeanum Hepper & P. Jaeger	Apple of Sodom	Nat	В	1
Prosopis pallida (Humb. & Bonpl. ex Willd.) Kunth Comma accidentatie (1-) 1 inb	kiawe	Nat	0 0			TILIACEAE Triumfetta rhomboidea Jacq.	1	Nat	C <3>	
Senna successions (L.) LILIN Senna surattensis (N.L. Burm.) H. Irwin & Barneby I AMIACEAE	conce semina scrambled egg plant	Nat	0 0			VERBENACEAE Lantana camara L. Stachytarpheta urticifolia	lantana 	Nat Nat	A <3>	
<i>Hyptis pectinata</i> (L.) Poit. <i>Leonotis nenetifolia</i> (L.) R Br	comb hyptis lion's ear	Nat Nat	0 0	<2, 3> <2>		Verbena litoralis Kunth	ōwi	Nat	R	
MALVACEAE Abutilon grandifolium (Willd.)	hairy abutilon	Nat	0	\$ \$		L STATUS = distributional status I Ind = indigenous, nati	legend to Table 1 for the Hawaiian Islands: ive to Hawaii, but not unique	e to the Har	waiian Islands.	
Sida fallax Walp	ʻilima	Ind	0 0	ć		Nat = naturalized, ex arrival of Coo	otic, plant introduced to t k Expedition in 1778, ar	the Hawaii nd well-es	an Islands since the tablished outside o	of
Sida spinosa L. Sida spinosa L.	 prickly sida	Nat	o z	<7>		cultivation. Orn = A cultivated pla	nt; a species not thought to	be natural	ized (spreading on its	ts
Urena lobata L. ^{Malthoria} indica l	the loss	Nat Ind	۲ N			own) in Hawai'i ABUNDANCE = occurrence ratii	i. ngs for plant species:			
MELIACEAE			، ۲			R – Rare U - Uncommon	seen in only one or perhap seen at most in several loca	s two locat ations.	ions.	
mena azeaaracn L. MYRTACEAE	Chinabery	Nat	×			U - Uccasional C - Common	seen with some regularity. observed numerous times	during the	survey.	
<i>Syzygium cumini</i> (L.) Skeels NYCTAGINACEAE	Java plum	Nat	D			A - Abundant NOTES: Not observed in <2> - Noted presen	found in large numbers; m: 1 Area No. 1. 11 in Area No. 2.	ay be locall	y dominant.	
<i>Boerhavia coccinea</i> Mill. OXALIDACEAE		Nat	ł	33		<3> - Noted present<4> - Plant lacking	nt in Area No. 3. key diagnostic characteristi	ics (flower,	fruit);	
Oxalis corniculata L. PLUMBAGINACEAE	yellow wood sorrel	Pol	К			identification	ı, therefore, uncertain.			1
Plumbago zeylanica POLYGONACEAE	ilie'e	Ind	n							
Antigonon leptopus Hook. & Arnott PORTULACEAE	Mexican creeper	Nat	:	<2>		Avian Fauna				
<i>Portulaca pilosa</i> L. PROTEACEAE	1	Nat	К	<2, 3>		A total of 239 individual birds	s of 15 species, represe	nting 10	separate families	s,
<i>Grevillea robusta</i> A. Cunn. ex R.Br.	silk oak	Nat	n			were recorded during point Plover (<i>Pluvialis fulva</i>), is an remaining 13 species recorde	counts. One species n indigenous migrator d are all established al	recorde y shore	d, Pacific Golden bird species. The ral snecies (Table	- e -
KUTACEAE <i>Murraya paniculata</i> W. Jack	mock orange	Nat	I	<2>		2).	ע מור מוו כסומטווסוורע מו			د
AECOS Inc. [Report No. 1613]				Page	11	AECOS Inc. [Report No, 1613]			Page 12	2

Table 2. Avian species detected at Nāʿālehu WWTP site		HU WWTP, KA
On November 5, 2020.	Table 2 (continued). Common Name Scientific Name	<i>ST</i> RA
me Scientific Name St	THRAUPIDAE - Tanagers Thraupinae - Core Tanagers	
PHASIANIDAE - Pheasants & Partridges	Sicolis flaveola Key to Table 2	A 0.50
Gallus gallus A 1.90 A 1.90	57 Status	
COLUMBIFORMES COLUMBIDAE - Pigeons & Doves	 A Alien – Introduced to the Hawaiian Islands by humans IM Indigenous migratory- Native migratory shorebirds species which breeds in the Mathematic structure of birds detected divided by the number of point 	e high Arctic it counts (10)
Streptopelia chinensis A 1.70 Geopelia striata A 2.50		
CHARADRIIFORMES CHARADRIIDAE - Lapwings & Plovers CHARADRIIDAE - Lapwings & Plovers	Avian diversity and densities were in keeping with the location of	the three site
ver Pluvialis fulva poccepterpontes	and the vegetation present. Three introduced species— G (<i>Acridotheres tristis</i>), House Finch (<i>Haemorhous mexicatus</i>), an (<i>Geopilia striata</i>)—accounted for 44% of the total number of b	mmon Myr. 1 Zebra Dov rds recordeo
ALAUDIDAE - Larks	Common Myna was the most commonly tallied species, which acc percent of the birds recorded during noint counts.	ounted for 2
Alauda arvensis A 1.40		
e Zosterops japonicus A 1.20	Mammals	
STURNIDAE - Starlings Acridotheres tristis A 5.20	Terrestrial mammalian species detected along with the method are presented in Table 3.	s of detectio
ESTRILDIDAE - Estrildid Finches Estrilda actrild		
Euodice contans A 0.60	Tahle 3 Mammalian eneries deterted at Na'alehu WWTD ei	te in 2020
Lonchura oryzivora A 0.70		
inia Lonchura punctulata A 1.10 pASSEBIDAE - Old Morld Snarrowe	Concession Marsea Calantific Marsea C	T
Passer domesticus A 0.60 A 0.60	Contrigent realize Carnivoraria	ו המומרו
FRINGILLIDAE - Fringiline and Carduline Finches & Allies	Canidae - Wolves, Jackals & Allies	
Carduelinae - Carduline Finches and Hawaiian	Domestic dog Canis lupus familiaris	V, A, S
Honeycreepers	Viverridae - Civets & Allies	:
Haemornous mexicanus A 2.70 narv Ceithaara mozambira A 1.60	Small Indian mongoose Herpestes javanicus Ealidaa. Pats	A V, A
	House cat <i>Felis cotus</i> renuaer- Lats	>
t No. 1613] Page 13	AECOS [nc. [Report No. 1613]	Page 1

latural Resources Survey	NĀ'Ā	ALEHU WV	VTP, KA'U	Natural Resources Survey NATURENU WV	/TP, KA'U
Table 3 (continued).				Although a much smaller area, the latter site is both highly d	sturbed
Common Name	Scientific Name	ST	Detection	(Increasing the count of futured as species) and subject to frequent intro of species on vehicles and rubbish loads.	rucuons
ATRIODACT	YLA - EVEN-TOED UNGULATES idae - Old World Swine			Faunal Resources	
Pig Sus scrofa		A	V, Sc, Tr, Si		
Bovidae-	- Hollow-horned Ruminants	<	V A Tr Sc	Insects	
		¢	V, A, II, JC	Several insects are now listed as endangered in the Hawaiian Island	s: seven
K	ey to Table 3			species of the yellow-faced bee (<i>Hylaeus anthracinus, H. assimulans, H.</i>	acilis, H.
67 Status				hilaris, H. kuakea, H. longiceps, and H. mana; USFWS, 2016) and Bla	ckburn's
A Alien – Introduced to the Hawaiian Isla	ands by hitmans			sphinx moth (<i>Manduca blackburnii</i> ; USFWS, 2000, 2005).	
V Visual – the animal was seen				-	-
Detection type				No yellow-faced bee species was observed during the survey and no	otential
A Audio – the anima: was heard				habitat or food sources were noted. The caterpillar of the sphinx mo	th feeds
Sc Scat – feces of the anima: were observ	ved			exclusively on plants in the Family Solanaceae. In particular, where the	moun IS
Tr Tracks – tracks of the animals were se				iouitu, cater pinats are most often associated with the wheely distribut mative tree tohacco mlant (<i>Nicotiana alauca</i>). The anky mlant of	cu, 11011- hearwad
Si Sign – browsing's signs, beds, rooting,	, dust baths etc. were observed			representing the Family Solanaceae was a single wild cherry toma	to nlant
)))				(Solanum Accobersicum var. cerasiforme). This species is not know	n to be
				utilized by the caterpillar. We would deem any threat to these insec	species
				due to Project activities as non-existent.	
Discussion and Recommend	lations			Terrestrial Birds	
Recommendations are partly ba	tsed on U.S. Fish and Wildlife	Service	, Animal	The findings of the avian survey are consistent with the vegetation pr	sent on
Avoidance and Minimization Mea	asures (USFWS-PIFWO, nd) Im	iplemen	tation of	the three sites. The WWTP site is mostly pasture that has been bull- where The other two complements is to be the being in the second second second second second second second second	tozed in
ule recommendations (provide contractor will minimize impact	the period as builded refined in the mass of the mass	oy uie naximiir	rrujeci n extent	places. The other two shirallet sites, located in town, have very lithin sheries dominated by alien species. Fourteen of the recorded hird spi	cu piant cies are
practicable.				alien, naturalized bird species. The only native bird, the Pacific Golden-I	lover, is
				an indigenous migratory shorebird commonly seen across the state	oetween
Floral Resources				late September and the end of April each year. This species migrates	o Arctic
				breeding grounds in late April or the first week of May.	
No plants of conservation conce	ern or enjoying statutory prof	tection	(that is,	It is nossible that the endangered Hawaiian Petrel (<i>Puffinus sand</i>)	ichesis).
the curvey and given the highly	eu; HULNK, 1998; USFWS, ZUZU V districted matrice of the site	u) were	noted In 1 not be	Band-rumped Storm-Petrel (<i>Hydrobates castro</i>), and the threatened	Newell's
ure survey and given die mgur. ovnoeted to ho anomina thene E	y unstantional mature of the site	currant.		Shearwater (Puffinus newelli) over-fly the Project area between April	and the
expected to be growing unere. F 66 species with 4 species as	ou comparative purposes, uns : s either indigenous or of es	arly Do	levealeu Ivnesian	middle of December each year in very small numbers A primary	cause of
introduction. The previous site	e	any io ang Mā	n malahoa	mortality in these listed seabirds in Hawai'i is thought to be predation	by alien
Highway 1 mi (16 km) east	falso nastiireland)—vielded	30 sn	eries of	mammalian species at inland nesting colonies (USFWS, 1983; Sim	ons and
introduced plants and 1 specie	es each of indigenous and es	arlv Po	lynesian	Hodges, 1998; Ainley et al., 2001). Collision with man-made structure of the structure of t	tures is
introduction. A survey of the mu	ich smaller Wai'öhinu Transfer	Station	(AECOS.	considered to be second-most as a cause of mortality; nocurnal conhirde conversibily flodelings on their unity to see in the symmetr and	y nying
2020) in Wai'ōhinu, 2 mi (3.4 l	km) to the west produced a l	list of	59 taxa.	seabirus, especially lieugings on their way to sea in the summer and	Iall, can
4 <i>ECOS</i> Inc. [Report No, 1613]			Page 15	AECOS Inc. [Report No, 1613]	age 16

AECOS Inc. [Report No, 16]

Natural Resources Survey N	AʿĀLEHU WWTP, Kaʻu	Natural Resources Survey NA'ÁLEHU WWTP, KA'U
become disoriented by exterior lighting. Disoriented seabird man-made structures and, if not killed outright, become opportunity for feral mammals (Hadley, 1961; Telfer, 1979; Si et al., 1985; Telfer et al., 1987; Cooper and Day, 1998; Pod Ainley et al., 2001; Hue et al., 2001; Day et al., 2003).	ls may collide with e easy targets of incock, 1981; Reed iolsky et al., 1998;	Project area on a seasonal basis. The potential impact to bats would be during the clearing and grubbing phase of the construction. The trimming or removal of foliage and/or trees within the construction areas may temporarily displace an individual bat utilizing the vegetation as a roost. However, a bat will use multiple roosts within their home territory, so the potential disturbance
The potential impact that the Project poses to protected increased threat that birds will be downed during the seabil (from September 15 through December 15) after becomin lights associated with the Project. The two main types of outdd a threat to nocturnally flying seabirds are: 1) night-time const	d seabirds is the rd fledging season ng disoriented by oor lighting posing rruction lights, and	resulting from the removal of the vegetation is likely to be minimal. During the pupping season, female carrying pups may be less able to rapidly vacate a roost site while vegetation is cleared. Additionally, adult female bats sometimes leave their pups in the roost tree while they themselves forage, and very small pups may be unable to flee a tree that is being felled.
 2) exterior lighting around Project infrastructure. Fully shield all outdoor lights so the bulb can only be bulb height and only use when necessary; install 	e seen from below automatic motion	• To avoid potential deleterious impacts to roosting bats with pups, it is recommended that no woody vegetation taller than 4.6 m (15 ft) be removed during the bat pupping season between June 1 and September 15 (USFWS, nd-c).
off lights when human activity is not occurring in th external lighting structures should be fully "darl (HDLNR-DOFAW, 2016).	e lighted area. All k sky compliant"	• The use of barbed wire to top fence lines may entangle flying bats and should be avoided (Zimpfer and Bonaccorso, 2010).
No habitats suitable for waterbird species are present on any Once the WWTP is constructed it is possible that one or mc attracted to ponds associated with the WWTP. No species o	✓ of the three sites. ore species will be currently listed or	Other Resources of Potential Concern <i>Critical Habitat</i>
proposed for listing as threatened or endangered under endangered species statutes is associated with the project si USFWS, nd-a). Mammalian Resources	· federal or state ite (HDLNR, 2015;	Federally delineated Critical Habitat is not present in the Project area (USFWS, nd-b). No equivalent designation exists under state law. Conservation zoning in Hawai'i is promulgated at the state level by state Conservation Districts. No Conservation Districts occur near the Project.
The five mammalian species recorded during these survey	s are all common	Jurisdictional Waters
introduced species. Although no rodents were recorded, it is more of the other four established alien Muridae found on H house mouse (<i>Mus musculus</i>), roof rat (<i>Rattus rattus</i>), b <i>norvegicus</i>), and black rat (<i>Rattus exulans hawaiiensis</i>)—use found within the general project area on a seasonal basis. It small Indian mongoose (<i>Herpestes javanicus</i>) use resource Project area. These human commensal species are drawn thabitation and activity. All of these introduced mammideleterious to native ecosystems and the native faunal spec	i likely that one or lawa'i'-European rown rat (<i>Rattus</i> various resources : is also likely that es in the general to areas of human alian species are cies dependent on	No streams or wetlands occur in the Project area, although the WWTP site itself is located within a shallow drainage swale and runoff flow will be directed around the WWTP infrastructure in a new channel. We observed no indications of water flow in the swale at the Project site. A concrete drainage channel crossed by the proposed force main on Melia Street is similar in directing runoff through town into shallow swale further downslope. Neither of these drainageways direct flows to streams or outlets on the ocean shore. Annual rainfall in the Na ² alehu area is on the order of 40 in (1020 mm; Giambelluca et al. 2013) and the brouse scholevis in conducive to stream formation.
them. No Hawaiian hoary bats or were detected during the course o within the realm of possibility that this species uses res	of our survey. It is ources within the	
AECOS'Inc. [Report No, 1613]	Page 17	AECOS Inc. [Report No, 1613] Page 18

Natural Resources Survey NA'ALEHU WWTP, J	VU Natural Resources Survey	NA'ALEHU WWTP, KA'U
References Cited	Hadley, T. H. 1961. Shearwat	er calamity on Kauai. ' <i>Elepaio</i> , 21: 60.
AECOS Inc. (AECOS). 2020. A natural resources assessment for improvement the Wai'ōhinu Transfer Station, Wai'ôhinu, Ka'u, Hawai'i (TMK 9 005:001 por.). Prep. for Wilson Okamoto Corporation. AECOS No. 16 14 pp.	to 5- Hawaii Department of Land Wildlife, Endangeree Introduced Wild Birds 8: of Hawaii, Administr March 02, 1998.	and Natural Resources (HDLNR). 1998. Indigenous 1 And Threatened Wildlife And Plants, And 5. Department of Land and Natural Resources. State ative Rule §13-134-1 through §13-134-10, dated
Ainley, D. G. R. Podolsky, L. Deforest, G. Spencer, and N. Nur. 2001. The Sta and Population Trends of the Newell's Shearwater on Kaua'i: Insig from Modeling, in: Scott, J. M, S. Conant, and C. Van Riper III (edit <i>Evolution, Ecology, Conservation, and Management of Hawaiian Bird</i> <i>Vanishing Avifauna. Studies in Avian Biology No.</i> 22. Coop Ornithological Society, Allen Press, Lawrence, Kansas. Pp. 108-123.	us ts 2015. Hawai'i Admini ts Natural Resources, S Chapter 124, Indigen A Injurious Wildlife, It -s	strative Rules, Title 13, Department of Land and ubtitle 5 Forestry and Wildlife, Part 2 Wildlife, ous Wildlife, Endangered and Threatened Wildlife, htroduced Wild Birds, and Introduced Wildlife. 5 pp.
Chesser, R. T., S. M. Billerman, K. J. Burns, C. Cicero, J. L. Dunn, A. W. Fratter, Lovette, N. A. Mason, P. C. Rassmusen, and K. Winker. 2019. Sixt Supplement to the Check-list of North American Birds. <i>The Ornithological Advances</i> , 136: 1-23.	J. Hawai'i Department of Land th Wildlife (HDLNR-DOI URL: http://dlnr.hawaii kebruary 21, 2020.	l and Natural Resources-Division of Forestry and AW). 2016. Wildlife Lighting. PDF available at .gov/wildlife/files/2016/03/DOC439.pdf; last retrieved
D. F. Stotz, and K. Winker. 2020. Sixty-first Supplement to the Check of North American Birds. American Ornithological Society. Avail online at URL: http://checklist.aou.org/taxa; last retrieved July 31, 2020.	r., Hue, D., C. Glidden, J. Lippert. St Use and Limiting Fa Petrels on Mauna Loa, III (editors) <i>Evoluti</i> <i>Hawaiian Birds: A Va</i>	L. Schnell, J. Maclvor and J. Meisler. 2001. Habitat ctors in a Population of Hawaiian Dark-rumped Hawai'i, <i>in</i> : Scott, J. M. S. Conant, and C. Van Riper on, <i>Ecology, Conservation, and Management of</i> <i>nishing Avifauna</i> . Studies in Avian Biology No. 22.
David, R. E., and E. B. Guinther. 2012. Biological Surveys Conducted for Natalehu Wastewater Treatment Plant Project, Ka'ŭ District, Island Hawai'i, Rana Biolovical Consulting, Inc. 18 nn.	he Cooper's Ornithologic of 242).	al Society, Allen Press, Lawrence, Kansas (Pg. 234-
Cooper, B. A. and R. H. Day. 1998. Summer Behavior and Mortality of D.	Imada, C. T. 2019. Hawaiiar k- 2019 update). <i>Bishop</i>	ı Naturalized Vascular Plants Checklist (February Museum Tech. Rept. 69. 209 pp.
rumpeu reucis anu newens sheatwatets at rower bines on na Colonial Waterbirds, 21 (1): 11-19.	Podolsky, R., D. G. Ainley, G. of Newell's Shearwate	Spencer, L. de Forest, and N. Nur. 1998. Mortality ers Caused by Collisions with Urban Structures on
Day, R. H., B. Cooper, and T. C. Telfer. 2003. Decline of Townsend's (New. Shearwaters (<i>Puffinus auricularis newelli</i>) on Kauai, Hawaii. <i>The J</i> 120: 669-679.	l's kaua I. <i>colonul Water</i> <i>k</i> , Reed, J. R., J. L Sincock, and J Procellariform Birde:	<i>birds</i> , z1: 20-34. I. P. Hailman 1985. Light Attraction in Endangered Reduction by Shielding Unward Radiation <i>The Auk</i>
Giambelluca, T. W., Q. Chen, A.G. Frazier, J.P. Price, YL. Chen, PS. Chu, Eisabeid and D.M. Polanes, 2013, Online Printed Mathematica	K. 102: 377-383.	
Liscretd, and D.M. Detpare. 2013: Unline Kannall Adas of Hawal 1. <i>Amer. Meteor. Soc.</i> 94: 313-316, doi: 10.1175/BAMS-D-11-0022 Online at URL: http://rainfall.geography.hawaii.edu/; last accessed on J 24, 2021.	11. Simons, T. R., and C. N. I 1. <i>phaeopygia</i>). <i>In</i> : A. I America, No. 345, Th and the American Orn	Hodges. 1998. Dark-rumped Petrel (<i>Pterodroma</i> 200le and F. Gill (editors). The Birds of North e Academy of Natural Sciences, Philadelphia, PA. ithologists Union, Washington, D.C.
AECOS Inc. [Report No. 1613] Page	PBCOS Inc. [Report No. 1613]	Page 20

NAʿALEHU WWTP, KAʿU NAʾALEHU WWTP, KAʿU	ring the Newell's Shearwater. Pages 76-78 in Proceedings Forestry and Wildllife Conference, 2-4 October 1980. Land and Natural Resources, State of Hawai'i. Honolulu. 183. University of Hawai'i Press. 1853 pp.	. Herbst. 2005. A Tropical Garden Flora. Plants Cultivated and 1999. Supplement to the Manual of the flowering plants of Hawaii. Planest, and S. H. Sohmer, Hawaii, pp. 1855-1918. In: Wagner, W. L., D. R. Herbst, and S. H. Sohmer, pp. 1812. In: Wagner, W. L., D. R. Herbst, and S. H. Sohmer, pp. 1813. In: Vagner, Manual of the flowering plants of Hawaii. Revised edition 2 vols. University of Hawaii Press and B.P. Bishop Museum.	ccessful Newell's Shearwater Salvage on Kauai. ' <i>Elepaio</i> , Wilson, D. E. and D. M. Reeder (eds.). 2005. Wilson & Reeder's Mammal Species	or the World Unite Edution). Available online at UKL: http://www.departments.bucknell.edu/biology/resources/msw3/browse.asp; last 11, 15: 406-413.	ervice (USFWS). 1983. Hawaiian Dark-Rumped Petrel & Shearwater Recovery Plan. USFWS, Portland, Oregon.	red and Threatened Wildlife and Plants: Determination of atus for Blackburn's Sphinx Moth from the Hawaiian <i>Register</i> , 65 (21; February 1, 2000): 4770-4779.	very Plan for Blackburn's Sphinx Moth (Manduca gust 26, 2005.	17. Final Rule: Endangered and Threatened Wildlife and ered Status for 49 Species from the Hawaiian Islands , 81 (190; Friday, September 30, 2016): 67786-67860.	a). USFWS Endangered Species. Available online at URL: gov/endangered/; last retrieved October 30, 2020.	o). Critical Habitat Portal. Available online at URL: jov/ecp/report/table/critical-habitat.html; last retrieved May	ervice-Pacific Islands Fish and Wildlife Office (USFWS- te (nd). Avoidance and Minimization Measures. Available https://ww.fws.gov/pacificislands/articles.cfm? id=149489720; Jv 31. 2020.
atural Resources Survey	Sincock, J. L. 1981. Saving the Newell' of the Hawaii Forestry and Department of Land and Natur	Staples, G. W. and D. R. Herbst. 2005. <i>in the Hawaiian Islands and</i> Honolulu. 908 pp.	Telfer, T. C. 1979. Successful Newell 39:71	J. L. Sincock, G. V. Byrd, and seabirds to lights: Conservat <i>Wildlife Soc. Bull.</i> , 15: 406-413.	U.S. Fish & Wildlife Service (USFWS) Newell's Manx Shearwater R February 1983.	2000. Endangered and Threatt Endangered Status for Black Islands, <i>Federal Register</i> , 65 (2	2005. Recovery Plan for blackburni). August 26, 2005.	2016. 50 CFR 17 Final Rule: Plants: Endangered Status foi Federal Register, 81 (190; Frid-	Undated (nd-a). USFWS End. https://www.fws.gov/endangered/	Undated (nd-b) Critical Ha https://ecos.fws.gov/ecp/report/ta 20, 2020	U.S. Fish & Wildlife Service-Pacific I PIFWO). No date (nd). Avoida online at URL: https://www.fws Jast Partierood Infu 21 2020

AECOS Inc. [Report No, 1613]



AECOS, Inc. 45-939 Kamehameha Highway, Suite 10-

45-939 Kamehameha Highway, Suite 104 • Kaneohe HI 96744 • Telephone: (808)234-7770 • • Email: guinther@aecos.com •

December 2, 2020

John Sakaguchi Wilson Okamoto Corp. 1907 So. Beretania Street, Suite 400 **Honolulu, HI 96826**

RE: Natural Resources Assessment for the Na'alehu Large Capacity Cesspools Replacement Project

Mr. Sakaguchi,

AECOS, Inc. biologists, Eric Guinther, David Miranda, and Reginald David conducted an initial assessment survey of the proposed site for the Na'alehu WWTP on February 18, 2020. We prepared an internal draft report. However, changes to the project layout of the site and additional offsite appurtenances (a sever pump station and force main) required that we undertake additional surveys for the project. These were completed by Reginald David and David Miranda on November 5, 2020. Our surveys (see Figure 1, attached) covered the proposed WTP site (Area No. 1), including the proposed drainage feature adjacent, the SPS site off the east end of Optika Street (Area No. 3), the service area neighborhood (outlined partly in red), and the location where Re force main will cross a concrete-lined channel at the west end of Melia Street over to Kaalaki Road (Area No. 2).

The vegetation is dominated almost to the exclusion of native species on all three sites. Avian resources were sparse on the WWTP site as the habitat is very disturbed. All of the proposed WWTP site is pasture land, in various states of use, covered by grasses and scrub growth. Scattered trees are present, mostly along the shallow drainage swale. The proposed location for the SPS is a pasture with tall Guinea grass and minimal other weedy species. No resources are present at any of the sites with respect to the endangered Blackburn's Sphinx moth. All but one avian species detected during the surveys are introduced, established alien species. The sole and use species regulden Plover (*Pluvialls fulva*)—is an indigenous migratory shorehird species regulden Plover to resources/environments present—with the exception of the stars stars stares share no aquatic resources/environments present—with the exception of the

ATTACHMENT E

crossing of the concrete drainage channel (Area No. 2)—and are not within federal Critical Habitat or a State of Hawai'i Conservation Zone.

SBC,

Eric Guinther AECOS Inc.



Figure 1. Specific areas surveyed in two visits to Na'alehu.

APPENDIX D:

Section 7 Consultation

U.S. Fish and Wildlife Service Response



United States Department of the Interior

FISH AND WILDLIFE SERVICE Pacific Islands Fish and Wildlife Office 300 Ala Moana Boulevard, Room 3-122 Honolulu, Hawaiʻi 96850



June 21, 2024

In Reply Refer To: 2024-0078740-S7-001

Chane Hayashida State of Hawai'i Department of Health, Wastewater Branch 2827 Waimano Home Road, Room 207 Pearl City, Hawai'i 96782

Subject: 2024-0078740-S7-001 Nāʿālehu Wastewater Collection, Treatment, Disposal, and Drainage System Modifications, Nāʿālehu, Hawaiʿi

Dear Chane Hayashida,

This letter is in response to your May 31, 2024, request for our concurrence with your determination that the proposed wastewater system modification actions in Nā'ālehu, Hawai'i, "may affect, but is not likely to adversely affect" federally listed species. Specifically, you requested informal consultation pursuant to section 7(a)(2) of the Endangered Species Act of 1973, as amended (16 U.S.C. 1531 *et seq.*) (ESA) for the following federally listed species:

- 'Ōpe'ape'a or Hawaiian hoary bat (*Lasiurus cinereus semotus*)
- Hawaiian seabirds, including Hawai'i Distinct Population Segment of the 'akē'akē or band-rumped storm-petrel (*Hydrobates castro*), 'a'o or Newell's shearwater (*Puffinus newelli*), 'ua'u or Hawaiian petrel (*Pterodroma sandwichensis*)

The U.S. Environmental Protection Agency (EPA) has designated the State of Hawai'i Department of Health (HDOH) as its non-federal representative for certain projects funded under the Clean Water State Revolving Fund (CWSRF) program. The County of Hawai'i Department of Environmental Management is planning to undertake the Nā'ālehu Wastewater Collection, Treatment, Disposal, and Drainage System Modifications Project. There is no designated critical habitat within the project area.

PACIFIC REGION 1

Idaho, Oregon*, Washington, American Sāmoa, Guam, Hawai'i, Northern Mariana Islands *partial The determination of "no effect" to other species, rests with the action agency. The U.S. Fish and Wildlife Service (Service) has no regulatory or statutory authority for concurring with "no effect" determinations, and no consultation with the Service is required. We recommend that the action agency document their analysis on effects to listed species and maintain that documentation as part of the project file.

The Service requests HDOH also consider impacts to avian species protected under the Migratory Bird Treaty Act (MBTA) of 1918, as amended (16 U.S.C. § 703–712) and implement avoidance and minimization measures. A list of birds protected under the MBTA implementing regulations is provided at 50 CFR § 10.13.

The Service also requests the HDOH continue to coordinate with us prior to construction activities that may create temporary water sources and/or, any exposed wastewater and water containment, settling, or drainage facilities, such as ponds, pools or other capture structures. Endangered Hawaiian waterbirds (ae'o, Hawaiian stilt, *Himantopus mexicanus knudseni*; 'alae ke'oke'o, Hawaiian coot, *Fulica alai*; and koloa maoli, Hawaiian duck, *Anas wyvilliana*) can be attracted to such reservoirs or open sources of water during construction, operation, and maintenance of the new facility.

The proposed action area and project activities may also attract nēnē or Hawaiian goose (*Branta sandvicensis*). Nēnē may be observed in a variety of habitats, but prefer open areas, such as pastures, golf courses, wetlands, natural grasslands and shrublands, and lava flows. Nēnē are vulnerable to vehicular strikes, human presence, and activities as they can result in injury or mortality associated with crushing nests hidden in vegetation, vehicular strikes, and human-caused disturbance, such as keeping adults from provisioning young in nests. We recommend HDOH survey the action area prior to beginning proposed Project activities and coordinate with us for future construction activities to avoid and minimize impacts to nēnē.

The action area of the Project is not limited to the footprint of the construction activities; rather, the action area includes the surrounding area impacted by the project activities. The creation of drainage channels related to Project construction of the wastewater treatment plant (WWTP) and supporting infrastructure should be included within the Project description and considered in all consultations.

The concurrence provided in this letter is confined to the species and avoidance and minimization actions incorporated into the project activities specifically described in your letter dated May 31, 2024, and does not relieve the EPA and its designee, HDOH, of the responsibility under the ESA for take of listed species for the proposed project.

Project Description

The EPA is cooperating with HDOH, to make wastewater system improvements that will replace existing collection, treatment, and disposal systems servicing the community of Nāʿālehu in the Kaʿū District on the island of Hawaiʿi. (Figure 1). The Hawaiʿi County Department of Environmental Management, Wastewater Division is proposing to construct a wastewater treatment and disposal system to treat sewage to allow the closure of three large capacity

cesspools currently used to dispose of untreated sewage. Collection and delivery systems will be within the developed community of Nāʿālehu. A force main pump station (SPS) east of the developed community will connect to the WWTP, cross a drainage channel, and, otherwise, follow existing roads. The WWTP will be located on approximately 28-acres (11.3-hectares) of State lands to the southwest of the town on the south side of Māmalahoa Highway (Figure 1, light blue area). Site work proposed includes grading of a drainage diversion swale, beginning from a vegetated diversion channel, and heading east within the Project area (Figure 1). No streams or wetlands occur in the Project area; however, the WWTP site itself is located within a shallow drainage swale and runoff flow will be directed around the WWTP infrastructure in a new channel.



Figure 1. Nāʿālehu Wastewater Collection, Treatment, Disposal, and Drainage System Modifications Project in Nāʿālehu, Hawaiʿi. (LLC = existing large capacity cesspool; SPS = proposed sewage pumping Station; light blue = proposed wastewater treatment plant)

Conservation Measures

There are no streams or wetlands within the proposed project footprint described in the May 31, 2024 letter. The Project will incorporate the Service recommended Avoidance and Minimization Measures (Attachment A) for federally listed Hawaiian seabird species and the 'ōpe'ape'a. To avoid and minimize impacts to 'ōpe'ape'a, the Project will not use barbed wire in fencing. A provision will be included in the construction bid requiring the contractor to provide a qualified

biologist to survey for 'ōpe'ape'a during the bat birthing and pupping season from June 1 through September 15 prior to the start of any construction activities. Woody plants taller than 15 feet (ft) will only be removed outside of the June 1 to September 15 time period.

Effects to Listed Species

Hawaiian seabirds

Hawaiian seabirds may traverse the project area at night during the breeding, nesting, and fledging seasons (March 1 to December 15). Outdoor night lighting can cause seabird disorientation, fallout, and injury or mortality when seabirds are attracted to lights and after circling the lights, they become exhausted and collide with nearby wires, buildings, fences, or other structures or they may land on the ground. Fledglings are particularly vulnerable to light attraction when they are traversing the project area between September 15 and December 15, as they are making their first flights from their mountain nests to the sea. Downed seabirds are subject to increased mortality due to collision with automobiles, starvation, and depredation by dogs, cats, and other predators.

No nighttime construction activities during the seabird fledging period, September 15 to December 15, is planned for this project. Any seabirds traversing the area at night are extremely unlikely to be exposed to construction-related activities because no work will occur at night. All outdoor lights associated with the Project structures and infrastructure will be fully shielded and only used when human activity is occurring in the lighted area. Motion detectors or timers will be deployed on all outdoor lighting. All required fencing installed at the Project that extends above vegetation will integrate three strands of polytape to increase visibility and reduce collision risk. We do not expect injury, mortality, or measurable disruptions to the normal behaviors of Hawaiian seabirds. Therefore, effects to Hawaiian seabirds are considered discountable.

'Ōpe 'ape 'a

Woody vegetation may be used by 'ōpe'ape'a for roosting, potentially year around. The 'ōpe'ape'a roosts in woody vegetation across all islands and will leave their young unattended in trees and shrubs when they forage. If trees or shrubs 15 ft or taller are cleared during the pupping season, June 1 through September 15, there is a risk that young bats could inadvertently be harmed or killed, since they are too young to fly or move away from disturbance.

Human presence and project activities may cause temporary disruptions to the normal behaviors of 'ōpe'ape'a near the Project area. If bats are present during the construction, we expect the disturbance may cause them to leave the site. We expect that these disturbances will be short term and intermittent and will not result in measurable disruptions of their normal behaviors, nor will there be reductions in the reproductive success or fitness of the bats. The high mobility of adult bats enables them to relocate to suitable vegetation nearby. No nonvolant young would be injured or killed because trees and vegetation taller than 15 ft will not be removed during the pupping season when the young cannot fly.

Avoidance and minimization measures will be implemented to avoid adverse effects to 'ōpe'ape'a. 'Ōpe'ape'a are not expected to be injured, killed, or to experience measurable disruptions to their normal behaviors. Therefore, effects to the 'ōpe'ape'a are discountable.

Summary

Based on the project description, conservation measures incorporated (Attachment A) and supporting biological rationale provided above, we expect effects to the four federally listed species to be discountable (extremely unlikely to occur). Therefore, the Service concurs with the determination that the proposed Project may affect but is not likely to adversely affect federally listed species.

This concludes section 7 consultation for the proposed Project. Reinitiation of this consultation is required by the Service, where discretionary Federal involvement or control over the proposed actions has been retained or is authorized by law and:

- 1) If new information reveals effects of the action that may affect listed species or critical habitat in a manner or to an extent not previously considered;
- 2) If the identified action is subsequently modified in a manner that causes an effect to listed species or critical habitat that was not considered herein; or
- 3) If a new species is listed or critical habitat designated that may be affected by the proposed actions.

The concurrence provided in this letter under our ESA section 7 authority is only applicable to the activities of the aforementioned Project. Please continue to coordinate with us as the Project enters new construction and permitting phases.

Thank you for participating with us in the protection of our endangered species. If you have any questions, please contact Diane Sether, Ph.D., at diane_sether@fws.gov or by telephone at 808-210-4919, or pifwo_admin@fws.gov. When referring to this project, please include this reference number: 2024-0078740-S7-001.

Sincerely,

Chelsie Javar-Salas Acting Island Team Manager for Maui Nui and Hawai'i Pacific Islands Fish and Wildlife Office

Attachment: A: Service Avoidance and Minimization Measures (Animals)

Attachment A: Animals – U.S. Fish and Wildlife Service Avoidance and Minimization Measures

FINAL Avoidance and Minimization Measures (AMMs) Final revised May 2023

ESA Listed Species

Endangered 'ōpe'ape'a (Hawaiian hoary bat, *Lasiurus cinereus semotus*): The Hawaiian hoary bat roosts in woody vegetation across all islands and will leave their young unattended in trees and shrubs when they forage. If trees or shrubs 15 feet or taller are cleared during the pupping season, June 1 through September 15, there is a risk that young bats could inadvertently be harmed or killed, since they are too young to fly or move away from disturbance. Hawaiian hoary bats forage for insects from as low as 3 feet to higher than 500 feet above the ground and can become entangled in barbed wire used for fencing.

To avoid and minimize impacts to the endangered Hawaiian hoary bat we recommend you incorporate the following applicable measures into your project description:

- Do not disturb, remove, or trim woody plants greater than 15 feet tall during the bat birthing and pup rearing season (June 1 through September 15).
- Do not use barbed wire for fencing.

Endangered 'ua'u (Hawaiian petrel, *Pterodroma sandwichensis*), Threatened 'a'o, (Newell's shearwater, *Puffinus newelli*), and Endangered Hawai'i Distinct Population Segment of the 'akē'akē (band-rumped storm-petrel, *Hydrobates castro*):

Hawaiian seabirds may traverse the project area at night during the breeding, nesting and fledging seasons (March 1 to December 15). Outdoor lighting could result in seabird disorientation, fallout, and injury or mortality. Seabirds are attracted to lights and after circling the lights they may become exhausted and collide with nearby wires, buildings, or other structures or they may land on the ground. Downed seabirds are subject to increased mortality due to collision with automobiles, starvation, and predation by dogs, cats, and other predators. Young birds (fledglings) traversing the project area between September 15 and December 15, in their first flights from their mountain nests to the sea, are particularly vulnerable to light attraction.

To avoid and minimize potential project impacts to seabirds we recommend you incorporate the following measures into your project description:

- Fully shield all outdoor lights so the bulb can only be seen from below.
- Install automatic motion sensor switches and controls on all outdoor lights or turn off lights when human activity is not occurring in the lighted area.
- Avoid nighttime construction during the seabird fledging period, September 15 through December 15.

Seabirds have been known to collide with fences, powerlines, and other structures near nesting colonies. To avoid and minimize the likelihood of collision we recommend you incorporate the following measures into your project description:

- Where fences extend above vegetation, integrate three strands of polytape into the fence to increase visibility.
- For powerlines, guy-wires and other cables, minimize exposure above vegetation height and vertical profile.

We recommend further coordination with our office to address specific project details and potential seabird interactions.

APPENDIX E:

Draft Archaeological Inventory Survey

Cultural Surveys Hawaii

Draft

Archaeological Inventory Survey Report for the Nāʿālehu Large Capacity Cesspool Closure Project, Kāhilipalinui and Kāwala Ahupuaʿa, Kaʿū District, Hawaiʿi Island, TMKs: (3) 9-5-007:016; 9-5-008:033, 045, 048; 9-5-010:001, 030; 9-5-021:015, 020, 999; 9-5-022:001; 9-5-024:001, 009, 010, 011, 069, 076, 077, 999; 9-5-025:002, 039, 999; 9-5-026:999

> Prepared for County of Hawai'i Department of Environmental Management, Wastewater Division

> > Prepared by Olivier M. Bautista, B.A., Samantha Purdy, B.A., Sarah Wilkinson, B.A., and Hallett H. Hammatt, Ph.D.

Cultural Surveys Hawaiʻi, Inc. Kailua, Hawaiʻi (Job Code: KAHILIPALINUI 6)

July 2024

Oʻahu Office		Hawai'i Office
P.O. Box 1114		Hualani St. #124
Kailua, Hawai'i 96734	www.culturalsurveys.com	Hilo, Hawai'i 96720
Ph.: (808) 262-9972		Ph.: (808) 965-6478
Fax: (808) 262-4950		Fax: (808) 965-6582

Management S	ummary	included a 1,500-foot (ft) extension of an exis Under this proposed action the environmental
Reference	Archaeological Inventory Survey Report for the Nāʿālehu Large Capacity Cesspool Closure Project, Kāhilipalinui and Kāwala Ahupuaʿa, Kaʿū District, Hawaiʿi Island, TMKs. (3) 9-5-007:016; 9-5-008:033, 048; 9-5-010:001, 030; 9-5-021:015, 020, 999; 9-5- 022:001; 9-5-024:001, 009, 010, 011, 069, 076, 077, 999; 9-5-025:002, 039, 999; 9-5-026:999 (Bautista et al. 2024)	review under Hawai'i Revised Statues (HRS) respectively, was initiated. Effective as of 22 August 2022, the EPA and 0 entered a Revised AOC, which identified four evaluation:
Date	July 2024	Alternauve #1: Fackage wastewater 11 collection system
Project Number(s)	Cultural Surveys Hawai'i, Inc. (CSH) Job Code: KAHILIPALINUI 3 HICRIS Project Number: 2019PR30811 CWSRF Project Numbers: C150090-06 (Nāʿālehu Sewage Pump Station and Force Main); C150090-07 (Nāʿālehu Wastewater Treatment and Disposal System and Drainage Modifications)	 Altemative #2: Package Wastewater Tr connected to the existing collection sys Altemative #3: Individual Wastewater's Maintenance program model Altemative #4: Individual Wastewater's
Investigation Permit Number	CSH completed the archaeological inventory survey (AIS) fieldwork under archaeological fieldwork permit numbers 20-07 and 21-10 for calendar years 2020 and 2021, respectively, issued by the Hawai'i State Historic Preservation Division (SHPD) per Hawai'i Administrative Rules (HAR) §13-13-282	program model. COH-DEM Wastewater Division (WWD) has treatment alternative options, considering the to impact, public input, legal challenges, cost, and selection of a treatment option. As a result. Cno
Agencies	Hawai'i State Department of Health (DOH); SHPD; County of Hawai'i Department of Environmental Management (COH-DEM)	selected option No. 1 as the preferred alternativ I inder Alternative 1 the County would constru
Land Jurisdiction	State of Hawai'ti/County of Hawai't/Private	system in the Na alehu community to replace t
Project Proponent	COH-DEM Ramzi Mansour, Director 345 K ekuanaoa Street, Suite 41 Hilo, HI 96720 Attention: Mark Grant Email: cohdem@hawaiicounty.gov	gravity lines that convey sewage to the three Luppopsed wastewater treatment and disposal faserve the former Brewer lots as well as newly a future connection to the WWTP. The potential parcels may be further assessed by the County During construction the County shall ensure the
Project Funding	State of Hawai'i (revolving fund); County of Hawai'i	maintain access to the sewer system at all time:
Project Location	This project is in the town of Nāʿālehu in the Kaʿū District of Hawai'i Island. The project area straddles the Māmalahoa Highway (State Route 11, also called Hawai'i Belt Road) in the traditional land divisions or <i>ahupua</i> 'a of Kāhilipalinui to the west and Kāwala to the east.	utilizing the existing collection system within t house lots and constructing new gravity sewers station (WWPS), and force main to transport se the new WWTP. Alternative 1 also involves th
Project Description	In June 2017 the United States Environmental Protection Agency (EPA) and the COH-DEM voluntarily entered an Administrative Order on Consent (AOC) to close three County owned large capacity cesspools (LCCs) in the town of Na [*] atehu. The original proposed project to close the LCCs included construction of a new wastewater collection and transmission system to convey wastewater to a new wastewater treatment facility for processing and disposal to a land application system planted with native trees, to be established on the project site for that purpose. The construction of the wastewater treatment plant also	sewers to replace the existing collection. Under this alternative, the County would perfor 1. Construct a new gravity sewer on Ka'al Spur Road to the WWTP located on a p (TMK) (3) 9-5-007:016. 2. Implement drainage improvements with WWTP within TMKs (3) 9-5-007:016 i 3. Construct a new pump station located o 9-5-008:048. and construct a new force
AISR for the Nā'ālehu Cesspo	ol Closure Project, Kähilipalinui and Käwala, Ka'ŭ, Hawai'i Island	AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Management Summary

Management Summary

ting drainage channel. and historic preservation §343 and §6E-8,

OH-DEM voluntarily project alternatives for

- eatment Plant and new
 - eatment Plant Plants em
 - system (IWS) –
- system IWS Operating

s evaluated the four technical, environmental d assorted factors for a OH-DEM and WWD have e option.

the existing system of LCCs and connect it to the acility. The WWTP would accessible parcels for al connection of additional y in the future. ct a new sewer collection

that residential units can tes. Alternative 1 involves in the Brewer Company rs, wastewater pump sewage from the LCCs to the installation of gravity

rm the following actions:

- a'iki Road and Nā'ālehu ortion of Tax Map Key
- thin the vicinity of the 5 and (3) 9-5-021:015. 0n a portion of TMK: (3) te main, which crosses an

:=

TMKs: multiple

TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Management Summary

	existing storm drainage channel at Melia Street, to connect to
	the Ka'ala'iki Road gravity sewer. Construct a new gravity sewer on Opukea Street and Ohai Road to intercept existing flow entering the LCCs and divert sewage to the WVPS and transport flows to the gravity sewer along Vertexity Docu
	Ka ata 'ist Road. 5. Install gravity severs within the streets to replace the existing collocion everem
	 Close and abandon the three LCCs. Accommodate future expansion of subsurface effluent disposal
	located within a portion of 1 MK: (5) 9-5-022:001. A restachnical survey will be conducted during the design phase to
	A geotectimical survey with occupated auring the design phase to inform a soil investigation report for understanding the subsurface
	conductors. Otass cutum and muchaning equipment with or during the down the tall grass so that a topographic survey can be conducted of the terrent at heave $(23, 0.5, 0.7)$ to $(23, 0.5, 0.7)$ to $(23, 0.5, 0.7)$.
	geotechnical survey work will be monitory by CSH and the results
Area of Potential	appended to this rule for the Na'alehu Large Capacity Cesspool Closure Project is
Effect (APE) and	approximately 80.4 acres (32.5 hectares). The APE includes:
AIS Project Area	• The proposed treatment plant and disposal facility, which will be located <i>makai</i> (seaward) of the Māmalahoa Highway at
	TMKs: (3) 9-5-007:016 and (3) 9-5-022:001. An associated drainage channel may be located on a portion of TMK: (3) 9-5-071:015.
	• The location(s) of proposed pump station at TMK: (3) 9-5- 008-064
	The proposed collection system primarily in the County streets of Marinehue.
	 The locations of three LCCs which will be decommissioned; and
	 Community parcels currently connected to the existing County system.
	The AIS project area comprises a 49.62-acre (20.08-hectare) portion of
	the overall AFE. I he project area comprises humerous tax parcels (in whole or part) and public robadway rights-of-way; a full list of project www.e.g. area incubic incommentations in an incubic of project
	I PARS and approximation introduction is provided in Appendix A. The ALS project area includes all the components of the APE listed above except
	for the currently connected community parcels. All project staging will be confined within the AIS project area limits at the proposed treatment plant and sewer pump station site.
Historic Preservation Regulatory Context	This AIS investigation was designed to be compliant with both Federal and Hawai's State environmental and historic meservation review
AISB for the Na'alan Casenoor	Province Vishilinelinei and Visuala Volis Hauraisi Island
ALNK FOT THE INA ALENH CESSPOO TMKS: multiple	I Closure Project, namipaintu and Nawaia, na u, Hawa I Istand

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Management Summary

	legislation. Due to funding from the Clean Water State Revolving Fund (CWSRF), this project is considered an undertaking requiring compliance with Section 106 of the National Historic Preservation Act (NHPA) and the National Environmental Policy Act (NEPA). As a county project within private, county, and state lands, the project is also subject to Hawai' i State environmental and historic preservation review legislation (Hawai' i State Statutes IHRS] §3-43 and HRS §6E- 8/Hawai'i Administrative Rules [HAR] §13-275, respectively).
	The Na [*] alehu Large Capacity Cesspool Closure Project (comprising the Na [*] alehu Sewage Pump Station and Force Main Project and Na [*] alehu Wastewater Treatment and Disposal System and Drainage Modifications Project) involves CWSRF funding and is therefore an undertaking subject to the NHPA and its implementing regulations. The EPA is the lead federal agency for this undertaking. The EPA
	administers the CWSRF Program, which authorizes capitalization grants to state agencies in Region 9, including the Hawai'i State DOH. In turn, the DOH provides sasistance to public and private community water systems as well as nonprofit noncommunity water systems for CWSRF projects. In October 2015 the EPA authorized the DOH to undertake consultation with the State Historic Preservation Officer (SHPO) and Native Hawaiian organizations (NHOs) for projects funded under the CWSRF (Appendix B).
	In consultation with the SHPD, this AlS investigation fulfills the requirements of HAR §13-13-276 and the <i>Secretary of the Interior's Standards for Archaeology and Historic Preservation</i> . It was conducted to assess any historic properties for integrity and site significance in accordance with HAR §13-275-6. This report is also intended to support any project-related historic preservation consultation with consulting parties, state and county agencies and interested Native Hawaiian Organizations (NHOs) and community groups, if applicable.
	In 2020 CSH prepared an archaeological literature review for the project's Environmental Assessment (Purdy et al. 2020). This investigation was designed—through detailed historical, cultural, and archaeological background research—to determine the likelihood that historic properties may be affected by the project and, based on findings, consider cultural resource management recommendations. It was also intended to support the COH-DEM's consultation with the SHPD regarding the project's necessary historic preservation review steps pursuant to HAR §13-275.
	On 22 October 2020 the County submitted a packet of materials (including a cover letter dated 16 October 2020) to SHPD initiating consultation under HRS §6E-8/HAR §13-275-5(b) and requesting concurrence with the need for an AIS for the project. SHPD replied in
or the Nā'ālehu Cesspool	i Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island

AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple
Management Summary

	correspondence dated 1 December 2020 (Log No.: 2020/02555; Doc. No.: 2012NM01; Appendix C) concurring with the DEM's recommendation that an AIS he conducted The SHPD letter also stated
	that proposed AIS testing strategy attached to the County's 22 October
	2020 submittal "appears sufficient in identifying historic properties within the current project area and determining the potential impacts to
	those sites should the AIS identify any."
	The DOH initiated Section 106 consultation with the State Historic Preservation Office and other parties in June 2024.
Fieldwork Effort	Fieldwork was conducted intermittently between 26 October 2020 and 29 January 2021 by CSH Archaeologist Samantha Purdy, B.A., and
	Project Director Olivier M. Bautista, B.A., under the general supervision of Principal Investigator Hallett H. Hammatt, Ph.D. This
Consultation	Work representation of the presentation will be compared and NHPA Section 106 consultation with community members, agencies, and NHOs is heine undertaken by the project monoments
Historic Properties Identified	This AIS identified four previously documented and seven newly documented historic properties in the project area.
	The four previously documented sites include the following:
	• SIHP # 50-10-74-29507 is a historic ranching complex. Pursuant to HAR §13-275-6 it is assessed as significant under Criterion d for the information it has yielded about historic ranching activity
	in the project area. It is evaluated as not eligible for listing on the National Register of Historic Places ("National Register").
	 SIHP # 50-10-47-30187 comprises the former and present alignments of the Māmalahoa Highway, also known as the Belt Road (variably State Routes 11, 19, 180, and 190). Pursuant to
	HAR §13-275-6 it is assessed as significant under Criterion a for its association with events that have made an important
	contribution to the development of transportation routes on Hawaii Island, and under Criterion d for having yielded
	information about historic transportation in the project area. Pursuant to 36 CFR 60.4 is evaluated as eligible for listing on
	the National Register under Criterion A (for its association with events that have made a significant contribution to the broad
	patterns of our history) and Criterion D (for having yielded, or being likely to yield, information important in history).
	 SIHP # 50-0-74-30929 is a pre-Contact activity area and historic ranching complex. It is assessed as significant under Criterion d for the information it has vielded about traditional and historic

AISR for the Na^{*}älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka^{*}ū, Hawai'i Island TMKs: multiple

>

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Management Summary

|--|

AISR for the Naʿālehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Kaʿū, Hawaiʾi Island TMKs: multiple

.5

Management Summary

Effect Recommendation	In accordance with federal regulations (36 CFR 800.5), the AIS results support the determination made by the Hawai'i State DOH on behalf of the EPA of "no adverse effect." In accordance with HAR §13-275-7, the Hawai'i State DOH and COH-DEM have determined the project effect is "effect, with proposed mitigation commitments." The project proponents request SHPD concurrence with these determinations.
	One historic property on the project area has been evaluated as eligible for inclusion on the National Register: SIHP # 50-10-47-30187 (Mämalahon Hiohwav) The 15-m (50-ft) nortion of this historic
	property within the project area retrievely protection only, as all the constructed elements of the original Māmalahoa Highway are no
	honger evident today. Write the project with affect the modern todaway elements, these elements will be replaced following construction, and the integrity and significance of the highway will not be diminished. Thereace, a the aveign will house or device active to the aveignt.
	This site is assessed as significant under Criterion a of the State of Hawai's significance criteria for its association with events that have
	made an important contribution to the development of transportation routes on Hawaii Island, and under Criterion d for having yielded information about historic transportation in the project area.
	Documentation of SLH1 [#] $=$ -9U18 / in this ALS has included instorteal research, GPS data collection, photographs, and written description. This ALS has adequately documented the location, extent, function, age, and construction methods of the portion of SIHP \pm -30187 in the project area, mitigating project-related impacts pursuant to HAR §13-275-8.
	The remaining ten historic properties in the project area (SIHP #s 50- 10-74-29507, -30929, -30930, -31268, -31269, -31270, -31271, -31272, -31273, and -31274) may also be impacted by project development.
	I here instorte properties have been valuated as not engule for inclusion on the National Register. These sites are assessed as significant under Criterion d of the State of Hawai' i significance criteria because they have vielded information important for research on
	history. Documentation of these historic properties in this AIS has included historical research, GPS data collection, photographs, plan view maps, and written descriptions. This AIS has adequately
	documented the location, extent, function, age, and construction methods of these ten historic properties, mitigating project-related impacts pursuant to HAR §13-284-8.
Mitigation Recommendations	The single historic property evaluated as eligible for listing on the National Register (SIHP # 50-10-47-30187, Mämalahoa Highway), will
	not be adversely impacted by the project as its integrity and significance will not be diminished. This AIS has adequately documented the location, extent, function, age, and construction methods of the 11
	historic properties within the project area, mitigating any project-related

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

. LI V

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Management Summary

impacts pursuant to HAR §13-284-8. No further work is recommended for these 11 historic properties. Archaeological monitoring is recommended during all project ground-

Archaeological monitoring is recommended during all project grounddisturbing activities to facilitate historic property identification for information purposes, especially of any subsurface lava tubes that may contain historic properties. Achaeological monitoring will proceed HAR §13-2794, which will be submitted to the SHPD for review and acceptance prior to project initiation.

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Table of Contents

Management Summary

....i

1.1 Project Background1
 Historic Preservation Regulatory Context and Document Purpose
1.3.1 Natural Environment
Section 2 Methods
2.1 Field Methods15
2.1.1 Pedestrian Survey15
2.1.2 Subsurface Testing15 2.2 Laboratory Methods19
2.2.1 Artifact Analysis
2.2.2 Energy-Dispersive X-ray Fluorescence (EDXRF) Analysis
2.2.4 Radiocarbon Analysis
2.2.5 Ursposition of Materials
2.4 Consultation Methods
3.1 Traditional and Historical Background
3.1.3. The Mathematics recovery a second sec
3.1.5 1900s
3.1.6 Contemporary Land Use
Section 4 Results of Fieldwork44
4.1 Pedestrian Inspection Results 44 4.2 Subsurface Testing Results 59 4.2.1 Mechanical Exploratory Trenching 59 4.2.2 Manual Test Excavations 103
Section 5 Results of Laboratory Analysis109
5.1 Artifact Analysis 10 5.1.1 Traditional Hawaiian Artifacts 10 5.2.1 Wood Taxa Analysis 112 5.2.2 Radiocarbon Analysis 112 5.2.2 Radiocarbon Analysis 112
Section 6 Site Descriptions
AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island iX TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

6.1 SIHP # 50-10-74-29507
6.2 SUHP # 50-10-4/-3018/
6.4 SIHP # 50-10-74-30930
0.5 SIHP # 50-10-74-51208141 6.6 SIHP # 50-10-74-31269145
6,7 SIHP # 50-10-74-31270
6.8 SIHP # 50-10-74-31271157 6 0 SIHP # 50-10-74-31272
6.10 SIHP # 50-10-74-31273
6.11 SIHP # 50-10-74-31274165
Section 7 Summary and Interpretation167
Section 8 Significance Assessments and Nation Register Eligibility Determinations 168
8.1 Significance Assessments under HRS §6E168
8.1.1 SIHP # 50-10-74-29507168 8 1 2 STHP # 50 10 47 30187
8.1.3 SIHP # 50-10-74-30929169
8.1.4 SIHP # 50-10-74-30930169
8.1.5 SIHP # 50-10-74-31268169 9.1.6 STHP # 50.10.74.31260
8.1.7 SIHP # 50-10-74-31279169
8.1.8 SIHP # 50-10-74-31271
8.1.9 SIHP # 50-10-74-31272170
8.1.11 SIHP # 50-10-74-31273
8.2 National Register Eligibility Determinations
8.2.1 SIHP # 50-10-74-29507
8.2.2 SIHP # 50-10-4/-5018/1/1 8 2 3 SIHP # 50-10-74-30039
8.2.4 SIHP # 50-10-74-30930
8.2.5 SIHP # 50-10-74-31268
8.2.6 SIHP # 50-10-74-31269172 8.2.5 SIHD # 50-10-74-31269
8.2.8 SIHP # 50-10-74-31271
8.2.9 SIHP # 50-10-74-31272.
8.2.10 SIHP # 50-10-74-31273
Section 9 Project Effect and Mitigation Recommendations.
0.1 Deviaet Britact
9.1 Riget Differentieren ander and
Section 10 References Cited
Appendix A Project Area Land Jurisdiction179
Appendix B EPA Section 106 Authorization180
Appendix C SHPD Correspondence182
AISB for the N3 3febra Cesswool Closure Proviect Kähtlinelinni and Käwala Ka'ü. Hawai'i Island
A A TMKs multiple

Appendix D EDXRF Analysis Report 184
Appendix E Radiocarbon Analysis Report186
Appendix F Site Descriptions from Previous Archaeological Studies
SIHP # 50-10-74-29507 (Clark et al. 2013:184–207)191
SIHP # 50-10-47-30187 (Clark et al. 2014:52–58)
SIHP # 50-10-74-30929 (Gastilo and Clark 2018: 145–149; Draft)
SIHP # 50-10-74-30930 (Gastilo and Clark 2018: 149–150; Draft)
Appendix G Select Channel Improvements Plans (SIHP # 50-10-74-31273)

AISR for the Na^{*}älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka^{*}ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

List of Figures

Figure 1. Portion of the 1995 Naalehu USGS 7.5-minute topographic quadrangle showing the
Figure 2. Tax Map Key (TMK) (3) 9-5-07 showing the project area; the proposed treatment facility would be located within a nortion of prosed IO16 (Hauvis); TMK 2011A) 3
Figure 3. TMK: (3) 9-5-008 showing portion of parent or of trawar 1 1000 2033, 045, and 048 (monosed minime portions of the project area within parels 033, 045, and 048 (monosed minime station station (1300) 1. TMK Service 2014)
Figure 4. TMK: (3) 9-5-021 showing the project area crossing the State-owned Māmalahoa Hiphway (Hawaii' TMK Service 2014)
Figure 5. TMK: (3) 9-5-024 showing portions of the project area within parcels 001, 009, 010, 011, 010, 011, 040, 076, 077, 073, and variance county modurase <i>I</i> chelotai Streast Vilities Streast Onlyce
Street, Obia Road, Kukui Road) (Hawai'i TMK Service 2014)
Figure 6. 1 MK: (3) 9-5-025 showing portions of the project area within parcels 002, 039, and county roadways (Milo Road, Melia Street) (Hawai'i TMK Service 2014)7
Figure 7. TMK: (3) 9-5-026 showing portion of the project area within various County roadways (Ohai Road, Nahele Road, Lokelani Road, Kukui Road) (Hawai'i TMK Service 2014)8
Figure 8. Aerial photograph of the project area (ESRI 2017-2018) showing the AIS project area within the Jaroer APF
Figure 9. Overlay of Soil Survey of the Island of Hawaii (Sato et al. 1973) indicating soil types
within and surrounding the project area (USDA SSURGO 2014)
trenches at the proposed wastewater treatment facility site at TMK: (3) 9-5-007:01617
Figure 11. Client provided map showing approximate proposed locations for three test
excavation trenches at the proposed at the proposed sewer pump station site at 1MK: (3) 9-5-008:048
Figure 12. Map of Hawai'i Island showing population as of 1853 (Coulter 1931:28)25
Figure 13. Portion of the 1995 Naalehu USGS 7.5-minute topographic quadrangle showing the location of I CAs within and in the immediate vicinity of the moviect area 27
Figure 14. Portion of W.A. Wall 1886 map of Hawai'i Island showing the project area in relation
to sugar mills, the landing at Honu'apo, and established transportation routes
Figure 15. Portion of M.S. Monsarrat 1887 map of Ka'ū showing the project area; note location
Prigure 16. Portion of J.M. Donn 1906 map of Hawai'i Island showing the project area within
sugar plantation lands
rigue 1.7. Fottout of the 1721 (varient USUS) 7.3-finitude topographic quantangle subwing the project area
Figure 18. Portion of the 1962 Naalehu USGS 7.5-minute topographic quadrangle showing the
project area
showing the project area
Figure 20. Aernal photograph (ESRI 2022) showing previous archaeological studies conducted and historic properties documented in the vicinity of the project area
Figure 21. Site distribution map from Clark et al. (2008:19) showing the locations of documented
historic properties
AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island Xii
TMKs: multiple

.x

Figure 22. Site distribution map from Clark et al. (2013:123) showing the locations of documented historic properties
Figure 2.5. Friotograph tooking down Kukut Koau Within the collection system portion of the project area; view to south
project area; view to east
Figure 26. Photograph overlooking the Opukea Street and Kukui Road intersection within the collection system portion of the project area; view to east
Figure 27. Photograph showing a portion of Ohai Road within the collection system portion of the project area; view to north47
Figure 28. Photograph overlooking a portion of the northern-most spur of the collection system above the Na alehu town; evidence of cattle grazing was observed in this area; view to
West
above the Na ⁷ älehu town; evidence of cattle grazing was observed in this area; view to north
Figure 30. Photograph overlooking the proposed pump station portion of the project area, at the site of the old Na ⁺ alehn Dairy: view to southwest.
Figure 31. Photograph overlooking the proposed pump station portion of the project area, at the
Figure 32. Photo overlooking the portion of the project area connecting the proposed sewer
pump station site to the collection system along Ohai Road; view to northeast
Māmalahoa Highway; view to north50
Figure 34. Photograph overlooking a portion of the proposed treatment facility site <i>makai</i> of Māmalahoa Highway; view to east50
Figure 35. Photograph overlooking a portion of the proposed treatment facility site <i>makai</i> of Mämalahoa Hiohwav: view to west 51
Figure 36. Photograph of cattle grazing in the proposed treatment facility site, in the SIHP # - 31360 vicinity: vicinity or and
Figure 37. Photograph taken within the natural depression bisecting the proposed treatment
facility portion of the project area; view to west
depression bisecting the proposed treatment facility portion of the project area; view to
horthwest
Street; view to southeast
Figure 40. Photograph of access to LCC 1 at LMK: (3) 9-5-124:011; when to southeast
Street; view to southwest
AISR for the Nāʿālehu Cesspool Closure Project, Kālnilīpalīnui and Kāwala, Kaʿū, Hawaiʿi Island xi <u>ii</u> TMKs. multīnie
LINLS: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Figure 43. Photograph showing the vicinity of LCC 3 at TMK: (3) 9-5-024:001 adjacent to the
Māmalahoa Highway and Ohai Road intersection; view to northwest
Project area
Potton of the project area
Figure 48. Photograph of TT-1 marked out with flagging tape prior to excavation; view to east 61 Figure 48. Photograph of TT-1 marked out with flagging tape prior to excavation; view to east 61 Figure 40. Photograph of TT-1 marked out with the onthwest
rigue 72.1 novegaph of 1.1.1 post-externant, 1.1.4 to not investigation of 1.1.1 post-externant control from 0.2 Figure 50. Photograph of TT-1 east sidewall profile; view to east
Figure 52. Photograph of TT-2 marked out with flagging tape prior to excavation; view to somtweet
Figure 53. Photograph of TT-2 post-excavation; view to southeast
rigue 54. ruougraph of 1.1-2 cast stuewan proute, view to cast
Figure 56. Photograph of TT-3 marked out with flagging tape prior to excavation; view to north
Figure 57. Photograph of TT-3 post-excavation; view to northeast
Figure 58. Photograph of TT-3 north sidewall profile; view to north
rigure 39. frome of 11-5 norm succean
Figure 61. Photograph of TT-4 post-excavation; view to northwest70
Figure 62. Photograph of TT-4 north sidewall, view to the north
Figure 64. Photograph of TT-5 marked out with flagging tape prior to excavation, view to west
Figure 65. Photograph of TT-5 post excavation; view to the northwest
Figure 66. Photograph of TT-5 west sidewall, view to west
Figure 68. Photograph of TT-6 marked out with flagging tape prior to excavation; view to
southeast
Figure 70. Photograph of TT-6 east sidewall profile, view to east
Figure 71. Frome of 1.1-0 day successful
Figure 73. Photograph of TT-7 post-excavation; view to north
Figure 74. Photograph of T1-7 north sidewall profile; view to north
Figure 76. Photograph of TT-8 marked out with flagging tape prior to excavation; view to west
70
AISR for the Närätehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island xiy TMKs: multiple

Figure 77. Photograph of TT-8 post-excavation; view to west
Figure 80. Photograph of TT-9 marked out with flagging tape prior to excavation; view to east 85 Figure 81 Dhistorraph of TT-9 marked out with flagging tape prior to excavation; view to east 85
Figure 81. Photograph of TT-9 west sidewall profile; view to west
Figure 84. Photograph of TT-10 marked out with flagging tape prior to excavation, view to west 84.
Figure 85. Photograph of TT-10 post-excavation, view to west
Figure 87. Profile of TT-10 west sidewall
Figure 89. Photograph of TT-11 post-excavation; view to north
Figure 91. Profile of TT-11 south sidewall
Figure 93. Photograph of TT-12 post-excavation; view to east
Figure 74. 1 Invogradu of 1.1-12 west stoewart profinely, view to west
Figure 90. Fnotograph of 1.1-1.5 marked out with flagging tape prior to excavation; view to northwest
Figure 97. Photograph of TT-13 post-excavation; view to northeast
Figure 99. Profile of TT-13 cast sidewall
Figure 100. Photograph of TT 14 marked out with flagging tape prior to excavation; view to south
Figure 101. Photograph of TT 14 post-excavation; view to southwest
Figure 102. Friorograph of 1.1.14 north suce wall profile; view to east
Figure 104. Photograph of TU-1 marked out with pin flags and string prior to excavation, view to southwest
Figure 105. Photograph of TU-1 post-excavation; view to southwest
Figure 100. Photograph of 1 U-1 soutinest sidewall profile; view to soutinwest 104 Figure 107. Profile of TU-1 (showing depths in cm below datum)
Figure 108. Photograph of TU-2 marked out with string prior to excavation, view to west106
Figure 109. Fnotograph of 1.0-2 post-excavauon; view to west
Figure 111. Profile of TU-2 west sidewall (showing depths in cm below datum) Figure 112 Photos of Cataloc #001 hasalt adze
Figure 113. Photos of Catalog # 003, coral abrader110
Figure 114. Bivariate plot showing Niobium (Nb) vs. Kubidium (Rb) ratios of "CSH Adze" in relation to other basalt samples from Ka'ū (courtesy UHH EDXRF Lab)11
AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawa''i Island Xv TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Figure 115. Bivariate plot showing Strontium (Sr) vs. Zirconium (Zr) ratios of "CSH Adze" in
Feation to other basait simples from it a u (courtes) UHH EDAKF Lab)
Figure 117. Calibration data and calibration curve for Sample Beta–592958 / 2b from TU-1 (contensor of Base Addition)
Figure 118. Plan view map of SIHP # -29507 from Clark et al. (2013:185) overlain with the <i>approximated</i> current project area extent (in pink) and showing the locations of newly
assigned Features E and F
Figure 120. Photograph of a portion of the SIHP # -29507 Feature E rock wall; view to northeast
Figure 121. Plan view of SIHP # -29507 Feature F, drainage channel
Figure 123. Photograph showing an earthen section of the SIHP # -29507 Feature F drainage
channet, whith the paratient write rence visione in background, view to southeastr Figure 124. Photograph showing SIHP # -29507 Feature G fence connecting to the southeastern
Figure 125. Photograph of a portion of SIHP #-29507 Feature G fence, showing typical
construction; view to west
Highway) in Hilea, view to south
rigure 127. Friotograph from Tucna and Framman (2017:32) of SHAF # -3016/ (Manatanoa Highway) in Ninole; view to north
Figure 128. Photograph overlooking the portion of SIHP # -30187 (Māmalahoa Highway) within the avoid mean view to methode
Tigure 129. Photograph overlooking the portion of SIHP # -30187 (Māmalahoa Highway) within
the project area; view to southwest128 Evine 130 Dian view of STHD # -30020 commlex
Figure 130. Flatt view 01 SIHP # -50252 Comptex
Figure 132. Photograph of SIHP # -30929 Feature A alignment; view to south133
Figure 133. Photograph of SIHP # -30929 Feature B paved area; view to north
Figure 134, Frienograph of SIHP # -30929 Feature D payed at ea, view to east Figure 135. Photograph of SIHP # -30929 Feature C rock mound; view to the northwest
Figure 136. Photograph overlooking the western section of the SIHP # -30930 rock wall; view to
Figure 137. Photograph showing typical construction along a section of the SIHP # -30930 rock
wall; view to the north
rock wall; view to north
Figure 139. Photograph of the breach in the SIHP # -30930 rock wall, and the modern fencing used for remain view to west 139
Figure 140. Plan view of SIHP # -31268, drainage ditch
AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island Xvj
TMKs: multiple

Figure 141. Photograph of SIHP # -31268 taken atop the earthen berm forming the eastern ditch
embankment; view to north
Figure 142. Photograph of SIHP #-51268, taken from within the ditch and looking toward the western herm (scale is leaning against interior herm wall): view to northwest
Figure 143. Photograph of SIHP #-31268 interior ditch channel, looking upslope from the
southern end of the ditch; view to north
Figure 144. Photograph of cartnen spiilway Within SLHP # - 51208 aiten; view to northwest 144 Finure 145 Dian view of STHD # _21260 remobilion commiley
Figure 146. Photograph overlooking the SIHP # -31209; teature A paddock from its northeastern
corner; view to north147
Figure 147. Photograph overlooking SIHP # -31270 jeep road extending through the eastern
portion of the SIHP # -31269 Feature A paddock; view to west
Figure 146. Fnotograph of water trougn along eastern side of SLIFT # -51209 Feature A paddock; view to north
Figure 149. Photograph overlooking the western portion of the SIHP # -31269 Feature A
paddock (SIHP # -31270 jeep road visible at center); view to east
Figure 1.30. Photograph of overlooking the SLHP # -31209 Feature B holding pens, view to how to have
Figure 151. Photograph of interior of the western SIHP # -31269 Feature B holding pen, with
concrete trough visible at center; view to east149
Figure 152. Photograph overlooking SIHP # -31269 Feature C corrals; view to east151
Figure 153. Photograph of the southern corral at SIHP # -31269 Feature C; view to northwest 151
Figure 1.34. Protograph of the corridor between the northern and southern corrals at SLHP $\#$ - 31360 Feature C vian to east north
Figure 155. Photograph showing the access to the northeastern corral at SIHP # -31269 Feature
C adjacent to the Feature D loading ramp; view to south152
Figure 156. Photograph overlooking the SIHP # -31269 Feature D loading ramp and squeeze
chute from within the northeastern Feature C corral; view to northeast
Figure 157. Photograph overlooking the SIHP # -31269 Feature D loading ramp and squeeze
cnue with realure B notaing pens visiole at background right, view to southeast125 Figure 158 Photograph overlooking the portion of SIHP # -31270 ison road near its intersection
with Na ² alehu Spur Road; view to south
Figure 159. Photograph overlooking the portion of SIHP # -31270 jeep road between the Taylor
Built Construction yard and the Onipa'a Ranch entry gate, view to north
Figure 100. Photograph overlooking a portion of the SLHP # -512/0 Jeep road makal of SLHP #S -30930 and -31360 view to south
Figure 161. Photograph of SIHP # -31270 jeep road passing through a gate along the SIHP #
-30930 rock wall and into the SIHP # -31269 Feature A paddock; view to northeast156
Figure 162. Photograph overlooking the portion of SIHP # -31271 (Nā'ālehu Spur Road) within
the project area; view to north
rigure 105. Protograph overlooking the portion of SLHP # -512/1 (iva alenti Spur Koad) within the project area: view to south
Figure 164. Photograph overlooking the portion of SIHP # -31272 (Ka'alaiki Road) within the
project area; view to north160
AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island Xviii
TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

AISR for the Naʿālehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Kaʿū, Hawaiʾi Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Introduction
List of Tables	Section 1 Introduction
Table 1. LCAs in the immediate vicinity of the project area. 28 Table 2. Land use within <i>kaleana</i> awards overlapping the project area. 28 Table 3. TT-15 Stratigraphic Description 66 Table 6. TT-15 Stratigraphic Description 66 Table 6. TT-15 Stratigraphic Description 66 Table 6. TT-15 Stratigraphic Description 67 Table 6. TT-15 Stratigraphic Description 67 Table 7. TT-5 Stratigraphic Description 67 Table 9. TT-5 Stratigraphic Description 78 Table 1. TT-7 Stratigraphic Description 78 Table 1. TT-15 Stra	 1.1 Project Background Alt request of County of Havari Department of Environmental Management (DEM), clintal Survey Elavari, Jac, CSH) has prepared this achieological invertory survey (AIS) for the N87 albub. Large Gondry Cleaner Project areas, which comprises numerous tax parcels and pholic protein Ray and the Narrow of the N87 albub. Large Gondry Cleaner Project areas, which comprises numerous tax parcels and pholic protein Ray and ST 7.5 minute topographic quadrage (Figure 8). The structure of the N995 U.S. Goodgraf Survey (SSS) 7.5 minute topographic quadrage (Figure 8). The project areas which comprises numerous tax parcels and pholic protein a pholic rotavy rights of two H and ST 7.2 minute topographic quadrage (Figure 8). The original phologenphic quadrage (Figure 7), and 2.017-2018 actal phologenphi (Figure 8). The original project phologen first and construction of a new vastewater collection and transmission system potention for the N87 (SSS 2). The structure of the norther of a new vastewater collection and the N019 CL 0.5 minute toposet. TMKs and applicable in whole or puttod and construction of a new vastewater collection and transmission system processing particular data dependent (LCGs) in the nort of N and Store (SSS). The structure of the nortex of the nortex of the project site first purpose. The construction of the wastewater nortex on the project site first physics of preservation system planed do with native trees, to be established on the environmental and historic preservation system planed do construction of the wastewater transmistic site (TRSS), S-3, S-3, S-3, S-3, S-3, S-3, S-3, S-3
AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kävala, Ka'ü, Hawai'i Island xix	AISR för the Nä älehu Cesspool Closure Project, Kähilipalinui and Kärvala, Ka'ū, Havai'i Island

AISR for the Nä'älehu Cesepool Closure Project, Kähilipalimui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

xix





Figure 1. Portion of the 1995 Naalehu USGS 7.5-minute topographic quadrangle showing the location of the project area

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple





Introduction





AISR for the Nä"älehu Cesspool Closure Project, Kähilipulinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

ŝ





Figure 3. TMK: (3) 9-5-408 showing portions of the project area within parcels 033, 045, and 048 (proposed pump station site) (Hawai'i TMK Service 2014)

AISR for the Nii älehu Cesspool Closure Project, Kähilipalinui and Kävatla, Ka'ü, Hawai'i Island TMKs: multiple



Introduction

Introduction



Figure 4. TMK: (3) 9-5-021 showing the project area crossing the State-owned Māmalahoa Highway (Hawai'i TMK Service 2014)

AISR for the Mi' äddu Cesspool Closure Project, Kähilipalinui and Kätvala, Ka'ü, Havva'i Island TMKs: multiple

4

S



Figure 5. TMK: (3) 9-5-024 showing portions of the project area within parcels 001, 009, 010, 011, 069, 076, 077, and various county roadways (Lokelani Street, Kilika Street, Opukca Street, Ohai Road, Kukui Road) (Hawai'i TMK Service 2014) 9

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILJPALJNUJ 6

Introduction

Introduction



Figure 6. TMK: (3) 9-5-025 showing portions of the project area within parcels 002, 039, and county roadways (Milo Road, Melia Street) (Hawai'i TMK Service 2014)

1

AISR for the Nii äldelu Cesspool Cleasure Project, Kähtlippdinui and Kätwala, Ka'ü, Hawai'i Island TMKs: multiple

Introduction

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Introduction



Figure 7. TMK: (3) 9:5-026 showing portion of the project area within various County roadways (Ohai Road, Nahele Road, Lokelani Road, Kukui Road) (Hawai'i TMK Service 2014)

AISR for the Nër äiden. Cesspool Closure Project. Kähilipulinni and Käwala, Ka'û, Huvui'i Island TMKs: muhipie

×



Figure 8. A erial photograph of the project area (ESRI 2017-2018) showing the AIS project area within the larger APE

AISR for the Nä'älehu Cesspool Closue Projeet, Kähilipalinui and Käwala, Ka'û, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Introduction	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Introduction
(WWPS), and force main to transport sewage from the LCCs to the new WWTP. Alternative 1 also involves the installation of gravity sewers to renlace the existing collection.	1.2 Historic Preservation Regulatory Context and Document Purpose
Under this alternative, the County would perform the following actions:	This AIS investigation was designed to be compliant with both Federal and Hawai'! State environmental and historic preservation review legislation. Due to funding from the Clean Water
 Construct a new gravity sewer on Ka'ala'iki Road and Na'alehu Spur Road to the WWTP located on a portion of Tax Man Key (TMK) (3) 9-5-007;016. 	State Revolving Fund (CWSRF), this project is considered an undertaking requiring compliance with Section 106 of the National Historic Preservation Act (NHPA) and the National
2. Implement drainage improvements within the vicinity of the WWTP within TMKs (3) 9-5-007:016 and (3) 9-5-021:015.	Environmental Policy Act (NEPA). As a county project within private, county, and state lands, the project is also subject to Hawari State environmental and historic preservation review legislation
Construct a new pump station located on a portion of TMK: (3) 9-5-008:048, and construct a new force main, which crosses an existing storm drainage channel at Melia	(Hawai'i Revised Statutes [HRS] §343 and HRS §6E-8/Hawai'i Administrative Rules [HAR] §13- 275, respectively).
Street, to connect to the Ka'ala'iki Road gravity sewer. 4. Construct a new gravity sewer on Opukea Street and Ohai Road to intercept existing flow entering the LCCs and divert sewages to the WWDS and transmort flows to the	The Nā alehu Large Capacity Cesspool Closure Project (comprising the Nā alehu Sewage Pump Station and Force Main Project and Nā alehu Wastewater Treatment and Disposal System
5. Install gravity sewers within the streets to replace the existing collection system.	and Drainage Modifications Project) involves CWSKF funding and is therefore an undertaking subject to the NHPA and its implementing regulations. The EPA is the lead federal agency for this undertaking. The EPA administers the CWSRF Program, which authorizes capitalization grants to
 Close and abandon the three LCCS. Accommodate future expansion of subsurface effluent disposal located within a portion 	state agencies in Region 9, including the Hawai'i State DOH. In turn, the DOH provides assistance to public and private community water systems as well as nonprofit noncommunity water systems
of 1MK: (5) 9-5-022:001. A geotechnical survey will be conducted during the design phase to inform a soil investigation	for CWSRF projects. In October 2015 the EPA authorized the DOH to undertake consultation with the State Historic Preservation Officer (SHPO) and Native Hawaiian organizations (NHOs) for
report for understanding the subsurface conditions. Grass cutting and mulching equipment will be	projects funded under the CWSRF (Appendix B).
utilized to cut down the tall grass so that a topographic survey can be conducted of the terrain at parcels TMKs: (3) 9-5-007:016, (3) 9-5-021:015, and (3) 9-5-022:001 prior to the geotechnical boring. The grass cutting and geotechnical survey work will be monitored by CSH and the results appended to this AIS report upon completion.	In consultation with the SHPD, this AIS investigation fulfills the requirements of HAR §13-13- 276 and the Secretary of the Interior's Standards for Archaeology and Historic Preservation. It was conducted to assess any historic properties for integrity and site significance in accordance with HAR 812-2756. This removes is also intended to summary any moior-related historic preservation
The federal Area of Potential Effect (APE) for the Na ale hu Large Capacity Cesspool Closure Project is approximately 80.4 acres (32.5 hectares). The APE is shown in Figure 8 and includes:	consultation with consulting parties, such as state and county agencies and interested Native Hawaiian Organizations (NHOs) and community groups, if applicable.
• The proposed treatment plant and disposal facility, which will be located <i>makai</i> (seaward) of the Māmalahoa Highway at TMKs: (3) 9-5-007:016 and (3) 9-5-022:001. An associated drainage channel may be located on a portion of TMK: (3) 9-5-021:015;	In 2020 CSH prepared an archaeological literature review for the project's Environmental Assessment (Purdy et al. 2020). This investigation was designed—through detailed historical, cultural, and archaeological background research—to determine the likelihood that historic
 The location(s) of proposed pump station at TMK: (3) 9-5-008:048; The proposed collection system primarily in the County streets of Nā[*]älehu; The locations of three LCCs which will be decommissioned; and Community parcels currently connected to the existing County system. 	properties may be attected by the project and, based on indings, consider cultural resource management recommendations. It was also intended to support the County of Hawai'i DEM's consultation with the SHPD regarding the project's necessary historic preservation review steps pursuant to HAR §13-275.
The AIS project area comprises a 49.62-acre (20.08-hectare) portion of the overall APE (see Fioure 8). The moviest area commrises numerous fax parcels (in whole or part) and nublic readway.	On 22 October 2020 the County submitted a packet of materials (including a cover letter dated 16 October 2020) to SHPD initiating consultation under HRS §6E-8/HAR §13-275-5(b) and
The AIS project area includes all the components of the APE listed above except for the currently connected community parcels. All project staging will be confined within the AIS project area limits at the proposed treatment plant and sewer pump station site.	requesting concurrence with the need for an AIS for the project. SHPD replied in correspondence dated 1 December 2020 (Log No.: 2020.02555; Doc. No.: 2012NM01; Appendix C) concurring with the DEM's recommendation that an AIS be conducted. The SHPD letter also stated that proposed AIS testing strategy attached to the County's 22 October 2020 submittal "appears sufficient in identifying historic properties within the current project area and determining the protomed in masters to those excession(the AIS king any.
	The DOH initiated Section 106 consultation with the State Historic Preservation Office and other parties in June 2024.
AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island 10 TMKs: multiole	AISR for the Nä älehu Cesspool Closure Project, Kähilipalimui and Kävala, Ka'ū, Havai'i Island 11 TMKs.multinle
and the state of t	L'AUXOS ANNAUGOS

- The proposed treatment plant (seaward) of the Māmalahoa Hi An associated drainage channel
 - The location(s) of proposed pu •
 - The proposed collection system The locations of three LCCs wh
 - •
- Community parcels currently community

Introduction

1.3 Environmental Setting

1.3.1 Natural Environment

This project area is situated on the southeastern slope of Mauna Loa, within the traditional land divisions of Kähilipalinui and Käwala. The project area ranges in elevation from approximately 195–247 m (640–810 ft) above mean sea level (AMSL). The town of Nä alehu receives an anual average rainfall of 1,062 mm (42 inches) (Giambelluca et al. 2013). The terrain in this area is characterized by 15–20% slopes to the southeast. The project area lands have been used stransvely for residential and commercial development, ranching, and agriculture which have significantly altered the natural environment.

According to the U.S. Department of Agriculture (USDA) Soil Survey Geographic (SSURGO) database (2001) and soil survey data gathered by Sato et al. (1973), the project area overlies soils from the Naalehu, Punaluu and Kaimu Series (Figure 9). The northern portion of the project area is situated on soils from the Naalehu series including Naalehu silty clay loam, 0%–10% slopes (NaC); Naalehu silty clay loam, 10%–20% slopes (NaD); Naalehu silty clay loam, 20%–35% slopes (NaC); and Naalehu very tocky silty clay loam, 6%–20% slopes (NaE); and Naalehu very tocky silty clay loam, 6%–20% slopes (NaE); and Naalehu very tocky silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are described as well-drained silty clay loam, 6%–20% slopes (Sae Figure 9). Naalehu series soils are used mostly for sugarcane or pasture (Sato et al. 1973:40).

The southwestern portion of the project area consists primarily of Punaluu extremely rocky peat, 6%-20% slopes (PYD), with smaller small portions overlapping areas of Kaimu extremely rocky peat 7%-20% (KED) and the miscellaneous land types phaloeboe (rLW), and Rock land (rRO) (see Figure 9). Punaluu and Kaimu series soils are described as well-drained, thin organic soils over lava bedrock (Sato et al. 1973:22, 48). Punaluu soils are used for pasture, and Kaimu soils are used for pasture, macadamia nuts, papaya, and cirrus fruits (Sato et al. 1973:22, 48). *Pathoehoe* lava flow is described as having "billowy, glass surface that is relatively smooth [...] Pathoehoe lava has no soil covering and is typically bare of vogetation except for mosses and lichnes" (Sato et al. 1973:32). Rock land is a miscellaneous land type that consists of *pinoehoe* lava bas and carcing and is upically bare of vogetation except for mosses and since bedrock covered by a thin layer of soil material. "The vegetation is confined mainly to the soil cover areas and cracks in the lava [...] Rock land is used for pasture, wildlife habitat, and watershed" (Sato et al. 1973:55).

Vegetation within the undeveloped portions of the project area consist primarily of guinea grass (*Megathyrsus maximus*), invasive vines, and lantana (*Lantana camara*) associated with use of these areas for pasture. A depression caused by a collapsed lava tube system bisects the proposed treatment facility portion of the project area in TMK: (3) 9-5-007:016. This depression trends east-west and contains several tree species including native *kubai* or candlenut (*Aleurites moluccanus*).

1.3.2 Built Environment

The collection system portion of the project area located *mauka* (upslope) of the Māmalahoa Highway comprises various active county residential roadways including 'Ohai Road, Kukui Road, Miol Road, Melia Street, Opukea Street, Nahele Street, Lokelani Street and Kilika Street (see Figure 7). It also includes two forcemain alignments across a concrete drainage channel west of the residential area, and associated infrastructure connecting to and along Ka'alaiki Road. The concrete drainage channel is fenced along its length and accessed from a gravel road off Ka'alaiki Road or from the western terminus of Melia Street. The three existing LCGs are located at TMKs:

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

12

Introduction

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	Section 2 Methods	2.1 Field Methods	CSH completed the fieldwork component of this AIS under archaeological permit numbers 2 07 and 21-10, issued by the SHPD pursuant to HAR §13-13-282. Fieldwork was conduct intermittently between 26 October 2020 and 29 January 2021 by CSH Archaeologist Samant Purdy, B.A., and Project Director Olivier M. Bautista, B.A., under the general supervision of Principal Investigator Hallett H. Hammatt, Ph.D. This work required approximately 19 person days to complete.	In general, fieldwork included 100% pedestrian inspection of the project area, GPS da collection, site documentation, and a program of subsurface testing.	A 100%-coverage pedestrian inspection of the project area was undertaken for the purpose in historic propenty identification and documentation. The pedestrian survey was accomplish through systematic sweeps spaced 5-10 m apart depending on the density of the vegetation are ground visibility.	Where a new historic property was encountered, the determination of its boundary was bass on factors including apparent age, architectural style, and the spatial and function interrelationships of both natural and man-made features.	At each historic property archaeologists recorded site locations using a Trimble Geo 7X ur (sub-meter accuracy), took digital photographs, and prepared a written description and plan vie maps of each site, or updated existing site descriptions and maps as applicable. Generally, pk view maps were not drawn for roads (SIHP #s-30187, -31271, -31272, -31273 or walls (-3127 -31276); instead, GPS points were taken at key locations and these sites and their routes we mapped in GIS for inclusion on an aerial image. A plan view map was also not prepared for th SIHP #-31274 drainage channet, as a copy of the original design plans was obtained for inclusio in this report.	2.1.2 Subsurface Testing	A program of subsurface testing was undertaken for the AIS in accordance with a testin strategy submitted to SHPD in October 2020 alongside materials initiating the project's histor preservation review (see Section 1.2). This testing program comprised both mechanical at manul accavation undertaken for two distinct purposes. Exploratory mechanical (i.e., "backhoe testing was conducted to assess the potential for subsurface archaeological features throughout th undeveloped portions of the project area, including but not limited to buried cultural deposi and/or culturally modified lava tubes. Hand excavation was completed at select historic proper features in an attempt to yield data to support assessments of site age and function.	2.1.2.1 Mechanical Excavation Methods	Mechanical excavation was conducted as proposed using a "mini" excavator to complete exploratory test trenches (TTs). A total of 11 trenches were excavated within the maj components of the treatment facility site (disposal groves, treatment lagoons, finishing wetlan	AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island
Introduction	These three parcels ences. The pronosed	 S) within a portion Igelsia Ni Cristo 	ad intersection and rcial establishments därätehu Spur Road Spur Road accesses tent facility will be	enced paddocks and								14
Cultural Surveys Hawai'i Job Code: KAHILIPALINUJ 6	(3) 9-5-024:011 (LCC 1), 9-5-024:010 (LCC 2), and 9-5-024:001 (LCC 3). have been mexionsly disturbed and the latter two narcels contain private reside	offection system may also include development of a sewer pump station (SF of TMK: (3) 9-5-008:048, which is bound to the south by residences and th	Church and has been used for pasture. The project area crosses the Mamalahoa Highway at the Ka'alaiki Ro; extends <i>makai</i> along Nā'ālehu Spur Road where it passes several commer including but not limited to the Nā'ālehu Shopping Center. The portion of N within the project area is paved. A dirt road extending south from Nā'ālehu ' Onipa'a Ranch and pasture lands in which the proposed wastewater treatm	located at TMK: (3) 9-5-007:016 (see Figure 2). These pasture lands contain fo other remnants of ranching activity.								AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island

Methods

stream channel diversion; Figure 10) and three trenches were excavated within the sewer pump station site (at the below grade pump station and force main corridor; Figure 11).

The center point of each test trench was recorded using Trimble sub-meter accurate GPS prior to excavation. In general, trenches measured approximately 5 m (16.5 ft) long and 1.0 m (3.3 ft) in width. All trenches were excavated to bedrock or the physical limitations of the backhoe, whichever was reached first.

The archaeologist carefully watched all machine excavations and inspected the soil sediments as they were removed from the trench and placed into an adjacent backfill pile. Mechanical excavations occurred one stratum or level at a time. If subsurface features or cultural layers were encountered, the machine was to cease excavation and excavation would proceed manually, expanding outward from the trench as necessary to expose the limits of the feature or cultural layer. However, no subsurface features or cultural layers were encountered. A stratigraphic profile of each test excavation was drawn and photographed. The observed sediments were described using standard USDA soil description observations/terminology. Soil sediment descriptions included Munsell color; texture; consistence; structure; plasticity; cementation; origin of sediments; descriptions of any inclusions, such as cultural material and/or roots; lower boundary distinctiveness and topography; and other general observations. Stratigraphic anomalies or potential cultural deposits were to be carefully represented on test excavation profile maps if encountered. If subsurface features or cultural deposits were nonutrered, plan view maps were to be completed at the top and/or base of each cultural layer, and at the top and base of each feature; and maps of the completed test units would include plotted locations of all points in each unit where GPS readings were taken. No such features were encountered.

Had any been encountered, all identified cultural materials and/or charcoal or soil samples would have been appropriately documented and described, and all collected materials catalogued, labeled, and delivered to the CSH laboratory for analysis.

Following documentation, the excavated materials were returned to their original provenience and the ground surface was restored as best as possible to its original condition.

2.1.2.2 Manual Excavation Methods

Manual or hand excavation was undertaken as applicable based on the findings of the pedestrian survey. Two 1.0-m-sq test units (TUs) were hand excavated at architectural features of -31270. The center point of each test unit was recorded using Trinble sub-meter accurate GPS prior to excavation. Both units were excavated to bedrock. Had subsurface features or cultural layers been encountered, the unit was to be expanded outward as necessary to expose the limits of the feature flayers were encountered. Soil-sedimentary deposits were removed by shovel, trowel, and whisk and dustpan. Excavations proceeded one stratum at a time, with internal levels added as needed (no internal level crossed a stratum boundary). All excavated soil-sediment was passed through a 1/8-inch mesh screen. All excavated architectural rock material and screened sediment was deposited in a mesh screen. All excavated architectural rock material and screened sediment was deposited in a and documentation. All identified cultural materials and/or charcoal or soil samples were and documentation. All identified cultural materials and/or charcoal or soil samples were

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Methods



Figure 10. Client provided map showing approximate proposed locations for 11 test excavation trenches at the proposed wastewater treatment facility site at TMK: (3) 9-5-007:016

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

16

٥
=
2
4
5
∢
2
=
≒
Э
-
ö
X
Э
ā
0
7
'n
ž
a
Ц
Ś
ъ
È
₽
ē
Ξ
Ξ
ី



Figure 11. Client provided map showing approximate proposed locations for three test excavation trenches at the proposed at the proposed sewer pump station site at TMK: (3) 9-5-008:048

AISR for the Nāʿālehu Cesspool Closure Project. Kābilipalinui and Kāwala, Kaʿū, Hawaiʻi Island TMKs: multiple

18

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Methods

appropriately documented and described, and collected materials were catalogued, labeled, and delivered to the CSH laboratory for analysis. All materials collected for analysis were collected during screening of excavated material and were not part of a distinct cultural feature or layer such as a hearth or trash pit. These materials were analyzed following the applicable methods described in Section 2.2.

The test units were documented following the methods described in Section 2.1.2.1. Following documentation, the excavated architectural materials and sediments were returned to their original provenience and the ground surface was restored as best as possible to its original condition.

Some cultural materials were collected from the screen during excavation of one of the two test units and returned to the laboratory for analysis.

2.2 Laboratory Methods

Materials collected during AIS fieldwork were identified and catalogued at CSH's laboratory facilities on Hawai'i Island. Analysis of collected materials was undertaken using standard archaeological laboratory techniques. Materials were washed, sorted, measured, weighed, described, and/or photographed (see Section 5). The analyzed materials included diagnostic (identifiable or datable) artifacts, and charcoal.

2.2.1 Artifact Analysis

In general, artifact analysis focused on establishing, to the greatest extent possible, material type, function, cultural affiliation, and age of manufacture. As applicable, artifacts were washed, sorted, measured, weighed, described, photographed, and catalogued. Diagnostic (dateable or identifiable) attributes of artifacts were researched.

The artifacts collected during excavation were types generally interpreted as traditional Hawaiian artifacts (an adze and a coral abrader). No historic-era artifactual material was encountered in the project area. The traditional Hawaiian artifactual material was identified, and forms and functions determined, using standard reference materials (e.g., Barrera and Kirch 1973; Brigham 1974; Buck 2003; Emory et al. 1968; and Graves and McElroy 2004). Analyzed materials were tabulated and are presented in Section 5.1.

2.2.2 Energy-Dispersive X-ray Fluorescence (EDXRF) Analysis

The use of energy-dispersive X-ray fluorescence (EDXRF) may make it possible to determine if lithic artifacts from within the project area are from similar sources, different sources, or from sources consistent with another island. Using an EDXRF spectrometer, Dr. Peter Mills of the University of Hawai'i at Hilo is working to establish EDXRF spectrometer, Dr. Peter Mills of the University of Hawai'i at Hilo is working to establish EDXRF spectrometer, Dr. Peter Mills of the University of Hawai'i at another risland. Using an EDXRF spectrometer, Dr. Peter Mills of the University of Hawai'i at an orbit of the stand spectrometer, Dr. Peter Mills of the University of Hawai'i at a Hilo is working to establish geochemical "fingerprints" of stone tools that Native Hawaiians quarried from various sites and track the extent to which that material was circulated on each island or throughout the island chain. The EDXRF analyzer allows archaeologists to conduct rapid and non-destructive analyses of stone artifacts to determine the extent and distance stone tools moved from the quarries. Attempts are made to match the lithic artifact samples with geochemical data collected on known prelistoric quarry areas, such as on Mauma Kea, and sources of volcanic glass, such as Pu'u Wa'awa'a located on the north slope of Huallali volcano on Hawai'i Island (MCOv et al. 2011). Samples that do not match known quarry sites may lead to the discovery of currently unknown quarry sites, or possibly to the identification of stone tools derived from other island groups such as Tahiti and the Marquess. By examining

AISR för the Nä*älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Methods	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Methods
vere derived from non-local sources, ovement of lithic artifacts through time ed over long distances of onen ocean.	arrangements with the landowner regarding the disposition of this material. Should the landowner request different archiving of material, an archive location will be determined in consultation with the SHPD. All data senerated during the course of the AIS are stored at the CSH offices.
, more information has become known	2.3 Research Methods
ss the Hawanan Islands. In support of ormation about artifacts found within a cpensive and non-destructive analysis	Background research included a review of previous archaeological studies on file at the SHPD; review of documents at Hamilton Library of the University of Hawai'i, the Hawai'i State Archives, the Mission Houses Museum Library, the Hawai'i Public Library, and the Bishop Museum
s was sent to the UHH Geoarchaeology gy beam of X-rays onto the sample and hen measured by an energy-dispersive	Archives; study of historic photographs at the Hawai'i State Archives and the Bishop Museum Archives; and study of historic maps at the Survey Office of the Department of Land and Natural Resources. Historic maps and photographs from the CSH library were also consulted. In addition, Mähele records were examined from the Waihona 'Aina database (Waihona 'Aina 2021).
y mito votage signals una are urgitarly), zirconium (zr), rubidium (Rb), and udy. The signals were plotted on Sr to m samples in the database. The results	This research provided the environmental, cultural, historic, and archaeological background for the project area. The sources studied were used to formulate a predictive model regarding the expected types and locations of cultural resources in the project area.
Dr Peter Mills. Following analysis, the	2.4 Consultation Methods
vations within the context of a historic s were sent to the Wood Identification esearch Institute, Inc. (IARII) for taxa under magnification of a dissecting tristics of known woods in the Pacific University of Hawai'i, as well as in oples provides useful information for oples provides useful information for alysis but was subsequently informed alysis but was subsequently informed alysis but was subsequently informed of a full analysis, CSH was provided a statied for radiocarbon dating, and the station for radiocarbon dating method. The conventional radiocarbon lendar ages using the INTCAL13 High ration program, Version 4.3, developed (ORAU).	NHPA Section 106 consultation with community members, agencies, and NHOs is being undertaken by the project proponents.
not include human remains or grave ice in Hilo Hawai'i. CSH will make	

the extent to which stone tools in various ahupua'a v archaeologists will be able to quantify Native Hawaiian m and space, and possibly identify some tools that were carri

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

As the UHH EDXRF database has expanded over time about specific sources for stone and volcanic glass acros furthering this research, and to gather potential source inf given project, CSH submits lithic samples for this inc whenever possible.

Lab for EDXRF analysis. The EDXRF focuses a high ener-uses atomic excitation to create X-ray emissions that are t The single lithic artifact collected during the current AI spectrometer. The spectrometer converts the X-ray energ, analyzed and plotted. The trace elements strontium (Sr niobium (Nb) were used as main discriminators for this st Zr and Rb to Nb bivariate plots in comparison to data fro were summarized in personal communication to CSH from artifact was returned to the CSH laboratory.

2.2.3 Wood Taxa Analysis

Laboratory (WIDL) at the International Archaeological R published descriptions. Taxa identification of wood san interpreting the environmental and cultural history of the pretime frame of land use. The identification can also be use to the WIDL for analysis and selection for radiocarbon ar Select samples of charcoal material from test unit exca property were analyzed for species identification. Sample identification. The samples are intended to be viewed microscope and then compared with anatomical characte Islands Wood Collection at the Department of Botany, material suitable for radiocarbon dating. In the case of this that no one was available to process the samples. In lieu brief, informal evaluation of which samples were likely be samples were returned to the CSH laboratory.

2.2.4 Radiocarbon Analysis

age determined by Beta Analytic, Inc. was calibrated to cal Probability Density Range Method (HPD) and OxCal calib Charcoal samples informally identified as short-lived p (see Section 2.2.3) were sent to Beta Analytic, Inc. of analysis using the Accelerator Mass Spectrometer (AMS) by the University of Oxford Radiocarbon Accelerator Uni

2.2.5 Disposition of Materials

goods) will remain temporarily curated at the CSH off Materials collected during the current AIS (which do

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

20

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Background Research	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Background Research
Section 3 Background Research	In contrast, Kahilipalinui contained "no settlements of prime importance," (Handy and Handy 1972:595).
3.1 Traditional and Historical Background 3.1.1 Traditional Background and Land Use	In the pre-Contact period, inter-district competition resulted in the shifting dominance of the <i>ali</i> ' <i>i</i> (chiefly class) of Ka'ū with the rulers of Puna and Kohala. The chief Kalani'opu'u ruled Ka'ū (and the entire island) during the eighteenth century just before the first European visitors began to record their early innerscins of the land and its neonle.
As u is the southernmost and largest district of riawait 1 island, known for its dynamic natural environment and fferce people. The project area is situated within the boundaries of two <i>dhupud</i> 'a, Kähilinalinui and Käwala Lorsted on the southern clones of Manna 1 oa and northesis of the	3.1.2 Early Historic Period
Kählippali literally translates as "[wind-] swept clift," while Kāwala translates as "to strike back."	Lt. James King of Captain Cook's expedition in 1779 recorded his impressions of this portion of the rocky Ka'ū coast as "a prospect of the most horrid and dreary kind" (King 1784:104). Regardless of his first impression, the rich fishing grounds on the coast and the inland plantations
Increase generating agreement that is not not the cartiest setted areas in the frawarian Islands, though there is less consensus regarding the timeframe of its initial settlement. Kirsch (1985:81–87) proposes settlement by the fourth of fifth century AD, and notes that Ka Lae would have been an attractive locale given its direct proximity to abundant deep-sea fishing grounds.	provided Hawainans with an abundance of resources, which supported a substantial population. Archibald Menzics (1920), a surgeon and naturalist on the 1794 voyage of Captain George Vancouver, and the Rev. William Ellis (1963), who traveled through Ka'ū in 1823, encountered productive and nonulous villaces at hearby Homivano, east of the proviect area and Wa'čhinu
Archaeological studies in the general region clearly indicate pre-Contact permanent habitation settlement along the coast as well as within inland portions of Ka Lae. Archaeological research conducted in the upland Ka Lae region (Cordy 1986; Cordy 1987; Spear and Rosendahl 1987; Tomonari-Tuggle and Tuggle 1991) signify a distinct inland settlement typically focusing on agricultural subsistence. An organized upland field system is known to have been present in Ka'ū but has not been investivated in any systemic way.	Protection and performent strugges at nearly 170 the first missionary to visit Ka'ū. During the west of the project area. Rev. Ellis may have been the first missionary to visit Ka'ū. During the 1830s Protestant missionary presence was not installed until the early 1840s when Catholic and Protestant missions were established in the district. In 1842, the Protestant minister John Paris reached Ka'alu'alu (located approximately 7.0 miles southwest at the coast) by schooner where he
Given the proximity of Wai'öhinu and Na'älehu to the inland Ka Lac region, and their relatively sheltered setting from lava flows, it is unsurprising that agriculture and associated settlement were firmly established in these places prior to European Contact. Radiocarbon dates from a lava tube site located in the upslope Wai'öhinu area indicate occupation between AD 1420 and 1655 (Robins et al. 1992). Handy and Handy (1972) describe pre-Contact Na'älehu as follows:	found, The shore was lined with hundreds of natives as our little boat neared the shore. I was taken up by a great strong native Samson, whose entire dress was a malo [male s loincloth] and who was iattooed from head to foot. He looked fietree but set me gently down on the pahochoe anil a crowd of natives Then came greetings from the multitude and some bisation whould and for work of my fear A
Na'alehu was a place of some importance in ancient times. There were two sayings that had reference to it. <i>Na'alehu i ka pala pa'a</i> ('Na'alehu of the thick-walled calabashes') means that folk of this locality were proudly thick skulled and	Totul the inductives, some existing inty nations and some taking note of my rece. A joyful 'Aloha inol' with a low wail, rose from the aged ones. [Paris 1926:89] Paris' account of the profusion offered in welcome by the Ka'ū natives illustrates the abundant resources available in the district:
stubborn even more so than the run of ka'u tolk. Another saying snow that they regarded themselves, or were regarded as, merciless fighters. <i>Na'alehu haehae poko</i> means 'Na'alehu tears to bits.' This saying comes down from the memory of a warrior who was so strong and so fierce that he tore his victims to bits with his bare hands.	two strong men, tattooed from head to foot, came in bearing a huge whole hog, baked entire minus hair and entrails. These bearers were followed by others, dressed in the same style bringing calabashes of various sizes filled with fish, poi, potatoes, then earne meloons, baranaas, and sugar came, and little gourds filled with
This was an area of sweet potatoes and gourds. There was a notable heiau, a <i>luakini</i> or war temple, the remains of which are in the rear of the present public-school site. About half a mile beyond this to seaward is a caved-in section of lava tube forming	goat's milk. All was spread out in royal Hawaiian style, a dozen kukuis [nuts from the Candlenut tree, <i>Aleutris moluccana</i>] burning and kahilis [feather standards] waving to and fro. [Paris 1926:90]
a sort of cave, on the wall of which are lightly drawn petroglyphs in the form of human figures. [Handy and Handy 1972:595–596]	Paris settled in Wai'ōhinu where he founded a church and school. Later, in 1843, a stone church was also built at Punalu'u to the northeast. Cordy (1986-21) postulates that around this time a settlement shift was occurring from costal to inland regions the result of denomlation and of
Handy and Handy (1972:359) note that Kāwala was considered "an <i>ahupua 'a</i> of importance because to it belonged to Na'alehu with its large population, rich planting areas; planting, healing and war <i>heiau</i> ; sports arena; a pavilion of the <i>ali</i> 'i in a grove of <i>kou</i> trees; and other distinctions."	efforts to gain access to the government road and to populate the economic center of Wai'öhinu. Mission station reports and censuses and accounts by visitors to Ka'ū during the mid-nineteenth century document the changes to the district brought about by natural forces and by the pressures
AISR for the Nāʿālehu Cesspool Closure Project, Kāħilīpalinui and Kāwala, Kaʿū, Hawaiʾi Island 22 TMKs: multiple	AISR for the Nā ⁻ ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka ⁻ ü, Hawai'i Island 23 TMKs: multiple

3.1 Traditional a

3.1.1 Traditional Back

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Background Research

of an increasing western presence. A visitor to Wai'öhinu and its environs in 1849 anonymously published an account describing the devastating effects of a drought and fire that had occurred three years earlier:

[W]e noticed many a tall, stately trunk, branchless and lifeless standing monumentlike, all over the country. On enquiry we ascertained that they were the remains of a noble forest, which, with the whole surrounding country, were burnt in 1846. In that year a severe drought visited the Island, the streams dried up, the grass withered, and fire swept over the whole district. [Sailor in Kelly 1980:89]

The author also describes an area above the settlement at Wai'oninu that, apparently undamaged by the 1846 fire, probably represents the idyllic setting that had drawn the Ka'u *ali'i* to the *ahupua'a*:

products of the temperate, with the rich and varied fruits of the tropic zones. But alas the farmers are wanting, the land lies in all the wild luxuriance of nature of about 5,000 feet were repaid with one of the richest scenes it was our privilege luxuriantly and attains the height of two or three feet. On this land we saw some noble upland kalo, and a number of very large banana trees. Several crystal springs take their rise on the summit, and might send, if rightly directed, a portion of their treasures through every man's fields. Behind this noble series of hills, timber desolate, there are no passable roads, except foot paths, to it, and no harbor at which [W]e ascended the hills back of the mission, and when we had reached an elevation to look upon. Below us lay, fashioned by the hand of nature, within a range of ten miles, six lovely terraces, on which one thousand dwellings might be placed, each of which should have a prospect of the sea, the rocky shore, the lava and the verdant upland. To each of these farms might be attached, of from 100 to 1500 acres of land, now lying utterly waste, that would repay bountifully the labor of the husbandman. The grass, with which most of the land was covered, grows abounds. So that there is to be found every thing desirable to make a rich farming country, and in a circuit of some fifteen miles, might be abundantly grown the best vessels could lie in safety, is found within many miles. [Sailor in Kelly 1980:89] Noticeably missing from this account is mention of any Hawaiians occupying and utilizing this verdant land "now lying utterly waste." Early census data provide insights into the dramatic shift in population between the 1830s and 1850s. Ant 1831-1832 census of Ka ü, the first taken within the district, records a total population of 5,800. In 1835 the total population is counted as 4,766. The first official government census, taken in 1847, records the population as having dropped to 3,010. Reversed John Paris work with in an 1848 mission station report (Paris 1848.3), a "distressing famine and fire which overran the country," the year 1845 the work of depopulation of Kau has gone on with fearful rapidity. "He notes, during the year 1845 the work of depopulation of Kau has gone on with fearful rapidity." He notes, during the year 1845 the work of depopulation of Kau has gone on with fearful rapidity. "He notes, during the year 1845 the work of depopulation of Kau has gone on with fearful rapidity." He notes, during the year 1845 the work of depopulation of Kau has gone on with fearful rapidity. "He notes, during the year 1845 the work of depopulation of Kau has gone on with fearful rapidity." He notes, during the years 1845 and 1846 (Paris 1848.3), a "distressing famine and fire which overran the country," the same disasters the anonymous visitor of 1849 mentioned. By the time of the 1853 government census onty 2,210 people are recorded in Kau. 100–150 people in the Nä*alehu vicinity. Despite a relatively low population of only about 100–150 people in the Nä*alehu vicinity. Despite a relatively low population of only about 100–150 people in the Nä*alehu vicinity. Despite a relatively low population density indicated in the vicinity of the project area, documents associated with the Mähele and subsequent Kuleana Act illustrate continued patterns of settlement and subsistence agriculture in the Na*alians.

AISR for the Nä ätehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 12. Map of Hawai'i Island showing population as of 1853 (Coulter 1931:28)

AISR for the Na'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

24

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Background Research

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

3.1.3 The Māhele and the Kuleana Act

In the mid-nineteenth century, during the time of Kamehameha III, a series of legal and legislative changes were brought about in the name of land reform (see the works of Jon Chinen 1958, 1971 for a thorough and well-written explanation). Before the Mähele, all land belonged to the *akua* (gods), held in trust for them by the paramount chief, and managed by subordinate chiefs.

Following the enactment of a series of new laws from the mid-1840s to mid-1850s, Kamehameha III divided the land into four categories: Crown Lands reserved for himself and the royal house; Government Lands; Konohiki Lands claimed by *ali* i and their *konohiki* (supervisors); and *kuleana*, small plots claimed by the *mada* claimed by *ali* i and their *konohiki* (supervisors); and *kuleana*, small plots claimed by the *mada* claimed to the claimet and vitnesses. A Royal Patent (RP), which quitclaimed the government's interest in the land, was sinceess. A Royal Patent (RP), which quitclaimed the government's interest in the land, was insuces on most Land Commission Awards (Chinen 1958:14). In some cases, more than one RP number was issued for an LCA, especially in cases where there were several widely separated *Taprana* (lots), such as an award with agricultural land in one *ahupua* 'a and a house lot in another. *Ali'i* were required to pay a commutation fee to the government for their confirmed Konohiki Land titles; this payment could be in cash or in the return of land to the government or crown. Many *ali'i* elected to return substantial portions of their awarded lands to avoid the one-third commutation cash fee. The Kuleana Act of 1850 allowed *maka Tänara*, in principle, to own land parcels where they were currently and actively cultivating and/or residing. In 1851, certain Government Lands became known as Royal Patent Grants or Land Grants.

LCA records provide insight into former land use within the project area and vicinity. The *ali'i* William Pitt Lelei'ohoku during the Mähele of 1848 retained the *ahupua'a* of Kahilipalinui as part of LCA 9971. Under the subsequent Kuleana Act a number of *kuleana* or commoner claims were also made within Kahilipalinui, but none were awarded in the vicinity of the project area. At the time of the Mähele, Kāwala Ahupua'a was not delineated as it is today. Rechtman (2011) notes

[...] there were at least four smaller *ahupua* 'a (Aemalo, Poupouwela, Papaikou, and Paukui; from west to east) surrounded by the larger Kāwala and Kaunāmano ahupua'a [Kaunāmano is to the east of Kāwala]. The names of these smaller *ahupua* 'a do not appear on later maps, and it appears as though they were subsumed by the larger Kaunāmano and Kāwala ahupua'a. [Rechtman 2011:19]

Over two dozen *kuleana* were awarded within Aemalo, Poupouwela, Papaikou, Paukui, and Kāwala (which typically appears in the records using the alternative spelling "Kowala"). Of these awards, ten are indicated in direct proximity to the project area; these awards along with LCA 9971 to Lelei'ohoku are depicted in Figure 13 and summarized in Table 1 (which provides separate entries for multiple '*apana* or lots belonging to the same award). Seven of the 12 proximate *kuleana* overlap the project area; these are fisted in Table 2 along with the forms of land use stressed for each in the associated claim records. Land use within the project area as expressed in the Mähele data consisted of subsistence agriculture; indicated crops include *kalo* (taro, *Colocasia esculenta*), *wauke* (paper mulberry, *Brousonetia papyrifera*), and banana (*Maia* sp.).

Subsequent to the Mähele many Land Grants were made throughout Nä'älehu; unfortunately. Land Grant records rarely contain specific land use information.

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka ù, Hawai'i Island TMKs: multiple

26



Figure 13. Portion of the 1995 Naalehu USGS 7.5-minute topographic quadrangle showing the location of LCAs within and in the immediate vicinity of the project area

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ū, Hawai'i Island TMKs: multiple

7314 8786:1 8786:2 8787B:2	Awardee Ku Kuaana Kuaana 2 Kupele	Ahupua'a Poupouwela [*Kāwala] Paukui, Poupouwela, Papa [*Kāwala] Paukui, Poupouwela, Papa [*Kāwala] Papaikou, Kaumāmano, Pa	uikou ukui ukui	Royal Patent # 7326 5313 5313 5313 3204	Acreage 13.9 3.4 5.26 2.6
8787B:	3 Kupele Kikoopua	Papaikou, Kaunāmano, Pa [*Kāwala] Poupouwela, Aemalo, Kov [*Kāwala]	ukui <i>w</i> ala	3204 6822	4.8 6.14
9112:1 9225B	Kahana Keawe	Poupouwela, Kowala [*Ka Poupouwela [*Kāwala]	āwala]	4169 7971	6.0 16.4
9971:10	D Leleiohoku,William Pitt(ali 'i award)	Kahilipalinui		6882	2,155
10310	Nunanuna	Poupouwela, Kahilipali, K [*Kāwala]	aunamano	7927	12.6
10304:1	l Nauka	Poupouwela, Aemalo [*Ki	āwala]	6660	5.0
10886	Nuinuimakua	Poupouwela [*Kāwala]		6455	7.18
11028	Peke, Samula	Poupouwela, Aemalo [*K	āwala]	7685	6.55
*Der Table 2.	notes post-Māhele Land use within <i>k</i>	ahupua'a name designation aleana awards overlapping	the project i	urca	
LCA #	Awardee	Date Awarded	Land Use		
7314	Ku	1848	[none give	[u	
8786:2	Kuaana	1848	Three kihi	pai [garden, culti-	vated patch]
8606	Kikoopua	1849	One taro, (me wauke, one fal	low
9225B	Keawe	1848	[none give	[u	
10304:1	l Nauka	1875	One (3) tai	0.	
10310	Nunanuna	1890	Four taro,	one banana	
10886	Nuinuimak	ua 1872	Six taro		

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Background Research

Ka[']ū, when Princess Ruth aimea (Hansen and Kelly the 1920s. Organized cattle material taken from the pillows; ultimately these):17). Other shorter-lived for export of agriculture ay and Wai'öhinu (Kelly ÷ hift in populations from the aartner John Costa founded village in Nā^{*}älehu ensued th only 7.5 acres of planted n fluctuates in the district tion 1990:2). Due to its

An 1886 map (Figure 14) pany mills at Nā'ālehu and cated in the vicinity of the Cane Haul Road"), both of o establish the Hilea Sugar organ 2000:108). In 1877 nd by the end of the 1870s nd the road heading south : and Morgan 2000:108– chinson Sugar Company, ū under the ownership of location of the mill and oirs, and animal pens.

wed by decommissioning tralized mill was built at Company (Dorrance and the cheapest sugar ports

of schools and post offices north and northwest of the project area. Figure 16 also illustrates the approximate boundaries of sugar plantation lands (in red) in relation to the forest lands *mauka* (in blue) and grazing lands west (in yellow) associated with Ka'alu'alu Ranch. This ranch was twentieth century. The the town; this area is still e 16) depicts the location

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

29







Figure 14. Portion of W.A. Wall 1886 map of Hawai'i Island showing the project area in relation to sugar mills, the landing at Honu'apo, and established transportation routes

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

30



AISR for the Mä älehu Cesspool Closure Project, Kähilipulinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Background Research	purchased by the Hutchinson Sugar Company in 1936; the ranch and a small dairy were "operated in conjunction with the plantation" (Hawaiian Sugar Planters' Association 1990:3). A 1921 U.S. Geological Survey (USGS) map (Figure 17) shows the locations of various workers camps above the town; it also includes the route of the "Hutchinson Plantation RR [Rairoad]", crossing the Volcano Road (the preusor to the Māmalahoa Highway). Notable is the absence on Figure 16 and Figure 17 of a clearly marked route following that of the upper cane haul road (Ka'alaiki Road), indicating this route may have been falling out of favor for general travel.	Nā'älehu School (also identified as Nā'älehu Elementary School) was established at its current location along the Māmalahoa Highway in 1928. The school is listed on the Hawai'i Register of Historic Places under the thematic group "Public Schools on the Island of Hawai'i" (State Inventory of Historic Places [SIHP] # 50-10-74-7522). A national register nomination form was completed for the school by Tonia Moy of SHPD in 2002.	Beginning the in 1920s flumes were utilized to transport cane to the railroad servicing the Honu'apo Mill and relied on a system of tunnels and reservoirs "dug 3,000 to 8,000 feet into the mountain," (Dorrance and Morgan 2000:109). These flumes were used into the 1940s when transport shifted completely over to trucks. The Hawaiian Agricultural Company located in Pähala to the east utilized a narrow-gauge railway to transport its milled sugar to Honu'apo for export. The Honu'apo landing was closed in 1942 due to the war.	In the 1940s the Māmalahoa Highway (Route 11) was constructed through Ka'ū along the former Volcano Road alignment. Portions of this roadway located around the island have been designated as SIHP $\#$ 50-10-47-30187. The Māmalahoa Highway is visible on the 1962 topographic map (Figure 18) as well as the cane haul road (Ka'alaiki Road), along which an airstrip is visible northeast of town. Figure 18 also depicts along the eastern side of town a driveway extending north from the highway toward a structure features are situated with what may have been at the time a new dairy. These driveway and structure features are situated within the parcel containing a proposed sever pump station (TMK: (3) 9-5-008:048); tax records indicate a milk parlor was constructed on the property in 1960. Figure 18 indicates that by the early 1960s residential development in Na'alehu town appeared much as it does today.	The concrete drainage channel and an associated debris basin located between Ka'alaiki Road and the residential neighborhood were completed in 1965. This curved feature is visible on a 1978 aerial photo (Figure 19), which also depicts further residential development south of the Mämalahoa Highway. 3.1.6 Contemporary Land Use	In 1971 Hutchinson Sugar Company was merged into the Hawaiian Agricultural Company to form the Kau Sugar Company, Inc., which would be renamed as the Kau Agribusiness Company in 1986 (Dorrance and Morgan 2000:112). Following the demise of the sugar industry in other parts of the island, Kau Agribusiness Company cesed its sugar operations in 1996 (Dorrance and Morgan 2000:112). Nai alenu continues to serve a rural population supported by predominately agricultural and livestock economics. The town is also used as a stor-over for fourists visiting	Punalu'u Beach located 8 miles northeast. South Point located 15 miles southwest, and/or travelling between Hilo and Kailua-Kona.
Background Research	CA P	Honu		Public Lada. Bonerad Statement: Track. Approximate Area of Graturg Landa. Approximate Area of Disarging Landa.	 Porter Reserves. Approximate Acts of Forent Lands soft in Neucres. Expressimate Acts of Wor Lands (Nice and Tras). Exbands. Port Offices. 	Scale 0 0.75 1.5 Kilometers N 0 0.5 1 Miles Cultural Surveys Traven's Inc.	wing the project area within
ultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	A O A			PALL PALL	ILLIPA AHAR U	Legend Project Arca Base Map. 1906 Donn Hawaii Tarritory Survey Map of Hawaii Island, with Land Use (RM 2060) Data Sources: CSH	'igure 16. Portion of J.M. Donn 1906 map of Hawai'i Island shov sugar plantation lands

TMKs: multiple

TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



0



AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

35

A CONTRACT

K

Figure 17. Portion of the 1921 Naalehu USGS 7.5-minute topographic quadrangle showing the project area

Cultural Surveys I Jawari, Inc.

Base Map: USGS Topographic Map Naalehu (1921) Data Sources: CSH

Project Area

Legend

600 Meter 2,000 Fee

Scale

1,000

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

34

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Background Research

Cultural Surveys Hawai' Job Code: KAHILIPALINUI 6 Background Research	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Background Research
	On 16 December 2019, the County of Hawai'i, State of Hawai'i Department of Land and Natural Resources' Legacy Land Conservation Program (LLCP), Ka'ti Mahi LLC, and The Trust for Public Land completed the sale and purchase of 2,317.844 acres of land called Waikapuna located at TMK: 9-5-007:016 adjacent to Na [*] älehu. A portion of funds for the transaction were provided by the County of Hawai'I's Public Access. Open Space, and Natural Resources Preservation (PONC) program. The Ala Rhakai Trail Association (ATA) is the fee simple owner of the narcol (TDK- 0, 2,070.1017). The transaction encumbered the maiority of land with a
	perpetual conservation execution of the function of the land to agricultural and cultural preservation uses. This is the first conservation easement purchased by the County under the PONC program. An area of approximately 29 acres in the <i>mauka</i> portion of the parcel was excluded from the conservation easement to site the Nä ^a alebu treatment and disposal facility.
PE	3.2 Previous Archaeological Research Five previous archaeological studies have been identified within or in the immediate vicinity of the current project area. These studies and the historic properties documented therein are depicted in relation to the project area in Figure 20 and summarized in Table 3.
Manatrool Hoy	In 2005 Haun and Associates conducted an archaeological inventory survey (AIS) of an approximately 1.0-acre area located southeast of the current project area (Haun and Henry 2005; see Figure 20). Extensive prior impacts by historic and modern ranching activity were observed. No archaeological features were documented, and no further work was recommended. Based on the lack of findings the study was termed an "archaeological assessment" pursuant to HAR §13-284-5(5[A]).
	In 2006 CSH undertook a literature review and field inspection (LRFI) for two Ka'ū District schools as part of a state-wide Department of Education (DOE) wastewater systems improvement project. One of the two schools was Nāʿalehu School, located across the Māmalahoa Highway from the southeastern portion of the project area (Hammatt and Shideler 2006; see Figure 20). The LFRI included data for LCAs in the vicinity of the school and noted that Nāʿalehu School is listed on the Hawai'i Register of Historic Places. An on-site archaeological monitoring program was recommended for the project.
Learnd Project Area	In 2008 Rechtman Consulting conducted an AIS of 42.5 acres overlapping the western portion of the current project area (Clark et al. 2008; see Figure 20). The study documented three newly identified historic properties associated with historic-era ranching and sugar plantation activities: SIHP #s 50-10-74-26408 (complex of stone and concrete lined trenches representing possible latrines). SIHP # -26409 (concrete water trough), and -26410 (core-filled boundary wall) (see Figure 20 and Figure 21). No further work was recommended for these sites. Of the documented sites, SIHP # -26409 appears closest to the current project area but does not overlap it.
0 500 1,000 Feet 1 Base Map: USGS Ontrophotoqued Naslehu (1978) Data Sources: CSH	In 2009 CSH undertook archaeological monitoring for the DOE wastewater improvements project at Ná'alehu School (Wilkinson et al. 2009; see Figure 20). No cultural materials were exposed during project excavations.
Figure 19. Portion of the 1978 USGS orthophotoquad aerial photo, Naalehu Quadrangle, showing the project area	In 2011 Rechtman Consulting completed an AIS of 2.24 acres located across the Māmalahoa Highway from the collection system portion of the current project area (Rechtman 2011; see Figure 20). The study documented six newly identified historic properties associated with historic land use: SIHP #s 50-10-74-28925 (post-1938 residential complex); -28926, -28927, and -28990
AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple	AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ü, Hawai'i Island TMK:s: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

29634 4 Hammatt & Shideler 2006 Wilkinson et al. 2009 - 29532 Rechtman 2011 28928 Haun & Henry 2005 **300 Meters** - 29536 1,000 Feet Clark et al. 2013 29533 28990 29528 29531 - 29513 29527 150 500 Scale 29530 26883 28925 26883 -29512 28926 - 29505 28927 -Nije Ba Krikni Ba Weis Si 29506 -29511 ---R Kaalaiki R Legend Project Area Archaeological Study Area Historic Property 26409 Clark et al. 2008

Figure 20. Aerial photograph (ESRI 2022) showing previous archaeological studies conducted and historic properties documented in the vicinity of the project area

Base Map: ESRI Aerial Imagery 2022 Data Sources: CSH

Cultural Surveys Hawarii, Inc.

AISR for the Nä ätehu Cesspool Closure Project, Kähilipalinui and Kāwala, Kaʿū, Hawaiʻi Island TMKs: multiple

38

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Background Research

Table 3. Previous	archaeological studie	es in the vicinity of th	e project area
Reference	Type of Study	Location	Results
Haun and Henry 2005	Archaeological inventory survey	1 acre in Kaunāmano Ahupua'a; TMK: (3) 9-5-011:001 por.	No historic properties identified; noted extensive prior impacts throughout project area
Hammatt and Shideler 2006	Archaeological literature review and field inspection	Nā`ālehu School; TMKs: (3) 9-5- 009:00-6, 015	Recommended on-site archaeological monitoring
Clark et al. 2008	Archaeological inventory survey	42.5 acres in Kāhilipalinui and Kāwala Ahupua'a; TMKs: (3) 9-5- 021:015 and 9-5- 022:001	Documented three newly identified historic properties: SIHP #\$ 50-10- 74-26408, -26400, and -26410, all associated with historic period land use; none of these historic properties appear to be located within the current project area
Wilkinson et al. 2009	Archaeological monitoring	Nā'ālehu School; TMKs: (3) 9-5- 009:006, 015	No cultural resources documented during monitoring
Rechtman 2011	Archaeological inventory survey	2.24 acres in Kāwala Ahupua'a; TMK: (3) 9-5- 009:003	Documented six newly identified historic properties: SIHP #s 50-10- 74-28925, -28926, -28927, -28928 and -28990
Clark et al. 2013	Archaeological inventory survey	1,044 acres in Kāhilipalinui and Kāwala Ahupua'a; TMK: (3) 9-5- 010:001	Documented 204 historic properties of which 12 were previously documented and 192 were newly array of pre-Contact and historic era features used for including habitation, windbreaks and shelters, burials, ceremony, petroglyph, <i>papamī</i> , agriculture, transportation, and ranching; in closest proximity to current project are ane SHIP #s 50-10-74-29505, historic wall, and 2-9507, historic ranching complex

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	(three discrete mid-nineteenth century stone walls); -28928 (core-fill-w-28929 (complex of historic rock clearing mounds) (see Figure 20). Arch, was recommended for initial project ground disturbance. In 2013 Rechtman Consulting completed an AIS of 1,044 acres locat 010:001 in Känhilpalinui and Käwala Alupua'a, bounding the proposed tt portion of the current project area to the east (Clark et al. 2013; see Figure 22). Contact and historic era features were recorded that functioned a habita shelters, burials, ceremony, petroglyph, <i>paquan</i> , agriculture, transportatio northwestem portion of the 2013 study area abuts the current project area; at (SIHP \pm 50-10-74-29507) was documented overlapping with parcel 016 in tranch wall (SIHP \pm 50-10-74-29507) was documented overlapping with parcel 016 in tranch wall (SIHP \pm 50-10-74-29507) was also recorded in the vicinity (s other sites recorded by Clark et al. (2013) are well away from the current project area; at (SIHP \pm 50-10-74-29507) was also recorded in the vicinity (s other sites recorded by Clark et al. (2013) are well away from the current project area; at throughout Walkapuna; of these, three are indicated to overlapping the parcel of 10 in tranch wall (SIHP \pm 50-10-74-29507) was also recorded in the vicinity (s other sites recorded by Clark et al. (2013) are well away from the current project area; the vicinity (s other sites recorded by Clark et al. (2013) are well away from the current project area; the parcel at TMK: (3) 9-5-007/016 (Gasin) and Clark 2018. Novelapping the parcel at TMK: (3) 9-5-007/016 (Gasin) and Clark 2018, werdapping the parcel at the reductive of these, three are indicated to overlapping the parcel at the reductive of these, three are indicated to overlap the current 50-10-74-29507 ranch complex also documented by Clark et al. 2013; -309.	Witapuna materiary materiary transmits compared wave materiary and accidition, has graciously provies ite description information for SIHP #s -30929 and -30930 for reference it ac description in the project area is within the district of Ka ⁺ on Hawai ⁺ I sland, which is one of the earliest settled areas in the Hawaiian archipelago. Despite its lo const, in the pre-Contact period Na ⁺ älehu supported a considerable populati of agricultural fields. The eastern side of present Na ⁺ älehu town, including he the focus of this settlement and also contained several types of <i>heiau</i> and Kahilipalimu to the east was likely utilized more for outlying agricultural fields. The eastern side of present Na ⁺ älehu town, including he the focus of this settlement and also contained several types of <i>heiau</i> and Kahilipalimu to the east was likely utilized more for outlying agricultural fields. The eastern side of present Na ⁺ älehu town including he the focus of this settlement and also contained several types of <i>heiau</i> and Kahilipalimu to the east was likely utilized more for outlying agricultural fields. The asset is the contained several types of <i>heiau</i> and Kahilipalimu to the east was likely utilized more for outlying agricultural fields in the Mächele, Kahilipalimu was retained as Konohiki Land, and no LCA vicinity of the project area. In contrast, the LCAs surround and in some cas of the project area within what is presently known as Kāwala Ahupua'a. Sugar plantations proved to be the formative industry in the Nä ⁺ älehu ar Nä ⁺ älehu in the 1860s, somewhat upslope from the current town center.
Research		qoen

TMKs: multiple

TMKs: multiple

	٥
5	Ξ
2	z
	-
2	9
- 5	3
- 6	₽
- 2	2
1	4
	<u>2</u>
	ğ
1	2
	ğ
1	-
	5
	S,
	Ξ.
	ŝ
	ž
¢	2
-	d,
	Ξ
1	E
- 6	- 1



area boundary

CEGEND

Existing fence line Archaeological sit opening

Existing road

Existing building

ini i

29510

00263 71567

0 10: 0

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Background Research

(Route 11) was constructed through Ka'ū along the former Volcano Road alignment. The town continued to grow along the highway into the 1960s, which is also when the large concrete drainage channel was completed. By this time sugar production in Ka'ū was winding down. The In the 1920s Na'alehu School was established, and in the 1940s the Māmalahoa Highway town continues to support an agrarian economy and is a popular tourist stopover.

Subsurface extent of lava tube

Archaeological featu

lava tube

Bedrock sink/depression

0

Historic roadway

Over a century of sugar cultivation, ranching, and town development have heavily impacted the pre-Contact environment in and around Nā'ālehu. Previous archaeological investigations in the vicinity of the project area have identified scattered and disturbed remnants of pre-Contact land use amongst historic-era habitation, agriculture, and ranching sites.

including an undocumented segment of Māmalahoa Highway (SIHP # 50-10-47-30187) and Ka'ala'iki Road, also known as Cane Haul Road. The concrete flood drainage channel crossed by the project area between Ka'ala'iki Road and Maile Street was built in 1965 and is therefore a The proposed wastewater treatment facility and sewer pump station sites are currently used for pasture and may contain remnants of ranching activities or other historic land use. A prior AIS (Clark et al. 2013) documented historic ranch features on the parcel immediately adjacent to the 29507) is understood to overlap with the current project area. Another prior AIS (Gastilo and Clark 2018) was conducted in TMK: (3) 9-5-007:016 and overlaps the current proposed wastewater treatment facility area; while some of the sites documented during this study are indicated in the current project area, the AIS report was not finalized with SHPD. The proposed wastewater treatment facility and sewer pump station sites may also contain pre-Contact features located proposed wastewater treatment facility, and a portion of one of these sites (SIHP # 50-10-47-The project area extends along or crosses at least two significant historic roadway alignments, historic property. Other structures within the project area may also be greater than 50 years old. within lava tubes or other areas not disturbed by agricultural or ranching activities.

65

12202

29613s

29619

- Inot - 52567

0819

18893

29559 29562 29560, B

89565

29202

29558 295

••)29551

153554

29553

B

64

0548

29515 29516 29516 29519

THEAT

30

29514

1067 11962

1.5.1

29637 ()

El al 29680

81962

29689

65

29690

29648

29636

590

\$29633

\$23622

9653 \$29625

500

400

100 200 300

-

Acters

AISR for the Na'alehu Cespool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

Figure 22. Site distribution map from Clark et al. (2013:123) showing the locations of documented historic properties

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

4

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

Section 4 Results of Fieldwork

Fieldwork was conducted between 26 October 2020 and 29 January 2021 by CSH archaeologist Samantha Purdy, B.A., and Project Director Olivier M. Bautista, B.A., under the general supervision of Principal Investigator Hallett H. Hammatt, Ph.D. This work required approximately 19 person-days to complete.

Fieldwork consisted of a 100% pedestrian of the project area and a program of subsurface testing. The results of the pedestrian survey are summarized in Section 4.1, and the results of the subsurface testing program are provided in Section 4.2.

4.1 Pedestrian Inspection Results

The pedestrian survey was accomplished with the two archaeologists walking transects throughout the project area spaced from 10 to 25 m apart depending on ground visibility. Overall, ground visibility was excellent, exhibited primarily by previously developed areas like roadways (Figure 23 through Figure 27) or by wide open pastures of lantana and guinea grass (Figure 28 through Figure 37). Or by wide open pastures of lantana and guinea grass (Figure 28 through Figure 36). A depression caused by a collapsed lava tube system within the proposed treatment facility portion of the project area in TMK: (3) 9-5-007:016 was carefully explored for lava tube openings and surface archaeological features (Figure 37). One lava tube opening was encountered within the depression but found to be culturally sterile (Figure 38). The three existing whrough Figure 44).

A total of eleven historic properties were documented during the survey, including portions or the entirety of four previously documented sites (SIHP #s -29507, ranching complex; -30187, Mamalahoa Highway; -30929, pre-Contact and historic complex; and -30930, historic ranch wall) and seven newly recorded sites assigned as SIHP #s -31268 through -31274 (Table 4). The distribution of these sites is shown in Figure 45 and Figure 46. Ten of the 11 documented sites are dated firmly within the historic era; the remaining site, SIHP # -30292, is a pre-Contact and historic complex identified within the natural depression bisecting the proposed treatment facility site. The remainder of the project area has been extensively impacted by development and/or historic ranching or agricultural endeavors, likely accounting for the lack of additional pre-Contact faures outside of the depression. The ten historic-era sites are associated with transportation (SIHP #s -30187, -31270, -31271, -31272), ranching (SIHP #s, -29507, -30390, -31269), water control (SIHP #s -31268, -31272), and y (SIHP #s, -29507, -30390, -31269), water control (SIHP #s -31268, -31272), and y (SIHP #s, -31274).

The historic properties identified within the project area were documented as described in Section 2.1.1. Detailed site descriptions are provided in Section 6. Manual test excavations were completed following the methods outlined in Section 2.1.2.2 at two features of SIHP # -30929. These manual test excavations were completed in an effort to yield data supporting assessments of site age and function (see Section 1.4.2.2). Materials collected from these test excavations were analyzed at the CSH laboratory in Hilo, Hawai't, the results of the lab analysis are provided in Section 5. No human remains or grave goods were encountered anywhere within the project area.

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

4

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 23. Photograph looking down Kukui Road within the collection system portion of the project area; view to south



Figure 24. Photograph of a section of Milo Street within the collection system portion of the project area; view to east

AISR for the Näʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawai'i Island TMK s: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 25. Photograph overlooking the Nahele Street and Ohai Road intersection within the collection system portion of the project area; view to east



Figure 26. Photograph overlooking the Opukea Street and Kukui Road intersection within the collection system portion of the project area; view to east

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

46

Figure 27. Photograph showing a portion of Ohai Road within the collection system portion of the project area; view to north



Figure 28. Photograph overlooking a portion of the northern-most spur of the collection system above the Nā'ālehu town; evidence of cattle grazing was observed in this area; view to west

AISR for the Nā ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Results of Fieldwork



Figure 29. Photograph overlooking a portion of the northern-most spur of the collection system above the Na'alehu town; evidence of cattle grazing was observed in this area; view to north



Figure 30. Photograph overlooking the proposed pump station portion of the project area, at the site of the old Nāʿālehu Dairy; view to southwest

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

48

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork





Figure 31. Photograph overlooking the proposed pump station portion of the project area, at the site of the old $N\bar{a}$ alehu Dairy; view to north



Figure 32. Photo overlooking the portion of the project area connecting the proposed sewer pump station site to the collection system along Ohai Road; view to northeast

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Results of Fieldwork



Figure 33. Photograph overlooking a portion of the proposed treatment facility site *makai* of Māmalahoa Highway; view to north



Figure 34. Photograph overlooking a portion of the proposed treatment facility site makai of Māmalahoa Highway; view to east

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

50

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 35. Photograph overlooking a portion of the proposed treatment facility site *makai* of Māmalahoa Highway; view to west



Figure 36. Photograph of cattle grazing in the proposed treatment facility site, in the SIHP # - 31269 vicinity; view to east

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 37. Photograph taken within the natural depression bisecting the proposed treatment facility portion of the project area; view to west



Figure 38. Photograph showing the opening to a culturally sterile lava tube within the natural depression bisecting the proposed treatment facility portion of the project area; view to northwest

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

52

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 39. Photograph showing the vicinity of LCC 1 at TMK: (3) 9-5-024:011 along Opukea Street; view to southeast



Figure 40. Photograph of access to LCC 1 at TMK: (3) 9-5-024:011; view to southeast

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

12 Mar 1

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 41. Photograph showing the vicinity of LCC 2 at TMK: (3) 9-5-024:010 along Opukea Street; view to southwest



Figure 42. Photograph showing the access to LCC 2 at TMK: (3) 9-5-024:010; view to southwest

AISR for the Nā alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

54

Figure 43. Photograph showing the vicinity of LCC 3 at TMK: (3) 9-5-024:001 adjacent to the Māmalahoa Highway and Ohai Road intersection; view to northwest



Figure 44. Photograph of access to LCC 3 at TMK: (3) 9-5-024:001; view to southwest

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 45. Aerial imagery (ESRI 2017-18) showing the location of historic properties in the project area

AISR for the Nä ätehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

56



Figure 46. Aerial image showing the historic property and features identified within the southern portion of the project area

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ū, Hawai'i Island TMKs: multiple
Table 4. Hist	oric properti	es documented wi	ithin the project	t area		4.2 Subsurface Testing Results
SIHP # (50-10-74)	CSH Site∮	# Formal Type	Number of Features	Function	Age	A subsurface testing program was undertaken comprising mechanical and manual excaval Exploratory mechanical (i.e., "backhoe") testing was conducted to assess the potential
-29507		Complex	7 (3 in current project area)	Ranching	Historic	subsurface archaeological features throughout the undeveloped portions of the project a including but not limited to buried cultural deposits and/or culturally modified lava tubes. locations of all project test excavations (mechanical test trenches or "TT" and manual test uni
[47]-30187		Māmalahoa Highway	1	Transportation	Historic	"TU") are depicted on Figure 47. The results of the mechanical testing are provided in Sec 4.2.1. Hand exervation was completed at -31270 in the proposed watewater treatment facility is 2.2. Hence to a summer to summer to summer the summer provided and the summer provided and the summer testing the summer provided and the summer testing tes
-30929	CSH 3	Complex	Э	Activity Area, Ranching	Pre-Contact/ Historic	in an autempt to yterd data to support assessments of sue age and function; triese results presented in Section 4.2.2.
-30930	CSH 8	Wall	1	Ranching	Historic	4.2.1 Mechanical Exploratory Trenching
-31268	CSH 1	Drainage Ditch	1	Water Control	Historic	The exploratory testing program involved the completion of 14 mechanically assi evaluratory test trenches (TT) distributed throughout the undevalued notions of the moviest to
-31269	CSH 2	Complex	4	Ranching	Historic	Eleven of the 14 completed trenches were excavated in the proposed wastewater treatment fac
-31270	CSH 4	Jeep Road	1	Transportation	Historic	site at TMK: (3) 9-5-007:016, and the remaining three trenches were excavated within
-31271	CSH 5	Nā'ālehu Spur Road	1	Transportation	Historic	proposed pump station site at TMK: (3) 9-5-008:048 (see Figure 47). The test trenches were excavated using a compact excavator to either bedrock or the mechan
-31272	CSH 6	Ka'alaiki Road	1	Transportation	Historic	limits of the machine, whichever was encountered first. The trenches measured approximate
-31273	CSH 7	Complex	2	Water Control	Historic	m (16.5 ft) long and 1 m (3.2 ft) wide. Bedrock was encountered at all but one trench (T) when a daths of each
-31274	CSH 9	Wall	1	Boundary	Historic	whete southents exceeded the react of excavator. In the outer 1.5 treatment facility not ranged from annoximately 20 cm below surface (cmbs) in the nonosed treatment facility not
		-				of the project area to approximately 100 cm bars in the proposed pump station portion of the pro- area. No new historic properties were encountered during the exploratory testing. A modern can was observed at approximately 40–60 cmbs (in Stratum I) in TT-13; other than this mo trash, no cultural materials were observed in any of the exploratory test trenches.
						The exploratory testing program generally revealed two distinct areas of sedimentary deposi within the project area. Sediments exposed during excavation of TTs -1–11 in the prop
						wastewater treatment facility portion of the project area <i>maka</i> i of Mämalahoa Highway were as expected. USDA soil survey data indicates this portion or the project area is characterize
						shallow, dark-colored peats belonging to the Punaluu and Kaimu series, and an area of rock in the contheastern nortion of the nronoced treatment facility site (see Section 1.3.1 and Fiorn
						While three units contain possible layers of natural peat sediments situated directly atop by
						bedrock (TTs -2, -7, and -10), all 11 trenches in this area contained substantial layers of yellow
						hued silty clay loams. The two most likely explanations for the widespread presence of vellowish silty clay hown through the area are that they accumulated across the area in strum run
						potentially originating from existing drainage channels (e.g., SIHP # -31268), or that the
						sediments formed naturally in volcanic ash. The Ka Lae vicinity is known to contain a
						characterized by deep, yellowish brown ash deposits commonly called "Pahala Ash," howev is unclear why the IISDA soil surveys would not have noted its measure in this area
						Codimonto avenand divine avenue of ATTO 10.14 in the second avenue derive avenue derive avenue derive avenue
						seminents exposed utuing externation of 115 -12-11 in the proposed puttip station pottor the project area <i>mauka</i> of Māmalahoa Highway exposed sediments consistent with the kn
						sediment type for this area (Naalehu silty clay loam; see Section 1.3.1 and Figure 9).

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

9	
Ξ	
5	
_	
H	
<	
Ξ	
$\overline{\mathbf{A}}$	
\simeq	
14	
÷	
-8	
~	
5	
5	
×Е.	
va	
8	
Ξ	
Ś	
െ	
2	
2	
~	
2	
3	
н	
C	



Figure 47. Aerial photograph (ESRI 2017 and 2018) showing the locations of mechanical test trenches (TT) and manual test units (TU) throughout the project area

AISR för the Näʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawaiʻi Island TMKs: multiple

60

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.1.1 Test Trench 1 (TT-1)

Test Trench I (TT-1) was located in the northem portion of the proposed treatment facility site where a disposal grove is planned for development (see Figure 10 and Figure 47). Figure 48 shows TT-1 marked out with orange flagging tape prior to excavation. TT-1 measured approximately 5m long by 1 m wide. It was excavated to a maximum depth of 262 cmbs through one layer of yellowish brown silty clay loam (Stratum I) and terminated at the physical limitations of the excavator (Figure 49 through Figure 51 and Table 5). No subsurface features or cultural materials were observed within TT-1. Modern rubish is scattered across the ground surface in proximity to TT-1, most likely washout from the nearby drainage channel (-31268). The sediments encountered at TT-1 do not appear to reflect the natural Punaluu extremely rocky peat, 6 to 20 percent slopes (rPYD) soil type anticipated in this area (see Section 1.3.1 and Figure 9). Sato et al. (1973:48) describe these as thin soils with dark-colored peat characteristics; the representative profile information provided by Sato et al. (1973:49) describe a layer of black peat only 4 inches thick atop bedrock. While we can expect to see some deviation in profile daracteristics within a grown silty clay loam. The deviance in soil characteristics may be due to prior gericultural use of the area and/or its location adjacent to an existing drainage channel, where overflow sediments may have accumulated over time. Evidence of the former interpretation is found in the presence of the scattered modem trash on the ground surrounding the trench. Alternatively, these sediments may represent deposit(s) of "Pahala Ash" known to be present in the Ka Lae area.



Figure 48. Photograph of TT-1 marked out with flagging tape prior to excavation; view to east

Results of Fieldwork



Figure 49. Photograph of TT-1 post-excavation; view to northwest



Figure 50. Photograph of TT-1 east sidewall profile; view to east

AISR för the Näʿātehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Kaʿū, Hawaʿi' Island TMKs: multiple

62

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 51. Profile of TT-1 east sidewall

Table 5. TT-1 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
_	0-262	10YR, 5/6, yellowish brown; silty clay loam; fine, granular structure;
		moist, very friable consistence; weak cementation; non plastic;
		terrigenous; lower boundary not exposed; few, fine roots; no cultural
		material present

4.2.1.2 Test Trench 2 (TT-2)

Test Trench 2 (TT-2) was located in the northern portion of the proposed treatment facility site where a disposal grove is planned for development (see Figure 10 and Figure 47). Figure 52 shows TT-2 marked out with orange flagging tape prior to excavation. TT-2 measured approximately 5m long by 1 m wide. It was excavated to a maximum depth of 180 cmbs through three layers pf yellowish brown silty clay loam (Strata I, II and III) and one layer of potential Punaluu series sediment (Stratum IV), and terminated at basalt bedrock (Figure 53 through Figure 55 and Table 6). No subsurface features or cultural materials were observed within TT-2.

The sediments encountered at TT-2 are similar to those observed at TT-1, except for the base (Stratum IV) layer of darker silt loam that may represent a form of natural Punaluu series sediment. Strata I and III exhibit essentially identical characteristics, potentially signifying typical sedimentary accumulation interrupted by a significant flooding event (gravelly Stratum II). If the Strata I and III sediments are in fact Pahala Ash, their relationship to Stratum II is unclear.



Figure 52. Photograph of TT-2 marked out with flagging tape prior to excavation; view to southwest

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

64

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 53. Photograph of TT-2 post-excavation; view to southeast



Figure 54. Photograph of TT-2 east sidewall profile; view to east



Figure 55. Profile of TT-2 east sidewall Table 6 TT-2 Strationanhic Description

Stratum	Depth	Description of Sediment
	(cmbs)	
I	0-45	10YR 4/4, dark yellowish brown; silty clay loam; fine, granular
		structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; clear, smooth lower boundary; fine to medium roots
		common; no cultural material present
П	45-52	10YR 5/6, yellowish brown; extremely gravely silty clay loam; fine,
		granular structure; moist, very friable consistence; weak cementation;
		non plastic; terrigenous; clear, smooth lower boundary; no roots
		observed; no cultural material present
III	50-180	10YR 4/4, dark yellowish brown; silty clay loam; fine, granular
		structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; clear, smooth lower boundary, terminating in northern end
		of trench on bedrock; no roots observed; no cultural material present
N	165-180	10YR 4/2, dark grayish brown; silty loam; fine, granular structure;
		moist, very friable consistence; weak cementation; non plastic;
		terrigenous; abrupt, smooth lower boundary, terminating at bedrock; no
		roots observed; no cultural material present

AISR för the Näʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawaiʿi Island TMKs: multiple

99

đ

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

4.2.1.3 Test Trench 3 (TT-3)

Test Trench 3 (TT-3) was located near the central portion of the proposed treatment facility site where a disposal grove is planned for development (see Figure 10 and Figure 47). Figure 56 shows TT-3 marked out with orange flagging tape prior to excavation. TT-3 mearured approximately 5m long by 1 m wide. It was excavated to a maximum depth of 120 cmbs through two layers of throws illy elay loam (Straa I and II) and terminated at basalt bedrock (Figure 57 through Figure 59 and Table 7). No charcoal or cultural materials were observed within TT-3.

The sediments observed at TT-3 are similar to those observed at TTs -1 and -2 in the vicinity, potentially representing sediment accumulation from drainage runoff or with Pahala Ash deposition(s).



Figure 56. Photograph of TT-3 marked out with flagging tape prior to excavation; view to north

Results of Fieldwork



Figure 57. Photograph of TT-3 post-excavation; view to northeast



Figure 58. Photograph of TT-3 north sidewall profile; view to north

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 59. Profile of TT-3 north sidewall Table 7. TT-3 Stratioraphic Description

)	
Stratum	Depth (cmhs)	Description of Sediment
I	0-138	10YR 5/8, yellowish brown; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; clear, wavy
		lower boundary; no roots observed; no cultural material present
II	90-120	10YR 4/4, dark yellowish brown; silty clay loam; fine, granular structure: moist. very friable consistence: weak cementation: non plastic:
		terrigenous; abrupt, smooth lower boundary, terminating at bedrock; no roots observed; no cultural material present

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

69

Results of Fieldwork

4.2.1.4 Test Trench 4 (TT-4)

Test Trench 4 (TT-4) was located near the central portion of the proposed treatment facility site where a disposal grove is planned for development (see Figure 10 and Figure 47). Figure 60 shows TT-4 marked out with orange flagging tape prior to excavation. TT-4 measured approximately 5m long by 1 m wide. It was excavated to a maximum depth of 100 cmbs through two layers of yellowish brown silty clay loam (Strata I and II) and terminated at basalt bedrock (Figure 61 through Figure 63 and Table 8). No subsurface features or cultural materials were observed within TT-4.

The sediments observed at TT-4 are similar to those observed at TTs -1-3 in the vicinity, potentially representing sediment accumulation from drainage runoff or with Pahala Ash deposition(s).



Figure 60. Photograph of TT-4 marked out with flagging tape prior to excavation; view to north

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka"ū, Hawai'i Island TMKs: multiple

70

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 61. Photograph of TT-4 post-excavation; view to northwest



Figure 62. Photograph of TT-4 north sidewall, view to the north



Figure 63. Profile of TT-4 north sidewall

Table 8. TT-4 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
I	06-0	10 YR 5/6, yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; clear, wavy lower boundary; many, tine roots; no cultural material present
П	40-100	10YR 3/4, dark yellowish brown; silty clay loam; fine, granular
		structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; abrupt, smooth lower boundary, terminating at bedrock;
		few, fine roots; no cultural material present

AISR för the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ü, Hawai'i Island TMKs: multiple

22

والمسمول

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.1.5 Test Trench 5 (TT-5)

Test Trench 5 (TT-5) was located in the northern-central portion of the proposed treatment facility site where the drainage channel realignment is planned for development (see Figure 10 and Figure 47). Figure 64 shows TT-5 marked out with orange flagging tape prior to excavation. TT-5 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 50 cmb sathrough a single layer of dark yellowish brown silty clay loam (Strata I) and terminated at basalt bedrock (Figure 65 through Figure 67 and Table 9). No subsurface features or cultural materials were observed whiln TT-

The sediments observed at TT-5 are similar to those observed at TTs -1-4 in the vicinity. While the sediment profile is considerably shallower, it lacks the darker-colored peat attributes associated with the Kaimu and Punaluu series sediments understood to characterize this area (see Section 1.3.1 and Figure 9). The sediments at TT-5 are therefore assessed as potentially representing sediment accumulation from drainage runoff or with Pahala Ash deposition(s).



Figure 64. Photograph of TT-5 marked out with flagging tape prior to excavation, view to west

Results of Fieldwork



Figure 65. Photograph of TT-5 post excavation; view to the northwest



Figure 66. Photograph of TT-5 west sidewall, view to west

AISR för the Närätehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Islaud TMKs multiple

74

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Tigue 07. FIGHE 01 11-9 west shuewan Table 9 TT-5 Strationanhic Description

non	~ ~
drinson	• •
hiic	¢
ouaugia	
2	6
mic 2.	
	į

Stratum	Depth (cmbs)	Description of Sediment
I	0-50	10YR 4/4, dark yellowish brown; silty clay loam; fine, granular
		structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; abrupt, smooth lower boundary, terminating at bedrock;
		many, fine/medium roots; no cultural material present

4.2.1.6 Test Trench 6 (TT-6)

Test Trench 6 (TT-6) was located in the eastern portion of the proposed treatment facility site where the realigned drainage channel outlet is planned for development (see Figure 10 and Figure 47). Figure 68 shows TT-6 marked out with orange flagging tape prior to excavation. TT-6 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 90 cmbs through one layer of dark yellowish brown silly clay loam (Stratum I) and one layer of dark brown silt loam (Stratum II) and terminated at basalt bedrock (Figure 69 through Figure 71 and Table 10). No subsurface features or cultural materials were observed within TT-6.

The sediments observed at TT-6 appear to include a variation of the yellowish brown silty clay loam seen at TTs -1–5 atop a darker, more stony silt loam that may represent a form of the natural Kaimu series soils understood to characterize this area (see Section 1.3.1 and Figure 9).



Figure 68. Photograph of TT-6 marked out with flagging tape prior to excavation; view to southeast

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 69. Photograph of TT-6 post-excavation; view to west



Figure 70. Photograph of TT-6 east sidewall profile, view to east

AISR for the Na adehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Results of Fieldwork



Figure 71. Profile of TT-6 east sidewall

Table 10. TT-6 Stratigraphic Description

) 	
Stratum	Depth (cmbs)	Description of Sediment
I	0-40	10YR 4/4, dark yellowish brown; stony silty clay loam; fine, granular
		structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; clear, smooth lower boundary; many, very fine/fine roots;
		no cultural material present
Π	40-90	10YR 3/3, dark brown; extremely stony silt loam; fine, granular
		structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; abrupt, wavy lower boundary, terminating at bedrock; no
		roots observed; no cultural material present

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Zieldwork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.1.7 Test Trench 7 (TT-7)

Test Trench 7 (TT-7) was located near the central portion of the proposed treatment facility site where a disposal grove is planned for development (see Figure 10 and Figure 47). Figure 72 shows TT- marked out with orange flagging tape prior to excavation. TT-7 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 95 cmbs through two layers of yellowish brown silty clay loam (Strata I and II) and one layer of dark grayish brown silt loam subsurface features or cultural materials were observed within TT-7.

The sediments observed at TT-7 appear to include variations of the yellowish brown silty clay loams seen at TTs - 1–6 atop a darker, more stony silt loam that may represent a form of the natural Punaluu series soils understood to characterize this area (see Section 1.3.1 and Figure 9).



Figure 72. Photograph of TT 7 marked out with flagging tape prior to excavation; view to north

AISR for the Närälehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'u, Hawai'i Island TMKs: multiple

Results of Fieldwork



Figure 73. Photograph of TT-7 post-excavation; view to north



Figure 74. Photograph of TT-7 north sidewall profile; view to north

AISR för the Nä älehu Cespool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

80

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 75. Profile of TT-7 north sidewall

Table 11. 7	lT-7 Stratign	aphic Description
Stratum	Depth (cmbs)	Description of Sediment
	0-20	10 YR 4/4, dark yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; clear, smooth lower boundary; few, fine roots; no cultural material present
П	20-85	10 YR 5/6, yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; clear, smooth lower boundary; no roots observed; no cultural material present
III	80-95	10 YR 4/2, dark grayish brown; extremely stony silt loam; fine, granular structure; moist, very friable consistence; weak cementation; slightly plastic; terrigenous; abrupt, smooth lower boundary, terminating at bedrock; many, fine roots; no cultural material present

4.2.1.8 Test Trench 8 (TT-8)

Test Trench 8 (TT-8) was located in the southern-central portion of the proposed treatment facility site where a wetland and operations buildings are planned for development (see Figure 10 and Figure 76 shows TT-8 marked out with orange flagging tape prior to excavation. TT-8 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 90 cmbs through to layers of silty clay loam (Strata I and II) and terminated at basalt bedrock (Figure 77 through Figure 79 and Table 12). No subsurface features or cultural materials were observed within TT-8.

The sediments observed at TT-8 are similar to those observed at TTs -1-7 in the vicinity, potentially representing sediment accumulation from drainage runoff or with Pahala Ash deposition(s).



Figure 76. Photograph of TT-8 marked out with flagging tape prior to excavation; view to west

AISR for the Nä älehu Cesspool Closure Project. Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

82

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 77. Photograph of TT-8 post-excavation; view to west



Figure 78. Photograph of TT-8 east sidewall profile; view to east



Figure 79. Profile of TT-8 east sidewall Table 12. TT-8 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
I	0-50	10 YR 5/8, dark yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic;
		terrigenous; clear, smooth lower boundary; few, fine roots; no cultural material present
П	45-95	10 YR 4/3, brown; silty clay loam; fine, granular structure; moist, very friable consistence: weak commutation: non plastic: tarritemone: abmut
		neory consistence, wear contraction, not placete, periodices, anothy, wavy lower boundary, terminating at bedrock; many, fine roots, no cultural metrial present

AISR for the Näʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawai'i Island TMKs: multiple

84

dwork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.1.9 Test Trench 9 (TT-9)

Test Trench 9 (TT-9) was located in the southern-central portion of the proposed treatment facility site where treatment lagoons are planned for development (see Figure 47). Figure 80 shows TT-9 marked out with orange flagging tape prior to excavation. TT-9 maskred approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 52 cmbs through a single layer of dark yellowish brown silty clay loam (Strata I) and terminated at besart beforek (Figure 81 through Figure 83 and Table 13). No charcoal or cultural materials were observed within TT-9.

The sediments observed at TT-9 are similar to those observed at TTs -1-8 in the vicinity, potentially representing sediment accumulation from drainage runoff or with Pahala Ash deposition(s).



Figure 80. Photograph of TT-9 marked out with flagging tape prior to excavation; view to east

Results of Fieldwork



Figure 81. Photograph of TT-9 post-excavation; view to east



Figure 82. Photograph of TT-9 west sidewall profile; view to west

AISR för the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ü, Hawai'i Island TMKs: multiple

86

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 83. Profile of TT-9 west sidewall

Table 13. TT-9 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
Ι	0-52	10YR 3/4, dark yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; abrupt, wavy lower boundary; few, very fine roots; no cultural material present

4.2.1.10 Test Trench 10 (TT-10)

Test Trench 10 (TT-10) was located near the southern corner of the proposed treatment facility site where a treatment lagoon is planned for development (see Figure 10 and Figure 47). Figure 84 shows TT-10 marked out with orange flagging tape prior to exeavation. TT-10 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 55 cmbs through one layer of yellowish brown silty clay loam (Stratum 1) and one layer of dark grayish brown silt loam (Stratum 11) and terminated at Basalt bedrock (Figure 85 through Figure 87 and Table 14). No subsurface features or cultural materials were observed within TT-10.

USDA soil survey data indicate TT-10 is within an area characterized by the miscellaneous land type "rock land" (*pbinochoc* bechock covered in places by a thin layer of soil, per Sato et al. 1972:51; see Section 1.3.1 and Figure 9). However, the sediments observed at TT-10 appear to include a variation of the yellowish brown silty clay loams seen at TTs-1–9 atop a darker, more sinoly silt loam that may represent a form of the natural Punaluu series soils, which are also understood to be present nearby to the north.



Figure 84. Photograph of TT-10 marked out with flagging tape prior to excavation, view to west

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

88

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 85. Photograph of TT-10 post-excavation; view to west



Figure 86. Photograph of TT-10 west wall; view to west

Results of Fieldwork



Figure 87. Profile of TT-10 west sidewall

Table 14. TT-10 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
I	0-40	10 YR 4/4, dark yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; clear to abrupt, smooth lower boundary, terminating at bedrock where not underlain by Stratum II; few, fine roots; no cultural material present
П	40-55	10 YR 4/2, dark grayish brown; extremely stony silt loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; abrupt, smooth lower boundary, terminating at bedrock; many, fine roots; no cultural material present

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.1.11 Test Trench 11 (TT-11)

shows TT-11 marked out with orange flagging tape prior to excavation. TT-11 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 35 cmbs through a single layer of dark yellowish brown silty clay loam (Stratum I) and terminated at basalt bedrock (Figure 89 through Figure 91 and Table 15). No subsurface features or cultural materials were observed within TT-11. site where a treatment lagoon is planned for development (see Figure 10 and Figure 47). Figure 88 Test Trench 11 (TT-11) was located in the southeastern corner of the proposed treatment facility

The sediments observed at TT-11 are similar to those observed at TTs-1-10 in the vicinity, potentially representing sediment accumulation from drainage runoff or with Pahala Ash deposition(s).



Figure 88. Photograph of TT-11 marked out with flagging tape prior to excavation; view to north

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

AISR for the Na alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

Results of Fieldwork



Figure 89. Photograph of TT-11 post-excavation; view to north



Figure 90. Photograph of TT-11 south sidewall, view to south

AISR för the Nä ätehu Cesspool Closure Project, Kähtlipalinui and Käwala, Ka'ü, Hawai'i Istand TMKs: multiple

92

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 91. Profile of TT-11 south sidewall Table 15. TT-11 Strationablic Description

Str

(cmbs)	
0-35	10YR 4/6, dark yellowish brown; silty clay loam; silty clay loam; fine,
	granular structure; moist, very friable consistence; weak cementation
	non plastic; terrigenous; abrupt, smooth lower boundary, terminating at
	bedrock; few, fine roots; no cultural material present

4.2.1.12 Test Trench 12 (TT-12)

Test Trench 12 (TT-12) was located within the central portion of the proposed pump station site at TMK: (3) 9-5-008:048, along the route of an access and utility easement planned for development (see Figure 11 and Figure 47). Figure 92 shows TT-12 marked out with orange flagging tape prior to excavation. TT-12 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 145 cmbs through two layers of natural Naalehu series sediment (Strata 1 and II) and terminated a baschock (Figure 93 through Figure 95 and Table 16). No subsurface features or cultural materials were observed within TT-12. The sediments observed at TT-12 were as expected given the natural soil type understood to characterize this portion of the project area.



Figure 92. Photograph of TT-12 marked out with flagging tape prior to excavation; view to east

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka"ū, Hawai'i Island TMKs: multiple

94

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 93. Photograph of TT-12 post-excavation; view to east



Figure 94. Photograph of TT-12 west sidewall profile; view to west



Figure 95. Profile of TT-12 west sidewall

Table 16. TT-12 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
Ι	0.45	10YR 4/4, dark yellowish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; diffuse, smooth lower boundary; many, fine roots; no cultural material present
П	40-145	SYR 5/6, yellowish red; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; abrupt, wavy lower boundary, terminating at bedrock; few, fine roots; no cultural material present

twork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.1.13 Test Trench 13 (TT-13)

Test Trench 13 (TT-13) was located within the central portion of the proposed pump station site at TMK: (3) 9-5-008:048, where pump station structures are planned for development (see Figure 11 and Figure 47). Figure 96 shows TT-13 marked out with orange flagging tape prior to excavation. TT-13 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 110 cmbs through two layers of natural Naalehu series sediments (Strata I and II) and terminated at a salt bedrock (Figure 97 through Figure 99 and Table 18). No subsurface features were observed within TT-13; a modern soda can was observed at approximately 40–60 cmbs (in Stratum I).

The sediments observed at TT-13 were similar to those observed at TT-12 and were as expected given the natural soil type understood to characterize this portion of the project area. The presence of a soda can within Stratum I indicates some level of prior disturbance in this area.



Figure 96. Photograph of TT-13 marked out with flagging tape prior to excavation; view to northwest

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ū, Hawai'i Island TMKs: multiple

96

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 97. Photograph of TT-13 post-excavation; view to northeast



Figure 98. Photograph of TT-13 east sidewall, view to the east

AISR for the Na'alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

98

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Table 17. TT-13 Stratigraphic Description Figure 99. Profile of TT-13 east sidewall

mondu	of Sodi
	intion
and a se	Docor
	4
2	Dent
	ĺ,
	tinn a
ŝ	Ť,

Stratum	Depth (cmbs)	Description of Sediment
	0-50	10YR 3/3, dark brown; silty clay loam; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; diffuse, wavy lower boundary; many, fine roots; no cultural material present
П	45-110	5YR 5/6, yellowish red; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; abrupt, wavy lower boundary, terminating at bedrock; few, fine roots; no cultural material present

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

4.2.1.14 Test Trench 14 (TT-14)

site at TMK: (3) 9-5-008:048, where pump station structures are planned for development (see Figure 11 and Figure 47). Figure 100 shows TT-14 marked out with orange flagging tape prior to excavation. TT-14 measured approximately 5 m long by 1 m wide. It was excavated to a maximum depth of 70 cmbs through two layers of natural Naalehu series sediment (Strata I and II) and terminated at basalt bedrock (Figure 101 through Figure 103 and Table 18). No subsurface features or cultural materials were observed within TT-14. Test Trench 14 (TT-14) was located within the central portion of the proposed pump station

The sediments observed at TT-14 were similar to those observed at TTs-12 and -13 and were as expected given the natural soil type understood to characterize this portion of the project area.



Figure 100. Photograph of TT 14 marked out with flagging tape prior to excavation; view to south

Results of Fieldwork

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 101. Photograph of TT 14 post-excavation; view to southwest



Figure 102. Photograph of TT 14 north sidewall profile; view to east

AISR for the Na alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

9	
5	
Ę.	
A	
e,	
曱	
KA	
ode:	
ŭ	
lob	
÷	
Jawai	
S	
ve,	
Sur	
Iral	
Cult	
_	



Figure 103. Profile of TT 14 north sidewall

Table 18. TT 14 Stratigraphic Description

Stratum	Depth (cmbs)	Description of Sediment
Ι	0-50	10YR 3/3, dark brown; silty clay loam; silty clay loam; fine, granular structure; dry, very friable consistence; weak cementation; non plastic; terrigenous; diffuse, wavy lower boundary; many, fine roots; no cultural material present
П	20-70	5YR 3/4, dark reddish brown; silty clay loam; fine, granular structure; moist, very friable consistence; weak cementation; non plastic; terrigenous; abrupt, smooth lower boundary, terminating at bedrock; few, fine roots; no cultural material present

AISR for the Nä älehu Cesspool Closure Project, Kähilipalimui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

102

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

4.2.2 Manual Test Excavations

Two 1-m by 1-m test units (TU) were excavated by hand at SIHP # -30929 in an attempt to yield data to support assessments of site age and function (see Figure 47). TU-1 at SIHP # -30929 Feature A contained cultural material which was collected for laboratory analysis (see Section 5). TU-2 at SIHP # -30929 Feature B was culturally sterile.

4.2.2.1 Test Unit 1 (TU-1)

Test Unit 1 (TU-1) was excavated within the central portion of SIHP #-30929 Feature A, which is located in the natural depression within the proposed treatment facility site (see Figure 47 and Section 6.3). The unit was placed within the central area of the feature where a notable slab type stone was observed to cover a void. Figure 104 shows TU-1 marked out with pin flags and string prior to excavation. TU-1 measured approximately 1-m-sq. TU-1 was excavated to a maximum depth of 78 cmbs through a single layer of very dark grayish brown stony silty loam (Stratum 1) containing basalt cobles and small boulders (Figure 105 through Figure 107 and Table 19). An adze fragment and a coral abrader were observed at approximately 64 cmbs (110 cm below datum (cmbdl) and 69 cmbs (115 cmbd), respectively, in pockets of sediment in between the larger subsurface rock materials. Small charcoal fragments or flecks were observed distributed evenly throughout the unit from 14 cmbs (60 cmbd) to the base of excavation, which terminated at basalt befrook. These materials. Including the largest pieces of charcoal identified, were collected for laboratory analysis (see Section 5).

The sedimentary profile observed at TU-1 indicates Feature A was constructed in a very rocky area exhibiting an accumulation of natural silt loam sediments.



Figure 104. Photograph of TU-1 marked out with pin flags and string prior to excavation, view to southwest

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork



Figure 105. Photograph of TU-1 post-excavation; view to southwest



Figure 106. Photograph of TU-1 southwest sidewall profile; view to southwest

AISR för the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 107. Profile of TU-1 (showing depths in cm below datum)

Table 19. TU 1 Stratigraphic Description

Stratum	Depth (cmbd)	Description of Sediment
	46-124	10YR 3/2, very dark grayish brown; very stony silty loam; fine, granular
		structure; dry, loose consistence; weak cementation; non plastic;
		terrigenous; abrupt, smooth boundary, terminating at bedrock; many
		very fine to coarse roots; cultural materials present including an adze
		fragment at 110 cmbd, coral abrader at 115 cmbd, and charcoal
		fragments from 60 cmbd to base of excavation

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Fieldwork

4.2.2.2 Test Unit 2 (TU-2)

Test Unit 2 (TU-2) was excavated near the center of SIHP # -30929 Feature B, which is located in the natural depression within the proposed treatment facility site (see Figure 47 and Section 6.3). The unit was placed in a portion of the feature containing both architectural materials and sedimentary deposit. Figure 108 shows TU-2 marked out with string prior to excavation. TU-2 measured approximately 1-m-sq. It was excavated to a maximum depth of 41 cmbs through on layer of very dark gravish bounders were not removed, but soil was excavated to bedrock. No cultural materials were observed within TU-2.

The sedimentary profile observed at TU-2 indicates the Feature B architecture was constructed atop basalt bedrock. The silt loam sediments represent a subsequent natural soil accumulation within the Feature B construction.



Figure 108. Photograph of TU-2 marked out with string prior to excavation, view to west

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 109. Photograph of TU-2 post-excavation; view to west



Figure 110. Photograph of TU-2 west sidewall profile; view to west

AISR for the Na adehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Sur	weys Hawai'i Job C	Code: KAHILIPALINUI 6	Results of Fieldwork	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Results of Laboratory Analysis
	0° TN	Î	((Section 5 Results of Laboratory Analysis
	0	20 CENTIMETERS TU-2	Rock Bedrock	Collection of artifacts and cultural materials for laboratory analysis was undertaken as part of the AIS investigation. All materials collected from the project area for laboratory analysis were
	5 5 5			recovered from the screen (1/8-inch mesh) during excavation of TU-1 at SIHP # -30929 Feature A. The artifacts and cultural materials encountered during screening were noted on historic property description and/or excavation forms and assessed in the field for datable and/or representative characteristics. The materials selected for laboratory analysis were washed and sorted in the laboratory as applicable.
Υ _μ ι.	40			This section describes the findings of the laboratory analysis. Section 5.1 addresses the artifacts collected for analysis. Section 5.2 provides wood taxa and radiocarbon analysis results for select charcoal samples collected during test excavations.
	1 1			5.1 Artifact Analysis
		Surface		Two artifacts were collected from TU-1 at SIHP $\#$ -30929 Feature A. These artifacts are presented in Table 21. The artifacts include a basalt adze and coral abrader, which are generally assessed as traditional Hawaiian artefactual materials. These artifacts are discussed in greater detail in Section 5.1.1. No historic-era artifacts were discovered during excavation.
	-1	1		5.1.1 Traditional Hawaiian Artifacts
	- 1	Str. 1		The artifacts encountered within the project area that could be associated with traditional or pre-contact land use include a lithic object (basalt adze) and a coral abrader (see Table 21).
-		Unexcavated		Catalog #001 is a basalt adze (Figure 112) collected during excavation of TU-1. The artifact was encountered at 64 cmbs (110 cmbd) near the center of the unit. The artifact has a polished appearance with some chipping present on the beveled edge. It measures 6.44 cm long, 3.88 cm wide, and 2.24 cm thick, with a weight of 111.1 g. The adze was sent to the UH-Hilo geoarchaeology lab for EDXRF analysis in an attempt to ascertain information about the source the lithic material found in the project area (Section 5.1.1.1).
-	120			Catalog #003 is a coral abrader (Figure 113) collected during excavation of TU-1. The artifact was encountered at 69 cmbs (115 cmbd) in the northeastern quadrant of the unit. The irregular-
Figure 1	11. Profile of	TU-2 west sidewall (showing dept	ths in cm below datum)	shaped artifact is 3.09 cm long, 2.41 cm wide, and 2.36 cm thick. It weighs 10.5 g. 5.1.1.1 Energy-Dispersive X-ray Fluorescence (EDXRF) Results
Table 20). TU-2 Stratig	graphic Description		The basalt adze found during excavation of TU-1 at SIHP $\#$ -30929 Feature A was submitted
Stratun	n Depth (cmbs)	Description of Sediment		to the UH-Hilo geoarchaeology lab for EDXRF analysis. The results of this analysis are summarized in Table 22 and in Figure 114 and Figure 115; full report is in Appendix D. Dr Peter
1	29-100	10Y R 3/2, very dark grayish brov granular structure; dry, loose con plastic; terrigenous; abrupt, smoc bedrock; many very fine to coars	wn; extremely stony sufry loam; fine, histence; weak cementation; non oth lower boundary, terminating at e roots, no cultural material present	Antus provides the following analysis of the results: As I mentioned, this adze falls out smack dab in the middle of "Group F" which is common in Ka'u and Kona. In the Excel file, I plotted it against a backdrop of Richoro Minseum arificates that we analyzed from Waiahukini (H8) and other sites at
]				Ka Lae [see Figure 114 and Figure 115], some of that data is presented in the attached Kahn et al. [2016] article, which also discusses Group F. Kilauea and Mauna Loa both produce lava flows with this range of geochemistry and EDXRF
AISR for t TMKs: mu	the Nā'ālehu Cesspo altiple	ool Closure Project, Kähilipalinui and Kāwala, K	ƙa'ŭ, Hawai'i Island 108	AISR for the Nāʿālehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Kaʿū, Hawaiʾi Island TMKs: multiple

Results of Laboratory Analysis

Results of Laboratory Analysis

Table 21. Artifact catalog (all materials collected from TU-1 at SIHP # -30929 Feature A)

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

atalog	Context (Stratum; depth)	Description	# of Pieces	Material Type	Dimensions (centimeters)	Weight (grams)	Photo Reference
_	Str. I; 64 cmbs	Basalt adze	1	Lithic	6.44 Long 3.88 Wide 2.24 Thick	111.1	Figure 112
	Str. I; 69 cmbs	Coral Abrader	1	Coral	3.09 Long 2.41 Wide 2.36 Thick	10.5	Figure 113



Figure 112. Photos of Catalog #001, basalt adze



Figure 113. Photos of Catalog # 003, coral abrader

AISR for the Nāʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Istand TMKs: multiple

Table 22. EDXRF Results for Basalt Adze (Catalog #001) from TU-1

Sr Zr Source	(ppm) (ppm) Interpretation	315 138 Group F	(Kona and Ka'ū)	
qN	(mdd)	11		
Rb	(undd)	8		
Description		Basalt adze		
EDXRF Sample ID		Kahilipalinui 3 Acc 1	TU1 -31270 fea b	adze fragment 111.1g







Figure 115. Bivariate plot showing Strontium (Sr) vs. Zirconium (Zr) ratios of "CSH Adze" in relation to other basalt samples from Ka'ū (courtesy UHH EDXRF Lab)

AISR for the Nä"äichu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ū, Hawai'i Island TMKs: multiple

11

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	Re	esults of Laboratory Analysis	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Results of Laboratory Analysis
is not great at sorting the two out in this 2021 email correspondence]	s geochemical range. [Dr Mil	lls, 16 March	for firewood by the Hawaiians (Hillebrand 1981:396) and the milky sap was once considered a possible source for rubber (Rock 1974:261). Some members of this
5.2 Charcoal Analysis	-		genus were once known as Channesyce but the Hawanian species have since been reassigned to the genus Euphorbia (Govearts et al. 2000; Steinman and Porter 2002; Voncend Berry 7011, 17, Minuschem; 2020-21
Procurement of charcoal samples for chron subsurface testing program. Charcoal was collec SIHP #-30929 Feature A (Table 23). Small charc the unit from 14 cmbs (60 cmbd) to the base of analysis.	mological analysis was one cted from the screen during ex- coal fragments or flecks were- f excavation; the largest piece	of the goals of the xcavation of TU-1 at observed throughout es were collected for	According to the Bishop Museum's online ethnobotany database, ' <i>dheahea</i> in an endemic shrub According to the Bishop Museum's online ethnobotany database, ' <i>dheahea</i> in an endemic shrub that is part of the Chenopodiaceae family "Occurring as a common or occasional element of dry habitats, ranging from 0–2520 m from coastal zones to dry forest and subalpine shrubland (Wagner et al. 1990:538) on most main islands and some NWHI [North West Hawaiian Islands]," (Bishop
A bulk sample comprising all of the col identification; individual pieces identified as sl	illected charcoal was submit ihort-lived wood species were	tted for wood taxa e then submitted for	Museum 2021). The plant was used traditionally for food and medicine, in the making of composite fishhooks, and for spiritual purposes (Bishop Museum 2021).
rautocarbon anatysis. Table 23. Charcoal catalog (all materials collect	ted from TU-1)		Some of the material within the bulk charcoal sample was identified as charced fragmental <i>huku</i> or candidenti (<i>Alexives moluccanus</i>) nutshell. The presence of <i>hukui</i> can be indicative of past land more advisition of the second
Catalog # Context (stratum; depth) Desc	cription	Weight (grams)	use and activity as tirese nues were tranitonany unitized tot tood, nguting, incurding, and outer uses (Handy and Handy 1972:231–232). In an agricultural context they may have been used to enrich
002 Str. I; 14 cmbs (60 cmbd) to Char	rcoal (9 pcs) 2	2.1	planting soils.
5.2.1 Wood Taxa Analysis			The bulk charcoal sample was returned to CSH. Of the sample, 1.7 g was marked as unidentified. The remaining 0.4 g of charcoal comprised the ' <i>uheuheut</i> , <i>ideolo</i> , and <i>kuku</i> nutshell samples, which were separated by type into small, labeled ziptop bags. Some of the charcoal pieces were shartered into smaller increase humoulong the moress of chiming of and identification.
All of the charcoal collected from TU-1 at SI to the Wood Taxa Identification Lab (WIDL) or taxa announciate for radiocarbon dating Unfortu-	IHP # -30929 Feature A (9 pic in O'ahu for the purpose of id imately. due to nersonnel issue	eces) were submitted lentifying short-lived se at the WIDL_CSH	activities. The <i>'aheahea</i> and <i>'akoko</i> samples were prepared for forwarding to Beta Analytic for radiocarbon analysis.
was unable to obtain a full analysis report. A br	rief email summary was provi	ided to CSH instead,	5.2.2 Radiocarbon Analysis
focused on identifying which samples might be	best suited for radiocarbon ar	nalysis:	Following the recommendations of Gail Murakami from the WIDL, two charcoal samples
I am not able to identify all of the charc for known historical introductions and couple of possibilities for radiocarbon native shrub previously known as (Chenopodium oahuense), another nativ vour site near Dahakuloa?) There is only vour site near Dahakuloa?)	coal but I have looked throug found none. I also managed dating. One is akoko (Euph- Clamassyce. The second we shrub which is a tree at Po vone nicer of the aberlas so.	h the sample to pick out a orbia spp), a is abeabea bhakuloa. (Is	representing short-lived wood species were submitted to Beta Analytic in Florida for radiocarbon analysis. These included the entirety of the ' <i>ukoko</i> (total = 0.3 g) and ' <i>uheuheu</i> (total = >0.1 g) charcoals from CSH Catalog # 002, collected from TU-1 at SIHP #09299 Feature A. The two samples were assigned respectively by the Beta Analytic lab as "Beta-592957" (Beta sample code 2a) and "Beta-59268" (Beta sample code 2b); the results of the radiocarbon analyses for these samples are summarized in Table 24. The full Beta Analytic report is in Appendix E.
you such the real routakuroa:) there is only to date the akoko instead. [Gail Murakar	mi, 27 April 2021 email corre	you may need sspondence]	A 2-sigma calibration of the radiocarbon analysis of Sample Beta-592957 / 2a suggested four
Notably, no historically introduced woods w noted to be suitable for radiocarbon analysis co (<i>Chenopodium oahuense</i>). The following descri	vere identified in the charcoal pnsisted of 'akoko (Euphorbia iption of 'akoko and its traditio	l sample. The pieces <i>a</i> spp.) and <i>'aheahea</i> mal uses comes from	possible atternation of the formation of the procession of the procession of the possible of the possible of the possible of the procession of the processi
results of a wood taxa analysis completed for a l EUPHORBIACEAE (Spurge family)	prior CSH project on Hawai 1	1 Island:	A 2-sigma calibration of the radiocarbon analysis of Beta-592958/2b suggested three possible date ranges, AD 1666 to 1783 (42.8%), AD 1796 to 1894 (33.8%), and AD 1903 to Post 1950 to be a suggested and a suggested
Euphorbia spp. ('Akoko)			(18.8%), with the late pre-Contact date range (1666 to $1/83$) being the most probable (see 1able 24 and Figure 117).
Distribution of the 15 endemic Hawaiian from coastal environments to upper fores 1990:602-617). Eight native species, and found on Hawai'i Island today in a wide	a shrubs and small trees in this st zones on the main islands (¹ d a number of naturalized intro range of habitats. ⁴ Akoko wa:	s genus range Wagner et al. oductions, are s once valued	
AISR for the Nāʿālehu Cesspool Closure Project, Kāhilipalinui an TMKs: multiple	nd Kāwala, Ka'ū, Hawai'i Island	112	AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka Tū, Hawai'i Island 113 TMKs: multiple

2.1	
Charcoal (9 pcs)	
Str. I; 14 cmbs (60 cmbd) to BOE	
02	

5.2.1 Wood Taxa An

Results of Laboratory Analysis

Table 24. Radiocarbon dating analysis (2-sigma calibration) for short-lived species charcoal collected from TU-1

Laboratory	Measured	13C/12C	Conventional	2-Sigma Calibration
Number /	Radiocarbon	Ratio	Radiocarbon	(95.4% Probability)
Sample Code	Age		Age	
Beta-592957 /	701.06 +/- 0.37	-8.8 0/00	180 +/- 30 BP	AD 1722 to 1814 (49.9%)
2a	pMC			AD 1656 to 1698 (19.2%)
				AD 1910 to Post 1950 (19.0%)
				AD 1836 to 1880 (7.3%)
Beta-592958 /	150 +/- 30 BP	-25.2 0/00	150+/-30	AD 1666 to 1783 (42.8%)
2b				AD 1796 to 1894 (33.8%)
				AD 1903 to Post 1950 (18.8%)

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Results of Laboratory Analysis



Figure 116. Calibration data and calibration curve for Sample Beta–592957 / 2a from TU-1 (courtesy of Beta Analytic)



Figure 117. Calibration data and calibration curve for Sample Beta–592958 / 2b from TU-1 (courtesy of Beta Analytic)

AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

114

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Site Descriptions

Section 6 Site Descriptions

6.1 SIHP # 50-10-74-29507

FORMAL TYPE:	Complex
FUNCTION:	Ranching
NUMBER OF FEATURES:	7 (3 newly recorded features in project area)
AGE:	Historic
TEST EXCAVATIONS:	None
TAX MAP KEY:	(3) 9-5-007:016; 9-5-010:001 (outside current project area)
PREVIOUS	Clark et al. 2013 (portion in 9-5-010:001)
DOCUMENTATION:	

SIHP # -29507 is an approximately 100-m-sq complex associated with Onipa'a Ranch. The site overlaps the far eastern extent of the proposed treatment plant facility portion of the project area (Figure 46). The site is within an area do the project treatent project in the second to the project area (Figure 46). The site is within an area do the H = -31270) and the Onipa'a Ranch. The site is within an area do the H = -31270). The site majority of the site likes in TMK: (3) 9-5-010:001 which bounds the current project area to the east; this portion of the site was first identified and documented by Clark et al. (2013). At that time, some additional associated features were also observed to extend into TMK: (3) 9-5-007:016 but were not fully documented or assigned feature designations at that time. Figure 118 is the SIHP # -29507 plan view map from Clark et al. (2013:185), adapted to show the extent of the documented features (Features A through D) in relation to the current project area bounds. As Figure 118 shows, portions of what are now assigned as Features E (rock wall) and F (concrete and earthen channel) were noted during the 2013 survey. A write fence fine associated with the site was also documented in the current survey as Feature G (see Figure 46).

The following are excerpts from the SIHP # -29507 site description in Clark et al. (2013); the detailed descriptions for Features A, B, C, and D are omitted below as these features are situated outside the current project area. The full site description is provided in Appendix F.

SIHP Site 29507

Site 29507 is a complex of old ranch buildings located in the northwestern corner of the project area near the junction of the Site 29505 and Site 29506 walls... The complex includes three main buildings (Features A, B, and D), and bjece of machinery from the Nä'afleh Sugar Mill on a stone platform (Feature C), and several associated features (i.e. fence lines, water lines, rock walls, gates, a water trough, a small shed, a kennel, a recent *imu* pit, a wooden corral, hitching posts, a gravel parking area, modern farm equipment, trash, etc.) within a roughly 100 meter Nä'älehu Spur Road in the northwestern corner of the study parcel. The gate is used anost exclusively to access the adjoining parcel to the west (TMK:3-9-5-07:016), as vehicular access in op possible from this gate to the remainder of the current project area located on the eastern side of Site 29506. Only one of the buildings

AISR for the Nä älehu Cesspool Closure Project, Kähilipalimui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



approximated current project area extent (in pink) and showing the locations of newly assigned Features E and F

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMK s: multiple

116

Site Descriptions

(Feature B) is situated wholly within the boundaries of the current study parcel, the other two are situated partially within the study parcel, and partially on the neighboring parcel to the west. A map of the entire site (prepared by Rechtman Consulting, LLC during fieldwork conducted on the neighboring parcel in 2007 [this study was never finalized]) is presented below, but only the site's features that recorded in detail...

The features of Site 29507 ... include a long metal and wood structure on a concret slab located at the southern end of the complex (Feature A), a small wooden building (Feature B) located at the southwestern end of Site 29505, a large wooden building at the northern extent of the complex next to the gravel parking area (Feature C), and an old piece of machinery from the original Na² alehu sugar mill that has been set up on a cobble platform (Feature C) at the intersection of Sites 29505 and 29506. All of the buildings are in various states of disrepair, and they do not appear to have been used (except for storage purposes) for a number of years. Features A, B, C, and D of Site 29507, which are all at least partially located within the boundaries of the current project area, are edscribed in detail below. Their locations relative to the project area are boundaries, Sites 29505 and 29506, and the other features of Site 29507 that are situated on the neighboring parcel to the west are depicted in [Figure 118]. [Clark et al. 2013:184]

The Site 29507 ranch building complex appears to have been built during the early twentieth century, by the ranching subsidiary of the Hutchinson Sugar Plantation (Ka'alu'alu Ranch/Nā'ālehu Ranch and Dairy, Inc.). The formal attributes of the building suggest their functions, and a review of the Hutchinson Sugar Plantation of Dampany Amual Reports for the years 1903 to 1969 (available at the University of Hawai'i at Hilo Mookin Library) provides additional information relating to the construction and use of the buildings. Feature A functioned as a piggery with an attached slaughterhouse for pigs; Feature B was constructed as a small stable with three stalls, but was later turned into a bunk house; and Feature D was the slaughterhouse for cattle, but was converted to a ranch headquarters and equipment storage area when the slaughterhouse shut down in the mid-1950s. Feature C, which is not a building, seems to be a piece of equipment removed from the original Nis Talehu Sugar Mill and placed on top of a cobble foundation at the junction of the Site 29506 ranch walls. The specific use of Feature C at Site 29507 is not clear; it may have been purely a decorative addition to the ranching complex.

The first mention of Site 29507 in the Hutchinson Sugar Plantation Company Annual Reports occurs in 1918 when it was reported by the plantation manager that 108 cattle were slaughtered at the company's own slaughterhouse (Feature D). Henke (1929) also mentions that the ranch had its own slaughterhouse, at which roughly 100 head of cattle were slaughtered annually. The location of the slaughterhouse is shown on a March 1931 map of the Na² alehu Ranch and Dairy, Inc. lands in the northwestern corner of the project area adjacent to Dairy Paddock # 8 ... and the Site 29505 and Site 29506 ranch walls. Feature A is specifically mentioned in the annual report for 1944 (H. P. S. C. 1944), when Managet J. S.

AISR för the Nä ätehu Cesspool Closure Project, Kähilipalinui and Kävala, Ka'ü, Hawa'i' Island TMKs: multiple

118

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Beatty reported on the expenses incurred by additions to buildings at the piggery. A 1944 date inscribed in concrete of the furrowing pens at the western end of the Feature A structure confirms the additions occurred in that year. No specific mention of Features B and C was found in the annual reports. The writing on Feature C, however, indicates that the machinery was produced in 1871 by W. A. McOnie Company of Scoland Street Engine works, Glasgow, Scoland (founded in 1840), who were the manufacturer's [sic] of sugar mills and engines that were shipped all over the world (Grace's Guide 2007). The year written on Feature C corresponds to the date the Naïafiehu Sugar Mill was erected, and the machinery was likely moved to the location of Site 29507 after the mill was dismantled in ca.

1910.

The Nä^{*}illehu slaughterhouse closed in the mid-1950s, and the ranch's beef was then marketed through the Hilo Meat Co. (The Gilmore Hawaii Sugar Manual 1954:46). After the closure of the slaughterhouse the Feature D building was converted to serve as an office building (ranch beadquarters), workshop and equipment storage area. It is likely around this time that the covered structures attached to its western end were added. Feature D is known to have served as the headquarters [for] several rannebes that ran eathe on the project area lands during the second half of the twenticth century, including the Hawaiian Ranch Co. (Ka*alu'alu Ranch; ca. 1967), Parker Ranch (ca. 1978), Kawaihae Ranch (ca. 1987), and Onipa'a Ranch (unknown date). The stable (Feature B) was converted to a bunkhouse at this time as well (probably by Onipa'a Ranch, which is inscribed in the concrete at Feature B). When Feature A ceased to be used as a piggery is unknown. [Clark et al. 2013;207]

Feature E is a stacked, core-filled rock wall located along the western side of the SIHP #-29507 complex (see Figure 46 and Figure 118 through Figure 120). The wall measures 59 m long overall, of which 36.5 m is in the current project area. The wall ranges in height from 100–120 cm and is approximately 100 cm wide. Un-milled wooden posts and modern metal t-posts support a single strand of barbed wire fence running along the surface of the wall. Portions of the wall exhibit minor collapse. One gate is present near the current southern project area boundary (see Figure 118).

Feature F is a concrete and earthen drainage channel located between Features E and G (see Figure 46, Figure 118, Figure 121, Figure 122, and Figure 123). Figure 118 indicates the channel inlet is statuated 2 or 3 m south of Feature D, outside the current project earce, this portion of the channel was found to be obscured by sediment and vegetation during the current fieldwork. The channel was visible beginning where it passes a rock wall. The channel trends downslope to the southeast, protected by a wire feare offset a distance of 3-5 m to the west. It is characterized as a concrete channel measuring 50 cm wide at the surface, tapering to 18 cm wide at its interior base, with a depth of 13 cm. This concrete section of the channel beginning at the rock wall is 50.4 m long; of this; 30.5 m is within the current project area. The concrete channel opens into a wider, earthen channel draining into a level field. The earthen portion of the channel is approximately suffin the project area.

AISR för the Nä'älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Site Descriptions

Site Descriptions

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 119. Photograph of a portion of the SIHP # -29507 Feature E rock wall; view to northeast



Figure 120. Photograph of a portion of the SIHP # -29507 Feature E rock wall; view to northeast

AISR for the Na'alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

120

Site Descriptions



Figure 121. Plan view of SIHP # -29507 Feature F, drainage channel

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Site Des	scriptions	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Site Descriptions
		Feature G is a wire fence line extending from the -31269 Feature B vicinity to the south and east back toward the main part of the SIHP \pm -29507 complex (see Figure 46, Figure 118, Figure 124 and Figure 125). Overall, 260 m of Feature G fence line was recorded during the survey, of which 115 m is within the current project area bounds. The fence is constructed with three to four strands of barbed wire supported with milled wooden post and modern metal t-posts. SIHP \pm -29507 is in remnant to good condition. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. Clark et al. (2013:1392, 1397) assessed SIHP \pm -29507 is in remnant to good condition. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. Clark et al. (2013:1392, 1397) assessed SIHP \pm -29507 is an significant under Criterion d for the information it has yielded about historic land use and recommended the site for no further work. The results of the current survey support the prior assessment made by Clark et al. (2013). The three newly documented features within the current project area are likewise assessed as significant under Criterion d for the information they have yielded about historic ranching activity (see Section 8.1). The available records suggest SIHP \pm -29507 has not been previously evaluated for cligibility to be listed on the National Register. Based on the findings of the current AIS the site is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and the site is recommended for no further work (see Section 8.2).
Figure 12. The tograph of a concrete section of the SIHP # -29507 Feature F drainage char view to southeast	inei:	
Figure 123. Photograph showing an earthen section of the SIHP # -29507 Feature F drainag channel, with the parallel wire fence visible in background; view to southeast	ංති	
AISR for the Nāʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawaiʻi Island	122	AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island

AISR for the Nāʿālehu Cesspool TMKs: multiple

TMKs: multiple

Site Descriptions	Cultural Surveys Hawai'i Job Code: KAHILJ	PALINUI 6 Site Descriptions
	6.2 SIHP # 50-10-47-30	187
	FORMAL TYPE:	Māmalahoa Highway
	FUNCTION:	Transportation
	NUMBER OF FEATURES:	
	AGE:	Historic
	TEST EXCAVATIONS:	None
	TAX MAP KEY:	9-5-021:999
	PREVIOUS DOCUMENTATION:	None in current project area; portions elsewhere documented by Clark et al. (2014), Belluomini and Hammatt (2017), LaChance et al. (2017), Yucha and Hammatt (2017)
outheastern	SIHP # -30187 comprises the known as the Belt Road (variabl ft) portion of the Mämalahoa H main crosses it at the Ka'alaiki.1 town (see Figure 46). The Mäma study (Clark et al. 2014; S1) con This SIHP designation has been LaChance et al. 2017, Yucha	former and present alignments of the Māmalahoa Highway, also y State Routes 11, 19, 180, and 1900. A very small (15-m or 50- ighway is in the current project area, where the proposed force Road/Nä ² älehu Spur Road intersection in the middle of Nä ² älehu lahoa Highway was first assigned as SIHP $\#$ -30187 under a prior ducted along Route 11 in the Captain Cook area of South Kona. used in subsequent studies (e.g., Belluomini and Hammatt 2017) to document portions of the Mämalahoa
	Highway around the island. Cl Belluomini and Hammatt (2017 SIHP # -30187 initially includec consisting of a stone revetment abandoned segment of roadway provide the following introductio	osest to the current project area is the portion documented by), in Hiflea Ahupua'a 6.0 mi (9.7 km) m northeast of Na'ālehu. I five features documented by Clark et al. (2014) in South Kona (Feature A), two stone retaining walls (Feature B and C), one (Feature D), one concrete culvert (Feature E). Clark et al. (2014) on to the site:
	[Mămalahoa Highway] P Maly and Maly 2001), improvements over more straightening, widening, systems that are upgrade projects. The first large fooppath/mule trail into o portion of the upper Gov early 1900s. By 1933 constructed over the sat advances in earth moving land more efficiently, an [Clark et al. 2014;52]	as developed from what was once a footpath (Kealaehu; into its current Highway 11 form through incremental than a century. The improvements come in the form of grade changes, and enhanced storm water drainage d routinely, but especially during large scale construction accale endeavor that transformed this roadway from a net automobiles could negotiate was the Kona to Ka'i emment Road, which was conducted in the late 1800s to the Mämalahoa Highway (Hawai'i Belt Road) was ne general corridor as the older Government Road, but g technology enabled engineers to cut and fill the sloping d thus were able to create a more direct, less curvy route.
ical (cal	The five features documenter Māmalahoa Highway. The full S in Appendix F. Belluomini and F	1 by Clark et al. (2014) are suggested to be fairly typical for the IHP $\#$ -30187 site description from Clark et al. (2014) is included fammatt (2017) provide the following description for the portions
- -	AISD ճշո դեռ M3-3Jahn (Casewood Channe Dr	ioo Kitibolimi oo Karolo Keis Hamii blaad
1/4	AIDIN 101 ULD ING MICHING COSCIENCE AND MICHINE AND	oject, Kaninpalnuu and Nawara, Na u, frawar i isianu

Figure 125. Photograph of a portion of SIHP # -29507 Feature G fence, showing typical construction; view to west

AISR for the Na'alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Figure 124. Photograph showing SIHP # -29507 Feature G fence connecting to the southeaster comer of SIHP # -31269 Feature B holding pens; view to west



Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

	ASSE V
ZÀ	

TMKs: multiple

Site Descriptions

of the Belt Road closest to the current project area, in the context of a project to replace the Hilea Stream Bridge:

The portions of SIHP # 50-10-47-30187 that extends through the current project APE and the nearby Yucha and Hammatt (2017) project APE for the Ninole Bridge Replacement Project are similar in condition. Highway 11 extends north to south through both project's APE' [sic] as a gently crowned, two-lane (one lane in each direction, approximately 11-feet wide) asphalt highway with white line shoulder striping and reflectors down the center [Figure 126 and Figure 127]. This portion of Highway 11 was constructed in the late 1930s and early 1940s, likely around the time of the construction of Hilea Bridge (SIHP # 50-10-74-30298) and Ninole Bridge (SIHP # 50-10-68-30299) in 1940 as a straightening of the Old Government Road that is northwest of the project APE. The roadway has been repaved and modified in moden times [see Figure 126 and Figure 127]. Some minor signs of wear such as cracking were observed. The raised pavement mostly mate and the road surface markings have some cracking, but have not faded. [Bellounni and Hammatt 2017;54]

The portion of the Māmalahoa Highway within the current project area is similar in character to those portions documented by Belluomini and Hammatt (2017) and Yueha and Hammatt (2017). Figure 128 and Figure 129 depict the 15-m wide portion of Māmalahoa Highway within the project area. No culverts or other architectural features are present along this portion of the highway. Crosswalks are employed along all sides of the intersection, and concrete sidewalks are present at the northeastern and southeastern concrets.

the constructed elements of the original Māmalahoa Highway are no longer evident today. SIHP # -30187 was previously assessed as significant by Clark et al. (2014:81) per HAR §13-284-6 under Criterion "A" for its association with "important late nineteenth and early twentieth events The 15-m (50-ft) portion of SIHP # -30187 (Māmalahoa Highway) within the current project area is in good, maintained condition as an active roadway. Based on the nature of the roadway in this specific location, comprising the historic corridor only and absent any other contributing elements such as bridges, culverts, or guardrails, the site retains integrity of location only, as all for its information potential. Belluomini and Hammatt (2017:54) note, "It is the understanding of this study that the Clark et al. (2014) assessment of significance was explicitly under HAR "\$13-284-6" as significant under Criteria a and d. This report supports the Clark et al. (2014) assessment of significance." The current study also supports these past assessments of significance. Therefore, pursuant to HAR §13-275-6, the portion of SIHP # -30187 in the current project area is assessed as significant under Criterion a for its association with events that have made an important contribution to the development of transportation routes on Hawaii Island, and under Criterion d in establishing a regional transportation network that has its roots in antiquity" and Criterion "D' for having yielded information about historic transportation in the project area.

Similarly, the findings of this AIS at SIHP # -30187 support the previous evaluation of eligibility for listing on the National Register of a similar portion of the roadway made by Belluomini and Hammatt (2017). This previous evaluation found that SIHP # -30187 is eligible for inclusion in the National Register under Criterion A (for its association with events that have made a significant contribution to the broad patterns of our history) and Criterion D (for having

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

126

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



Figure 126. Photograph from Belluomini and Hammatt (2017:39) of SIHP # -30187 (Māmalahoa Highway) in Hīlea; view to south



Figure 127. Photograph from Yucha and Hammatt (2017:35) of SIHP # -30187 (Māmalahoa Highway) in Nīnole; view to north

	N.	
		/

Figure 128. Photograph overlooking the portion of SIHP # -30187 (Māmalahoa Highway) within the project area; view to northeast



Figure 129. Photograph overlooking the portion of SIHP # -30187 (Māmalahoa Highway) within the project area; view to southwest

AISR for the Nāʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawaiʻi Island TMKs: multiple

128

Site Descriptions

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

yielded, or being likely to yield, information important in history). As noted above, the subject portion of this site retains integrity of location only, as all the constructed elements of the original Māmalahoa Highway are no longer evident today. While the project will affect the modern roadway elements, these elements will be replaced following construction, and the integrity and significance of the highway will not be diminished. Therefore, the project will have no adverse effect on SIHP # -30187 and no further work is recommended.
9
Ē
¥
Ξ.
1
Ē.
E.
Ξ
\mathbf{A}
\mathbf{z}
ö
В
C
q
5
÷٩
Ň
Ę.
1
- Ši
ž
-
112
Ę.
2
-

6.3 SIHP # 50-10-74-30929

Complex	Activity area, ranching	S: 3	Pre-Contact/Historic	2 (TU-1 and TU-2)	(3) 9-5-007:016	Gastilo and Clark (2018; Draft)
FORMAL TYPE:	FUNCTION:	NUMBER OF FEATURES	AGE:	TEST EXCAVATIONS:	TAX MAP KEY:	PREVIOUS DOCUMENTATION:

Т

SIHP #-30929 is a pre-Contact and historic complex located along the interior edge of a natural depression in the central-western portion of the proposed wastewater treatment facility site (see Figure 46). The SIHP #-31270 jeep road is situated approximately 10 m to the west. The natural depression is sheded by a dense canopy of Christmasberry and *hukui* trees, and contains a ground depression is invasive vines, and grasses. The ground surface within the depression is uneven, with numerous basalt outcoppings and scattered basalt boulders and cobbles. The topography surrounding the site slopes generally towards the north and east into the depression.

SIHP # -30929 was previously documented by ASM (Gastilo and Clark 2018) as follows:

Site 30929 consists of an enclosure, an L-shaped pavement, and a pathway ... that are located approximately 150 meters northeast of Site 30933 in the northern portion of the current study area ... The terrain surrounding Site 30929 consists of undulating pähochoc bedrock, numerous bedrock outcrops, and some areas of soil accumulation. To the immediate south of Site 30929 is a large section of the current study area that has been heavily impacted by bulldozing activities over the past several decades. These activities have occurred as recently as 2015. Vegetation in the area consists of koa haole, Christmas-berry, lantana, Java plum, and Guinea grass. Site 30929 is described in detail below. Based on its formal attributes, Site 30929 was likely used for Precontact habitation and storage purposes, as well as for Historic ranching purposes.

The enclosure is located to the immediate southwest of the L-shaped pavement and is constructed of small to large pāhoehoe cobbles ... The northern, eastern, and southern edges of the enclosure form a semi-circle while its westem edge is formed by a pile of large cobbles. It measures roughly three meters long by three meters wide, and ranges between 0.2 to 0.6 meters tall. The interior of the enclosure contains a thin layer of soil on top of bedrock, losse cobbles, and koa haole. Noted along the northwestem corner of the enclosure are two flat-laid slabs that appear to be covering a sub-surface void that is approximately 0.3 meters deep The L-shaped pavement is located roughly one to 1.5 meters northeast of the enclosure and is constructed of small to large cobbles and slabs ... In total, the pavement measures 11.1 meters long by 2.3 to 3.8 meters wide, and ranges between 0.2 to 1.4 meters tall. However, the northern edge of the western pavement

AISR för the Nä ätehu Cesspool Closure Project, Kähilipalinui and Kävala, Ka'ü, Hawa'i' Island TMKs: multiple

130

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

terminates at a steep cobble slope which elevates the pavement up to 1.4 meters above the adjacent ground surface to the north. These cobbles were likely never stacked and may be naturally occurring. Noted to the cast of the pavement is what appears to be a possible pathway. It extends to the north and south across a shallow portion of the surrounding depression and along the level area on which the pavement is constructed. The pathway measures roughly 1.5 meters wide, extends an unknown distance to the north and south, and consists of soil, loose cobbles, and exposed bedrock. In some sections the pathway is lined with loosely piled cobbles, and exposed bedrock. In some sections the pathway is lined with loosely piled cobbles. Noted between the pavement and the pathway is a subsurface void that measures approximately 0.2 meters deep. To the east of the pathway is a 6.2-meter-long section of a galvanized nutural materials were noted at Site 30929. Based on its formal attributes, Site however, the voids may have been utilized for storage or burial purposes however, the voids may have been utilized for storage or burial purposes. The pathways and the galtwarined metal pipe studied for farcing and the area was modified for ranching purposes. [Gastilo and Clark 2018:145–147]

The full Gastilo and Clark (2018) draft site description is provided in Appendix F for reference. The photos included with the description indicate significant changes in vegetation have occurred since that time, as the vegetation density has since increased. Two other notable differences were observed in the current documentation: the section of metal pipe noted by Gastilo and Clark (2018:147) along the eastern side of the site is no longer present; and an additional feature, a rock mound, was newly identified to the south of Feature B. Based on the current site observations. SIHP # -30929 is described as a complex comprising three discrete features: a semi-circular alignment of stones (Feature A) correlating with ASM's enclosure feature; a large, paved area with features; and a stone mound (Feature C) (Figure 130). Overall, the site measures approximately 20 m long (N/S) by up to 13 m wide (E/W). No portable artifacts were observed on the ground surface. Feature A is a small semicircular alignment of stones located approximately 10 m east of the SIHP # -31270 road (see Figure 130, Figure 131, and Figure 132). The feature is generally constructed of a 1–2 course circular alignment of placed basalt cobbles and small boulders and incorporates natural bedrock formations to the west. A void along the western interior edge of the feature appeared partially covered by basalt slabs. Feature A measures approximately 4.3 m (N/S) by 3.5 m (E/W), with heights ranging from 10–56 cm. The feature is in fair condition with some potential damage from tree roots. Excavation potential was assessed as good, and a 1–m-sq test unit (TU-1) was excavated near the center of Feature A, incorporating the void area, in an attempt to yield data that may support the assessment of feature function and age (see Section 4.2.2.1). Testing yielded cultural materials including an adze fragment, coral abrader, and charcoal. These materials were collected for laboratory analysis (see Section 5).

Feature B is a large, paved area comprising the majority of the SIHP # -30929 site footprint (see Figure 130, Figure 133, and Figure 134). It is situated approximately 1 m west of Feature A. The feature is comprised of basalt cobbles and boulders placed on a natural basalt outcrop to form a paved, level area. The paved area is roughly triangular in shape, widest along the northern

AISR for the Nä ätelen Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Figure 131. Photograph of SIHP # -30929 Feature A alignment; view to southeast Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Site Descriptions (06)

Figure 130. Plan view of SIHP # -30929 complex

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Na alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka 'ū, Hawai'i Island

TMKs: multiple

132







9	
Ξ	
ž	
Ξ	
×	
Ξ.	
Ħ	
4	
Ы	
ö	
В	
Q	
9	
2	
Ъ,	
M	
Ξ	
VS.	
ve.	
Ę.	
\mathbf{s}	
12	
lfu.	
õ	



Figure 133. Photograph of SIHP # -30929 Feature B paved area; view to north



Figure 134. Photograph of SIHP # -30929 Feature B paved area; view to east

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

134

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

potential was assessed as good, and a 1-m-sq test unit (TU-2) was excavated near the southern end exposed outcrop edge and tapering to a curved point at the southern end. The eastern side of the feature exhibits a sloped area containing fewer stones that may have served as a ramped pathway Feature B measures approximately 13.5 m (N/S) by up to 12.5 m (E/W), with heights ranging from 5 cm to up to 90 cm at stacking along the eastern outcrop edge. Feature B is in fair condition with some collapse visible at the southern end, and potential tree root damage throughout. Excavation of Feature B in an attempt to yield data that may support the assessment of feature function and into the depression. The space between Features A and B is similarly devoid of stones. Overall age (see Section 4.2.2.2). Aside from the architectural paving stones no cultural materials were encountered Feature C is an oblong stone mound located approximately 5 m south of Feature B (see Figure 130 and Figure 135). The mound consists of a loose pile of basalt cobbles and boulders atop a bedrock outcrop. Feature C measures 3.8 m long (N/S) by 2.0 m wide (E/W) with heights ranging southern sides. Excavation potential was assessed as poor due to lack of sedimentary deposit in this location and given the presence of visible voids revealing the interior of the mound, which were observed to contain no cultural material. It is unlikely that ASM would not have noticed this mound during their documentation, indicating it may have been constructed since that time, or from 30-80 cm. It is in fair condition, exhibiting some areas of collapse on the northern and attributed to the bulldozing noted to the south by Gastilo and Clark (2018:145).

The presence of traditional artefactual materials (adze, coral abrader) and radiocarbon analysis work environment. Furthermore, research about land use in this part of the project area indicates pre-Contact habitation was likely focused to the east of Kahilipalinui, with agricultural activity being the more predominant land use in the proposed treatment facility vicinity. The section of galvanized metal pipe observed at the site previously by ASM may have been coincidental to the presence of the nearby jeep road, particularly as no other remnants of a historic pipeline were observed. Regardless, the site has very likely experienced impacts and modification from historic SIHP # -30929 is assessed as a pre-Contact activity area, likely associated with agriculture, with of charcoal found within TU-1 at SIHP # -30929 Feature A indicate the site is likely pre-Contact or early post-Contact in age. While the kukui fragments found in TU-1 may have been naturally and ongoing modern land use. Based on the past (Gastilo and Clark 2018) and current observations, deposited by trees within the depression, the remaining materials are indicative of a traditional continued historic use.

related disturbance. The site retains integrity of location, design, setting, materials, and workmanship. The prior assessment of significance by Gastilo and Clark (2018) was not available. Based on the current findings it is assessed as significant under Criterion d for the information it SIHP # -30929 is in overall fair condition, based on some areas of minor collapse and treehas provided about traditional and historic land use in the project area (see Section 8.1). The available records suggest SIHP # -30929 has not been previously evaluated for eligibility to be listed on the National Register. Based on the findings of the current AIS the site is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and the site is recommended for no further work (see Section 8.2).

AISR for the Na alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

6.4 SIHP # 50-10-74-309.	30
FORMAL TYPE:	Rock Wall
FUNCTION:	Ranching
NUMBER OF FEATURES:	1
AGE:	Historic
TEST EXCAVATIONS:	None
TAX MAP KEY:	(3) 9-5-007:016
PREVIOUS DOCUMENTATION:	Gastilo and Clark (2018; Draft)
SIHP # -30930 is a historic co facility portion of the project area in an area used for pasture, where outcrops, and Christmasberry and the SIHP # -3 1269 ranching comp	re-filled rock wall bisecting the proposed wastewater treatment (see Figure 46 and Figure 136 through Figure 139). The wall is $:$ the terrain is gently sloping to the south with exposed bedrock lantana are common. SIHP # -30930 provides the backbone for olex.
SIHP # -30930 was previously	documented by ASM (Gastilo and Clark 2018) as follows:
Site 30930 is a core-fille northern section of the ci stacked pähoehoe cobbles 1.1 meters wide and up to with Site 26410 while it: collapsed in some areas an Site 30929 Based on its fi	d wall that is located near a bulldozed pasture in the urrent study area The wall is constructed of large and extends 920 meters to the east/west It measures o 1.2 meters tall. The western end of the wall intersects is eastern termination adjoins Site 29507. The wall is disinterrupted by a metal gate in an area that is north of and attributes. Site 30970 was likely used for ranchine
purposes during the first p	art of the 20th century. [Gastilo and Clark 2018:149]
The full Gastilo and Clark (201 CSH observed SIHP # -30930 to west for an overall distance of a filled construction, with large bas a fill of smaller coboles. The wall an SIHP # -30930 continues west wall terminates just west of SIH1 ranch wall, SIHP # -29505 may have been of ranch buildings. Several other bre of the wall at the eastern end of S 31206 feature D canned heavidt	8) draft site description is provided in Appendix F for reference. be as described by ASM. SIHP $\#$ -30930 trends roughly east- proximately 375 m in the project area. It utilizes classic core- alt cobbles and small boulders forming the outer wall faces and 1 is an average of 1.0 m wide with heights ranging from 1.1–1.3 t beyond the project area an unknown distance. To the east, the # -29507. Clark et al. (2013) documented another core-filled ding east from SIHP $\#$ -29507; at one time SIHP $\#$ -30930 and ne continuous wall that was interrupted by development of the eaches were observed within the current project area. A portion SIHP $\#$ -31260 was repurposed for construction of the SIHP $\#$ mp (see Section 6.6). Another hered is present where the SIHP
20-m-wide breach is present appl	roximately 60 m west of the SIHP # -31270 passage; this area

contains considerable amounts of modern trash and may have been used as a modern dump site. The breach at this latter location has been repaired using modern wire fencing and t-posts, indicating the wall still functions as a livestock fence. Aside from this modern trash, no cultural CSH observed SIF west for an overal filled construction, a fill of smaller col of the wall at the e 31269 Feature D c # -31270 jeep road m. SIHP # -3093(wall terminates ju ranch wall, SIHP SIHP # -29505 m The full Gastil ranch buildings. 20-m-wide bread

AISR for the Na alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

136

Site Descriptions

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



Figure 135. Photograph of SIHP # -30929 Feature C rock mound; view to the northwest

Site Descriptions



Figure 136. Photograph overlooking the western section of the SIHP # -30930 rock wall; view to northeast



Figure 137. Photograph showing typical construction along a section of the SIHP # -30930 rock wall; view to the north

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

138

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 138. Photograph showing core-filled construction along a section of the SIHP # -30930 rock wall; view to north



Figure 139. Photograph of the breach in the SIHP # -30930 rock wall, and the modern fencing used for repair; view to west

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Site Descriptions	Cultural Surveys Hawai't Job Code: KAHILIPAL	LINUI 6 Site Descriptions
material was observed on or around the site. Excavation would be unlikely to yield information	6.5 SIHP # 50-10-74-3126	8
unpacturg trie assessment of site age and function. STHP # .30030 is a historic ranchine wall likely constructed as a cattle fence. The wall me-	FORMAL TYPE: I	Ditch
dates the SIHP # -31269 ranching complex and, based on the presence of a similar wall (SIHP #	FUNCTION:	Storm Drainage
-29505) on the parcel to the west, it may also pre-date the SIHP # -29507 ranching complex.	NUMBER OF FEATURES: 1 AGE: 1	1 Historic
intermittent breaches and repurposing of component material elsewhere, the site retains integrity	TEST EXCAVATIONS:	None
of location, design, setting, materials, workmanship, feeling, and association. The prior assessment of significance by Gastilo and Clark (2018) was not available. Based on the current findings it is assessed as significant under Criterion d for the information it has provided about historic ranching activity in the project area (see Section 8.1).	TAX MAP KEY: PREVIOUS DOCUMENTATION:	(3) 9-5-022:001 None
The available records suggest SIHP $\#$ -30930 has not been previously evaluated for eligibility to be listed on the National Register. Based on the findings of the current AIS the site is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and the site is recommended for no further work (see Section 8.2).	SIHP # -31268 is a historic earl proposed treatment facility site (see of this ditch in order to divert sto southeast (see Figure 10). The ditch the <i>makai</i> side of Māmalahoa Hig portion of the ditch is outside the pr area dominated by tall guinea grass	then drainage ditch located in the northwestern portion of the Figure 45 and Figure 46). Project plans call for redevelopment form water away from the proposed treatment facility to the h extends from the vicinity of the Latter-Day Saints church on ghway downslope to the south-southwest; the northernmost roject area limits (see Figure 46). The site is situated within an s and scattered trees.
	SIHP # -31268 measures 280 m defined by two parallel, wide earthe width from approximately 3-7 m. surrounding ground surface; and ur trees runs along the northern half of mildly to the south-southwest. An o channel: the spillway has a height o the site.	1 long by 15-20 m wide (Figure 140 through Figure 144). It is en berms set 5-8 m apart. The gradually sloping berms range in 1. The exterior berm edges are generally 1–2 m above the p to 2.8 m above the interior channel floor. A series of small of the eastern berm. The interior of the ditch is level and slopes earthen spillway is located at about the midpoint of the ditch of up to 40 cm. No cultural material was observed on or near
	SIHP # -31268 is likely historic Figure 19). While the site contains a material used for berm construction within the ditch would likely be sw assessed as low. Two exploratory t 31268 but did not yield any cultural	c in age. It appears to be visible on the 1978 orthophoto (see areas of deep sediment, these sediments are primarily local fill on and runoff accumulation. Any cultural materials deposited wept away by storm water. Therefore, excavation potential is test trenches (TT-1 and TT-2) were excavated near SIHP $\#$ - II materials.
	SIHP #-31268 is in good conditi retains integrity of location, design, assessed as significant under Criter control activity in the project area (s	tion and continues to function to channel storm water. The site , setting, materials, workmanship, feeling, and association. It is rion d for the information it has provided about historic water (see Section 8.1).
	SIHP # -31268 is evaluated as ne content has been adequately docum 8.2).	tot cligible for listing on the National Register. Its information neuted, and it is recommended for no further work (see Section
AISR for the Nä ⁻ älehu Cesspool Closure Project, Kähtlipalinui and Käwala, Ka ⁻ ŭ, Hawai ⁻ i Isłand 140 TMKs: multiple	AISR for the Nä älehu Cesspool Closure Project TMKs: multiple	ct. Kähilipalimui and Kāwala, Ka'ū, Hawai'i Island

Site Descriptions



Figure 140. Plan view of SIHP # -31268, drainage ditch

AISR for the Nā ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

142

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



Figure 141. Photograph of SIHP # -31268 taken atop the earthen berm forming the eastern ditch embankment; view to north



Figure 142. Photograph of SIHP # -31268, taken from within the ditch and looking toward the western berm (scale is leaning against interior berm wall); view to northwest

AISR for the Nã älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Site Descriptions



Figure 143. Photograph of SIHP # -31268 interior ditch channel, looking upslope from the southern end of the ditch; view to north



Figure 144. Photograph of earthen spillway within SIHP # -31268 ditch; view to northwest

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ù, Hawai'i Island TMKs: multiple

144

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

6.6 SIHP # 50-10-74-31269

FORMAL TYPE:	Complex
FUNCTION:	Ranching
NUMBER OF FEATURES:	4
AGE:	Historic
TEST EXCAVATIONS:	None
TAX MAP KEY:	(3) 9-5-007:016
PREVIOUS	None
DOCUMENTATION:	

SIHP # -31269 is a historic ranching complex located in the eastern portion of the proposed wastewater treatment facility site (see Figure 46). The site is situated in a relatively level, grassy area with occasional bedrock outcrops used for pasture (see Figure 36). It is comprised of four interconnected features including a large, bisected paddock (Feature A), adjoining holding pens and contals (Features D) (rigure 145). Overall, SIHP # -31269 measures approximately 65 m (NS) by 185 m (E/W). Its component features are constructed along the historic SIHP # -3093 to ck wall, which continues east and west beyond the limits of SIHP # -31269 (see Figure 46 and Figure 145). The SIHP # -31270 jeep road extends through the interior of the Feature A paddock (see Figure 46 and Figure 145). No portable artifacts or cultural deposits were encountered throughout the site, though scattered modern trash was observed. Some abandoned vehicles are present outside the Feature A paddock to the north.

Feature A is a large, irregularly shaped livestock paddock formed by a series of fences abutting the northern side of the SIHP # -30930 rock wall (see Figure 46 and Figure 145 through Figure 149). The paddock measures approximately 180 m long (E/W) by 30 m wide (N/S) with fence heights of 140 cm. It is bisceted by an interior fence with an access gate at its northern end. The fence lines comprising the paddock are constructed of hog and barbed wire, milled and un-milled wooden posits, metal "T" posits, and a galvanized metal gates, and occasionally incorporate other through one of two access gates at the northeastern comer, the gate in the bisecting fence line, and another gate situated within the SIHP # -31070 jeep road extends through the paddock through one of two access gates at the northeastern comer, the gate in the bisecting fence line, and another gate situated within the SIHP # -30930 wall at the southwestern corner of the paddock (see Figure 145). A functioning metal water trough is situated along the easternmost fence, approximately 10 m south of the gate at the northeastern paddock comer; this trough measures 3.10 m long, 0.85 m wide, and 0.64 m high.

Feature B consists of two adjoining rectangular pens situated east and south of Features C and D, respectively (see Figure 46, Figure 145, Figure 150, and Figure 151). The pens, which served to hold cattle waiting to be loaded onto the Feature D ramp, are constructed using large metal posts and tubes. The western pen measures approximately 7 m long (N/S) by 5 m wide (E/W). The pens have similar hields for approximately 2 m long (N/S) by 5 m wide (E/W). The pens have similar hields for approximately 2 m long (N/S) by 6 m wide (E/W). The pens have similar hields for approximately 2 m long (N/S) by 6 m wide (E/W) is situated in the northwestern comer of the eastern pen (N/S) by 0.6 m wide (E/W) is situated in the northwestern comer of the eastern pen (S/S) and has been recorded as Feature G of that site (see Figure 151). A wire fence extending from the southeastern comer of fracture B circles back around toward SIHP # -29507 and has been recorded as Feature G of that site (see Figure 46 and Section 6.1).

AISR för the Nä*älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



ALSR for the NA' Mehn (TMKs multiple

146



Figure 146. Photograph overlooking the SIHP # -31269 Feature A paddock from its northeastern corner; view to north



Figure 147. Photograph overlooking SIHP # -31270 jeep road extending through the eastern portion of the SIHP # -31269 Feature A paddock; view to west

AISR for the Na alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka 'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 148. Photograph of water trough along eastern side of SIHP # -31269 Feature A paddock; view to north



Figure 149. Photograph overlooking the western portion of the SIHP # -31269 Feature A paddock (SIHP # -31270 jeep road visible at center); view to east

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

148

Figure 150. Photograph of overlooking the SIHP # -31269 Feature B holding pens; view to northwest



Figure 151. Photograph of interior of the western SIHP # -31269 Feature B holding pen, with concrete trough visible at center; view to east

AISR for the Nä"alchu Casspool Closure Project, Kähilipalinui and Käwala, Ka"u, Hawai'i Island TMKs: multiple

Site Descriptions

Feature C is a series of rectangular, gated corrals situated south of Feature A and directly west of Features B and D (see Figure 46, Figure 145, and Figure 152 through Figure 155). It comprises three discrete corrals: two parallel corrals accessed from within the Feature A paddock, and another corral to the south accessed only from outside of Feature A. Overall, Feature C measures approximately 30 m long (N/S) by up to 14 m wide. Heights range from 160–200 cm. The northwestem corral measures approximately 18 m long by up to 6 m wide. The southern corral measures approximately 10-m-set. A 1-m-wide gated corridor separates the southern corral measures approximately 10-m-set. A 1-m-wide gated corridor separates the southern corral measures approximately 10-m-set. A 1-m-wide gated corrals, range wire, and repuposed gates.

Feature D is a concrete and stone cattle loading ramp and associated squeeze chute located in the southeastern portion of the SIHP # -31269 complex (see Figure 46, Figure 155) through Figure 157). The rectangular-shaped ramp is constructed with stacked subangular basalt cobles and boulders capped with concrete. The basalt stone material appears to have been sourced from an adjacent portion of the SIHP # -30930 rock wall. Cattle would have been fumeled onto the ramp from the Feature B holding pens via the squeeze chute and loaded onto a vehicle parked within the Feature A paddock. The ramp portion of Feature D measures 5.5 m long (NW/SE) by 2.5 m wide (NE/SW). The ramp portion of the clutte leading from the feature B ports to the ramp bortion is approximately 6 m long by 3 m wide; it is constructed of metal posts and measures utilized on the platform. The loading ramp incorporates an extension of the clutte, with plywood sidewalls to 1.6 m high. The loading ramp incorporates an extension of the clutte, with plywood sidewalls utilized on the platform.

SIHP # -31269 is a historic ranching complex, likely part of the larger defunct Onipa'a Ranch. The ground surface within the site had been heavily impacted by livestock and vehicular activity, as indicated on the 1978 orthophoto (see Figure 19). Excavation would be unlikely to yield information impacting the assessment of site age and function. Three exploratory test trenches first. 4.5, and -7) were excavated in areas surrounding but not within SIHP # -31269 and did not yield any cultural materials.

Overall, SIHP # -31269 is in good, maintained condition. The site exhibits evidence of repurposing of materials throughout. The presence of cattle at the site indicates it is still in active use to some extent. Despite continued maintenance and incorporation of repurposed and modern material in places, the site retains integrity of location, design, setting, materials, workmanship, feeling, and association. It is assessed as significant under Criterion 4 for the information it has provided about historic land ranching activity the project area (see Section 8.1).

SIHP # -31269 is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and it is recommended for no further work (see Section 8.2).

AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

150

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



Figure 152. Photograph overlooking SIHP # -31269 Feature C corrals; view to east



Figure 153. Photograph of the southern corral at SIHP # -31269 Feature C; view to northwest

AISR för the Nä*älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Figure 154. Photograph of the corridor between the northern and southern corrals at SIHP # - 31269 Feature C; view to east north



Figure 155. Photograph showing the access to the northeastern corral at SIHP # -31269 Feature C adjacent to the Feature D loading ramp; view to south

AISR for the Nā ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

152

Figure 156. Photograph overlooking the SIHP # -31269 Feature D loading ramp and squeeze chute from within the northeastern Feature C corral; view to northeast



Figure 157. Photograph overlooking the SIHP # -31269 Feature D loading ramp and squeeze chute with Feature B holding pens visible at background right; view to southeast

AISR for the Nä'alehu Cesspool Clesure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Site Descriptions

6.7 SIHP # 50-10-74-31270

FORMAL TYPE:	Road
FUNCTION:	Transportation
NUMBER OF FEATURES:	1
AGE:	Historic
TEST EXCAVATIONS:	None
TAX MAP KEY:	(3) 9-5-007:016 and 9-5-010:001
PREVIOUS	None
DOCUMENTATION:	

SIHP # -31270 is a historic jeep road that extends along a portion of the proposed gravity sever line corridor and through the proposed treatment facility site in the southern portion of the project area (see Figure 46, Figure 147, Figure 149, and Figure 158 through Figure 161). The road is accessed from Na² alehu Spur Road (SIHP # -31271) just south of the shopping center, extending through the gated Taylor Built Construction yard, and continuing south through another gate into through the gated Taylor Built Construction yard, and continuing south through another gate into through the SIHP # -31269 Feature A paddock and southwest through the SIHP # -30930 rock wall, continuing downslope outside of the project area to the southwest (see Figure 46). Vegetation in the areas surrounding through the ond the rode for pasture. These areas are actively used for pasture. SIHP # -31270 is a "jeep" road with a predominately dirt and gravel travel surface measuring approximately 3.4 m wide. The portion of the road extending between the Taylor Built Construction yard and the Onipa'a Ranch gate exhibits patches of remnant asphalt but has generally devolved into a dirt roadway. An approximately 775 m-long portion of the road is within the project area. No cultural material was observed on or near the site. Excavation potential is assessed as poor due to the nature of the site (dirt road).

SIHP # -31270 is used for transportation, specifically to access the Onipa'a Ranch and other areas downslope from the Nā'alehu Spur Road (SIHP # -31271). The road is in good condition with continued use and maintenance. The road appears as a 'jeep trail" on the 1962 Nā'alehu UGGS topographic map (see Figure 18), meaning it is an historic road alignment. It is also visible on a the 1978 Nā'alehu orthophoto quad map (see Figure 19). The initial date of constructor for durins road is unknown, but it likly was developed in association with the historic reath and associated infrastructure in this area around the turn of the twentieth century (see Section 6.1).

SIHP # -31270 is in good condition with continued maintenance and use. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. It is assessed as significant under Criterion d for the information it has provided about historic transportation in the project area (see Section 8.1).

SIHP # -31270 is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and it is recommended for no further work (see Section 8.2).

AISR for the Näʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawai'i Island TMKs: multiple

154

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



Figure 158. Photograph overlooking the portion of SIHP # -31270 jeep road near its intersection with Na^{*}älehu Spur Road; view to south



Figure 159. Photograph overlooking the portion of SIHP # -31270 jeep road between the Taylor Built Construction yard and the Onipa's Ranch entry gate; view to north

AISR for the Na^{*}älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'u, Hawai'i Island TMKs: multiple

	6.8 SIHP # 50-10-74-312	12
4	FORMAL TYPE:	Nāʻālehu Spur Road
	FUNCTION:	Transportation
12 - 2 - 1 - 1 - 1 - 1 - 1 - 1 - 1 - 1 -	NUMBER OF FEATURES:	
	AGE:	Historic
and the second se	TEST EXCAVATIONS:	None
	TAX MAP KEY:	(3) 9-5-010:001
	PREVIOUS DOCUMENTATION:	None
	SIHP # -31271 is the historic Figure 46, Figure 162, and Figure of the proposed sewer line corric passes a shopping center and out project area into ranch lands used	Vă'ătehu Spur Road located <i>makai</i> of Māmalahoa Highway (see 163). A small section of this roadway corresponds with a portion or. From the Māmalahoa Highway intersection, SIHP #-31271 er commercial establishments, continuing southeast beyond the 1 for pasture.
	An approximately 81.5-m-lon, of the road exhibits a $6-7$ m-wide	g portion of SIHP # -31271 is within the project area. This section asphalt travel surface with no striping or sidewalks except at the
oad <i>makai</i> of SIHP #s	Māmalahoa Highway intersectio Ranch Road and veers southeast travel surface with no striping	n. Where the road diverges from the SIHP # -31270 Onipa'a outside of the project area, it narrows to a 5-6-m-wide asphalt or sidewalks. This section of the road extends southwest
	approximately 200 m, at which po No cultural material was observe to the nature of the site (asphalt r	bint it continues an unknown distance as an unimproved dirt road. I on or near the site. Excavation potential is assessed as poor due bad).
	SIHP # -31271 is used for tran and ranch lands located <i>makai</i> of on both the 1071 and 1967 Na ⁵	sportation, specifically to access the commercial establishments Māmalahoa Highway. It appears as a substantial road alignment siehu TISGS tonoorashic mans (see Figure 17 and Figure 18)
	Notably, Figure 18 (and the 1978 at that time turned back east tow	Natistic for the production of the production of the product of the SIHP # -29507 Onipata Ranch complex, with a jeep road
	also continuing southeast along indicates this former portion of t	the current alignment. Modern aerial imagery (see Figure 8) he road, which is located outside the project area bounds, is no
	longer in use. The initial date of o was developed in association wit early twentieth century prior to 1	onstruction for the Na' alenu Spur Koad is unknown, but it likely a the historic plantation town and subsequent ranches during the 221 (see Section 3.1.5).
7	SIHP # -31271 is in fair to goo integrity of location, design, se assessed as significant under C	d condition with continued maintenance and use. The site retains ting, materials, workmanship, feeling, and association. It is riterion d for the information it has provided about historic
	transportation in the project area	(see Section 8.1).
	SIHP # -31271 is evaluated as content has been adequately docu	not eligible for listing on the National Register. Its information mented, and it is recommended for no further work (see Section
s along the SIHP # ; view to northeast	8.2).	
156	AISR for the Nä älehu Cesspool Closure Pro	ect, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island
	1 IVINS: Intutpre	

Figure 161. Photograph of SIHP # -31270 jeep road passing through a gate -30930 rock wall and into the SIHP # -31269 Feature A paddock;

AISR for the Na'alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Figure 160. Photograph overlooking a portion of the SIHP # -31270 jeep ro -30930 and -31269; view to south



ultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	Site Descriptions	Cultural Surveys Hawai'i Job Code: KAHILJI	PALINUI 6 Site Descripti	otions
		6.9 SIHP # 50-10-74-312	272	
		FORMAL TYPE:	Road	
		FUNCTION:	Transportation	Τ
		NUMBER OF FEATURES:		
		AGE:	Historic	[
	A	TEST EXCAVATIONS:	None	
	×1.	TAX MAP KEY:	(3) 9-5-008:033	
		PREVIOUS DOCUMENTATION:	None	
		SIHP # -31272 is the histori Figure 46, Figure 164, and Figure of the proposed sewer force ma extends between the Māmalahoa drainage channel, passing severa beyond the project area limits to	c Ka'ala'iki Road located mauka of Māmalahoa Highway (s 165). A small section of this roadway corresponds with a portion in corridor. The section of this roadway within the project at I elighway and the gravel road used to access the SIHP # -312 al commercial establishments and residences. The road continu the northeast.	(see tion area 273 nues
Figure 162. Photograph overlooking the portion of SIHP # -31271 (Nāʿālehu Spur R the project area; view to north	aod) within	An approximately 165-m-lon of the road exhibits a 7–8-m-wid is worn and in need of repaving- potential is assessed as poor due	g portion of SIHP $\#$ -31272 is within the project area. This secti e asphalt travel surface with no striping or sidewalks. The asph No cultural material was observed on or near the site. Excavati to the nature of the site (asphalt road).	tion halt tion
		SIHP # -31272 is used for th businesses and homes <i>mauka</i> of Pähala to the east when the high major transportation route betwe and operations of sugar plantatio this alignment may be visible on	ansportation. The road is currently utilized primarily to accc Māmalahoa Highway, or as a detour route between Nāʿālehu a hway is closed due to flooding. For a time, this roadway was en Nāʿālehu and Pāhala and was instrumental in the developme ns in the area (see Section 3.1.4). Some of the earliest sections maps from the late 1800s (see Figure 14 and Figure 15).	cess and as a nent is of
		SIHP #31272 is in fair to goo integrity of location, design, s assessed as significant under C transportation in the project area	od condition with continued maintenance and use. The site retai etting, materials, workmanship, feeling, and association. It Triterion d for the information it has provided about histor (see Section 8.1).	ains lt is oric
	7/	SIHP # -31272 is evaluated a content has been adequately doct 8.2).	s not eligible for listing on the National Register. Its informati umented, and it is recommended for no further work (see Secti	tion
Figure 163. Photograph overlooking the portion of SIHP # -31271 (Nāʿālehu Spur R. the project area; view to south	oad) within			
AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple	158	AISR for the Nāʿālehu Cesspool Closure Pr TMKs: multiple	jeet, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island	159

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

9	
Б	
Z	
5	
P	
E.	
Н	
Ξ	
3	
-	
qe	
ŭ	
à	
Jo	
5	
va	
Iav	
1	
S	
Ň	
Su	
- Te	
nr	
ult	
0	



Figure 164. Photograph overlooking the portion of SIHP # -31272 (Ka'alaiki Road) within the project area; view to north



Figure 165. Photograph overlooking the portion of SIHP # -31272 (Ka'alaiki Road) within the project area; view to southwest

AISR for the Nä älehu Cesspool Closure Project, Kähilipalimui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

160

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions

6.10 SIHP # 50-10-74-31273

FORMAL TYPE:	Complex
FUNCTION:	Water control
NUMBER OF FEATURES:	2
AGE:	Historic
TEST EXCAVATIONS:	None
TAX MAP KEY:	(3) 9-5-025:002
PREVIOUS DOCUMENTATION:	None

SIHP # -31273 is a historic water control complex overlapped by the central portion of the project area west of the main residential neighborhood (see Figure 45). The site is crossed by the proposed sewer force main corridor. The site comprises two features: Feature A is concrete drainage channel used to carry storm runoff from upslope safely through the town, and Feature B is a foobridge that crosses the channel. The proposed sewer force main corridor will extend across HIPH \pm -3173 in the vicinity of the footbridge. The site is surrounded by residential properties. No cultural material was observed within the portions of the site overlapped by the project area.

Feature A is an open concrete drainage channel extending downslope from a debris catch basin. Feature A plans are provided in Appendix G. Storm runoff accumulates in the catch basin until it overflows into a spillway serving as the channel inlet. Water flows down the channel past the main residential area, through a culvert under the Mämalahoa Highway, and through a continuation of the open concrete channel along the eastern side of Na² albih Park. The concrete channel past the along either side of the channel and the eastern side of Na² albih Park. The concrete channel prevent unauthorized access. The channel is of m wide with vertical sidewalls and has a depth of up 0.12 m below the surrounding ground surface. An approximately 40-m-long section of the overall channel is situated within the project area (Figure 166 and Figure 167). The concrete is cracked in places and weeds are growing along the edges of the channel. The archaeologists did not have access to the interior of the channel during fieldwork.

Feature B is a concrete footbridge crossing the Feature A channel (Figure 168 and Figure 169). The footbridge is located between the western terminus of Melia Street and the gravel access road located on the eastern side of Ka'alaiki Road. The footbridge comprises a concrete slab measuring 8 m long by 2 m wide and 10 cm thick. On the southern side of the footbridge is a stylized concrete railing measuring 1.1 m high and 15 cm wide. On the northern side of the footbridge is a metal handrail measuring 1.2 m high.

SIHP # -31273 is a historic water diversion channel and associated footbridge. The original channel plans (see Appendix G) indicate the site was constructed in the 1960s, making it over 50 years old. Excavation potential was assessed as poor due to the nature of the site.

SIHP #-31273 is in overall good condition with continued maintenance and use. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. It is assessed as significant under Criterion d for the information it has provided about historic water control activity in the project area (see Section 8.1).

AISR for the Na'älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Site Descriptions



Figure 166. Photograph overlooking the portion of SIHP # -31273 Feature A drainage channel within the project area; view is to north from Feature B footbridge as archaeologists did not have access to the channel



Figure 167. Photograph overlooking the portion of SIHP # -31273 Feature A drainage channel within the project area; view is to southeast from Feature B footbridge as archaeologists did not have access to the channel

AISR för the Nå ädehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

162

Suc

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Site Descriptions



Figure 168. Photograph overlooking the SIHP # -31273 Feature B footbridge toward Melia Street in the background; view to northeast



Figure 169. Photograph showing the stylized concrete railing at the SIHP # -31274 Feature B footbridge; view to southeast

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Site Descriptions	Cultural Surveys Hawai'i Job Code: KAHILJ	IPALINUI 6 Site Descriptions
ional Register. Its information	6.11 SIHP # 50-10-74-3	1274
or no further work (see Section	FORMAL TYPE:	Rock Wall
	FUNCTION:	Boundary
	NUMBER OF FEATURES:	
	AGE:	Historic
	TEST EXCAVATIONS:	None
	TAX MAP KEY:	(3) 9-5-008:048
	PREVIOUS DOCUMENTATION:	None
	SIHP # -31274 is a historic w proposed pump station site at TN 45, Figure 170, and Figure 1711. which is delineated by a fence co area is fairly level fallow pastur eastern portion of the walt, and	vall located along a portion of the property boundary between the M(K: (3) 9.5-008/048 and the Igtesia Ni Cristo Church (see Figure The wall appears to be located 1 m inside the parcel 048 boundary, norstructed of repurposed metal roofing material. The terrain in this re, with head-high guine grass. A large avocado tree shades the the westem portion was obscured by dead grasses.
	SIHP # -31274 is a low, lineau and small boulders. It measures I strand barbed wire has been ple barbed wire is supported with ru of collapse. No cultural materia lack of surface artifacts and asse	r wall constructed of $2-3$ courses of loosely stacked basalt cobbles 6.2 m long (E/W), 1.0 m wide, with heights of $30-70$ cm. A four- aced along the northern side of the wall facing the pasture. The sisted metal t-posts. The wall is in poor condition, with some areas $.1$ was observed on or near the site. Excavation is poor due to the essed function of the site.
	SIHP # -31274 is a historic v the church property. This wall a wire appears to have been adde impacts to the wall occurred. It is a remnant of what was once a	wall likely constructed to keep grazing livestock from impacting appears to predate the metal roofing fence behind it. The barbed cd later to keep livestock off the low wall, potentially after some is unclear why this wall extends only a short distance; perhaps it longer wall.
	Despite its poor condition, materials, and workmanship. It has provided about historic land	SIHP # -31274 retains integrity of location, design, setting, is assessed as significant under Criterion d for the information it luse in the project area (see Section 8.1).
	SIHP # -31274 is evaluated a content has been adequately doc 8.2).	is not eligible for listing on the National Register. Its information sumented, and it is recommended for no further work (see Section
sland 164	AISR for the Nā älehu Cesspool Closure Pr TMK s: multiple	roject, Kähilipalinui and Kāwala, Ka'ū, Hawai'i İsland 165

SIHP # -31273 is evaluated as not eligible for listing on the Natic content has been adequately documented, and it is recommended for 8.2).

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Ka'ū, Hawai'i Isk TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Site Descriptions	Cultural Surveys Hawai'i Job Code: KAHIL/IPAL/NUI 6 Summary and Interpretation
	Section 7 Summary and Interpretation
	Fieldwork was conducted between 26 October 2020 and 29 January 2021. This work required approximately 19 person-days to complete. Fieldwork consisted of a 100% pedestrian survey of the project area, GPS data collection, and a program of subsurface testing. Overall, ground visibility was excellent throughout the project area, which comprises previously developed areas like roadways and wide-open pastures impacted by longstanding ranching activity.
	A total of eleven historic properties were documented during the survey, including portions or the entirety of four previously documented sites (SIHP #s -29507, ranching complex; -30187, Mamālahoa Highway; -30929, pre-Contact and historic complex; and -30930, historic ranch wall) and seven newly recorded sites (SIHP #s -31268, historic earthen drainage ditch; -31269, historic ranching complex; -31270, historic jeep road; -31271, historic Na [*] ālehu Spur Road; -31272, historic Ka'alaiki Spur Road; -31273, historic storm drainage complex; and -31276, historic
	boundary wall). That ten of the 11 documented sites date to the historic era is significant in illustrating the impacted nature of the project area lands, given the extent of land use in the Na'alehu area during pre-Contact times. The single site exhibiting pre-Contact origins, SIHP # -30929, was identified within the natural depression bisecting the proposed treatment facility site that appears to have avoided extensive impact by historical land use. The site does exhibit some evidence of later modification.
Figure 170. Photograph of SIHP # -31274 rock wall; view to southeast	Test excavations at SIHP #-30929 yielded traditional artifacts including an adze fragment made with basalt likely sourced from Kona or Ka'ū and a coral abrader. Charcoal was also collected during excavation at SIHP # -30929 and, following a cursory wood taxa analysis, two samples representing short-lived wood species were submitted for radiocarbon analysis. Notably, the two samples returned results with overlapping highest probability radiocarbon analysis. Notably, the two 1814 [Beta-592957 / 2a] and AD 1666 to 1783 [Beta-592958 / 2b]). These results support the interpretation that SIHP # -30929 originated as a pre-Contact activity area, likely associated with agricultural activity in Kahilipalinui.
	In addition to the manual test excavations at SHP # -30929, a program exploratory mechanical testing was conducted to assess the potential for subsurface archaeological features throughout the undeveloped portions of the project area. The exploratory testing program involved excavation of 14 test trenches, including 11 trenches in the proposed wastewater treatment facility site at TMK: (3) 9-5-007:016, and three trenches in the proposed pump station site at TMK: (3) 9-5-008:048. All but one trench was excavated to bedrock; at TT-1 sedimentary deposits exceeded the reach of the excavator (>260cmbs), likely due to its location near an earthen drainage channel (SIHP \pm -31268). In the other 13 trenches the depths of sediments ranged from approximately 20 cmbs in the proposed pump station of the project area to approximately 100 cmbs in the proposed pump station of the project area. No new historic properties or cultural deposits were encountered during the exploratory testing. Exploratory testing within the proposed
Figure 171. Photograph showing construction of SIHP # -31274 rock wall with barbed wire; view to south	wastewater treatment facility site exposed substantial layers of yellowish-hued silty clay loams throughout that are characteristically inconsistent with the rocky peats understood to comprise this area. Exploratory testing within the proposed pump station portion of the project area exposed sediments consistent with the Naalehu silty clay loam sediment type known to be present in this area.
AISR for the Näʿälchu Cesspool Closure Project, Kähilipalinui and Kāwala, Kaʿū, Hawaiʻi Island TMKs: multiple	AISR for the Nä älehu Cespool Closure Project, Kähilipalinui and Kāwala, Ka ū, Hawai'i Island 167 TMKs: multiple

(AHILIPALINUI 6 Significance Assessments and Eligibility Determinations	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Significance Assessments and Eligibility Determinations
ance Assessments and Nation Register vibility Determinations	under Criterion "A" for its association with "important late nineteenth and early twentieth events in establishing a regional transportation network that has its roots in antiquity" and Criterion "D" for the information activation of Dalucation of Dalucation of the second second for the second se
	for its information potential. Beliatomin and frammat (2017.34) note. It is the understanding of this study that the Clark et al. (2014) assessment of significance was explicitly under HAR "\$13-
IT previously documented and seven newly documented historic ca, for a total of 11sites in the project area. Section 8.1 provides der HRS §6E, while Section 8.2 provides National Register eligibility	284-6" as significant under Criteria a and d. This report supports the Clark et al. (2014) assessment of significance. The current study also supports these past assessments of significance. Therefore, pursuant to HAR §13-275-6, the portion of SIHP # -30187 in the current project area is assessed as significant under Criterion a for its association with events that have made an important
ssments under HRS §6E	contribution to the development of transportation routes on Hawaii Island, and under Criterion d for having yielded information about historic transportation in the project area.
storic property to be significant under HAR §13-275-6 (applicable to	8.1.3 SIHP # 50-10-74-30929
istoric property should possess integrity of location, design, setting, eling, and/or association, and meet one or more of the following	SIHP # -30929 (pre-Contact activity area and historic ranching complex) is in overall fair condition, based on some areas of minor collapse and tree-related disturbance. The site retains
it with events that have made an important contribution to the broad in history;	integrity of location, design, setting, materials, and workmanship. The prior assessment of significance by Gastilo and Clark (2018) was not available. Based on the current findings it is assessed as significant under Criterion d for the information it has provided about traditional and
I with the lives of persons important in our past;	ILISTOTIA TARIA USA ILI ALA ALAA. 0.1.4. CITTIA # 20.10.7.4. 2000.00
distinctive characteristics of a type, period, or method of construction,	
	SITT #
, or is likely to yield, information important for research on prehistory	use. Despite its internitient oreacities and repurposing of component material elsewhete, the suc- retains integrity of location, design, setting, materials, workmanship, feeling, and association. The
ortant value to the native Hawaiian people or to another ethnic group the to associations with cultural practices once carried out, or still	prior assessment or significance by Castulo and Clark (2016) was not available. based on the current findings it is assessed as significant under Criterion d for the information it has provided about historic ranching activity in the project area
ture property of oue to associations with traditional benetics, events of s- 	8.1.5 SIHP # 50-10-74-31268
ity.	SIHP # -31268 (historic earthen drainage ditch) is in good condition and continues to function
	to channel storm water. The site retains integrity of location, design, setting, materials, workmanship, feeline, and association. It is assessed as significant under Criterion d for the
ranching complex) is in remnant to good condition. The site retains setting materials workmanshin feeling and association Clark et al	information it has provided about historic water control activity in the project area.
SIHP # -29507 as significant under Criterion d for the information it	8.1.6 SIHP # 50-10-74-31269
nd use and recommended the site for no further work. The results of	SIHP # -31269 (historic ranching complex) is in overall good, maintained condition. The site
the prior assessment made by Clark et al. (2013). The three newly the current project area are likewise assessed as significant under	exhibits evidence of repurposing of materials throughout. The presence of cattle at the site indicates it is still in active use to some extent. Despite continued maintenance and incorporation
on they have yielded about historic ranching activity.	of repurposed and modern material in places, the site retains integrity of location, design, setting, materials, workmaship, feeling, and association. It is assessed as significant under Criterion d for the information is the material device the integral device and integrate and integrate and integrate and for
of SIHP # -30187 (Māmalahoa Highway) within the current project condition as an active roadway. Based on the nature of the roadway in	the functionation is have been about instoric tand randing activity the project at ea. 8.1.7 SHIP # 50-10-74-31270
rising the historic corridor only and absent any other contributing	SIHP $\#$ -31270 (historic jeep road) is in good condition with continued maintenance and use.
lverts, or guardrails, the site retains integrity of location only as all the original Māmalahoa Highway are no longer evident today. SIHP # ssed as significant by Clark et al. (2014;81) per HAR "§13-284-6"	The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. It is assessed as significant under Criterion d for the information it has provided about historic transportation in the project area.
	V.D. F4. NS-4. 1. S
ure Project, Kähilipalmui and Käwala, ka ti, Hawar'i Istand	AbsK for the Na alebu Cesspool Closure Project, Kabilipalmui and Kawata, Ka'u, Hawa'i I Istand TMKs: multiple

Signific Ξ Section 8

properties in the project an significance assessments und determinations. This AIS identified for

8.1 Significance Ass

Under HRS §6E, for a hi government projects), the h materials, workmanship, fe significance criteria:

- Be associated patterns of our в
 - Be associated Ъ
- ပ
- Embody the d represent the v
- Have yielded, or history; or р
 - Have an impo of the state d carried out, at oral accounts cultural identi o

8.1.1 SIHP # 50-10-74-2950

SIHP # -29507 (historic ra integrity of location, design, st (2013:1392, 1397) assessed S has yielded about historic land the current survey support th documented features within t Criterion d for the information

8.1.2 SIHP # 50-10-47-3018

area is in good, maintained co this specific location, compri elements such as bridges, culv constructed elements of the o -30187 was previously assess The 15-m (50-ft) portion

tural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Significance Assessments and Eligibility Determinations	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Significance Assessments and Eligibility Determinations
8 SIHP # 50-10-74-31271 SIHP # -31271 (historic Nāʿālehu Spur Road) is in fair to good condition with continued ntenance and use. The site retains integrity of location, design, setting, materials, kmanship, feeling, and association. It is assessed as significant under Criterion d for the	eligibility to be listed on the National Register. Based on the findings of the current AIS the site is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and the site is recommended for no further work. 8.2.2 SIHP # 50-10-47-30187
ormation it has provided about historic transportation in the project area. 9 SIHP # 50-10-74-31272	As noted in Section 8.1.2, the 15-m (50-ft) section of SIHP #-30187 (Māmalahoa Highway) in the project area retains integrity of location only as all of the constructed elements of the original
SIHP # -31272 (historic Ka'ala'iki Road) is in fair to good condition with continued intenance and use. The site retains integrity of location, design, setting, materials, rkmanship, feeling, and association. It is assessed as significant under Criterion d for the ormation it has provided about historic transportation in the project. 10 SIHP #50-10-74-31273	Mämalahoa Highway are no longer evident today. SIHP # -30187 was previously evaluated as eligible for inclusion in the National Register (per 36 CFR 60.4) by Belluomini and Hammatt (2017.62) under Criterion A (for its association with events that have made a significant contribution to the broad patterns of our history) and Criterion D (that having yielded, or being likely to yield, information important in history). This evaluation of eligibility is based on the site's association with "important in history). This evaluation of eligibility is based on the site's
SIHP # -31273 (historic water control complex) is in overall good condition with continued internance and use. The site retains integrity of location, design, setting, materials, orkmanship, feeling, and association. It is assessed as significant under Criterion d for the cormation it has provided about historic water control activity in the project area. 11 SIHP #50-10-74-31274	transportation network that has its roots in antiquity." (Clark et al. 2014.81). The findings of the current AIS support these this prior evaluation of eligibility. While the project will affect the modern roadway elements, these elements will be replaced following construction, and the integrity and significance of the highway will not be diminished. Therefore, the project will have no adverse effect on SIHP $\#$ -30187 and no further work is recommended.
Despite its poor condition, SIHP #-31274 (historic boundary wall) retains integrity of location, eign, setting, materials, and workmanship. It is assessed as significant under Criterion d for the ormation it has provided about historic land use in the project area.	As noted in Section 8.1.3 SIHP # -30929 (pre-Contact activity area and historic ranching complex) is in vorteal fair condition, based on some areas of minor collapse and tree-related distributions. The cite activity refloating the following activity activity and mathematical produced frequencies.
2 National Register Eligibility Determinations	available records suggest SIHP # -30929 has not been previously evaluated for eligibility to be
Under Section 106, historic property significance is evaluated as eligibility for listing on the tional Register of Historic Places ("National Register") pursuant to 36 CFR 60.4. To be asidered eligible for listing on the National Register, a historic property should possess integrity described in Section 8.1 above, and meet one or more of the following broad significance	listed on the National Register. Based on the findings of the current AIS the site is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and the site is recommended for no further work. 8.2.4 SIHP #50-10-74-30930
terna. A That are associated with events that have made a significant contribution to the broad patterns of our history:	As noted in Section 8.1.4, SIHP # -30930 (historic ranch wall) is in overall fair condition with continued maintenance and use. Despite its intermittent breaches and repurposing of component material elsewhere, the site retains integrity of location, design, setting, materials, workmanship,
B That are associated with the lives of persons significant in our past;	feeling, and association. The available records suggest SIHP # -30930 has not been previously evaluated for elizibility to be listed on the National Register. Based on the findings of the current
C That embody the distinctive characteristics of a type, period, or method of construction, or that represent that work of a master, or that possess high artistic values, or that represent a significant and distinguishable entity whose components may lack individual distinction;	AIS the site is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and the site is recommended for no further work. 8.2.5 SIHP # 50-10-74-31268
D That have yielded, or may be likely to yield, information important in prehistory or history.	As noted in Section 8.1.5, SIHP # -31268 (historic earthen drainage ditch) is in good condition and continues to function to channel storm water. The site retains integrity of location, design, setting materials workmanshin feeling and association SIHP # -31268 is evaluated as not
2.1 SIHP# 50-10-74-29507	eligible for listing on the National Register. Its information content has been adequately
As noted in Section 8.1.1, SIHP # -29507 (historic ranching complex) is in remnant to good adition and retains integrity of location, design, setting, materials, workmanship, feeling, and ociation. The available records suggest SIHP # -29507 has not been previously evaluated for	documented, and it is recommended for no further work.
SR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island Mss: multiple	AISR for the Nä ätehu Cesspool Closure Project, Kähilipalimui and Kāwala, Ka'ū, Hawai'i Island 171 TMK si multiple

Cultural Surveys Hawai'i Job

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Significance Assessments and Eligibility Determinations	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 Project Effect and Mitgation Recommendations
8.2.6 SIHP # 50-10-74-31269	Section 9 Project Effect and Mitigation Recommendations
As noted in Section 8.1.6, SHPP # -31269 (historic ramehing complex) is in overall good, maintained condition. The site exhibits evidence of repurposing of materials throughout. The measures of onth or the chicken is critic action are to come over the context context.	9.1 Project Effect
presence of caure at the site indicates it is still in active use to some extent. Despite continued maintenance and incorporation of repurposed and modern material in places, the site retains integrity of location, design, setting, materials, workmanship, feeling, and association. SIHP # -31269 is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and it is recommended for no further work.	In accordance with federal regulations (36 CFR 800.5), the AIS results support the determination made by the Hawai'i State DOH on behalf of the EPA of "no adverse effect." In accordance with HAR §13-275-7, the Hawai'i State DOH and COH-DEM have determined the project effect is "effect, with proposed mitigation commitments." The project proponents request SHPD concurrence with these determinations
8.2.7 SIHP # 50-10-74-31270	
As noted in Section 8.1.7, SIHP # -31270 (historic jeep road) is in good condition with continued maintenance and use and retains integrity of location, design, setting, materials, workmanship, feeling, and association. SIHP # -31270 is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and it is recommended for no further work.	One historic property on the project area has been evaluated as eligible for inclusion on the National Register: SIHP $\#$ 50-10-47-30187 (Mämalahoa Highway). The 15-m (50-ft) portion of this historic property within the project area retains integrity of location only, as all the constructed elements the integrity are no longer evident today. While the project will affect the modern roadway elements, these elements will be replaced following construction, and the integrity and significance of the highway will not be diminished. Therefore, the project will
8.2.8 SIHP # 50-10-74-31271	have no adverse effect on SIHP # -30187. This site is assessed as significant under Criterion a of
As noted in Section 8.1.8, SIHP # -31271 (historic Nā'ālehu Spur Road) is in fair to good condition with continued maintenance and use. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. SIHP # -31271 is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and it is recommended for no further work. 8.2.9 SIHP # 50-10-74-31272	the State of Hawai'i significance criteria for its association with events that have made an important contribution to the development of transportation nucles on Hawaii Island, and under Criterion d for having yielded information about historic transportation in the project area. Documentation of SITP #-30187 in this AIS has included historical research, GPS data collection, photographs, and written description. This AIS has adequately documented the location, extent, function, age, and construction methods of the portion of SIHP # -30187 in the project area, mitigating project-related impacts pursuant to HAR §13-275-8.
As noted in Section 8.1.9, SIHP # -31272 (historic Ka'ala'iki Road) is in fair to good condition with continued maintenance and use. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. SIHP # -31272 is evaluated as not eligible for listing on the National Register. Its information content has been adequately documented, and it is recommended for no further work.	The remaining ten historic properties in the project area (SIHP #s 50-10-74-29507, -30929, - 30930, -31268, -31269, -31271, -31272, -31273, and -31274) may also be impacted by project development. These historic properties have been evaluated as not eligible for inclusion on the National Register. These sites are assessed as significant under Criterion d of the State of Hawai't significance criteria because they have yielded information important for research on
8.2.10 SIHP # 50-10-74-31273	history. Documentation of these historic properties in this AIS has included historical research, CDS data collection abstrametic also view mass and united dataging This AIS has
As noted in Section 8.1.10, SIHP # -31273 (historic water control complex) is in overall good condition with continued maintenance and use. The site retains integrity of location, design, setting, materials, workmanship, feeling, and association. SIHP # -31273 is evaluated as not	Or's data contection, protographs, plan view maps, and written descriptions. This Als has adequately documented the location, extent, function, age, and construction methods of these ten historic properties, mitigating project-related impacts pursuant to HAR §13-284-8. 0.7 Mitigation Bocommandations
cligible for listing on the National Register. Its information content has been adequately documented and it is recommended for no further work	
documentory, and it is recommended to include work. 8.2.11 SHPP # 50-10-74-31274	The single historic property evaluated as eligible for listing on the National Register (SIHP # 50-10-47-30187, Māmalahoa Highway), will not be adversely impacted by the project as its
As noted in Section 8.1.11, despite its poor condition, SIHP # -31274 (historic boundary wall) retains integrity of location, design, setting, materials, and workmanship. SIHP # -31274 is evaluated as not eligible for listing on the National Register. Its information content has been adecurately documented, and it is recommended for no further work.	integrity and significance will not be diminished. This AIS has adequately documented the location, extent, function, age, and construction methods of the 11 historic properties within the project area, mitigating any project-related impacts pursuant to HAR §13-284-8. No further work is recommended for these 11 historic properties.
	Archaeological monitoring is recommended during all project ground-disturbing activities to facilitate historic property identification for information purposes, especially of any subsurface lava tubes that may contain historic properties. Archaeological monitoring will proceed under an
AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Kävala, Ka'ū, Hawai'i Island TMKs: multiple	AISR for the Nāʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island 173 TMKs: multiple
Addition to a second	

8.2.7 SIHP # 50-10-74-312

8.2.8 SIHP # 50-10-74-312

8.2.9 SIHP # 50-10-74-312

8.2.10 SIHP # 50-10-74-31

8.2.11 SIHP # 50-10-74-31

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6	Project Effect and Mitigation Recommendations	Cultural Surveys Havai'i Job Code: KAHILIPALINUI 6 References Cited
archaeological monitoring plan that meets the require submitted to the SHPD for review and acceptance prior	ments of HAR §13-279-4, which will be t to project initiation.	Section 10 References Cited
		Belluomini, Scott A. and Hallett H. Hammatt 2017 Archaeological Inventory Survey Report for the Hilea Bridge Replacement Project, Hilea Ahupua'a, Ka'ā District, Hawai'i Island, Federal Highway Administration/Central Federal Lands Highway Division (FHWA/CFLHD) contract DTFH68-13-R-00027 TMKs: (3) 9-5-017:007 por., 008 por., and 9-5-017 Hawai'i Belt Road Māmalahoa Highway Right-of-Way. Cultural Surveys Hawai'i, Inc. Kailua, Hawai'i.
		Bishop Museum 2021 Aheahea. <i>Hawaiian Ethnobotany Online Database</i> . Available online at http://data.bishopmuseum.org/ethnobotanydb/ethnobotany.php?b=d&ID=aheahea
		Clark, Matthew R., Herbert Poepoe, and Robert B. Rechtman 2008 An Archaeological Inventory Survey of TMKs: 3-9-5-021:015 and 3-9-5-021;015 Kähilipali Nui and Käwala Ahupua'a, Ka'ü District, Island of Hawai 'i. Rechtman Consulting, Hilo, Hawai'i.
		Clark, Matthew R., J. David Nelson, Amy L. Ketner, Ashton Dircks Ah Sam, Lauryl K. Zenobi, and Robert B. Rechtman 2013 An Archaeological Inventory Survey of TMK: 3-9-5-010:001, Kāwala Ahupua'a, Kai't District Island of Howari'i Rechtman Consultine Hilo Hawai'i
		 Clark, Matthew R., J. David Nelson and Robert B. Rechtman 2014 An Archaeological Inventory Survey of Sections of the Mämalahoa Highway and Napo'opo'o Road Rights-of-Way (and Portions of Adjacent Parcels) for the Nastruction of the Proposed Mämalahoa Bypass Interchange Portions of TMKs: (3) 8-1-06.20, 704, 188, 8-1-08, 003, 004, 005, 021; and 8-1-09: 006, 007, 010, 011, 012, 013, 050, 057, 999, Ka awaloa Ahupua'a, South Kona District, Island of Hawai'i, Rechtman Consulting, LLC, Kea'an, Hawai'i.
		Coulter, John Wesley 1931 Population and Utilization of Land and Sea in Hawaii, 1853. Bishop Museum Bulletin 88. Bishop Museum Press, Honolulu.
		Donn, John M. 1906 Hawaii Territory Survey map of Hawaii Island (colorized). Registered Map 2060. Hawai'i Land Survey Division, Department of Accounting and General Services, Honolulu.
		Dorrance, William H. and Francis S. Morgan 2000 Sugar Islands: The 165-Year Story of Sugar in Hawai'i. Mutual Publishing, Honolulu.
		Ellis, William 1963 Journal of William Ellis: Narrative of a Tour of Hawai'i with Remarks on the History and Traditions. Advertising Publishing Company, Honolulu.
AISR for the Nāʿātehu Cesspool Closure Project, Kāhilipalinui and Kāwala. TMKs: multiple	. Ka'û, Hawai'î Island 174	AISR för the Nāʿālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Kaʿū, Hawai'i Island 175 TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 References Cited	Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6 References Cited
Elwell, Marge, and Dennis Elwell	King, James
2005 Historic Na'alehu: Hawaii's Deep South. The Paragon Agency, Orange,	1784 A Voyage of the Pacific Ocean. G. Nicol and T. Cadell, London.
Cautouna.	LaChance, Fred, Sarah Wilkinson, and Hallett H. Hammatt
2015 Images of America Kau District. Arcadia Publishing, Charleston, South Carolina.	2017 Archaeological Inventory Survey for the Mämalahoa Highway (Route 19)
ESRI	Widening Project, Waikoloa Ahupua'a, South Kohala District, Island of Hawai'i,
2015 World Imagery. Esri, Redlands, California. Available online at www.	<i>TMK: (3) 6-4.</i> Cultural Surveys Hawai'i, Inc. Kailua, Hawai'i
arcgisonline.com/maps/World_Imagery.	McCoy, Mark D., Peter R. Mills, Steven Lundblad, Tim Rieth, Jennifer G. Kahn, and
Giambelluca, T.W., Q. Chen, A.G. Frazier, J.P. Price, YL. Chen, PS. Chu, J.K.	Rowan Gard
Eischeid, and D.M. Delparte	2011 A Cost Surface Model of Volcanic Glass Quarrying and Exchange in Hawai'i.
2013 Online Rainfall Atlas of Hawai'i. Bulletin of the American Meteorological Society	Journal of Archaeological Science 36:234/-2500.
volume 94, pp. 313-316, doi: 10.1175/BAMS-D-11-00228.1. Electronic document,	Menzies, Archibald
http://rainfall.geography.hawaii.edu.	1920 Hawai'i Nei 128 Years Ago. W.F. Wilson, editor. New Freedom Press, Honolulu.
Hammatt, Hallett H. and David W. Shideler	Mills, Peter M., Steven P. Lundblad, Ken Hon, Jadelyn J. Moniz Nakamura, Elizabeth
2006 Archaeological Literature Review and Field Check Study of Two DOE Schools,	L. Kahahane, Arian Drake-Raue, Tanya M. Souza, and Richard Weil
Ka'ū District, Island of Hawai'i, Hawai'i Inter-Island DOE Cesspool Project,	2011 Reappraising Craft Specialisation and Exchange in Pre-Contact Hawai'i through
TMK: (3) 9-6-005:008, 039; 95-009:006, 015. Cultural Surveys Hawai'i, Inc.,	Non-destructive Sourcing of Basalt Adze Debitage. <i>Journal of Pacific</i>
Kailua, Hawai'i.	<i>Archaeology</i> , Volume 2, Number 2, 2001, pp. 79-92. New Zealand Archaeological
Handy, E.S. Craighill and Elizabeth G. Handy	Association, University of Otago, New Zealand.
1972 Native Planters in Old Hawaii: Their Life, Lore, and Environment. Bishop Museum	Monsarrat, M.S.
Bulletin 233. Bishop Museum Press, Honolulu.	1887 Map of Kau from Kiolakaa to Punaluu. Registered Map 1455. Hawai'i Land Survey Divition Stata of Haveni'i Damertment of Accounting and General Services
Hansen, Violet, and Marion Kelly	Division, plate of nawal 1, Department of Accounting and Octicial Schrees,
1972 Cultural and Historical Survey of Ka'alu'alu, District of Ka'u, Island of Hawaii.	Honolulu. Available online at http://dags.Hawai'i.gov/survey/search.php.
Bernice P. Bishop Museum Department of Anthropology, Honolulu.	Murakami, Gail M.
Haun, Alan E. and Dave Henry	2020 Analysis of Charcoal from Cultural Surveys Hawaii PUAA 5 Project, North Kona,
2005 Archaeological Inventory Survey, TMK: 9-5-5:017, 018 and 9-6-6:004, Lands of	Hawai i (TNK (3)7-5-014:001 por.). Wood Identification Laboratory, International
Kaunamano, Ka'u District, Island of Howai't. Haun & Associates, Kea'au,	Archaeological Research Institute, Inc., Honolulu.
Hawai'i.	Pukui, Mary Kawena, Samuel H. Elbert, and Esther T. Mookini
Hawaiian Sugar Planters' Association	1974 <i>Place Names of Hawaii.</i> Revised and expanded edition. University of Hawaii Press,
1990 Hawaiian Sugar Planters Association, Plantation Archives, Kau Sugar Company	Honolulu.
(ruucinnson sugar riantauon company) inaatenu, riawan 1670-1731. Accession: 84-07.	2011 An Archaeological Survey of TMK: 3-9-5-09:003 in Compliance with Section 106
Hawaiʻi TMK Service	of the National Historic Preservation Act, Kawaia Ampua a, Ka'u District, Island
2014 Tax Map Key: (3) 9-5-007, 008, 024, 025, and 026. Hawaiʻi TMK Service,	of Hawai'i. Rechtman Consulting, Hilo, Hawai'i.
Honolulu.	Sato, H., Warren Ikeda, Robert Paeth, Richard Smythe, and Minoru Takehiro, Jr.
Kelly, Marian	1973 Soil Survey of the Island of Hawaii, State of Hawaii. U.S. Department of
1980 Majestic Ka'u: Mo'olelo of Nine Ahupua'a. Bernice P. Bishop Museum	Agriculture, Washington, D.C.
Department of Anthropology, Honolulu.	USDA (U.S. Department of Agriculture)
Kahn, Jennifer G., Steven P. Lundblad, Peter R. Mills, Vvonne Chan, Ken Longdecker,	2001 Soil Survey Geographic (SSURGO) database. U.S. Department of Agriculture,
Vortuge Generoe	Natural Resources Conservation Service. Fort Worth. Texas.
2016 Settlement Chronologies and Shifting Resource Exploitation in Ka'ū District, Hawaiian Islands. <i>Asian Perspectives</i> , Volume 55, Number 2, 2016, pp. 184-207. University of Hawai'i Press, Honolulu.	http://www.ncgc.nrcs.usda.gov/products/datasets/ssurgo/.
AISR for the Nä alchu Cesspool Closure Project, Kähilipalinui and Käwala, Ka ü, Hawai'i Island 176	AISR för the Nä ⁺ älchu Cesspool Closure Project, Kähilipalinui and Kävala, Ka'ü, Hawai'i Island 177
TMKs: multiple	TMKs: multiple

Appendix A

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

References Cited

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

- USGS (U.S. Geological Survey) 1921 Naalehu USGS 7.5-minute series topographic quadrangle. USGS Information Services, Denver, Colorado.
- Naalehu USGS 7.5-minute series topographic quadrangle. USGS Information Services, Denver, Colorado. 1962
- USGS Orthophotoquad (Aerial photograph), Pahala Quadrangle. USGS Information Services, Denver, Colorado. 1978
- Naalehu, USGS 7.5-minute series topographic quadrangle. USGS Information Services, Denver, Colorado. 1995

Yucha, Trevor and Hallett H. Hammatt

Project, Ninole Ahupua'a, Ka'ii District, Hawai'i Island, Federal Highway Administration/Central Federal Lands Highway Division (FHWA/CFLHD) Contract DTFH68-13-R-00027 TMKs: (3) 9-5-019:011, 016, 024, 035 por., 9-5-027:020 por., and 9-5-019, 9-5-027 Hawai'i Belt Road Māmalahoa Highway Right-of-Way: Cultural Surveys Hawai'i, Inc. Kailua Hawai'i. 2017 Archaeological Inventory Survey Report for the Ninole Bridge Replacement

Waihona 'Aina

2020 The Mähele Database. Electronic document, http://waihona.com.

Project Area Land Jurisdiction Appendix A

Project TMK: (3)	Owner	Parcel	Acreage Portion in
		Acreage	Project Area
9-5-007:016	Ala Kahakai Trail	2,317.844*	30.805**
	Association		
9-5-008:033	County of Hawai'i	2.28	1.136
9-5-008:045	DWD & THD Family Trust	37.555	0.487
9-5-008:048	Kuahiwi Contractors Inc.	12.266	2.019
9-5-010:001	Kuahiwi Contractors Inc.	204.13	1.424
9-5-010:030	Kuahiwi Contractors Inc.	1.30	0.08
9-5-021	Road (State)	-	0.067
9-5-021:015	TOT Trust	15.97	1.37
9-5-021:020	State	-	0.014
9-5-022:001	Kau Valley LLC	26.46	3.601
9-5-024	Road (County)	-	3.882
9-5-024:001	Jerry Yukio Egami	0.30	0.301
9-5-024:009	Mildred Andrade	0.291	0.068
9-5-024:010	Tina Edmunds	0.33	0.333
9-5-024:011	County of Hawai'i,	0.24	0.238
9-5-024:069	Kuahiwi Contractors Inc.	0.128	0.124
9-5-024:076	Richard Oba	0.423	0.077
9-5-024:077	James Manoha	0.398	0.071
9-5-025	Road (County)	-	0.718
9-5-025:002	County of Hawai'i,	3.09	0.24
9-5-025:039	La'i LLC	0.88	0.329
9-5-026	Road (County)	-	2.248

*Parcel acreage as listed in the County of Hawai'i Real Property Tax records, and as shown in the executed Grant of Conservation Easement between Ala Kahakai Trails Association and the County dated 16 December 2019

**For the purpose of this archaeological study, this acreage is slightly larger than the approximately 29 acress excluded from the conservation easement for the Nā^{*}ālehu treatment and disposal facility due to project boundary smoothing in GIS

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

178

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple





Appendix C

Appendix C SHPD Correspondence



AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix C

Due to the potential for historic properties located within the current project area, SHPD request that an AIS be conducted in the project area. Additionally, SHPD indicated that current project area, SHPD request strategy appears actionation in distription and addition and the current project area and determining the potential impacts to those astes about the AIS detaily any. The current antiminal additesses in the SHPD in project area and determining the potential impacts to those astes about the AIS detaily any. The current antiminal additesses in the SHPD in project area and detarmining the potential impacts to those astes about the AIS detaily any. The current antiminal additesses in the SHPD in project area and detarmining CD and GD and GD area to be addressed provide current astimination letter. The next step in the roview process is conducting an AIS as per theorem and the AIS addition and the AIS addition and the roview process is conducting an AIS as per theorem. The County of Hawai'I DEM requests the SHPD concurrence with the determination "of the need for an AIS for the project." Additionally, the County is requesting concurrence with the proposed testing strategy and an accheological permit for the work **SHD** yieldnates the AIS must be conducted by a firm permitted to conduct archeological work the Shared's website for a list of permitted times. A literature review titled *Archicalogical Literature Review (apper) for the NT 2014 Literator Capacity Cosposis* Replacement Project, Khilipatini and Khwala Aluqua X, Kai Dharris, Hawai 'Likand TMK's multiple (Purdy et al. August 2020) was completed for the project's Environmental Assessment (EA) portion of the project. The literature review indicates undocumental harburs proprinter may be present (EA) portion of the project. The segment of the Manifundian Highway (21):PF 3-10.14/-3-2015) and a protion of a histore concrete (holdbarned) The current subminuit inducts an AIS strategy plan for the proposed project. The AIS will follow the standards accounting to HZAR 912-576. The puyers will reacture 100% as unless areas on this as documentation, mapping and proceedings. The thermal exact and the proposed for 1 4 sophoratory traches that will span approximately 50 meters by 0.60 to 1.0 meters much exact which to behaviors of the physical limits of the backhore. Additional manual exactingness will see the additional exact and the backets of the physical limits of the backhore. Additional manual exactingness will see the additional for the backets of the physical limits of the backhore. Additional manual exactingness will see the additional holicities properties during the AIS. SHPDs shall notify the County when the AIS, and any additional documentation if needed (e.g., archaeological monitoring plan) has been accepted and the permit issuance process may proceed. Please contact Nicole A. Mello, Hawa'i Island Historic Preservation IV, at <u>Nicole Mello@hawaii.gov</u>, for matters regarding archaeological resources or this letter. SHPD looks forward to receiving an AIS that meets the requirements of HAR §13-276 for review. and (6) Verification of completion of detailed mitigation plan. Information regarding any federal component, such as US EPA funding or approval. Starth Witchen, CSH, synthynaerigeuthaureness com William Fok, CSH, WiTch@geuhumhanrenss.com Joy Hoyfenspegt, CSH, JhnAfensperged.collutinaturess.com Caragi Josten, Isrown and Carlowell, Gikstreaufd.com John Scheguchi, Jaskaguchi/Grilson-Barndo.com Alan S. Downer, PhD Administrator, State Historic Preservation Division Deputy State Historic Preservation Officer William Kucharski December 1, 2020 Page 2 Alan Downer eri Aloha, ::))

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

182

Appendix D

Pb (0.35) ppm BHOL2 stata 2:11-31 MOL2 stata 2:11-31 MOL2 2:121 % MOL2 2:211 % ADD3 7:759 % ADD3 7:759 % ADD3 7:759 % ADD3 7:753 %ADD3 7:753 % ADD3 7:753 % ADD3

0

Appendix D

Appendix D EDXRF Analysis Report

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

eet. 03/1621_13/07.17 allinetect. 03/1621_13/07.02 re version: 7.2 (Build 134)	uto acuum redum uto secum fedum	uto acuum teulum	uto acuum ledium	
seoarchaeology Lab Anayz so2:1-Ka'u mise.mth softwa	Current Current Counte Limit Counte Limit Count Rade Current Counts Limit Counts Limit Count Rade	Current Cours Limit Couns Limit Afmosphere Count Rate M	Current Countent A Counts Limit A Aumosphere Count Rate	Ba-kground (spainA) 11.19 11.10 11.1
UH Hilpelinui 3 uue: Linear Vocuments Wethods/221 Method	50 kV 250 seconds C 20 mick a text a seconds 9 seconds 20 seconds	28 KV 280 seconds 250 seconds 40 keV 0 seconds	6 kV 200 seconds No Filter 10 keV 0 seconds	Denomination Peak (cps6mA) Denomination Peak (cps6mA) De
Sample List: CS Analvisis Techni Method File: CA. Conditions	High Zh Voftage Livetime Filter Maximum Energy Warmup time Mid Za Voftage Livetime Filvetime Filvetime Maximum Energy	warnup unto Voltage Livetime Filter Maximum Energy Warmup time	Low Za Voitage Livetime Filter Maximum Energy Warmup time	Results Kallipalmurt Nazo Nazo Nazo Nazo Nazo Nazo Nazo Nazo

AISR for the Nāʿālehu Cesspool Closure Project, Kāfnlipalīnui and Kāwalā, Kaʿū, Hawaiʻi Island TMKs: multiple



AISR for the Nã alchu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Appendix E

Radiocarbon Analysis Renord Appendix E

BETR Beta Analytic	Beta Analytic, Inc. 4985 SW17/4° Court Miant. 1: 23315 USA Tet: 305-667-5167 Fax: 305-667-994 into@betalabservices.com
ISO/IEC 17025:2017-Accredited Testing Labor	atory
May 27, 2021	
Mr. Oriver Bautisa Octumed Surveys Hewaii 3980 Huana 1St. Sula 124 Hub. Hewaii 9770	
RE: Radiocarbon Dating Results Near M. Paniteta	
Deal Mil. Dadusia,	
Enclosed are the radiocarbon deling results for two samples recently sent to us. As usua not report vine results and calination data is provided where applicable. The Convert been concreted for trait radionation effects and where applicable, calination was performed (cled on the graph pages).	I, the method of analysis is listed onal Radiocarbon Ages have all using 2020 calibration databases
The web directory containing the table of results and PDF download also contains pictur option and a quality assurance report containing expected vs. measured values for 3-5 workin simultaneously with your samples.	s. a cvs spreadsheet download ig standards analyzed
Reported results are accredited to ISO/IEC 17025.2011 Testing Accreditation PJLA #59 enformed there in our laboratory and counted in our own accelerators here. Since Bela is not graduate trained to strict protocols of the ISO/IEC 17025.2017 Testing Accreditation PJLA # Trailyses.	23 standards and all chemistry was a teaching laboratory, only 59423 program participated in the
As always Conventional Radiocarbon Ages and sigmas are rounded to the nearest 10 yr International Radiocarbon Conference. When counting statistics produce sigmas lower than - pp is clued for the result unless otherwise requested. The reported of 30 values were measu arbo mass spectrometery. They are NOT the AMS of 30 values word include fractionation et MIS induced sources.	ars per the conventions of the 1977 - 30 years, a conservative +- 30 red separately in an IRNIS (solope tects from natural, chemistry and
When interpreting the results, please consider any communications you may have had w	ith us regarding the samples.
The cost of analysis was previously involced. As always, if you have any questions or w don't hesitate to contact us.	build like to discuss the results,
Alio, Dipu apana al	Patrick
Chris Patrick Vice President of L	boratory Operations
Page 1 of 5	

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

	BETA	Beta Analytic		4905 SW 74" Court Miami, FL 33155 USA Tal: 305,447,5147
SOIED TADGE TRADEORATION ACCORDING Laboratory CONTRACTORDING TANDACTORDING LANCES CONTRACTORDING TANDACTORDING TANDACTORD ONE BAURSI ONE BAURSI CONTRACTORDING TANDACTORDING TANDACTORD ONE BAURSI CONTRACTORDING TANDACTORDING TANDACTORD CONTRACTORDING TANDACTORD CONTRACTORDING TANDACTORD CONTRACTORD		ESTING LABORATORY		Fax: 305-663-0964 info@betalabservices.com
Construction Report Data May 1, 202, 201 China Shunga Hawai Report Data May 1, 202, 201 China Shunga Hawai Report Data May 1, 202, 201 China Shunga Hawai Report Data May 1, 202, 201 Laboratory Numbei Sangle Code Numbei Report Data May 20, 202 Laboratory Numbei Sangle Code Numbei Conventional Redocation Age (BP) of Sandary 20, 202 Laboratory Numbei Sangle Code Numbei Conventional Redocation Age (BP) of Sandary 20, 202 Laboratory Numbei Sandary Sanda		ISO/IEC 17025:2017-Accredited Te	sting Laboratory	
Okine Bautista Report Date: May 27, 2021 Johuna Survoys Howai Material Received: May 27, 2021 Johuna Survoys Howai Johuna Survoys Howai Material Received: Material Received: <td>ш</td> <td>REPORT OF RADIOCARBON D</td> <td>DATING ANALYSI</td> <td>8</td>	ш	REPORT OF RADIOCARBON D	DATING ANALYSI	8
Cubrand Structorys Housel Mantel Received. May 20, 2021 Lationation Number Sample Code Number Conventional Radiocantion Age (BP) or Pervent Muchanic Cautori (MOI) S Salable Isolutions Jeba - 555597 Jeba - 502597 Jeb - 4005 RMS 61702. 818 61702. <td>Olivier Bautista</td> <td></td> <td>Report Date:</td> <td>May 27, 2021</td>	Olivier Bautista		Report Date:	May 27, 2021
Laboratory Number Sample Code Number Conventional Radiocation Age (EP) of Percent Modernic caturus (MOV 6: Saulte Source) Jaho - 50057 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 50059 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 50059 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 50059 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 5005 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 5005 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 5005 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 5005 Jaho - 500 P IRNS 67130: -86.8 0/00 Jaho - 5005 Jaho - 500 P IA00 - 500 P Jaho - 5005 Jaho - 5005 Jaho - 5005 Jaho - 5005 Jaho - 5005 Jaho - 5007 Jaho - 5005 Jaho - 5007 Jaho - 5007 Jaho - 5005 Jaho - 5007 Jaho - 5007 Jaho - 5005 Jaho - 5007 Jaho - 5007 Jaho - 5005 Jaho - 5007 Jaho - 5007 Jaho - 5005 Jaho - 5007 Jaho - 5007 Jaho - 5005 Jaho	Cultural Surveys Hawaii		Material Received:	May 20, 2021
Bits - 58367 Data 180 + 30 - 90 - 180 - 180 - 100	Laboratory Number	Sample Code Number	Conventional Ra Percent Modern Carb	tdiocarbon Age (BP) or win (pMC) & Stable Isutupes
(49.6)() 1722-1914 cal AD (221-130 cal BP) (12.2)() 1555-1596 cal AD (221-230 cal BP) (12.3)() (120)() (120)() (121-230 cal BP) (12.3)() (120)() (120)() (121-230 cal BP) (12.3)() (120)() (121-10) cal BP) (120)() (120)() (120)() (121-10) cal BP) (11-10) cal BP) Similar Material: Chancell (110)() (11-10) cal BP) (11-10) cal BP) Analyzed Material: Chancell (110)() (110)() (110)() (110)() Analyzed Material: Chancell (110)() (110)() (110)() (110)() (110)() Analyzed Material: Chancell (110)()<	Beta - 592957	2a	180 +/- 30 BP	IRMS 513C: -8.8 a/oa
Submitter Material: Charcoal Pereteainment: Charcoard metalia) addiata/add Antykor Material: Charcoard metalia) addiata/add Antykors: Savora MS-Symated delivery Pereteainment: Charlon: By778 +0.037 Markor Savora Anto: 2014b, 25216 + 356 500 Anto: 3014b; 30250 + 0.037 Markor Savora Anto: 3014b; 30250 + 0.037 Markor Savora Anto: 3014b; 30250 + 0.037 Markor Savora Anto: 3014b; 30250 + 0.037 Markor Markor Markor Anto: 3014b; 305 500 Anto: 3014b; 305 500 Anto: 3014b; 305 500 Anto: 3014b; 305 3021) Markor Markor Markor Markor Anto: 3014b; 305 500 Anto: 3014b; 305 2021) Markor Markor Anto: 3014b; 305 500 Anto: 3014b; 3014		(43.9%) 1722 - 1814 cal AD (19.2%) 1656 - 1698 cal AD (19.0%) 1910 - Pert AD 1950 (1.3%) 1936 - 1890 cal AD	(228 - 136 cal BP) (294 - 252 cal BP) (10 - Peet BP 0) (114 - 70 cal BP)	
Reach: an iDORC/TOD20071econded, los sub-condending or should labor was used in the analyse. Al work was cone al Beat in 4 hohuse NRC accelerator mass processments after the second second second second and the analyses. Al work was cone al Beat in 4 hohuse NRC accelerator mass used for takena: offence, marks and acceleration of the mean of the second	2	Submitter Material: Charceal Submitter Material: Charceal Analyzed Material: Charted material acts Analyzed Material: Charted material Analyzed Material: Charted material Analyzed Material: Charted Galo D14(2, -22:16.4:, 365.000 A14(2, -32:26:4:, 365.000 A14(2, -32:26:4:, 365.000 A14(2, -32:26:4:, 365.000 A14(2, -32:26:4:, 365.000 A14(2, -32:26:4:, 365.000 A14(2, -32:26:4:, 365.000) Calibration: BataCal4.20:HD me	//alkal/acid y sso_2021) on: 101.06 -/ 0.37 pMC thoud INTCAL20	
returbs an (SOIC)-(TOSCOI)) accorded to sub-controlling of should lacer test and the sub-part AI work war form all that in 4 horbitam MCL accordence mast percentants and 4 Therman (SOIL). The Constraint of the second to the sub-part of the SOI and the SOIL and the SOIL accordence and sup- percentants and a thread of the SOIL accordence age, was calculated using the SOIP wards. SO 1050, SOIL and the solution when applicable. The Age is rounded to the mean (1) you are all to solution and super- tending applicable. The Age is rounded to the mean (1) you are all to solutions and super- ted and the solution when applicable. The Age is rounded and are solved when the converted and a wards the solution activity and the solution (2) house according and the Advector and a solution (2) house a solution (2) here according to the converted and according to the SOIL (2) constrained and the round means are pointed applies less than 30 BP on the Converted Biolecation Age are constrained, provide up to 30, according to the point in according solution. Calculated applies less than 30 BP on the Converted Biolecation Age are calculated up to 30, acto wards are available (red) to 400, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 410, 400, 410, 410, 410, 410, 410, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 410, 400, 410, 410, 410, 410, 410, 410, 410, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 4100, 410, 41				
	Reaults are ISOAEC-1702-2017 accr spectrometer and 4 Thermon IPMAs. Lased for televisar californi where Reaults grade han the modern role (cosific dad). Outoed errors are 15 4150 values are on the meterial it califordion grade pages.	etted. An sub-contraling or statient laker was used in this a The "Constrained Rediscretor Age" was calculated using the problem, The Statistical Rediscretoria Age" was calculated using the neuron service and an entropy of the statistical and an and beneric as reported as provided signing uses then 30 the or- physic country statistics. Collardied Signing uses the 30 the or- angle country statistics. Collardied Signing uses the 30 the or- angle country statistics. Collardied Signing uses the 30 the or- physic country statistics. Collardied Signing uses the 30 the or- ticle.	Interprets. All work wars donte al Bet Libby marking (SoB) years; bis control it trippind as real-control years 95%, the Conventional Radiocation Age the Conventional Radiocation Age to VCOC+. References for cokendar	a in 4 inheure NEC accentrator mass click for load lange fraction and was re preset (197), present = A0 1950, the 1.42 optimum of NET 35A4900C and the standard up to 30, calibrations are start at the batton of
Page 2 of 5		Page 2 of 5		

186

AISR for the Na^{*}älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka^{*}ū, Hawai'i Island TMKs: multiple

Appendix E

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Beta Analytic, Inc. 4905 SW 74th Court 1410 SW 23155 USA Tel: 305-667-5167 Fax: 305-663-9956 fax: 306-663-30956 info@betalabservices.com Realls are ISORE-1782/2017 accretion. No subcentracting or student laker was used in the amelyers. All work was done at Boats in 4h-house NEC accretionation are accurate to the contract and the contrest and the contract and the contrest and the contract and the Conventional Radiocarbon Age (BP) or Pervent Modern Carbon (pMC) & Stable Isotopes IRMS 513C: -25.2 o/oo Report Date: May 27, 2021 May 20, 2021 REPORT OF RADIOCARBON DATING ANALYSES Material Received: (284 - 167 cal BP) (154 - 56 cal BP) (17 - Poet BP 0) 150 +/- 30 BP ISO/IEC 17025:2017-Accredited Testing Laboratory A14C. -28.89 +J. 3 67 oloc (1950.2021) Measured Radiocarbon Age. (without d13C correction): 150 +J- 30 BP Calibration: Belacal4.20. HPD method. INTCAL20 Submitter Material: Charcoal Prefreatment: (charred material) acid/alkali/acid Analyzed Material: Charred material Analyses Service: AMS-Standard delivery Percent Modern Cartoon: 98.15 +/. 0.37 pMC Fraction Modern Carbon: 0.9815 +/- 0.0037 D14C: -18.50 +/- 3.67 o/oo 1666 - 1783 cal AD 1796 - 1894 cal AD 1903 - Poet AD 1950 Page 3 of 5 2b Sample Code Number Beta Analytic (42.8%) (33.8%) (18.8%) Cultural Surveys Hawaii BETR Laboratory Number Beta - 592958 Olivier Bautista

AISR för the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ú, Hawai'i Island TMKs: multiple

188



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

Appendix E

Appendix E



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

190

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F

Site Descriptions from Previous Archaeological Studies **Appendix F**

.... ì . SIHP

10-74-29307 (Clark et al. 2013:184-207)	IIP Site 2997 the 2507 is a complex of old much buildings located in the northwestern corner of the project area near e particulo of the Site 2505 and Site 25906 wild Site Fegure 40. The complex diadous three main addram (Festure 7), and second should be the form the Närlahu Skara Mill on a store addram frame and the Site 2505 wild site of machinery from the Närlahu Skara Mill on a store addram frame any investigation of the site 2506 wild site of the source of the project area near addram (Festure 7), and second search thran site 10. The complex distribution site of ough, a site of the second search starts is the site of the site of the site of the site of mobile thram engineeut, the site of a start and start of the site of the site of the site of mobile site is and and at the gate is a rescent thrap ify a worden come of the sardy pared, and pared the dust or starts and site of the current by site thrap set of site site is and panet (the libuiding (Festure 16) site interval) with the stardy pared, and pared is of the curve a started partially with the stardy pared, and partially can the distork conduct on the neighboring pared in 2007 is presented below. Thu only the site's feature that distork conduct on the neighboring pared in 2007 is presented below. Thu only the site's feature that can will be accurated and the curve as signed feature disgnations and tectorles that distork conducted on the neighboring pared in 2007 is presented below. The features faster by rechanal Constitue, LLC in 2007 as also or storest estab- tern will be curved as the condition. LLC in 2007 as also or storest estab- tern will be complexed as the condition. LLC in 2007 as also or storest estab- tern will be complexed as the condition. LLC in 2007 as also or storest estab- tern will be complexed by flat or dissonal and wood structure on a concret estab- tion of the curved will be complex (Feature 5) stored to the curve the storest and the concret stability LLC in 2007 as also by concret of bot.	autiversement of Site 25005, a flager worden unlight at the northem reletint of the complex next to the rance) parking area (Fearmer C), and an off price or manimumy from the rorginal NE34ban sugar mult hat is been set up on a cobble pattern (Fearmer C) at the intersection of Sites 25005 and 100 the allifting are trained as a cobble pattern (Fearmer C) at the intersection of Sites 25005 and 100 the allifting are in warrows tasked of disreptiar, and here you not appear to have been used (except K6 stronger parese). For a muration state of the current L) at the scale of the state in the interval protect for an under of years. Earnes A, B, C, and D of Site 2505, which are all labely at the strainly reader whith the boundaries Sites 25015 and 25906, and the other features of Site 25070 that are tured on the neighboring pared to the west are depicted in Figure 118.
r # 50-10-/4-	SITP Site 29 Site 2050 The 205	southwestern growel practication has been set to buildings are purpares for the proper of the relative to the situated on the

AISR for the Na alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island TMKs: multiple





AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

192

Appendix F

Appendix F



AISR for the Nä älehu Cesspool Closure Project, Kähilipalimui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

194

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



Appendix F





TMKs: multiple 196 AISR for the Na'alehu Cesspool Closure Project, Kahilipalinui and Kawala, Ka'ū, Hawai'i Island

TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island

Appendix F



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

198

AISR for the Nä'älehu Cesspool Closure Project, Kähilipalimui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple



Appendix F

Appendix F



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

200



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Appendix F
Appendix F



AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

202

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple



Appendix F

Appendix F



204 AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Feature C is an old piece of sugar mill equipment that has been placed on a cobole foundation at the punction of the Site 25/30 and 25/96 with a couplity where meters onthered or feature B (see Figure 118). The item is a submeter with couplity where the meters onthered of Feature B (see Figure 118). The item is a submeter with couplity where the counters and large development of the state and in states 92 estimates all. On the extrince panels of Feature C are the inscriptions "W, & A. M/OME (LLASCOW 1871," in the center of each of the long sides (Figure 138) and "500 GALLs" in the center of the static 92 estimates all. On the extrince panels of Feature C are the inscriptions "W, & A. M/OME (LLASCOW 1871," in the center of each of the long sides (Figure 138) and "500 GALLs" in the scatter of the static field state of the three of the long sides (Figure 138) and "500 GALLs" in the scatter of the static field state of the long sides (Figure 138) and "500 GALLs" in the scatter of the static field state of the long sides (Figure 138) and "500 GALLs" in the scatter of the static field state (LLASCOW 1871," in the scatter of the long sides (Figure 138) and "500 GALLs" in the scatter of the static state are bolic in 1871. The stater mode to the work ong state state scatter developed with statily beet to grave the state related in the scatter of the work ong state state in the scatter of the scatter state state of the state scatter of the state mode of the work ong state state scatter developed with state scatter 2056 walls where their meet. This colded in maternase 4.1 meters long by 2.6 meters wide and 2056 walls where their near state state in the scatter of spate to state state state at state of the state work of the state state of the scatter state state state state state state 2056 walls where their near state sta Feature D is a large wooden structure with a metal root that is losted 15 meters northwest of Feature B at the northwestern stear of the Site S2050 complox (see Fague 110). In DOT when this structure was first recorded by Rehuma Consulting, LLIC it had four adjoining rooted sections that formed a building with the overall dimensions of 53 meters logo by 23 meters used, feguer 111). The sections at Feature D indicated (1) animit there norm structures and by 23 meters used, feguer 111). The section at Feature D to four dimensions of 53 meters logo by 23 meters ado (2011). The sections that formed as the structure structures and 13 and 14 meters and a contract stable (2) a small gable root covered area attrached to the extern each of the num structure. All 3 collarsed to be more or a three grand meter that A of the cortex withing. however, the small gable root covered area to the west of that A of the cortex withing. however, the small gable root covered area to the section that A of the contract root and a contract stable (2) a small gable root covered area to the west of that A of the contract with a collapsed to be more use at the vesture at the structure of that A of the contract with a structure and the order of all have also determines have the structure attriation of the nortex structure of the overed structure (Figure 142). The condition of the two interving in an error of the structure and the covered structure (Figure 142). The condition of the two purposes of the current structure and the covered structure (Figure 142). The structure attracture attracture field attracture attracture and the structure and the structure of the two purposes of the current structure and the covered structure (Figure 142). The structure attracture the the parameter structure and the interving the the structure and the structure attracture (Figure 142). The structure attracture the structure attracture at the vesture attracture (Figure 142) attracture the structure attracture attracture attracture (Figure 142), attracture (Fi The central room, which is accessed through the other set of double doors in the north wall of Feature B, meases 7.2 mices long (norther selections) by 5.5 metrics wide. This room is mary and has no interior medications except for a shelf. A burk covered by 8 shell not excertise out it which appears to be a research for the second through its framer who experime within to prove. This manu, which appears to be a research and the second through its framework who are shown and the second the second through the second through a second through a three second through a second through This central room appears to have been two rooms at one point in time. Running morth/south acress the room is the outline of a former ranked concrete all that has heave allow the have all have divided the room into two equal 3.5 x 3.5 mears eachors. Each section of concrete float, one there said of the removed sail, is sloped to a damin hole through the base of the concrete sail wall on the sould such of the structure. In the northern wall of the room at its western end or a second set of double doors is present, the doors have been removed and walled off. A levatory that measures 3.5 meters long (oorth/south) by 1.2 meters wide (east/west) is constructed of the vest end of Takung E. This constructions are port of the equival instruction (as vicineed) by the inconstant concrete floor), but has been built so that the coeffine and with that match has of the inconstant concrete floor), but has been built so that the coeffine and with that match has of original structure. The startory contents a total, cash, and concrete shower stall 1 and 1 and drain 2bth, doors in the vestern wall of feature E lacit of the largory. The formal attributes of Feature E stages of that a trive storphard is entance of the largory. The formal attributes of peature B stages that doring the statum of the poyetry). RC-0719 198 Feature C Feature D

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

Appendix F

Appendix F

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F





AISR for the Na alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka 'ū, Hawai'i Island TMKs: multiple

206

AISR for the Nä'ālehu Cesspool Closure Project, Kāhilipalimui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Appendix F



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

208

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple



Appendix F





AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

ē

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

210

Appendix F

Appendix F



Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F



AISR for the Nä'älehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple

212

Appendix F

The first mention of Shte 25907 in the Hutchinson Sligar Plantation Company Annual Reports occurs in 1918 when it was reported by the plantation manager that 108 statis was an adjusted that the company's own sligalphenologic (feature D). Hence (1929) also mentions that the ranch had its own sligalphenologics, at which roughly 100 had of caller even sligalphet annually. The location to the slaughbenches is shown on a Adach 1031 mesh of the NGT adjust Branch and Duriy. The location to the slaughbenches is shown on Shach 1031 mesh of the NGT adjust Branch and Duriy. The location the slaughbenches is shown on Shach 1031 mesh of the NGT adjust Branch and Duriy. The location the slaughbenches is shown on Shach 1031 mesh of the NGT adjust Branch and Duriy. The location the slaughbenches is shown on Shach 1031 mesh of the NGT adjust Branch and Duriy. The location the slaughbenches is shown on Shach 1031 mesh of the NGT adjust Branch and Duriy. The location the slaughbenches is shown on Shach 1031 mesh the annual the annual the Remark Share 2506 mesh walls. Feature A is specifically metricoust that the moder to Hay 111 + S. C. 13404, Wort McDurat A is specifically metricoust plate methods are adjuster to the fragmer Share confirms the additions corrected in the graves the Wester and the Hay 11400, who were the manifecturer's of signar mills and engines that the moder of Datager Share confirms of South 2007). The year writing on feature C, however, minister shat the moder of 11870, Wort were the manifecturer's of signar mills and engines that was distributed and the Nate Scate Scate Share multis and engines that the moder of 118400, Wort were the manifecturer's of signar mills and engines that the moder State State Scate State mills and engines that the Nate State State State Company of correstor to the enginest that the word (Care scated, and the mathiney was likely mored to the tot correstor to the product of the scate State State State State multis and engines that was distributed and the state State State State State The Site 29507 and/building complex appears to have been built during the early wentieth century, by the maching assisting of the intuination Signar Flankanio (Kash and Man) and Kanohy Shah Rawh and Dary, fisc.) The formal attributes of the building suggest their functions, and a review of the Hukinsson Signar Plankator Company Annual Repeats for the years 1930 to 1950 (waitable at the University of Hawari at Histor Company Annual Repeats for the years 1930 to 1950 (waitable at the University of Hawari 2014) Aloskini Liberary provides additionation information relaring to the construction and see of the buildings. Feature A. Intercioned as a pageory with an atabed shaughtense for pigest the was interesting to a mached shaughtense for grant events the shaughtenhouse for earlied. In wascincered to a much heddangement and equipment transverse dates and down in the mid-1950k. Feature C. which is not a building, seems to be a piece of equipment removed from the organital state was allow state and search of an interim of the Ray S950 mech walks. The specific use of Feature C at Site 29507 is not dear, it may have been purely a decorative addition to the ranching complex. The Nif sidehu slaughterhouse closed in the mid-1950s, and the ranch's beef was then marketed through the Hilo Mate Co. (The Chinnee Haven, Sugar Mannen 1954-46). Alther desared to He sidualytherhouse of the Feature D building was converted to serve as an office building (mach headquarters), workshop and equipment storage rear. It is lakely around this unde math the overeal control statisticant equipment storage rear. It is lakely around this turn that the overeal control statistic math equipment storage rear. It is lakely around this turn that the overeal control statistic math equipment storage rear. It is lakely around this turn that the overeal control statistic math equipment storage rear. It is alkely around this turn that the overeal control statistic project area lands during the second life of the varient entrol. This is alkely around the form of the transform that is the stable (Feature E) was converted to a bunkhouse at this time as well (probably by Onlys a kendy, which is inscribed in the concrete at Feature B). When Feature A cested to be used as a project of antherwain. RC-0719

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F

SIHP # 50-10-47-30187 (Clark et al. 2014:52-58)

SIHP Site 30187

Site 30187 consists of the features Minnalahoa Highway (its past and current alignments) located within the current project area (see Figure 10). As atlacd in the Chinne-Historical context section with the root, this root/at his developed from what was core a floophath (Kaahahu, Mby and Maly 2001), into its current Highway 11 form through interemental improvements over more all compared. The improvements come in the factor fut in the rootway has project areas (see Figure 10). As a staticat the Chinne-Historical content science in the rootway from a factor project area (see Figure 10). The improvements come in the factor of static project maps grade doubles, and transmoster over more than area upgraded routinely, but specially during large calls construction projects. The first large scale accounce that are upgraded routinely, but specially during large conduction project area (see Figure 20). As a post to the area (see a throw of the counce of which was conducted in the for 1000, so early 1006. By 1933 the Manus Kaya and a scale scale area from the atlant area (project and the area of the active scale area for a scale area are conducted in the for 1000, so early 1006. By 1933 the Manus Highway (He Hawi Ti-Bit Read) was constructed engines to can define the offer (Soure area for a dravess) are an or afferd. Here, area doubd engines to varia at the looping that more efficiently, and thus were able to results a more direct, less envy route. Within the current study area, along the costing Minhahoa Highway alignment, one store reventent (Feature E) were recorded, and are described in detail hole.

Feature A

Feature A is a sloped rectament consisting of large cobbles and small boulders located along the *makini* side of an elevated protein of the Higherd away from the read staticts at a roughly 40 degree angle. The lowest point of the adjacent ground surfaces is at the northwestern end writes at a roughly 40 degree angle. The lowest point of the adjacent ground surfaces is at the northwestern end writes at a roughly 40 degree angle. The lowest point of the adjacent ground surfaces is at the northwestern end where the base of the reventment measures 5.6 meters below the road surface and typers up the southestern end where the base of the reventment measures 5.6 meters below the road surface and typers up the southestern end where the base of the reventment measures 5.6 meters below the road surface and boulders, underlying dark frowm soil was observed. At its southest end, Feature A, Reminda et Mi 20 at 2021 as a based by the rown soil was observed. At its southest end, Feature A, Reminda et Mi 20 at 222 and boulders, underlying dark frowm soil was observed. At its southest end, Feature A, Reminda et Mi 20 at 222 at the southest the rown soil was observed. At its southest end, Feature A, Reminda et Mi 20 at 222 at the southest the rown soil was observed. At its southest end, Feature A, Reminda et Mi 20 at 222 at 222 at boulders, underlying dark frowm soil was observed. At its southest end, Figure 23, Nust southest end is stated with the rown southest between TMK (5) 8-149011 at at 0571 that terminates were mediated in the base of Feature A (Figure 5). This loped current of the base of the rown at 10 meters a

AIS for the Māmalahoa Bypass Interchange, Ka'awaloa Ahupua'a, South Kona, Hawai'i

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Appendix F



Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island

TMKs: multiple

216

AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Appendix F



AISR for the Nā'ālehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F



AISR for the Na'alehu Cesspool Closure Project, Kähilipalinui and Kāwala, Ka'ū, Hawai'i Island TMKs: multiple

Appendix F



AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F



AISR for the Nä"alehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"u, Hawai'i Island TMKs: multiple

Appendix F



Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F



AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka"ü, Hawai'i Island TMKs: multiple

Appendix F

SIHP # 50-10-74-30929 (Gastilo and Clark 2018:145-149; Draft)

Site 50-10-74-30929

Site 3/929 consists of an exclosure, an L-shared pavement, and a padhway (Figure 92) that are located approximately and the structure of 20 and 20 and the structure structure structure of the 30 and 20 and

The endoance is located to the immediate southwest of the L-shaped pavement and is constructed of small to stage photohole cololes (figure 93). The northow asserts, and southern degar of the endoance in an sami-carde what is avastent degar is formed by a pit of large cololes. It measures roughly three meters long by three meters well, and mage stewent 0.1 to 65 means at IT. The micror of the endoance and and mayer of soil on top of bodies, those only as and *koan hande*. Noted along the northwestern corner of the endoance are two flat laid slabs that appear to be covering a sub-surface void that is approximately 0.3 meters deep (P-gure 94).

AIS of TMK: (3) 9-5-007:016, Kahilipali Nui, Kahilipali Iki, and Kahaea, Ka'u, Hawai'i

145

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6



AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nä"älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

224





Hawai' I Island 226 AISR for the Na⁺2 TMKs: multiple

TMKs: multiple

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F



AISR for the Na alehu Cesspool Closure Project, Kāhilipalinui and Kāwala, Ka 'ū, Hawai'i Island

Appendix F



Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix F

SIHP # 50-10-74-30930 (Gastilo and Clark 2018:149-150; Draft)

Site 50-10-74-30930

Site 30930 is a score filled wall that is located near a buildozed partner in the northern section of the current study area (see Figure 49). The wall is constructed of Targe stackord, *photodne* coptes and *cheadare* 2020 meters to the estivates (see Figure 99). That measures 1.1 measures



AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka'ū, Hawai'i Island TMKs: multiple

AISR for the Nä älehu Cesspool Closure Project, Kähilipalinui and Käwala, Ka ü, Hawai'i Island TMKs: multiple

229







Cultural Surveys Hawai'i Job Code: KAHILJPALJNUJ 6

Appendix G







232

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix G



AISR for the Nä äldetu Cesspool Closure Project, Kähtlipalinui and Käwala, Ka'ü, Hawai'i Island TMKs: multiple







234

Cultural Surveys Hawai'i Job Code: KAHILIPALINUI 6

Appendix G



AISR for the Nii äldelu Cesspool Cleasure Project, Kähtlippdinui and Kätwala, Ka'ü, Hawai'i Island TMKs: multiple







Appendix G





Appendix G



AISR for for he Nii'ādah Cesspool Closure Project, Kāhilpalinti and Kāvala, Ka'ū, Hawa'r bland TMKs: muliple

APPENDIX F:

Section 106 Consultation

Initiation Letter

Alan S. Downer, PhD, Administrator June 3, 2024 Page 2 of 5

Overview of the Undertaking

In June 2017 the EPA and the COH-DEM voluntarily entered an Administrative Order on Consent (AOC) to close three County-owned large-capacity œsspools (LCCs) in the town of Nä'alehu. The original proposed project to close the LCCs included the construction of a new watewater collection and transmission system to convey wastewater to a new wastewater treatment facility for processing and clisposal to a land application system planted with native trees, to be established on the project site for that purpose. The construction of the wastewater treatment plant also included a 1,500-foot (ft) extension of an existing drainage channel.

Effective as of August 22, 2022, the EPA and COH-DEM voluntarily entered a Revised AOC, which identified four project atternatives for evaluation:

- Alternative #1: Package Wastewater Treatment Plant and new collection system
 Alternative #2: Package Wastewater Treatment Plant Plants connected to the existing
 - collection system
 Alternative #3: Individual Wastewater System (IWS) Maintenance program model
 - Alternative #4: Individual Wastewater System IWS Operating program model

COH-DEM Wastewater Division (WWD) has evaluated the four treatment alternative options, considering the technical, environmental impact, public input, legal challenges, cost, and assorted factors for the selection of a treatment option. As a result, COH-DEM and WWD have selected Alternative 1 as the preferred option. Under Alternative 1, the County would construct a new sewer collection system in the Na'älehu community to replace the existing system of gravity lines that convey sewage to the three LCCs and connect it to the proposed wastewater treatment and disposal facility (*Attachment A*, *Figure 1*). The WWTP would serve the former therever lots as well as newly accessible parcels for *U* ture connection to the WWTP (*Attachment A*, *Figure* 2). The potential connection of additional parcels may be further assessed by the County in the future.

During construction, the County shall ensure that residential units can maintain access to the sewer system at alt times. Alternative 1 involves utilizing the existing collection system within the Brewer Company house lots and constructing new gravity sewers, wastewater pump station (WWPS), and force main to transport exage from the LCCs to the new WWTP. Alternative 1 also involves the installation of gravity sewers to replace the existing collection.

Under this alternative, the County would perform the following actions:

- Construct a new gravity sewer on Ka'ala iki Road and Na'alehu Spur Road to the WWTP located on a portion of Tax Map Key (TMK) (3) 9-5-007:016.
 Implement drainage improvements within the vicinity of the WWTP within TMKs (3) 9-5-
 - Implement drainage improvements within the vicinity of the WWTP within TMKs (3) 9-5-007:016, (3) 9-5-022:001 and (3) 9-5-021:015.
 Construct a new pump station located on a portion of TMK: (3) 9-5-008:048, and
 - Construct a new pump station located on a portion of 1MK: (3, 9-5-008:048, and construct a new force main, which crosses an existing storm drainage channel at Melia Street, to connect to the Ka'ala'ik Road gravity sewer.
 Construct a new cravity sewer on Onukea Street and Ohai Road to intercent evicting
 - Construct a new gravity sewer on Opukea Street and Ohai Road to intercept existing flow entering the LCCs and divert sewage to the WWPS and transport flows to the gravity sewer along Ka'ala'iki Road.

Alan S. Downer, PhD, Administrator June 3, 2024 Page 3 of 5

- Install gravity sewers within the streets to replace the existing collection system.
 Close and abandon the three LCCs.
- Accommodate future expansion of subsurface effluent disposal located within a portion of TMK: (3) 9-5-022:001.

The newly accessible and potentially newly accessible lots shown in *Attachment A, Figure 2* are not considered part of the undertaking nor the APE, as the respective lot owners would be responsible for the design and completion of these connections and for the proper closure of their individual wastewater systems with their respective private properties.

Area of Potential Effect (APE) and Archaeological Inventory Survey (AIS) Project Area

The APE for the Naʿalehu LCCs replacement project is approximately 80.4 acres. The APE, which is depicted on a 2022 aerial map (*Attachment A, Figure 3*), includes:

- The proposed treatment plant and disposal facility which will be located makai (seaward) of the Māmalahoa Highway at TMKs: (3) 9-5-007:016 and (3) 9-5-022:001. An associated drainage channel may be located on a portion of TMK: (3) 9-5-021:015;
 - The location(s) of proposed pump station(s);
- The proposed collection system primarily in the County streets of Na alehu;
 - The locations of three LCCs which will be decommissioned; and
- Community parcels currently connected to the existing County system.

The AIS for the project area comprises a 49.62-acre portion of the overall APE (*Attachment A*, *Figure 3*). The AIS project area includes all the components of the APE listed above except for the currently connected community parcels. All project staging will be confined within the AIS project area limits at the proposed treatment plant and sewer pump station site(s).

Archaeological Background

Six archaeological studies predating the current undertaking have been identified within or in the immediate vicinity of the APE. These studies are depicted in relation to the project area in *Attachment A, Figure 4* and summarized in *Attachment B, Table 1*. Twenty-six historic properties documented during three of these prior studies are within 1 km of the current APE; these sites are shown in *Attachment A, Figure 4* and summarized in *Attachment B, Table 2*.

In 2005 Haun and Associates conducted an AIS of an approximately 1.0-acre area located southeast of the current APE (Haun and Henry 2005). Extensive prior impacts by historic and modern ranching activity were observed. No archaeological features were documented, and no further work was recommended. Based on the lack of findings the study was termed an "archaeological assessment" pursuant to HAR §13-284-5(5[A]).

In 2006 Cultural Surveys Hawai'i (CSH) undertook a literature review and field inspection (LRFI) for two Ka'ū District schools as part of a state-wide Department of Education (DOE) wastewater systems improvement project. One of the two schools was Nā'ālehu Elementary and Intermediate School, located across the Māmalahoa Highway from the southeastem portion of the current APE (Hammatt and Shideler 2006). The LRFI included data for Land Commission Awards in the vicinity of the school and noted that Nă'ālehu Elementary School is listed on the Hawai'i Register of Historic Places. An on-site archaeological monitoring program was

Alan S. Downer, PhD, Administrator June 3, 2024 Page 4 of 5 recommended. In 2009 CSH undertook archaeological monitoring for the project (Wilkinson et al. 2009); no cultural materials were exposed during project excavations. In 2008 Rechtman Consulting conducted an AIS of 42.5 acres overlapping the western portion of the current APE (Clark et al. 2008). The study documented three newly identified historic properties associated with historic-era ranching and sugar plantation activities: SIHP #s 50-10-74-26408 (complex of stone and concrete-lined trenches representing possible latrines), -26409 (concrete water trough), and -26410 (core-filled boundary wall). No further work was cloomended for these sites. Of the documented sites, SIHP #-26409 (a concrete trough) is closest to the current APE but does not overlap it (*Attachment A*, *Figure 4*).

In 2011 Rechtman Consulting completed an AIS of 2.24 acres located across the Māmalahoa Highway from LCCs 3 and 4 in the current APE (Rechtman 2011). The study documented six newly identified historic properties associated with historic land use, shown in *Attachment* A, *Figure 4*: SIHP #s 50-10-74-28925 (post-1938 residential complex): -28926, -28927, and -28990 (the discrete mid-nineteenth century stone walls): -28928 (core-fill-walled enclosure); and -28029 (complex of historic rock clearing mounds). Archaeological monitoring was recommended for the initial project ground disturbance. In 2013 Rechtman Consulting completed an AIS of 1,044 acres at TMK: (3) 9-5-010:001, east of the proposed treatment plant facility portion of the current APE (Clark et al. 2013). The survey, which extended down to the coast, documented 204 historic properties, including 12 previously which extanded down to the coast, documented 204 historic properties, including 12 previously which extanded down to the coast, documented 204 historic properties, including 12 previously documented sites and 1927 newly documented sites. Sites documented within 1 km of the current APE are shown in Attachment A, *Figure 4*. A wide array of pre-Contact and historic era features were recorded for habitation, windbreaks and shelters, burials, coremonies, petroglyphs, paparuñ, agriculture, transportation, and ranching. The northwestern corner of the 2013 study abuts the current APE along the proposed gravity sewer corridor near the WWTP plant site. A historic ranch complex (SIHP # 50-10-74-29507) was documented overlapping with parcel 016 just downslope; a portion of this site is situated within the northeastern corner of the 2013 study abuts the current APE along the proposed dravity sewer corridor near the WWTP plant site. A historic ranch complex (SIHP # 50-10-74-29507) was documented overlapping with parcel 016 just downslope; a portion of this site is situated within the northeastern corner of the 2013 study abuts the current APE.

CSH completed a literature review analysis for the current undertaking (Purdy et al. 2020) to determine the likelihood that historic properties may be affected by the project and, based on findings, consider cultural resource management recommendations to facilitate project planning and support the project's historic preservation and environmental review compliance. The literature review determined the project area extends along or crosses at least two significant historic roadway alignments, including an undocumented segment of Manalahoa Highway (SIHP # 50-10-47-30187) and Ka'ala'lik Road, also known as the "Cane Haul Road." The concrete flood drainage channel crossed by the project area between Ka'ala'lik Road and Maile Street was built in 1965 and is therefore a historic property. Other constructions within the project area aublicted to prior archaeological survey and may contain remants of the WVTP has not be greater than 50 years old. Purdy et al. (2020) also noted the subjected than so other historic land use, or pre-Contact features located within lava tubes or other areas not disturbed by historic activities. The results of the literature review indicate additional archaeological work is warranted for the NVTP has no content historic Passesment is in accord with the SHPD 21 January 2020 S6E-8 Historic Preservation Review letter (Log. No.: 2019.01550; Doc. No.: 2001SND; requesting additional information about the

Alan S. Downer, PhD, Administrator June 3, 2024 Page 5 of 5 project. An AIS will be completed to fulfill SHPD's request to identify and inventory historic properties within the project area.

Consultations

Section 106 consultation letters have also been sent to Native Hawaiian organizations, consulting parties, and/or interested persons that might attach significance to this area and have invited them to participate in the process. The mailing list is provided in *Attachment C*.

We welcome any comments that you may have on this project's proposed improvements.

We are particularly interested in any information you may have on the historic and cultural sites that have been recorded in the area. In addition, if you are acquainted with any persons or organizations that are knowledgeable about the proposed project area or any descendants with ancestral, lineal, or cultural ties to, cultural knowledge or concerns for, and/or cultural or ancestral, lineal, or cultural ties to, cultural knowledge or concerns for, and/or cultural or eligious attachment to the proposed project area, then we would appreciate receiving their names and contact information.

We would appreciate a written response within thirty (30) calendar days from receipt of this letter. Please address any written comments to email: <u>Chane.Hayashida@doh.hawaii.gov</u> or the following address:

Attn: Chane Hayashida Department of Health, Wastewater Branch 2827 Waimano Home Road, Room 207 Pearl City, HI 96782 Should you have any questions, please contact Chane Hayashida at (808) 586-4294.

Sincerely.

JONATHAN NAGATO, P.E., ACTING CHIEF Wastewater Branch

Attachments

CH:jn

C: Ramzi Mansour (via email at Ramzi Mansour@hawaiicounty.gov) Chris Sparber (via email at <u>Chris.Sparber.@hawaiicounty.gov</u>) Mark Grant (via email at <u>Mark.J.Grant@hawaiicounty.gov</u>)



ATTACHMENT A



Figure 3. Aerial photograph (ESRI 2022) showing the locations of the Area of Potential Effect (APE), AIS project area, and existing Large Capacity Cesspools (LCCs)



Figure 2. Map showing the proposed WWTP service area (courtesy Wilson Okamoto Corp.)



Figure 4. Aerial photograph (ESRI 2022) showing previous archaeological studies and historic properties documented 1 km of the APE

ATTACHMENT B

	,		
Reference	Type of Study	Location	Results
Haun and Henry 2005	Archaeological inventory survey	1 acre in Kaunāmano Ahupua'a; TMK: (3) 9-5-011:001 por.	No historic properties identified; noted extensive prior impacts throughout project area
Hammatt and Shideler 2006	Archaeological literature review and field inspection	Nā'ālehu Elementary School; TMKs: (3) 9- 5-009:006, 015	Recommended on-site archaeological monitoring
Clark et al. 2008	Archaeological inventory survey	42.5 acres in Kāhilipalinui and Kāwala Ahupua'a; TMKs: (3) 9-5- 021:015 and (3) 9-5- 022:001	Documented three newly identified historic properties: SIHP #s 50-10-74-26408, -26409, and -26410, all associated with historic period land use; none of these historic properties appear to be located within the current project area
Wilkinson et al. 2009 Rechtman 2011	Archaeological monitoring Archaeological inventory survey	Na [*] ālehu Elementary School; TMKs: (3) 9- 5-009:006, 015 2-24 acres in Kāwala Ahupua'a; TMK: (3) 9-5-009:003	No significant resources documented during monitoring Documented six newly identified historic properties: SIHP #s 50-10-14-28925, -28926, -28927, -28928 and -28990
Clark et al. 2013	Archaeological inventory survey	1,044 acres in Kāhilipalinui and Kāwala Ahupua'a; TMK: (3) 9-5-010:001	Documented 204 historic properties of which 12 were previously documented and 192 were newly recorded; sites comprised a wide array of pre- Contact and historic are features used for including habitation, windbreaks and shelters, burials, ceremony, petroglyph, papamū, agriculture, transportation, and agriculture, transportation, and ranching; in closest proximity to current project area are SIHP #s 50-10-74-29505, historic wall, and -29507, historic ranching combex

Table 1. Previous archaeological studies in the vicinity of the APE

Table 2. Historic properties previously documented within 1 km of the current APE

		•			
SIHP # (50-10-74-)	Formal Type	# of Features	Function	Age	Reference
26408	Concrete trenches	3	Ranching	Historic	Clark et al. 2008
26409	Concrete trough	1	Ranching	Historic	Clark et al. 2008
26882	Enclosure	1	Habitation	Precontact	Clark et al. 2013
26883	Platform	1	Marker	Historic	Clark et al. 2013
28925	Residential complex	8	Habitation	Historic	Rechtman 2011
28926	Wall	1	Former enclosure	Historic/modern	Rechtman 2011
28927	Wall	1	Former enclosure	Historic/modern	Rechtman 2011
28928	Wall	1	Enclosure	Historic/modern	Rechtman 2011
28929	Rock piles	4	Clearing	Historic	Rechtman 2011
28990	Wall	1	Boundary	Historic/modern	Rechtman 2011
29505	Core-filled wall	1	Ranching/boundary	Historic	Clark et al. 2013
29506	Core-filled wall	1	Ranching/boundary	Historic	Clark et al. 2013
29507	Complex	7	Ranching	Historic	Clark et al. 2013
29508	Concrete trough	1	Ranching	Historic	Clark et al. 2013
29509	Modified outcrop	3	Habitation	Precontact	Clark et al. 2013
25910	Lava tube	1	Habitation	Precontact	Clark et al. 2013
29511	Terrance remnant	1	Indeterminate	Unknown	Clark et al. 2013
29512	Wall remnant	1	Indeterminate	Unknown	Clark et al. 2013
29513	Pavement remnant wall/modified	1	Habitation	Precontact	Clark et al. 2013
	outcrop				
29527	Wall remnant	1	Boundary	Historic	Clark et al. 2013
29528	Core-filled wall	1	Ranching/boundary	Historic	Clark et al. 2013
29530	Modified sink	1	Possible burial	Precontact	Clark et al. 2013
29531	Enclosure remnant	1	Habitation	Precontact	Clark et al. 2013
29532	Modified sink	1	Agriculture	Precontact/historic	Clark et al. 2013
29533	Modified sink/lava tube	1	Habitation	Precontact	Clark et al. 2013
29534	Enclosure	1	Ceremonial	Precontact	Clark et al. 2013
29536	Enclosure/enclosure	2	Habitation	Precontact/historic	Clark et al. 2013

Native Hawaiian Organization	s (NHOs) and Other Con	sulted Parties
Contact	Affiliation	Contact Info
SHPO		
Jr. Downer, Alan	SHPO Administrator	Kakuhihewa Buliding 601 Kamokila Blvd., Suite 555
		Kapolei, HI 96707
		Phone: 808-692-8015
		Fax: 808-692-8020
		E-mail: alan.s.downer@hawaii.gov
NHOS and Individuals	Ľ	
Akana, Paula	Executive Director, Friends	
		HONOIUIU, HI 90604-2239 1908) 522 0822
		(ouo) 322-uozz info@iolanipalace.org
Andrade Lehua	Executive Director Hui	1438 Kilanea Avenue
	Mālama Ola Nā 'Õiwi	Hilo HI 96720
		(808) 969-9220
		lehua@hmono.org
Ayau, E. Halealoha	Kumu, Hui Iwi Kuamoʻo	144 Kulana Street
		Hilo, HI 96720
		(808) 646-9015
		halealohahapai64@gmail.com
Brown, Samson L.	President, Au Puni o	21 Pohai Street
	Паман	ПІО, ПІ 90/20
		aponiz1@gmail.com (808) 315-2141
Crysdale, Shaylan	Kamaʻāina of Kaʻū, worked	scrysdale@TNC.org
	for The Nature	
	Conservancy in Ka'ū	
Cypher, Mahealani	Secretary, Koʻolau	Po Box 4749
	Foundation	Kaneohe, HI 96744
		(808) 226-4195
		malamapono744@aol.com
Dr. Kana'iaupuni, Shawn	President and CEO,	2040 Bachelot Street
	Partners in Development	Honolulu, HI 96817
	Foundation	(808) 595-2752 bid@bidfoundation.om
Faulkner Kiersten	Executive Director Historic	680 Iwilei Road Suite 690
	Hawai'i Foundation	Honolulu. HI 96817
		(808) 523-2900
		preservation@historichawaii.org
[–] ergerstrom, Hanalei	Spokesperson, Na Kupuna	P.O. Box 951
	Moku U Keawe	Kurtistown, HI 96760
		(808) 938-9994 hankhawaiian@vahoo.com
⁻ ontes, Ann	President, Friends of Ka'ū	PO Box 400
	Libraries	Pahala, HI 96777
		(808) 987-7748
		afontes@hawaiiantel.net
Hong, Lea	Hawai'i Island Director,	1003 Bishop Street
	The Trust for Public Land	Pauahi Tower, Suite 740
		10001111, 114Wall 80010 10001 571 0560

ATTACHMENT C

Contact	Affiliation	Contact Info
Hopoi, M. Tonga	Manager of Government Relations, Kamehameha Schools	567 South King Street Honolulu, HI 96813 (808) 523-6368 mihopol@ksbe.edu
Dr. laukea, Lesley K.	Acting History & Culture Branch Chief, SHPD	(808) 692-8023 lesley.iaukea@hawaii.gov
Ferreira, Stacy Kealohalani Ca: Kamakan Ferreira (Lead Compliance Specialist), Lauren Morawski (Compliance Archaeologist), Shane Nelson (Community Outreach Advocate)	CEO, Office of Hawaiian Affairs	560 N. Nimitz Highway, Ste 200 Honoluu, IH 96718 (808) 594-1835 kamakan@oha.org laurenm@oha.org shanen@oha.org
Jackson, Anela	President, 'Aha Mālama, Corp.	91-200 Leimao Place Kapolei, HI 96707 (808) 694-9232 ahamalama.anela@omail.com
Ka'awa, Nohea	<i>Kama'āina</i> of Ka'ū and works for the DLNR	kaawa.nohea@gmail.com
Scott Mahoney c/o Jordan Calpito and Christian Omerod, SHPD Burial Sites	Chair, Hawai'i Island Burial Council	SHPD Hilo Office 40 Po'okela Street Hilo, HI 96720
Cc: Desmon Haumea (Kaʻū Rep)		christian.omerod@hawaii.gov
Kalili, Dre	President, Association of Hawaiian Civic Clubs	P.O. Box 1135 Honolulu, HI 96807 ahcc.nuhou@gmail.com
Kawachi, Wayne	<i>Kama'äina</i> of Ka'ū and President of 'O Ka'ū Kākou organization	'O Ka'ŭ Kākou PO Box 365 Pahala, HI 96777 okaukakou.org@gmail.com
Kuluwaimaka, Elizabeth	Aha Moku Council, Ka'ū Representative	(808) 990-9327 nawai1040@aol.com
Makuakane, Paul	Aha Moku Council, Ka'ū Representative	pmaku808@yahoo.com
Paik, Linda Kaleo	Secretary, Ala Kahakai Trail Association	Ala Kahakai Trail Association P.O. Box 2338 Kamuela, H) 96743 (808) 354-7765 kateopatk@yahoo.com
Penn, David	Program Specialist, DLNR Legacy Land Conservation Program	1151 Punchbowl Street, Room 325 Honolulu, HI 96813 (808) 586-0921
Ragsdale, Dennis W.	Advocate General, Order of Kamehameha I	1777 Ala Moana Blvd, #142-102 Honolulu, HI 96815-1603 (808) 235-2425 order@kamehaneha-1.org
Steiner, William W. Moekahi	Pacific Agricultural Land Management Systems	PO Box 4565 Hilo, HI 96720 (808) 294-0750 (2006) 294-0750 com

Contact	Affiliation	Contact Info
Trask, Mililani	Convenor, Na Koa Ikaika	PO Box 6377
	Ka Lahui Hawaii	Hilo, HI 96720
		(808) 961-4811
		mililani.trask@icllchawaii.com
Vierra, Darlyne	Aha Moku Council, Ka'ū	(808) 640-8740
	Representative	dpvierra@yahoo.com
Wise, Taffi	Executive Director, Kanu o	PO Box 6511
	ka 'Āina Learning 'Ohana	Kamuela, HI 96743
		(808) 887-1117
		taffi@kalo.org
Yokoyama Jr., Melvin K.	Chairman of the Board,	P.O. Box 273
	Malama Ka'u Foundation	Pahala, HI 96777
		(619) 855-5709
		mel yokoyama@akamai-
		intelligence.com
Yoshida, Berkeley	President, Hawaiian Civic	hawaiicivicclubkau@gmail.com
	Club of Kaʻū	

ATTACHMENT D

2008 An Archaeological Inventory Survey of TMKs: 3-9-5-021:015 and 3-9-5-022:001, Kähilipali Nui and Kāwala ahupua'a, Ka'ū District, Island of Hawai'i. Rechtman Clark, Mathew R., Herbert Poepoe, and Robert B. Rechtman

Consulting, Hilo, Hawai'i.

Clark, Matthew R., J. David Nelson, Amy L. Ketner, Ashton Dircks Ah Sam, Lauryl K. Zenobi, and Robert B. Rechtman

- 2013 An Archaeological Inventory Survey of TMK: 3-9-5-010:001, Kāwala Ahupua'a, Ka'ü District, Island of Hawai'i. Rechtman Consulting, Hilo, Hawai'i.
 - at www. California. Available online 2022 World Imagery. Esri, Redlands, C arcgisonline.com/maps/World_Imagery.

Hammatt, Hallett H. and David W. Shideler

2006 Archaeological Literature Review and Field Check Study of Two DOE Schools, Ka'ū District, Island of Hawa'ï, Hawa'ï Inter-Island DOE Cesspool Project, TMK: (3) 9-6-005:008, 039; 95-009:006, 015. Cultural Surveys Hawai'i, Inc., Kailua, Hawai'i.

Haun, Alan E. and Dave Henry 2005 Archaeological Inventory Survey, TMK: 9-5-5:017, 018 and 9-6-6:004, Lands of Kaunamano, Ka'u District, Island of Hawai'i. Haun & Associates, Kea'au, Hawai'i.

Purdy, Samantha, Sarah Wilkinson, and Hallett H. Hammatt

2020 Archaeological Literature Review Report for the Na alebu Large Capacity Cesspools Replacement Project, Kāhilipalinui and Kāwala Ahupua'a, Ka 'ū District, Hawai'i Island, TMKs: multiple. Cultural Surveys Hawai'i, Inc., Kailua, Hawai'i.

Rechtman, Robert B.

An Archaeological Survey of TMK:3-9-5-09:003 in Compliance with Section 106 of the National Historic Preservation Act, Kāwala Ahupua'a, Ka'ū District, Island of Hawai'i. Rechtman Consulting, Hilo, Hawai'i. 2011

Wilkinson, Sarah, Rosanna Runyon, Aulii Mitchell, and Hallett H. Hammatt

Archaeological Monitoring Representation of National Intermediate School, Hawai'i Inter-Island DOE Cesspool Project, Kaunāmano Ahupura, Ka'ū District, Island of Hawai'i, TMK: [3] 9-5-006:015. Cultural Surveys Hawai'i, Inc., Kailua, Hawaii 2009

APPENDIX G:

Early Consultation Package Comment and Response Letters

JOSH GREEN, M.D. GOVERNOR KE KIA'ÅINA



KEITH A. REGAN COMPTROLLER KA LUNA HO'OMALU HANA LAULĀ

MEOH-LENG SILLIMAN DEPUTY COMPTROLLER KA HOPE LUNA HO'OMALU HANA LAULÂ

STATE OF HAWAI'I | KA MOKU'ÃINA O HAWAI'I DEPARTMENT OF ACCOUNTING AND GENERAL SERVICES | KA 'OIHANA LOIHELU A LAWELAWE LAULA P.O. BOX 119, HONOLULU, HAWAII 96810-0119

(P)24.067

APR 1 2 2024

Keola Cheng Wilson Okamoto Corporation 1907 South Beretania Street. Suite 400 Honolulu, Hawaii 96826

APR 15 2024 RED ON DRAWATO CORPORATION

Dear Keola Cheng:

Subject: Environmental Information Document and Environmental Assessment **Consultation Package** Naalehu LCC Closure Naalehu, Hawaii Island, Hawaii

Thank you for the opportunity to provide comments on the subject project. The Department of Accounting and General Services has control and management of a property identified Tax Map Key (3) 9-5-021:010, which is in the general area of the proposed project. The Naalehu State Office Building as well as the Naalehu Public Library are located on this property. We have some concerns with Alternative 1 for potential odors given the proximity of the proposed location for the wastewater treatment plant to the Naalehu State Office Building and Public Library. We look forward to receiving additional details on the proposed project when it is available.

If you have any questions, your staff may call David DePonte of the Planning Branch at 586-0492, or email at david.c.deponte@hawaii.gov.

Sincerely,

GORDON S. WOOD Acting Public Works Administrator

DD:mo Roger Ross – DAGS Hawaii

c:





10345-05 September 8, 2024

Mr. Gordan Wood Department of Accounting and General Services State of Hawaii P.O. Box 119 Honolulu, HI 96810-0119

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nāʿālehu, Hawaiʿi Island, Hawaiʿi

Dear Mr. Wood:

Thank you for your letter dated April 12, 2024, regarding the regarding the subject Early Consultation Package for for the Na'alehu Large Capacity Cesspool Closure Project on the Island of Hawai'i. We acknowledge your comments and they have been considered in the preparation of the Draft EA with regard to meeting content requirements prescribed in Hawai'i Administrative Rules, Title 11, Chapter 200.1, Section 18. A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

We acknowledge your comments and note that the Proposed wastewater treatment measures shall include means and methods for odor control. As described in Section 2.1.1 of the Draft EA / EID, Alternative 1 shall include a preliminary treatment method which is intended to minimize potential orders as well as an odor control system which shall further filter odorous air.

Please note that the Draft EA has been published and made available for review and comment in the current issue of the State of Hawai'i's Environmental Review Program's (ERP) The Environmental Notice.

We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning

cc: Mr. Mark Grant, Wastewater Division Project Coordinator

JOSH GREEN, M.D. GOVERNOR | KE KIA'ĀINA

SYLVIA LUKE LIEUTENANT GOVERNOR | KA HOPE KIA'ĀINA





DAWN N. S. CHANG CHAIRPERSON BOARD OF LAND AND NATURAL RESOURCES COMMISSION ON WATER RESOURCE MANAGEMENT

STATE OF HAWAI'I | KA MOKU'ĀINA 'O HAWAI'I DEPARTMENT OF LAND AND NATURAL RESOURCES KA 'OIHANA KUMUWAIWAI 'ĀINA LAND DIVISION

P.O. BOX 621 HONOLULU, HAWAII 96809

April 26, 2024

Wilson Okamoto Corporation Attn: Mr. Keola Cheng Director of Planning 1907 South Beretania Street, Suite 400 Honolulu, Hawaii 96826

via email: publiccomment@wilsonokamoto.com

Dear Mr. Cheng:

SUBJECT: Environmental Information Document and Environmental Assessment Consultation Package for the **Na'alehu Large Capacity Cesspool Closure** located at Na'alehu, Ka'u District, Island of Hawaii; Various TMKs, on behalf of County of Hawaii, Department of Environmental Management

Thank you for the opportunity to review and comment on the subject matter. The Land Division of the Department of Land and Natural Resources (DLNR) distributed or made available a copy of your request pertaining to the subject matter to DLNR's Divisions for their review and comments.

At this time, enclosed are comments from the Land Division-Hawaii District on the subject matter. Should you have any questions, please feel free to contact Darlene Nakamura at (808) 587-0417 or email: <u>darlene.k.nakamura@hawaii.gov</u>. Thank you.

Sincerely,

Russell Tsuji

Russell Y. Tsuji Land Administrator

Enclosure cc: Central Files
JOSH GREEN, M.D. GOVERNOR | KE KIA AINA

SYLVIA LUKE





DAWN N. S. CHANG CHAIRPERSON BOARD OF LAND AND NATURAL RESOURCES COMMISSION ON WATER RESOURCE MANAGEMENT

STATE OF HAWAI'I | KA MOKU'ĀINA 'O HAWAI'I DEPARTMENT OF LAND AND NATURAL RESOURCES KA 'OIHANA KUMUWAIWAI 'ĂINA LAND DIVISION

P.O. BOX 621 HONOLULU, HAWAII 96809

April 2, 2024

MEMORANDUM

TO:

DLNR Agencies:

Div. of Aquatic Resources

Div. of Boating & Ocean Recreation

- X Engineering Division (DLNR.ENGR@hawaii.gov)
- X Div. of Forestry & Wildlife (rubyrosa.t.terrago@hawaii.gov)
- Div. of State Parks

X Commission on Water Resource Management (DLNR.CWRM@hawaii.gov) Office of Conservation & Coastal Lands

X Land Division - Hawaii District (gordon.c.heit@hawaii.gov)

X Aha Moku Advisory Committee (leimana.k.damate@hawaii.gov)

Russell Y. Tsuji, Land Administrator Russell Tsuji FROM: Environmental Information Document and Environmental Assessment SUBJECT: Consultation Package for the Na'alehu Large Capacity Cesspool Closure LOCATION: Na'alehu, Ka'u District, Island of Hawaii; Various TMKs APPLICANT: Wilson Okamoto Corporation on behalf of County of Hawaii, Department of Environmental Management

Transmitted for your review and comment is information on the above-referenced subject matter. Please submit comments by April 26, 2024.

If no response is received by the above date, we will assume your agency has no comments. Should you have any questions about this request, please contact Darlene Nakamura at darlene.k.nakamura@hawaii.gov. Thank you.

BRIEF COMMENTS:

We have no objections.

We have no comments.)

We have no additional comments.)

) Comments are included/attached.

Sic

Signed:	Gordon Heit	
Print Name:	Gordon Heit	
Division:	Land Division	
Date:	April 19, 2024	

Attachments CC: Central Files



Mr. Gordan Heit Department of Land and Natural Resources Land Division State of Hawaii P.O. Box 621 Honolulu, HI 96809

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nāʿālehu, Hawaiʿi Island, Hawaiʿi

Dear Mr. Heit:

Thank you for your letter dated April 2, 2024, regarding the regarding the subject Early Consultation Package for for the Na'alehu Large Capacity Cesspool Closure Project on the Island of Hawai'i. We acknowledge that the Department of Land and Natural Resources Land Division does not have any comments or objections. A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

Please note that the Draft EA has been published and made available for review and comment in the current issue of the State of Hawai'i's Environmental Review Program's (ERP) The Environmental Notice.

We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning

JOSH GREEN, M.D. GOVERNOR | KE KIA'ĀINA

SYLVIA LUKE LIEUTENANT GOVERNOR | KA HOPE KIA'ĀINA





DAWN N. S. CHANG CHAIRPERSON BOARD OF LAND AND NATURAL RESOURCES COMMISSION ON WATER RESOURCE MANAGEMENT

STATE OF HAWAI'I | KA MOKU'ĀINA 'O HAWAI'I DEPARTMENT OF LAND AND NATURAL RESOURCES KA 'OIHANA KUMUWAIWAI 'ĀINA LAND DIVISION

P.O. BOX 621 HONOLULU, HAWAII 96809

May 13, 2024

Wilson Okamoto Corporation Attn: Mr. Keola Cheng Director of Planning 1907 South Beretania Street, Suite 400 Honolulu, Hawaii 96826

via email: publiccomment@wilsonokamoto.com

Dear Mr. Cheng:

SUBJECT: Environmental Information Document and Environmental Assessment Consultation Package for the **Na'alehu Large Capacity Cesspool Closure** located at Na'alehu, Ka'u District, Island of Hawaii; Various TMKs, on behalf of County of Hawaii, Department of Environmental Management

Thank you for the opportunity to review and comment on the subject matter. In addition to our previous comments dated April 26, 2024, enclosed are comments from the Division of Forestry & Wildlife on the subject matter. Should you have any questions, please feel free to contact Darlene Nakamura at (808) 587-0417 or email: <u>darlene.k.nakamura@hawaii.gov</u>. Thank you.

Sincerely,

Russell Tsuji

Russell Y. Tsuji Land Administrator

Enclosure cc: Central Files JOSH GREEN, M.D. GOVERNOR | KE KIA'ÄINA

SYLVIA LUKE LIEUTENANT GOVERNOR | KA HOPE KIA'ÄINA





DAWN N. S. CHANG CHAIRPERSON BOARD OF LAND AND NATURAL RESOURCES COMMISSION ON WATER RESOURCE MANAGEMENT

STATE OF HAWAI'I | KA MOKU'ĀINA 'O HAWAI'I DEPARTMENT OF LAND AND NATURAL RESOURCES KA 'OIHANA KUMUWAIWAI 'ĀINA LAND DIVISION

P.O. BOX 621 HONOLULU, HAWAII 96809

April 2, 2024

MEMORANDUM

FROM:	DLNR Agencies: Div. of Aquatic Resources Div. of Boating & Ocean Recreation X Engineering Division (DLNR.ENGR@hawaii.gov) X Div. of Forestry & Wildlife (rubyrosa.t.terrago@hawaii.gov) Div. of State Parks X Commission on Water Resource Management (DLNR.CWRM@hawaii.gov) Office of Conservation & Coastal Lands X Land Division – Hawaii District (gordon.c.heit@hawaii.gov) X Aha Moku Advisory Committee (leimana.k.damate@hawaii.gov)
TO:	Russell Y. Tsuji, Land Administrator Russell Tsuji
SUBJECT:	Environmental Information Document and Environmental Assessment Consultation Package for the Na'alehu Large Capacity Cesspool Closure
LOCATION:	Na'alehu, Ka'u District, Island of Hawaii; Various TMKs
APPLICANT:	Wilson Okamoto Corporation on behalf of County of Hawaii, Department of Environmental Management

Transmitted for your review and comment is information on the above-referenced subject matter. Please submit comments by **April 26, 2024.**

If no response is received by the above date, we will assume your agency has no comments. Should you have any questions about this request, please contact Darlene Nakamura at <u>darlene.k.nakamura@hawaii.gov</u>. Thank you.

BRIEF COMMENTS:

- () We have no objections.
-) We have no comments.
- () We have no additional comments.

 $(\mathbf{\nabla})$ Comments are included/attached.

Lindsey Nietmann

Signed:

Print Name: Lindsey Nietmann, Acting Wildlife Prog. Mgr.

May 9, 2024

Date:

JOSH GREEN, M.D. GOVERNOR | KE KIA AINA

SYLVIA LUKE LIEUTENANT GOVERNOR | KA HOPE KIA'ÄINA



DAWN N.S. CHANG CHAIRPERSON BOARD OF LAND AND NATURAL RESOURCES COMMISSION ON WATER RESOURCE MANAGEMENT

> RYAN K.P. KANAKA'OLE FIRST DEPUTY

DEAN D. UYENO ACTING DEPUTY DIRECTOR - WATER

AQUATIC RESOURCES BOATING AND OCEAN RECREATION BUREAU OF CONVEYANCES COMMISSION ON WATER RESOURCE MANAGEMENT CONSERVATION AND CASTAL LANDS CONSERVATION AND RESOURCES ENFORCEMENT ENGIREERING FORESTRY AND WILDLIFE HISTORIC PRESERVATION KAHOOLAWE ISLAND RESERVE COMMISSION LAND STATE PARKS

Log no. 4502



STATE OF HAWAI'I | KA MOKU'ĀINA 'O HAWAI'I DEPARTMENT OF LAND AND NATURAL RESOURCES KA 'OIHANA KUMUWAIWAI 'ĀINA

> DIVISION OF FORESTRY AND WILDLIFE 1151 PUNCHBOWL STREET, ROOM 325 HONOLULU, HAWAII 96813

> > May 8, 2024

MEMORANDUM

- TO: RUSSEL Y. TSUJI, Administrator Land Division
- **FROM:** LINDSEY NIETMANN, Acting Wildlife Program Manager Division of Forestry and Wildlife
- **SUBJECT:** Environmental Information Document and Environmental Assessment Consultation Package for the Na'alehu Large Capacity Cesspool Closure on Hawai'i Island

The Department of Land and Natural Resources, Division of Forestry and Wildlife (DOFAW) has received your request for consultation regarding the Environmental Information Document and Environmental Assessment Consultation Package for the Na'alehu Large Capacity Cesspool Closure in the Ka'u District on the Island of Hawai'i. The proposed project involves the construction of facilities which would allow the County to close the three Large Capacity Cesspools (LCC). The proposed project area includes approximately 204 discrete tax parcels (in whole or in part) and portions of multiple County of Hawai'i roadways in Na'alehu. The four alternatives include a package plant with a newly constructed sewer collection system in the Na'alehu community, a package plant utilizing the existing 80-year-old collection system, construction of an individual wastewater system (IWS) in which the county maintains the IWS, or construction of an IWS in which the individual homeowner maintains the IWS. By selecting one of these alternatives, it will allow the county to close the LCCs and provide a new SDWA compliant solution for handling wastewater generated by the Na'alehu Community.

The State listed 'ōpe'ape'a or Hawaiian Hoary Bat (*Lasiurus cinereus semotus*) could potentially occur at or in the vicinity of the project and may roost in nearby trees. Any required site clearing should be timed to avoid disturbance to bats during their birthing and pup rearing season (June 1 through September 15). During this period woody plants greater than 15 feet (4.6 meters) tall should not be disturbed, removed, or trimmed. Barbed wire should also be avoided in any construction as bats can become ensnared and killed by such fencing material during flight.

Artificial lighting can adversely impact seabirds that may pass through the area at night by causing them to become disoriented. This disorientation can result in their collision with

manmade structures or the grounding of birds. For nighttime work that might be required, DOFAW recommends that all lights used be fully shielded to minimize the attraction of seabirds. Nighttime work that requires outdoor lighting should be avoided during the seabird fledging season, from September 15 through December 15, when young seabirds make their maiden voyage to sea.

If nighttime construction is required during the seabird fledgling season (September 15 to December 15), we recommend that a qualified biologist be present at the project site to monitor and assess the risk of seabirds being attracted or grounded due to the lighting. If seabirds are seen circling around the area, lights should then be turned off. If a downed seabird is detected, please follow DOFAW's recommended response protocol by visiting https://dlnr.hawaii.gov/wildlife/seabird-fallout-season/.

Permanent lighting also poses a risk of seabird attraction, and as such should be minimized or eliminated to protect seabird flyways and preserve the night sky. For illustrations and guidance related to seabird-friendly light styles that also protect seabirds and the dark starry skies of Hawai'i please visit https://dlnr.hawaii.gov/wildlife/files/2016/03/DOC439.pdf.

State-listed waterbirds such as ae'o or Hawaiian stilt (*Himantopus mexicanus knudseni*), 'alae ke'oke'o or Hawaiian coot (*Fulica alai*), and nēnē or Hawaiian Goose (*Branta sandvicensis*) could potentially occur at or in the vicinity of the proposed project site. It is against State law to harm or harass these species. If any of these species are present during construction, all activities within 100 feet (30 meters) should cease and the bird or birds should not be approached. Work may continue after the bird or birds leave the area of their own accord. If a nest is discovered at any point, please contact the Hawai'i Branch DOFAW Office at (808) 974-4221 and establish a buffer zone around the nest.

DOFAW is concerned about the wastewater treatment facility attracting vulnerable birds to areas that may host nonnative predators such as cats, rodents, and mongooses. We therefore recommend taking action to minimize predator presence, i.e., remove cats, place bait stations for rodents and mongoose, and provide covered trash receptacles. Implementing additional

The endemic pueo or Hawaiian Short-Eared Owl (*Asio flammeus sandwichensis*) could potentially nest in the project area. Pueo nest on the ground and active nests have been found year-round. Before any potential vegetative alteration, especially ground-based disturbance, we recommend that line transect surveys are conducted during crepuscular hours through the project area. If a pueo nest is discovered, a minimum buffer distance of 100 meters from the nest should be established until chicks are capable of flight.

The State listed 'io or Hawaiian Hawk (*Buteo solitarius*) may occur in the project vicinity. Prior to undertaking vegetation clearing, DOFAW recommends that pre-construction surveys of the area be conducted by a qualified biologist following appropriate survey methods (Gorressen et al., 2008) to ensure no Hawaiian Hawk nests are present, which may occur during the breeding season from March to September. The survey should be conducted at least 10 days prior to the start of construction. If an 'io nest is detected, a buffer zone of 100 meters (330 feet) should be established around it where no construction shall occur until the chick or chicks have fledged, or the nest is abandoned and DOFAW staff should be immediately notified. If adult individuals are detected in the area during construction, all activities within 30 meters (100 feet) of the bird should cease. Work may continue when the bird has left the area on its own.

DOFAW recommends using native plant species for landscaping that are appropriate for the area; i.e., plants for which climate conditions are suitable for them to thrive, plants that historically occurred there, etc. Please do not plant invasive species. DOFAW also recommends referring to www.plantpono.org for guidance on the selection and evaluation of landscaping plants and to determine the potential invasiveness of plants proposed for use in the project.

DOFAW recommends minimizing the movement of plant or soil material between worksites. Soil and plant material may contain detrimental fungal pathogens (e.g., Rapid 'Ōhi'a Death), vertebrate and invertebrate pests (e.g., Little Fire Ants, Coqui Frogs, etc.), or invasive plant parts (e.g., African Tulip, Octopus Tree, Trumpet Tree, etc.) that could harm our native species and ecosystems. We recommend consulting the Big Island Invasive Species Committee (BIISC) at (808) 933-3340 to help plan, design, and construct the project, learn of any high-risk invasive species in the area, and ways to mitigate their spread. All equipment, materials, and personnel should be cleaned of excess soil and debris to minimize the risk of spreading invasive species.

The invasive Coconut Rhinoceros Beetle (CRB) or Oryctes rhinoceros is found on the islands of O'ahu, Hawai'i Island, Maui and Kaua'i. On July 1, 2022, the Hawai'i Department of Agriculture (HDOA) approved Plant Quarantine Interim Rule 22-1. This rule restricts the movement of CRB-host material within or to and from the island of O'ahu, which is defined as the Quarantine Area. Regulated material (host material or host plants) is considered a risk for potential CRB infestation. Host material for the beetle specifically includes a) entire dead trees, b) mulch, compost, trimmings, fruit and vegetative scraps, and c) decaying stumps. CRB host plants include the live palm plants in the following genera: Washingtonia, Livistona, and Pritchardia (all commonly known as fan palms), Cocos (coconut palms), Phoenix (date palms), and Roystonea (royal palms). When such material or these specific plants are moved there is a risk of spreading CRB because they may contain CRB in any life stage. For more information regarding CRB, please visit https://dlnr.hawaii.gov/hisc/info/invasive-species-profiles/coconut-rhinoceros-beetle/.

Gorresen, P. M., R. J. Camp, J. L. Klavitter, and T. K. Pratt. 2008. Abundance, distribution and population trend of the Hawaiian Hawk: 1998-2007. Hawai'i Cooperative Studies Unit Technical Report HCSU-009. University of Hawai'i at Hilo. 53 pp., incl. 8 figures, 3 tables & 1 appendix.

We recommend that Best Management Practices are employed during and after construction to contain any soils and sediment with the purpose of preventing damage to near-shore waters and marine ecosystems.

We appreciate your efforts to work with our office for the conservation of our native species. These comments are general guidelines and should not be considered comprehensive for this site or project. It is the responsibility of the applicant to do their own due diligence to avoid any negative environmental impacts. Should the scope of the project change significantly, or should it become apparent that threatened or endangered species may be impacted, please contact our staff as soon as possible. If you have any questions, please contact Katherine Cullison, Protected Species Habitat Conservation Planning Coordinator at <u>katherine.cullison@hawaii.gov</u>.

Sincerely,

Lindsey Nietmann

LINDSEY NIETMANN Acting Wildlife Program Manager



Ms. Lindsey Nietmann Department of Land and Natural Resources Division of Forestry and Wildlife State of Hawaii 1151 Punchbowl Street, Room 325 Honolulu, HI 96813

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nāʿālehu, Hawaiʿi Island, Hawaiʿi

Dear Ms. Nietmann:

Thank you for your letter dated May 8, 2024 regarding the Early Consultation Environmental Assessment for the Na'alehu Large Capacity Cesspool Closure Project on the Island of Hawai'i. We acknowledge your comments and they have been considered in the preparation of the Draft EA with regard to meeting content requirements prescribed in Hawai'i Administrative Rules, Title 11, Chapter 200.1, Section 18. A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

We acknowledge your comments and note that a Natural Resource Assessment has been completed for the project area. The results of the assessment and best management practices are described in Section 4.6 of the Draft EA.

Please note that the Draft EA has been published and made available for review, and comment in the current issue of the State of Hawai'i's Environmental Review Program's (ERP) The Environmental Notice. We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning



CUST 3-3-1 H-W/G

April 11, 2024

Wilson Okamoto Corporation 1907 South Beretania Street, Suite 400 Honolulu, HI 96826 Attention: Mr. Keola Cheng

To Whom It May Concern:

SUBJECT: Draft Environmental Information Document (EID)/ Environmental Assessment (EA) Nā'ālehu Large Capacity Cesspool (LCC) Nā'ālehu, Hawai'i Island, Hawai'i Tax Map Key: 9-5-xxx:various

Thank you for the opportunity to comment on the subject's Draft Environmental Information Document (EID) and Environmental Assessment (EA). Hawaiian Electric will be able to provide electrical service to the proposed development in Nā'ālehu. A detailed analysis will be performed after the receipt of the consultant's detailed design drawings and estimated load. The following is a summary of our comments:

- 1. Generation capacity As of January 2024, Hawaiian Electric's current system peak load is 187.3MW-net and our total firm generation system capability is 261.4MW-net.
- Electrical Substation The area is served by our existing Punalu'u and South Point Substations and a 12,470 volt overhead distribution along Mamalahoa Highway and in surroundings areas. The capacity of our existing substation may be adequate to serve the anticipated load.
- Off-Site Electrical Distribution System The existing off-site 12,470 volt distribution system may not be adequate to serve the some of the proposed alternatives for this project.
- On-Site Electrical Distribution System On-site distribution line extensions and easements may be required on the developer's property to serve the anticipated load.

After the development's detailed loading and civil plans are submitted, Hawaiian Electric will prepare a firm cost to provide electrical power to this development.

Hawaiian Electric recommends energy efficient and conservation measures to reduce the maximum electrical demand and energy consumption. The developer may call Hawai'i Energy at (808) 537-5577 for questions or details on available programs.

Nā'ālehu LCC Page 2 April 11, 2024

It is encouraged that the developer's electrical consultant open a pre-service request with Hawaiian Electric via our <u>link to our Interconnection Tool</u> as soon as practicable to ensure timely electrical facility installation. For more information and instructions here is the link to our website: <u>Electrical Services > Builder, Developers & Contractors.</u>

If you have any questions, please contact (808) 969-0311 or email us at <u>Cl@hawaiianelectric.com</u>.

Me ke aloha pumehana,

Shelley Doctor <u>shelley.doctor@hawaiianelectric.com</u> Designer III Transmission and Distribution Engineering

email: D. Demichelis



Ms. Shelley Doctor Planner III Hawaiian Electric 74-5519 Kaiwi Street Kailua-Kona, HI 96740

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nāʿālehu, Hawaiʿi Island, Hawaiʿi

Dear Ms. Doctor:

Thank you for your letter dated April 11, 2024, regarding the Early Consultation Environmental Assessment (EA) for the Nāʿālehu Large Capacity Cesspool (LCC) Closure Project on the Island of Hawaiʿi. We appreciate your input on the project's electrical service needs.

We acknowledge your comments and have carefully considered them in the preparation of the Draft Environmental Assessment (EA). A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

Please note that the Draft EA has been published and made available for review and comment in the current issue of the State of Hawai'i's Environmental Review Program's (ERP) The Environmental Notice.

We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning

Mitchell D. Roth Mayor



Benjamin T. Moszkowicz Police Chief

> Reed K. Mahuna Deputy Police Chief

April 10, 2024

 POLICE DEPARTMENT

 349 Kapi'olani Street
 • Hilo, Hawai'i 96720-3998

 (808) 935-3311
 • Fax (808) 961-2389

County of Hawai`i

Mr. Keola Cheng Planning Director Wilson Okamoto Corporation 1907 South Beretania Street, Suite 400 Honolulu, HI 96826



Dear Mr. Cheng:

SUBJECT: ENVIRONMENTAL INFORMATION DOCUMENT AND ENVIRONMENTAL ASSESSMENT CONSULTATION PACKAGE FOR NĀ'ĀLEHU LCC CLOSURE NĀ'ĀLEHU, HAWAI'I ISLAND, HAWAI'I

We appreciate you allowing us the opportunity to review and provide commentary on the proposed Nā'ālehu Large Capacity Cesspool (LLC) Closure project (Proposed Project) located in the Ka'ū District.

The Hawai'i Police Department has reviewed the proposal and does not have any comments or concerns at this time.

Should you have any questions or concerns, please contact Acting Captain Pernell Hanoa, Commander of our Ka'ū District, at phone number (808) 939-2520 or via email at <u>Pernell.Hanoa@hawaiicounty.gov</u>.

Sincerely,

BENJAMIN T. MOSZKOWICZ POLICE CHIEF CHAD/BASOUE ASSISTANT POLICE CHIEF AREA TH-OPERATIONS

PH/jaj 24HQ0392



Mr. Benjamin Moszkowicz Hawaii Police Department State of Hawaii 349 Kapi'olani Street Hilo, HI 96720-2389

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nāʿālehu, Hawaiʿi Island, Hawaiʿi

Dear Mr. Moszkowicz:

Thank you for your letter dated April 10, 2024, regarding the regarding the subject Early Consultation Package for the Na'alehu Large Capacity Cesspool Closure Project on the Island of Hawai'i. We acknowledge that the County of Hawai'i Police Department does not have any comments at this time. A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

Please note that the Draft EA has been published and made available for review and comment in the current issue of the State of Hawai'i's Environmental Review Program's (ERP) The Environmental Notice.

We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning



STATE OF HAWAI'I OFFICE OF PLANNING & SUSTAINABLE DEVELOPMENT

235 South Beretania Street, 6th Floor, Honolulu, Hawai'i 96813 Mailing Address: P.O. Box 2359, Honolulu, Hawai'i 96804

SYLVIA LUKE LT. GOVERN

MARY ALICE EVANS DIRECTOR

Telephone: Fax: Web:

(808) 587-2846 (808) 587-2824 https://planning.hawaii.gov/

DTS202404030917NA

April 24, 2024

Environmental Review Program

Land Use Commission

Special Plans Branch

State Transit-Oriented Development

Statewide Geographic Information System

Statewide Sustainability Branch Mr. Keola Cheng **Planning Director** Wilson Okamoto Corporation 1907 South Beretania Street, Suite 400 Honolulu, Hawai'i 96826

Dear Mr. Cheng:

Subject: Environmental Information Document and Environmental Assessment Early Consultation for Nā'ālehu Large Capacity Cesspool Closure Project, Nāʿālehu, Hawaiʿi Island, Hawaiʿi

The Office of Planning and Sustainable Development (OPSD) is in receipt of your early consultation request, received April 3, 2024, on the preparation of an Environmental Information Document (EID) and Environmental Assessment (EA), for the proposed Nā'ālehu Large Capacity Cesspool (LCC) Closure Project, located in the Ka'ū District on the island of Hawai'i.

According to the request, the community of Nā'ālehu is located approximately 64 miles southwest of Hilo in the Ka'ū District, and 2.06 miles from the shoreline. The proposed project consists of two parts: Nāʿālehu Large Capacity Cesspool Conversion and Nā'ālehu Wastewater Collection System. The project will construct facilities to allow the County of Hawaii to close the three LLCs in Nāʿālehu for the compliance requirements of the Amended Administrative Order on Consent (AOC) between the U.S. Environmental Protection Agency (EPA) and the County of Hawaii, and the applicable portions of the Clean Water Act. The recently Amended AOC, which took effect on August 22, 2022, requires the three LLCs to be closed no later than December 31, 2027. The Amended AOC 31.a. requires evaluation of following alternatives:

• Alternative 1 - A package plant and new collection system. The County would construct a new sewer collection system to replace the existing system of gravity lines that convey sewage to the three LCCs, and connect to the proposed wastewater treatment and disposal facility.

Management

Program

Coastal Zone

Land Use Division

Mr. Keola Cheng April 24, 2024 Page 2

- *Alternative 2* A package plant connected to the existing collection system. The County would perform the same actions under Alternative 1. However, the new wastewater treatment and disposal facility and pump station would be connected to the existing 80-year-old collection system.
- Alternative 3 A maintenance contract model Individual Wastewater System (IWS) program. The County funds and manages project construction of the IWS. The maintenance program includes county personnel being responsible for necessary maintenance to the IWS such as pumping the septic tanks.
- Alternative 4 An operating permit model IWS program. The County is to fund design and construct new IWS systems and administer an operating permit program for the IWS systems to the homeowners. The homeowners would be responsible for maintenance scheduling, contracting and paying for a service provider to conduct maintenance and/or responding to trouble calls, monitoring and record keeping of maintenance.
- Alternative 5 No action. The EPA has determined that the County of Hawaii, as the current owner and/or operator of two LCCs that serve approximately 109 private residences in the community of Pahala and three LCCs that serve approximately 164 private residences in the community of Nā'ālehu, violated and continues to violate the Safe Drinking Water Act and its Underground Injection Control program requirements for existing LCCs.

The OPSD has reviewed the subject request, and has the following comments to offer:

- The EA shall discuss the proposing agency or agencies, the requirements of the National Environmental Policy Act, and the triggers set forth in Hawaii Revised Statutes (HRS) Chapter 343, and list all required permits and approvals for the proposed Nāʿālehu LCC closure project.
- 2. The State of Hawaii Coastal Zone Management (CZM) Area encompasses the entire state. The Hawaii CZM Law, HRS Chapter 205A, requires all state and county agencies to enforce the CZM objectives and policies. The subject EA should include an assessment with mitigation measures, if needed, as to how the proposed development conforms to each of the CZM objectives and supporting policies set forth in HRS Chapter 205A-2, as amended.
- 3. Under Alternative 2, the new wastewater treatment and disposal facility and pump station would be connected to the existing 80-year-old collection system. The EA should evaluate the lifespan of the existing collection system, and assess whether such an old collection system will be compatible with new wastewater treatment and disposal facility as critical infrastructure.

Mr. Keola Cheng April 24, 2024 Page 3

- 4. OPSD recommends that preservation and burial treatment plans should be prepared for any identified archaeological features. Should any archaeological or cultural resources, or burials be discovered during ground excavation, all construction work shall be ceased immediately. Subsequent work shall proceed only upon an archaeological clearance from the State Historic Preservation Division, Department of Lands and Natural Resources.
- 5. Pursuant to Hawaii Administrative Rules (HAR) § 11-200.1-18(d), the EA needs to consider alternatives and assess their potential impacts. The OPSD recommends that the site-specific Best Management Practices shall be developed and implemented to prevent any runoff, sediment, soil and debris potentially resulting from associated construction activities from adversely impacting the coastal ecosystems and the State waters as specified in HAR Chapter 11-54.
- 6. The OPSD is the lead state agency with the authority to conduct CZM Act federal consistency reviews. If the proposed LCC closure project receives federal grants and assistance or requires federal permit or license approval the applicant should consult with the OPSD on the applicability of CZM Act federal consistency review.

If you respond to this comment letter, please include DTS202404030917NA in the subject line. For any questions regarding this letter, please contact Shichao Li of our office at (808) 587-2841 or by email at shichao.li@hawaii.gov.

Sincerely,

Mary Alice Evans

Mary Alice Evans Director



Ms. Mary Alice Evans Office of Planning and Sustainable Development State of Hawaii P.O. Box 2359 Honolulu, HI 96804

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nā'ālehu, Hawai'i Island, Hawai'i DTS202404030917NA

Dear Ms. Evans:

Thank you for your letter dated April 24, 2024 regarding the Early Consultation Environmental Assessment for the Na'alehu Large Capacity Cesspool Closure Project on the Island of Hawai'i. We appreciate your comments and concerns regarding potential impacts to sensitive wildlife species and habitats.

We acknowledge your comments and have carefully considered them in the preparation of the Draft Environmental Assessment (EA). A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning

From:	Cole, Colleen
To:	Public Comment
Cc:	Asman, Lindsy; PIFWO Admin, FW1
Subject:	Comments on EID/EA for Nā'ālehu Large Capacity Cesspool Closure, Ka'ū, Hawai'i
Date:	Friday, April 5, 2024 2:43:15 PM
Attachments:	IPaC Info Letter Species List Instructions PIFWO 20Apr2022 Final.pdf
	Plant Avoidance and Minimization Measures FINAL May 2023.docx
	Naalehu LCC Animal Avoidance and Minimization Measures - FINAL May 2023.docx

Aloha Keola Cheng,

The Pacific Islands Fish and Wildlife Office received your request for comments on the proposed Nā'ālehu Large Capacity Cesspool Closure project located in the Ka'ū District on Hawai'i Island on April 2, 2024. We reviewed the proposed project summarized in the Environmental Information Document you provided. At this time, we recommend that the project planning for any of the four alternative actions include avoidance and minimization measures (AMMs) for endangered species that may be affected by project activities.

You can obtain an official species list in the <u>Information for Planning and</u> <u>Consultation (IPaC)</u> online tool. Please see the attached pdf with detailed directions on how you obtain an official species list in IPAC.



Once you have entered basic project information, including a map of the project, IPaC will generate a species list comprised of all federally listed species that may occur in the project area. Each species includes a link in which you will find avoidance and minimization measures (AMMs) for that species.

Attached to this email are AMMs that you will likely encounter when you obtain an official species list for the project.

Please feel free to contact me if you need additional assistance.

Mahalo, **Colleen Cole Biologist - Maui Nui & Hawai'i Island Team** Pacific Islands Fish and Wildlife Office U.S. Fish and Wildlife Service 154 Waiānuenue Avenue Suite 103 PO Box 10225 Hilo, Hawaiʻi 96720-2452

Cell Phone: 808-859-1002 Email: colleen_cole@fws.gov

FINAL Avoidance and Minimization Measures (AMMs) Final revised May 2023

ESA Listed Species

Endangered 'ōpe'ape'a (Hawaiian hoary bat, *Lasiurus cinereus semotus*): The Hawaiian hoary bat roosts in woody vegetation across all islands and will leave their young unattended in trees and shrubs when they forage. If trees or shrubs 15 feet or taller are cleared during the pupping season, June 1 through September 15, there is a risk that young bats could inadvertently be harmed or killed, since they are too young to fly or move away from disturbance. Hawaiian hoary bats forage for insects from as low as 3 feet to higher than 500 feet above the ground and can become entangled in barbed wire used for fencing.

To avoid and minimize impacts to the endangered Hawaiian hoary bat we recommend you incorporate the following applicable measures into your project description:

- Do not disturb, remove, or trim woody plants greater than 15 feet tall during the bat birthing and pup rearing season (June 1 through September 15).
- Do not use barbed wire for fencing.

Endangered 'ua'u (Hawaiian petrel, *Pterodroma sandwichensis*), Threatened 'a'o, (Newell's shearwater, *Puffinus newelli*), and Endangered Hawai'i Distinct Population Segment of the 'akē'akē (band-rumped storm-petrel, *Hydrobates castro*):

Hawaiian seabirds may traverse the project area at night during the breeding, nesting and fledging seasons (March 1 to December 15). Outdoor lighting could result in seabird disorientation, fallout, and injury or mortality. Seabirds are attracted to lights and after circling the lights they may become exhausted and collide with nearby wires, buildings, or other structures or they may land on the ground. Downed seabirds are subject to increased mortality due to collision with automobiles, starvation, and predation by dogs, cats, and other predators. Young birds (fledglings) traversing the project area between September 15 and December 15, in their first flights from their mountain nests to the sea, are particularly vulnerable to light attraction.

To avoid and minimize potential project impacts to seabirds we recommend you incorporate the following measures into your project description:

- Fully shield all outdoor lights so the bulb can only be seen from below.
- Install automatic motion sensor switches and controls on all outdoor lights or turn off lights when human activity is not occurring in the lighted area.
- Avoid nighttime construction during the seabird fledging period, September 15 through December 15.

Threatened nēnē (Hawaiian goose, Branta (Nesochen) sandvicensis): Nēnē are found on the islands of Hawai'i, Maui, Moloka'i, and Kaua'i. They are observed in a variety of habitats, but prefer open areas, such as pastures, golf courses, wetlands, natural grasslands and shrublands, and lava flows. Threats to the species include introduced mammalian and avian predators, wind facilities, and vehicle strikes.

To avoid and minimize potential project impacts to nēnē we recommend you incorporate the following measures into your project description:

- Do not approach, feed, or disturb nēnē.
- If nēnē are observed loafing or foraging within the project area during the breeding season (September through April), have a biologist familiar with nēnē nesting behavior survey for nests in and around the project area prior to the resumption of any work. Repeat surveys after any subsequent delay of work of 3 or more days (during which the birds may attempt to nest).
- Cease all work immediately and contact the Service for further guidance if a nest is discovered within a radius of 150 feet of proposed project, or a previously undiscovered nest is found within the 150-foot radius after work begins.
- In areas where nēnē are known to be present, post and implement reduced speed limits, and inform project personnel and contractors about the presence of endangered species on-site.

A 4(d) rule was established at the time the nēnē was downlisted to threatened status. Under the 4(d) rule, the following actions are not prohibited under the Act, provided the additional measures described in the downlisting rule are adhered to:

- Take by landowners, or their agents, conducting intentional harassment in the form of hazing or other deterrent measures not likely to cause direct injury or mortality, or nēnē surveys.
- Take that is incidental to conducting lawful control of introduced predators or habitat management activities for nēnē.
- Take by authorized law enforcement officers for the purpose of aiding or euthanizing sick, injured, or orphaned nēnē; disposing of dead specimens; and salvaging a dead specimen that may be used for scientific study.

Endangered Hawaiian waterbirds (ae'o, Hawaiian stilt, *Himantopus mexicanus knudseni*; 'alae ke'oke'o, Hawaiian coot, *Fulica alai*, koloa maoli, Hawaiian duck, *Anas wyvilliana*): Hawaiian waterbirds are currently found in a variety of wetland habitats including freshwater marshes and ponds, coastal estuaries and ponds, artificial reservoirs, kalo or taro (*Colocasia esculenta*) lo'i or patches, irrigation ditches, sewage treatment ponds, and in the case of the Hawaiian duck, montane streams and marshlands. Hawaiian stilts may also be found wherever ephemeral or persistent standing water may occur. Threats to these species include non-native predators, habitat loss, and habitat degradation. Hawaiian ducks are also subject to threats from hybridization with introduced mallards.

Based on the project details provided, your project may result in the creation of standing water or open water that could attract Hawaiian waterbirds to the project site. In particular, the Hawaiian stilt is known to nest in sub-optimal locations (e.g., any ponding water), if water is present. Hawaiian waterbirds attracted to sub-optimal habitat may suffer adverse impacts, such as predation and reduced reproductive success, and thus the project may create an attractive nuisance. Therefore, we recommend you work with our office during project planning so that we may assist you in developing measures to avoid impacts to listed species (e.g., fencing, vegetation control, predator management).

To avoid and minimize potential project impacts to Hawaiian waterbirds we recommend you incorporate the following measures into your project description:

- In areas where waterbirds are known to be present, post and implement reduced speed limits, and inform project personnel and contractors about the presence of endangered species on-site.
- If water resources are located within or adjacent to the project site, incorporate applicable best management practices regarding work in aquatic environments into the project design (see enclosure).
- Have a biological monitor that is familiar with the species' biology conduct Hawaiian waterbird nest surveys where appropriate habitat occurs within the vicinity of the proposed project site prior to project initiation. Repeat surveys again within 3 days of project initiation and after any subsequent delay of work of 3 or more days (during which the birds may attempt to nest). If a nest or active brood is found:
 - Contact the Service within 48 hours for further guidance.
 - Establish and maintain a 100-foot buffer around all active nests and/or broods until the chicks/ducklings have fledged. Do not conduct potentially disruptive activities or habitat alteration within this buffer.
 - Have a biological monitor that is familiar with the species' biology present on the project site during all construction or earth moving activities until the chicks/ducklings fledge to ensure that Hawaiian waterbirds and nests are not adversely impacted.

Endangered Blackburn's sphinx moth (Manduca blackburni):

The adult Blackburn's sphinx moth feeds on nectar from native plants, including beach morning glory (*Ipomoea pes-caprae*), 'ilie'e (*Plumbago zeylanica*), maiapilo (*Capparis sandwichiana*), and others. Blackburn's sphinx moth larvae feed on non-native tree tobacco (*Nicotiana glauca*) and native 'aiea (*Nothocestrum* spp.). To pupate, the larvae burrow into the soil and can remain in a state of torpor for a year or more before emerging from the soil. Soil disturbance can result in death of the pupae.

We offer the following survey recommendations to assess whether the Blackburn's sphinx moth occurs within the project area:

- A biologist familiar with the species should survey areas of proposed activities for Blackburn's sphinx moth and its larval host plants prior to work initiation.
 - Surveys should be conducted during the wettest portion of the year (usually November-April or several weeks after a significant rain) and within 4-6 weeks prior to construction.
 - Surveys should include searches for adults, eggs, larvae, and signs of larval feeding (chewed stems, frass, or leaf damage).
 - If moths, eggs, larvae, or native 'aiea or tree tobacco over 3 feet tall, are found during the survey, please contact the Service for additional guidance to avoid impacts to this species.

If no Blackburn's sphinx moth, 'aiea, or tree tobacco are found during surveys, it is imperative that measures be taken to avoid attraction of Blackburn's sphinx moth to the project location and

prohibit tree tobacco from entering the site. Tree tobacco can grow greater than 3 feet tall in approximately 6 weeks. If it grows over 3 feet, the plants may become a host plant for Blackburn's sphinx moth. We therefore recommend that you:

- Remove any tree tobacco less than 3 feet tall.
- Monitor the site every 4-6 weeks for new tree tobacco growth before, during, and after the proposed ground-disturbing activity.
 - Monitoring for tree tobacco can be completed by any staff, such as groundskeeper or regular maintenance crew, provided with picture placards of tree tobacco at different life stages.

Endangered Hawksbill sea turtle (*Eretmochelys imbricata*): The Service consults on sea turtles and their use of terrestrial habitats (beaches where nesting and/or basking is known to occur), whereas the National Oceanic and Atmospheric Administration (NOAA) Fisheries consults on sea turtles in aquatic habitats. We recommend that you consult with NOAA Fisheries regarding the potential impacts from the proposed project if it may affect off-shore or open ocean habitats.

Green sea turtles may nest on any sandy beach area in the Pacific Islands. Hawksbill sea turtles exhibit a wide tolerance for nesting substrate (ranging from sandy beach to crushed coral) with nests typically placed under vegetation. Both species exhibit strong nesting site fidelity. Nesting occurs on beaches from May through September, peaking in June and July, with hatchlings emerging through November and December.

To avoid and minimize project impacts to sea turtles and their nests we recommend you incorporate the following measures into your project description:

- No vehicle use on or modification of the beach/dune environment during the sea turtle nesting or hatching season (May to December).
- Do not remove native dune vegetation.
- Incorporate applicable best management practices regarding Work in Aquatic Environments (see enclosed) into the project design. Have a biologist familiar with sea turtles conduct a visual survey of the project site to ensure no basking sea turtles are present.
 - If a basking sea turtle is found within the project area, cease all mechanical or construction activities within 100 feet until the animal voluntarily leaves the area.
 Cease all activities between the basking turtle and the ocean.
- Remove any project-related debris, trash, or equipment from the beach or dune if not
- actively being used.
- Do not stockpile project-related materials in the intertidal zone, reef flats, sandy beach and adjacent vegetated areas, or stream channels.

Optimal sea turtle nesting habitat is a dark beach free of barriers that restrict sea turtle movement. Nesting turtles may be deterred from approaching or laying successful nests on lighted or disturbed beaches. They may become disoriented by artificial lighting, leading to exhaustion and placement of a nest in an inappropriate location (such as at or below the high tide line). Hatchlings that emerge from nests may also be disoriented by artificial lighting. Inland areas visible from the beach should be sufficiently dark to allow for successful navigation by hatchlings to the ocean.

To avoid and minimize project impacts to sea turtles from lighting we recommend incorporating the following applicable measures into your project description:

- Avoid nighttime work during the nesting and hatching season (May to December).
- Minimize the use of lighting on or near beaches and shield all project-related lights so the light is not visible from any beach.
 - If lights can't be fully shielded or if headlights must be used, fully enclose the light source with light filtering tape or filters.
- Incorporate design measures into the construction or operation of buildings adjacent to the beach to reduce ambient outdoor lighting such as:
 - tinting or using automatic window shades for exterior windows that face the beach;
 - reducing the height of exterior lighting to below 3 feet and pointed downward or away from the beach; and
 - minimize light intensity to the lowest level feasible and, when possible, include timers and motion sensors.

Avoidance, Minimization, and Conservation Measures for listed plants in the Pacific Islands

can result in higher incidence or intensity of fire. Activities such as grazing, use of construction equipment and vehicles, and increased Project activities may affect listed plant species by causing physical damage to plant parts (roots, stems, flowers, fruits, seeds, etc.) as Soil disturbance or removal has the potential to negatively impact the soil seed bank of listed plant species if such species are present moisture, temperature), damaging or destroying the listed plants and also increasing the risk of invasion by nonnative plants, which absorption of water and nutrients and damage plant root systems and may result in reduced growth and/or mortality of listed plants. reproduction. Cutting and removal of vegetation surrounding listed plants has the potential to alter microsite conditions (e.g., light, human traffic (i.e., trails, visitation, monitoring), can cause ground disturbance, erosion, and/or soil compaction, which decrease well as impacts to other life requisite features of their habitat, which may result in reduction of germination, growth and/or or historically occurred in the project area.

modified sites is proposed, conduct a botanical survey for listed plant species within the project action area, defined as the area where direct and indirect effects are likely to occur. Surveys should be conducted by a knowledgeable botanist with documented experience minimizing disturbance outside of existing developed or otherwise modified areas. When disturbance outside existing developed or conducted during the wettest part of the year (typically October to April) when plants and identifying features are more likely to be In order to avoid or minimize potential adverse effects to listed plants that may occur on the proposed project site, we recommend in identifying native Hawaiian and Pacific Islands plants, including listed plant species. Botanical surveys should optimally be visible, especially in drier areas. If surveys are conducted outside of the wet season, the Service may assume plant presence.

The boundary of the area occupied by listed plants should be marked with flagging by the surveyor. To avoid or minimize potential adverse effects to listed plants, we recommend adherence to buffer distances for the activities in the Table below. Where disturbed Whenever possible we recommend using native plants for landscaping purposes. The following websites are good resources to use areas do not need to be maintained as an open area, restore disturbed areas using native plants as appropriate for the location. (https://hawaiiscape.com/Publications), Native Hawaiian Plants for Landscaping, Conservation, and Reforestation (<u>https://www.ctahr.hawaii.edu/oc/freepubs/pdf/of-30.pdf</u>), and Best Native Plants for Landscapes when choosing landscaping plants: Landscape Industry Council of Hawai'i Native Plant Poster (https://www.ctahr.hawaii.edu/oc/freepubs/pdf/OF-40.pdf)

If listed plants occur in a project area, the avoidance buffers are from project activities. However, where project activities will o is required. The impacts to the plants of concern within the buff at the boundary of the disturbance, as far from the affected plan	recommended to reduce direct and indirect impacts to listed plants ccur within the recommended buffer distances, additional consultation fer area may be reduced by placing temporary fencing or other barriers its as practicable.
The above guidelines apply to areas outside of designated critic unit boundaries, additional consultation is required.	al habitat. If project activities occur within designated critical habitat
All activities, including site surveys, risk introducing nonnative ensure that all equipment, personnel, and supplies are properly	s species into project areas. Specific attention needs to be made to checked and are free of contamination (weed seeds, organic matter, or
other contaminants) before entering project areas. Quarantines a species proximal to project areas need to be considered or adequated or adequated or species proximal to project areas areas areas areas a species proximal to project areas areas areas areas a species proximal to project areas areas areas areas a species provide a species areas a species a species a species a species areas a species areas a species a species areas a species a species a species areas a species	and or management activities occurring on specific priority invasive uately addressed. This information can be acquired by contacting local
experts such as those on local invasive species committees (Kan Nui: <u>https://mauiinvasive.org/</u> ; and Hawai'i: <u>https://www.biisc.</u>	ua'i: <u>https://www.kauaiisc.org/;</u> O'ahu: <u>https://www.oahuisc.org/</u> ; Maui <u>org/</u>
Table 1. Recommended buffer distances to minimize and avoid	potential adverse impacts to listed plants from activities listed below.
	Buffer Distance (feet (meters)) – Keep Project Activity This

	5	
-	H	
_	g	
Ę	_	
	ŏ	
	S	
÷	Ξ	
	Ś	
•	E	
•	Ξ	
•	1	
1	\mathbf{c}	
	ದ	
	Ξ	
	5	
د	ㅂ	
	\sim	
	g	
	ਫ਼	
	Ω	-
-	0	Ì
	ŏ	
	S	
÷	Ξ	
	0	
1	 ~	
	ž	
	g	
	õ	-
	Ξ	
•	Ξ	
	ŝ	
	ñ	
	5	
-	þ	
	9	
-	g	
•		
	ž	
	Ĕ	
	2	
	~	۰.
-	_	
:	g	
	701d	
	avoid	
	d avoid	
	nd avoid	
	and avoid	
-	te and avoid	
· · ·	ize and avoid	
-	mize and avoid	
- -	nimize and avoid	
- -	unimize and avoid	
- -	minimize and avoid	
- -	o minimize and avoid	
· · · · · ·	to minimize and avoid	
- -	es to minimize and avoid	
- -	a ces to minimize and avoid	
	ances to minimize and avoid	
· · · · ·	stances to minimize and avoid	
· · · · · · ·	distances to minimize and avoid	
· · · · · ·	· distances to minimize and avoid	
· · · · · · · · · · · · · · · · · · ·	er distances to minimize and avoid	
·· · · · · · · · · · · · · · · · · · ·	ther distances to minimize and avoid	
	witter distances to minimize and avoid	
	buffer distances to minimize and avoid	
	ed butter distances to minimize and avoid	
	ded butter distances to minimize and avoid	
	inded butter distances to minimize and avoid	
	hended buffer distances to minimize and avoid	
	mended buffer distances to minimize and avoid	
	mmended buffer distances to minimize and avoid	
	commended buffer distances to minimize and avoid	
	ecommended buffer distances to minimize and avoid	
	Recommended buffer distances to minimize and avoid	
	. Recommended buffer distances to minimize and avoid	
	1. Recommended butter distances to minimize and avoid	
	le 1. Recommended buffer distances to minimize and avoid	
	ble 1. Recommended butter distances to minimize and avoid	
	able 1. Recommended butter distances to minimize and avoid	

Action	Buffer Distance (feet (meters) Far Away fron) – Keep Project Activity This m Listed Plant
	Grasses/Herbs/Shrubs and Terrestrial Orchids	Trees and Arboreal Orchids
Walking, hiking, surveys	3 ft (1 m)	3 ft (1 m)
Cutting and Removing Vegetation By Hand or Hand Tools (e.g., weeding)	3 ft (1 m)	3 ft (1 m)
Mechanical Removal of Individual Plants or Woody Vegetation (e.g., chainsaw, weed eater)	3 ft up to height of removed vegetation (whichever greater)	3 ft up to height of removed vegetation (whichever greater)
Removal of Vegetation with Heavy Equipment (e.g., bulldozer, tractor, "bush hog")	2x width equipment + height of vegetation	820 ft (250 m)

	Action	Buffer Distance (feet (meters) Far Away froi) – Keep Project Activity This n Listed Plant
		Grasses/Herbs/Shrubs and Terrestrial Orchids	Trees and Arboreal Orchids
	Ground-based Spray Application; hand application (no wand applicator; spot treatment)	10 ft (3 m)	Crown diameter
	Ground-based Spray Application; manual pump with wand, backpack	50 ft (15 m)	Crown diameter
Use of Approved Herhicides	Ground-based Spray Application; vehicle-mounted tank sprayer	50 ft (15 m)	Crown diameter
(following label)	Aerial Spray (ball applicator)	250 ft (76 m)	250 ft (76 m)
1	Aerial Application – herbicide ballistic technology (individual plant treatment)	100 ft (30 m)	Crown diameter
	Aerial Spray (boom)	Further consultation required	Further consultation required
Use of Insecticides (p	ollinators, seed dispersers)	Further consultation required	Further consultation required
Ground/Soil Disturb. e.g., shovel, 'õ'õ; Sm:	ance/Outplanting/Fencing (Hand tools, all mechanized tools, e.g., auger)	20 ft (6 m)	2x crown diameter
Ground/Soil Disturb	ance (Heavy Equipment)	328 ft (100 m)	820 ft (250 m)
Surface Hardening/	Soil Trails (e.g., human, ungulates)	20 ft (6 m)	2x crown diameter
compaction	Roads/Utility Corridors, Buildings/Structures	328 ft (100 m)	820 ft (250 m)

Action	Buffer Distance (feet (meters) Far Away froi) – Keep Project Activity This m Listed Plant
	Grasses/Herbs/Shrubs and Terrestrial Orchids	Trees and Arboreal Orchids
Prescribed Burns	Further consultation required	Further consultation required
Farming/Ranching/Silviculture	820 ft (250 m)	820 ft (250 m)

Definitions (Wagner et al. 1999)

Crown: The leafy top of a tree.

Herb: A plant, either annual, biennial, or perennial, with the non-woody stems dying back to the ground at the end of the growing season.

Shrub: A perennial woody plant with usually several to numerous primary stems arising from or relatively near the ground. Tree: A woody perennial that usually has a single trunk

References Cited

USFWS. 2010. Endangered and threatened wildlife and plants; determination of endangered status for 48 species on Kauai and designation of critical habitat. Federal Register 75: 18960–19165.

. 2012. Endangered and threatened wildlife and plants; endangered status for 23 species on Oahu and designation of critical habitat for 124 species; final rule. Federal Register 77: 57648–57862.

. 2013a Endangered and threatened wildlife and plants; determination of endangered status for 38 species from Molokai, Lanai, and Maui. Federal Register 78: 32014–32065.

. 2013b. Endangered and threatened wildlife and plants; determination of endangered species status for 15 species on Hawaii Island. Federal Register 78: 64638–64690.

. 2016. Endangered and threatened wildlife and plants; determination of endangered status for 49 species from the Hawaiian Islands. Federal Register 81: 67786–67860.

. 2016. USFWS Rare plant database. Unpublished.

Wagner, W.L., Sohmer, S., and D.R. Herbst. 1999. Manual of the flowering plants of Hawaii, revised edition. Honolulu, Hawaii. University of Hawaii and Bishop Museum Press. 1,919 pp.



Ms. Coleen Cole U.S. Fish and Wildlife Service Pacific Islands Fish and Wildlife Office 150 Waiānuenue Avenue Suite 103 Hilo, Hawaiʻi 96720

Subject: Environmental Assessment Early Consultation for the Na'alehu Large Capacity Cesspools Closure Nāʿālehu, Hawaiʿi Island, Hawaiʿi

Dear Ms Cole:

Thank you for your e-mail dated April 8, 2024 regarding the Early Consultation Environmental Assessment for the Na'alehu Large Capacity Cesspool Closure Project on the Island of Hawai'i. We acknowledge your comments and they have been considered in the preparation of the Draft EA with regard to meeting content requirements prescribed in Hawai'i Administrative Rules, Title 11, Chapter 200.1, Section 18. A record of your comments, along with this response, have been produced and are appended to the Draft EA in Appendix G.

We acknowledge your comments and note that a Natural Resource Assessment has been completed for the project area. The results of the assessment and best management practices are described in Section 4.6 of the Draft EA.

Please note that the Draft EA has been published and made available for review, and comment in the current issue of the State of Hawai'i's Environmental Review Program's (ERP) The Environmental Notice.

We appreciate your participation in the EA review process.

Sincerely,

Keola Cheng

Keola Cheng Director - Planning

Draft Environmental Information Document & Environmental Assessment Nā'ālehu Large Cesspool Closure

